IDEX CORP /DE/
Form 10-K
February 19, 2016

UNITED STATES SECURITIES AND EXCHANGE COM	<b>MISSION</b>
Washington, D.C. 20549	
Form 10-K	2 OD 15(4) OF THE SECUDITIES EXCUANCE ACT
b OF 1934	3 OR 15(d) OF THE SECURITIES EXCHANGE ACT
For the Fiscal Year Ended December 31, 2015	
	ON 13 OR 15(d) OF THE SECURITIES EXCHANGE
For the Transition Period From to	
Commission file number 1-10235	
IDEX CORPORATION	
(Exact Name of Registrant as Specified in its Charter)	
Delaware	36-3555336
(State or other jurisdiction of incorporation or organization)	(I.R.S. Employer Identification No.)
1925 West Field Court, Lake Forest, Illinois	60045
(Address of principal executive offices)	(Zip Code)
Registrant's telephone number: (847) 498-7070	
Securities Registered Pursuant to Section 12(b) of the Act:	
Title of Each Class	Name of Each Exchange on Which Registered
Common Stock, par value \$.01 per share	New York Stock Exchange and Chicago Stock Exchange
Securities Registered Pursuant to Section 12(g) of the Act: None	
Indicate by check mark if the registrant is a well-known sea Act. Yes b No "	soned issuer, as defined in Rule 405 of the Securities
Indicate by check mark if the registrant is not required to fil Act. Yes "No b	e reports pursuant to Section 13 or Section 15(d) of the
Indicate by check mark whether the registrant: (1) has filed the Securities Exchange Act of 1934 during the preceding 1 required to file such reports), and (2) has been subject to suc Indicate by check mark whether the registrant has submitted any, every Interactive Data File required to be submitted an the preceding 12 months (or for such shorter period that the files). Yes b No "	2 months (or for such shorter period that the registrant was ch filing requirements for the past 90 days. Yes þ No " d electronically and posted on its corporate Web site, if d posted pursuant to Rule 405 of Regulation S-T during
Indicate by check mark if disclosure of delinquent filers pur herein, and will not be contained, to the best of registrant's incorporated by reference in Part III of this Form 10-K or an Indicate by check mark whether the registrant is a large accor or a smaller reporting company. See the definitions of "larg company" in Rule 12b-2 of the Exchange Act. (Check one):	knowledge, in definitive proxy or information statements ny amendment to this Form 10-K. belerated filer, an accelerated filer, a non-accelerated filer, e accelerated filer," "accelerated filer" and "smaller reporting con-accelerated filer " Smaller reporting company "

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes "No b

The aggregate market value, as of the last business day of the registrant's most recently completed second fiscal quarter, of the common stock (based on the June 30, 2015 closing price of \$78.58) held by non-affiliates of IDEX Corporation was \$6,085,231,271.

The number of shares outstanding of IDEX Corporation's common stock, par value \$.01 per share, as of February 16, 2016 was 75,929,397.

#### DOCUMENTS INCORPORATED BY REFERENCE

Portions of the proxy statement with respect to the IDEX Corporation 2016 annual meeting of stockholders (the "2016 Proxy Statement") are incorporated by reference into Part III of this Form 10-K.

PART I.		
Item 1.	Business	<u>1</u>
Item 1A.	Risk Factors	<u>8</u>
Item 1B.	Unresolved Staff Comments	<u>10</u>
Item 2.	Properties	<u>10</u>
Item 3.	Legal Proceedings	<u>11</u>
Item 4.	Mine Safety Disclosures	<u>11</u>
PART II		
т. с	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of	10
Item 5.	Equity Securities	<u>12</u>
Item 6.	Selected Financial Data	<u>14</u>
Item 7.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>15</u>
Item 7A.	Quantitative and Qualitative Disclosures About Market Risk	<u>23</u>
Item 8.	Financial Statements and Supplementary Data	<u>25</u>
Item 9.	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	<u>67</u>
Item 9A.	Controls and Procedures	<u>67</u>
Item 9B.	Other Information	<u>67</u>
PART II	I.	
Item 10.	Directors, Executive Officers and Corporate Governance	<u>68</u>
Item 11.	Executive Compensation	<u>68</u>
Itam 10	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder	60
Item 12.	Matters	<u>68</u>
Item 13.	Certain Relationships and Related Transactions, and Director Independence	<u>68</u>
Item 14.	Principal Accountant Fees and Services	<u>68</u>

Item 14. Principal Accountant Fees and Services

PART IV.

Item 15. Exhibits and Financial Statement Schedules <u>Signatures</u>

Exhibit Index

<u>69</u> <u>70</u> <u>71</u>

#### PART I

Cautionary Statement Under the Private Securities Litigation Reform Act

This report contains "forward-looking" statements within the meaning of the Private Securities Litigation Reform Act of 1995, as amended. These statements may relate to, among other things, capital expenditures, acquisitions, cost reductions, cash flow, revenues, earnings, market conditions, global economies and operating improvements, and are indicated by words or phrases such as "anticipate," "estimate," "plans," "expects," "projects," "forecasts," "should," "could," " "management believes," "the company believes," "the company intends," and similar words or phrases. These statements are subject to inherent uncertainties and risks that could cause actual results to differ materially from those anticipated at the date of this report. The risks and uncertainties include, but are not limited to, the following: economic and political consequences resulting from terrorist attacks and wars; levels of industrial activity and economic conditions in the U.S. and other countries around the world; pricing pressures and other competitive factors, and levels of capital spending in certain industries - all of which could have a material impact on order rates and IDEX's results, particularly in light of the low levels of order backlogs it typically maintains; its ability to make acquisitions and to integrate and operate acquired businesses on a profitable basis; the relationship of the U.S. dollar to other currencies and its impact on pricing and cost competitiveness; political and economic conditions in foreign countries in which the company operates; interest rates; capacity utilization and the effect this has on costs; labor markets; market conditions and material costs; and developments with respect to contingencies, such as litigation and environmental matters. The forward-looking statements included here are only made as of the date of this report, and management undertakes no obligation to publicly update them to reflect subsequent events or circumstances, except as may be required by law. Investors are cautioned not to rely unduly on forward-looking statements when evaluating the information presented here.

#### Item 1. Business.

IDEX Corporation ("IDEX," the "Company," "us," "our," or "we") is a Delaware corporation incorporated on September 24, 1987. The Company is an applied solutions business that sells an extensive array of pumps, flow meters and other fluidics systems and components and engineered products to customers in a variety of markets around the world. All of the Company's business activities are carried out through wholly-owned subsidiaries.

The Company has three reportable business segments: Fluid & Metering Technologies ("FMT"), Health & Science Technologies ("HST") and Fire & Safety/Diversified Products ("FSDP"). Within our three reportable segments, the Company maintains fifteen platforms, where we focus on organic growth and strategic acquisitions. Each of our fifteen platforms is also a reporting unit, where we annually test for goodwill impairment.

During the third quarter of 2015, the Company announced the appointment of Eric Ashleman as Chief Operating Officer. While there were no changes to the reportable segments or movement of businesses between the reportable segments, the Company no longer delineates between "platforms" and "groups" and made the following changes to how certain businesses are managed internally:

Created the Valves platform as a result of the Alfa Valvole acquisition in June 2015.

Eliminated the Diaphragm and Dosing Pump Technology ("DDPT") platform.

Created the Industrial platform from the businesses previously reported within Chemical, Food & Process (Richter, Viking, and Aegis) plus the Warren Rupp and Trebor businesses from DDPT.

Created the Water platform from the businesses previously reported within Water Services & Technology (ADS, IETG, and iPEK) plus the Pulsafeeder and Knight businesses from DDPT.

The Fluid & Metering Technologies segment contains the Energy (comprised of Corken, Faure Herman, Liquid Controls, SAMPI and Toptech), Valves (comprised of Alfa Valvole), Water (comprised of Pulsafeeder, Knight, ADS, IETG, and iPEK), Industrial (comprised of Richter, Viking, Aegis, Warren Rupp, and Trebor), and Agriculture (comprised of Banjo) platforms. The Health & Science Technologies segment contains the Scientific Fluidics (comprised of Eastern Plastics, Rheodyne, Sapphire Engineering, Upchurch Scientific, ERC, and CiDRA Precision Services), IDEX Optics & Photonics (comprised of CVI Melles Griot, Semrock, and AT Films), Sealing Solutions

(comprised of Precision Polymer Engineering, FTL Seals Technology, and Novotema), Gast, Micropump, and Material Processing Technologies (comprised of Quadro, Fitzpatrick, Microfluidics, and Matcon) platforms. The Fire & Safety/Diversified Products segment is comprised of the Fire Suppression (comprised of Class 1, Hale and Godiva), Rescue (comprised of Dinglee, Hurst Jaws of Life, Lukas, and Vetter), Band-It, and Dispensing platforms. IDEX believes that each of its reporting units is a leader in its product and service areas. The Company also believes that its strong financial performance has been attributable to its ability to design and engineer specialized quality products, coupled with its ability to identify and successfully consummate and integrate strategic acquisitions.

#### FLUID & METERING TECHNOLOGIES SEGMENT

The Fluid & Metering Technologies segment designs, produces and distributes positive displacement pumps, flow meters, injectors, and other fluid-handling pump modules and systems and provides flow monitoring and other services for the food, chemical, general industrial, water & wastewater, agriculture and energy industries. Fluid & Metering Technologies application-specific pump and metering solutions serve a diverse range of end markets, including industrial infrastructure (fossil fuels, refined & alternative fuels, and water & wastewater), chemical processing, agriculture, food & beverage, pulp and paper, transportation, plastics and resins, electronics and electrical, construction & mining, pharmaceutical and bio-pharmaceutical, machinery and numerous other specialty niche markets. Fluid & Metering Technologies accounted for 43%, 42% and 43% of IDEX's sales in 2015, 2014 and 2013, respectively, with approximately 44% of its 2015 sales to customers outside the U.S. The segment accounted for 43%, 43% and 47% of IDEX's operating income in 2015, 2014 and 2013, respectively.

Energy. Energy consists of the Company's Corken, Faure Herman, Liquid Controls, SAMPI and Toptech businesses. Energy is a leading supplier of flow meters, electronic registration and control products, rotary vane and turbine pumps, reciprocating piston compressors, and terminal automation control systems. Headquartered in Lake Bluff, Illinois (Liquid Controls products), Energy has additional facilities in Longwood, Florida and Zwijndrecht, Belgium (Toptech products); Oklahoma City, Oklahoma (Corken products); La Ferté Bernard, France (Faure Herman products); and Altopascio, Italy (SAMPI products). Applications for Liquid Controls and SAMPI positive displacement flow meters, electronic, registration and control products include mobile and stationary metering installations for wholesale and retail distribution of petroleum and liquefied petroleum gas, aviation refueling, and industrial metering and dispensing of liquids and gases. Corken products consist of positive-displacement rotary vane pumps, single and multistage regenerative turbine pumps, and small horsepower reciprocating piston compressors. Toptech supplies terminal automation hardware and software to control and manage inventories, as well as transactional data and invoicing, to customers in the oil, gas and refined-fuels markets. Faure Herman is a leading supplier of ultrasonic and helical turbine flow meters used in the custody transfer and control of high value fluids and gases. Approximately 44% of Energy's 2015 sales were to customers outside the U.S.

Valves. Valves consists of the Company's Alfa Valvole ("Alfa") business. Alfa is a leader in the design, manufacture and sale of specialty valve products for use in the chemical, petro-chemical, energy and sanitary markets. Located in Casorezzo, Italy, Alfa's products are used in various industrial fields for fluid control, in both gas and liquid form, in all sectors of plant engineering, cosmetics, detergents, food industry, electric energy, pharmaceutical, chemical plants, petrochemical plants, oil, heating/air conditioning and in all markets worldwide and also on ships, ferries and marine oil platforms. 100% of Alfa's 2015 sales were to customers outside the U.S.

Water. Water consists of the Company's ADS, IETG, iPEK, Knight and Pulsafeeder businesses. Water is a leading provider of metering technology, flow monitoring products and underground surveillance services for wastewater markets, alloy and non-metallic gear pumps, peristaltic pumps, transfer pumps, as well as dispensing equipment for industrial laundries, commercial dishwashing and chemical metering. ADS's products and services provide comprehensive integrated solutions that enable industry, municipalities and government agencies to analyze and measure the capacity, quality and integrity of wastewater collection systems, including the maintenance and construction of such systems. IETG's products and services enable water companies to effectively manage their water distribution and sewerage networks, while its surveillance service specializes in underground asset detection and mapping for utilities and other private companies. iPEK supplies remote controlled systems used for infrastructure inspection. Knight is a leading manufacturer of pumps and dispensing equipment for industrial laundries, commercial dishwashing and chemical metering. Pulsafeeder products (which also include OBL products) are used to introduce precise amounts of fluids into processes to manage water quality and chemical composition, as well as peristaltic pumps. Its markets include water & wastewater treatment, oil & gas, power generation, pulp & paper, chemical and hydrocarbon processing, and swimming pools. Water maintains operations in Huntsville, Alabama and various other locations in the United States and Australia (ADS products and services); Leeds, England (IETG products and services); Hirschegg, Austria, and Sulzberg, Germany (iPEK products); Rochester, New York, Punta Gorda, Florida and Milan, Italy (Pulsafeeder products); Lake Forest, California, Mississauga, Ontario, Canada, and Lewes, England, (Knight products); and a maquiladora in Ciudad Juarez, Chihuahua, Mexico (Knight products). Approximately 40% of

Water's 2015 sales were to customers outside the U.S.

Industrial. Industrial consists of the Company's Richter, Viking, Aegis, Warren Rupp, and Trebor businesses. Industrial is a producer of fluoroplastic lined corrosion-resistant magnetic drive and mechanical seal pumps, shut-off, control and safety valves for corrosive, hazardous, contaminated, pure and high-purity fluids, as well as rotary internal gear, external gear, vane and rotary lobe pumps, custom-engineered OEM pumps, strainers, gear reducers and engineered pump systems. Richter's products offer superior solutions for demanding and complex pump applications in the process industry. Viking's products consist of external gear pumps, strainers and reducers, and related controls used for transferring and metering thin and viscous liquids sold under the Viking and Wright Flow brands. Viking products primarily serve the chemical, petroleum, pulp & paper, plastics, paints, inks, tanker trucks, compressor, construction, food & beverage, personal care, pharmaceutical and biotech

markets. Aegis is a leader in the design, manufacture and sale of specialty chemical processing valves for use in the chemical, petro-chemical, chlor-alkali, pharmaceutical, semiconductor and pulp & paper industries. Warren Rupp products (which also include Pumper Parts and Versa-Matic products) are used for abrasive and semisolid materials as well as for applications where product degradation is a concern or where electricity is not available or should not be used. Warren Rupp products, which include air-operated diaphragm pumps, primarily serve the chemical, paint, food processing, electronics, construction, utilities, oil & gas, mining and industrial maintenance markets. Trebor is a leader in high-purity fluid handling products, including air-operated diaphragm pumps and deionized water-heating systems. Trebor products are used in manufacturing of semiconductors, disk drives and flat panel displays. Industrial maintains operations in Kempen, Germany and Suzhou, China (Richter products); Cedar Falls, Iowa (Viking, Wright Flow, and Richter products); Eastbourne, England (Wright Flow products); and Shannon, Ireland (Viking and Blagdon products); Geismar, Louisiana (Aegis products); Mansfield, Ohio (Warren Rupp products); Salt Lake City, Utah (Trebor products). Industrial primarily uses independent distributors to market and sell its products. Approximately 51% of Industrial's 2015 sales were to customers outside the U.S.

Agriculture. Agriculture consists of the Company's Banjo business. Banjo is a provider of special purpose, severe-duty pumps, valves, fittings and systems used in liquid handling. Banjo is based in Crawfordsville, Indiana with a facility in Didam, The Netherlands, and its products are used in agriculture and industrial applications. Approximately 15% of Banjo's 2015 sales were to customers outside the U.S.

#### HEALTH & SCIENCE TECHNOLOGIES SEGMENT

The Health & Science Technologies segment designs, produces and distributes a wide range of precision fluidics, rotary lobe pumps, centrifugal and positive displacement pumps, roll compaction and drying systems used in beverage, food processing, pharmaceutical and cosmetics, pneumatic components and sealing solutions, including very high precision, low-flow rate pumping solutions required in analytical instrumentation, clinical diagnostics and drug discovery, high performance molded and extruded, biocompatible medical devices and implantables, air compressors used in medical, dental and industrial applications, optical components and coatings for applications in the fields of scientific research, defense, biotechnology, aerospace, telecommunications and electronics manufacturing, laboratory and commercial equipment used in the production of micro and nano scale materials, precision photonic solutions used in life sciences, research and defense markets, and precision gear and peristaltic pump technologies that meet exacting original equipment manufacturer specifications. Health & Science Technologies accounted for 36%, 35% and 35% of IDEX's sales in 2015, 2014 and 2013, respectively, with approximately 55% of its 2015 sales to customers outside the U.S. The segment accounted for 33%, 31% and 30% of IDEX's operating income in 2015, 2014 and 2013, respectively.

Scientific Fluidics. Scientific Fluidics consists of the Company's Eastern Plastics, Rheodyne, Sapphire Engineering, Upchurch Scientific, ERC, and CiDRA Precision Services ("CPS") businesses. Scientific Fluidics has facilities in Rohnert Park, California (Rheodyne products); Bristol, Connecticut (Eastern Plastics products); Middleboro, Massachusetts (Sapphire Engineering products); Oak Harbor, Washington (Upchurch Scientific products); Kawaguchi, Japan (ERC products); and Wallingford, Connecticut (CPS products). Eastern Plastics products, which consist of high-precision integrated fluidics and associated engineered manifolds, are used in a broad set of end markets including medical diagnostics, analytical instrumentation, and laboratory automation. Rheodyne products consist of injectors, valves, fittings and accessories for the analytical instrumentation market. These products are used by manufacturers of high pressure liquid chromatography ("HPLC") equipment servicing the pharmaceutical, biotech, life science, food & beverage, and chemical markets. Sapphire Engineering and Upchurch Scientific products consist of fluidic components and systems for the analytical, biotech and diagnostic instrumentation markets, such as fittings, precision-dispensing pumps and valves, tubing and integrated tubing assemblies, filter sensors and other micro-fluidic and nano-fluidic components, as well as advanced column hardware and accessories for the high performance liquid chromatography market. The products produced by Sapphire Engineering and Upchurch Scientific primarily serve the pharmaceutical, drug discovery, chemical, biochemical processing, genomics/proteomics research, environmental labs, food/agriculture, medical lab, personal care, and plastics/polymer/rubber production markets. ERC manufactures gas liquid separations and detection solutions for the life science, analytical instrumentation and clinical chemistry markets. ERC's products consist of in-line membrane vacuum degassing solutions, refractive index detectors and

ozone generation systems. CPS products consist of microfluidic components serving the life science, health and industrial market. Approximately 55% of Scientific Fluidics' 2015 sales were to customers outside the U.S. IDEX Optics and Photonics ("IOP"). IOP consists of the Company's CVI Melles Griot ("CVI MG"), Semrock, and AT Films (including Precision Photonics products) businesses. CVI MG is a global leader in the design and manufacture of precision photonic solutions used in the life sciences, research, semiconductor, security and defense markets. CVI MG's innovative products are focused on the generation, control and productive use of light for a variety of key science and industrial applications. Products consist of specialty lasers and light sources, electro-optical components, specialty shutters, opto-mechanical assemblies and components. In addition, CVI MG produces critical components for life science research, electronics manufacturing, military and other industrial applications including lenses, mirrors, filters and polarizers. These

components are utilized in a number of important applications such as spectroscopy, cytometry (cell counting), guidance systems for target designation, remote sensing, menology and optical lithography. CVI MG has manufacturing sites located in Albuquerque, New Mexico; Carlsbad, California; Rochester, New York; Leicester, England; Kyongki-Do, Korea; Tamagawa, Japan; and Didam, The Netherlands. Semrock is a provider of optical filters for biotech and analytical instrumentation in the life sciences markets. Semrock's optical filters are produced using state-of-the-art manufacturing processes which enable it to offer its customers significant improvements in instrument performance and reliability. Semrock is located in Rochester, New York. AT Films specializes in optical components and coatings for applications in the fields of scientific research, defense, aerospace, telecommunications and electronics manufacturing. AT Films' core competence is the design and manufacture of filters, splitters, reflectors and mirrors with the precise physical properties required to support their customers' most challenging and cutting-edge optical applications. The Precision Photonics portion of its business specializes in optical components and coatings for applications of the support their customers' most challenging and cutting-edge optical applications in the fields of scientific research, aerospace, telecommunications and electronics manufacturing. AT Films business specializes in optical components and coatings for applications of its business specializes in optical components and coatings for applications. The Precision Photonics portion of its business specializes in optical components and coatings for applications. The Precision Photonics portion of the business specializes in optical components and coatings for applications in the fields of scientific research, aerospace, telecommunications and electronics manufacturing. AT Films is headquartered in Boulder, Colorado. Approximately 54% of IOP's 2015 sales were to customer

Sealing Solutions. Sealing Solutions consists of the Company's Precision Polymer Engineering ("PPE"), FTL Seals Technology ("FTL"), and Novotema businesses. PPE is a provider of proprietary high performance seals and advanced sealing solutions for a diverse range of global industries and applications, including hazardous duty, analytical instrumentation, semiconductor, process technologies, oil & gas, pharmaceutical, electronics, and food applications. PPE is headquartered in Blackburn, England with an additional manufacturing facility in Brenham, Texas. FTL, located in Leeds, England, specializes in the design and application of high integrity rotary seals, specialty bearings, and other custom products for the mining, power generation, and marine markets. Novotema, located in Villongo, Italy, is a leader in the design, manufacture and sale of specialty sealing solutions for use in the building products, gas control, transportation, industrial and water markets. Approximately 78% of Sealing Solutions' 2015 sales were to customers outside the U.S.

Gast. Gast consists of the Company's Gast and Jun-Air businesses. The Gast business is a leading manufacturer of air-moving products, including air motors, low-range and medium-range vacuum pumps, vacuum generators, regenerative blowers and fractional horsepower compressors. Gast products are used in a variety of long-life applications requiring a quiet, clean source of moderate vacuum or pressure. Gast products primarily serve the medical equipment, environmental equipment, computers and electronics, printing machinery, paint mixing machinery, packaging machinery, graphic arts, and industrial manufacturing markets. The Jun-Air business is a provider of low-decibel, ultra-quiet vacuum compressors suitable for medical, dental and laboratory applications. Based in Benton Harbor, Michigan, Gast also has a logistics and commercial center in Redditch, England. Approximately 26% of Gast's 2015 sales were to customers outside the U.S.

Micropump. Micropump, headquartered in Vancouver, Washington, is a leader in small, precision-engineered, magnetically and electromagnetically driven rotary gear, piston and centrifugal pumps. Micropump products are used in low-flow abrasive and corrosive applications. Micropump products primarily serve the continuous ink-jet printing, medical equipment, chemical processing, pharmaceutical, refining, laboratory, electronics, textiles, peristaltic metering pumps, analytical process controllers and sample preparation systems markets. Approximately 72% of Micropump's 2015 sales were to customers outside the U.S.

Material Processing Technologies ("MPT"). MPT consists of the Company's Quadro, Fitzpatrick, Microfluidics and Matcon businesses. Quadro is a leading provider of particle control solutions for the pharmaceutical and bio-pharmaceutical markets. Based in Waterloo, Canada, Quadro's core capabilities include fine milling, emulsification and special handling of liquid and solid particulates for laboratory, pilot phase and production scale processing. Fitzpatrick is a global leader in the design and manufacture of process technologies for the pharmaceutical, food and personal care markets. Fitzpatrick designs and manufactures customized size reduction, roll compaction and drying systems to support their customers' product development and manufacturing processes. Fitzpatrick is headquartered in Elmhurst, Illinois. Microfluidics is a global leader in the design and manufacture of laboratory and commercial equipment used in the production of micro and nano scale materials for the pharmaceutical and chemical markets. Microfluidics is the exclusive producer of the Microfluidizer family of high shear fluid

processors for uniform particle size reduction, robust cell disruption and nanoparticle creation. Microfluidics is based in Waterloo, Canada and has offices in Newton, Massachusetts. Matcon is a global leader in material processing solutions for high value powders used in the manufacture of pharmaceuticals, food, plastics, and fine chemicals. Matcon's innovative products consist of the original cone valve powder discharge system and filling, mixing and packaging systems, all of which support its customers' automation and process requirements. These products are critical to its customers' need to maintain clean, reliable and repeatable formulations of prepackaged foods and pharmaceuticals while helping them achieve lean and agile manufacturing. Matcon is located in Evesham, England. Approximately 59% of MPT's 2015 sales were to customers outside the U.S.

#### FIRE & SAFETY/DIVERSIFIED PRODUCTS SEGMENT

The Fire & Safety/Diversified Products segment produces firefighting pumps and controls, rescue tools, lifting bags and other components and systems for the fire and rescue industry, engineered stainless steel banding and clamping devices used in a variety of industrial and commercial applications, and precision equipment for dispensing, metering and mixing colorants and paints used in a variety of retail and commercial businesses around the world. The Fire & Safety/Diversified Products segment accounted for 21%, 23% and 22% of IDEX's sales in 2015, 2014 and 2013, respectively, with approximately 52% of its 2015 sales to customers outside the U.S. The segment accounted for 24%, 26% and 23% of IDEX's operating income in 2015, 2014 and 2013, respectively.

Fire Suppression. Fire Suppression consists of the Company's Class 1, Hale and Godiva businesses, which produce truck-mounted and portable fire pumps, stainless steel valves, foam and compressed air foam systems, pump modules and pump kits, electronic controls and information systems, conventional and networked electrical systems, and mechanical components for the fire, rescue and specialty vehicle markets. Fire Suppression's customers are primarily OEMs. Fire Suppression is headquartered in Ocala, Florida (Class 1 and Hale products), with additional facilities located in Warwick, England (Godiva products). Approximately 38% of Fire Suppression's 2015 sales were to customers outside the U.S.

Rescue. Rescue consists of the Company's Dinglee, Hurst Jaws of Life, Lukas and Vetter businesses, which produce hydraulic, battery, gas and electric-operated rescue equipment, hydraulic re-railing equipment, hydraulic tools for industrial applications, recycling cutters, pneumatic lifting and sealing bags for vehicle and aircraft rescue, environmental protection and disaster control, and shoring equipment for vehicular or structural collapse. Rescue's customers are primarily public and private fire and rescue organizations. Rescue has facilities in Shelby, North Carolina (Hurst Jaws of Life products); Tianjin, China (Dinglee products); Erlangen, Germany (Lukas products); and Zulpich, Germany (Vetter products). Approximately 71% of Rescue's 2015 sales were to customers outside the U.S. Band-It. Band-It is a leading producer of high-quality stainless steel banding, buckles and clamping systems. The BAND-IT brand is highly recognized worldwide. Band-It products are used for securing exhaust system heat and sound shields, industrial hose fittings, traffic signs and signals, electrical cable shielding, identification and bundling, and in numerous other industrial and commercial applications. Band-It products primarily serve the automotive, transportation equipment, oil & gas, general industrial maintenance, electronics, electrical, communications, aerospace, utility, municipal and subsea marine markets. Band-It is based in Denver, Colorado, with additional operations in Staveley, England. Approximately 36% of Band-It's 2015 sales were to customers outside the U.S. Dispensing. Dispensing produces precision equipment for dispensing, metering and mixing colorants and paints used in a variety of retail and commercial businesses around the world. Dispensing is a global supplier of precision-designed tinting, mixing, dispensing and measuring equipment for auto refinishing and architectural paints. Dispensing products are used in retail and commercial stores, hardware stores, home centers, department stores, automotive body shops as well as point-of-purchase dispensers. Dispensing is headquartered in Sassenheim, The Netherlands with additional facilities in Wheeling, Illinois; Unanderra, Australia; and Milan, Italy, as well as IDEX shared manufacturing facilities in India and China. Approximately 61% of Dispensing's 2015 sales were to customers outside the U.S.

# INFORMATION APPLICABLE TO THE COMPANY'S BUSINESS IN GENERAL AND ITS SEGMENTS Competitors

The Company's businesses participate in highly competitive markets. IDEX believes that the principal points of competition are product quality, price, design and engineering capabilities, product development, conformity to customer specifications, quality of post-sale support, timeliness of delivery, and effectiveness of our distribution channels.

Principal competitors of the Fluid & Metering Technologies segment are the Pump Solutions Group (Maag, Blackmer and Wilden products) of Dover Corporation (with respect to pumps and small horsepower compressors used in liquified petroleum gas distribution facilities, rotary gear pumps, and air-operated double-diaphragm pumps); Milton Roy LLC (with respect to metering pumps and controls); and Tuthill Corporation (with respect to rotary gear pumps). Principal competitors of the Health & Science Technologies segment are the Thomas division of Gardner Denver, Inc. (with respect to vacuum pumps and compressors); Thermo Scientific Dionex products (with respect to analytical

instrumentation); Parker Hannifin (with respect to sealing devices); Valco Instruments Co., Inc. (with respect to fluid injectors and valves); and Gooch & Housego PLC (with respect to electro-optic and precision photonics solutions used in the life sciences market).

The principal competitors of the Fire & Safety/Diversified Products segment are Waterous Company, a unit of American Cast Iron Pipe Company (with respect to truck-mounted firefighting pumps); Holmatro, Inc. (with respect to rescue tools); CPS

Color Group Oy (with respect to dispensing and mixing equipment for the paint industry); and Panduit Corporation (with respect to stainless steel bands, buckles and clamping systems).

Customers

The principal customers for our products are discussed immediately above by product category in each segment. None of our customers in 2015 accounted for more than two percent of net sales.

Employees

At December 31, 2015, the Company had 6,801 employees. Approximately 7% of employees were represented by labor unions, with various contracts expiring through June 2020. Management believes that the Company has a positive relationship with its employees. The Company historically has been able to renegotiate its collective bargaining agreements satisfactorily, with its last work stoppage in March 1993. Suppliers

The Company manufactures many of the parts and components used in its products. Substantially all materials, parts and components purchased by the Company are available from multiple sources.

Inventory and Backlog

The Company regularly and systematically adjusts production schedules and quantities based on the flow of incoming orders. Backlogs typically are limited to one to one and a half months of production. While total inventory levels also may be affected by changes in orders, the Company generally tries to maintain relatively stable inventory levels based on its assessment of the requirements of the various industries served.

Raw Materials

The Company uses a wide variety of raw materials which are generally available from a number of sources. As a result, shortages from any single supplier have not had, and are not likely to have a material impact on operations. Shared Services

The Company has production facilities in Suzhou, China and Vadodara, India that support multiple business units. IDEX also has personnel in China, India, Dubai, Latin America and Singapore that provide sales and marketing, product design and engineering, and sourcing support to its business units, as well as personnel in various locations in South America, the Middle East and Japan to support sales and marketing efforts of IDEX businesses in those regions. Segment Information

For segment financial information for the years 2015, 2014 and 2013, including financial information about foreign and domestic sales and operations, see "Management's Discussion and Analysis of Financial Condition and Results of Operations" and Note 11 of the Notes to Consolidated Financial Statements in Part II, Item 8, "Financial Statements and Supplementary Data."

#### Executive Officers of the Registrant

Set forth below are the names of the executive officers of the Company, their ages, years of service, the positions held by them, and their business experience during the past five years.

		Years	
Name	Age	of	Position
	_	Service	
Andrew K. Silvernail	45	7	Chairman of the Board and Chief Executive Officer
Heath A. Mitts	45	10	Senior Vice President and Chief Financial Officer
Eric D. Ashleman	48	7	Senior Vice President and Chief Operating Officer
Denise R. Cade	53	<1	Senior Vice President, General Counsel and Corporate Secretary
Daniel J. Salliotte	49	11	Senior Vice President-Corporate Strategy, Mergers and Acquisitions and Treasury
Michael J. Yates	50	10	Vice President and Chief Accounting Officer
Jeffrey D. Bucklew	45	4	Senior Vice President-Chief Human Resources Officer
M 011 111	1		

Mr. Silvernail has served as Chief Executive Officer since August 2011 and as Chairman of the Board since January 2012. Prior to that, Mr. Silvernail was Vice President-Group Executive Health & Science Technologies, Global Dispensing and Fire & Safety/Diversified Products from January 2011 to August 2011. From February 2010 to December 2010, Mr. Silvernail was Vice President-Group Executive Health & Sciences Technologies and Global Dispensing. Mr. Silvernail joined IDEX in January 2009 as Vice President-Group Executive Health & Science Technologies.

Mr. Mitts has served as Senior Vice President and Chief Financial Officer since March 2011. Mr. Mitts joined IDEX as Vice President-Corporate Finance in September 2005.

Mr. Ashleman has served as Senior Vice President and Chief Operating Officer since July 2015. Mr. Ashleman joined IDEX in 2008 as the President of Gast Manufacturing.

Ms. Cade has served as Senior Vice President, General Counsel and Corporate Secretary since joining IDEX in October 2015. Prior to joining IDEX, Ms. Cade was Senior Vice President, General Counsel, Corporate Secretary and Chief Compliance Officer for SunCoke Energy, Inc. from March 2011 to October 2015 and held various roles at PPG Industries before joining SunCoke.

Mr. Salliotte has served as Senior Vice President-Mergers, Acquisitions and Treasury since February 2011.

Mr. Salliotte joined IDEX in October 2004 as Vice President-Strategy and Business Development.

Mr. Yates has served as Vice President and Chief Accounting Officer since February 2010. Mr. Yates joined IDEX as Vice President-Controller in October 2005.

Mr. Bucklew has served as the Senior Vice President-Chief Human Resources Officer since joining IDEX in March 2012. Prior to joining IDEX, Mr. Bucklew served as the Vice President of Human Resources for Accretive Health from March 2009 to March 2012.

The Company's executive officers are elected at a meeting of the Board of Directors immediately following the annual meeting of stockholders, and they serve until the meeting of the Board immediately following the next annual meeting of stockholders, or until their successors are duly elected and qualified or until their death, resignation or removal. Public Filings

Copies of the Company's annual report on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K and amendments to those reports are made available free of charge at www.idexcorp.com as soon as reasonably practicable after being filed electronically with the United States Securities and Exchange Commission (the "SEC"). Our reports are also available free of charge on the SEC's website, www.sec.gov. Information on the Company's website is not incorporated into this Form 10-K.

Item 1A. Risk Factors.

For an enterprise as diverse and complex as the Company, a wide range of factors present risks to the Company and could materially affect future developments and performance. In addition to the factors affecting specific business operations identified in connection with the description of our operations and the financial results of our operations elsewhere in this report, the most significant of these factors are as follows:

Changes in U.S. or International Economic Conditions Could Adversely Affect the Sales and Profitability of Our Businesses.

In 2015, 50% of the Company's sales were derived from domestic operations while 50% were derived from international operations. The Company's largest end markets include life sciences and medical technologies, fire and rescue, oil & gas, paint and coatings, chemical processing, agriculture, water & wastewater treatment and optical filters and components. A slowdown in the U.S. or global economy and, in particular, any of these specific end markets could reduce the Company's sales and profitability.

Conditions in Foreign Countries in Which We Operate Could Adversely Affect Our Business.

In 2015, approximately 50% of our total sales were to customers outside the U.S. We expect our international operations and export sales to continue to be significant for the foreseeable future. Our sales from international operations and our sales from export are both subject in varying degrees to risks inherent in doing business outside the U.S. These risks include the following:

possibility of unfavorable circumstances arising from host country laws or regulations;

risks of economic instability;

currency exchange rate fluctuations and restrictions on currency repatriation;

potential negative consequences from changes to taxation policies;

disruption of operations from labor and political disturbances;

• changes in tariff and trade barriers and import or export licensing requirements; and,

political instability, terrorism, insurrection or war.

Any of these events could have an adverse impact on our business and operations.

Our Inability to Continue to Develop New Products Could Limit Our Sales Growth.

The Company's sales were down 6% in 2015. Approximately 8% of our 2015 sales were derived from new products developed over the past three years. Our ability to continue to grow organically is tied in large part to our ability to continue to develop new products.

Our Growth Strategy Includes Acquisitions and We May Not be Able to Make Acquisitions of Suitable Candidates or Integrate Acquisitions Successfully.

Our historical growth has included, and our future growth is likely to continue to include, acquisitions. We intend to continue to seek acquisition opportunities both to expand into new markets and to enhance our position in existing markets throughout the world. We may not be able to successfully identify suitable candidates, negotiate appropriate acquisition terms, obtain financing needed to consummate those acquisitions, complete proposed acquisitions or successfully integrate acquired businesses into our existing operations. In addition, any acquisition, once successfully integrated, may not perform as planned, be accretive to earnings, or otherwise prove beneficial to us.

Acquisitions involve numerous risks, including the assumption of undisclosed or unindemnified liabilities, difficulties in the assimilation of the operations, technologies, services and products of the acquired companies and the diversion of management's attention from other business concerns. In addition, prior acquisitions have resulted, and future acquisitions could result, in the incurrence of substantial additional indebtedness and other expenses.

The Markets We Serve are Highly Competitive and this Competition Could Reduce our Sales and Operating Margins. Most of our products are sold in competitive markets. Maintaining and improving our competitive position will require continued investment by us in manufacturing, engineering, quality standards, marketing, customer service and support, and our distribution networks. We may not be successful in maintaining our competitive position. Our competitors may develop products that are superior to our products, or may develop methods of more efficiently and effectively providing products and

services or may adapt more quickly than us to new technologies or evolving customer requirements. Pricing pressures may require us to adjust the prices of our products to stay competitive. We may not be able to compete successfully with our existing competitors or with new competitors. Failure to continue competing successfully could reduce our sales, operating margins and overall financial performance.

We are Dependent on the Availability of Raw Materials, Parts and Components Used in Our Products. While we manufacture certain parts and components used in our products, we require substantial amounts of raw materials and purchase some parts and components from suppliers. The availability and prices for raw materials, parts and components may be subject to curtailment or change due to, among other things, suppliers' allocations to other purchasers, interruptions in production by suppliers, changes in exchange rates and prevailing price levels. Any change in the supply of, or price for, these raw materials or parts and components could materially affect our business, financial condition, results of operations and cash flow.

Significant Movements in Foreign Currency Exchange Rates May Harm Our Financial Results.

We are exposed to fluctuations in foreign currency exchange rates, particularly with respect to the Euro, Canadian Dollar, British Pound, Indian Rupee and Chinese Renminbi. Any significant change in the value of the currencies of the countries in which we do business against the U.S. Dollar could affect our ability to sell products competitively and control our cost structure, which could have a material adverse effect on our results of operations. For additional detail related to this risk, see Part II, Item 7A, "Quantitative and Qualitative Disclosure About Market Risk." An Unfavorable Outcome of Any of Our Pending Contingencies or Litigation Could Adversely Affect Us. We currently are involved in legal and regulatory proceedings. Where it is reasonably possible to do so, we accrue estimates of the probable costs for the resolution of these matters. These estimates are developed in consultation with outside counsel and are based upon an analysis of potential results, assuming a combination of litigation and settlement strategies. It is possible, however, that future operating results for any particular quarter or annual period could be materially affected by changes in our assumptions or the effectiveness of our strategies related to these proceedings. For additional detail related to this risk, see Item 3, "Legal Proceedings."

Our Intangible Assets, Including Goodwill, are a Significant Portion of Our Total Assets and a Write-off of Our Intangible Assets Would Adversely Impact Our Operating Results and Significantly Reduce Our Net Worth. Our total assets reflect substantial intangible assets, primarily goodwill and identifiable intangible assets. At December 31, 2015, goodwill and intangible assets totaled \$1,396.5 million and \$287.8 million, respectively. These assets result from our acquisitions, representing the excess of cost over the fair value of the tangible net assets we have acquired. Annually, or when certain events occur that require a more current valuation, we assess whether there has been an impairment in the value of our goodwill and identifiable intangible assets. If future operating performance at one or more of our reporting units were to fall significantly below forecast levels, we could be required to reflect, under current applicable accounting rules, a non-cash charge to operating income for an impairment. Any determination requiring the write-off of a significant portion of our goodwill or identifiable intangible assets would adversely impact our results of operations and net worth. See Note 4 in Part II, Item 8, "Financial Statements and Supplementary Data" for further discussion on goodwill and intangible assets.

A Significant or Sustained Decline in Commodity Prices, Including Oil, Could Negatively Impact the Levels of Expenditures by Certain of Our Customers.

Demand for our products depends, in part, on the level of new and planned expenditures by certain of our customers. The level of expenditures by our customers is dependent on, among other factors, general economic conditions, availability of credit, economic conditions within their respective industries and expectations of future market behavior. Volatility in commodity prices, including oil, can negatively affect the level of these activities and can result in postponement of capital spending decisions or the delay or cancellation of existing orders. The ability of our customers to finance capital investment and maintenance may also be affected by the conditions in their industries. Reduced demand for our products could result in the delay or cancellation of existing orders or lead to excess manufacturing capacity, which unfavorably impacts our absorption of fixed manufacturing costs. This reduced demand could have a material adverse effect on our business, financial condition and results of operations.

Our Success Depends on Our Executive Management and Other Key Personnel.

Our future success depends to a significant degree on the skills, experience and efforts of our executive management and other key personnel and their ability to provide the Company with uninterrupted leadership and direction. The loss of the services of any of our executive officers or a failure to provide adequate succession plans for key personnel could have an adverse impact. The availability of highly qualified talent is limited, and the competition for talent is robust. However, we provide long-term equity incentives and certain other benefits for our executive officers which provide incentives for them to make a long-term commitment to our Company. Our future success will also depend on our ability to have adequate succession plans in place and to attract, retain and develop gualified personnel. A failure to efficiently replace executive management members and other key personnel and to attract, retain and develop new qualified personnel could have an adverse effect on our operations and implementation of our strategic plan. Our Business Operations May Be Adversely Affected by Information Systems Interruptions or Intrusion. We depend on various information technologies throughout our Company to administer, store and support multiple business activities. If these systems are damaged, cease to function properly, or are subject to cyber-security attacks, such as those involving unauthorized access, malicious software and/or other intrusions, we could experience production downtimes, operational delays, other detrimental impacts on our operations or ability to provide products and services to our customers, the compromising of confidential or otherwise protected information, destruction or corruption of data, security breaches, other manipulation or improper use of our systems or networks, financial losses from remedial actions, loss of business or potential liability, and/or damage to our reputation. While we attempt to mitigate these risks by employing a number of measures, including employee training, technical security controls, and maintenance of backup and protective systems, our systems, networks, products and services remain potentially vulnerable to known or unknown threats, any of which could have a material adverse effect on our business, financial condition or results of operations.

Failure To Comply with the U.S. Foreign Corrupt Practices Act, the U.K. Bribery Act or Other Applicable Anti-bribery Laws Could Have an Adverse Effect on Our Business.

The U.S. Foreign Corrupt Practices Act, the U.K. Bribery Act and similar anti-bribery laws in other jurisdictions generally prohibit companies and their intermediaries from making improper payments for the purpose of obtaining or retaining business. Recent years have seen a substantial increase in anti-bribery law enforcement activity with more frequent and aggressive investigations and enforcement proceedings by both the Department of Justice and the SEC, increased enforcement activity by non-U.S. regulators and increases in criminal and civil proceedings brought against companies and individuals. Our policies mandate compliance with all anti-bribery laws. However, we operate in certain countries that are recognized as having governmental and commercial corruption. Our internal control policies and procedures may not always protect us from reckless or criminal acts committed by our employees or third-party intermediaries. Violations of these anti-bribery laws may result in criminal or civil sanctions, which could have a material adverse effect on our business, financial condition and results of operations.

Item 1B. Unresolved Staff Comments. None.

#### Item 2. Properties.

The Company's principal plants and offices have an aggregate floor space area of approximately 4.5 million square feet, of which 2.9 million square feet (63%) is located in the U.S. and approximately 1.7 million square feet (37%) is located outside the U.S., primarily in the U.K. (8%), Germany (7%), Italy (6%), China (4%), India (3%) and The Netherlands (2%). Management considers these facilities suitable and adequate for the Company's operations. Management believes the Company can meet demand increases over the near term with its existing facilities, especially given its operational improvement initiatives that usually increase capacity. The Company's executive office occupies 36,588 square feet of leased space in Lake Forest, Illinois and 4,420 square feet of leased space in Chicago, Illinois.

Approximately 3.0 million square feet (67%) of the principal plant and office floor area is owned by the Company, and the balance is held under lease. Approximately 1.9 million square feet (41%) of the principal plant and office floor area is held by business units in the Fluid & Metering Technologies segment; 1.4 million square feet (31%) is held by business units in the Health & Science Technologies segment; and 1.0 million square feet (21%) is held by business units in the Fire & Safety/Diversified Products segment.

### Item 3. Legal Proceedings.

The Company and four of its subsidiaries are presently named as defendants in a number of lawsuits claiming various asbestos-related personal injuries and seeking money damages, allegedly as a result of exposure to products manufactured with components that contained asbestos. These components were acquired from third party suppliers, and were not manufactured by the Company or any of the defendant subsidiaries. To date, the majority of the Company's settlements and legal costs, except for costs of coordination, administration, insurance investigation and a portion of defense costs, have been covered in full by insurance subject to applicable deductibles. However, the Company cannot predict whether and to what extent insurance will be available to continue to cover its settlements and legal costs, or how insurers may respond to claims that are tendered to them. Claims have been filed in jurisdictions throughout the United States. Most of the claims resolved to date have been dismissed without payment. The balance have been settled for various insignificant amounts. Only one case has been tried, resulting in a verdict for the affected business unit. No provision has been made in the financial statements of the Company for these asbestos-related claims, other than for insurance deductibles in the ordinary course, and the Company does not currently believe these claims will have a material adverse effect on it.

The Company is also party to various other legal proceedings arising in the ordinary course of business, none of which is expected to have a material adverse effect on it.

Item 4. Mine Safety Disclosures. Not applicable.

### PART II

Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities.

The principal market for the Company's common stock is the New York Stock Exchange, but the common stock is also listed on the Chicago Stock Exchange. As of February 16, 2016, there were approximately 6,760 stockholders of record of our common stock and there were 75,929,397 shares outstanding.

The high and low sales prices of the common stock per share and the dividends paid per share during the last two years are as follows:

	2015			2014		
	High	Low	Dividends	High	Low	Dividends
First Quarter	\$78.85	\$69.44	\$0.28	\$79.27	\$68.58	\$0.23
Second Quarter	80.31	73.80	0.32	80.85	69.17	0.28
Third Quarter	79.61	66.88	0.32	81.82	72.27	0.28
Fourth Quarter	79.59	69.40	0.32	78.97	65.91	0.28

Our payment of dividends in the future will be determined by our Board of Directors and will depend on business conditions, our earnings and other factors.

For information pertaining to securities authorized for issuance under equity compensation plans and the related weighted average exercise price, see Part III, Item 12, "Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters."

The Company's purchases of common stock during the quarter ended December 31, 2015 are as follows:

			Total Number of	Maximum Dollar
	Total Number of	Average Price	Shares Purchased as	Value that May Yet
Period	Shares Purchased	U	Part of Publicly	be Purchased Under
	Shares I urchaseu	I alu per Share	Announced Plans	the Plans
			or Programs <sup>(1)</sup>	or Programs <sup>(1)</sup>
October 1, 2015 to October 31, 2015	189,470	\$74.60	189,470	\$ 351,872,224
November 1, 2015 to November 30, 2015		—	—	351,872,224
December 1, 2015 to December 31, 2015	219,803	76.94	219,803	634,960,648
Total	409,273	\$75.86	409,273	\$ 634,960,648

On December 1, 2015, the Company's Board of Directors approved an increase of \$300.0 million in the authorized (1) level for repurchases of common stock. This followed the prior Board of Directors approved repurchase authorization of \$400.00 million, announced by the Company on November 6, 2014. These authorizations have no

(\*) authorization of \$400.00 million, announced by the Company on November 6, 2014. These authorizations have no expiration date.

Performance Graph. The following table compares total stockholder returns over the last five years to the Standard & Poor's (the "S&P") 500 Index, the S&P Midcap Industrials Sector Index and the Russell 2000 Index assuming the value of the investment in our common stock and each index was \$100 on December 31, 2010. Total return values for our common stock, the S&P 500 Index, S&P Midcap Industrials Sector Index and the Russell 2000 Index were calculated on cumulative total return values assuming reinvestment of dividends. The stockholder return shown on the graph below is not necessarily indicative of future performance.

	12/10	12/11	12/12	12/13	12/14	12/15
IDEX Corporation	\$100.00	\$94.76	\$118.94	\$188.78	\$198.98	\$195.83
S&P 500 Index	\$100.00	\$100.00	\$113.40	\$146.97	\$163.71	\$162.47
S&P Midcap 400 Industrials Sector Index	\$100.00	\$98.18	\$117.97	\$168.05	\$171.14	\$161.34
Russell 2000 Index	\$100.00	\$94.55	\$114.43	\$148.48	\$153.73	\$144.95
12						

Item 6. Selected Financial Data.<sup>(1)</sup>

(Dollars in									
thousands,	2015	2014	2013	2012	2011				
except per share	2013	2014	2013	2012	2011				
data)									
<b>RESULTS OF</b>									
OPERATIONS									
Net sales	\$2,020,668	\$2,147,767	\$2,024,130	\$1,954,258	\$ 1,838,451				
Gross profit	904,315	949,315	873,364	803,700	738,673				
Selling, general									
and	479,408	504,419	477,851	444,490	421,703				
administrative		••••		•••,••					
expenses									
Gain on sale of	(18,070)	)							
business	· · · ·								
Restructuring	11,239	13,672	_	32,473	12,314				
expenses									
Asset				198,519	—				
impairments Operating									
income	431,738	431,224	395,513	128,218	304,656				
Other (income)									
expense — net	(2,243)	(3,111)	178	(236)	1,443				
Interest expense	41 636	41,895	42,206	42,250	29,332				
Provision for									
income taxes	109,538	113,054	97,914	48,574	80,024				
Net income	282,807	279,386	255,215		\$ 3,744,642	\$ 3,596,436	\$ 8,479,786	\$ 1,327,326	\$ 2,

(1) Includes the results of Cherry Hill acquired effective January 1, 2005 and Rudolph and Sletten acquired October 3, 2005.

(2) Income available for common stockholders includes adjustments to net income for (a) accrued dividends on our \$21.25 Preferred Stock, or \$2.125 Depositary Shares, (b) the reversal of previously accrued and unpaid dividends in the amount of approximately \$7.3 million applicable to 440,627 of the \$2.125 Depositary Shares purchased and retired by us in June 2003, (c) the reversal of previously accrued and unpaid dividends in the amount of approximately \$2.3 million applicable to 374,185 of the \$2.125 Depositary Shares purchased and retired by us in November 2005, and (d) the \$0.3 million excess of fair value over carrying value upon redemption of the remaining outstanding \$2.125 Depositary Shares in May 2006.

- (3) Includes a \$23.6 million after-tax charge related to an adverse judgment received in the WMATA matter.
- (4) A construction project is included in our backlog at such time as a contract is awarded or a letter of commitment is obtained and adequate construction funding is in place. Backlog is not a measure defined in accounting principles generally accepted in the United States of America, or GAAP, and our backlog may not be comparable to the backlog of other companies. Management uses backlog to assist in forecasting future results.
- (5) New business awarded consists of the original contract price of projects added to our backlog in accordance with Note (4) above plus or minus subsequent changes to the estimated total contract price of existing contracts. Management uses new business awarded to assist in forecasting future results.

Tutor-Saliba Selected Historical Financial Information

The selected historical financial data set forth below is derived in part from and should be read in conjunction with Tutor-Saliba s consolidated financial statements, the related notes and Management s Discussion and Analysis of Financial Condition and Results of Operations included elsewhere in this proxy statement. The consolidated statement of income data for each of the years ended December 31, 2005, 2006 and 2007 and the consolidated balance sheet data as of December 31, 2006 and 2007 were derived from Tutor-Saliba s audited consolidated financial statements appearing in the annexes to this proxy statement. The consolidated statement of income data for the three-month periods ended March 31, 2007 and 2008 and the consolidated balance sheet data as of March 31, 2007 and 2008 were derived from Tutor-Saliba s unaudited consolidated financial statements appearing in the annexes to this proxy statement. This information is unaudited but, in Tutor-Saliba management s opinion, has been prepared on the same basis as the audited consolidated financial statements and related notes included elsewhere in this proxy statement and includes all adjustments, consisting only of normal recurring adjustments, that Tutor-Saliba s management considers necessary for a fair presentation of the information for the periods presented. The consolidated statement of income data for the years ended December 31, 2003 and 2004 and the consolidated balance sheet data as of December 31, 2003, 2004 and 2005 were derived from Tutor-Saliba s audited consolidated financial statements not included herein. Historical results are not necessarily indicative of results to be expected for future periods.

	Three Mor Marc 2008		2007	r 31, 2004	2003		
Consolidated statement of income data:			(III thous	ands, except sl	lare data)		
Revenues:							
Domestic civil	\$ 34,196	\$ 56,135	\$ 174,277	\$ 141,875	\$ 117,046	\$ 194,082	\$ 208,543
Domestic building	341,470	160,346	896,124	320,403	228,905	250,836	204,275
International	19,297	15,579	81,423	52,179	41,639	71,118	66,126
Total	394.963	232.060	1.151.824	514,457	387,590	516.036	478,944
Cost of revenues	394,903	232,000	1,131,824	485,434	371,334	495,103	448,338
Cost of revenues	500,559	211,404	1,005,005	405,454	571,554	495,105	440,550
Gross profit	34,604	20,656	88,221	29,023	16,256	20,933	30,606
Gain on sale of property and equipment	(149)	(50)	(446)	(454)	(7,524)		
Cost of contract litigation settlement					14,652(4)		
General and administrative expenses	13,557	7,754	36,237	26,823	26,352	22,739	21,702
Income (loss) from operations	21,196	12,952	52,430	2,654	(17,224)	(1,806)	8,904
Other income, net (1)	1,263	12,142	99,206	11,479	40,504	17,114	11,593
Interest expense	(779)	(1,005)	(4,197)	(5,257)	(3,414)	(3,750)	(4,519)
Income before minority interest and income							
taxes	21,680	24,089	147,439	8,876	19,866	11,558	15,978
Provision for income taxes (3)	(619)	(797)	(4,399)	(1,663)	(849)	(309)	(2,166)
Income before minority interest	21,061	23,292	143,040	7,213	19,017	11,249	13,812
Minority interest	21,001	(20)	(111)	(40)	(16)	(204)	(29)
Willoffty Interest		(20)	(111)	(40)	(10)	(204)	(29)
Income from continuing operations	21,061	23,272	142,929	7,173	19,001	11,045	13,783
Income (loss) from discontinued operations, net of taxes (2)		(39)	226	(36)	1,407	11,803	(79)
Net income	\$ 21,061	\$ 23,233	\$ 143,155	\$ 7,137	\$ 20,408	\$ 22,848	\$ 13,704
Income per share from continuing operations	\$ 23.40	\$ 25.85	\$ 158.80	\$ 7.97	\$ 21.11	\$ 12.27	\$ 15.31
Income (loss) per share from discontinued operations, net of taxes		(0.04)	0.25	(0.04)	1.56	13.11	(0.09)
operations, net of areas		(0.04)	0.25	(0.04)	1.50	13.11	(0.07)
Net income per share	\$ 23.40	\$ 25.81	\$ 159.05	\$ 7.93	\$ 22.67	\$ 25.38	\$ 15.22
Weighted average number of shares	900,043	900,043	900,043	900,043	900,043	900,043	900,043

(1) Primarily reflects unrealized holding gains and gains on sales of marketable securities.

(2) Reflects the results of operations associated with equipment and real estate operations unrelated to construction activities, including a gain on sale of equipment of \$11.8 million in 2004.

(3) Since January 1, 1996, Tutor-Saliba has been a subchapter S corporation and has been exempt from paying federal income taxes. In addition, from and after the day Tutor-Saliba elected or was otherwise treated as a subchapter S corporation for state tax purposes, Tutor-Saliba has paid certain state income taxes at a reduced rate.

(4) Represents charge recorded related to settlement of contract litigation on the San Francisco International Airport project.

	March 31,			December 31,							
		2008		2007		2007		2006	2005	2004	2003
						(iı	n th	ousands)			
Consolidated balance sheet data:											
Working capital	\$	96,311	\$	151,352	\$	147,541	\$	126,885	\$ 133,659	\$ 137,381	\$ 47,305
Current ratio		1.26x		1.61x		1.45x		1.60x	1.80x	1.94x	1.26x
Long term debt, less current maturities		26,905		60,038		53,617		60,221	38,456	50,398	61,582
Shareholders equity		184,515		188,096		211,630		161,713	164,939	159,948	131,740
Ratio of long-term debt to equity		.15x		.32x		.25x		.37x	.23x	.32x	.47x
Total assets		603,980		503,741		601,493		441,757	382,748	364,956	379,476
Other data:											
Backlog at end of period (1)	\$1	,309,996	\$2	2,151,760	\$	1,591,522	\$	2,282,318	\$ 508,634	\$ 500,231	\$741,293
New business awarded (2)	\$	113,437	\$	101,502	\$	461,028	\$	2,288,091	\$ 396,043	\$ 274,974	\$557,007

- (1) A construction project is included in our backlog at such time as a contract is awarded or a letter of commitment is obtained and adequate construction funding is in place. Backlog is not a measure defined in accounting principles generally accepted in the United States of America, or GAAP, and our backlog may not be comparable to the backlog of other companies. Management uses backlog to assist in forecasting future results.
- (2) New business awarded consists of the original contract price of projects added to our backlog in accordance with Note (1) above plus or minus subsequent changes to the estimated total contract price of existing contracts. Management uses new business awarded to assist in forecasting future results.

#### Selected Unaudited Pro Forma Condensed Combined Financial Information

The merger will be accounted for by using the purchase method of accounting in accordance with GAAP. The tangible and intangible assets and liabilities of Tutor-Saliba will be recorded as of the closing date of the merger, at their respective fair values, and assumed by and added to those of Perini. For a detailed description of the purchase accounting method, please see The Merger Anticipated Accounting Treatment beginning on page 66.

The following selected unaudited pro forma condensed combined balance sheet information as of March 31, 2008 and the selected unaudited pro forma condensed combined statements of income information for the year ended December 31, 2007 and the three months ended March 31, 2008 are based on the separate historical consolidated financial statements of Perini and Tutor-Saliba, and reflect the merger and related events and apply the assumptions and adjustments described in the notes to the unaudited pro forma condensed combined financial statements. The pro forma adjustments are more fully described in the notes to the unaudited pro forma condensed combined financial statements beginning on page 108 of this proxy statement. The selected unaudited pro forma condensed combined balance sheet as of March 31, 2008 reflects the merger and related events as if they had been consummated on March 31, 2008. The selected unaudited pro forma condensed combined statements of income for the year ended December 31, 2007 and the three months ended March 31, 2008 reflect the merger and related events as if they had been consummated on Perini s 2007 fiscal year.

The pro forma adjustments are based upon available information and assumptions that the managements of Perini and Tutor-Saliba believe reasonably reflect the merger. The selected unaudited pro forma condensed combined financial statements do not include the effects of the costs associated with any restructuring or

integration activities resulting from the transaction. In addition, the selected unaudited pro forma condensed combined financial statements do not include the realization of any cost savings from operating efficiencies or synergies resulting from the transaction, nor do they include any potential incremental revenues and earnings that may be achieved with the combined capabilities of the companies. The final purchase price allocation, which will be determined subsequent to the closing of the merger, and its effect on results of operations may differ significantly from the pro forma amounts included in the selected unaudited pro forma condensed combined financial statements. These amounts represent the managements best estimate as of the date of this proxy statement. We present the unaudited pro forma condensed combined financial statements for informational purposes only. The selected pro forma condensed combined financial statements are not necessarily indicative of what our financial position or results of operations actually would have been had we completed the merger as of the dates indicated. In addition, the selected unaudited pro forma condensed combined financial statements do not purport to project the future financial position or operating results of the combined company.

The following selected unaudited pro forma condensed combined financial information (a) has been derived from, and should be read in conjunction with, the unaudited pro forma condensed combined financial statements and the accompanying notes included in this proxy statement beginning on page 118 and (b) should be read in conjunction with the consolidated financial statements of Perini and Tutor-Saliba and other information filed by Perini with the SEC and incorporated by reference in this proxy statement. Please see Where You Can Find Additional Information beginning on page 152.

G&A Expense       40,718       159,495         Income From Construction Operations       58,791       205,162         Other Income (Expense), Net       2,768       114,630(1)         Interest Expense       (1,822)       (8,962)         Income Before Minority Interest and Income Taxes       59,737       310,830         Provision for Income Taxes       59,737       310,830         Income Before Minority Interest       37,398       194,730         Minority Interest       37,398       194,619         Per Share of Common Stock:       3830       194,619         Per Share of Common Stock:       50,75       \$ 3,91         Diluted Earnings       \$ 0.75       \$ 3,91         Diluted Earnings       \$ 0.74       \$ 3,86(2)         Weighted Average Common shares Outstanding:       3836         Basic       50,132       49,806		Three Months	
PRO FORMA STATEMENTS OF INCOME DATA:           Revenues         \$ 1,640,186         \$ 5,875,295           Cost of Operations         1,540,677         5,510,638           Gross Profit         99,509         364,657           G&A Expense         40,718         159,495           Income From Construction Operations         58,791         205,162           Other Income (Expense), Net         2,768         114,630(1)           Interest Expense         (1,822)         (8,962)           Income Before Minority Interest and Income Taxes         59,737         310,830           Provision for Income Taxes         (22,339)         (116,100)           Income Before Minority Interest         37,398         194,730           Minority Interest         37,398         194,619           Per Share of Common Stock:         Basic Earnings         \$ 0.75         \$ 3,91           Diluted Earnings         \$ 0.74         \$ 3,86(2)           Weighted Average Common shares Outstanding:         U         U         1,830		Mar. 31, 2008 (In thousands, ex	Dec. 31, 2007 cept per share
Revenues       \$ 1,640,186       \$ 5,875,295         Cost of Operations       1,540,677       5,510,638         Gross Profit       99,509       364,657         G&A Expense       40,718       159,495         Income From Construction Operations       58,791       205,162         Other Income (Expense), Net       2,768       114,630(1)         Interest Expense       (1,822)       (8,962)         Income Before Minority Interest and Income Taxes       59,737       310,830         Provision for Income Taxes       59,737       310,830         Provision for Income Taxes       (22,339)       (116,100)         Income Before Minority Interest       37,398       194,730         Minority Interest       (111)       (111)         Net Income       \$ 37,398       \$ 194,619         Per Share of Common Stock:       Basic Earnings       \$ 0.75       \$ 3.91         Diluted Earnings       \$ 0.75       \$ 3.91       3.86(2)         Weighted Average Common shares Outstanding:       Basic       50,132       49,806	DDO EODMA CTATEMENTE OF INCOME DATA.	data	ı)
Cost of Operations       1,540,677       5,510,638         Gross Profit       99,509       364,657         G&A Expense       40,718       159,495         Income From Construction Operations       58,791       205,162         Other Income (Expense), Net       2,768       114,630(1)         Interest Expense       (1,822)       (8,962)         Income Before Minority Interest and Income Taxes       59,737       310,830         Provision for Income Taxes       59,737       310,830         Provision for Income Taxes       (22,339)       (116,100)         Income Before Minority Interest       37,398       194,730         Minority Interest       37,398       194,619         Per Share of Common Stock:       9       0.75       \$ 3.91         Diluted Earnings       \$ 0.75       \$ 3.91       3.86(2)         Weighted Average Common shares Outstanding:       50,132       49,806		¢ 1 640 196	¢ 5 975 205
Gross Profit       99,509       364,657         G&A Expense       40,718       159,495         Income From Construction Operations       58,791       205,162         Other Income (Expense), Net       2,768       114,630(1)         Interest Expense       2,768       114,630(2)         Income Before Minority Interest and Income Taxes       59,737       310,830         Provision for Income Taxes       (22,339)       (116,100)         Income Before Minority Interest       37,398       194,730         Minority Interest       37,398       194,619         Per Share of Common Stock:       Basic Earnings       \$       0.75       \$       3.91         Diluted Earnings       \$       0.74       \$       3.86(2)			
G&A Expense       40,718       159,495         Income From Construction Operations       58,791       205,162         Other Income (Expense), Net       2,768       114,630(1)         Interest Expense       (1,822)       (8,962)         Income Before Minority Interest and Income Taxes       59,737       310,830         Provision for Income Taxes       59,737       310,830         Income Before Minority Interest       37,398       194,730         Minority Interest       37,398       194,619         Per Share of Common Stock:       3830       194,619         Per Share of Common Stock:       50,75       \$ 3,91         Diluted Earnings       \$ 0.75       \$ 3,91         Diluted Earnings       \$ 0.74       \$ 3,86(2)         Weighted Average Common shares Outstanding:       3836         Basic       50,132       49,806	Cost of Operations	1,540,677	5,510,638
G&A Expense       40,718       159,495         Income From Construction Operations       58,791       205,162         Other Income (Expense), Net       2,768       114,630(1)         Interest Expense       (1,822)       (8,962)         Income Before Minority Interest and Income Taxes       59,737       310,830         Provision for Income Taxes       59,737       310,830         Income Before Minority Interest       37,398       194,730         Minority Interest       37,398       194,619         Per Share of Common Stock:       \$       0.75       \$         Basic Earnings       \$       0.74       \$       3.86(2)         Weighted Average Common shares Outstanding:       #       #       #       #         Basic       50,132       49,806       #       \$       9,806	Gross Profit	99,509	364.657
Other Income (Expense), Net2,768114,630(1)Interest Expense(1,822)(8,962)Income Before Minority Interest and Income Taxes59,737310,830Provision for Income Taxes(22,339)(116,100)Income Before Minority Interest37,398194,730Minority Interest37,398194,730Minority Interest(111)Net Income\$ 37,398\$ 194,619Per Share of Common Stock:	G&A Expense		
Other Income (Expense), Net2,768114,630(1)Interest Expense(1,822)(8,962)Income Before Minority Interest and Income Taxes59,737310,830Provision for Income Taxes(22,339)(116,100)Income Before Minority Interest37,398194,730Minority Interest37,398194,730Minority Interest(111)Net Income\$ 37,398\$ 194,619Per Share of Common Stock:			
Interest Expense(1,822)(8,962)Income Before Minority Interest and Income Taxes59,737310,830Provision for Income Taxes(22,339)(116,100)Income Before Minority Interest37,398194,730Minority Interest37,398194,619Net Income\$37,398194,619Per Share of Common Stock: Basic Earnings\$0.75\$3.91Diluted Earnings\$0.74\$3.86(2)Weighted Average Common shares Outstanding: Basic50,13249,806			
Income Before Minority Interest and Income Taxes59,737310,830Provision for Income Taxes(22,339)(116,100)Income Before Minority Interest37,398194,730Minority Interest37,398194,619Net Income\$37,398194,619Per Share of Common Stock: Basic Earnings\$0.75\$3.91Diluted Earnings\$0.74\$3.86(2)Weighted Average Common shares Outstanding: Basic50,13249,806			
Provision for Income Taxes(22,339)(116,100)Income Before Minority Interest37,398194,730Minority Interest(111)Net Income\$ 37,398\$ 194,619Per Share of Common Stock: Basic Earnings\$ 0.75\$ 3.91Diluted Earnings\$ 0.75\$ 3.91Diluted Earnings\$ 0.74\$ 3.86(2)Weighted Average Common shares Outstanding: Basic50,13249,806	Interest Expense	(1,822)	(8,962)
Income Before Minority Interest 37,398 194,730 Minority Interest 1111 Net Income \$ 37,398 194,619 Per Share of Common Stock: Basic Earnings \$ 0.75 \$ 3.91 Diluted Earnings \$ 0.74 \$ 3.86(2) Weighted Average Common shares Outstanding: Basic 50,132 49,806	Income Before Minority Interest and Income Taxes	59,737	310,830
Minority Interest(111)Net Income\$ 37,398\$ 194,619Per Share of Common Stock: Basic Earnings\$ 0.75\$ 3.91Diluted Earnings\$ 0.74\$ 3.86(2)Weighted Average Common shares Outstanding: Basic50,13249,806	Provision for Income Taxes	(22,339)	(116,100)
Minority Interest(111)Net Income\$ 37,398\$ 194,619Per Share of Common Stock: Basic Earnings\$ 0.75\$ 3.91Diluted Earnings\$ 0.74\$ 3.86(2)Weighted Average Common shares Outstanding: Basic50,13249,806	Income Before Minority Interest	37.398	194,730
Per Share of Common Stock: Basic Earnings \$ 0.75 \$ 3.91 Diluted Earnings \$ 0.74 \$ 3.86(2) Weighted Average Common shares Outstanding: Basic 50,132 49,806	Minority Interest	· · · · · ·	(111)
Basic Earnings\$0.75\$3.91Diluted Earnings\$0.74\$3.86(2)Weighted Average Common shares Outstanding: Basic50,13249,806	Net Income	\$ 37,398	\$ 194,619
Basic Earnings\$0.75\$3.91Diluted Earnings\$0.74\$3.86(2)Weighted Average Common shares Outstanding: Basic50,13249,806			
Diluted Earnings \$ 0.74 \$ 3.86(2) Weighted Average Common shares Outstanding: Basic 50,132 49,806			
Weighted Average Common shares Outstanding: Basic 50,132 49,806	Basic Earnings	\$ 0.75	\$ 3.91
Basic 50,132 49,806	Diluted Earnings	\$ 0.74	\$ 3.86(2)
, , ,	Weighted Average Common shares Outstanding:		
Diluted 50,640 50,406	Basic	50,132	49,806
	Diluted	50,640	50,406

(1) Includes \$94,105 non-recurring gain on sale of marketable securities.

(2) Pro forma diluted earnings per common share excluding the non-recurring gain on sale of marketable securities (see Note (1) above) is \$2.70.

#### **Closing Price of Perini Common Stock**

The following table includes the closing prices per share of Perini common stock as reported on the NYSE Composite Transaction Tape on:

April 2, 2008, the date of the public announcement of the execution of the Merger Agreement after the end of the full trading day, and

[], 2008, the most recent practicable date prior to the mailing of this proxy statement to the Perini shareholders.

	Perini Common Stock Closing Price Per Shar	
April 2, 2008	\$ 38.25	5
[], 2008	\$	[]

#### NOTE REGARDING FORWARD-LOOKING STATEMENTS

We have made forward-looking statements in this proxy statement, and in documents that are incorporated by reference in this proxy statement, that are subject to risks and uncertainties. These statements are based on the current beliefs and assumptions of our management. Generally, forward-looking statements include information concerning possible or assumed future actions, events or results of operations of Perini, Tutor-Saliba or the combined company. Forward-looking statements include the information in this proxy statement regarding:

management forecasts;

regulatory matters;

financial projections and estimates;

efficiencies/cost avoidances;

cost savings;

income and margins;

earnings per share;

growth;

economies of scale;

combined operations;

the economy;

future economic performance;

conditions to, and the timetable for, completing the merger;

future acquisitions and dispositions;

litigation;

potential and contingent liabilities;

management s plans;

business portfolios;

taxes; and

merger and integration-related expenses.

These statements may be preceded by, followed by or otherwise include the words may, will, should, could, would, potential, possible, expects, anticipates, intends, plans, estimates, hopes or similar expressions. We claim the protection of the safe harbor for forward-looking statements contained in the Private Securities Litigation Reform Act of 1995 for all forward-looking statements.

Forward-looking statements are not guarantees of performance. You should understand that the following important factors, in addition to those discussed in Risk Factors beginning on page 26 and elsewhere in this proxy statement, and in the documents which are incorporated by reference in this proxy statement, could affect the future results of Perini, and of the combined company after the completion of the merger, and could cause those results or other outcomes to differ materially from those expressed or implied in our forward-looking statements:

the ability of Perini and Tutor-Saliba to satisfy all conditions precedent to the completion of the merger;

the ability of Perini and Tutor-Saliba to integrate their operations successfully;

the timing of the integration of Perini and Tutor-Saliba necessary to achieve enhanced earnings or realize cost savings;

the retention of existing, and continued attraction of additional customers and employees;

an unsolicited offer by another company to acquire the assets or capital stock of Perini;

unexpected costs or unexpected liabilities related to the merger, or the effects of purchase accounting that may be different from our current expectations;

the effects of uncertainty surrounding the merger that may cause the business of Perini to suffer;

other economic, business and competitive factors;

the costs and other effects of legal proceedings;

the impact on the trading price of Perini common stock of resales in the public market of shares of Perini common stock received by Tutor-Saliba shareholders in the merger;

the ability of Mr. Tutor to exert significant influence over corporate decisions as a result of his ownership of Perini common stock following the merger, his position as chairman and chief executive officer of the combined company and his right to designate up to two nominees for election as directors of the combined company;

changes in accounting policies, practices or their interpretations; and

#### the factors described in Perini s reports filed with the SEC.

Because forward-looking statements are subject to assumptions and uncertainties, actual results may differ materially from those expressed or implied by such forward-looking statements. Shareholders are cautioned not to place undue reliance on such statements, which speak only as of the date of this proxy statement or the date of any document incorporated by reference.

All written and oral forward-looking statements concerning the merger or other matters addressed in this proxy statement or incorporated by reference in this proxy statement and attributable to Perini or any person acting on its behalf are expressly qualified in their entirety by the cautionary statements contained or referred to in this section. Except to the extent required by applicable law or regulation, Perini undertakes no obligation to release any revisions or updates to such forward-looking statements to reflect events or circumstances after the date of this proxy statement or to reflect the occurrence of unanticipated events.

#### ANNUAL MEETING OF THE SHAREHOLDERS

#### OF PERINI CORPORATION

This proxy statement is furnished in connection with the solicitation of proxies by the Perini board of directors for use at our annual meeting of shareholders to be held on [], 2008, and at any adjournment or postponements thereof.

#### Date, Time and Place

The annual meeting of shareholders will be held at the [ ], on [ ], 2008, at [ ], local time.

#### Proposals to be Considered at the Annual Meeting

At the annual meeting, and any adjournment or postponement of the annual meeting, Perini shareholders will be asked to consider and vote upon the following proposals:

- 1. a proposal to approve the issuance of 22,987,293 shares of Perini common stock in the merger contemplated by the Merger Agreement;
- 2. a proposal to amend our amended and restated articles of organization to increase the number of authorized shares of Perini common stock from 40 million to 75 million shares;
- 3. the election of four (4) Class III Directors, to hold office for a three-year term, expiring at our 2011 annual meeting of shareholders and until their successors are duly elected and qualified;
- 4. a proposal to ratify the selection of Deloitte & Touche LLP, independent registered public accountants, as auditors of Perini for the fiscal year ending December 31, 2008;
- 5. a proposal to amend our 2004 Stock Option and Incentive Plan to increase the number of shares authorized for issuance under the plan from 3 million to 5.5 million;
- 6. a proposal to approve the adjournment or postponement of the annual meeting if necessary to solicit additional proxies or as otherwise necessary; and

## 7. such other business as may properly come before the meeting. Shareholders Entitled to Vote

Only our shareholders of record at the close of business on July 30, 2008, also referred to in this proxy statement as the record date, will be entitled to notice of, and to vote at, the annual meeting and any adjournment or postponement thereof. Notwithstanding the record date specified above, our stock transfer books will not be closed and shares may be transferred subsequent to the record date. However, all votes must be cast in the names of shareholders of record on the record date.

On the record date, there were approximately 27.3 million shares of Perini common stock issued and outstanding held by approximately 784 shareholders of record.

#### **Shareholder Votes Required**

*Proposal 1*: Approval of the share issuance proposal requires the affirmative vote of the holders of a majority of Perini common stock casting votes at the annual meeting (provided that the total votes cast on the proposal represent over 50% of the outstanding shares of Perini common stock entitled to vote on the proposal at the annual meeting). **The approval of Proposal 1 is a condition to the merger, and thus a vote against this proposal effectively will be a vote against the merger.** 

*Proposal 2*: Approval of the articles amendment proposal requires the affirmative vote of the holders of a majority of the outstanding shares of Perini common stock entitled to vote at the annual meeting. The approval of Proposal 2 is a condition to the merger, and thus a vote against this proposal effectively will be a vote against the merger.

*Proposal 3*: Election of each of the nominees for director requires the affirmative vote of a plurality of the votes cast at the annual meeting. You may vote FOR any or all director nominees and/or WITHHOLD your vote from any or all of the director nominees.

*Proposal 4*: Ratification of the selection of Deloitte & Touche, LLP as our independent auditors for 2008 requires the affirmative vote of a majority of the votes cast on the proposal at the annual meeting.

*Proposal 5*: Approval of the plan amendment proposal requires the affirmative vote of the holders of a majority of Perini common stock casting votes at the annual meeting (provided that the total votes cast on the proposal represent over 50% of the outstanding shares of Perini common stock entitled to vote on the proposal at the annual meeting).

*Proposal 6*: Approval of a proposal to adjourn or postpone the annual meeting requires the affirmative vote of the holders of a majority of the votes cast on the proposal at the annual meeting, whether or not a quorum is present.

### Quorum

The presence, in person or by proxy, of outstanding shares of Perini common stock representing a majority of the shares entitled to vote is necessary to constitute a quorum for the transaction of business at the annual meeting. Shares that reflect abstentions or broker non-votes will be counted for purposes of determining whether a quorum is present for the transaction of business at the annual meeting. Stock owned directly or indirectly by Perini, if any, is not considered outstanding for purposes of determining a quorum.

### Abstentions and Broker Non-Votes

An abstention occurs when a shareholder sends in a proxy with explicit instructions to decline to vote regarding a particular matter. For purposes of establishing a quorum, abstentions in person and proxies received but marked as abstentions as to any or all matters to be voted on count as present.

Abstentions with respect to the articles amendment proposal (Proposal 2) will have the same effect as a vote against the proposal. Abstentions with respect to the share issuance proposal (Proposal 1) or the plan amendment proposal (Proposal 5) will have no effect on the outcome of the proposal, except that it will be more difficult for Perini to ensure that the total votes cast on each of the share issuance proposal and the plan amendment proposal represent over 50% in interest of all Perini common stock entitled to vote on each such proposal, which is necessary for the proposals to be approved. Abstentions have no effect on the election of directors (Proposal 3), the ratification of auditors proposal (Proposal 4) or the postponement or adjournment proposal (Proposal 6).

If your shares are held in street name, your brokerage firm, under certain circumstances, may vote your shares for you if you do not return your proxy. Brokerage firms have authority under the rules of the NYSE to vote customers unvoted shares on certain routine matters. If you do not give a proxy to your brokerage firm to vote your shares, your brokerage firm may either vote your shares on routine matters or leave your shares unvoted. The election of directors (Proposal 3), the ratification of Deloitte & Touche LLP as the Perini s independent auditors for the fiscal year ending December 31, 2008 (Proposal 4) and the proposal to adjourn or postpone the annual meeting (Proposal 6) are considered routine matters, therefore, your brokerage firm may

vote your shares for you if you do not return your proxy. The share issuance proposal (Proposal 1), the articles amendment proposal (Proposal 2) and the plan amendment proposal (Proposal 5) are not considered routine matters, therefore, your brokerage firm may not vote your shares for you if you do not return your proxy to your brokerage firm. These votes are commonly referred to as broker non-votes. Broker non-votes will not be counted as having been voted on the proposal.

Broker non-votes with respect to the articles amendment proposal (Proposal 2) will have the same effect as a vote against the proposal. Broker non-votes with respect to the share issuance proposal (Proposal 1) or the plan amendment proposal (Proposal 5) will have no effect on the outcome of the proposal, except that it will be more difficult for Perini to ensure that the total votes cast on each of the share issuance proposal and the plan amendment proposal represent over 50% in interest of all Perini common stock entitled to vote on each such proposal, which is necessary for the proposals to be approved.

Regardless of whether you are a record holder of your shares or hold your shares in street name, we encourage you to provide voting instructions to your brokerage firm by returning your completed proxy. This ensures your shares will be voted at the meeting according to your instructions. Record holders may complete the proxy card enclosed with this proxy statement and return it to us. If your shares are held in street name, you should receive directions from your brokerage firm about how to submit your proxy to them at the time you receive this proxy statement.

### **Proxy Solicitation**

In addition to solicitation by mail, our directors, officers and employees may solicit proxies from Perini shareholders by telephone, facsimile or other electronic means of communication. These persons will not receive additional or special compensation for such solicitation services. We will, upon request, reimburse brokers, banks and other nominees for their expenses in sending proxy materials to their customers who are beneficial owners and obtaining their voting instructions.

Perini has retained Innisfree M&A Incorporated, a proxy solicitation firm, to assist it in the solicitation of proxies for the annual meeting. Perini will pay Innisfree a fee of \$50,000, and an additional fee of \$25,000 if shareholders approve the merger proposals, for its services. In addition, Perini will pay Innisfree additional fees depending on the extent of additional services requested by Perini and will reimburse Innisfree for expenses Innisfree incurs in connection with its engagement by Perini.

### **Revocation of Proxies**

You may revoke your proxy at any time before the vote is taken at the annual meeting. To revoke your proxy, you must either advise our Corporate Secretary in writing, submit a proxy by telephone, via the Internet or mail dated after the date of the proxy you wish to revoke or attend the annual meeting and vote your shares in person. Attendance at the annual meeting will not by itself constitute revocation of a proxy.

Please note that if you have instructed your broker to vote your shares, the options for revoking your proxy described above do not apply and instead you must follow the directions provided by your broker to change those instructions.

#### Voting

Perini shareholders of record and many shareholders who hold their shares through a broker or bank will have the option to submit their proxies or voting instructions via the Internet or by telephone. There are separate arrangements for using the Internet and telephone to submit your proxy depending on whether you are a shareholder of record or your shares are held in street name by your broker.

Perini shareholders of record have three options for submitting their proxy.

You may submit a proxy by mail by executing and returning the enclosed proxy card in the enclosed envelope, which requires no postage if mailed in the United States. All shares of Perini common stock represented by a proxy properly signed and received at or prior to the annual meeting, unless subsequently revoked, will be voted in accordance with the instruction on the proxy.

You may also vote electronically until 2:00 a.m. eastern standard time on [], 2008 by logging on to the Internet at *www.investorvote.com/PCR* and following the instructions.

You may vote telephonically until 2:00 a.m. eastern standard time on [ ], 2008 by calling 1-800-652-VOTE (8683) and following the recorded instructions.

By casting your vote in any of the three ways listed above, you are authorizing the individuals listed on the proxy to vote your shares in accordance with your instructions. If a proxy is signed and returned without indicating any voting instructions, the shares of Perini common stock represented by the proxy will be voted (1) *FOR* each director nominee and *FOR* approval of Proposals 1, 2, 4, 5 and 6 and (2) as to any matters for which Perini did not have notice on or before [], 2008, properly brought before the annual meeting, in the sole discretion of the Perini board of directors as to such matters.

If your shares are held in street name, you should check the voting instructions card provided by your broker, bank or other nominee to see which options are available and the procedures to be followed. Please follow their instructions carefully. Also, please note that if the holder of record of your shares is a broker, bank or other nominee and you wish to vote in person at the annual meeting, you must request a legal proxy from your bank, broker or other nominee that holds your shares and present that proxy and proof of identification at the annual meeting.

### Voting by Perini Directors and Executive Officers

On the record date, directors and executive officers of Perini and their affiliates beneficially owned or had the right to vote shares of Perini common stock representing approximately 1.7% of the shares of Perini common stock outstanding on the record date. To Perini s knowledge, directors and executive officers of Perini and their affiliates intend to vote their shares of Perini common stock in favor of the merger proposals. A more detailed description of the ownership of Perini common stock by certain beneficial owners and Perini s directors and executive officers begins on page 143.

#### **Adjournments and Postponements**

Although it is not currently expected, the annual meeting may be adjourned or postponed, including for the purpose of soliciting additional proxies, by action of the Perini shareholders if approved by shareholders or by the presiding officer of the annual meeting in accordance with Perini s bylaws. In addition, the Perini board of directors may postpone and reschedule the annual meeting prior to the meeting in accordance with Perini s bylaws. Any adjournment may be made without notice, other than by an announcement made at the annual meeting of the time, date and place of the adjourned meeting, regardless of whether or not a quorum is present. The annual meeting may be adjourned or postponed by the Perini shareholders by the affirmative vote the holders of a majority of the votes cast on the proposal at the annual meeting, regardless of whether or not a quorum is present.

Any adjournment or postponement of the annual meeting for the purpose of soliciting additional proxies will allow the Perini shareholders who have already sent their proxies to revoke them any time prior to their use at annual meeting as adjourned or postponed.

### **RISK FACTORS**

Perini shareholders should carefully read and consider the following risk factors, which we believe are all of the significant risks related to the merger and the anticipated business of the combined company, as well as the other information contained and referred to in this proxy statement, before voting at the annual meeting.

### **Risks Relating to the Merger**

### If the public markets assign lower values to the Tutor-Saliba business than the values used in negotiating the terms of the merger, the trading price of Perini common stock may decline.

The stock of Tutor-Saliba is not publicly traded, so there is no current market-based valuation for Tutor-Saliba s business. We also do not believe that there are closely comparable companies with publicly traded equity that provide a reasonable, comparative basis of valuation for Tutor-Saliba. In negotiating the merger, we used what we believe to be a reasonable valuation for Tutor-Saliba. The public markets may not value the Tutor-Saliba business in the same manner as we have valued it for purposes of negotiating the terms of the merger. In addition, in negotiating the economic terms of the merger, we relied upon implied valuations of Perini and Tutor-Saliba that were based upon projections of their respective future financial performance. These projections were based on estimates, assumptions and judgments with respect to, among other things, future economic, competitive and financial market conditions and future business decisions, which are difficult to predict and which are subject to numerous risks and uncertainties. The future financial performances of Perini and Tutor-Saliba could be significantly different from those contemplated by the financial projections. If either Perini s future financial performance is materially better than projected (and Tutor-Saliba does not also perform materially better), or if Tutor-Saliba s future financial performance is materially lower than projected (and Perini s performance is not similarly lower), the market may conclude that the value assigned to Tutor-Saliba in the merger was too high. In any of these events, the trading price of Perini common stock may decline.

### The merger may not be completed, which could adversely affect Perini s business operations and stock price.

To complete the merger, Perini shareholders must approve the issuance of shares of Perini common stock as contemplated by the Merger Agreement and the amendment to the Perini amended and restated articles of organization.

In addition, the Merger Agreement contains additional closing conditions, which are described in the section The Merger Agreement beginning on page 73. These conditions may not be satisfied or waived. If we are unable to complete the merger, Perini would be subject to a number of risks, including the following:

Perini would not realize the benefits of the merger, including any synergies from combining the two companies; and

the trading price of Perini common stock may decline to the extent that the current market price reflects a market assumption that the merger will be completed.

The occurrence of any of these events individually or in combination could have a material adverse effect on the results of operations, financial position and cash flows of Perini or the trading price of our common stock.

## The combined company may not realize some or all of the expected benefits of the merger that were considered in negotiating the terms of the merger.

Perini entered into the Merger Agreement with the expectation that the merger would result in various benefits, including, among other things, synergies, cost savings and operating efficiencies. In negotiating the merger, we used what we believe to be a reasonable estimates of benefits the combined company will start to realize from the merger in 2009. These estimates were based on assumptions and judgments, which are difficult to predict and which are subject to numerous risks and uncertainties. As a result, the combined company may not

realize all of these benefits within the timeframes or in the amounts expected. If the combined company does not realize the expected benefits from the merger within the timeframes or in the amounts expected, the trading price of Perini common stock may decline.

# Ronald N. Tutor s ownership interest in the combined company, along with his management position and his right to designate up to two nominees to serve as members of the Perini board of directors, will provide him with significant influence over corporate matters and may make a third party s acquisition of the combined company (or its stock or assets) more difficult.

Immediately following completion of the merger, two trusts controlled by Mr. Tutor will own approximately 43% of the outstanding shares of Perini common stock. In addition, Mr. Tutor will be the chairman and chief executive officer of the combined company and will have the right to designate up to two nominees for election as members of the Perini board of directors. (Mr. Tutor and his two designees would be 3 of 11 directors.) Although the Shareholders Agreement imposes significant limits on Mr. Tutor s right to vote the shares of Perini common stock that will be held by the Tutor Group, or to take specified actions that may facilitate an unsolicited acquisition of control of Perini by Mr. Tutor or his affiliates (all as described in The Shareholders Agreement beginning on page 86), Mr. Tutor will nonetheless still be able to exert significant influence over the outcome of a range of corporate matters, including significant corporate transactions requiring a shareholder vote, such as a merger or a sale of the combined company or its assets. This concentration of ownership and influence in management and board decision-making also could harm the price of Perini common stock following completion of the merger by, among other things, discouraging a potential acquirer from seeking to acquire shares of Perini common stock (whether by making a tender offer or otherwise) or otherwise attempting to obtain control of the combined company.

## The issuance of shares of Perini common stock in the merger will substantially reduce the percentage interests of current Perini shareholders in the earnings, voting power and market value of combined company.

Perini will issue approximately 23 million shares of Perini common stock in the merger. Upon completion of the merger and the issuance of these shares, based on the amount of shares of Perini common stock currently outstanding, Perini shareholders prior to the merger will own approximately 55% of the combined company s outstanding shares of common stock and the former Tutor-Saliba shareholders will own approximately 45% of the combined company s outstanding shares of common stock. The issuance of shares of Perini common stock in the merger will cause a significant reduction in the relative percentage interests of current Perini shareholders in earnings, voting power and market value of the combined company.

The merger is expected to be dilutive in 2008 (assuming completion in the third quarter of 2008) and accretive to Perini shareholders after 2008, and the accretive nature of the transaction is expected to result in increased earnings per share over time. The extent and duration of any dilution will depend on several factors, including the amount of merger-related expenses it incurs that are charged against its earnings, the time required to realize the benefits expected from the merger and the results of operations of Tutor-Saliba, which will not be known until after the merger is completed. If expenses charged against earnings are higher than the Special Committee expected, the benefits expected from the merger are slower to be realized than the Special Committee expected or Tutor-Saliba does not achieve the revenue and earnings growth projected by the Special Committee, the amount of dilution in 2008 could be greater than currently anticipated and the merger may not turn out to be accretive to current Perini shareholders (or may be less accretive than currently anticipated). In such event, the trading price of Perini common stock may decline.

After reflecting the pro forma adjustments as described in Unaudited Pro Forma Condensed Combined Financial Statements beginning on page 113, the pro forma diluted earnings per common share of the combined company for the three months ended March 31, 2008 is \$0.74 and the pro forma diluted earnings per common share of the combined company for the year ended December 31, 2007 is \$2.70 (excluding Tutor-Saliba s non-recurring gain on the sale of marketable securities). These unaudited pro forma condensed combined financial statements apply certain assumptions and adjustments which are based upon available information and assumptions that the managements of Perini and Tutor-Saliba believe reasonably reflect the merger.

### Perini will incur significant transaction, compliance, restructuring and other merger-related fees and costs.

Perini expects to incur costs associated with combining the operations of its business with those of Tutor-Saliba, as well as transaction fees and other costs related to the merger. The total cost to consummate the transaction is estimated to be approximately \$19.2 million. The amount of transaction costs expected to be incurred is a preliminary estimate and subject to change. The combined company also will incur restructuring and integration costs in connection with the merger. Based on the experience of other merged companies and initial management estimates, costs to integrate computer systems, facilities and insurance arrangements, as well as equipment sizing, employee relocation, retention and severance costs, will be approximately \$2 million over a period of 2 years. These merger-related and restructuring cost estimates depend on decisions to be made during integration. As with any merger, these estimates are based on preliminary assessment of integration activities and are likely to change. Additional unanticipated costs may be incurred in the integration of the businesses of the two companies. Depending on the nature of the restructuring activity, the cost may be included in the purchase price allocation to the extent known or as an expense in the period incurred. In addition, it is expected that the combined company s costs related to legal and regulatory compliance may increase substantially, at least in the near term, because Tutor-Saliba has not previously been required to comply with the reporting, internal control, public disclosure and similar legal and regulatory compliance obligations and requirements applicable to publicly traded companies. Although Perini expects that the elimination of duplicative costs, as well as the realization of other efficiencies related to the integration of the businesses, may offset incremental transaction, merger-related and restructuring costs over time, this net benefit may not be achieved in the near term or at all.

## The public resale by former Tutor-Saliba shareholders of Perini common stock received in the merger could have a negative effect on the trading price of Perini common stock following completion of the merger.

In the merger, we will issue a total of approximately 22.1 million shares of Perini common stock to two trusts controlled by Mr. Tutor and approximately 900,000 shares to the other shareholders of Tutor-Saliba. None of these shares will be registered under the Securities Act of 1933, and they will only be able to be resold pursuant to a separate registration statement or an applicable exemption from registration (under both federal and state securities laws). The shares will be subject to contractual restrictions under the terms of the Shareholders Agreement. Under those restrictions, none of those shares will be permitted to be resold for six months after completion of the merger (except with the consent of the Perini board of directors or in a registered offering). Thereafter, the trusts will be permitted to sell, in the aggregate, a maximum of approximately 6.6 million shares of Perini common stock through the fifth anniversary of the completion of the merger (unless the Perini board of directors allows otherwise). Mr. Tutor has indicated that he intends to cause the Tutor Group to sell a portion of the shares of Perini common stock received in the merger as soon as the initial six-month period lapses, subject to market conditions and the terms of the Shareholders Agreement. In addition, if we propose to register any of our shares in a registered public offering, the Tutor Group has a right to include its shares in such offering through a valid piggyback registration of shares, subject to the right of the underwriters of an offering to limit the number of shares included in such registration. Please see The Shareholders Agreement beginning on page 86 for a description of the terms of the Shareholders Agreement.

If all or a substantial portion of these shares of our common stock issued in the merger are resold into the public markets, such transactions may cause a decline in the trading price of our common stock.

#### The combined company will record goodwill that could become impaired and adversely affect its operating results.

Under United States generally accepted accounting principles (which we refer to in this proxy statement as GAAP), the merger will be accounted for under the purchase method of accounting as a purchase by Perini of Tutor-Saliba. Under the purchase method of accounting, the total implied purchase price paid by Perini in the merger will be allocated to Tutor-Saliba stangible assets and liabilities and identifiable intangible assets based on their fair values as of the date of completion of the merger. The excess of the purchase price over those fair values

will be recorded as goodwill. We expect that the merger will result in the creation of goodwill based upon the application of purchase accounting. The unaudited pro forma condensed combined financial statements contained in this proxy statement beginning on page 113 reflect an estimate of goodwill resulting from the merger amounting to \$746.4 million. As a result of the merger, total pro forma goodwill of \$773.7 million would represent 24% of the total combined pro forma assets of \$3.2 billion. Please see Note 1, Goodwill of Notes to Unaudited Pro Forma Condensed Combined Financial Statements beginning on page 120 for a discussion of the factors contributing to the estimated amount of goodwill resulting from the merger. To the extent the value of goodwill or intangibles becomes impaired, the combined company may be required to incur material charges relating to such impairment. Such a potential impairment charge could have a material and adverse impact on the combined company s operating results.

### If Tutor-Saliba issues notes to the former Tutor-Saliba shareholders in connection with the merger, Tutor-Saliba s excess cash flows will be dedicated to repaying those notes and will not be generally available to Perini until the notes are repaid.

In connection with the merger, Tutor-Saliba may issue aggregate dividends of up to \$120 million of cash and notes, subject to limitations contained in the Merger Agreement, in addition to the other distributions permitted by the Merger Agreement. These notes are being issued to the former Tutor-Saliba shareholders in essence to loan to the combined company, in order to support its surety requirements, funds that Tutor-Saliba would have otherwise distributed as dividends to its shareholders prior to the merger. Pursuant to the terms of the form of note attached to this proxy statement as *Annex D*, Tutor-Saliba is required to prepay the notes with 100% of its excess cash flow following the completion of the merger. Thus, until the notes are repaid, all excess cash flow generated by Tutor-Saliba following completion of the merger will be dedicated to repaying any notes issued by Tutor-Saliba in connection with the merger. Please see Pre-Closing Distribution of Property beginning on page 70 for a detailed description of the terms of the notes.

## The combined company will have continuing contractual obligations with Mr. Tutor, which may create conflicts of interest or may not be practical to enforce on Perini s behalf.

The Merger Agreement includes obligations of Perini and the former Tutor-Saliba shareholders, including Mr. Tutor, that will continue following completion of the merger. These obligations include indemnification obligations, which may entitle Perini to seek recovery from the former Tutor-Saliba shareholders for losses related to pre-merger actions or omissions of Tutor-Saliba. In addition, the Employment Agreement, the Shareholders Agreement and the notes to be issued by Tutor-Saliba also include obligations that will be in effect after the completion of the merger, including the restrictions on competitive activities, several of which may be impacted by the operating performance of Perini or Tutor-Saliba or the activities of Mr. Tutor.

In light of the important role Mr. Tutor is expected to serve for the combined company, it may be more difficult, impractical or unadvisable for Perini to enforce or assert defenses with respect to these contractual obligations against Mr. Tutor than against an unaffiliated third party, which may create a conflict of interest for Perini or Mr. Tutor. (Other former Tutor-Saliba shareholders are also expected to have continuing roles with the combined company, and a similar conflict of interest may arise, although their interests in the combined company will be significantly less than Mr. Tutor s.) If Perini determines that these contractual obligations should not be enforced even if there is a valid claim for enforcement or a valid defense to the enforcement of these obligations, Perini may not get the entire benefit for which it negotiated in these agreements, including recovery for certain losses related to Tutor-Saliba for which it otherwise would be entitled to indemnification. At the time the Special Committee and the Perini board of directors approved the merger, neither the Special Committee nor the Perini board of directors were aware of any material claims for which we would expect to exercise the remedies described in this risk factor.

## Perini, Perini s directors and Tutor-Saliba are named parties to a lawsuit relating to the merger that could delay or enjoin the completion of the merger and could cause Perini to incur substantial costs.

A class-action complaint on behalf of a putative Perini stockholder class against Perini, Perini s current directors and Tutor-Saliba relating to the merger is pending in Massachusetts state court. Depending on the

outcome, the action and any future similar actions could delay or enjoin the completion of the merger and could result in substantial costs to Perini. In addition, the cost to Perini of defending the action and any future similar actions, even if resolved in Perini s favor, could be substantial and could divert the attention of Perini management and its resources in general. See The Merger Litigation Related to the Merger beginning on page 72.

### Risks Relating to the Combination of Perini and Tutor-Saliba

#### The combined business will have a substantially increased backlog and may not fully realize the revenue value of such backlog.

As of March 31, 2008, our backlog of uncompleted construction work was approximately \$7.2 billion. On a pro forma basis, assuming completion of the merger, the backlog of the combined company would be approximately \$8.5 billion as of March 31, 2008, which is 18% higher than Perini s backlog on a stand-alone company basis.

We include a construction project in our backlog at such time as a contract is awarded or a letter of commitment is obtained and adequate construction funding is in place. The revenue projected in our backlog may not be realized or, if realized, may not result in profits. For example, if a project reflected in our backlog is terminated, suspended or reduced in scope, it would result in a reduction to our backlog which would reduce, potentially to a material extent, the revenues and profits we expected to receive from contracts in backlog. If a customer cancels a project, we may be reimbursed for certain costs and profit thereon, but typically we have no contractual right to the unearned revenues reflected in our backlog. Significant cancellations or delays of projects in our backlog could have a material and adverse effect on our revenues, cash flows and profits.

### The combined company will assume all of the risks associated with businesses that are now structured as joint ventures between Perini and Tutor-Saliba, including risks related to pending material litigation.

As of March 31, 2008, Tutor-Saliba was our partner in a total of 3 joint ventures for projects with a backlog of \$74.8 million. Upon completion of the merger, the combined company will own 100% of these projects (and of any additional joint ventures we may commence with Tutor-Saliba prior to completion of the merger). The result is that the combined company will have a greater share of both the potential benefits and the potential risks of such projects and will no longer enjoy the benefit of reduced financial and operational risks that accompany joint ventures with third parties. One of our joint ventures with Tutor-Saliba is involved in litigation with the Los Angeles County Metropolitan Transportation Authority, or LAMTA, regarding work done on various tunnel and station projects in the 1990s. We are a 40% minority partner in that joint venture and Tutor-Saliba is the 60% managing partner. Upon completion of the merger, the combined company will be exposed to 100% of the risk of any adverse outcome of that litigation.

### The growth prospects and future earnings of the combined company may be adversely affected, and the anticipated benefits of the merger may not be fully realized, if the combined company is unable to retain the services of Mr. Tutor.

A substantial benefit of the merger is expected to be the continued service of Mr. Tutor as the full-time chairman and chief executive officer of the combined company. Losing Mr. Tutor s services could adversely affect the business of the combined company until a suitable replacement could be found. In addition, a loss of Mr. Tutor s services in the near term is likely to have a negative impact on the combined company s ability to fully realize the anticipated benefits of the merger. Replacing Mr. Tutor quickly with an executive of equal experience and capabilities would likely be difficult. Although Mr. Tutor will be bound by the Employment Agreement following completion of the merger, as described in The Employment Agreement beginning on page 89, and his significant ownership interest in the combined company is likely to resign. In addition, we could lose the services of Mr. Tutor if he were to become disabled or otherwise physically unable to continue to work. Thus, despite the Employment Agreement, Mr. Tutor s continued employment with Perini cannot be assured.

Any reduction in financing for, or changes in regulation of, the hospitality and gaming industry, particularly in Las Vegas, Nevada, that reduces the number of large construction projects available to us could adversely affect the future financial results and growth prospects of the combined company.

Approximately 58% (or \$3.4 billion) of the combined company s pro forma revenues for the year ended December 31, 2007 were derived from construction projects in the hospitality and gaming industry in Las Vegas, Nevada. Any decline in this market or in the growth of this industry could materially and adversely affect the future financial results and growth prospects of the combined company.

# A significant portion of the combined company s operations will be concentrated in California and Nevada, and any adverse change to the economy or business environment in California or Nevada could adversely affect the combined company s future earnings and growth prospects.

Following completion of the merger, a significant portion of the combined company s operations will be concentrated in California and Nevada. As a result, the combined company will be susceptible to fluctuations caused by adverse economic or other conditions in this region, including as a result of natural or other disasters. If either of these states were to experience an economic slowdown or recession, private developers might curtail building construction activities as vacancies increase and state and local governments might reduce spending on building and civil projects due to revenue shortfalls. Thus, a stagnant or depressed economy in California or Nevada could adversely affect our business, results of operations and financial condition both immediately and over a number of years after any recovery in those markets.

An economic slowdown or an uncertain economic outlook may have a particularly adverse affect on consumer spending in the hospitality and gaming industry, as that spending is discretionary and may decline during or in anticipation of economic downturns, when consumers have or expect to have less disposable income. We expect that the combined company will derive a significant portion of its revenues from the construction, expansion and renovation of hospitality and gaming facilities in Las Vegas, Nevada, so that any actual or anticipated economic slowdown could result in the cancellation or delay of projects in Las Vegas, Nevada.

# If Black Construction s opportunity to win significant business from the expansion of the United States military s operations on the island of Guam does not develop as anticipated, the growth prospects, revenues and earnings of the combined company could be adversely affected.

A significant portion of the future revenues and growth prospects of the Black Construction, a subsidiary of Tutor-Saliba, over the next several years is expected to involve the construction of facilities for the expansion of the United States military s base on the island of Guam. This construction is dependent upon the continued implementation of the United States military s announced plan to relocate 8,000 marines and other military personnel from Okinawa, Japan to the island of Guam by 2014. The continued implementation of the United States military s plan, and the amount of work that Black Construction wins and performs in connection with the expansion of the United States military s base on the island of Guam, depends upon a number of factors, including:

competition from other construction companies operating on the island of Guam;

the political environment in the United States and Japan;

the financial and other terms agreed upon between the United States and Japan with respect to the relocation;

the United States military s and the Japanese government s availability of funds for the continued funding of the expansion and relocation in light of funding demands for other national priorities and commitments;

political, military and terrorist activities that affect the United States foreign policy;

### Edgar Filing: IDEX CORP /DE/ - Form 10-K

the ability of the combined company to invest sufficiently, and on favorable terms, in expanding Black Construction s capabilities on the island of Guam, including hiring and relocating necessary personnel, acquiring land (including for warehousing and barracks) and acquiring and relocating equipment; and

economic, political and other risks relating to business outside of the United States (despite the fact that the island of Guam is a United States territory), as discussed below in The combined business will have significant operations outside of the United States, and we expect these non-U.S. operations to increase. They will be subject to economic, political and other risks, which could adversely affect our revenues and earnings.

Any of these factors could result in a delay or cancellation of some or all of the anticipated work on the island of Guam, which would have an adverse effect on the growth prospects, future revenues and future earnings of the combined company.

### A decrease in government funding of infrastructure and other public projects could reduce the revenues of the combined company.

Approximately 7% (or \$409 million) of the combined company s pro forma revenues for the year ended December 31, 2007 were derived from construction projects involving civil construction contracts. Civil construction markets are dependent on the amount of infrastructure work funded by various governmental agencies which, in turn, depends on the condition of the existing infrastructure, the need for new or expanded infrastructure and federal, state or local government spending levels. A slowdown in economic activity in any of the markets that the combined company will serve may result in less spending on public works projects. In addition, a decrease or delay in government funding of infrastructure projects or delays in the implementation of voter-approved bond measures could decrease the number of civil construction projects available and limit our ability to obtain new contracts, which could reduce revenues within the civil construction segment of the combined company. In addition, budget shortfalls in California and other states in which both Perini and Tutor-Saliba are involved in significant infrastructure projects could curtail or delay the funding of future projects.

Perini s building construction segment also is involved in significant construction projects for public healthcare facilities, primarily in California, and public education facilities, primarily in Florida and California. These projects also are dependent upon funding by various federal, state and local governmental agencies. A decrease in government funding of public healthcare and education facilities, particularly in California and Florida, could decrease the number and/or size of construction projects available and limit our ability to obtain new contracts in these markets, which could further reduce the revenues and earnings of the combined company.

### The combined business will have significant operations outside of the United States, and we expect these non-U.S. operations to increase. They will be subject to economic, political and other risks, which could adversely affect our revenues and earnings.

We derived approximately 3% of our revenues and approximately 35% of our income from construction operations for the year ended December 31, 2007 from our work on projects located outside of the United States, including projects in Iraq and Afghanistan. We expect non-U.S. projects to represent a meaningful portion of the revenues and earnings of the combined business for the foreseeable future. This includes the anticipated future business activities of Tutor-Saliba s Black Construction subsidiary on the island of Guam, which is a United States territory that has its own local government and is subject to only certain United States laws. The non-U.S. operations of the combined business will expose it to risks inherent in doing business in regions outside the United States (and especially in hostile regions such as Iraq and Afghanistan), including:

political risks, including risks of loss due to civil disturbances, guerilla activities and insurrection;

acts of terrorism and acts of war;

unstable economic, financial and market conditions;

potential incompatibility with foreign subcontractors and vendors;

foreign currency controls and fluctuations;

trade restrictions;

variations in taxes; and

changes in labor conditions, labor strikes and difficulties in staffing and managing international operations. Any of these factors could harm our non-U.S. operations and, consequently, the business and operating results of the combined company. Specifically, failure to successfully manage risks associated with non-U.S. operations could result in higher operating costs than anticipated or could delay or limit our ability to generate revenues and income from construction operations in key markets outside of the United States.

## The combined company will include several new lines of businesses recently acquired by Tutor-Saliba. These acquisitions expose us to additional risks that, if realized, could adversely affect our future financial performance and operations.

In 2007 and 2008, Tutor-Saliba acquired two businesses Powerco Electric Corp., an electrical construction subcontractor, and Desert Plumbing & Heating Co., Inc., a plumbing and mechanical (including HVAC) subcontractor and certain material mining contracts and material stockpiles in Nevada (which are referred to in this proxy statement as the aggregates business). Although Tutor-Saliba has some experience managing and operating these types of businesses, we do not have any experience in these lines of business. Such new lines of business involve additional risks, such as those associated with entry into new markets, new operating activities, risks associated with integrating the operations of the acquired business into existing operations, managerial challenges and risks associated with marketing and delivering the goods and services provided by these new businesses.

There is no assurance that these new businesses will be operated successfully, will be integrated into the operations of Tutor-Saliba (and of the combined company) or will produce the financial and operating benefits that Tutor-Saliba expected, in making these acquisitions, or that we anticipated in negotiating the terms of the merger and valuing Tutor-Saliba. If the acquired businesses do not perform as expected, or if they are not successfully integrated and managed, the financial performance of the combined business may be adversely affected.

## Following the merger, we intend to continue to pursue acquisition opportunities, which may be difficult to integrate into our combined operations.

Following the merger, we intend to continue to pursue acquisitions as part of our growth strategy. The process of managing and integrating new acquisitions into our combined operations may result in unforeseen operating difficulties and may require significant financial, operational and managerial resources that would otherwise be available for the operation, development and expansion of the existing business of the combined company. To the extent that we misjudge our ability to integrate and properly manage acquisitions, we may have difficulty achieving our operating, strategic and financial objectives.

Acquisitions also may involve a number of special financial, business and operational risks, such as:

difficulties in integrating diverse corporate cultures and management styles;

additional or conflicting government regulation;

disparate company policies and practices;

client relationship issues;

diversion of our management s time, attention and resources;

### Edgar Filing: IDEX CORP /DE/ - Form 10-K

decreased utilization during the integration process;

loss of key existing or acquired personnel;

increased costs to improve or coordinate managerial, operational, financial and administrative systems;

dilutive issuances of equity securities, including convertible debt securities to finance acquisitions;

the assumption of legal liabilities;

amortization of acquired intangible assets; and

potential write-offs relating to the impairment of goodwill.

In addition to the integration challenges mentioned above, acquisitions of non-U.S. companies offer distinct integration challenges relating to non-U.S. GAAP financial reporting, foreign laws and governmental regulations, including tax and employee benefit laws, and other factors relating to operating in countries other than the United States, which are discussed above in the discussion regarding the difficulties the combined company may face operating outside of the United States.

## If Tutor-Saliba is unable to sustain its recent, significant rate of growth, the growth prospects and future results of the combined company are likely to be adversely affected.

Over the past two years, Tutor-Saliba has undergone substantial revenue and earnings growth. This growth has come from a combination of organic growth and the effects of recent acquisitions. These acquisitions also are expected to contribute significantly to the future performance and growth prospects of the Tutor-Saliba business. Because this growth has not occurred over a sustained period of time, and because it is expected to be partially dependent upon the performance of recent acquisitions, there is no assurance that Tutor-Saliba will be able to continue this rapid pace of growth in the future. Such growth also could be negatively affected by many factors, including future construction industry and capital market conditions, the effects of integration with the Perini business or failures to integrate and successfully manage recent acquisitions (as discussed in the risk factor The combined company will include several new lines of businesses recently acquired by Tutor-Saliba. These acquisitions expose us to additional risks that, if realized, could adversely affect our future financial performance and operations. ). If, following the merger, Tutor-Saliba s growth rate slows, or if it fails to grow at the pace anticipated by Perini, the growth prospects and future results of the combined company are likely to be adversely affected.

### THE MERGER

The Companies

Perini Corporation

Trifecta Acquisition LLC

73 Mt. Wayte Avenue

Framingham, Massachusetts 01701

### (508) 628-2000

Perini is a leading construction services company offering diversified general contracting, construction management and design-build services to private clients and public agencies throughout the world. We offer general contracting, preconstruction planning and comprehensive project management services, including the planning and scheduling of the manpower, equipment, materials and subcontractors required for a project. We also offer self-performed construction services, including site work, concrete forming and placement and steel erection. During 2007, we performed work on approximately 185 construction projects for over 100 federal, state and local government agencies or authorities and private customers.

Our business is conducted through three primary segments: building, civil and management services. Our building segment (comprised of Perini Building Company, James A. Cummings, Inc. and Rudolph and Sletten, Inc.) focuses on large, complex projects in the hospitality and gaming, sports and entertainment, educational, transportation, corrections, healthcare, biotech, pharmaceutical and high-tech markets in New York, Connecticut, New Jersey, Massachusetts, Florida, Washington, D.C., Arizona, Nevada and California. Our civil segment is comprised of Perini Civil Construction and Cherry Hill Construction, Inc. and focuses on public works construction primarily in the northeastern and mid-Atlantic United States, including the repair, replacement and reconstruction of public infrastructure such as highways, bridges, mass transit systems and wastewater treatment facilities. Our management services segment provides diversified construction, design-build and maintenance services to the United States military and government agencies as well as surety companies and multi-national corporations in the United States and overseas.

Perini was incorporated in 1918 as a successor to businesses that had been engaged in providing construction services since 1894. Our headquarters are in Framingham, Massachusetts, and we have twelve other principal offices throughout the United States.

Shares of Perini common stock are traded on the NYSE under the symbol PCR.

For additional information about Perini and its business, please see Incorporation by Reference beginning on page 152 and Where You Can Find Additional Information beginning on page 152.

Trifecta Acquisition LLC is a California limited liability company and a wholly owned subsidiary of Perini. Trifecta Acquisition LLC was formed solely for the purpose of entering into the Merger Agreement with Tutor-Saliba and completing the merger. It has not conducted any business operations other than those contemplated by the Merger Agreement.

Tutor-Saliba Corporation

15901 Olden Street

Sylmar, California 91342

(818) 362-8391

Tutor-Saliba is a leading civil infrastructure and commercial building construction company that focuses on large, complex projects, usually ranging from \$100 million to \$1 billion or more in size. Tutor-Saliba manages all aspects of these projects, including design-build, design-bid-build and pre-construction services for its customers, the project owners.

Tutor-Saliba operates in three segments: domestic building, domestic civil and international. Tutor-Saliba s domestic building operations are predominately in Nevada and California where it maintains large offices. Tutor-Saliba s domestic civil operations have been historically focused primarily in California and New York. Tutor-Saliba s international operations are conducted primarily on the island of Guam and in the Philippines.

Tutor-Saliba s domestic building operations focus on large, complex buildings in the gaming and hospitality, sports and entertainment, transportation, education and healthcare markets, as well as for governmental agencies. Tutor-Saliba s domestic civil operations focus on large, complex public infrastructure construction, including highways, bridges, airports, wastewater treatment facilities and mass transit systems. Tutor-Saliba s primary customers are federal and state government agencies and local municipalities. Tutor-Saliba s international operations are conducted through its subsidiary, Black Construction Corporation, which has been in operation for over 50 years and is the largest contractor on the island of Guam. Black Construction also operates in the Philippines and Micronesia. This segment focuses on both civil and building projects, serving the United States military, local governments and private companies.

Tutor-Saliba is a privately held corporation. Its principal shareholders are two trusts controlled by Mr. Tutor.

For additional information about Tutor-Saliba and its business, please see Information about Tutor-Saliba beginning on page 92 and Where You Can Find Additional Information beginning on page 152.

### **The Merger Proposals**

At the annual meeting, among other matters, the holders of Perini common stock will be asked to consider and vote on the following proposals related to the merger:

*Proposal 1*: A proposal to approve the issuance of 22,987,293 shares of Perini common stock in the merger contemplated by the Agreement and Plan of Merger, dated as of April 2, 2008, by and among Perini, Trifecta Acquisition LLC, Tutor-Saliba, Mr. Tutor and shareholders of Tutor-Saliba, as amended by Amendment No. 1 thereto, dated as of May 28, 2008; and

*Proposal 2*: A proposal to amend our amended and restated articles of organization to increase the number of authorized shares of Perini common stock from 40 million to 75 million shares.

### THE MERGER WILL NOT BE COMPLETED UNLESS, AMONG OTHER THINGS, THE PERINI SHAREHOLDERS APPROVE BOTH (1) THE SHARE ISSUANCE PROPOSAL AND (2) THE ARTICLES AMENDMENT PROPOSAL. UNDER THE MERGER AGREEMENT, COMPLETION OF THE MERGER IS SUBJECT TO THE SATISFACTION (OR, IF LEGALLY PERMITTED, WAIVER) OF SPECIFIED CLOSING CONDITIONS. APPROVAL BY THE PERINI SHAREHOLDERS OF THE MERGER PROPOSALS IS SUCH A CONDITION, WHICH MAY NOT BE WAIVED BY EITHER PERINI OR TUTOR-SALIBA.

The share issuance proposal is intended to satisfy the listing requirements of the NYSE.

The articles amendment proposal is intended to satisfy the requirements of Massachusetts law and the Perini amended and restated articles of organization for shareholder approval of amendments to the amended and restated articles of organization.

### **Background of the Merger**

Perini and Tutor-Saliba have worked closely together since approximately 1977. The two companies have worked as joint venture partners on over 35 projects with aggregate project value of over \$2.9 billion since that time. This includes recent joint ventures such as the Tappan Zee Bridge Repair, Brooklyn Queens Expressway Rehabilitation and the Airtran Terminal at Jamaica Station.

Tutor-Saliba has been an investor in Perini for much of the past 17 years, over which period Ronald N. Tutor has been the chairman and chief executive officer of Tutor-Saliba. Mr. Tutor (through trusts controlled by him) is the principal shareholder of Tutor-Saliba, owning approximately 96% of the outstanding stock of Tutor-Saliba.

In January 1997, a group led by Blum Capital Partners, which included Tutor-Saliba as a limited partner, invested approximately \$30 million in Perini to help Perini avoid bankruptcy. In connection with that investment, Mr. Tutor became a director of Perini and its chief operating officer, pursuant to a management agreement between Perini and Tutor-Saliba. Michael R. Klein, another investor in that group, also joined the Perini board of directors.

In 2000, Perini obtained additional \$40 million in equity capital from a group led by Tutor-Saliba, American International Group Inc, or AIG, and O&G Industries, in which Tutor-Saliba again participated as an investor, and, in connection with this investment, Mr. Tutor became Chairman of the Perini board of directors. Tutor-Saliba subsequently acquired the shares acquired by AIG directly from AIG.

In March 2000, Mr. Tutor became Perini s chief executive officer. He continues to serve Perini as both its chairman and chief executive officer, positions he has also continuously held at Tutor-Saliba. Mr. Klein was elected vice chairman of the Perini board of directors in September 2000, and its Lead Outside Director in 2004.

From time to time since 1997, Mr. Klein and other members of the Perini board of directors have raised with Mr. Tutor the possibility of combining Perini with Tutor-Saliba. None of these discussions resulted in any substantial negotiations because Mr. Tutor routinely indicated that he preferred to keep Tutor-Saliba a separate, privately owned company.

On November 13, 2007, at a meeting of the Compensation Committee of the Perini board of directors, Mr. Tutor explained, as reflected in public filings made by Tutor-Saliba and Mr. Tutor, that Tutor-Saliba had been selling its shares of Perini common stock. He said that Tutor-Saliba intended to sell the remainder of its shares of Perini common stock in order to separate the two businesses because he was contemplating taking Tutor-Saliba public in the Spring of 2008 through an initial public offering. He went on to state that, if he proceeded with this initial public offering, he would resign his executive roles with Perini. As reflected in public filings made by Tutor-Saliba and Mr. Tutor, following this meeting, Tutor-Saliba continued to sell the remainder of its shares of Perini common stock until, in December 2007, it ceased to directly or indirectly own any shares of Perini common stock. (Mr. Tutor continued to own (and still owns) restricted stock units covering 150,000 shares of Perini common stock, which shares vested by their terms on June 30, 2008.)

Following this disclosure, the non-management directors of Perini who were present at the meeting of the Compensation Committee (Mr. Klein, Peter Arkley, Willard W. Brittain, Jr., Robert A. Kennedy and Raymond R. Oneglia) convened separately to discuss the implications for Perini of Mr. Tutor s statements. They determined to begin to explore alternatives on behalf of Perini, including, whether to again propose that Mr. Tutor consider, as an alternative to an initial public offering of Tutor-Saliba, combining Perini and Tutor-Saliba, on terms that were acceptable to Perini. These directors discussed the possibility that a new chief executive officer might need to be identified if a business combination transaction between Perini and Tutor-Saliba could not be negotiated. They decided to begin to search for potential candidates to prepare for that possibility. These directors also discussed the possibility that unsolicited proposals for the acquisition of Perini might be presented to Perini, including those that might arise either once a business combination with Tutor-Saliba was publicly announced or after the merger was completed. They also discussed potential bidders that they thought were most likely to be interested in Perini. However, they did not reach any conclusions regarding how Perini should react to any such proposals or whether interest in any such proposals should be solicited. Recognizing that any business combination transaction with Tutor-Saliba would present a conflict of interests for Mr. Tutor, the directors asked Mr. Klein to inform the other non-management members of the Perini board of directors of these recent developments and then to organize a committee of independent and disinterested directors to be responsible for Perini s interests. They also discussed several administrative matters relating to forming such a committee.

Over the next several weeks, Mr. Klein discussed these matters with the other non-management members of the Perini board, who generally indicated that a business combination transaction with Tutor-Saliba would merit consideration, if Mr. Tutor was interested in considering that as an alternative to taking Tutor-Saliba public. These discussions related primarily to ministerial and administrative matters related to organizing an independent committee of the Perini board of directors to consider this kind of a transaction. They also agreed that the group of directors that had informally convened to discuss the transaction on November 13, 2007 (other than Mr. Oneglia) should likely serve on such an independent committee, subject to an outside counsel assessment of their independence and disinterestedness. Mr. Klein also discussed with those directors the process for retaining and negotiating arrangements with potential legal and financial advisors.

In late November and early December 2007, Mr. Klein approached potential legal and financial advisors to assist a board committee that would be formed if Mr. Tutor were to be interested in discussing a business combination transaction between Perini and Tutor-Saliba. Mr. Klein contacted Kirkland & Ellis LLP to serve as the committee s legal counsel and UBS Securities LLC (referred to in this proxy statement as UBS) to serve as the committee s financial advisor, with the retention of each to be formally determined if and when a special committee was formed.

Mr. Klein arranged and had a dinner meeting with Mr. Tutor on the evening of December 11, 2007, in New York City to assess his receptivity to a business combination between Tutor-Saliba and Perini as an alternative to an initial public offering of Tutor-Saliba because he believed it was in Perini s best interests to pursue negotiations with Mr. Tutor to determine whether a business combination transaction with Tutor-Saliba could be entered into on terms that were acceptable to Perini. At that dinner, Mr. Klein discussed with Mr. Tutor reasons why he believed that a business combination transaction between the two companies would be an attractive alternative for Mr. Tutor. Mr. Klein suggested to Mr. Tutor that both an initial public offering of Tutor-Saliba and a business combination transaction between Tutor-Saliba and Perini would achieve the same economic goal for Tutor-Saliba shareholders permitting them to monetize their interests in Tutor-Saliba at public market values. However, he suggested to Mr. Tutor that a business combination with Perini would be a faster, less expensive and less risky alternative than an initial public offering of Tutor-Saliba to successfully create value for the Tutor-Saliba shareholders. He also noted that the benefits expected to be realized from a combination of the businesses of Tutor-Saliba and Perini would not be present in an initial public offering. Mr. Klein indicated that he thought that there was interest on the part of the Perini board of directors in entering into discussions regarding such a business combination transaction. Mr. Tutor responded that he was willing to consider such a business combination transaction along with his continuing pursuit of an initial public offering for Tutor-Saliba, although he did not express any conclusions regarding the likelihood of entering into the business combination transaction. He indicated that he expected that Tutor-Saliba would be prepared to file its registration statement for its initial public offering within a few months and that prospective advisors had indicated that the initial public offering would be well received by the market. Mr. Klein advised Mr. Tutor that, given his relationships with both entities, any negotiations of a transaction should adhere to stringent procedures appropriate to the negotiation of transactions involving conflicts of interest. He noted that such procedures would include the formation of an independent committee to negotiate the transaction on behalf of Perini, which could hire independent legal and financial advisors, restrictions on Mr. Tutor s communications with the other members of the Perini board of directors and management regarding the transaction, restrictions on Mr. Tutor s access to any information relevant to Perini s negotiation of the transaction by the independent committee and abstentions by Mr. Tutor from any actions that might affect the negotiations. Mr. Tutor stated that he understood and agreed to observe these procedures. Mr. Klein then reported the results of this dinner conversation to the other members of the Perini board of directors with whom he discussed the proposed transaction on November 13, 2007, and they generally agreed to proceed to organize a special committee of independent and disinterested directors to explore this business combination transaction on behalf of Perini, with the assistance of outside independent legal and financial advisors.

On January 3, 2008, Mr. Tutor sent a letter to Mr. Klein and the other members of the Perini board of directors in which he outlined the terms on which he would be interested in a business combination transaction between Tutor-Saliba and Perini. The letter stated that he would expect Perini to be the surviving public company in the

transaction and that he would continue as the chairman of the Perini board of directors and chief executive officer of Perini after the closing of the transaction. Mr. Tutor proposed that, after certain distributions of non-core Tutor-Saliba assets prior to the closing of the transaction, based on his assessment of the value of Tutor-Saliba, the shareholders of Tutor-Saliba should own approximately 55% of the pro forma outstanding shares of the combined company, although he suggested that he might be willing to structure that equity interest to include 49% of the pro forma outstanding shares of common stock of the combined company, plus non-voting preferred stock of Perini with a value of \$250 million. Mr. Tutor suggested a meeting to discuss the proposed transaction on or about January 15, 2008. Shortly after receiving this letter, to which he did not respond, Mr. Klein received from Tutor-Saliba a copy of marketing materials prepared by advisors to Tutor-Saliba containing, among other information, various preliminary valuation indications of Tutor-Saliba in connection with a potential initial public offering of, and certain assumed strategic investments in, Tutor-Saliba. These materials had been prepared for the sole purpose of being delivered to several prospective underwriters in connection with the initial public offering. Mr. Klein subsequently conveyed these materials to other members of the Perini board of directors.

On January 7, 2008, the Perini board of directors held a meeting at Perini s headquarters in Framingham, Massachusetts and by teleconference. All of the directors (including Mr. Tutor) were in attendance in person or by telephone. At the meeting, the Perini board of directors created the Special Committee to explore and evaluate potential strategic transactions that might be available to Perini, including in particular a business combination transaction with Tutor-Saliba, and to discuss and negotiate the terms of any such transactions. The board discussed the need to ensure that all members of the Special Committee were both independent and disinterested. Mr. Klein reported that he had received advice and assistance assessing the qualifications of the Perini directors to serve as members of the Special Committee from Kirkland & Ellis. After discussion of such qualifications, the board decided that the Special Committee would be comprised of four independent, disinterested directors who had begun to discuss the proposed transaction with Tutor-Saliba on November 13, 2007 Messrs. Arkley, Brittain, Kennedy and Klein. The board of directors directed the Special Committee to report back to the full board with its recommendations and conclusions at such time as they were prepared to either recommend a specific transaction or advise the full board that, in their judgment, no strategic transaction available to Perini should be pursued. Mr. Klein was elected the initial Chairman of the Special Committee, subject to further confirmation by the Special Committee.

Following the meeting, Mr. Klein scheduled a meeting with Mr. Tutor and other representatives of Tutor-Saliba for January 17, 2008 in Los Angeles, California.

Prior to a meeting of the Special Committee on January 16, 2008, Mr. Klein and UBS reached tentative agreement on the financial terms of UBS engagement by the Special Committee.

On January 16, 2008, the Special Committee held a meeting in Los Angeles, California, together with its financial and legal advisors. The Special Committee discussed with these advisors the dynamics defining the possibility of negotiating a transaction with Tutor-Saliba on terms that might be acceptable to both parties, in preparation for the meeting scheduled for the following day with representatives of Tutor-Saliba. UBS presented its preliminary financial analyses of Tutor-Saliba and Perini and a general overview of the financial market for engineering and construction companies. The Special Committee and its advisors discussed the degree to which an impediment to the ability of Perini to offer a transaction that would be attractive to Mr. Tutor and acceptable to Perini, in that the shares of Perini common stock were trading at prices well below historic levels for Perini and multiples of other public companies in the engineering and construction industry and discussed events that might explain that to Mr. Tutor. As they had previously discussed on November 13, 2007, the Special Committee also discussed the possibility of business combinations with other participants in the industry, including proposals that might arise in connection with the pursuit or announcement of a transaction with Tutor-Saliba, and the pursuit of alternative leadership in the event Mr. Tutor resigned from Perini to pursue an initial public offering of Tutor-Saliba. After this discussion, the Special Committee decided not to initiate exploration of alternative transaction swith other participants in the industry until it had first explored more completely a combination with Tutor-Saliba or received a proposal for an alternative transaction on an unsolicited basis. The

Special Committee reached this decision based on several factors, including (i) its belief that it should proceed with the consideration and negotiation of a transaction with Tutor-Saliba before it determined if soliciting interest in proposals to acquire Perini or enter into an alternative business combination transaction involving Perini was necessary, appropriate or in the best interests of Perini and its shareholders, (ii) the current trends and competitive developments in the industry, including the overall decline in market value of public companies in the industry, which created unfavorable conditions in which to seek to sell Perini, (iii) the fact that the transaction with Tutor-Saliba appeared to be, based on the Special Committee s knowledge of the business, market and potential acquirors of Perini, the most likely transaction that would increase shareholder value, (iv) the risk that Perini might lose the current opportunity to pursue a business combination transaction with Tutor-Saliba if Tutor-Saliba became aware of the initiation of efforts to identify and solicit alternative transactions, (v) the Special Committee s belief that, in any event, any final transaction agreement with Tutor-Saliba would include provisions that would allow the Special Committee and the Perini board of directors to pursue a superior offer if it were to materialize, and (vi) the Special Committee s belief that completion of the merger would not preclude the pursuit of any potential strategic alternatives for Perini in the future (other than remaining as a stand-alone company without the acquisition of the Tutor-Saliba business) and, in fact, might enhance the availability and/or attractiveness of such alternatives. However, the Special Committee agreed that it could revisit this issue in the future. The Special Committee ultimately did not opt to pursue other strategic alternatives during its consideration of the merger, primarily because it determined that the merger was in the best interests of the Perini shareholders, the Merger Agreement included the expected provisions that would allow the Special Committee and the Perini board of directors to pursue an alternative transaction proposal reasonably likely to lead to a superior proposal if one materialized (or change the recommendation of the Perini board of directors to the Perini shareholders pursuant to its fiduciary obligations), the Special Committee determined that soliciting interest in alternative transactions was not necessary or appropriate in light of the nature of the merger in which Perini shareholders would continue to hold their interests in Perini and no unsolicited offers for alternative transactions had been made known to the Special Committee during the period of consideration of the Merger Agreement.

On January 17, 2008, the Special Committee held a meeting in Los Angeles, California, together with its legal advisors. At the meeting, the Special Committee discussed several organizational matters and basic procedures to be followed by the Special Committee. Topics expected to be discussed with representatives of Tutor-Saliba also were reviewed. The Special Committee confirmed that Mr. Klein would serve as the Chairman of the Special Committee and that the Special Committee would formally retain UBS as the Special Committee s financial advisor (subject to the negotiation of an appropriate engagement letter) and Kirkland & Ellis as the Special Committee s legal counsel. The representatives of Kirkland & Ellis discussed several legal considerations with the Special Committee, including the fiduciary obligations of Perini directors with respect to any business combination with Tutor-Saliba and how the terms of any business combination agreement with Tutor-Saliba could affect those obligations.

Later in the day on January 17, 2008, the Special Committee and its financial and legal advisors met at UBS offices in Century City, California with representatives of Tutor-Saliba and its advisors, including Deutsche Bank Securities Inc. (referred to in this proxy statement as Deutsche Bank), Tutor-Saliba s financial advisor, and Tutor-Saliba s outside legal counsel, Latham & Watkins LLP and Nomi Castle. In attendance on behalf of Tutor-Saliba were Mr. Tutor, James A. Frost, executive vice president and chief operations officer of Tutor-Saliba, and William B. Sparks, senior vice president and chief financial officer of Tutor-Saliba. Early in the discussions that ensued at the meeting, Mr. Tutor agreed that, in light of the dialogue between the parties concerning the then-current market price of Perini common stock, that any transaction could be negotiated on the basis of the implied relative values of the two companies, rather than being tied to the fluctuating market value of Perini common stock. This agreement in principle among the members of the Special Committee and Mr. Tutor reflected a consensus among them that Perini common stock was then under-valued, so that any transaction in which Tutor-Saliba shareholders would receive an interest in Perini based on the current trading prices for Perini common stock would necessarily undervalue Perini, and that if the terms of a transaction were appropriately negotiated, the benefits ultimately would be reflected in a rise in the market value of Perini over the long term. Accordingly, the negotiations thereafter proceeded toward finding terms of a business combination that reflected

the relative implied values of the two companies (one to the other), based on both their present and projected performance, rather than looking to the fluctuating market price of Perini common stock. As a result, Tutor-Saliba agreed to make financial and legal due diligence materials regarding Tutor-Saliba available to the Special Committee and its advisors and that the parties could reconvene to discuss possible financial terms of a potential transaction once the Special Committee was able to review these materials and discuss the results of its work. A representative of Kirkland & Ellis also reminded the parties about the importance of adhering to the Special Committee s guidelines for the negotiation of a transaction.

Tutor-Saliba made financial and legal due diligence materials regarding Tutor-Saliba available to the Special Committee and its advisors, beginning on January 19, 2008 and continuing until shortly before the execution of the Merger Agreement. This included providing access to an online data room, making Tutor-Saliba s management and advisors available for various meetings and providing supplemental information at the request of the Special Committee and its advisors.

Over the next several days, representatives of Kirkland & Ellis and Latham & Watkins discussed appropriate procedures for the sharing of information between the companies and negotiated a confidentiality and standstill agreement that both companies signed on January 25, 2008.

On January 23, 2008, representatives of UBS met with management of Tutor-Saliba, including Messrs. Tutor, Frost and Sparks, and representatives of Deutsche Bank at Tutor-Saliba s headquarters near Los Angeles, California. At the meeting, the parties discussed certain projected financial information of Tutor-Saliba.

On January 25, 2008, management of Perini, including Robert Band, president and chief operating officer of Perini, and Kenneth R. Burk, senior vice president and chief financial officer of Perini, and representatives of UBS met with management of Tutor-Saliba, including Messrs. Tutor, Frost and Sparks, and representatives of Deutsche Bank at Tutor-Saliba s headquarters near Los Angeles, California. At the meeting, the parties discussed certain alternatives regarding post-closing operations of the companies (which did not include any discussion of post-closing management of the combined company).

On February 7, 2008, the Special Committee held a meeting in Washington, D.C., together with its financial and legal advisors. Representatives of UBS provided an update to the Special Committee on the work UBS had done since the January 17, 2008 meetings with Tutor-Saliba, including its preliminary financial analyses of Tutor-Saliba, Perini and the proposed transaction. The Special Committee and its advisors discussed the preliminary analyses and several open items in them. The Special Committee reviewed and discussed the projections of financial performance for each of Perini and Tutor-Saliba proposed by each company, respectively, and the comments on the reasonableness of the projections provided by each company s management of the other company s projections. (Mr. Tutor did not participate in the preparation of Perini s base case projections). In addition, in a series of separate calls, the Special Committee discussed these items with management of Tutor-Saliba (including Messrs. Tutor and Sparks), management of Perini (including Messrs. Tutor, Band and Burk), and other members of the Perini board of directors. Following these calls, the Special Committee further discussed the various projections with UBS and provided further input on its views of the projections in an effort to create a set of projections for each company and an estimate of synergies for the combined company that would reflect the best currently available estimates and judgments of the Special Committee, taking into account its independent views, input from the two companies senior management and input from the other directors of Perini, of the future financial performance of the companies and the estimated synergies for the combined company. In addition, the Special Committee discussed the potential impact on Perini and its shareholders of the discontinuation of Mr. Tutor s leadership of Perini (which they expected would occur if no business combination were negotiated and Mr. Tutor was able to complete an initial public offering of Tutor-Saliba). They concluded that his absence could have significant adverse effects on Perini s future success on a stand-alone basis, particularly on Perini s civil segment. Accordingly, the Special Committee adjusted the projected financial performance of Perini as a stand-alone company operating without Mr. Tutor s involvement to reflect this view. Finally, the Special Committee also discussed preliminarily terms and conditions for the retention of Mr. Tutor as the chief executive officer and substantial shareholder of the combined company for an extended period after any business combination.

Following the February 7, 2008 meeting of the Special Committee, Mr. Klein contacted Mr. Tutor and indicated that he expected to be able to meet in about a week to discuss the possible financial terms of a business combination.

Shortly after the February 7, 2008 meeting of the Special Committee, Perini and the Special Committee executed an engagement letter with UBS that had been approved by the Special Committee at the meeting.

Between February 8 and February 10, 2008, the Special Committee continued to review, discuss and refine the projections by and of the two companies separately and combined, and the estimated synergies for a combined company.

On February 12, 2008, the Special Committee met by teleconference, together with its financial and legal advisors. Representatives of UBS provided an update to the Special Committee on its work since the previous meeting of the Special Committee and presented a revised preliminary financial analyses of Tutor-Saliba, Perini and business combination transaction which reflected guidance provided by the Special Committee at and since the previous meeting of the Special Committee. The Special Committee discussed the relative implied values of Perini and Tutor-Saliba, as reflected in the preliminary financial analyses presented by UBS. Based on the relative implied values of Perini and Tutor-Saliba reflected in the analyses presented to the Special Committee, there was general agreement among the Special Committee members that Mr. Tutor s proposal that Perini shareholders should own less than a majority of the total shares of the combined company was not acceptable. Based on the relevant information presented and discussed, the Special Committee discussed various approaches to the amount of relative ownership of the combined company by Perini and Tutor-Saliba shareholders. It sought an approach that would be beneficial to Perini shareholders, sufficiently attractive to Mr. Tutor to cause him to want to continue pursuing the proposed transaction and consistent with the relative implied valuations of Tutor-Saliba and Perini (taking into account the estimated synergies). In light of these discussions, the Special Committee generally agreed, on a tentative basis, that they would propose a business combination in which the Tutor-Saliba shareholders would receive a percentage of the fully-diluted shares of the combined company in the mid-40 s as a basis for further discussion of the transaction with Mr. Tutor. The Special Committee also reviewed compensation packages for chief executive officers at peer firms, with a view to developing an appropriate employment agreement with Mr. Tutor in the event that a transaction were to be completed. Representatives of Kirkland & Ellis discussed with the Special Committee certain non-financial terms for which the Special Committee should consider negotiating as a material part of the negotiation of the material terms of the entire transaction, including restrictions on Mr. Tutor with respect to his ownership, voting and sale of shares of Perini common stock received in any transaction. The Special Committee discussed various negotiation strategies and ranges of acceptable values and authorized Mr. Klein to meet with Mr. Tutor to proceed to negotiate the material terms of a possible transaction.

Mr. Klein subsequently asked Mr. Arkley to join him at the meeting with Mr. Tutor.

On February 14, 2008, Mr. Klein, accompanied at times by Mr. Arkley, met with Mr. Tutor in Century City, California. During the course of this meeting, representatives of UBS and Kirkland & Ellis were available to advise the Special Committee members and representatives of Deutsche Bank and Latham & Watkins were available to advise the representatives of Tutor-Saliba. Following the decision regarding the combined company ownership ratios proposed by the Special Committee, Mr. Klein initiated the negotiation by proposing to Mr. Tutor a transaction in the form of an all-stock merger in which shareholders of Tutor-Saliba would receive shares of Perini common stock equal to 42% of the fully diluted shares of the combined company in exchange for all shares of Tutor-Saliba, after the distribution of certain Tutor-Saliba assets. Mr. Klein suggested that the Special Committee believed that Tutor-Saliba would be viewed as contributing approximately 42% of the implied value of the combined company in a merger, taking into account estimated synergies that were expected to be realized as a result of the transaction, so that Tutor-Saliba shareholders should, in the merger, receive approximately that percentage of the shares of common stock of Perini that would be outstanding at that time. Mr. Tutor responded that he felt that percentage failed adequately to reflect the immediate and expected longer term contributions of Tutor-Saliba to the combined enterprise and requested, obtained and then responded further

to a presentation of the rationale for the proposed exchange ratio. After a series of proposals and counter-proposals, and after consultations with their respective outside advisors, Mr. Klein and Mr. Tutor reached a tentative agreement on the principal terms of a merger transaction, subject to further due diligence, confirmation of valuations, the negotiation of definitive agreements and the receipt of necessary approvals, including that of the Special Committee and the full Perini board of directors. The parties tentatively agreed that:

Perini would acquire Tutor-Saliba (excluding cash and certain non-core physical assets of Tutor-Saliba which Tutor-Saliba would distribute to its shareholders before the transaction) in a stock-for-stock transaction structured as a merger;

the shareholders of Tutor-Saliba would receive shares of Perini common stock as merger consideration equal to approximately 45% of the pro forma outstanding shares of the combined company, subject to the Special Committee receiving an opinion from its financial advisor that the corresponding exchange ratio was fair from a financial point of view to Perini;

the merger agreement would include a fiduciary out to permit the Perini board of directors to change its recommendation to Perini shareholders and/or terminate the merger agreement to accept a superior proposal;

Mr. Tutor would sign a 5-year employment agreement to serve as the chief executive officer of the combined company and would get certain board representation rights; and

Mr. Tutor and the other shareholders of Tutor-Saliba who would receive shares of Perini common stock in the merger would agree to be bound by transfer, voting, and standstill restrictions with respect to those shares in a shareholders agreement to be negotiated separately.

The parties also discussed whether the agreements should contemplate that Perini would be reincorporated in Delaware or Nevada and move its headquarters to Las Vegas, Nevada.

Following this meeting, Mr. Klein instructed Kirkland & Ellis to prepare draft agreements to reflect the proposed transaction discussed at the February 14 meeting.

On February 22, 2008, Kirkland & Ellis delivered a draft of a merger agreement, a shareholders agreement and an employment agreement for Mr. Tutor to Tutor-Saliba and its advisors. Following the delivery of these agreements, Mr. Tutor contacted Mr. Klein and raised concerns about various terms the parties had identified upon review of these draft agreements, including whether the post-closing indemnification of Perini would be secured by certain of the shares of Perini common stock to be issued in the merger, the right of Perini to respond to unsolicited alternative takeover proposals for Perini, the transfer restrictions imposed with respect to the shares of Perini common stock to be received by the Tutor-Saliba shareholders in the merger and unresolved compensation amounts in the employment agreement with Mr. Tutor. Messrs. Klein and Tutor also discussed a timetable for the negotiation of the agreements.

On February 28, 2008, representatives of Kirkland & Ellis held a teleconference with management of Tutor-Saliba, including Messrs. Tutor and Sparks, and representatives of Latham & Watkins to discuss the draft agreements delivered on behalf of the Special Committee. The parties discussed, among other things, the deal structure, various terms of the employment agreement with Mr. Tutor, the scope of the right of the Perini board of directors to change its recommendation to Perini shareholders or terminate the merger agreement, the amount of a termination fee and expense reimbursement that Perini would be required to pay to Tutor-Saliba following the termination of the merger agreement in certain circumstances, whether the post-closing indemnification of Perini would be secured by certain of the shares of Perini common stock to be issued in the merger, the terms on which Tutor-Saliba would be permitted to make pre-closing dividends of various non-core assets and available cash balances and the terms of the transfer, standstill and voting restrictions in the proposed shareholders agreement.

Over the course of the next several weeks, representatives of Kirkland & Ellis and Latham & Watkins continued to exchange drafts of the transaction agreements and engaged in several teleconferences to negotiate the terms of the agreements at the direction and on behalf of their respective clients. In addition, Messrs. Klein

### Edgar Filing: IDEX CORP /DE/ - Form 10-K

and Tutor directly negotiated various issues that arose regarding terms of these agreements. The parties continued to discuss many of the same items discussed on the February 28, 2008 teleconference, including the transaction structure (as to which the parties discussed several different options to effect the transaction in a tax-efficient manner and also discussed the possible reincorporation of the combined company in Delaware), the term of the transfer, standstill and voting restrictions in the shareholders agreement (which would extend for longer periods than initially agreed if the Tutor Group continued to hold a significant portion of the shares), the limitations on the indemnification obligations of the parties and the timing for Perini s receipt of audited financial statements of Tutor-Saliba. Throughout the time of these continuing negotiations, the Special Committee, with the assistance of its advisors, continued to collect and assess additional information regarding Tutor-Saliba, Perini and the proposed transaction and to review and refine the projections of the future financial performance of the companies based on information from the management of Tutor-Saliba and Perini and additional input from the Special Committee members.

On March 18, 2008, the Special Committee held a meeting at Perini s headquarters in Framingham, Massachusetts, together with its financial and legal advisors. Prior to the meeting, drafts of the Merger Agreement, Shareholders Agreement and Employment Agreement were circulated to the Special Committee members. At the meeting, Mr. Klein and representatives of Kirkland & Ellis provided an update to the other members of the Special Committee on the status of the negotiations to date. Kirkland & Ellis also reviewed the drafts of the agreements provided to the Special Committee. The Special Committee provided further guidance to the representatives of Kirkland & Ellis on proposed revisions to the terms of the agreements. In addition, the Special Committee discussed the terms on which Tutor-Saliba would be permitted to make the contemplated pre-closing dividends, including the expected amount of cash Tutor-Saliba would distribute to its owners by dividend prior to the closing and the anticipated impact of these dividends on the combined company s surety bonding capacity. Representatives of Kirkland & Ellis again reviewed with the Special Committee on the work UBS had done since the previous meeting of the Special Committee and presented its preliminary financial analyses of Tutor-Saliba, Perini and of the proposed transaction in light of the economic terms reflected in the current drafts of the transaction agreements. The Special Committee determined that it would update the Perini board of directors regarding the status of the proposed transaction at a regularly scheduled meeting of the Perini board of directors to take place on March 19, 2008.

Following this meeting of the Special Committee, Messrs. Klein and Arkley met with Mr. Tutor at Perini s headquarters in Framingham, Massachusetts to discuss the desire of the Special Committee that Tutor-Saliba s dividends prior to closing not result in the combined company lacking a level of cash at closing adequate to support anticipated bonding needs. As a result of these discussions, among other changes, Mr. Tutor agreed that the dividends that Tutor-Saliba would distribute to its shareholders prior to the closing would be limited to the amount of cash held by Tutor-Saliba and that the balance of any dividend would be paid in the form of notes.

On March 19, 2008, the Special Committee met at Perini s headquarters in Framingham, Massachusetts, together with its financial and legal advisors. Messrs. Klein and Arkley updated the other members of the Special Committee on the outcome of the prior day s negotiations with Mr. Tutor and the changes to which he had agreed to address the concerns with respect to available cash raised by the Special Committee. The Special Committee also asked additional questions of the representatives of UBS regarding its preliminary financial analyses of Tutor-Saliba, Perini and the proposed transaction presented at the last meeting of the Special Committee. The members of the Special Committee discussed Perini management s preliminary estimates of the projected effects of the proposed transaction on the earnings per share of the combined company.

Later that day, the Perini board of directors held a regularly scheduled meeting at Perini s headquarters in Framingham, Massachusetts, together with the Special Committee s financial and legal advisors. (Mr. Tutor did not attend the portion of the meeting during which the proposed transaction with Tutor-Saliba and all related matters were discussed.) At the meeting, Mr. Klein and representatives of Kirkland & Ellis reported to the other members of the Perini board of directors the status of negotiations and general terms of the proposed transaction between Perini and Tutor-Saliba. Mr. Klein indicated that the Special Committee expected to be able to make a formal recommendation to the board of directors within a few weeks.

Following these meetings, the negotiation of the transaction agreements and due diligence and financial analyses of the transaction continued among the representatives of Tutor-Saliba, the Special Committee and their respective advisors. In addition, each of the companies prepared and circulated disclosure schedules required by the draft agreements.

Based on the status of the negotiations of the proposed transaction, Mr. Klein convened meetings of the Special Committee and the Perini board of directors to take place on March 27, 2008 in New York City to formally consider the terms of the proposed transaction. However, before those meetings commenced, Messrs. Tutor and Klein met and decided that the transaction structure should be altered to be simpler and to avoid raising complications under applicable securities laws that might delay completion of the transaction or have certain short-term negative securities law effects on Perini. Specifically, they agreed that, rather than creating a new holding company that would own both Perini and Tutor-Saliba, Tutor-Saliba would merge with and into a new wholly owned subsidiary of Perini and Perini (rather than a holding company) would continue as the surviving, publicly traded parent corporation in the transaction. This change was made in light of the discussions over the preceding weeks in which Perini, Tutor-Saliba and their respective advisors sought to determine the optimum structure of the transaction that would, among other things, (i) result in the transaction qualifying as a tax-free transaction to the parties, (ii) have the greatest speed and highest likelihood of obtaining shareholder approval and satisfying the other closing conditions, and (iii) minimize any adverse effects to the combined company under the federal securities laws (which would be more significant for a holding company structure). None of the discussions regarding the structure of the transaction involved any changes to the consideration to be paid in the transaction. Mr. Klein and Mr. Tutor also discussed and tried to finalize (which they did through a series of phone calls over the ensuing days) the ownership percentages at which the voting and transfer restrictions in the Shareholders Agreement would lapse and the amount of shares that the Tutor Group would be entitled to vote in its discretion until such time under the voting restric

Later on March 27, 2008, the Special Committee held a meeting in New York City, together with its financial and legal advisors. Mr. Klein and representatives of Kirkland & Ellis reported to the other members of Perini board of directors on the recent developments regarding the transaction structure and other negotiations regarding the transaction terms. In light of the revisions to the draft agreements required to implement the decision to revise the proposed transaction structure, the Special Committee determined to delay formal consideration of the transaction.

At the conclusion of the meeting of the Special Committee, the Perini board of directors held a meeting in New York City, together with its financial and legal advisors. (Mr. Tutor did not participate in this meeting.) At the meeting, Mr. Klein and representatives of Kirkland & Ellis reported to the other members of the Special Committee on the recent developments regarding the transaction structure and other negotiations regarding the transaction terms. Mr. Klein indicated that he expected that the Special Committee would make a formal recommendation to the board of directors during the following week.

During the next several days, the negotiation of the transaction agreements (including refining and incorporating the revised transaction structure) and due diligence and financial analyses of the transaction continued among the representatives of Tutor-Saliba, the Special Committee and their respective advisors.

On the morning of April 2, 2008, Mr. Tutor and Mr. Klein met to finalize the negotiation of the remaining unresolved items in the transaction agreements, including the calculation of the precise number of shares to be issued in the merger, the terms on which Tutor-Saliba would be permitted to make the previously contemplated pre-closing dividends, the terms on which the transfer restrictions in the shareholders agreement would continue after five years and various terms of the employment agreement.

Later that day, the Special Committee held a meeting in New York City, together with its financial and legal advisors. Prior to the meeting, drafts of the Merger Agreement, Shareholders Agreement and Employment Agreement were circulated to the Special Committee members. At the meeting, Mr. Klein and representatives of

Kirkland & Ellis reported to the other members of the Special Committee on the most recent developments regarding the transaction structure and other negotiations, reviewed the terms of the agreements and summarized the final terms of the proposed transaction. The representatives of Kirkland & Ellis again reviewed with the Special Committee its fiduciary duties in connection with its consideration of the transaction and reviewed with the Special Committee the results of its due diligence review of Tutor-Saliba. The representatives of UBS presented to the Special Committee UBS financial analyses of Tutor-Saliba, Perini and the proposed transaction and UBS delivered its oral opinion to the Special Committee, subsequently confirmed by delivery of UBS written opinion, dated April 2, 2008, to the effect that, as of such date, and based upon and subject to various assumptions, matters considered and limitations described in the opinion, the exchange ratio provided for in the merger was fair, from a financial point of view, to Perini, as more fully described below under Opinion of UBS Securities LLC beginning on page 59.

During a recess in the meeting of the Special Committee, the Compensation Committee of Perini s board of directors met to consider the terms of the Employment Agreement that had been negotiated with Mr. Tutor. Messrs. Arkley, Brittain and Kennedy also participated in the discussion, at the invitation of the Compensation Committee. After this discussion, the Compensation Committee directed Mr. Oneglia, the Chairman of the Compensation Committee, to discuss with Mr. Tutor certain changes to the Employment Agreement that members of the Committee thought would be desirable. Mr. Oneglia then met with Mr. Tutor but was unable to obtain Mr. Tutor s assent to the requested changes. The Compensation Committee then re-convened, and after further discussion of the terms of the Employment Agreement and the value to the combined company of having Mr. Tutor committee for the next five years, unanimously approved the Employment Agreement.

The Special Committee then reconvened and unanimously determined that the Merger Agreement and the transactions contemplated thereby, including the merger, the issuance of shares of Perini common stock in the merger and the amendment of the Perini amended and restated articles of organization, are advisable, fair to and in the best interests of Perini and Perini shareholders and recommended that the Perini board of directors approve the Merger Agreement and the transactions contemplated thereby. The factors considered by the Special Committee in making this determination and recommendation are more fully described below under Reasons for the Merger Special Committee beginning on page 47.

At the conclusion of the meeting of the Special Committee, the Perini board of directors held a meeting in New York City, together with representatives of UBS and Kirkland & Ellis and the outside legal counsel of the Perini board of directors. At the meeting, Mr. Klein informed the Perini board of directors of the determinations and recommendations of the Special Committee with respect to the proposed transaction, reviewing all its material terms. The representatives of UBS presented to the Perini board of directors UBS financial analyses of Tutor-Saliba, Perini and the proposed transaction and informed the board of directors that UBS had rendered an opinion to the Special Committee to the effect that, as of April 2, 2008, and based upon and subject to various assumptions, matters considered and limitations described in the opinion, the exchange ratio provided for in the merger was fair, from a financial point of view, to Perini. The representatives of Kirkland & Ellis separately summarized the final terms of the transaction and the transaction agreements and reviewed with the directors their fiduciary duties in connection with their consideration of the transaction. Following deliberations, the board of directors unanimously (excluding Mr. Tutor, who did not participate in the meeting) determined that the Merger Agreement and the transactions contemplated thereby, including the merger, the issuance of shares of Perini common stock in the merger and the amendment of the Perini amended and restated articles of organization, are advisable, fair to and in the best interests of Perini and Perini shareholders, approved the Merger Agreement and the transactions contemplated thereby and recommended that Perini shareholders vote for the proposal to approve the issuance of shares of Perini common stock in the merger and the proposal to approve the amendment of the Perini amended and restated articles of organization. The factors considered by the Perini board of directors in making this determination and recommendation are more fully described below under Reasons for the Merger Perini Board of Directors beginning on page 54.

Following these meetings, the Merger Agreement, the Shareholders Agreement and the Employment Agreement were executed. Perini and Tutor-Saliba issued a joint press release announcing the merger and the execution of the agreements on the evening of April 2, 2008.

During the week of May 19, 2008, representatives of Tutor-Saliba asked representatives of the Special Committee whether they agreed that the Merger Agreement as executed did not adequately reflect Tutor-Saliba s ability to make tax distributions to its shareholders through the closing date of the merger. In response, the parties agreed that the Merger Agreement as executed did not expressly permit Tutor-Saliba shareholders to receive tax distributions intended to cover all such income taxes, as had been agreed orally. Following discussion of this issue among the parties and their respective financial and legal advisors, the parties agreed that this issue should be clarified in an amendment to the Merger Agreement to more clearly and fully reflect that understanding of the parties.

On May 26, 2008, Kirkland & Ellis delivered to Tutor-Saliba and its advisors a draft of an amendment to the Merger Agreement to make this clarification.

On May 27, 2008, the Special Committee held a meeting by teleconference, together with its financial and legal advisors. Following discussion of the amendment and an update by the Special Committee s advisors on the discussions regarding the amendment, the Special Committee determined that the proposed amendment to the Merger Agreement and the transactions contemplated thereby properly reflected the original business understandings and intentions of the parties with respect to tax distributions by Tutor-Saliba to its shareholders and were advisable, fair to and in the best interests of Perini and Perini shareholders and recommended that the Perini board of directors approve the amendment to the Merger Agreement. At the conclusion of the meeting of the Special Committee, the Perini board of directors held a meeting by teleconference, together with the Special Committee s legal advisors. At the meeting, Mr. Klein, Chairman of the Special Committee, informed the Perini board of directors of the determinations and recommendations of the Special Committee (and the discussions between the Special Committee and its advisors) with respect to the proposed amendment to the Merger Agreement. Following discussion of the amendment, the Perini board of directors (excluding Mr. Tutor, who did not participate in the meeting) determined that the proposed amendment to the Merger Agreement and the transactions contemplated thereby were advisable, fair to and in the best interests of Perini shareholders and approved the amendment and the transactions contemplated thereby were advisable, fair to and in the best interests of Perini shareholders and approved the amendment and the transactions contemplated thereby.

The amendment to the Merger Agreement was executed on May 28, 2008.

#### **Recommendation of the Perini Board of Directors**

The Perini board of directors unanimously (excluding Mr. Tutor, who did not participate in the meeting) determined that the Merger Agreement and the transactions contemplated thereby, including the merger, the issuance of shares of Perini common stock in the merger and the amendment of the Perini amended and restated articles of organization, are advisable, fair to and in the best interests of Perini and Perini shareholders and approved the Merger Agreement and the transaction contemplated thereby. *THE PERINI BOARD OF DIRECTORS RECOMMENDS THAT PERINI SHAREHOLDERS VOTE FOR THE SHARE ISSUANCE PROPOSAL AND THE ARTICLES AMENDMENT PROPOSAL*.

#### **Reasons for the Merger**

#### **Special Committee**

The Special Committee, with the advice and assistance of its independent legal and financial advisors, evaluated and negotiated the terms of the merger on behalf of Perini, including the terms and conditions of the Merger Agreement and the other related agreements, over the course of almost four months of deliberations. On April 2, 2008, the Special Committee unanimously determined that the Merger Agreement and the transactions contemplated thereby, including the merger, the issuance of shares of Perini common stock in the merger and the

amendment of the Perini amended and restated articles of organization, are advisable, fair to and in the best interests of Perini and the Perini shareholders and recommended that the Perini board of directors approve the Merger Agreement and the transactions contemplated thereby.

In reaching its decision to recommend that the Perini board of directors approve the Merger Agreement and the transactions contemplated thereby, the Special Committee consulted with Perini management, other members of the Perini board of directors (excluding Mr. Tutor) and the Special Committee s independent advisors in connection with the merger, including UBS, the Special Committee s financial advisor, and Kirkland & Ellis LLP, the Special Committee s outside legal counsel, and considered various material factors described below. Among the material information and factors considered by the Special Committee were the following:

*Strategic Considerations*. The Special Committee considered a number of factors pertaining to the strategic rationale for the merger, including the following:

*Increased Size and Scale and Management Capabilities; Potential Impact on Margins.* The combined company will have enhanced size and scale, including a pro forma backlog at the end of 2007 in excess of \$9 billion and over \$40 billion of targeted projects. In addition, the Special Committee considered the faster rate of projected growth of Tutor-Saliba, as compared to other comparable companies in the construction industry (including Perini on a stand-alone basis), and considered the likelihood that such projected growth would be achievable. The combination also is expected to enhance Perini s growth prospects significantly by adding substantial management capacity, client relationships and other capabilities to Perini, such as increased ability to perform its own subcontracting work through Tutor-Saliba s plumbing and mechanical and electrical subcontracting subsidiaries, and cost efficiencies. The larger size and scale of the combined company also should result in enhanced opportunities for growth, improved gross margins and greater operational flexibility because of additional access to markets, greater depth of resources of the combined company (including enhanced project management and estimating capabilities and a larger equipment fleet), enhanced focus on project process and execution and stronger experience in change management. The Special Committee also noted that larger publicly traded engineering and construction companies.

*Complementary Geographic Scope of Businesses.* Perini has focused its civil business segment on the Northeast and Southeast of the United States and the Middle East. Tutor-Saliba s civil business segment has focused more of its business on the West Coast and Southwest of the United States, the island of Guam and the Philippines. Both companies have commercial operations in the West, particularly in California and Nevada. Accordingly, the Special Committee determined that the merger will allow Perini to increase the diversification of its civil business through multiple geographic regions, which is expected to permit the combined company to compete for a larger number of projects, increase Perini s profile and local experience in public works projects in geographic regions that it has not historically served, decrease the costs of performing public works projects in regions not historically served by Perini and decrease Perini s exposure to regional economic downturns. Based on expectations for future public works and infrastructure spending, the Special Committee determined that the geographic areas in which Tutor-Saliba has more of a historical focus than Perini would likely be key growth markets for the combined company.

*Complementary Assets and Areas of Expertise; Expanding and Improving Civil Operations.* The merger will combine companies with complementary assets and areas of expertise, particularly Perini s larger building operations and Tutor-Saliba s expertise in civil projects. The combined company is expected to be able to draw upon the intellectual capital, technical expertise and experience of a deeper and more diverse workforce. In particular, the merger is expected to improve the performance and scope of Perini s civil segment once Tutor-Saliba s highly successful civil construction business can be integrated with Perini s civil segment. Tutor-Saliba s civil construction business has historically achieved substantially higher margins on its projects

than similar Perini civil projects. Over a five year period Tutor-Saliba has averaged 15% gross margins on its civil projects. With the integration of these businesses, sharing of management experience and best practices and expansion of capabilities, the parties expect to improve the operation and performance of both businesses, and Perini s civil businesses in particular.

Increased Ability to Leverage Combined Surety Capacity; Operational Flexibility from Strong Combined Balance Sheet. The combined company is expected to have an increased ability to obtain surety bonding, which will permit the combined company to compete for and perform a wider range of civil projects than Perini has historically. The combined company will be able to use the stronger balance sheet of the combined company (in particular as compared to the historical strength of Tutor-Saliba s balance sheet) to facilitate obtaining greater surety capacity, which will support larger civil projects in which Tutor-Saliba has significant experience but which Tutor-Saliba might otherwise be limited in pursuing given its independent surety capacity. In addition, the Special Committee noted that the all-stock nature of the transaction would permit Perini to retain the cash on its balance sheet without using it for acquisition consideration. (In evaluating these matters, the Special Committee took into account that under the terms of the Merger Agreement Tutor-Saliba would be permitted to dividend a portion of its available cash balances to its shareholders prior to completion of the merger, so that the merger would not materially add to the combined company is cash balances upon closing.)

*Consolidation of Mr. Tutor s Management Activities.* As discussed below, the merger will consolidate Mr. Tutors management activities on the growth and development of the combined company, and eliminate any conflicts of interest that existed as a result of his role as an executive of two companies with responsibilities to each company.

*Expected Ability to Integrate Operations.* Perini is very familiar with the operations of Tutor-Saliba, as Mr. Tutor is an executive of both companies and the companies have collaborated on numerous joint ventures for many years. In addition, both Perini and Tutor-Saliba have each completed recent acquisitions of businesses and have a strong record of accomplishment of successfully integrating companies in various business combination transactions. The Special Committee took note of these considerations in evaluating the likelihood that the operations of the two companies could be integrated successfully and efficiently.

Continued Role of Ronald N. Tutor; Terms of the Employment Agreement and Shareholders Agreement. In connection with the execution of the Merger Agreement, Mr. Tutor signed an Employment Agreement with Perini, to take effect upon the closing of the merger, in which he has agreed to serve on a full-time basis as the chairman of the Perini board of directors and chief executive officer of Perini following the completion of the merger. (He has held these positions on a part-time basis to date, sharing his time between Perini and Tutor-Saliba). The Special Committee determined that Mr. Tutor s potential departure from Perini could have significant adverse effects on Perini s future success on a stand-alone basis, and that his full-time employment with Perini in connection with the completion of the merger would mitigate the risk of those adverse effects on Perini s operations being realized. In the Special Committee s opinion, Mr. Tutor s continued participation was important because of (i) his proven track-record in successfully bidding on and profitably managing large construction projects, (ii) his relationships with principals in the surety and bonding industry, which provides enhanced access to bonding and insurance for Perini s construction projects, and (iii) his success in managing large civil and private construction projects on a fixed price basis. The Employment Agreement provides for an initial term of five years. The Special Committee also insisted that a significant amount of Mr. Tutor s personal net worth, including through two trusts controlled by him, would be tied to the performance of the combined company. Subject to certain exceptions, the Shareholders Agreement requires that he retain all of the shares of Perini common stock he will receive in the merger (either personally or through the Tutor Group) for six months after the closing of the merger and that 70% of those shares be retained for at least five years after the closing of the merger. Please see The Employment Agreement beginning on page 89 for a detailed discussion of the terms

and conditions of the Employment Agreement and The Shareholders Agreement beginning on page 86 for a detailed discussion of the terms and conditions of the Shareholders Agreement.

*Fixed Exchange Ratio; Basis of Negotiation of Price.* The Special Committee considered that the number of shares to be issued in the merger will not adjust to compensate for changes in the market price of Perini common stock prior to the closing of the merger. The Special Committee determined this to be particularly important in light of recent declines in the trading price of Perini common stock at the time of the negotiations of the merger, as compared to historic prices, the trading history and multiples of comparable companies and recent negative announcements regarding the commercial construction business, particularly including some of the markets in which Perini is heavily involved (such as the hospitality and gaming industry in Las Vegas, Nevada). The Special Committee noted that the economic terms of the merger were negotiated on the basis of the implied relative values of the companies, rather than valuing the consideration to be paid in the merger to the Tutor-Saliba shareholders on the basis of the market price for Perini common stock, which the Special Committee considered to be depressed.

*Financial Considerations*. The Special Committee considered the expected financial impact of the merger on Perini, including that the merger is expected to be dilutive in 2008 (assuming completion in the third quarter of 2008) and accretive to Perini shareholders after 2008 and that the accretive nature of the transaction would be expected to result in increased earnings per share of Perini common stock. The Special Committee also considered the historic financial condition, operating results and businesses of Perini and Tutor-Saliba, including information with respect to their respective earnings histories. Please see Projected Financial Information beginning on page 55.

*Synergies*. The Special Committee considered that, although no assurance can be given that any particular level of cost savings and other synergies will be achieved, Perini and Tutor-Saliba management had identified estimated synergies of approximately \$39 million of additional pre-tax earnings in 2009 (assuming the merger is completed in the third quarter of 2008), with increasing amounts in subsequent period projected to grow to approximately \$72 million of additional pre-tax earnings in 2012, that it expected could be realized following completion of the merger, most of which would begin to be realized in calendar year 2009 (assuming the merger closes in the third quarter of 2008). These include the ability to maintain or improve the earnings from operations of Perini s civil business (which the Special Committee believed would not add to the overall profitability of Perini if Mr. Tutor ceased to be employed by Perini), benefits from the shared access to management from both companies and enhanced operating margin through increased self-performance capabilities. The Special Committee took note of the fact that the synergy amounts were estimates, that they could change and that achieving the synergies would be subject to a number of uncertainties. In addition, the Special Committee did not compare the estimated synergies related to the merger to synergies of other potential transactions because Perini was not actively considering any other potential business combination transaction.

*Financial Presentation and Opinion of UBS Securities LLC*. The Special Committee considered the financial analyses and presentation of UBS, as presented to the Special Committee on April 2, 2008, and the opinion of UBS rendered orally to the Special Committee on April 2, 2008, which was subsequently confirmed by delivery of UBS written opinion, dated April 2, 2008, to the effect that, as of such date, and based upon and subject to various assumptions, matters considered and limitations described in the opinion, the exchange ratio provided for in the merger was fair, from a financial point of view, to Perini, as more fully described below under Opinion of UBS Securities LLC beginning on page 59.

*Post-Merger Corporate Governance; Terms of the Shareholders Agreement.* The Special Committee considered the corporate governance provisions of the Merger Agreement and the Shareholders Agreement, including that, upon completion of the merger, the Perini board of directors will continue to be comprised of a majority of independent directors selected by the Corporate Governance and Nominating Committee of the Perini board of directors (all of which are currently expected to be directors elected by Perini shareholders prior to the merger, including the new Class III director nominees) and that Mr. Tutor will continue to serve as chairman of the Perini board of directors and chief executive officer of the combined company. In addition, the Shareholders Agreement requires

that, for at least three years following the merger (and longer if Mr. Tutor (either personally or through the Tutor Group) continues to hold at least 20% of the outstanding shares of Perini common stock), all of the shares of Perini common stock that Mr. Tutor will receive in the merger (either personally or through the Tutor Group) will be voted in favor of this slate of directors, Mr. Tutor (either personally or through the Tutor Group) will exercise voting control of no more than 20% of the outstanding shares of Perini common stock on all other matters and Mr. Tutor (personally and on behalf of trusts controlled by him) will refrain from taking certain actions that could facilitate an unsolicited acquisition of control of Perini by Mr. Tutor or his affiliates. In addition, the Shareholders Agreement includes transfer restrictions that will limit the right of Mr. Tutor to recognize a change of control premium on the sale of the shares of Perini common stock he will receive in the merger (either personally or through trusts controlled by him) without all other shareholders also receiving that premium. Collectively, these provisions are expected to ensure that all shareholders of the combined company continue to have broad representation and voting authority with respect to the combined company. Please see

Post-Merger Governance and Management beginning on page 90 and The Shareholders Agreement beginning on page 86 for further information and a detailed discussion of the terms and conditions of the Shareholders Agreement.

*Terms of the Merger Agreement.* The Special Committee reviewed and considered the terms of the Merger Agreement and considered that, in its view, the material terms of the Merger Agreement, taken as a whole, were reasonable for an arms -length acquisition transaction. In particular, the Special Committee considered the representations and warranties made by Tutor-Saliba and its shareholders in the Merger Agreement, the restrictions on the operation of the Tutor-Saliba business from the signing of the Merger Agreement until the closing of the merger and the other covenants of Tutor-Saliba and its shareholders in the Merger Agreement, the conditions to each party s obligation to complete the merger and the rights of indemnification of each party to the Merger Agreement for losses as a result of breaches of the Merger Agreement. The Special Committee also considered that the Merger Agreement permitted the Perini board of directors and the Special Committee to consider and engage in negotiations regarding unsolicited potentially superior proposals, to withdraw or otherwise change its recommendation to Perini shareholders in favor of the proposals related to the merger for any reason if the failure to take such action would be reasonably likely to result in a breach of its fiduciary obligations and to terminate the Merger Agreement to accept a superior proposal. Please see The Merger Agreement beginning on page 73 for a detailed discussion of the terms and conditions of the Merger Agreement.

*Likelihood of Completion of the Merger*. The Special Committee considered the likelihood that the merger will be completed on a timely basis, including the likelihood that the merger will receive all necessary regulatory approvals without unacceptable conditions.

*Due Diligence*. The Special Committee considered the scope of the due diligence investigation of Tutor-Saliba conducted by members of Perini management and Perini s and the Special Committee s outside advisors and evaluated the results thereof. The Special Committee also took account Perini s familiarity with the Tutor-Saliba business, including that from a history of joint ventures between the two companies.

*Strategic Alternatives*. The Special Committee considered the trends and competitive developments in the industry and the range of strategic alternatives available to Perini, including the possibility of business combinations with other participants in the industry or continuing to operate as a stand-alone entity. At the time of its consideration of the merger, Perini was not involved in discussions regarding any other business combination transaction. The Special Committee initially decided not to pursue alternative transactions with other participants in the industry until it had first explored more completely a combination with Tutor-Saliba or received a proposal for an alternative transaction on an unsolicited basis. The Special Committee reached this decision on January 16, 2008 based on several factors, which are described in Background of the Merger beginning on page 36. The Special Committee ultimately did not opt to pursue other strategic alternatives during its consideration of the merger, primarily because it determined that the merger was in the best interests of the Perini shareholders, the Merger Agreement included expected provisions that would allow the Special Committee and the Perini board of directors to pursue an alternative transaction proposal reasonably

likely to lead to a superior proposal if one materialized (or change the recommendation of the Perini board of directors to the Perini shareholders pursuant to its fiduciary obligations), the Special Committee determined that soliciting interest in alternative transactions was not necessary or appropriate in light of the nature of the merger in which Perini shareholders would continue to hold their interests in Perini and no unsolicited offers for alternative transactions had been made known to the Special Committee during the period of consideration of the Merger Agreement.

*Impact of the Merger on Customers and Employees.* The Special Committee evaluated the expected impact of the merger on Perini s customers and employees and the benefits that would be derived from the merger by enhancing operations and by providing more opportunities for employees in a larger, more competitive company.

The Special Committee also considered potential risks associated with the merger, including the following:

*Tutor-Saliba Business Risks*. The Special Committee considered certain risks associated with Tutor-Saliba s business and operations, including the likelihood of winning future significant projects, risks associated with recent acquisitions completed by Tutor-Saliba (including uncertainty about Tutor- Saliba s ability to operate and grow the acquired businesses successfully), the fact that Tutor-Saliba has grown significantly over the last two years, the fact that a small number of customers account for a large percentage of Tutor-Saliba s revenues and Tutor-Saliba, as compared to other comparable companies (including Perini on a stand-alone basis), and considered the risks that such projected growth would not be achievable and, if not achieved, the adverse effects that could have on the Special Committee s expectations for the combined company s performance and the corresponding decrease in the expected equity value of shares of Perini common stock. In addition, the Special Committee considered the fact that, as a result of the merger, certain business risks related to Tutor-Saliba may have the effect of increasing the effect of existing risks related to Perini for the combined company) from the hospitality and gaming industry, particularly in Las Vegas, Nevada, as Tutor-Saliba is subject to similar risks based on its operations in Las Vegas, Nevada. Please see Risk Factors beginning on page 26 for additional information on business risks relating to Tutor-Saliba s impact on the combined business.

*Uncertainty of Projections and Judgments About Implied Relative Values.* The Special Committee took account of the fact that the projections and valuation methodologies used in assessing the relative values of Tutor-Saliba and Perini, developing estimates of potential synergies and negotiating the economic terms of the merger are inherently uncertain, as they involve estimates, judgments, assumptions and predictions about future events that may differ significantly from what actually occurs in the future. While the Special Committee took into consideration, in evaluating the projections, synergy estimates and valuations, the views of the management teams of the two companies, the advice of its outside advisors and the Special Committee members own knowledge of and experience with the industry and the businesses, the Special Committee carefully considered and discussed the risks and uncertainties associated with all of these matters and also considered various disagreements and differences of views between different constituencies about the future prospects for each of Perini and Tutor-Saliba, both on a stand-alone basis and as part of a combined company. Ultimately, while the Special Committee made determinations about what it believed were the most reasonable and appropriate assumptions and estimates, it did so with a full understanding of the uncertainties and risks inherent in such determinations. For a further discussion of these risks, please see Risk Factors beginning on page 26.

*No Public Market Value for Tutor-Saliba*. Tutor-Saliba is a privately held corporation and, accordingly, there is no public equity market valuation of Tutor-Saliba. In the opinion of the Special Committee (after consideration of information provided by its financial advisor), there are not any publicly traded

companies that are sufficiently closely comparable with Tutor-Saliba to provide a reasonably, comparative basis of valuation for Tutor-Saliba. Furthermore, any valuation of Tutor-Saliba is dependent on an estimate of the extent to which Tutor-Saliba will be able to achieve its forecasted operating results, which include high growth rates that may or may not be obtained, as well as significant anticipated growth for recently acquired businesses and for new markets (particularly in the island of Guam). While the Special Committee recommended that the Perini board of directors approve the Merger Agreement (including the consideration to be paid in the merger), the Special Committee considered these uncertainties and limitations and took into account the possibility that the public markets might reach a different conclusion in assessing the value of Tutor-Saliba or might be uncertain (or skeptical) about any value because of the informational limits and inherent uncertainties in valuing a privately owned business projecting substantial future growth. The Special Committee also considered that the actual operating results of Tutor-Saliba could be significantly different from its forecasted operating results, and accordingly the market price of Perini common stock could be particularly volatile as a result of the merger. Please see Risk Factors If the public markets assign lower values to the Tutor-Saliba business than the values used in negotiating the terms of the merger, the trading price of Perini common stock may decline. beginning on page 26.

*Termination Fee; Alternative Proposals.* The Special Committee considered the restrictions that the Merger Agreement imposes on actively soliciting alternative business combination transactions, the provisions of the Merger Agreement that require the reimbursement by Perini of up to \$5 million of transaction expenses of Tutor-Saliba following certain events of termination of the Merger Agreement and the provisions of the Merger Agreement that require the potential payment by Perini of a termination fee of \$30 million following a termination of the Merger Agreement and the consummation of an alternative transaction. The Special Committee understood that these provisions may have the effect of discouraging proposals for alternative business combinations for Perini until after the completion of the merger (or the earlier termination of the Merger Agreement) and may make it less likely that the transactions related to such alternative proposals would be negotiated or pursued during such period, even if they might be more favorable to the shareholders of Perini than the merger. Please see The Merger Agreement Termination Termination Fees/Reimbursement of Expenses beginning on page 85 for further information regarding such restrictions and fees and expenses.

*Employee Matters*. The Special Committee considered the impact that business uncertainty pending completion of the merger could have on the ability to attract, retain and motivate key personnel until the merger is completed.

Additional Interests of Executive Officers and Directors. The Special Committee considered that certain executive officers and directors of Perini (in particular, Mr. Tutor) have or may have interests with respect to the merger in addition to their interests as shareholders of Perini. Please see Additional Interests of Directors, Executive Officers and Certain Beneficial Owners beginning on page 70 for further information on these additional interests.

*Restrictions on Interim Operations.* The Special Committee considered the provisions of the Merger Agreement placing restrictions on Perini s operations during the period between the signing of the Merger Agreement and the completion of the merger. Please see The Merger Agreement Covenants Restrictions on the Parties Business Prior to the Closing beginning on page 77 for further information on these restrictions.

*Diversion of Management*. The Special Committee considered the possible diversion of management s time and attention from Perini s ongoing business due to the substantial time and effort necessary to complete the merger and plan for and implement the integration of the operations of Perini and Tutor-Saliba.

The Special Committee believed that, overall, the potential benefits of the merger to Perini and its shareholders outweighed the risks.

On May 27, 2008, the Special Committee determined that an amendment to the Merger Agreement to clarify the right of Tutor-Saliba to make certain tax distributions and the transactions contemplated thereby were advisable, fair to and in the best interests of Perini and Perini shareholders and recommended that the Perini board of directors approve the amendment to the Merger Agreement and the transactions contemplated thereby. In reaching its decision, the Special Committee consulted with Perini management and the Special Committee s financial and legal advisors. The Special Committee determined that (i) the proposed amendment and the transactions contemplated thereby properly reflected the original business understandings and intentions of the parties with respect to tax distributions by Tutor-Saliba to its shareholders, (ii) the revisions contemplated by the amendment, if present in the Merger Agreement as initially executed, would not have changed its decision on April 2, 2008 with respect to the Merger Agreement, and (iii) agreeing to the amendment was in the best interests of the long-term relationship of the parties.

Please see Additional Information about Factors Considered by the Special Committee and the Perini Board of Directors beginning on page 55 for further information about the considerations of the Special Committee.

### **Perini Board of Directors**

On April 2, 2008, the Perini board of directors unanimously (excluding Mr. Tutor, who did not participate in the meeting) determined that the Merger Agreement and the transactions contemplated thereby, including the merger, the issuance of shares of Perini common stock in the merger and the articles amendment proposal, are advisable, fair to and in the best interests of Perini and Perini shareholders, approved the Merger Agreement and the transaction contemplated thereby and recommended that Perini shareholders vote *FOR* the share issuance proposal and the articles amendment proposal.

In reaching its decision to approve the Merger Agreement and the transactions contemplated thereby and recommend that Perini shareholders vote *FOR* the share issuance proposal and the articles amendment proposal, the Perini board of directors consulted with Perini management, the Special Committee, and the separate outside counsel to Perini, considered various material factors described below, and received presentations from Kirkland & Ellis regarding the Merger Agreement and the related transaction documents and from UBS regarding its financial analyses, and had an opportunity to ask questions of the Special Committee, its advisors and Perini management. Among the material information and factors considered by the Perini board of directors were the following:

*Negotiation by Special Committee.* The Perini board of directors considered the fact that the Merger Agreement and the transactions contemplated thereby were negotiated on behalf of Perini by the Special Committee with the assistance of its independent financial and legal advisors, and the support of Perini management. The Special Committee was comprised entirely of directors who are not officers of Perini or Tutor-Saliba or affiliated with Tutor-Saliba or Mr. Tutor and will not personally benefit from the completion of the merger in a manner different from the unaffiliated Perini shareholders.

*Recommendation of the Special Committee*. The Perini board of directors considered the unanimous determination of the Special Committee that the Merger Agreement and the transactions contemplated thereby, including the merger, the issuance of shares of Perini common stock in the merger and the amendment of the Perini amended and restated articles of organization, are advisable, fair to and in the best interests of Perini and Perini shareholders and the unanimous recommendation of the Special Committee that the Perini board of directors approve the Merger Agreement and the transactions contemplated thereby.

*Factors Considered by the Special Committee*. The Perini board of directors discussed with the Special Committee and considered the factors considered by the Special Committee in reaching its recommendation, including the positive factors and potential benefits of the merger and the risks and other potentially negative factors concerning the merger, as described above. The Perini board of directors also took note of the fact that the Special Committee received advice from its legal and

financial advisors and considered the content of the presentations made directly to the Perini board of directors by the Special Committee s advisors.

The Perini board of directors believed that, overall, the potential benefits of the merger to Perini and its shareholders outweighed the risks.

On May 27, 2008, the Perini board of directors (excluding Mr. Tutor, who did not participate in the meeting) determined that an amendment to the Merger Agreement to clarify the right of Tutor-Saliba to make certain tax distributions and the transactions contemplated thereby were advisable, fair to and in the best interests of Perini and Perini shareholders and approved the amendment and the transactions contemplated thereby. In reaching its decision, the Perini board of directors consulted with Perini management, the Special Committee and the Special Committee and the proposed amendment and the transactions contemplated thereby properly reflected the original business understandings and intentions of the parties with respect to tax distributions by Tutor-Saliba to its shareholders, (ii) the revisions contemplated by the amendment, if present in the Merger Agreement as initially executed, would not have changed its decision on April 2, 2008 with respect to the Merger Agreement, and (iii) agreeing to the amendment was in the best interests of the long-term relationship of the parties.

# THE PERINI BOARD OF DIRECTORS RECOMMENDS THAT PERINI SHAREHOLDERS VOTE FOR THE SHARE ISSUANCE PROPOSAL AND THE ARTICLES AMENDMENT PROPOSAL.

Please see Additional Information about Factors Considered by the Special Committee and the Perini Board of Directors beginning on page 55 for further information about the considerations of the Perini board of directors.

#### Additional Information about Factors Considered by the Special Committee and the Perini Board of Directors

The foregoing discussion summarizes the material factors considered by the Special Committee and the Perini board of directors in their respective considerations of the merger, however, it is not intended to be exhaustive. In view of the wide variety of factors considered by the Special Committee and the Perini board of directors in connection with their respective evaluations of the merger, neither the Special Committee nor the Perini board of directors considered it practical to, nor did it attempt to, quantify, rank or otherwise assign relative weights to the specific factors that it considered in reaching its decisions. In considering the factors described above, individual members of the Special Committee or the Perini board of directors may have given different weight to different factors. The Special Committee and the Perini board of directors each considered the information and factors to be favorable to, and in support of, its determinations and recommendations.

The Special Committee and the Perini board of directors each realized that there can be no assurance about future results, including results considered or expected as described in the factors listed above, such as assumptions regarding potential synergies. It should be noted that this explanation of the factors considered by the Special Committee and the Perini board of directors and all other information presented in this section are forward-looking in nature and, therefore, should be read in light of the factors discussed under the heading Note Regarding Forward-Looking Statements beginning on page 20.

#### **Projected Financial Information**

Although Perini periodically may issue limited guidance to investors concerning its expected financial performance, Perini does not as a matter of course publicly disclose detailed financial projections. However, in connection with the negotiation of the merger, Perini management provided the Special Committee and its financial advisor with certain non-public financial projections that were prepared by the Perini management for internal planning purposes and not for public disclosure. Mr. Tutor did not participate in the preparation of

Perini s base case projections. Similarly, Tutor-Saliba management provided the Special Committee and its financial advisor with certain non-public financial projections that were prepared by the Tutor-Saliba management for internal planning purposes and not for public disclosure.

The Special Committee reviewed, challenged and independently assessed the projections of financial performance for each of Perini and Tutor-Saliba proposed by each company s management, respectively. The Special Committee refined and revised its assessment of the projections of financial performance of each company from time to time prior to the April 2, 2008 signing of the Merger Agreement, taking into account discussions with each company s management, its own views and the views of the other directors of Perini, to create a set of projections for each company and an estimate of synergies for the combined company that reflected the best currently available estimates and judgments of the Special Committee of the future financial performance of the two companies and the combined company (reflecting the estimated synergies). The projections reflecting all refinements and revisions by the Special Committee prior to the April 2, 2008 signing of the Merger Agreement are referred to as the Special Committee case projections.

The Special Committee case projections, together with other relevant information regarding the two companies and the proposed transaction, were provided to UBS prior to the delivery of UBS opinion, and the Special Committee directed UBS to utilize the Special Committee case projections for purposes of UBS analysis. Information regarding the material portions of the Special Committee case projections are summarized below.

The Special Committee case projections were prepared based upon a variety of estimates and assumptions. The estimates and assumptions underlying the financial projections of Perini, Tutor-Saliba and the synergies for the combined company included in the Special Committee case projections involve judgments with respect to, among other things, future economic, competitive, regulatory and financial market conditions and future business decisions, which are difficult to predict. While the financial projections summarized below were prepared in good faith by the Special Committee, no assurance can be given regarding future events, many of which are beyond Perini s control. Therefore, these financial projections may not be reliable or predictive of future operating results, and this information should not be relied on as such. Please see Risk Factors beginning on page 26 for additional information regarding the risks that the financial projections may not be achieved. At the time the merger was considered, the Special Committee and the Perini board of directors were aware of the foregoing limitations in respect of the Special Committee case projections.

The financial projections in this section were not prepared with a view toward public disclosure or with a view toward complying with the guidelines established by the American Institute of Certified Public Accountants with respect to prospective financial information or published guidelines of the SEC regarding financial projections, but, in the view of the Special Committee, were reasonably prepared on a basis reflecting the best currently available estimates and judgments as of the date of their preparation. Neither Perini s nor Tutor-Saliba s independent auditors, nor any other independent accountants, have compiled, examined or performed any procedures with respect to these financial projections, nor have they expressed any opinion or any other form of assurance on such information s achievability, and assume no responsibility for, and disclaim any association with, these financial projections. In addition, neither Perini s nor Tutor-Saliba s respective financial advisors expressed any opinion or any other form of assurance on such information s achievability. These financial projections are not historical fact and should not be relied upon as being necessarily indicative of actual future results. In light of the foregoing, and considering that the Perini annual meeting at which the merger proposals will be considered and voted upon will be held more than [1] months after the date that the underlying financial projections from each management team were provided to the Special Committee and more than [1] months after the date that the Special Committee last revised the Special Committee case projections, as well as the uncertainties inherent in any financial projections, Perini shareholders are cautioned not to unduly rely on these financial projections.

The inclusion of the descriptions below of the Special Committee case projections should not be interpreted as an indication that Perini considers this information necessarily reliable for predicting actual future results, and

this information should not be relied on for that purpose. In particular, due to the passage of time since the preparation of these financial projections, the projected events for certain of the periods presented have been superceded by events that have occurred to date, and accordingly, the estimates and assumptions underlying the Special Committee case projections are outdated. These projections are not included in this document in order to induce any Perini shareholder to vote to approve the merger proposals or to impact any investment decision with respect to the Perini common stock. Please see Note Regarding Forward-Looking Statements beginning on page 20.

#### PERINI DOES NOT INTEND TO UPDATE OR OTHERWISE REVISE THESE PROJECTIONS TO REFLECT CIRCUMSTANCES EXISTING SINCE THEIR PREPARATION OR TO REFLECT THE OCCURRENCE OF SUBSEQUENT EVENTS EVEN IN THE EVENT THAT ANY OR ALL OF THE UNDERLYING ASSUMPTIONS ARE NO LONGER APPROPRIATE.

#### Perini

Set forth below is a summary of the Special Committee case projections of financial performance of Perini on a stand-alone basis:

#### Summary of Perini Stand-Alone Financial Projections

(all amounts are in millions and are approximate)

	2008	2009	2010	2011	2012
Revenue	\$ 5,485	\$ 5,378	\$ 5,523	\$ 5,747	\$ 5,980
EBITDA (1)	\$ 157	\$ 151	\$ 165	\$ 176	\$ 181
Income From Construction Operations (EBIT)	\$ 145	\$ 136	\$ 151	\$ 152	\$ 154
Operating Margin	2.6%	2.5%	2.7%	2.6%	2.6%

(1) EBITDA is defined as income from construction operations plus depreciation and amortization.

The projected financial information set forth above reflects certain assumptions, which may cause the initial financial projections to vary significantly from actual financial results. In addition to those assumptions noted above, significant assumptions underlying the Special Committee case projections of financial performance of Perini on a stand-alone basis set forth above include the following:

Mr. Tutor would not continue as the chairman of the Perini board of directors and chief executive officer of Perini if the merger were not completed, and his services would otherwise cease to be available to Perini;

Without Mr. Tutor s leadership, the financial performance of Perini s civil business segment would produce break-even profit margins during 2008 and subsequent periods;

Perini s building segment would be able to maintain levels of backlog approximately equal to existing backlog and recognize the revenue resulting from that backlog, including by being awarded new work not previously included in backlog as existing projects are performed;

given expected changes in the markets in which Perini Management Services, Inc. (referred to in this proxy statement as PMSI) has operations, the profitability of PMSI s operations would decline from that of its recent profitability, although the profitability would remain positive through subsequent periods; and

Perini would not acquire any additional businesses.

#### Tutor-Saliba

Set forth below is a summary of the Special Committee case projections of financial performance of Tutor-Saliba on a stand-alone basis:

#### Summary of Tutor-Saliba Stand-Alone Financial Projections

(all amounts are in millions and are approximate)

	2008	2009	2010	2011	2012
Revenue	\$ 1,476	\$ 1,942	\$ 2,588	\$ 2,313	\$ 2,464
EBITDA (1)	\$ 100	\$ 162	\$ 213	\$ 207	\$ 224
Income From Operations (EBIT)	\$ 96	\$ 155	\$ 206	\$ 196	\$ 210
Operating Margin	6.5%	8.0%	8.0%	8.5%	8.5%

(1) EBITDA is defined as income from operations plus depreciation and amortization.

The projected financial information set forth above reflects certain assumptions, which may cause the initial financial projections to vary significantly from actual financial results. In addition to those assumptions noted above, significant assumptions underlying the Special Committee case projections of financial performance of Tutor-Saliba on a stand-alone basis set forth above include the following:

Tutor-Saliba s domestic building segment s revenue base would have strong growth reflecting the growth plans of Perini s clients and other participants in the Las Vegas, Nevada building market for the next several years, taking into account projects for which Tutor-Saliba was in advanced discussions, and thereafter Tutor-Saliba s domestic building segment would be able to maintain levels of backlog approximately equal to its then existing backlog and recognize the revenue resulting from that backlog, including by being awarded new work not previously included in backlog as then existing projects are performed;

Tutor-Saliba s civil business segment s revenue base would have strong near-term growth by maintaining its traditional market in what is expected to be a growing market for infrastructure spending in areas in which Tutor-Saliba has civil operations;

businesses acquired by Tutor-Saliba in 2007 and early 2008 Powerco Electric Corp., an electrical construction subcontractor, Desert Plumbing & Heating Co., Inc., a plumbing and mechanical (including HVAC) subcontractor, and the aggregates business would have increasing revenues as a result of favorable market factors and integration of their operations into Tutor-Saliba s other businesses;

revenues of Black Construction would grow steadily until 2010, when revenues would expand significantly with the award of large awards in 2010 and continuing for subsequent years, which periods correspond to the timetable for the expansion of the United States military s base on the island of Guam; and

Tutor-Saliba would not acquire any additional businesses. *Estimated Synergies* 

The Special Committee case projections included an estimate of synergies for the combined company. The Special Committee expected SG&A cost savings across both businesses of approximately \$5 million per year. In addition, the Special Committee expected most of the synergies from the merger would be in the form of incremental earnings. These items included:

maintaining the otherwise projected return to profitability of Perini s civil business as a result of drawing upon enhanced management capabilities from the combination of both companies management teams;

having incremental earnings resulting from increased bonding capacity of the combined company; and

having incremental earnings from the increased ability for Perini to self-perform concrete and other services following the merger (made possible in part due to Tutor-Saliba s recent acquisition of the aggregates business).

These synergies were projected to result in approximately \$39 million of additional pre-tax earnings in 2009 (assuming the merger is completed in the third quarter of 2008), with increasing amounts in subsequent periods projected to grow to approximately \$72 million of additional pre-tax earnings in 2012.

#### **Opinion of UBS Securities LLC**

On April 2, 2008, at a meeting of Perini s Special Committee held to evaluate the proposed merger, UBS delivered its oral opinion to the Special Committee, which was subsequently confirmed by delivery of UBS written opinion, dated April 2, 2008, to the effect that, as of such date, and based upon and subject to various assumptions, matters considered and limitations described in the opinion, the exchange ratio provided for in the merger was fair, from a financial point of view, to Perini. The exchange ratio in the Merger Agreement determines the number of shares of Perini common stock that would be received by the former Tutor-Saliba shareholders in the merger in exchange for each share of Tutor-Saliba common stock that is outstanding immediately prior to the effective time of the merger (other than certain excluded shares specified in the Merger Agreement).

The full text of UBS opinion describes the assumptions made, procedures followed, matters considered and limitations on the review undertaken by UBS. This opinion is attached as *Annex E* and is incorporated by reference in this proxy statement. **UBS opinion is directed only to the fairness, from a financial point of view, to Perini of the exchange ratio provided for in the merger. The opinion does not address the relative merits of the merger as compared to other business strategies or transactions that might be available to Perini or Perini s underlying business decision to effect the merger. The opinion does not constitute a recommendation to any shareholder as to how such shareholder should vote or act with respect to the merger. We encourage you to read UBS opinion carefully in its entirety.** The summary of UBS opinion presented below is qualified in its entirety by reference to the full text of the opinion.

In arriving at its opinion, UBS:

reviewed certain publicly available business and financial information relating to Perini;

reviewed certain internal financial information and other data relating to the businesses and financial prospects of Perini that were provided to UBS by Perini and not publicly available, including the financial forecasts and estimates prepared by the management of Perini and the Special Committee that the Special Committee directed UBS to utilize for purposes of UBS analysis;

reviewed certain internal financial information and other data relating to the businesses and financial prospects of Tutor-Saliba that were provided to UBS by Perini and not publicly available, including the financial forecasts and estimates prepared by the management of Perini, the management of Tutor-Saliba and the Special Committee that the Special Committee directed UBS to utilize for purposes of UBS analysis;

reviewed certain estimates of synergies prepared by the management of Perini, the management of Tutor-Saliba and the Special Committee, that were provided to UBS by Perini and not publicly available, that the Special Committee directed UBS to utilize for purposes of UBS analysis;

conducted discussions with the Special Committee and members of the senior managements of Perini and Tutor-Saliba concerning the businesses and financial prospects of Perini and Tutor-Saliba;

reviewed publicly available financial and stock market data with respect to certain other companies that are generally in the industry in which Perini and Tutor-Saliba operate;

compared the financial terms of the merger with the publicly available financial terms of other transactions involving certain other companies that are generally in the industry in which Perini and Tutor-Saliba operate;

performed discounted cash flow analyses of Perini and Tutor-Saliba in which UBS analyzed the future cash flows of Perini and Tutor-Saliba using the financial forecasts and estimates prepared by the management of Perini, the management of Tutor-Saliba and the Special Committee that the Special Committee directed UBS to utilize for purposes of UBS analysis;

reviewed current and historical market prices of Perini common stock;

considered certain pro forma effects of the merger on Perini s financial statements;

reviewed drafts of the Merger Agreement and the Shareholders Agreement, each dated April 2, 2008; and

conducted such other financial studies, analyses and investigations, and considered such other information, as UBS deemed necessary or appropriate.

In connection with its review, with the consent of the Special Committee, UBS assumed and relied upon, without independent verification, the accuracy and completeness in all material respects of the information provided to or reviewed by UBS for the purpose of its opinion. In addition, with the consent of the Special Committee, UBS did not make any independent evaluation or appraisal of any of the assets or liabilities (contingent or otherwise) of Perini or Tutor-Saliba, nor was UBS furnished with any such evaluation or appraisal. With respect to the financial forecasts, estimates, synergies and pro forma effects referred to above, UBS assumed, at the direction of the Special Committee, that they had been reasonably prepared on a basis reflecting the best currently available estimates and judgments of the Special Committee as to the future financial performance of Perini and Tutor-Saliba and such synergies and pro forma effects. In addition, UBS assumed, with the approval of the Special Committee, that the financial forecasts and estimates, including synergies, referred to above would be achieved at the times and in the amounts projected. UBS also assumed, with the consent of the Special Committee, that the merger would qualify for United States federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended. UBS opinion was necessarily based on economic, monetary, market and other conditions as in effect on, and the information available to UBS as of, the date of UBS opinion.

At the direction of the Special Committee, UBS was not asked to, nor did it, offer any opinion as to the terms, other than the exchange ratio to the extent expressly specified in UBS opinion, of the Merger Agreement or the form of the merger. UBS did not express any opinion as to the fairness of the amount or nature of, or any other aspects relating to, any compensation to be received by any officers, directors or employees of any parties to the merger, or any class of such persons, relative to the exchange ratio. UBS did not express any opinion as to what the value of Perini common stock would be when issued pursuant to the merger or the prices at which Perini common stock would trade at any time. In rendering its opinion, UBS assumed, with the consent of the Special Committee, that (i) the final executed form of each of the Merger Agreement and the Shareholders Agreement would not differ in any material respect from the April 2, 2008 drafts referred to above, (ii) the parties to the Merger Agreement would comply with all material terms of the Merger Agreement, and (iii) the merger would be consummated in accordance with the terms of the Merger Agreement without any adverse waiver or amendment of any material term or condition thereof. UBS also assumed that all governmental, regulatory or other consents and approvals necessary for the consummation of the merger would be obtained without any material adverse effect on Perini, Tutor-Saliba or the merger. Except as described above, the Special Committee imposed no other instructions or limitations on UBS with respect to the investigations made or the procedures followed by UBS in rendering its opinion. The issuance of UBS

In connection with rendering its opinion to the Special Committee, UBS performed a variety of financial and comparative analyses, which are summarized below. The following summary is not a complete description of all the analyses performed and factors considered by UBS in connection with its opinion. The preparation of a financial opinion is a complex process involving subjective judgments and is not necessarily susceptible to partial analysis or summary description. These analyses necessarily involve complex considerations and judgments concerning financial and operating characteristics and other factors that could affect the public trading or acquisition values of the companies concerned.

UBS believes that its analyses and the summary below must be considered as a whole and that selecting portions of the analyses and factors, without considering all analyses and factors or the narrative description of the analyses, could create a misleading or incomplete view of the processes underlying UBS analyses and opinion. UBS did not draw, in isolation, conclusions from or with regard to any one factor or method of analysis for purposes of its opinion but rather arrived at its ultimate opinion based on the results of all the analyses undertaken by it and assessed as a whole. As the Special Committee was aware, the financial and operating characteristics of Perini and Tutor-Saliba (including the business segments in which they operate) cause their financial results to have limited comparability, for valuation purposes, to those of other companies with publicly available financial and stock market data that UBS reviewed and of other transactions that UBS reviewed for which financial terms were publicly available, and, accordingly, UBS relied primarily on discounted cash flow analyses of the forecasts and estimates referred to above for purposes of UBS opinion.

The estimates of the future performance of Perini and Tutor-Saliba referred to above that the Special Committee directed UBS to utilize for purposes of UBS analyses are not necessarily indicative of future results or values, which may be significantly more or less favorable than those estimates. In performing its analyses, UBS considered industry performance, general business and economic conditions and other matters, many of which are beyond Perini or Tutor-Saliba s control. Estimates of the financial value of companies do not purport to be appraisals and do not necessarily reflect the prices at which companies actually may be sold.

The exchange ratio was determined through negotiation between Perini and Tutor-Saliba and the decision by Perini to enter into the merger was solely that of the Perini board of directors. UBS opinion and financial analyses were one of many factors considered by each of the Special Committee and the Perini board of directors in their respective evaluations of the merger and should not be viewed as determinative of the views of the Special Committee or the Perini board of directors with respect to the merger or the exchange ratio.

The following is a summary of the material financial analyses performed by UBS and reviewed with the Special Committee in connection with UBS opinion relating to the merger and presented to the Perini board of directors. The financial analyses summarized below include information presented in tabular format. In order to understand fully UBS financial analyses, the tables must be read together with the text of the summary. The tables alone do not constitute a complete description of the financial analyses. Considering the data below without considering the full narrative description of the financial analyses, including the methodologies and assumptions underlying the analyses, could create a misleading or incomplete view of UBS financial analyses.

#### Discounted Cash Flow Analyses

UBS performed a discounted cash flow analysis of (i) each of Perini and Tutor-Saliba using financial forecasts that the Special Committee directed UBS to utilize for purposes of its analysis for 2008 to 2012 (or, in the case of Tutor-Saliba s Black Construction business for 2008 to 2017) and (ii) the estimated after-tax synergies that Perini s management and the Special Committee estimated would result from the merger.

*Perini.* UBS performed a discounted cash flow analysis to calculate the estimated present value of the projected unlevered, after-tax free cash flow of Perini on a stand-alone basis from February 1, 2008 through December 31, 2012 using financial forecasts prepared by Perini s management and the Special Committee. UBS calculated a range of terminal values for Perini by applying terminal multiples of the last 12 months (LTM) earnings before interest, taxes, depreciation and amortization, commonly referred to as EBITDA, ranging from 6.0 to 8.0 to estimated EBITDA of Perini for the twelve months ending December 31, 2012. The cash flows and terminal values with respect to the Perini financial data were then discounted to present value as of January 31, 2008 using after-tax discount rates ranging from 12.5% to 14.5% based on its estimated weighted average cost of capital. This analysis indicated an implied stand-alone range of equity values for Perini common stock of approximately \$1,175 million to \$1,440 million and an implied stand-alone range of equity values per share of Perini common stock of approximately \$41.80 to \$51.30.

*Tutor-Saliba*. UBS performed a discounted cash flow analysis to calculate the estimated present value of (i) the projected unlevered, after-tax free cash flow of Tutor-Saliba s United States operations from February 1, 2008 through December 31, 2012 and (ii) the projected unlevered, after-tax free cash flow of Tutor-Saliba s Black Construction business from February 1, 2008 through December 31, 2017, in each case on a stand-alone basis using financial forecasts prepared by Perini s management, Tutor-Saliba s management and the Special Committee. UBS calculated a range of terminal values for Tutor-Saliba s United States operations by applying terminal LTM EBITDA multiples ranging from 6.0 to 8.0 to estimated EBITDA of Tutor-Saliba s United States operations for the twelve months ending December 31, 2012. UBS calculated a range of terminal values for Tutor-Saliba s Black Construction business by applying perpetuity growth rates ranging from 0.0% to 4.0% for periods subsequent to calendar year 2017. The cash flows and terminal values with respect to the Tutor-Saliba financial data were then discounted to present value as of January 31, 2008 using after-tax discount rates ranging from 12.5% to 14.5% based on its estimated weighted average cost of capital. The resulting implied stand-alone range of equity values for Tutor-Saliba s Black Construction business. This analysis indicated an implied stand-alone range of equity values for Tutor-Saliba s Black Construction business.

*Estimated Synergies.* UBS performed a discounted cash flow analysis to calculate the estimated present value of the estimated after-tax synergies that Perini s management, Tutor-Saliba s management and the Special Committee forecasted would result from the merger. For purposes of calculating a range of terminal values for the estimated after-tax synergies, UBS assumed a 0.0% perpetuity growth rate for estimated after-tax synergies for periods subsequent to calendar year 2012, the final year of the synergy forecasts provided to UBS. The cash flows and terminal values with respect to the estimated after-tax synergies were then discounted to present value as of January 31, 2008 using after-tax discount rates ranging from 12.5% to 14.5% based on the estimated weighted average cost of capital for Perini and Tutor-Saliba. This analysis indicated an implied range of values of the estimated after-tax synergies of approximately \$280 million.

UBS then calculated the implied range of equity values per share of Perini pro forma for the merger, by (i) summing the implied stand-alone range of equity values for Perini, the implied stand-alone range of equity values for Tutor-Saliba and the implied range of values of the estimated after-tax synergies and (ii) dividing that sum by the total number of shares of Perini common stock pro forma for the merger, based on the exchange ratio, to derive an implied range of equity values per share of Perini common stock pro forma for the merger of approximately \$43.30 to \$54.90, which represents an implied premium of approximately 3.6% to 7.0% to the implied stand-alone range of equity values per share of Perini common stock of approximately \$41.80 to \$51.30.

*Implied DCF Equity Value Contribution.* UBS performed an analysis to compare the implied relative contributions of Perini and Tutor-Saliba to the implied range of equity values of Perini pro forma for the merger. UBS calculated the relative contributions of Perini and Tutor-Saliba as a percentage of the implied range of equity values of Perini pro forma for the merger by dividing the implied stand-alone range of equity values for each of Perini and Tutor-Saliba described above, plus the estimated after-tax synergies that Perini s management and the Special Committee estimated would result from the merger attributable to each of Perini and Tutor-Saliba, by the implied range of equity values of Perini pro forma for the merger, assuming for purposes of the first calculation below that 55% of the estimated after-tax synergies that Perini s management and the Special Committee estimated would result from the merger were attributable to Perini, and 45% of such estimated after-tax synergies were attributable to Tutor-Saliba, and alternatively assuming for purposes of the second calculation below that 100% of the estimated after-tax synergies that Perini s management and the Special Committee estimated would result from the merger were attributable to Tutor-Saliba alone. This analysis indicated the following implied relative contributions of Perini and Tutor-Saliba to the implied range of equity values of Perini pro forma for the merger, after giving effect to the estimated after-tax synergies that Perini s

management and the Special Committee estimated would result from the merger, as compared to the approximately 45% pro forma ownership of the combined company by Tutor-Saliba shareholders, based on the exchange ratio:

		Perini		Tutor-Saliba	
Implied relative equity value contribution reference range (assumi	ng 55% of estimated present value of				
synergies attributable to Perini and 45% of estimated present value	e of synergies attributable to Tutor-Saliba)	53%	64%	36%	47%
Implied relative equity value contribution reference range (assumi	ng 100% of estimated present value of				
synergies attributable to Tutor-Saliba)		47%	57%	43%	53%
Other Factors					

In connection with rendering its opinion to the Special Committee, UBS also reviewed selected trading multiples of Perini on a stand-alone basis and selected trading multiples of selected publicly traded engineering and construction companies. Given the limited comparability of Perini, Tutor-Saliba and the proposed merger to other companies and transactions in the engineering and construction industry with publicly available data, UBS relied primarily on the discounted cash flow analyses summarized above for purposes of UBS opinion and not on the information presented below regarding the selected companies and selected transactions that UBS reviewed.

*Selected Public Companies.* In connection with rendering its opinion to the Special Committee, UBS reviewed selected trading multiples of Perini on a stand-alone basis and selected trading multiples of the following 12 companies: Comfort Systems USA, Inc.; EMCOR Group, Inc.; Granite Construction Incorporated; Insituform Technologies, Inc.; Layne Christensen Company; Matrix Service Company; Michael Baker Corporation; and Sterling Construction Company, Inc. (the preceding eight companies are referred to below as the Small Cap General & Specialty Contractors ); and Fluor Corporation; Jacobs Engineering Group Inc.; The Shaw Group Inc.; and URS Corporation (the preceding four companies are referred to below as the Other Engineering & Construction Firms ).

UBS reviewed, among other things, the enterprise values of the selected companies and Perini on a stand-alone basis, calculated as equity market value, plus debt at book value, preferred stock at liquidation value and minority interests at book value, less cash, as multiples of EBITDA for calendar year 2007 and estimated EBITDA for calendar years 2008 and 2009. UBS also reviewed closing stock prices of the selected companies and Perini on a stand-alone basis as a multiple of earnings per share, commonly referred to as EPS, for calendar year 2007 and estimated EDS for calendar years 2008 and 2009. Equity market value and multiples for the selected companies and Perini on a stand-alone basis were based on closing stock prices on April 1, 2008. Financial data for the selected companies were based on publicly available research analysts estimates, public filings and other publicly available information. Estimated financial data for Perini were provided to UBS by Perini s management and the Special Committee and were obtained from the Institutional Brokers Estimate System (I/B/E/S). This analysis indicated the following implied high, mean, median and low multiples for the selected public companies and for Perini on a stand-alone basis:

	Implied Multiples for Selected Public Companies				Implied Multiples for Perini on Stand-Alone Basis Special Committee I/B/E/S		
	High	Mean	Median	Low	Estimates	Estimates	
Enterprise Value as Multiple of EBITDA							
Small Cap General & Specialty Contractors:							
Calendar year 2007A	15.2x	8.4x	7.5x	4.8x	5.2x	5.2x	
Calendar year 2008E	10.0x	6.3x	5.5x	4.8x	5.0x	4.8x	
Calendar year 2009E	8.4x	5.5x	4.8x	4.5x	5.2x	4.9x	
Other Engineering & Construction Firms:							
Calendar year 2007A	20.6x	16.7x	16.9x	12.4x	5.2x	5.2x	
Calendar year 2008E	13.3x	10.7x	11.1x	7.3x	5.0x	4.8x	
Calendar year 2009E	11.0x	8.9x	9.0x	6.5x	5.2x	4.9x	
Price/EPS							
Small Cap General & Specialty Contractors:							
Calendar year 2008E	27.2x	14.1x	12.8x	6.4x	10.7x	10.2x	
Calendar year 2009E	21.3x	13.1x	11.5x	8.4x	11.4x	10.1x	
Other Engineering & Construction Firms:							
Calendar year 2008E	26.4x	20.4x	20.6x	14.2x	10.7x	10.2x	
Calendar year 2009E	21.5x	16.6x	16.6x	11.8x	11.4x	10.1x	

The selected public companies reviewed by UBS had limited comparability to Perini and Tutor-Saliba. The selection of the companies reviewed by UBS involved numerous complex subjective and qualitative judgments concerning financial and operating characteristics and other factors that UBS deemed appropriate. The selected public companies were chosen because they are either (i) U.S.-listed, U.S.-based construction companies or contractors that, based on their primary businesses, end markets, operations and size, as measured by revenue or market capitalization, could be considered, for purposes of analysis, generally to be in certain businesses or end markets in which Perini or Tutor-Saliba operates or (ii) larger, more diversified U.S.-listed, U.S.-based engineering and construction companies that have exposure to certain businesses, end markets and operations that, for purposes of analysis, could generally be considered to be in certain businesses or end markets in which Perini or Tutor-Saliba operates. UBS did not exclude any public company that UBS identified as meeting the above selection criteria. UBS did not identify any public company whose principal businesses, included, like Perini and Tutor-Saliba, construction services related to the hospitality and gaming industry in Las Vegas, Nevada, provided predominantly on a cost-plus basis. Given the limited comparability of Perini and Tutor-Saliba to other public companies in the engineering and construction industry, UBS did not draw any conclusions regarding the fairness from a financial point of view of the exchange ratio in the proposed merger from or with respect to the selected companies that UBS reviewed.

*Selected Transactions*. In connection with rendering its opinion to the Special Committee, UBS reviewed selected financial information for each of the following 12 selected transactions announced between June 11, 2002 and September 25, 2007: Hochtief Aktiengesellschaft/Flatiron Construction Corp.; Balfour Beauty PLC/Centex Corporation; Tenaska, Inc./InfrastruX Group; Energy*Solutions*, Inc./Duratek, Inc.; Dycom Industries, Inc./Prince Telecom Inc.; Perini/Rudolph and Sletten, Inc.; Ferrovial Agroman S.A./W.W. Webber, Inc.; Layne Christensen Company/Reynolds Inc.; Worley Group Limited/Parsons E&C Corporation; Madison Dearborn Partners, LLC/Great Lakes Dredge & Dock Company; Bilfinger Berger AG/Abigroup Limited; and Koninklijke BAM Groep NV/Hollandsche Beton Groep N.V.

UBS reviewed transaction values in the selected transactions, calculated as the purchase price paid for the target company s equity, plus debt at book value, preferred stock at liquidation value and minority interests at book value, less cash, as a multiple of LTM revenue, LTM EBITDA and LTM earnings before interest and taxes, commonly referred to as EBIT, of the target company, to the extent such data were publicly available at the time of announcement of the relevant transaction. This analysis indicated the following implied high, mean, median and low multiples for the selected transactions:

	Impli	Implied Multiples for Selected				
	High		actions Median	Low		
Purchase price as a multiple of LTM revenue	1.3x	0.6x	0.6x	0.1x		
Purchase price as a multiple of LTM EBITDA	9.6x	7.5x	8.0x	3.9x		
Purchase price as a multiple of LTM EBIT	24.1x	12.0x	10.7x	6.0x		

The selected transactions reviewed by UBS had limited comparability to the proposed merger. The selection of transactions reviewed by UBS involved numerous complex subjective and qualitative judgments concerning financial and operating characteristics and other factors that UBS deemed appropriate. The selected transactions were chosen because they involved construction companies or contractors that were acquired in transactions with publicly available data announced in the six years prior to the date of the UBS opinion and, based on their primary businesses, end markets and operations, as measured by revenue or purchase price, could be considered, for purposes of analysis, generally to be in certain businesses or end markets in which Perini or Tutor-Saliba operates. UBS did not exclude any transaction with publicly available data that UBS identified as meeting the above selection criteria. UBS did not identify any transaction with publicly available data announced in the six years prior to the date of the UBS opinion that involved the acquisition of a company whose principal businesses included, like Perini and Tutor-Saliba, construction services related to the hospitality and gaming industry in Las Vegas, Nevada provided predominantly on a cost-plus basis. Given the limited comparability of Perini, Tutor-Saliba and the proposed merger to other companies and transactions in the engineering and construction industry with publicly available data, UBS did not draw any conclusions regarding the fairness from a financial point to view of the exchange ratio in the proposed merger from or with respect to the selected transactions that UBS reviewed.

*Miscellaneous.* Under the terms of UBS engagement, Perini has agreed to pay UBS a fee of \$8.25 million for its services in connection with the merger, a portion of which was payable in connection with rendering its opinion and \$6.05 million of which is contingent upon consummation of the merger. In addition, Perini has agreed to reimburse UBS for its reasonable expenses, including fees, disbursements and other charges of counsel, and to indemnify UBS and related parties against liabilities, including liabilities under federal securities laws, relating to, or arising out of, its engagement. In the past, UBS and its affiliates have provided investment banking services to Perini and Tutor-Saliba unrelated to the proposed merger, for which UBS and its affiliates received compensation, including having acted as underwriter in an offering of 10,341,909 shares of Perini common stock in December 2005, including 4,240,229 shares of Perini common stock that were beneficially owned by Tutor-Saliba and Ronald N. Tutor, and as agent in September 2006 in a private placement of 400,000 shares of Perini common stock that were beneficially owned by Tutor-Saliba and Ronald N. Tutor. In addition to the fees Perini has agreed to pay UBS for its services in connection with the merger, UBS has received approximately \$4.4 million of fees for investment



banking services provided to Perini since December 2005. UBS and its affiliates may also provide investment banking services to Perini and Tutor-Saliba in the future, for which UBS and its affiliates may receive compensation, although there are no such services currently contemplated. In the ordinary course of business, UBS and its affiliates may hold or trade, for their own accounts and the accounts of their customers, securities of Perini and, accordingly, may at any time hold a long or short position in such securities.

The Special Committee selected UBS as its financial advisor in connection with the merger because UBS is an internationally recognized investment-banking firm with substantial experience in similar transactions and in the construction industry. UBS is continually engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, leveraged buy-outs, negotiated underwritings, competitive bids, secondary distributions of listed and unlisted securities and private placements.

#### **Anticipated Accounting Treatment**

The merger will be accounted for using the purchase method of accounting in accordance with accounting principles generally accepted in the United States under Statement of Financial Accounting Standards No. 141, *Business Combinations*. Perini will be the acquiring entity for financial reporting purposes. Under the purchase method of accounting, the cost of the transaction will be allocated to the tangible and intangible assets and liabilities assumed by the acquired entity based on their estimated fair values, with any excess being recognized as goodwill. Under Statement of Financial Accounting Standards No. 142, *Goodwill and Other Intangible Assets*, goodwill will not be amortized, but will be subject to an annual impairment test.

#### Material United States Federal Income Tax Consequences to Existing Perini Shareholders

The following is a summary of material United States federal income tax considerations relating to the merger, but does not purport to be a complete analysis of all potential tax considerations. This summary is based on the provisions of the United States Internal Revenue Code of 1986, as amended, the Treasury regulations promulgated thereunder, judicial authority, published administrative positions of the United States Internal Revenue Service (IRS) and other applicable authorities, all as in effect on the date of this document, and all of which are subject to change, possibly on a retroactive basis. We have not sought any ruling from the IRS with respect to the statements made and the conclusions reached in the following summary and there can be no assurance that the IRS will agree with our statements and conclusions.

This summary deals only with beneficial owners of shares of common stock in Perini that hold such shares of stock as capital assets within the meaning of section 1221 of the Internal Revenue Code (generally, property held for investment). This summary does not purport to deal with all aspects of United States federal income taxation that might be relevant to particular holders in light of their personal investment circumstances or status, nor does it address tax considerations applicable to holders that may be subject to special tax rules, such as certain financial institutions, tax-exempt organizations, subchapter S corporations, partnerships or other pass-through entities for United States federal income tax purposes or investors in such entities, insurance companies, broker-dealers, dealers or traders in securities or currencies, certain former citizens or residents of the United States subject to section 877 of the Internal Revenue Code and taxpayers subject to the alternative minimum tax. This summary also does not discuss shares of Perini common stock held as part of a hedge, straddle, synthetic security or conversion transaction, or situations in which the functional currency of a holder is not the United States dollar. Moreover, the effect of any applicable federal estate or gift, state, local or non-United States tax laws is not discussed.

The following discussion is not a substitute for careful tax planning and advice. Each holder of shares of Perini common stock should consult its own tax advisors with respect to the application of the United States federal income tax laws to its particular situation, as well as any tax consequences arising under the federal estate or gift tax laws or the laws of any state, local or non-United States taxing jurisdiction or under any applicable tax treaty.

At the closing of the merger, Kirkland & Ellis LLP, counsel to the Special Committee of the board of directors of Perini, will deliver to Perini its legal opinion, and Latham & Watkins LLP, counsel to Tutor-Saliba, will deliver to Tutor-Saliba its legal opinion each to the effect that, on the basis of the facts, assumptions and representations set forth in such opinion and the representations and covenants set forth in certificates obtained from officers of Perini, Trifecta Acquisition LLC and Tutor-Saliba, the merger will be treated as a tax-free reorganization within the meaning of Section 368(a) of the Internal Revenue Code. These opinions will be based on the law in effect on the date that these opinions are rendered. Any change in currently applicable law, which may or may not be retroactive, or the failure of any factual representation or assumption to be true, correct and complete in all material respects, could affect the validity of the Kirkland & Ellis LLP opinion and/or the Latham & Watkins LLP opinion. An opinion of counsel represents counsel s best legal judgment and is not binding on the Internal Revenue Service or on any court.

It is a condition to the obligations of Perini, Trifecta Acquisition and Tutor-Saliba to consummate the merger that Perini receive the opinion of Kirkland & Ellis LLP, and that Tutor-Saliba receive the opinion of Latham & Watkins LLP, both to the effect that the merger will be treated as a tax-free reorganization within the meaning of Section 368(a) of the Internal Revenue Code. In accordance with the tax opinions:

Perini shareholders will not recognize any gain or loss in the merger;

neither Perini nor Tutor-Saliba will recognize any gain or loss in the merger;

Perini will take a basis in the assets of Tutor-Saliba equal to the basis of those assets in the hands of Tutor-Saliba; and

Perini s holding period in the assets of Tutor-Saliba will include the holding period of Tutor-Saliba in those assets. Amendment to Perini Articles of Organization

The Perini board of directors has adopted, subject to shareholder approval and subject to the completion of the merger, a resolution recommending that Perini s amended and restated articles of organization be amended to increase the number of authorized shares of Perini common stock to 75 million from 40 million shares. A copy of the proposed amendment to our amended and restated articles of organization is attached as *Annex F* to this proxy statement.

The increase in the number of authorized shares of common stock is recommended by the Perini board of directors to provide sufficient common stock for issuance of common stock to the Tutor-Saliba shareholders in connection with the merger. Perini presently is authorized to issue 40 million shares of common stock. As of the record date, approximately [] shares of Perini common stock were issued and outstanding and approximately [] shares of common stock were issuable upon exercise of stock options or upon vesting of restricted stock units. At the effective time, we will issue 22,987,293 shares of Perini common stock to the Tutor-Saliba shareholders as consideration for the merger.

The actions contemplated by Proposal 2, even if approved by our shareholders, will not occur unless we complete the merger.

Except for the issuance of Perini common stock in connection with the merger, Perini s management has no arrangements, agreements, understandings or plans at the present time to issue additional shares of common stock for any purpose.

#### **Regulatory Approval**

Under the Hart-Scott-Rodino Antitrust Improvement Act of 1976, or HSR Act, and the rules and regulations promulgated thereunder, certain transactions, including the merger between Perini and Tutor-Saliba, may not be

consummated until required information and materials have been furnished to the Department of Justice, or DOJ, and the Federal Trade Commission, or FTC, and certain waiting period requirements have expired or been terminated. On April 16, 2008, each of Perini and Tutor-Saliba filed a Pre-merger Notification and Report Form pursuant to the HSR Act with the DOJ and FTC. The statutory waiting period under the HSR Act expired on May 16, 2008. At any time before the completion of the merger, the DOJ, the FTC or others could take action under the antitrust laws with respect to the merger, including to seek to enjoin the completion of the merger, to rescind the merger or to require divestiture of certain assets of Perini or Tutor-Saliba. There can be no assurance that a challenge to the acquisition on antitrust grounds will not be made or, if such challenge is made, that it would not be successful.

#### **Federal Securities Law Consequences**

In the merger, we will issue a total of approximately 22.1 million shares of Perini common stock to the two trusts controlled by Mr. Tutor and approximately 900,000 shares to the other shareholders of Tutor-Saliba. None of these shares will be registered under the Securities Act of 1933. It is intended that such shares will be issued pursuant to a private placement exemption under Section 4(2) of the Securities Act or other available exemptions, and they will only be able to be resold pursuant to a separate registration statement or an applicable exemption from registration (under both federal and state securities laws). The shares that will be received will be subject to contractual restrictions under the terms of the Shareholders Agreement. These restrictions are described in detail in The Shareholders Agreement Transfer Restrictions beginning on page 88.

#### NYSE Listing

It is a condition to the merger that the shares of Perini common stock to be issued in the merger be approved for listing on the NYSE, subject to official notice of issuance. Shares of Perini common stock will continue to be traded on the NYSE under the symbol PCR immediately following the completion of the merger.

#### **Appraisal Rights**

Holders of Perini common stock do not have dissenters or appraisal rights under Massachusetts law in connection with the merger.

Holders of Tutor-Saliba common stock have waived any dissenter s or appraisal rights they may have under California law in connection with the merger by the delivery of a unanimous written consent in favor of the approval of the Merger Agreement and the transactions contemplated thereby.

#### **Business Relationships Between Perini and Tutor-Saliba**

#### **Management Agreement**

In January 1997, we entered into a management agreement with Tutor-Saliba, and Mr. Tutor, in his capacity as chief executive officer of Tutor-Saliba, to provide certain management services. The management agreement has been renewed annually by our Compensation Committee, which consists entirely of independent directors, under the same basic terms and conditions as the initial agreement except that the amount of the fee payable thereunder by us to Tutor-Saliba has been increased. In March of 2007, the Compensation Committee voted to increase the annual payment to \$1 million.

We are entitled to terminate the management agreement at any time upon the determination by our board of directors that Mr. Tutor has not fulfilled his obligations under the agreement, or that it is inadvisable for Mr. Tutor to continue performing services for us pursuant to the agreement. We expect that the management agreement will be terminated upon the completion of the merger by mutual agreement of the parties.

#### Joint Ventures

Historically, we have participated in joint ventures with Tutor-Saliba and currently participate in certain joint ventures with Tutor-Saliba. Some of these joint venture projects have been managed by Tutor-Saliba and others have been managed by us. These joint ventures generated total revenues of \$96 million in 2007, of which our share contributed \$70.6 million to our consolidated revenues for the fiscal year ended December 31, 2007. The joint ventures generally act as an entity through a management committee comprised of the joint venture members, with voting proportionate to ownership interest and with one party named to manage the day-to-day operations of the joint venture.

The following are summaries of the joint ventures between Tutor-Saliba and us that are currently involved in construction activity.

*Tappan Zee Bridge Repair*: On November 4, 2005, we entered into a joint venture agreement with Tutor-Saliba for the purpose of submitting a joint bid and obtaining the contract for the repair of the Tappan Zee Bridge on the New York State Thruway. Our proportionate share of the joint venture is 70%. Tutor-Saliba holds a 30% interest in the joint venture. Pursuant to the agreement, we receive a management fee of \$15,000 per month in addition to our proportionate share in consideration for managing the joint venture. The aggregate amount payable to the joint venture under this contract is \$163.7 million. The construction contract calls for completion of the construction process by October 3, 2008.

*Brooklyn-Queens Expressway Rehabilitation*: On November 5, 2004, we entered into a joint venture agreement with Tutor-Saliba for the purpose of submitting a joint bid and obtaining the contract for the reconstruction of Interstate 278 (the Brooklyn-Queens Expressway). Our proportionate share of the joint venture is 80%. Tutor-Saliba holds a 20% interest in the joint venture. Pursuant to the agreement, we receive a management fee of \$27,500 per month in addition to our proportionate share in consideration for managing the joint venture. The aggregate amount payable to the joint venture under the contract is \$123 million. The agreement terminates on August 31, 2008, which date is the date the project is required to be completed.

*Airtran Terminal at Jamaica Station*: On August 22, 2000, we entered into a joint venture agreement with Tutor-Saliba for the purpose of submitting a joint bid and obtaining the contract for the construction of the Airtran Terminal at Jamaica Station. Our proportionate share of the joint venture is 50%. Tutor-Saliba holds a 50% interest in the joint venture. The aggregate amount payable to the joint venture under the contract is \$376 million. The construction work is completed; however, the joint venture is entitled to \$3.2 million of revenue backlog in connection with the project.

In addition to the joint ventures listed above, we entered into joint ventures with Tutor-Saliba during fiscal years 2005 and 2006 to submit bids on contracts for the construction of the Vegas 888 condominium project in Las Vegas and the Croton Water Treatment Plant at the Mosolu Golf Course in New York. Neither of these joint ventures secured a construction contract.

In fiscal years 2000 and 2001, we entered into joint venture agreements with Tutor-Saliba for the purpose of submitting joint bids and obtaining contracts to construct the UCLA Westwood Replacement Hospital and the UCLA Santa Monica/Orthopaedic Replacement Hospital, respectively. Subsequently, Tutor-Saliba purchased our interests in these two joint ventures and agreed to indemnify us from losses we might suffer as a result of participating in the bidding process. As of the date of this proxy, we have not suffered any losses or sought indemnification from Tutor-Saliba in connection with these two projects.

#### **Business Relationship with Desert Plumbing**

Following Tutor-Saliba s acquisition in January 2008 of Desert Plumbing & Heating Co., Inc., a subcontractor with whom Perini has had a business relationship for several years, Perini has continued to subcontract with Desert Plumbing on the same commercial third party terms as existed prior to the acquisition.

For additional information on Perini s business relationship with Desert Plumbing, please see Certain Relationships and Related Party Transaction Subcontractor beginning of page 141.

#### Additional Interests of Directors, Executive Officers and Certain Beneficial Owners

In considering the Perini board of directors recommendation to approve Proposal 1 and Proposal 2 in connection with the proposed merger, our shareholders should be aware that Mr. Tutor, who is our chairman and chief executive officer, has interests that are different from, and in addition to, the interests of Perini shareholders generally. These interests present actual conflicts of interest, and these interests are described below. The Special Committee and the Perini board of directors were aware of these conflicts of interest and considered them, among other matters, in reaching their decisions, as applicable, to approve the Merger Agreement and the merger and to recommend that our shareholders vote in favor of adopting Proposal 1 and Proposal 2 in connection with the merger. These conflicts of interest were the reason why our board of directors formed the Special Committee, as discussed above in Background of the Merger beginning on page 36. In addition, in light of these conflicts of interest, Mr. Tutor did not participate on behalf of Perini in any of the deliberations of the Perini board of directors or in any vote by the directors on any matters relating to the proposed merger, the Merger Agreement or the related agreements. He did, however, negotiate on behalf of Tutor-Saliba and its shareholders and express his views on the prospects of both the Perini and Tutor-Saliba businesses as and when requested by the Special Committee and its advisors.

#### **Beneficial Ownership of Perini Common Stock**

Mr. Tutor, through two trusts that he controls, is the beneficial owner of approximately 96% of the outstanding shares of Tutor-Saliba common stock. In the merger, these trusts will receive approximately 22.1 million shares of Perini common stock (representing approximately 96% of the shares Perini common stock issued in the merger) with a value of approximately \$845.6 million based on the closing price per share of Perini common stock of \$38.25 on April 2, 2008. As a result of the merger, Mr. Tutor, through these two trusts, will be the beneficial owner of approximately 43% of the outstanding shares of Perini common stock upon completion of the merger. These shares will be subject to the restrictions contained in the Shareholders Agreement. They also will have the benefit of the registration rights provided for in that agreement. Please see The Shareholders Agreement beginning on page 86 for a detailed summary of the terms of the Shareholders Agreement. A copy of the Shareholders Agreement is attached as *Annex B* to this proxy statement.

#### **Pre-Closing Distribution of Property**

In connection with the completion of the merger, Tutor-Saliba has distributed or will distribute to its shareholders (and Perini will not acquire in the merger) the following assets:

all of its interest in a residence in Ketchum, Idaho, with a market value of \$3.5 million;

cash for tax payments as described below; and

additional aggregate dividends of up to \$120 million of cash and notes (of which \$10 million in cash has already been distributed), subject to limitations contained in the Merger Agreement.

Because Tutor-Saliba is a subchapter S corporation, its taxable income is attributed to the shareholders of Tutor-Saliba for federal income tax purposes meaning that its shareholders are responsible for paying the income taxes on their proportionate share of the income. The Merger Agreement permits Tutor-Saliba to make cash distributions to its shareholders that are intended to cover their income tax obligations for Tutor-Saliba s income from January 1, 2008 through the completion of the merger. The combined company will retain such income. These distributions for taxes will not take into account any taxes with respect to income or gain realized in respect of the transactions contemplated by the Merger Agreement or otherwise not in the ordinary course of Tutor-Saliba s business or any gain realized on the distribution of appreciated assets, the taxes for which will be solely the responsibility of the Tutor-Saliba shareholders. In April 2008, Tutor-Saliba distributed approximately

\$11.6 million of cash to its shareholders in respect of their April 2008 income tax payment obligations arising from Tutor-Saliba s operations. Shortly before completion of the merger, the balance of the permissible tax distributions will be determined, and Tutor-Saliba will make additional cash distributions to its shareholders that are intended to cover their income tax obligations not covered by the April distribution. For a discussion of the reasons for the additional contemplated distribution, which will be Tutor-Saliba s final Subchapter S distributions, please see Subchapter S Dividend; Termination of Subchapter S Status beginning on page 95.

The Merger Agreement acknowledges that Perini will not acquire in the merger Tutor-Saliba s previously owned interests in a high-rise office building in San Pedro, California, commonly known as Pacific Place. Tutor-Saliba and an entity owned by Mr. Tutor completed an exchange transaction that resulted in Mr. Tutor s entity receiving a distribution of Tutor-Saliba s interests in Pacific Place and related assets and liabilities, with an aggregate net value (in excess of the assets received by Tutor-Saliba in the transaction) of \$21.3 million, as described more fully in Recent Developments and Expected 2008 Events Distribution of Commercial Real Estate; Distribution of Residential Real Estate and in Note 14 of the Audited Financial Statements of Tutor-Saliba attached as *Annex H* to this proxy statement.

The two trusts that Mr. Tutor controls have received or will receive distributions from Tutor-Saliba, with an aggregate net value to Mr. Tutor of to approximately \$153.3 million. To the extent these trusts receive notes as dividends, after completion of the merger, they will be creditors of Tutor-Saliba (which will be a wholly owned subsidiary of Perini) and will be entitled to receive payments of principal and interest from Tutor-Saliba. Pursuant to the Merger Agreement, these notes will include the following terms:

interest will accrue on the principal amount of the notes at a rate of 5% per annum;

the unpaid principal balance of the notes and all accrued and unpaid interest thereupon will become due and payable in full on June 30, 2012;

the notes may be prepaid at any time; and

Tutor-Saliba will be required to prepay the notes with 100% of excess cash flow of Tutor-Saliba (as described in the note) for the preceding fiscal year (or, for the year in which the closing of the merger occurs, the partial fiscal year from the closing through the end of such fiscal year) within 90 days of the end of each fiscal year until the principal and interest of the notes has been repaid.

# The Employment Agreement

In connection with the merger, we executed the Employment Agreement with Mr. Tutor. This agreement will take effect only upon completion of the merger. Please see The Employment Agreement beginning on page 89 for a detailed summary of the key terms of Mr. Tutor s Employment Agreement. The compensation and other benefits he will receive under the Employment Agreement provide him with additional interests in the completion of the merger.

#### Nominees to Perini Board of Directors

The Shareholders Agreement provides that the shareholder representative under the Shareholders Agreement (which will initially be Mr. Tutor) has the right to designate up to two nominees for election to the Perini board of directors (subject to certain conditions and limitations) and that, for so long as he serves as the chief executive officer of Perini, Mr. Tutor will be nominated for election to the Perini board of directors. Mr. Tutor has designated C.L. Max Nikias as one of the shareholder representative s board designees, and he will be appointed as a new member of the Perini board of directors upon the completion of the merger. As of the date of this proxy statement, Mr. Tutor has not elected to exercise his right to nominate a second director for election, although he has not waived the right to do so in the future. Please see The Shareholders Agreement Composition of the Board of Directors beginning on page 86 for a detailed summary of Mr. Tutor s rights with respect to the composition of the Perini board of directors.

#### **Personal Guaranties**

Pursuant to the Merger Agreement, we have agreed to cooperate with Tutor-Saliba to obtain consents and seek amendments to certain guaranties of surety bonds and banking agreements issued by Mr. Tutor in his personal capacity on behalf of Tutor-Saliba to remove Mr. Tutor as an obligor, guarantor or surety. As of June 30, 2008, Mr. Tutor had aggregate guarantee obligations to banking institutions for approximately \$115 million for the benefit of Tutor-Saliba. Mr. Tutor s indemnity with respect to surety bonds represents additional contingent obligations that fluctuate based on the value of operating contracts relating to the bonds in effect from time to time. As of the date of this proxy statement, we are not aware of any defaults under these surety bonds.

#### **Interests of Other Directors and Executive Officers**

In connection with the merger, none of our directors or executive officers will receive any transaction bonuses, none of their existing equity awards will vest or become payable on an accelerated basis, and no director or executive officer has any change of control arrangement under an employment agreement or any pension or other benefit plan that would entitle such director or officer to additional compensation or other benefits.

Although we expect to consider implementing new employment and incentive compensation arrangements for certain members of our management team and of Tutor-Saliba s management team (possibly including employment agreements), we have not had any discussions with any of our executive officers regarding such matters. Such matters, if pursued, are subject to consideration by the Compensation Committee of our board of directors, evaluation by the full board of directors and, if proposed, negotiation with the applicable management members. We may implement new employment and incentive compensation arrangements prior to the closing of the merger.

As described above, the Perini board of directors created a Special Committee to explore and evaluate potential strategic transactions that might be available to Perini, including in particular a business combination transaction with Tutor-Saliba, and to discuss and negotiate the terms of any such transactions. The members of the Special Committee were Messrs. Klein, Brittain, Arkley and Kennedy, with Mr. Klein serving as the Chair of the committee. The Perini board of directors agreed to pay the members of the Special Committee a fee of \$50,000 (\$60,000 for the Chair) plus \$2,000 per meeting attended or each other day on which they spent more than three hours working on any such transaction. The Perini board of directors determined that these fees were reasonable and customary for directors serving on a special committee in connection with this type of proposed transaction. The fees that the directors receive for serving on the Special Committee are not contingent upon success of the proposed transaction.

#### Litigation Related to the Merger

#### Weitman v. Tutor.

Ms. Nina Weitman, purporting to be a shareholder of Perini, seeks to bring a class action in the Massachusetts Superior Court Department of the Trial Court of Middlesex County on behalf of all shareholders of Perini regarding the merger, in a complaint dated June 19, 2008. The complaint alleges that Perini, its board of directors and Tutor-Saliba breached fiduciary duties to Perini in agreeing to enter into the Merger Agreement. Specifically, the complaint alleges that the proxy statement does not provide shareholders with enough information regarding the merger; the exchange ratio is not fair to the Perini shareholders; and the Perini board of directors breached its fiduciary duties by, among other things, failing to examine strategic alternatives to the proposed merger. The complaint seeks, among other forms of relief, certification of the case as a class action, injunctive relief to enjoin the merger, rescission in the event that the merger is consummated before a judgment in the case is entered and damages.

The plaintiff has filed a motion seeking expedited procedures for its lawsuit. Perini has served plaintiff with a motion to dismiss the complaint as to Perini. Tutor-Saliba has also served plaintiff with a motion to dismiss the complaint as to Tutor-Saliba. As of July 24, 2008, all three motions are still pending.

Perini plans to defend itself vigorously against this lawsuit. Additional lawsuits pertaining to the proposed merger could be filed in the future. Although we are unable at this time to determine the ultimate outcome of this lawsuit or any future similar lawsuits, injunctive relief or an adverse determination could affect our ability to complete the merger.

#### The Merger Agreement

The following is a summary of the material terms of the Merger Agreement. This summary does not purport to describe all the terms of the Merger Agreement and is qualified by reference to the complete Merger Agreement, the executed version of which is attached as Annex A to this proxy statement and an amendment thereto is attached as Annex AA to this proxy statement, both of which are incorporated by reference in this proxy statement. All shareholders of Perini are urged to read the Merger Agreement carefully and in its entirety.

The Merger Agreement is being summarized in this proxy statement and has been included as an annex to this proxy statement to provide you with information regarding its terms. The Merger Agreement contains representations and warranties that the parties thereto made to each other as of specific dates. The assertions embodied in those representations and warranties were made solely for purposes of the contract between and the transactions and agreements contemplated thereby among the respective parties thereto and may be subject to important qualifications and limitations agreed to by Perini and Tutor-Saliba in connection with negotiating the terms thereof. Moreover, some of those representations and warranties may not be accurate or complete as of any specified date, may be subject to a contractual standard of materiality different from those generally applicable to shareholders or may have been used for the purpose of allocating risk among the parties to the Merger Agreement rather than establishing matters as facts. Perini will provide additional disclosure in its public reports to the extent that it is aware of the existence of any material facts that are required to be disclosed under federal securities law and that might otherwise contradict the representations and warranties contained in the Merger Agreement and will update such disclosure as required by federal securities laws.

#### General

The Merger Agreement provides for the merger of Tutor-Saliba with and into Trifecta Acquisition LLC, a newly formed, wholly owned subsidiary of Perini, which is referred to in this proxy statement as Merger Sub, with Merger Sub continuing as the surviving entity. As a result of the merger, Merger Sub will continue as a wholly owned subsidiary of Perini.

#### **Closing Matters**

*Closing*. Unless the parties agree otherwise, the closing of the merger will take place on the third business day after all closing conditions have been satisfied or waived (other than any condition that by its nature cannot be satisfied until the closing), unless the Merger Agreement has been terminated. Please see Conditions to the Completion of the Merger beginning on page 83 for a more complete description of the conditions that must be satisfied or waived prior to the closing of the merger.

*Effective Time of the Merger*. As soon as practicable after the satisfaction or waiver of the conditions to the merger, we and Tutor-Saliba will file a certificate of merger with the Secretary of State of the State of California in accordance with the relevant provisions of the General Corporation Law of the State of California and the Limited Liability Company Act of the State of California and make all other required filings or recordings. The merger will become effective when the certificate of merger is filed or at such later time as Tutor-Saliba and we agree and specify in the certificate of merger.

The parties are working to complete the merger by the third quarter of 2008. However, because completion of the merger is subject to the receipt of regulatory approvals and the satisfaction or waiver of other conditions, we cannot predict the actual timing of the completion of the merger.

Amendment to Perini Articles of Organization. At the closing of the merger, we will amend our amended and restated articles of organization to increase the number of authorized shares of Perini common stock by filing articles of amendment with the Secretary of State of the State of Massachusetts in accordance with applicable provisions of the Massachusetts Business Corporations Act. The full text of the amendment to the amended and restated articles of organization of Perini is attached as *Annex F* to this proxy statement.

*Composition of the Perini Board of Directors at the Effective Time.* The parties will take all actions required such that as of the effective time of the merger the Perini board of directors composition will be in compliance with the Shareholders Agreement. Please see The Shareholders Agreement Composition of the Board of Directors beginning on page 86 for additional information on the composition of the Perini board of directors following the closing of the merger.

#### Consideration to be Paid Pursuant to the Merger

The Merger Agreement provides that, at the completion of the merger, each share of Tutor-Saliba common stock issued and outstanding immediately prior to the completion of the merger, but excluding shares of Tutor-Saliba common stock owned by us or Tutor-Saliba or any of our or their respective subsidiaries, will be converted into the right to receive a number of shares of Perini common stock equal to the quotient obtained by dividing (i) 22,987,293 shares of Perini common stock by (ii) the total number of shares of Tutor-Saliba common stock issued and outstanding as of immediately prior to the effective time of the merger. The number of shares of Perini common stock is subject to proportional adjustment in the event of certain significant changes to the capital stock of Perini prior to the closing of the merger.

#### Exchange of Tutor-Saliba Stock Certificates Pursuant to the Merger

Before the closing of the merger, we may either appoint a banking or financial institution mutually agreed to by us and Tutor-Saliba as the exchange agent to effectuate the exchange of Tutor-Saliba stock certificates for certificates representing shares of Perini common stock, or alternatively we or any of our subsidiaries may serve as the exchange agent with the consent of Tutor-Saliba. As soon as reasonably practicable after the closing of the merger, the exchange agent will send a letter of transmittal to each former Tutor-Saliba shareholder who holds one or more stock certificates. The letter of transmittal will contain instructions explaining the procedure for surrendering Tutor-Saliba stock certificates to the exchange agent.

Tutor-Saliba shareholders who surrender their stock certificates, together with a properly completed letter of transmittal, will receive a certificate representing that number of whole shares of Perini common stock into which their shares of Tutor-Saliba common stock were converted pursuant to the merger.

After the merger, each certificate that previously represented shares of Tutor-Saliba common stock will only represent the right to receive certificates representing the shares of Perini common stock into which those shares of Tutor-Saliba common stock have been converted; and dividends or other distributions, if any, of Perini common stock which Tutor-Saliba shareholders are entitled to under the terms of the Merger Agreement.

We will not pay dividends declared with a record date after the completion of the merger to any holder of any Tutor-Saliba stock certificates until the holder surrenders the Tutor-Saliba stock certificates. However, once those certificates are surrendered, we will pay to the holder, without interest, any dividends that have been declared and paid after the closing date of the merger on the shares into which those Tutor-Saliba shares have been converted.

After the completion of the merger, Tutor-Saliba will not register any transfers of shares of Tutor-Saliba common stock.

#### Covenants

Perini and Tutor-Saliba have each undertaken certain covenants in the Merger Agreement concerning the conduct of their respective businesses between the date the Merger Agreement was signed and the completion of the merger or the termination of the Merger Agreement. The following summarizes the more significant of these covenants:

*No Solicitation.* Each of Tutor-Saliba and Perini has agreed that it will not, and will not permit its respective officers, directors, employees, agents or representatives to:

solicit, initiate, knowingly encourage or knowingly induce the making, submission or announcement of, an acquisition proposal of the type described below;

furnish any non-public information in connection with an acquisition proposal;

participate or engage in discussions or negotiations with any person with respect to an acquisition proposal; or

enter into any agreement, letter of intent or other similar instrument with any person providing for an acquisition transaction. Each party agreed to provide notice to the other on receipt of any acquisition proposal or any request for information that is reasonably expected to lead to an acquisition proposal within one business day of receiving such acquisition proposal or request and keep the other party informed of developments related to the acquisition proposal or request.

Notwithstanding these restrictions, pursuant to the Merger Agreement, Perini is permitted to participate in negotiations with any person that has made a bona fide acquisition proposal and/or and furnish non-public information related to Perini pursuant to a confidentiality agreement if:

the acquisition proposal was received other than as a result of a material breach of the restrictions on solicitation activities noted above;

the Special Committee of Perini board of directors reasonably determines in good faith that an acquisition proposal constitutes or is reasonably likely to lead to a superior proposal of the type described below;

Perini has provided 48 hours of prior written notice to Tutor-Saliba of its intention to engage or participate in such activities; and

contemporaneously with furnishing any non-public information to such person, Perini furnishes the non-public information to Tutor-Saliba if it has not been previously made available to Tutor-Saliba.

An *acquisition proposal* means any bona fide inquiry, proposal or offer from any person relating to:

any direct or indirect acquisition or purchase of more than 20% of the total outstanding voting securities of the party or any of its subsidiaries or any tender offer or exchange offer that if completed would result in any person beneficially owning 20% or more of the total outstanding voting securities of the party or any of its subsidiaries;

any merger, consolidation, business combination or other similar transaction involving the party or any of its subsidiaries pursuant to which the shareholders of such party immediately prior to the consummation of such transaction would hold less than 80% of the equity interests in the surviving or resulting entity of such transaction immediately after consummation thereof;

any sale, exchange, transfer, license, acquisition or disposition (in each case other than in the ordinary course of business) of more than 20% of the assets of the party and/or its subsidiaries; or

any liquidation, dissolution, recapitalization or other significant corporate reorganization of the party.

In addition, any issuance of shares of Tutor-Saliba or transaction that would be prohibited by the restrictions described in Tutor-Saliba Shareholder Lock-Up below.

A *superior proposal* means any *bona fide* acquisition proposal (as defined above, provided however that 20% and 80% should be substituted for all references to 45% and 55% in the definition above) which the Perini board of directors determines in good faith (after consultation with a financial advisor of nationally-recognized reputation and its outside legal counsel) to be more favorable to Perini s shareholders than the transactions contemplated by the Merger Agreement after taking into account, among other things, all financial considerations and aspects of the acquisition proposal, all strategic considerations, all legal and regulatory considerations, the identity of the offeror, the conditions and likelihood of completion, the likelihood of the imposition of obligations in connection with obtaining necessary regulatory approval, any revisions to the terms of the Merger Agreement proposed by Tutor-Saliba and any other factors deemed by Perini board of directors in good faith to be appropriate.

*Perini Board of Directors* Covenant to Recommend. The Perini board of directors has recommended that Perini shareholders approve the share issuance proposal and the amendment proposal in connection with the merger.

The Perini board of directors agreed in the Merger Agreement that it may not withdraw or modify its recommendation to its shareholders or approve or adopt, or enter into any agreement providing for, an acquisition proposal unless (i) it determines in good faith, after consulting with outside counsel, that failure to take such action would be reasonably likely to result in a breach of the fiduciary obligations of its board of directors and (ii) if such change of recommendation is for the purposes of accepting or adopting an acquisition proposal or entering into an acquisition agreement (A) such acquisition proposal constitutes a superior proposal and (B) and the Perini board of directors has provided to Tutor-Saliba three business days prior written notice of its intent to change recommendation and, if requested by Tutor-Saliba, negotiated in good faith with Tutor-Saliba during such three business day period regarding revisions to the Merger Agreement that would avoid such change of recommendation.

Notwithstanding the foregoing, unless the Merger Agreement is terminated prior to the annual meeting, Perini is required to hold the annual meeting and permit its shareholders to vote on the merger proposals (Proposal 1 and Proposal 2), even if Perini receives an unsolicited acquisition proposal from a third party, engages in negotiations regarding that proposal with the third party and/or changes its recommendation in favor of the merger proposals.

*Tutor-Saliba Shareholder Lock-Up*. From the date of the Merger Agreement until the earlier of the effective time of the merger or termination of the Merger Agreement, the Tutor-Saliba shareholders may not:

transfer any of their shares of Tutor-Saliba common stock except pursuant to the transactions contemplated by the Merger Agreement;

deposit any of their shares of Tutor-Saliba common stock in a voting trust, grant any proxies with respect to their shares of Tutor-Saliba common stock or subject any of their shares of Tutor-Saliba common stock to any arrangement with respect to voting (except pursuant to the transactions contemplated by the Merger Agreement);

solicit proxies or become a participant in a solicitation (as such terms are defined in Regulation 14A under the Securities Exchange Act of 1934) in opposition to or competition with the consummation of the merger or assist any party in taking or planning any action which could compete with, impede, interfere with or attempt to discourage or inhibit the timely consummation of the merger;

encourage, initiate or cooperate in a shareholders vote or action by consent of Perini shareholders in opposition to or in competition with the consummation of the merger;

become a member of a group (as such term is used in Rule 13d-5 under the Securities Exchange Act of 1934) with respect to Perini common stock for the purpose of opposing or competing with the consummation of the merger; or

acquire, offer to acquire, or agree to acquire, any of the shares of Perini common stock (except pursuant to the transactions contemplated by the Merger Agreement or other arrangement in effect as of the date of the Merger Agreement). *Restrictions on the Parties Business Prior to the Closing.* In general, until either the completion of the merger or the termination of the Merger Agreement, each of Perini and Tutor-Saliba and their respective subsidiaries are required to carry on their businesses in all material respects in the ordinary course and to use commercially reasonable efforts to preserve their business organizations substantially intact and maintain their existing relations with customers, suppliers, employees and other business associates. Each of Perini and Tutor-Saliba has also agreed that (except as contemplated or permitted by the Merger Agreement), without the prior written consent of the other party, it will not and will not permit any of its subsidiaries or other controlled entities to:

issue, deliver or sell or authorize or propose the issuance, delivery or sale of any equity interests (in the case of Perini, other than in the ordinary course of business consistent with past practices or pursuant to existing equity interests);

split, combine, repurchase or reclassify any equity interests, declare, set aside or pay any dividend or other distribution (whether in cash, shares or property or any combination thereof), or make any other actual, constructive or deemed distribution in respect of any equity interests;

acquire or agree to acquire in any manner a substantial portion of the equity securities or assets of any business or person (other than acquisitions of assets with a value not greater than \$5 million in the aggregate);

wind-up, liquidate or dissolve, adopt or approve a plan of complete or partial wind-up, liquidation or dissolution;

enter into, consummate, or adopt resolutions providing for or authorizing, any merger, consolidation, restructuring, recapitalization or other reorganization, other than the transactions contemplated by the Merger Agreement;

modify or amend any right of any holder of outstanding equity interests;

amend its certificate of incorporation or bylaws or other comparable organizational documents;

make, change or rescind material tax elections, change accounting periods, adopt or change tax accounting methods, file amended tax returns, enter into closing agreements, settle tax claims or assessments, surrender any right to claim a tax refund or consent to any extension or waivers of the limitation period applicable to any tax claim or assessment; or

agree to take any of the foregoing actions.

In addition, Tutor-Saliba has also agreed that (except as contemplated by the Merger Agreement), without the prior written consent of Perini, it will not and will not permit any of its subsidiaries or other controlled entities to:

incur, assume or otherwise become liable for any debt other than in the ordinary course of business consistent with past practices;

incur or commit to any material capital expenditures outside of the ordinary course of business consistent with past practices (excluding any expenditures in connection with Tutor-Saliba s performance of services for any third party);

cancel, compromise, release or assign any material liability owed to it or any material claims held by it, other than in the ordinary course of business consistent with past practice;

enter into or accelerate, terminate, modify, renew or cancel any material contract other than in the ordinary course of business consistent with past practice;

sell, lease, transfer, license, mortgage, pledge or otherwise dispose of or encumber, except for any permitted liens, any of its material properties or assets (other than the sale of assets in the ordinary course of business consistent with past practice or as contemplated by the Merger Agreement);

implement any layoffs that could implicate the Worker Adjustment and Retraining Notification Act of 1988 or any similar applicable law;

change its auditor or materially change its methods of accounting in effect as of the date of the Merger Agreement except as required by changes in GAAP or applicable law or by a governmental entity;

revoke or terminate its status as an a validly electing subchapter S corporation within the meaning of the Internal Revenue Code;

settle or compromise any legal proceeding on terms which require it to pay, incur or assume any liability in excess of \$5 million;

enter into, amend, modify or renew any contract regarding employment, consulting, severance or similar arrangements with any of its directors, officers, employees or consultants, or grant any salary, wage or other increase in compensation or increase any employee benefit (including incentive, profit sharing or bonus payments), other than, in the case of employees or consultants, in the ordinary course of business consistent with past practice;

grant any severance or termination pay or benefit to any directors, officers, employees or consultants except pursuant to written agreements or severance policies in effect as of the date of the Merger Agreement, other than in the ordinary course of business consistent with past practice;

adopt or amend any employee plan except to the extent required by law or to the extent the adoption or amendment of such employment plan does not materially increase the aggregate cost of the employee plans; or

enter into or modify any transaction or arrangement with, or for the benefit of any of its affiliates or any of its directors, former directors, officers or shareholders.

These restrictions are subject to qualifications. In particular, the restrictions on Tutor-Saliba s payment of dividends and distribution of assets are subject to exceptions described below under Pre-Closing Distributions of Property and Pre-Closing Payment of Dividends; Dividend Note beginning on page 79.

*Reasonable Best Efforts to Complete the Merger*. Each of the parties agreed to cooperate with each other and to use their (and cause their respective subsidiaries to use) reasonable best efforts to take or cause to be taken all actions and do or cause to be done all things necessary, proper or advisable under the Merger Agreement and applicable laws to complete the merger and the other transactions contemplated by the Merger Agreement in the most expeditious manner practicable, including using reasonable best efforts to satisfy conditions, obtaining all necessary regulatory approvals (including the expiration or termination of the waiting periods under the Hart-Scott-Rodino Antitrust Improvements Act of 1976), and to use commercially reasonable efforts to obtain all necessary consents under contracts or permits in connection with the merger.

The parties also have agreed to take all action necessary to ensure that no state anti-takeover statutes or similar regulation is or becomes applicable to the merger and, if any state anti-takeover statute becomes applicable, to take all necessary actions to ensure that the merger and

# Table of Contents

other provisions of the Merger Agreement be completed as soon as reasonably practicable and to minimize the effect of such statute on the merger.

*Non-Competition; Non-Hire.* Each Tutor-Saliba shareholder and Mr. Tutor have agreed that for five years after the closing date, they will not engage in any competing operation (other than through a passive ownership interest) in any country in the world in which Tutor-Saliba or any of its subsidiaries or Perini or any of its

subsidiaries currently conducts business or conducted business in the prior two years. A competing operation means any contracting or subcontracting of, or any other provision or promotion of, construction services or products, in either case including general contracting, contracting, plumbing, electrical, concrete forming or placing, steel erection, engineering, construction management, design-build services, and self-performed construction services. In addition, each Tutor-Saliba shareholder and Mr. Tutor have agreed that for five years after the closing date, they will not (i) solicit or hire employees of Tutor-Saliba or any of its subsidiaries (with certain limited exceptions), or (ii) cause any of Tutor-Saliba s customers or suppliers to cease doing business or reduce the level of business with Tutor-Saliba or any of its subsidiaries.

*Pre-Closing Distributions of Property.* The Merger Agreement provides that, notwithstanding the restrictions on distributions of assets by Tutor-Saliba prior to the closing of the merger, Tutor-Saliba may declare and distribute certain specified assets to the Tutor-Saliba shareholders prior to the closing. Those assets are:

all of its partnership interests in several partnerships that jointly own a high-rise office building in San Pedro, California, commonly known as Pacific Place;

all of its interest in Ketchum, Idaho; and

cash for tax payments as described below.

Because Tutor-Saliba is a subchapter S corporation, its taxable income is attributed to the shareholders of Tutor-Saliba for federal income tax purposes meaning that its shareholders are responsible for paying income taxes on their proportionate share of the income. The Merger Agreement permits Tutor-Saliba to make cash distributions to its shareholders that are intended to cover their income tax obligations for Tutor-Saliba s income from January 1, 2008 through the completion of the merger. The combined company will retain such income. These distributions for taxes will not take into account any taxes with respect to income or gain realized in respect of the transactions contemplated by the Merger Agreement or otherwise not in the ordinary course of Tutor-Saliba s business or any gain realized on the distribution of appreciated assets, the taxes for which will be solely the responsibility of the Tutor-Saliba shareholders. In April 2008, Tutor-Saliba distributed approximately \$11.6 million of cash to its shareholders in respect of their April 2008 income tax payment obligations arising from Tutor-Saliba s operations. Shortly before the completion of the merger, the balance of the permissible tax distributions will be determined, and Tutor-Saliba will make additional cash distributions its shareholders that are intended to cover their income tax obligations not covered by the April tax distribution. If Perini and Tutor-Saliba are unable to agree on the amount of the tax distribution or either party later determines that the adjustment was incorrect, the amount of the tax distribution will be adjusted in accordance with a dispute resolution provision provided in the Merger Agreement following the completion of the merger.

In addition, Tutor-Saliba may pay the dividends described in Pre-Closing Payment of Dividends; Dividend Notes below.

*Pre-Closing Payment of Dividends; Dividend Notes.* The Merger Agreement provides that, notwithstanding the restrictions on the payment of dividends by Tutor-Saliba prior to the closing of the merger, Tutor-Saliba may declare and distribute specified assets as dividends to the Tutor-Saliba shareholders prior to the closing of the merger in an amount up to \$120 million. Tutor-Saliba may not distribute such dividends in cash unless the amount of unrestricted cash held by Tutor-Saliba as of the closing of the merger (after giving effect to such dividends, borrowings under Tutor-Saliba s credit agreement and ongoing operations of Tutor-Saliba) will be at least zero and in any event may not distribute in excess of \$50 million in cash dividends prior to the date that is one business day prior to the closing of the merger.

No earlier than one business day prior to the closing of the merger, Tutor-Saliba may declare and distribute a dividend payable in the form of one or more notes. The maximum principal amount of the notes is the amount by which \$120 million exceeds the aggregate amount of the cash dividends declared by Tutor-Saliba between the date of the Merger Agreement and the closing, less amounts of indebtedness to be repaid to Tutor-Saliba prior to the effective time of the merger by the former Tutor-Saliba shareholders which have not been repaid by the date of the issuance of the note.

A form of the note is attached as *Annex D* to this proxy statement, although the form contemplates that it may be changed to reflect requests made by Perini or as are necessary in connection with seeking any consent under, amendment to or replacement of any credit facilities or borrowings of Perini or Tutor-Saliba in connection with the transactions contemplated by the Merger Agreement. The note provides that the unpaid principal balance of the notes and all accrued and unpaid interest thereupon will become due and payable in full on June 30, 2012. Tutor-Saliba may prepay any amounts outstanding under the notes at any time. In addition, Tutor-Saliba will be required to prepay the notes with 100% of excess cash flow of Tutor-Saliba (as described in the note) for the preceding fiscal year (or, for the year in which the closing of the merger occurs, the partial fiscal year from the closing through the end of such fiscal year) within 90 days of the end of each fiscal year until the principal and interest of the notes has been repaid.

*Substitute Guarantees.* Perini and Tutor-Saliba agreed to cooperate to modify various contracts in order to remove or release Mr. Tutor and his affiliates as obligors, guarantors or sureties of any obligations of Tutor-Saliba or its subsidiaries under the applicable contracts or any similar obligations incurred from and after the date of the Merger Agreement.

Other Covenants. The Merger Agreement contains certain other covenants, including covenants relating to public announcements, access to information and cooperation on tax matters.

#### Listing of Perini Stock

We have agreed to use our reasonable best efforts to cause the shares of Perini common stock to be issued in connection with the merger to be approved for listing on the NYSE, subject to official notice of issuance. Approval for listing on the NYSE of the shares of Perini common stock issuable to Tutor-Saliba shareholders pursuant to the merger, subject only to official notice of issuance, is a condition to the obligations of Perini and Tutor-Saliba to complete the merger.

#### **Representations and Warranties**

The Merger Agreement contains representations and warranties made by each of Perini, Tutor-Saliba and the Tutor-Saliba shareholders to the other parties thereto as of specified dates.

The representations and warranties of Tutor-Saliba and Perini relate to:

corporate existence, qualification to conduct business and corporate standing and power;

subsidiaries;

corporate authority to enter into, and carry out the obligations under, the Merger Agreement and the enforceability of the Merger Agreement;

absence of a breach of the certificate of incorporation and bylaws (or their equivalents), applicable law or material agreements as a result of the merger;

required approvals and consents as a result of the merger;

capital structure;

financial statements and disclosure controls and procedures and, in the case of Perini, filings with the SEC;

absence of undisclosed liabilities;

absence of certain changes or events;

legal proceedings;

assets;

material contracts;

governmental contracts;

real property;

intellectual property;

insurance;

environmental matters;

tax matters;

labor and employee matters;

employee benefit plans;

interested party transactions;

the existence of certain unsuitable payments;

indemnification obligations;

required shareholder votes;

applicability of anti-takeover statutes;

payment of fees to finders or brokers in connection with the Merger Agreement;

in the case of Perini, the opinion of UBS;

in the case of Perini, the formation and activities of Merger Sub; and

in the case of Tutor-Saliba, ownership of shares of Perini common stock. The representations and warranties of the Tutor-Saliba shareholders relate to:

their ownership of Tutor-Saliba shares;

their authority to enter into the Merger Agreement;

the absence of conflicts with applicable law and other contracts;

an acknowledgement of the nature of their investment in Perini common stock and their accredited investor status; and

their status as valid shareholders of a subchapter S corporation within the meaning of the Internal Revenue Code. Several of the representations and warranties in the Merger Agreement are subject to important qualifications and limitations agreed to by Perini and Tutor-Saliba in connection with negotiating the terms thereof, including in certain cases the qualification that only items that would result in a material adverse effect would be deemed to make the representation and warranty untrue or other similar qualifications.

As used in the Merger Agreement, the term material adverse effect means, when used with respect to any party, any fact, circumstance, change or effect that, individually or when taken together with all other facts, circumstances, changes or effects in the aggregate, has had or would be reasonably expected to have or result in a material adverse effect on:

the business, assets, liabilities, properties, financial condition, or results of operations of such party and its subsidiaries, taken as a whole, other than facts, circumstances, changes or effects resulting from:

general market, economic or political conditions (including any changes arising out of acts of terrorism or war, weather conditions or other force majeure events) in the jurisdictions in which such party or any of its subsidiaries does business that do not disproportionately affect such party and its subsidiaries, taken as a whole, in any material respect compared to other Persons operating similar businesses in such jurisdictions;

general market, legal or economic conditions in the engineering or construction industries that do not disproportionately affect such party and its subsidiaries, taken as a whole, in any material respect compared to other participants in such industries;

the condition of the financial or securities markets in the countries in which such party or any of its subsidiaries does business that do not disproportionately affect such party and its subsidiaries, taken as a whole, in any material respect compared to other persons in such markets;

any change in the stock price or trading volume of equity interests of such party traded or listed on any national securities exchange or quotation system (but not any circumstance, change or effect that may be underlying such change to the extent such circumstance, change or effect would otherwise constitute a material adverse effect);

the pendency of the Merger Agreement or any of the transactions contemplated thereby, including the merger, or the taking of any action contemplated by the Merger Agreement or expressly permitted by the other party to the Merger Agreement; and

changes in GAAP; or

the ability of such party to perform its obligations pursuant to the Merger Agreement and the other related agreements and to consummate the transactions contemplated by the Merger Agreement and the other related agreements in a timely manner. **Indemnification** 

The Merger Agreement provides that following completion of the merger we, on the one hand, and the former Tutor-Saliba shareholders, on the other, will indemnify each other against losses, fines, costs and expenses incurred as a result of a breach of certain representations and warranties or violations of covenants contained in the Merger Agreement, subject to exceptions.

The respective indemnification obligations of each party are subject to customary limitations. No claim may be made as a result of a breach of the representations and warranties unless the loss with respect thereto is greater than \$250,000 and the aggregate losses as a result of breach of the representations and warranties exceeds \$5 million (and recovery may only be received for losses in excess of that amount). In addition, no party may recover in excess of \$60 million for losses as a result of breaches of the representations and warranties. In addition, no party may recover for losses as a result of a breach of the representations and warranties and warranties and warranties and warranties and warranties are used for losses as a result of a breach of the representations and warranties after the end of the survival period for the applicable representation and warranty. Most of the representations and warranties contained in or made pursuant to the Merger Agreement will survive the closing of the merger for a twelve-month period. Many of these limitations are subject to additional exceptions.

The Merger Agreement also limits the ability of any party to the Merger Agreement to be indemnified for losses as a result of, or otherwise rely upon for purposes of closing conditions, termination rights or other rights under the Merger Agreement, information known to the party seeking to assert a claim for indemnification or otherwise exercise the remedy under the Merger Agreement.

The trusts controlled by Mr. Tutor guarantee the indemnification obligations of the former Tutor-Saliba shareholders pursuant to the Merger Agreement.

The right to indemnification is the sole and exclusive remedy of the parties to the Merger Agreement in connection with losses arising out of matters set forth in the Merger Agreement, other than in the case of fraud or willful breach of a representation or warranty or the right to seek injunctive or equitable relief.

#### Conditions to the Completion of the Merger

The parties respective obligations to complete the merger are subject to the satisfaction or, to the extent legally permissible, the waiver of the following conditions in the Merger Agreement:

the receipt of the approval of the Merger Agreement by Tutor-Saliba shareholders by delivery of the written consent of Tutor-Saliba shareholders (which was delivered in connection with the execution and delivery of the Merger Agreement), and the receipt of the approval by Perini shareholders of the share issuance proposal and the articles amendment proposal;

the receipt of the required statutory approvals, including the expiration or termination of the waiting periods under the Hart-Scott-Rodino Antitrust Improvements Act of 1976;

the absence of any law, judgment, injunction, decree or other order by a governmental entity that is in effect challenging the merger or the other transactions contemplated by the Merger Agreement, having the effect of making the merger or the other transactions contemplated by the Merger Agreement illegal, otherwise preventing the consummation of the merger or any of the other transactions contemplated by the Merger Agreement or that seeks to retrain, prohibit or effect the rescission of the merger or the other transactions contemplated by the Merger Agreement;

the receipt of the approval for listing by the NYSE of Perini common stock to be issued pursuant to the merger, subject to official notice of issuance; and

the articles of amendment of the Perini amended and restated articles of organization having been filed with the Massachusetts Secretary of State and declared effective.

In addition, individually, the parties respective obligations to effect the merger are subject to the satisfaction or, to the extent legally permissible, the waiver of the following additional conditions:

the representations and warranties of the other party being true and correct as of the date of the Merger Agreement and as of the closing date of the merger unless, other than in the case of representations and warranties regarding capitalization, such failures to be true and correct (without giving effect to any materiality or material adverse effect qualifications or exceptions) in respect of those representations and warranties have not had and are not reasonably likely to result in a material adverse effect on the other party;

the other party having performed and complied in all material respects with all covenants required to be performed and complied with by it under the Merger Agreement;

the absence of a material adverse effect on the other party since the date of the Merger Agreement;

the receipt by both parties of all consents under any contracts or permits in connection with the Merger Agreement and the transactions contemplated thereby, except those consents the failure of which to obtain would not have a material adverse effect;

the execution and delivery by Perini and each shareholder of Tutor-Saliba of the Shareholders Agreement, and the Shareholders Agreement being in full force and effect;

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

the execution and delivery by Mr. Tutor and Perini of the Employment Agreement, and there being no constructive or anticipatory notice of termination of service thereunder;

in the case of Perini, the receipt by Perini of a certification by Tutor-Saliba to the effect that Tutor-Saliba is not and has not been a United States Real Property Holding Corporation within the meaning of Section 897(c)(2) of the Internal Revenue Code;

the receipt of an opinion of the party s counsel which provides that the merger will qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code;

in the case of Perini, the receipt by Perini of certain Tutor-Saliba audited financial statements by April 30, 2008, which audited financial statements do not differ in any material respect from the unaudited financial statements previously made available to Perini, except as provided in the Merger Agreement; and

the receipt by each party of a certificate from an executive officer of the other party with respect to the satisfaction of certain of the closing conditions.

The Merger Agreement provides that neither Perini nor Tutor-Saliba may rely on the failure of any closing condition to be satisfied if the failure was caused by its failure to use its reasonable best efforts to satisfy the closing conditions or any other material breach of any other provision of the Merger Agreement.

#### **Termination of the Merger Agreement**

Right to Terminate. The Merger Agreement may be terminated at any time prior to the completion of the merger in any of the following ways:

by mutual written consent of us and Tutor-Saliba;

by either us or Tutor-Saliba:

if the merger has not been completed by September 30, 2008; provided that this date is automatically extended to December 31, 2008 if the conditions relating to the receipt of shareholder approval for the merger proposals, the receipt of required statutory and regulatory approvals and/or the absence of injunctions or other legal or regulatory restraints and the filing of the amendment to the articles of organization have not been fulfilled but all other conditions to closing have been satisfied or waived or are then capable of being satisfied; provided, further, that a party may not terminate the Merger Agreement pursuant to this provision if its failure to fulfill in any material respect any of its obligations under the Merger Agreement has resulted in the failure of the merger to have been completed by the applicable termination date;

if a governmental entity has enacted, issued, promulgated, enforced or entered into a law, judgment, injunction, decree or other order that is in effect, has become final and non-appealable and has the effect of making the merger or the other transactions contemplated by the Merger Agreement illegal, otherwise preventing the consummation of the merger or any of the other transactions contemplated by the Merger Agreement or that seeks to restrain or prohibit the consummation of, or effect the rescission of the merger or the other transactions contemplated by the Merger Agreement pursuant to this provision if such party failed to use its reasonable efforts to resist or prevent the entry or passage, or to seek the removal or lifting of such law, judgment, injunctions, decree or order; or

if the Perini shareholders fail to approve the share issuance proposal or the amendment proposal at the annual meeting or any adjournment or postponement thereof at which a vote on the merger proposals was taken; or

#### by Perini:

if there has been an uncured breach or failure to perform in any material respect of any representation, warranty, covenant or other agreement made by Tutor-Saliba or any of its shareholders in the Merger Agreement, and the breach or failure to perform would result in the applicable closing condition to the merger not being satisfied; or

if the Perini board of directors has effected a change of recommendation if such change of recommendation is for the purposes of accepting or adopting an acquisition proposal or entering into an acquisition agreement; or

by Tutor-Saliba:

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

if there has been an uncured breach or failure to perform in any material respect of any representation, warranty, covenant or other agreement made by Perini in the Merger Agreement, and the breach or failure to perform would result in the applicable closing condition to the merger not being satisfied; or

if the Perini board of directors has effected a change of recommendation. *Termination Fees/Reimbursement of Expenses.* Perini will be required to pay a termination fee of \$30 million to Tutor-Saliba:

immediately upon termination of the Merger Agreement by Perini after it effects a change of recommendation for the purposes of accepting or adopting an acquisition proposal or entering into an acquisition agreement;

within one business day following the consummation of an acquisition transaction if (i) Perini or Tutor-Saliba terminates the Merger Agreement due to any Perini change of recommendation other than a Perini change of recommendation described in the bullet immediately above, and (ii) Perini enters into an acquisition agreement, or an acquisition transaction is otherwise consummated, within twelve months following the termination of the Merger Agreement (provided however that for these purposes 20% and 80% in the definition of acquisition transaction and acquisition agreement should be substituted with 35% and 65%); or

within one business day following the consummation of an acquisition transaction if (i) Perini or Tutor-Saliba terminates the Merger Agreement due to a failure to receive Perini shareholder approval at the Perini shareholders meeting and (ii) after the date of the Merger Agreement and before the Perini shareholders meeting, an acquisition proposal has been publicly announced and not withdrawn, and (iii) Perini enters into an acquisition agreement, or an acquisition transaction within twelve months following the termination of the Merger Agreement (provided however that for these purposes 20% and 80% in the definition of acquisition transaction agreement should be substituted with 35% and 65%).

Perini will reimburse Tutor-Saliba for up to \$5 million of Tutor-Saliba s reasonable, documented, out-of-pocket expenses if (i) Perini or Tutor-Saliba terminate the Merger Agreement due to a failure to receive Perini shareholder approval at the Perini shareholders meeting or (ii) Perini or Tutor-Saliba terminates the Merger Agreement due to any Perini change of recommendation. The reimbursement of expenses will not limit the obligation of Perini to pay the termination fee described above, if applicable.

#### The Shareholder Representative

Each Tutor-Saliba shareholder has appointed Mr. Tutor as the shareholder representative to serve as its and their true and lawful agent and attorney-in-fact. In such capacity, the shareholder representative may take any action on behalf of the shareholders of Tutor-Saliba, and may bind them under the Merger Agreement and any other related agreement.

#### Amendments, Extensions and Waivers

*Amendments*. The Merger Agreement may be amended by the parties, except that any amendment after the Perini shareholders meeting that by law requires approval by shareholders may not be made without such approval. The Merger Agreement may not be amended by Perini without the approval of the Special Committee or, if such committee shall no longer be constituted, a majority of the independent directors of Perini. All amendments to the Merger Agreement must be in writing signed by each party.

*Extensions and Waivers*. At any time prior to the completion of the merger, any party to the Merger Agreement may extend the time for the performance of any of the obligations or other acts of any other party; waive any inaccuracies in the representations and warranties of any other party contained in the Merger Agreement or in any document delivered pursuant to the Merger Agreement; or subject to the restrictions described in the immediately preceding paragraph, waive compliance by any other party with any of the agreements or conditions contained in the Merger Agreement. All extensions and waivers must be in writing and

signed by the party against whom the extension or waiver is to be effective. Any waiver of a condition to the merger by Perini would be made by Perini only if, in the judgment of the Perini board of directors, the waiver was in the best interests of Perini shareholders. Perini does not have any current intention to waive any condition to the merger and cannot predict the circumstances under which it would do so. Perini would re-solicit the votes of its shareholders if it decided to waive a material condition to the merger, including the condition that Perini receive an opinion of its counsel which provides that the merger will qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code.

*Governing Law.* The Merger Agreement is governed by the laws of New York, other than with respect to certain provisions and actions related thereto that are governed by the laws of California and Massachusetts, as appropriate.

#### The Shareholders Agreement

Simultaneously with the execution of the Merger Agreement, we entered into the Shareholders Agreement with Mr. Tutor, as the shareholder representative, and each of the Tutor-Saliba shareholders who will become shareholders of Perini upon completion of the merger. The Shareholders Agreement will become effective upon the completion of the merger.

The following is a summary of the material terms of the Shareholders Agreement. This summary does not purport to describe all the terms of the Shareholders Agreement and is qualified by reference to the complete Shareholders Agreement, which is attached as Annex B to this proxy statement and incorporated by reference in this proxy statement. All shareholders of Perini are urged to read the Shareholders Agreement carefully and in its entirety.

#### The Shareholder Representative

Mr. Tutor has agreed to serve as the shareholder representative of the former Tutor-Saliba shareholders under the Shareholders Agreement. He shall serve in such capacity until his resignation as the shareholder representative or his death or mental or physical incapacity. If Mr. Tutor ceases to serve as the shareholder representative under the Shareholders Agreement, the Tutor Group shall select a replacement shareholder representative.

#### **Composition of the Board of Directors**

The Shareholders Agreement provides that the shareholder representative will have the right to designate two nominees for election to the Perini board of directors for so long as the Tutor Group owns at least 22.5% of the outstanding shares of Perini common stock and one nominee if the Tutor Group owns less than 22.5% but more than 11.25% of the outstanding shares of Perini common stock. In addition, for so long as Mr. Tutor serves as the chief executive officer of Perini, he will be nominated for election to the Perini board of directors. The remaining members of the Perini board of directors will be nominated by the Nominating and Governance Committee of the Perini board of directors. At each meeting of Perini shareholders following the completion of the merger at which directors of Perini are to be elected, Perini has agreed to nominate for election to the Perini board of directors and recommend the election of the shareholder representative s designees and Mr. Tutor (as long as he serves as Perini s chief executive officer), subject to certain limitations to comply with law, governance requirements or eligibility for listing on a securities exchange or if a nominee is deemed to be unfit to serve as a director of an NYSE-listed company or otherwise does not meet applicable eligibility criteria.

The Tutor Group is expected to own approximately 43% of the outstanding shares of Perini common stock immediately following the closing of the merger. Pursuant to the Shareholders Agreement, the Perini board of directors at the effective time of the merger would be comprised of Mr. Tutor, two designees of the shareholder representative and eight other directors nominated by the Nominating and Governance Committee of the Perini board of directors (which are expected to include all of Perini s existing directors). Mr. Tutor has designated C.L. Max Nikias as one of the shareholder representative s board designees, and he will be appointed as a new

member of the Perini board of directors upon the completion of the merger. As of the date of this proxy statement, Mr. Tutor has not elected to exercise his right to nominate a second director for election, although he has not waived the right to do so in the future. Accordingly, Perini expects that, at the effective time of the merger, the Perini board of directors will be comprised of Mr. Tutor, Mr. Nikias and nine other directors recommended by the Nominating and Governance Committee of the Perini board of directors.

To the extent permitted by applicable law and the rules of the principal stock exchange on which Perini common stock is listed and to the extent determined by the Nominating and Governance Committee of the Perini board of directors, the shareholder representative will be entitled to have one of his designees serve on each committee of the Perini board of directors.

#### **Voting Restrictions**

Pursuant to the Shareholders Agreement, the Tutor Group will vote all of their shares of Perini common stock in support of the Perini board of directors slate of directors as described above.

In addition, on all other matters to be voted on by shareholders of Perini, the Shareholders Agreement provides that the Tutor Group will vote their shares of Perini common stock that are, in the aggregate, equal to up to 20% of the voting power of the outstanding shares of Perini common stock, in their discretion and the balance of their shares of Perini common stock in the same proportions as all other shares of Perini common stock (excluding the Tutor Group) are voted on such matter.

The Tutor Group has agreed not to grant any proxies with respect to Perini common stock or enter into any arrangement with respect to the voting of or taking action with respect to shares of Perini common stock other than as provided under the Shareholders Agreement.

These restrictions on voting remain in effect until the later of the third anniversary of the effective time of the merger or the date on which the Tutor Group owns less than 20% of the aggregate issued and outstanding shares of Perini common stock.

#### Standstill

Pursuant to the Shareholders Agreement, until the later of the third anniversary of the effective time of the merger or the date on which the Tutor Group owns less than 20% of the outstanding shares of Perini common stock, the Tutor Group may not take certain actions that may be deemed to be actions to obtain control of Perini, including:

acquiring or offering to acquire shares of Perini common stock that will result in the Tutor Group collectively owning shares of Perini common stock equal to more than the percentage of the total outstanding shares of Perini common stock to be held by them at the effective time of the merger (approximately 43%);

directly or indirectly soliciting proxies;

forming a group within the meaning of the federal securities laws;

granting any proxies or voting power with respect to their shares or depositing any shares in a voting trust;

initiating shareholder proposals;

seeking election of new board members or replacement of current board members;

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

seeking to call shareholder meetings;

making any public announcement or proposal with respect to any form of business combination transaction involving Perini; or

seeking publicly to have Perini waive, amend or modify any of the standstill provisions contained in the Shareholders Agreement. These standstill restrictions will not prohibit or restrict any action taken by a director or designee of the shareholder representative as a member of the Perini board of directors or the exercise of any voting rights with regard to shares of Perini common stock.

### **Transfer Restrictions**

The Shareholders Agreement provides that for six months after the completion of the merger, none of the former Tutor-Saliba shareholders may transfer or dispose of the shares of Perini common stock acquired pursuant to the merger other than to certain affiliated persons or pursuant to the exercise of piggyback registration rights described below following the decision by Perini to register shares of Perini common stock.

After the six-month anniversary of the completion of the merger, the Tutor Group will not be permitted to transfer shares of Perini common stock unless after doing so they continue to collectively own at least 70% of the shares of Perini common stock acquired by them pursuant to the merger. This restriction on the transfer of shares continues until the later of the fifth anniversary of the effective time of the merger or the date on which the Tutor Group owns less than 20% of the aggregate issued and outstanding shares of Perini common stock. After the fifth anniversary of the effective time of the merger or following the termination of Mr. Tutor s employment without Cause pursuant to the Employment Agreement, such restrictions lapse and the Tutor Group may transfer shares of Perini to any person or group. In addition, all transfer restrictions under the Shareholders Agreement terminate on the date that is the later of the fifth anniversary of the completion of the merger and such time as the Tutor Group collectively ceases to own 20% of the aggregate issued and outstanding shares of Perini common stock. Notwithstanding the foregoing, the former Tutor-Saliba shareholders may transfer or dispose of shares of Perini common stock in any transactions approved by a majority of the Perini board of directors, excluding Mr. Tutor and the directors designated by him in his capacity as the shareholder representative.

### **Registration Rights**

Pursuant to the Shareholders Agreement, Perini has agreed to give the Tutor-Saliba shareholders who will become Perini shareholders pursuant to the merger certain registration rights with respect to the shares of Perini common stock acquired pursuant to the merger.

*Demand Registration.* Following the six-month anniversary of the effective time of the merger and subject to the continuing effect of the transfer restrictions set forth in the Shareholders Agreement noted above, the shareholder representative may require Perini to register shares of Perini common stock issued to the Tutor Group in connection with the merger for resale under the Securities Act in an underwritten offering. The shareholder representative may exercise this demand registration right on up to three occasions. Perini will use its reasonable best efforts to qualify for registration on Form S-3. These demand registration rights are subject to customary conditions and limitations. Perini is responsible for paying the expenses of any such registration.

*Piggyback Registration.* If we propose to register any securities under the Securities Act, each Tutor-Saliba shareholder who will become Perini shareholders pursuant to the merger must receive notice of the registration and to include its shares of Perini common stock in the registration. These piggyback registration registration rights are subject to customary conditions and limitations, including the right of the underwriters of an offering to limit the number of shares included in such registration and Perini s right not to effect a requested registration. Perini is responsible for paying the expenses of any such registration.

Governing Law. The Shareholders Agreement is governed by the laws of Massachusetts.

### The Employment Agreement

Simultaneously with the execution of the Merger Agreement, we entered into the Employment Agreement with Mr. Tutor. The Employment Agreement will become effective upon the completion of the merger.

The following is a summary of the material terms of the Employment Agreement. This summary does not purport to describe all the terms of the Employment Agreement and is qualified by reference to the complete Employment Agreement, which is attached as Annex C to this proxy statement and incorporated by reference in this proxy statement. All shareholders of Perini are urged to read the Employment Agreement carefully and in its entirety.

*Position; Compensation and Benefits.* Under the Employment Agreement, Mr. Tutor will serve as chief executive officer of Perini, and as a member and the chairman of the Perini board of directors. Mr. Tutor has agreed to devote his reasonable best efforts and full business time to the performance of his duties under the Employment Agreement and the advancement of the business and affairs of Perini, other than his service on other boards of directors and charitable organizations and the maintenance of his family investments. He will receive an initial annual base salary of \$1.5 million, subject to annual review and upward adjustment, an annual performance-based cash bonus opportunity equal to 175% of his base salary if target performance levels established by the Compensation Committee of the Perini board of directors are satisfied (with greater or lesser amounts paid if performance levels are above or below such target level), and will be eligible to participate in Perini s equity incentive plan to be adopted after the effective time of the merger.

*Term.* The initial term of the Employment Agreement is five years, commencing on the effective date of the merger. The term extends automatically for successive one-year periods, unless either party notifies the other at least 90 days in advance of the expiration of the Employment Agreement that it does not intend to renew the Employment Agreement.

*Termination*. If (i) we terminate Mr. Tutor s employment for any reason other than for Cause or disability, or (ii) Mr. Tutor terminates his employment for Good Reason (as such terms are defined in the Employment Agreement), Mr. Tutor will be entitled to receive a severance package consisting of the following payments and benefits:

any earned but unpaid base salary and all other accrued benefits through the date of employment termination will be paid;

Mr. Tutor will receive a pro rata bonus for the calendar year in which his employment ends equal to the amount of such bonus that would have been paid to him had he remained with Perini for that entire calendar year, pro rated for the portion of the year he worked prior to his termination;

a cash payment equal to two times the sum of his base salary and target bonus for the year of termination;

immediate vesting of all outstanding equity awards Mr. Tutor holds; and

continued benefits for Mr. Tutor and his dependents for a period of 24 months.

If Mr. Tutor s employment is terminated other than for Cause or Disability during the two years following a Change of Control of Perini (as such terms are defined in the Employment Agreement) Mr. Tutor is entitled to the payments above, except that instead of receiving the cash payment equal to two times the sum of his base salary and target bonus for the year of termination, Mr. Tutor would receive a cash payment equal to three times the sum of his base salary and target bonus, and the continued benefits for Mr. Tutor and his dependents will continue for the greater of 36 months and the remaining term of his Employment Agreement.

If Mr. Tutor s employment is terminated due to his death or due to his Disability, then (i) any earned but unpaid base salary and all other accrued benefits through the date of employment termination will be paid, and (ii) all outstanding equity awards held by him shall immediately vest.

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

Perini will generally have Cause to terminate Mr. Tutor s employment in the following circumstances: (i) his conviction of, or plea of nolo contendere to, a felony; (ii) his willful and continued failure to substantially perform his essential job functions; (iii) his material act of fraud or willful and material misconduct to Perini; (iv) his willful and material breach of the employment contract; (v) a material breach by him of any material written Perini policy; or (vi) a failure by him to cooperate in any investigation or audit regarding the accounting practices, financial statements, or business practices of Perini. For purposes of this provision, no act or failure to act, on the part of Mr. Tutor, shall be considered willful unless it is done, or omitted to be done, by Mr. Tutor in bad faith or without reasonable belief that his action or omission was in the best interest of Perini. Any termination for Cause generally requires written notice to Mr. Tutor and providing him with 10 days to cure the conduct after such notice. The Perini board of directors must also vote affirmatively that Mr. Tutor is to be terminated for Cause after giving him an opportunity to be heard by the full board.

Mr. Tutor will generally have Good Reason to terminate his employment under any of the following circumstances: (i) any adverse change in his titles; (ii) any reduction in his base salary; (iii) a material diminution in his authority, responsibilities or duties; (iv) the assignment of duties materially inconsistent with his position; (v) a relocation of his place of employment to a location more than 50 miles further from the current offices near Los Angeles, California; (vi) any other material breach of the terms the Employment Agreement or (vii) the failure of Perini to have his contract assumed after a merger, consolidation, sale or similar transaction. In order to invoke a termination for Good Reason, Mr. Tutor must notify Perini of the existence of the event of Good Reason within 90 days of its occurrence, Perini must fail to cure the event within 30 days of the notice, and Mr. Tutor must terminate his employment within 10 days of the expiration of such period.

*Perquisites and Benefits.* Mr. Tutor is entitled to various perquisites and benefits as set forth in the Employment Agreement; including, (i) 150 hours of flying time per calendar year of personal use of Perini s business jet, with any unused balance being carried forward to subsequent years while employed; (ii) use of an automobile and driver, and use of an apartment in Las Vegas, Nevada, in each case on terms and conditions to be determined by the Perini board of directors; (iii) participation in all fringe benefits and perquisites made available generally to senior executives of Perini, generally on the same terms and conditions, (iv) 30 days vacation; (v) participation in all pension, retirement, profit sharing, savings, 401(k), income deferral, life insurance, disability insurance, accidental death and dismemberment protection, travel accident insurance, hospitalization, medical, dental, vision and other employee benefit plans, programs and arrangements made available generally to other senior executives of the Perini, to the extent eligible. Furthermore, Perini will obtain, on behalf of Mr. Tutor, life insurance coverage under term or ordinary life insurance polici(es) (at his choice) with an aggregate annual premium cost not to exceed \$175,000.

*Non-Competition.* Mr. Tutor has agreed that during the term of his employment with Perini and for two years after the end of his employment (unless his employment is terminated by Perini without Cause or he terminates his employment for Good Reason ), he will not compete with Perini or solicit certain of its employees. Mr. Tutor has also agreed to be bound by customary restrictions on disclosure of confidential information.

*Tax Gross Up.* If Mr. Tutor s benefits and payments become subject to an excise tax under Section 4999 of the Internal Revenue Code in connection with a change in control of Perini, he will be entitled to an additional gross up payment to compensate him for the amount of this additional excise tax.

Governing Law. The Employment Agreement is governed by the laws of California.

### Post-Merger Governance and Management

#### **Board of Directors**

As described above, upon the completion of the merger, Mr. Tutor (as the shareholder representative for the former Tutor-Saliba shareholders) will have the right to designate two nominees for election to the Perini board of directors and, for so long as Mr. Tutor serves as our chief executive officer, he will be nominated for election to the Perini board of directors. Please see The Shareholders Agreement beginning on page 86 for a description of the composition of the Perini board of directors following completion of the merger.

The Shareholders Agreement provides that the Perini board of directors at the effective time of the merger will be comprised of Mr. Tutor, two designees of the shareholder representative and eight other directors nominated by the Nominating and Governance Committee of the Perini board of directors (which are expected to include all of Perini s existing directors). It is expected that the eight current members of the board will be joined by Marilyn A. Alexander, Donald D. Snyder and Mr. Nikias. Mr. Tutor has designated Mr. Nikias as one of the shareholder representative s board designees, and he will be appointed as a new member of the Perini board of directors upon the completion of the merger. As of the date of this proxy statement, Mr. Tutor has not elected to exercise his right to nominate a second director for election, although he has not waived the right to do so in the future. Accordingly, Perini expects that at the effective time of the merger, the Perini board of directors will be comprised of Mr. Tutor, Mr. Nikias and nine other directors recommended by the Nominating and Governance Committee of the Perini board of directors. Mr. Tutor has advised the Perini board of directors that should he choose to designate a second person for appointment to the board of directors at a time when the board already included eleven members, he would support a temporary expansion of the board to twelve members to accommodate such additional member. Such expansion would continue until the next meeting of shareholders at which directors are elected, at which time he would expect the size of the board to be reduced back to eleven members (as contemplated by the Shareholders Agreement) and the slate of nominees for election to be adjusted accordingly.

Following the closing of the merger, and assuming the Perini shareholders elect the four director-nominees listed in Proposal 3 and assuming that Mr. Tutor appoints Mr. Nikias to serve as a director, the Perini board of directors will be classified in accordance with our By-Laws as follows:

Class I directors (with terms expiring in 2009): Robert Band, Michael R. Klein, and Robert L. Miller.

Class II directors (with terms expiring in 2010): Williard W. Brittain, Jr., Robert A. Kennedy, C.L. Max Nikias and Ronald N. Tutor.

Class III directors (with terms expiring in 2011): Marilyn A. Alexander, Peter Arkley, Raymond R. Oneglia and Donald D. Snyder.

#### **Voting Restrictions**

Pursuant to the Shareholders Agreement, Mr. Tutor has agreed to certain restrictions on his rights to vote the shares of Perini common stock that he will control following the completion of the merger. Please see The Shareholders Agreement beginning on page 86 for a description of the restrictions applicable to Mr. Tutor s voting rights pursuant to the Shareholders Agreement.

#### Management

Other than the Employment Agreement, as of the date of this proxy statement, neither we nor any of our affiliates has entered into any employment agreements with our management in connection with the merger, nor amended or modified any existing employment agreements. Upon completion of the merger, Mr. Tutor will continue to serve as chief executive officer and as a member and chairman of the Perini board of directors, subject to the terms and conditions of the Employment Agreement. We currently intend to retain members of our management team following the merger.

### INFORMATION ABOUT TUTOR-SALIBA

#### **Tutor-Saliba Business**

Tutor-Saliba is a leading civil infrastructure and commercial building construction company that focuses on large, complex projects, usually ranging from \$100 million to \$1 billion or more in size. Tutor Saliba manages all aspects of these projects, including design-build, design-bid-build and pre-construction services for project owners. These capabilities, together with its significant capacity to self-perform critical construction specialties such as concrete forming and placement, site excavation and support of excavation, and electrical and mechanical services, are the core strengths of Tutor-Saliba.

Tutor-Saliba operates in three segments: domestic building, domestic civil and international. Its domestic building operations are predominately in Nevada and California, where Tutor-Saliba maintains large offices. Its domestic civil operations have been historically focused primarily in California and New York, and it plans to expand by pursuing attractive infrastructure opportunities elsewhere in the United States. Its international operations are conducted primarily on the island of Guam and in the Philippines.

Tutor-Saliba disposed of the property comprising its commercial real estate segment in January 2008, as discussed below. The real estate operations are classified in discontinued operations. Accordingly, Tutor-Saliba did not discuss the impact of that segment on its overall performance.

Tutor-Saliba s growth has been driven by its success in winning contracts and effectively performing public infrastructure development and public and private sector building construction projects over many years. In 2007, Tutor-Saliba experienced growth in revenues and earnings in all of its business segments.

Below is a list of key factors that Tutor-Saliba believes will affect its future performance:

In its domestic building segment, the availability of opportunities for major construction projects depends on significant capital spending commitments by project sponsors in the markets served by Tutor-Saliba. For example, Tutor-Saliba s business in recent periods has benefited significantly from two ongoing projects in Las Vegas (the Wynn Encore and Planet Hollywood Tower), which were awarded in 2006. These projects will continue to drive Tutor-Saliba s operating performance in 2008. Tutor-Saliba believes that it can capitalize on its success on these projects and win significant additional casino hotel project opportunities in Las Vegas and elsewhere. The number and size of these future opportunities will be determined in part by consumer demand for upscale casino hotel gaming experiences, and by the availability of financing in the capital markets for project sponsors. Tutor-Saliba believes there will be continuing opportunities for public and private sector building construction in other fields as, for example, inefficient government buildings are replaced (such as the Los Angeles Police Headquarters building Tutor-Saliba is completing) and as the continuing growth of the healthcare sector and the aging of the baby boom creates demand for modern hospitals.

The pace of investment in infrastructure and the timing and availability of opportunities for its domestic civil segment to win new contracts in any particular period is a function of many factors including annual appropriations, development schedules, the availability of federal matching funds, administrative approvals and governmental priorities, which can in turn be affected by state and local budget conditions. There is widespread consensus that infrastructure in the United States has been substantially undercapitalized, as evidenced by major legislative initiatives in the United States Congress, California and other states and municipalities in recent years. These legislative initiatives have earmarked significant funds for highway, bridge, airport and other infrastructure projects that Tutor-Saliba believes will present it with significant opportunities for continued expansion.

Tutor-Saliba is expanding its international segment on the island of Guam, where it is the largest contractor in the market, to take advantage of the announced planned relocation of 8,000 United States Marines and other military personnel from Okinawa, Japan. These Guam activities will generate an estimated \$10.3 billion of projects for the construction industry over a period of approximately seven years beginning in 2009.

Tutor-Saliba s senior management team is directly involved in each bidding process, and no major bid is submitted and no major contract is entered into without the approval of Tutor-Saliba s senior management, including its president and chief executive officer, Ronald N. Tutor. Tutor-Saliba s approach is to bid selectively on projects that are consistent with its profitability goals, and not to pursue low-margin projects or projects that have schedules or other requirements that it believes pose unacceptable performance risks. As a result of its experience in bidding, negotiating and performing large, complex projects over many years, its familiarity with key suppliers and subcontractors in its markets, its ability to self-perform many construction specialties, and its track record of performing its projects on time and on budget, Tutor-Saliba has been able to achieve substantial returns for its shareholders.

#### Management s Discussion and Analysis of Financial Condition and Results of Operations

Most of Tutor-Saliba s business is conducted under fixed price and guaranteed maximum price contracts. Tutor-Saliba s profitability therefore depends to a large degree on its ability to accurately forecast costs and timetables for a potential project when it bids or negotiates the contract, and to successfully control project costs and meet project schedules when it performs the contract. Tutor-Saliba s ability to accurately estimate and provide for the necessary workforce, materials and subcontractors for each project directly affects its contract profitability. If Tutor-Saliba underestimates its costs, or if it fails to meet its projected schedules, its performance can be adversely affected, and the impact on a particular quarter or year can be significant under percentage-of-completion accounting, as discussed under Critical Accounting Policies beginning on page 97.

Tutor-Saliba s estimating process is extensive, and it maintains separate estimating departments for its domestic civil, domestic building and international segments. Tutor-Saliba s estimating function benefits from its long relationships with union and non-union labor providers and subcontractors in Tutor-Saliba s key markets, and Tutor-Saliba generally contracts for its principal estimated material needs (e.g., concrete and steel) at the beginning of each job. In some of Tutor-Saliba s contracts, it is able to pass certain cost increases (through contingency and escalation clauses) along to the customer. Tutor-Saliba requires many subcontractors to obtain performance bonds, and Tutor-Saliba often obtains bonding to protect against materials delivery defaults (through materials supply bonds) and subcontractor non-performance (through contractor default insurance), with premiums for any such bonds and insurance included in its bids, subject to deductibles at its risk. Tutor-Saliba owns substantially all of its equipment fleet, and the depreciation and amortization expense it incurs for equipment is substantially less than what rental cost would be for the same fleet. Tutor-Saliba does not maintain a separate sales force, as the size of its projects generally requires direct, extensive management involvement with project sponsors in each bid proposal.

Tutor-Saliba s ability to estimate performance schedules is also of critical importance, as it can earn additional profit if it completes a project ahead of schedule, whether through contractual bonuses and/or avoiding anticipated costs. If Tutor-Saliba fails to meet contractual performance schedules it may be subject to additional costs that reduce its profitability and/or liquidated damages to the project owner.

Because Tutor-Saliba works on very large projects that are generally completed over a period of 30 to 42 months, its revenues and earnings at any point in time are frequently concentrated in a few projects in each segment. For example, in the three months ended March 31, 2008, Tutor-Saliba s domestic building segment generated 56.1% and 19.0% of revenues, from the Las Vegas Wynn Encore and Las Vegas Planet Hollywood Tower projects, respectively. Similarly, in that same period, Tutor-Saliba s domestic civil segment generated 39.2% of its revenues from the Los Angeles International Airport Runway and Taxiway Improvements project. Tutor-Saliba s customers are either government agencies or large, well-financed private entities and it has not historically had any meaningful customer credit issues and does not presently maintain any reserve for doubtful accounts. Tutor-Saliba s government customers generally require contractors bidding on major projects to prequalify based on financial strength, experience and other factors, and it has been able to consistently pre-qualify and win repeat contracts with Caltrans and other state agencies and educational institutions when it has been the

low bidder. For private sector work, Tutor-Saliba maintains strong relationships with clients who appreciate Tutor-Saliba s ability to deliver projects. Tutor-Saliba continuously evaluates new opportunities with customers that have major building needs and selects the best opportunity to replace its large existing projects as they are completed in order to replenish its backlog and continue expanding.

In certain of its projects, particularly in its domestic civil segment, Tutor-Saliba enters into joint venture arrangements with other contractors. Since 1977, Tutor-Saliba has been involved in 35 joint ventures with Perini with aggregate project values over \$3 billion. Each of the joint venture participants is usually committed to supply a predetermined percentage of capital and bonding capacity, as required, and to share in a predetermined percentage of the income or loss of the project. Depending on geographic location, one of the venture partners generally takes operating control of the project, with the other partners primarily providing financial support. Tutor-Saliba typically has management responsibility for joint ventures located in the western United States, while other partners have had operational control of joint ventures in the eastern United States. Historically, Tutor-Saliba accounted for joint ventures using the equity method of accounting. In 2007, Tutor-Saliba changed its method of reporting its interests in joint ventures to the proportionate consolidation method, with its financial statements reflecting its proportionate interest in the assets, liabilities and results of operations of each venture. Ordinarily, joint ventures are structured as general partnerships, and each participant is fully liable for the obligations of the joint venture. Tutor-Saliba will continue to seek out attractive joint venture opportunities in its existing markets and throughout the United States.

Tutor-Saliba elected to be taxed as a subchapter S corporation effective January 1, 1996. The provision for taxes reflected in Tutor-Saliba s historical financial statements in this proxy statement relates primarily to its subsidiary, Black Construction Corporation, which conducts Tutor-Saliba s international segment operations and is not subject to Tutor-Saliba s subchapter S corporation election. Tutor-Saliba will cease to be a subchapter S corporation upon the completion of the merger.

#### **Recent Developments and Expected 2008 Events**

#### Acquisitions

In late 2007 and early 2008, Tutor-Saliba completed two acquisitions of subcontractors and the acquisition of certain assets to establish an aggregates production operation. It intends to continue to consider acquisition opportunities that expand its ability to self-perform construction specialties and to expand its geographic reach in the United States. When Tutor-Saliba acquires subcontractors, it expects that they will generally continue to subcontract for jobs on projects other than Tutor-Saliba s projects as an additional source of revenue.

*Desert Plumbing & Heating Co., Inc.* On January 4, 2008, Tutor-Saliba completed the stock purchase of Desert Plumbing & Heating Co., Inc. (DPH), located in Las Vegas, Nevada, for \$35 million in cash. In addition to this cash consideration, Tutor-Saliba has also agreed to pay the former owner of DPH a percentage of DPH s earnings over the next three years, up to a maximum of \$4 million annually. Upon the completion of the merger, this amount will become fixed at \$12 million in the aggregate and will no longer be contingent upon DPH s earnings and will be payable in annual installments of \$4 million in each of 2008, 2009 and 2010 (assuming completion of the merger in 2008). Tutor-Saliba has the right to pay this \$12 million amount in cash or stock. Based on preliminary purchase accounting estimates, at December 31, 2007, DPH had \$24.7 million in liabilities, making the total value of the transaction \$59.7 million (before the contingent payment), resulting in intangible assets (including goodwill) of \$27.5 million (plus \$12 million for the contingent payment on a pro forma basis) and additional tangible assets of \$32.2 million, subject to final completion of purchase accounting adjustments.

*Powerco Electric Corp.* On September 12, 2007, Tutor-Saliba completed the stock purchase of Powerco Electric Corp., an electrical subcontractor, for \$3.3 million in cash. The acquisition was effective as of September 30, 2007 and included the assumption of \$25.7 million in Powerco s liabilities for a total consideration of \$29 million, which included intangibles (including goodwill) of \$7 million and tangible assets of

<sup>94</sup> 

\$22 million. Substantially all of Powerco s work prior to acquisition was as a subcontractor for Tutor-Saliba, and most of Powerco s contracts with Tutor-Saliba were not profitable to Powerco. Tutor-Saliba had previously advanced working capital funds to Powerco starting in May 2005 to help mitigate its cash flow difficulties on several of its subcontracts on Tutor-Saliba s projects. At the effective date of the acquisition, the sum of these advances had reached \$17.8 million. These liabilities and Tutor-Saliba s related receivable were eliminated on consolidation.

*North Valley Commerce Center, Building G.* On October 30, 2007, Tutor-Saliba entered into an agreement to purchase the North Valley Commerce Center, Building G, in Sylmar, California for \$6.5 million in cash. The property will include an office and warehouse facility, which are currently under construction. As of January 25, 2008, Tutor-Saliba had made initial deposits for the property totaling \$0.3 million into an escrow account. The purchase is expected to close upon substantial completion of the office and warehouse facility.

*Aggregates Business.* On March 7, 2008, Tutor-Saliba entered into an agreement for the transfer and purchase of certain mineral material mining contracts, equipment, and material stockpiles used to operate All Star Aggregate, Inc. s existing business in Sloan Pit, which is located in Clark County, Nevada, for a purchase price of \$5.2 million in cash. In connection with that transaction, on March 12, 2008, Tutor-Saliba entered into a ten-year renewable contract with the Bureau of Land Management (BLM) for the extraction of 19.5 million tons of limestone aggregate on land located at Mount Diablo Meridian, Nevada. Pursuant to Tutor-Saliba s September 12, 2007 bid of \$0.90 per ton and a \$0.01 per ton administrative fee, the contract price totals approximately \$17.7 million and requires a 5% deposit of \$0.9 million prior to beginning extraction. The balance of the contract price will be paid monthly over the contract term based upon the value of materials removed in the prior month. In addition, Tutor-Saliba is obligated under a separate agreement to pay a third-party finders fee related to the transaction in the amount of \$0.3 million.

### Amended and Restated Credit Agreement

On September 17, 2007, Tutor-Saliba entered into a new Senior Secured Revolving Credit Facility with Comerica Bank, as administrative agent, and two participant lenders. Subsequently, on May 12, 2008, Tutor-Saliba executed the First Amendment to Revolving Credit Agreement. For a description of Tutor-Saliba s new agreement, please see Liquidity and Capital Resources Cash and Working Capital beginning on page 101. Tutor-Saliba expects to enter into an amendment to this credit agreement, or enter into a new credit agreement, in connection with the merger.

### Subchapter S Dividends; Termination of Subchapter S Status

In January and April of 2008, Tutor-Saliba distributed \$27.9 million and approximately \$11.6 million, respectively, to its shareholders to facilitate their payment of income taxes attributed to them as a result of Tutor-Saliba s status as a subchapter S corporation. In connection with the consummation of the merger, Tutor-Saliba will distribute an estimated \$120 million in cash or notes to its shareholders, as its final subchapter S distribution. This amount represents a return of invested capital (with respect to certain shareholders), earnings and profits from years prior to Tutor-Saliba s S election and amounts of its net income that have been or will be subject to taxes to its shareholders in respect of periods prior to December 31, 2007, less distributions previously made to its shareholders with respect to such net income. Tutor-Saliba will cease to be a subchapter S corporation upon the consummation of the merger.

#### Distribution of Commercial Real Estate; Distribution of Residential Real Estate

Effective January 1, 2008, Tutor-Saliba transferred its interest in its commercial real estate segment, comprised of an interest in an office building, of which one of its subsidiaries was the majority owner, in exchange for the 2% beneficial interest in Black Construction Company that it did not already own. The property was transferred to the minority interest holder, which is owned by Mr. Tutor. The net book value of the property and equipment transferred was \$26.8 million at December 31, 2007 and the minimum future rental income as of

December 31, 2007 was \$18.3 million. The transferee is assuming mortgage debt of \$24.4 million. As a result of this transaction, the assets, liabilities and operating results associated with this business will be reflected in Tutor-Saliba s consolidated financial statements as discontinued operations. Prior to completion of the merger, Tutor-Saliba will also distribute to its shareholders a residential property in Ketchum, Idaho for its fair market value of approximately \$3.5 million.

#### 2008 Compensation Charge for 2007 Sale of Stock

In October, 2007, a trust controlled by Mr. Tutor sold an aggregate of 34,500 shares of Tutor-Saliba common stock, representing 3.8% of Tutor-Saliba s outstanding common equity, to certain Tutor-Saliba executives for an aggregate purchase price of \$9.6 million, paid in the form of recourse notes payable by such executives. There are buy-sell agreements between the executives and the trust that allow the trust to repurchase the shares at cost or such other price as the trust may determine, and these provisions will lapse upon completion of the merger with Perini. This lapse will result in a non-cash compensation charge to Tutor-Saliba s 2008 earnings in the quarter in which the merger is completed. The determination of such a charge and its amount, which could be significant, will be calculated in accordance with Statement of Financial Accounting Standards No. 123R, *Share-Based Payments*.

#### Backlog Analysis for 2005, 2006 and 2007

Tutor-Saliba includes a project in its backlog when the contract has been signed and financing is in place. Tutor-Saliba reduces backlog as revenue is recognized in accordance with its application of percentage-of-completion accounting. In Tutor-Saliba s domestic operations, it received significant new contract awards in 2006 for the \$1.3 billion Las Vegas Wynn Encore and the \$488 million Las Vegas Planet Hollywood Tower, scheduled for completion in 2009. These projects increased Tutor-Saliba s year-end backlog to a record of \$2.3 billion at December 31, 2006, up from \$508.7 million at December 31, 2005. These large projects required Tutor-Saliba to establish a permanent office in Las Vegas and otherwise expand to provide the necessary administrative and logistical support necessary to meet its contractual obligations and production schedules. These projects were in full production during all of 2007 and should be substantially completed by the end of 2008 (even though the Las Vegas projects are not scheduled for final completion until 2009). Tutor-Saliba continues to seek additional major projects in 2008 to build up its backlog for work in 2008 and beyond. In Tutor-Saliba s international operations, it was awarded three major projects in 2007 totaling \$106.4 million (Home Depot Guam, Old Apra Housing and USN Global Hawk Aircraft Maintenance Facility), with all of these projects having started construction in 2007 and continuing into 2008 and beyond. In Tutor-Saliba s civil operations, its largest new project is the \$241 million Los Angeles International Airport Runway and Taxiway Improvements project, awarded in late 2005.

The following tables provide a summary of Tutor-Saliba s backlog by segment for the years ended December 31, 2005, 2006 and 2007, and for the three months ended March 31, 2008. Tutor-Saliba expects that 95% of its March 31, 2008 backlog will be recognized as revenue within the next 18 months.

	Backlog at December 31, 2007	New Business Awarded (1) (in tl	Backlog at March 31, 2008		
Domestic Building	\$ 1,396,493	\$ 88,377	\$ 341,470	\$	1,143,400
Domestic Civil	95,429	24,935	34,196		86,168
Total Domestic	1,491,922	113,312	375,666		1,229,568
International	99,600	125	19,297		80,428
Total	\$ 1,591,522	\$ 113,437	\$ 394,963	\$	1,309,996

	Backlog at December 31, 2006	New Business Awarded (1) (in th	Revenue Recognized housands)	Backlog at mber 31, 2007
Domestic Building	\$ 2,034,037	\$ 258,580	\$ 896,124	\$ 1,396,493
Domestic Civil	212,933	56,773	174,277	95,429
Total Domestic	2,246,970	315,353	1,070,401	1,491,922
International	35,348	145,675	81,423	99,600
Total	\$ 2,282,318	\$ 461,028	\$ 1,151,824	\$ 1,591,522

	Backlog at December 31, 2005	New Business Awarded (1) (in t	Revenue Recognized housands)	acklog at nber 31, 2006
Domestic Building	\$ 210,411	\$ 2,144,029	\$ 320,403	\$ 2,034,037
Domestic Civil	278,560	76,248	141,875	212,933
Total Domestic	488,971	2,220,277	462,278	2,246,970
International	19,713	67,814	52,179	35,348
Total	\$ 508,684	\$ 2,288,091	\$ 514,457	\$ 2,282,318

	Backlog at January 1, 2005	New Business Awarded (1) (in tl	Backlog at December 31, 2005		
Domestic Building	\$ 363,314	\$ 76,002	\$ 228,905	\$ 210,411	
Domestic Civil	108,954	286,652	117,046	278,560	
Total Domestic	472,268	362,654	345,951	488,971	
International	27,963	33,389	41,639	19,713	

Edgar Filing: IDEX CORP /DE/ - Form 10-K						
Total	\$ :	500,231 \$	396,043	\$ 387,590	\$	508,684

 New business awarded consists of the original contract price of projects added to Tutor-Saliba s backlog plus or minus subsequent changes to the estimated total contract price of existing contracts and the balance of existing contracts obtained through acquisitions.
 Critical Accounting Policies

Tutor-Saliba s financial statements have been prepared in conformity with GAAP, which requires management to make estimates, judgments and assumptions that affect the reported amounts of assets and liabilities and disclosures of contingent assets and liabilities at the date of the financial statements as well as the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Although Tutor-Saliba s significant accounting policies are described in Note 1, Significant Accounting Policies, of the Notes to Tutor-Saliba s Consolidated Financial Statements attached as *Annex H* to this proxy statement, the following discussion is intended to describe those accounting policies most critical to the preparation of its consolidated financial statements.

*Method of Accounting for Contracts.* Revenues and profits from Tutor-Saliba s contracts and construction joint venture contracts are recognized by applying percentages of completion for the period to the total estimated profits for the respective contracts. Percentage-of-completion is determined by relating the actual cost of the work performed to date to the current estimated total cost of the respective contracts. When the estimate on a contract indicates a loss, the entire loss is recorded during the accounting period in which it is estimated. In the ordinary course of business, at a minimum on a quarterly basis, Tutor-Saliba prepares updated estimates of the total forecasted revenue, cost and profit or loss for each contract. The cumulative effect of revisions in estimates of the total forecasted revenue and costs, including unapproved change orders and claims, during the course of the work is reflected in the accounting period in which the facts that caused the revision become known. The financial impact of these revisions to any one contract is a function of both the amount of the revision and the percentage-of-completion of the contract. An amount equal to the costs incurred that are attributable to unapproved change orders and claims is included in the total estimated revenue when realization is probable. Profit from unapproved change orders and claims is recorded in the accounting period such amounts are resolved.

Tutor-Saliba s balance sheet account Deferred Contract Revenue represents the excess of billings to date over the amount of contract costs and profits (or contract revenue) recognized to date on the percentage-of-completion accounting method. Another balance sheet account entitled Unbilled Work represents the excess of contract costs and profits (or contract revenue) recognized to date on the percentage-of-completion accounting method over billings to date. Unbilled Work results when (1) the appropriate contract revenue amount has been recognized in accordance with the percentage-of-completion accounting method, but a portion of the revenue recorded cannot be billed currently due to the billing terms defined in the contract and/or (2) costs, recorded at estimated realizable value upon determination that they are probable of collection, related to unapproved change orders or claims are incurred. For unapproved change orders or claims that cannot be resolved in accordance with the normal change order process as defined in the contract, Tutor-Saliba may employ other dispute resolution methods, including mediation, binding and non-binding arbitration, or litigation.

*Use of Estimates*. Among the many estimates, judgments and assumptions that management makes in preparing its financial statements in conformity with GAAP, there are none more critical than those made in the accounting for Tutor-Saliba s contracts. The key to establishing profitable contracts lies in management s ability to estimate expected costs and revenues on a contract by contract basis. Once contracts are in place, Tutor-Saliba accounts for them by using the percentage-of-completion method. Due to the long-term nature of Tutor-Saliba s contracts, cost and revenue estimates are updated each reporting period. The updated estimates relate to the projecting of total forecasted construction contract revenues, costs and profits in accordance with accounting for long-term contracts, and the recognition of potential liabilities in conjunction with certain contingencies, including the outcome of pending or future litigation, arbitration or other dispute resolution proceedings relating to contract claims. Actual results could differ from these estimates, and such differences could be material.

Tutor-Saliba believes, based on its experience, that its current systems of management and accounting controls allow management to produce materially reliable estimates of total contract revenue and cost during any accounting period. However, many factors can and do change during a contract performance period which can result in a change to contract profitability from one financial reporting period to another. Some of the factors that can change the estimate of total contract revenue and cost include differing site conditions (to the extent that contract remedies are unavailable), the availability of skilled contract labor, the ability of major material suppliers to deliver on time, the performance of major subcontractors, unusual weather conditions and the accuracy of the original bid estimate. Because Tutor-Saliba has multiple contracts in process at any given time, these changes in estimates can offset each other without impacting overall profitability. However, large changes

in cost estimates on larger, more complex construction projects can have a material impact on Tutor-Saliba s financial statements and are reflected in its results of operations when they become known. This impact could include reversal of profits recorded in prior periods.

When recording revenue on contracts relating to unapproved change orders and claims, Tutor-Saliba includes in revenue an amount equal to the amount of costs incurred to date for contract price adjustments that Tutor-Saliba sought to collect from customers for delays, errors in specifications or designs, change orders in dispute or unapproved as to scope or price, or other unanticipated additional costs, in each case when recovery of the costs is considered probable. When determining the likelihood of eventual recovery, Tutor-Saliba considers such factors as evaluation of entitlement, settlements reached to date and Tutor-Saliba s experience with the customer. The settlement of these issues often takes years depending upon whether the item can be resolved directly with the customer or involves litigation or arbitration. When new facts become known, an adjustment to the estimated recovery is made and reflected in the current period results.

The amount of unapproved change order and claim revenue is included in Tutor-Saliba s balance sheet as part of Unbilled Work. The amount of Unbilled Work relating to unapproved change orders and claims included in Tutor-Saliba s balance sheet at March 31, 2008 and December 31, 2007 is summarized below:

#### Unbilled Work at December 31, 2007 and March 31, 2008

	December 31, 2007	Mare	ch 31, 2008			
	(in tho	(in thousands)				
Unbilled Costs and Profits	\$ 3,545	\$	6,849			
Unapproved Change Orders	14,123		13,855			
Claims	11,351		11,682			
Total	\$ 29,019	\$	32,386			

Of the balance of unapproved change orders and claims included in Unbilled Work at March 31, 2008, approximately half belong to Tutor-Saliba s recently purchased, wholly owned subsidiary, Powerco Electric Corp., which was acquired in September 2007. These amounts are management s estimate of the probable recovery from the disputed claims considering the factors noted above evaluation of entitlement, settlements reached to date and Tutor-Saliba s experience with the customer. In the event that future facts and circumstances, including the resolution of disputed claims, cause a reduction in the aggregate amount of Tutor-Saliba s estimated probable recovery from the disputed claims, the amount of such reduction against future earnings will be recorded in the relevant period.

*Purchase Price Allocation.* In accordance with Statement of Financial Accounting Standards (SFAS) No. 141, *Business Combinations*, the purchase price of acquired properties is allocated to tangible and identified intangible assets and liabilities based on their respective fair values.

Tutor-Saliba s Powerco and Desert Plumbing acquisitions were stock purchases in which the price paid exceeded the value of the net tangible balance sheet assets purchased. The total price paid consisted of Tutor-Saliba s cash outlay and the assumption of the acquired balance sheet liabilities and, in the case of Desert Plumbing, contingent payments of \$4 million for each of three years upon satisfaction of specific performance targets, or upon completion of the merger.

Tutor-Saliba works with an independent valuation firm in identifying and evaluating the intangible assets associated with its acquisitions. Any intangible asset that cannot be assigned a value within the framework of SFAS No. 141, is considered goodwill. The subsequent amortization of intangible assets, including the impairment testing of goodwill, is done in accordance with SFAS No. 142, *Goodwill and Other Intangible Assets*.

Accounting for Income Taxes. Tutor-Saliba accounts for income taxes in accordance with SFAS No. 109, Accounting for Income Taxes, which requires the use of an asset and liability method of accounting for income taxes. Deferred income taxes are provided to reflect the tax effect of differences between the financial statement carrying amounts and the tax bases of assets and liabilities. Tutor-Saliba elected to be treated as a subchapter S corporation effective January 1, 1996. As a result, taxable income, loss and credits flow directly to the shareholders and tax related assets and liabilities of Tutor-Saliba become the obligation of the shareholders of the subchapter S corporation and are no longer reflected in the financial statements. The deferred tax assets, liabilities and provision reflected in the financial statements are those that do not flow through to the shareholders, as they relate to taxable subsidiaries. The subchapter S corporation status will terminate on the effective date of the merger. As a result of that termination, taxable income, loss, credits and related deferred taxes will be reflected in the financial statements for periods reported after the completion of the merger.

Information relating to Tutor-Saliba s provision for income taxes and the status of its deferred tax assets and liabilities is presented in Note 6, Income Taxes of Notes to Tutor-Saliba s 2007 Consolidated Financial Statements attached as *Annex H* to this proxy statement.

Accounting for Joint Ventures. Prior to 2007, Tutor-Saliba s non-controlling interests in construction joint ventures were accounted for on an equity method in Tutor-Saliba s Consolidated Balance Sheets and Consolidated Statements of Cash Flow and on the proportionate consolidation method in the Consolidated Statements of Income. Beginning in 2007, construction joint venture interests are accounted for using the proportionate consolidation method in the Consolidated Balance Sheets as well as the Consolidated Statements of Income whereby Tutor-Saliba s proportionate share of each joint venture s assets, liabilities, revenues and cost of operations are included in the appropriate classifications in the consolidated financial statements. Tutor-Saliba believes the change, which results in presenting all joint venture activity using a consistent methodology in both Consolidated Balance Sheets and Consolidated Statements of Income, is preferable.

The change had no impact on the Consolidated Statements of Income for any period presented. Although the change impacted various classifications within the Consolidated Balance Sheets and Consolidated Statements of Income, there was no impact to Shareholders Equity. Prior years Consolidated Balance Sheets and Consolidated Statements of Cash flows have been changed to conform to the 2007 presentation.

### Results of Operations Three Months Ended March 31, 2008 Compared to Three Months Ended March 31, 2007

	Revenue	es For The		
	Three Mo	nths Ended		
	Mar	ch 31,	Chang	e
	2007	2008	\$	%
	(in tho	usands)		
Domestic Building	\$ 160,346	\$ 341,470	\$181,124	113.0%
Domestic Civil	56,135	34,196	(21,939)	(39.1)%
International	15,579	19,297	3,718	23.9%
Total Revenues	\$ 232,060	\$ 394,963	\$ 162,903	70.2%

*Revenues*. For the three months ended March 31, 2008, total revenues were \$395 million, compared to \$232 million in the three months ended March 31, 2007, an increase of 70.2%. The revenue increase was driven by Tutor-Saliba s domestic building segment, which primarily benefited from the two major projects (Wynn Encore and Planet Hollywood Tower in Las Vegas) that were awarded in 2006 and which were in process throughout the 2007 period and performing at peak production levels during the first quarter of 2008, in addition to \$29.6 million in revenues provided by Desert Plumbing & Heating and Powerco Electric, both of which Tutor-Saliba acquired since the first quarter of 2007. Domestic building revenues were \$341 million in the 2008 period compared to \$160 million in the 2007 period. Domestic civil revenues were \$34 million in the 2007 period. Domestic civil segment revenues decreased during the period

primarily due to the work on the Los Angeles International Airport Runway and Taxiway Improvements project, which started construction in late 2005 and was in the later stages of completion during 2008. International revenues increased to \$19.3 million (or 24%) in the first three months of 2008 from \$15.6 million in the first three months of 2007 primarily as a result of the performance of the British Embassy project in the Philippines that was awarded in 2007 and is scheduled to be completed in the first half of 2008.

*Gross Profit.* Overall gross profit increased to \$34.6 million in the three months ended March 31, 2008 from \$20.7 million in the prior year, with gross margin decreasing slightly to 8.8% for the 2008 three-month period from 8.9% in the prior year. Gross margin declined due primarily to a larger proportion of the revenues in the first three months of 2008 being from the domestic building segment, which has a lower overall gross margin as compared to the domestic civil segment. The domestic civil segment had a decline in revenues in the first three months of 2008 as compared to 2007.

*Depreciation and Amortization Expense*. Depreciation and amortization expense increased to \$2.9 million in the three months ended March 31, 2008 from \$1.1 million in the prior year period. This increase is primarily in the domestic segments and results from a full period of depreciation from assets acquired in 2007, which were necessary to support Tutor-Saliba s increased volume of work, and the amortization of \$1.2 million of intangible assets recorded in connection with the purchase accounting for the Powerco Electric and Desert Plumbing & Heating acquisitions.

*General and Administrative Expense*. Tutor-Saliba s general and administrative expenses increased to \$13.6 million (3.4% of revenue) in the three months ended March 31, 2008 from \$7.8 million (3.3% of revenue) in the three months ended March 31, 2007. Nonetheless, Tutor-Saliba s general and administrative expenses increased at a lower rate than revenue indicating that it was able to leverage its corporate overhead over a much larger revenue base. The 2008 increase included an increase in costs in legal, accounting and other professional services in the amount of \$1.8 million, primarily related to the work performed in preparation for the merger and other potential strategic transactions.

The most significant costs in Tutor-Saliba s G&A expenses are employment-related.

In addition to a \$1.1 million increase in salaries and other related benefit costs paid in 2008 driven by the increase in number of personnel necessary to support Tutor-Saliba s expanding operations and increases in salaries and bonuses necessary to reward and retain existing personnel, there was a compensation charge incurred in the amount of \$1.1 million as a result of distributions made to certain shareholder executives being recorded as compensation pursuant to the provisions of Statement of Financial Accounting Standards No. 123R, Share-Based Payments.

The revenue and expense trends discussed above resulted in the following trends in income from operations performance.

	Income Fre For The Thre Ma	-	Chan	ge	
	2007		2008	\$	%
	(in th	ousands	)		
Domestic Building	\$ 6,722	\$	21,644	\$ 14,922	222.0%
Domestic Civil	6,308		6,560	252	4.0%
International	1,555		1,269	(286)	(18.4)%
Subtotal	\$ 14,585	\$	29,473	\$ 14,888	102.1%
Less: Corporate	(1,633)		(8,277)	(6,644)	406.9%
Total Income From Operations	\$ 12,952	\$	21,196	\$ 8,244	63.7%

*Other Income*. Total other income, net decreased to \$0.5 million in the three months ended March 31, 2008 from \$11.1 million in the three months ended March 31, 2007, primarily as a result of a gain on sale of marketable securities of \$11.4 million being recognized in the 2007 period.

*Interest and Other Income*. Interest and other income increased to \$1.4 million in the first three months of 2008 from \$0.8 million in the first three months of 2007 as a result of higher cash balances made possible by the sale of marketable securities, which increased available balances for investment in the 2008 period. The increase in available investment balances more than offset lower interest yields during the period.

*Interest Expense*. Interest expense decreased to \$0.8 million in the first three months of 2008 from \$1.0 million in the first three months of 2007 as Tutor-Saliba used a portion of its excess cash to reduce debt, including paying off a \$10 million term loan prior to entering into a new credit agreement. Interest expense also benefited from lower rates in the 2008 period as compared to the same period in 2007.

*Income Taxes*. Tutor-Saliba s provision for income taxes decreased to \$0.6 million in the three months ended March 31, 2008 from \$0.8 million in the three months ended March 31, 2007 due to a decrease in taxable income in 2008, primarily in Tutor-Saliba s foreign subsidiaries. Although Tutor-Saliba Corporation is a subchapter S corporation that passes most of its taxes through to its shareholders, a decrease in subsidiary income in Guam, taxed at 35%, resulted in the majority of the tax provision decrease. Tutor-Saliba will cease to be a subchapter S corporation upon the consummation of the merger, which it expects will have the impact described in the pro forma financial statements included elsewhere in this proxy statement.

*Income from Continuing Operations*. Tutor-Saliba s improved operating performance in the three months ended March 31, 2008 partially offset the \$11.4 million net gain recognized on the sale of marketable securities in the same period in 2007, resulting in income from continuing operations of \$21.1 million for the period in 2008, as compared to \$23.3 million for the same period in 2007.

#### Results of Operations Year Ended December 31, 2007 Compared to Year Ended December 31, 2006

Unless stated otherwise, all references to 2007, 2006, and 2005 refer to our fiscal years ended, or the dates as of December 31, 2007, December 31, 2006, and December 31, 2005.

	Revenue	<b>Revenues For The</b>				
	Year Ended	Year Ended December 31,				
	2006	2007	\$	%		
	(in the	ousands)				
Domestic Building	\$ 320,403	\$ 896,124	\$ 575,722	179.7%		
Domestic Civil	141,875	174,277	32,402	22.8%		
International	52,179	81,423	29,243	56.0%		
Total Revenues	\$ 514,457	\$ 1,151,824	\$ 637,367	123.9%		

*Revenues*. For the year ended December 31, 2007, total revenues were \$1.15 billion, compared to \$514.5 million in the year ended December 31, 2006, an increase of 124%. The revenue increase was driven primarily by Tutor-Saliba s domestic building segment, which had revenue of \$896.1 million in 2007, as compared to \$320.4 million in 2006. The increase was primarily due to two major Las Vegas projects, the Wynn Encore (\$581.2 million) and the Planet Hollywood Tower (\$112.9 million), that Tutor-Saliba was awarded in 2006 and which were in progress throughout 2007. In 2006, only the Wynn Encore had significant activity and produced revenue in the last six months of 2006 totaling \$120.2 million.

Domestic civil revenues were \$174.3 million in 2007 compared to \$141.9 million in 2006. The increase was primarily due to the full-period work on the Los Angeles International Airport runway project performed at full capacity, which generated \$90.6 million of revenue in 2007, as compared to \$64.7 in 2006. This project started construction in late 2005 and only produced significant revenue in the second half of 2006. This 2007 increase more than offset the impact of projects that were completed, or were in the later stages of completion, during 2006 and 2007.

International revenues increased to \$81.4 million in 2007 from \$52.2 million in 2006 primarily as a result of two projects, the Alpha Bravo Wharf Improvements project which was awarded in 2006 and which produced a

full-period of work in 2007, and the Home Depot project which was awarded in early 2007 and produced revenues of \$22.2 million during the fiscal year.

*Gross Profit.* Total gross profit increased to \$88.2 million in 2007 from \$29 million in 2006, with gross margin improving to 7.7% for 2007 from 5.6% the prior year. Gross margin improved due primarily to the higher margins obtained on the new contracts acquired in 2006 in both the domestic and international segments, the increased revenues in the higher margin domestic civil segment and the larger proportion of domestic building segment revenues in 2007 from higher margin projects than Tutor-Saliba s 2006 project mix.

*Depreciation and Amortization Expense*. Depreciation and amortization expense increased to \$6.1 million in 2007 from \$3.7 million in 2006. This increase is primarily in the domestic segments and results from a full period of depreciation from assets acquired in 2006 and the additional depreciation from additional assets acquired in 2007, all of which were necessary to support Tutor-Saliba s increased volume of work, and the amortization of \$1.2 million of intangible assets recorded in the purchase accounting for the acquisition of Powerco Electric.

*General and Administrative Expense*. Tutor-Saliba s general and administrative expenses increased to \$36.2 million (3.1% of revenue) in 2007 from \$26.8 million (5.2% of revenue) in 2006. Tutor-Saliba s general and administrative expenses increased at a lower rate than revenue as it was able to leverage its corporate overhead over a much larger revenue base. The 2007 increase included the cost of expanding Tutor-Saliba s Las Vegas office to add support for its expanding operations in that city, increased costs incurred in accounting, legal and other professional services in preparation for a contemplated capital markets transaction and the negative impact of fuel costs during the period. The 2006 period was significantly impacted by the cost of starting up Tutor-Saliba s Las Vegas operations and opening an office there. The most significant cost in Tutor-Saliba s general and administrative expenses are employment-related. In addition to a \$1.1 million increase in bonuses paid in 2007, salaries and other related benefit costs increased by \$2.8 million. These increases were driven by the increase in number of personnel necessary to support Tutor-Saliba s expanding operations and increases in salaries and bonuses necessary to reward and retain existing personnel.

The revenue and expense trends discussed above resulted in the following trends in operating income performance:

	For The Y	n Operations 'ear Ended ber 31,	Chai	nge
	2006 2007 \$ (in thousands)			%
Domestic Building	\$ 4,840	\$ 39,651	\$ 34,811	719.2%
Domestic Civil	9,061	26,135	17,074	188.4%
International	4,796	7,832	3,036	63.3%
Subtotal	\$ 18,697	\$ 73,618	\$ 54,921	293.7%
Less: Corporate	(16,043)	(21,189)	(5,146)	32.1%
Total Income From Operations	\$ 2,654	\$ 52,429	\$ 49,775	1,875.5%

*Other Income*. Total other income, net increased to \$95 million in 2007 from \$6.2 million in 2006 as a result of a gain on sale of marketable securities of \$94.1 million in 2007, generated from the sale of Tutor-Saliba s remaining 3 million shares of Perini common stock.

*Interest and Other Income.* Interest and other income increased to \$6.1 million in 2007 from \$3.4 million in 2006 as a result of higher cash balances made possible by the sale of marketable securities and the increased available balances for investment in 2007.

*Interest Expense.* Interest expense decreased to \$4.2 million in 2007 from \$5.3 million in 2006 as a portion of excess cash was used to reduce debt, including paying off a \$10 million term loan prior to entering into a new credit agreement.

*Income Taxes.* The provision for income taxes increased to \$4.4 million in 2007 from \$1.7 million in 2006 due to the increase in taxable income in 2007. Although Tutor-Saliba is a subchapter S corporation that passes most of its taxes through to its shareholders, increases in subsidiary income in Guam, taxed as a C corporation at 35%, resulted in the tax provision increase. Tutor-Saliba will cease to be a subchapter S corporation upon completion of the merger, which it expects will have the impact described in its pro forma financial statements included elsewhere in this proxy statement.

*Income from Continuing Operations*. Overall improved operating performance and gains recognized on the sale of Perini common stock resulted in income from continuing operations of \$142.9 million for 2007, as compared to \$7.2 million in 2006.

#### Results of Operations Year Ended December 31, 2006 Compared to Year Ended December 31, 2005

	<b>Revenues For The</b>		
	Year Ended December 31,	Change	
	2005 2006 \$	8 %	
	(in thousands)		
Domestic Building	\$ 228,905 \$ 320,403 \$ 91	,498 40.0%	
Domestic Civil	117,046 141,875 24	,829 21.2%	
International	41,639 52,179 10	0,540 25.3%	
Total Revenues	\$ 387,590 \$ 514,457 \$ 126	5,867 32.7%	

*Revenues.* Total revenues increased to \$514.5 million in 2006 from \$387.6 million in 2005. This increase was due mainly to an increase in domestic building construction revenues of \$91.5 million (or 40.0%) to \$320.4 million in 2006 from \$228.9 million in 2005, reflecting the increased volume of work in the hospitality and gaming market where Tutor-Saliba commenced work on two major projects (Wynn Encore and Planet Hollywood Towers) in Las Vegas in 2006 as well as the start of construction on a major building project in Los Angeles (Los Angeles Police Headquarters). Revenues earned in 2006 from the Wynn Encore project totaled \$120.2 million.

Civil construction revenues increased by \$24.8 million (or 21.2%) to \$141.9 million in 2006 from \$117 million in 2005 due primarily to \$64.7 million of revenue produced from a full year s production at the Los Angeles International Airport Runway and Taxiway Improvements project during 2006. The runway project more than offset the impact of projects that were completed, or were in the later stages of completion, during 2005 and 2006.

International revenues increased by \$10.5 million to \$52.2 million in 2006 from \$41.6 million in 2005 as the island of Guam operations rebounded from a relatively down year in 2005 when several major projects either were completed, or were in the latter stages of completion. Operations benefited from revenue totaling \$18 million from the start of the Alpha Bravo Wharf improvements project on the island of Guam as well as obtaining additional new work on the surrounding Micronesian Islands.

*Gross Profit.* Total gross profit increased to \$29 million in 2006 from \$16.3 million in 2005, with gross margin improving to 5.6% for 2006 from 4.2% in 2005 primarily due to higher margins on contracts entered into in 2006 in the domestic and international segments. Gross profit margin is a function of the mix of projects ongoing throughout each period. Typically, civil projects, which are usually performed for state and local governments and are typically obtained through competitive bidding, tend to produce higher gross profit margins than do building projects, which include both public and private customers, and are either competitively bid or are awarded under negotiated contract arrangements.

*Depreciation and Amortization Expense*. Depreciation and amortization expense increased to \$3.7 million in 2006 from \$2.9 million in 2005. The increase reflects an expansion in Tutor-Saliba s equipment fleet necessary to support its increased workload from major contracts acquired in 2006, primarily the Wynn Encore and Planet Hollywood Towers in Las Vegas. In 2006 we purchased \$22.1 million of property and equipment, compared to \$6.7 million in 2005.

*General and Administrative Expense*. Tutor-Saliba s general and administrative expenses increased to \$26.8 million (5.2% of revenue) in 2006 from \$26.4 million (6.8% of revenue) in 2005. Tutor-Saliba s general and administrative expenses increased at a lower rate than revenue as it was able to leverage its corporate overhead over a much larger revenue base.

*Litigation Settlement Expense*. In 2005, Tutor-Saliba recorded a charge of \$14.7 million in its domestic building segment to settle contract litigation relating to its completed project at the San Francisco International Airport. Tutor-Saliba had no litigation settlement related costs for this matter in 2006.

The revenue and expense trends discussed above resulted in the following trends in operating income performance.

	In	Income (Loss) From Operations For The Year Ended December 31,				ze
		2005		2006	\$	%
		(in th	ousands	)		
Domestic Building	\$	(14,838)	\$	4,840	\$ 19,678	n.a.
Domestic Civil		5,894		9,061	3,167	53.7%
International		1,476		4,796	3,320	224.9%
Subtotal	\$	(7,468)	\$	18,697	\$ 26,165	n.a.
Less: Corporate		(9,756)		(16,043)	(6,287)	64.4%
Total Income (Loss) From Operations	\$	(17,224)	\$	2,654	\$ 19,878	n.a.

*Other Income.* Total other income, net decreased to \$6.2 million in 2006 primarily due to the gain on sale of marketable securities of \$8.9 million in 2006 being far less than the combined gains of \$38.3 million on sale of marketable securities and derivatives relating to such marketable securities in 2005.

Interest and Other Income. Interest and other income increased to \$3.4 million in 2006 from \$2.7 million in 2005 as a result of increased amounts in retention escrow accounts and higher yields on funds invested.

*Interest Expense*. Interest expense increased to \$5.3 million in 2006 from \$3.4 million in 2005 as Tutor-Saliba increased borrowings, particularly in the second half of 2006 to support its increased working capital requirements associated with starting new large projects as discussed above, and the effect of higher interest rates during the period.

*Income Taxes.* Our provision for income taxes increased to \$1.7 million in 2006 from \$0.8 million in 2005. Although Tutor-Saliba Corporation is a subchapter S corporation that passes most of its taxes through to its shareholders, increases in subsidiary income in Guam, taxed as a C corporation at 35%, resulted in the tax provision increase.

*Income from Continuing Operations.* Income from continuing operations decreased to \$7.2 million in the year ended December 31, 2006 from \$19 million in the year ended December 31, 2005, as the decline in the other income more than offset the impact of Tutor-Saliba s improved operating performance.

### Liquidity and Capital Resources

### **Cash and Working Capital**

Cash and cash equivalents as reported in Tutor-Saliba s Consolidated Statements of Cash Flows consist of amounts held by Tutor-Saliba as well as its proportionate share of amounts held by construction joint ventures. Cash held by Tutor-Saliba is available for general corporate purposes while cash held by construction joint ventures is available only for joint venture-related uses. Cash held by construction joint ventures is distributed from time to time to Tutor-Saliba and to the other joint venture participants in accordance with their respective percentage interests after the joint venture partners determine that a cash distribution is prudent. Cash distributions received by Tutor-Saliba from construction joint ventures are then available for Tutor-Saliba s general corporate purposes. At March 31, 2008 and 2007, cash held by Tutor-Saliba and available for general corporate purposes was \$110.3 million and \$51.6 million, respectively, and Tutor-Saliba s proportionate share of cash held by joint ventures and available only for joint venture-related uses was \$11.5 million and \$23 million, respectively. In the first quarter of 2008, Tutor-Saliba used \$35 million in cash to acquire Desert Plumbing & Heating and \$27.9 million to pay a distribution to its shareholders in respect of 2007 income attributable to them as owners of a subchapter S corporation, offset in part by continuing strong cash flow from operations. Tutor-Saliba expects to use a substantial portion of its available cash balance to fund a final subchapter S distribution to Tutor-Saliba s shareholders prior to consummation of the merger.

Billing procedures in the construction industry generally are based on the specific billing terms of each contract. For example, billings may be based on various measures of performance, such as cubic yards excavated, architect s estimates of completion, costs incurred on cost-plus type contracts or weighted progress from a cost loaded construction time schedule. Billings are generally on a monthly basis and are reviewed and approved by the customer prior to submission. Therefore, once a bill is submitted, Tutor-Saliba is generally able to collect amounts owed to it in accordance with the payment terms of the contract. In addition, receivables of a contractor usually include retentions, or amounts that are held back until contracts are completed or until specified contract conditions or guarantees are met. Retentions are governed by contract provisions and are typically a fixed percentage (for example, 5% or 10%) of each billing. Tutor-Saliba generally follows the policy of paying its vendors and subcontractors on a particular project after it receives payment from its customer.

Tutor-Saliba s primary uses of cash have been for increases in working capital to fund new projects and investments in joint ventures, purchases of equipment and distributions to shareholders. When Tutor-Saliba commences a new project, it generally uses cash for approximately two months as it obtains materials and incurs payroll and other costs pending its first billing to and payment from the customer. In September 2007 and January 2008, Tutor-Saliba used cash resources to acquire two subcontractors (Powerco Electric Corp. and Desert Plumbing & Heating), in the amount of \$3.3 million and \$35 million, respectively. Tutor-Saliba s primary sources of cash have been from operations, particularly in periods where it is able to reduce working capital as existing projects move beyond their startup phase, borrowings, and the sale of marketable securities. A summary of cash flows for each of the three months ended March 31, 2008 and 2007 is set forth below:

	Marc 2007	Three Months Ended March 31, 2007 2008 Unaudited		
	(in mi	llions)		
Cash flows from:				
Operating activities	\$ (0.4)	\$ 2.2		
Investing activities	19.9	(51.3)		
Financing activities	(5.5)	(33.4)		
Net increase (decrease) in cash	\$ 14.0	\$ (82.5)		
Cash at beginning of period	37.6	192.8		
Cash at end of period	\$ 51.6	\$ 110.3		

In the first three months of 2008, Tutor-Saliba generated \$2.2 million in cash flow in operating activities and used \$51.3 million in cash flow from investing activities and \$33.4 million in financing activities for a net decrease in its cash balance of \$82.5 million. Tutor-Saliba s positive cash flow from operating activities primarily reflects net income less the increase in receivables of \$39.7 million and the increase in other assets of \$2 million, partially offset by depreciation and amortization expense of \$2.9 million, a \$21.2 million increase in payables resulting from costs incurred on its construction projects, and \$1.7 million of unbilled work on other projects. Cash flow from investing activities primarily reflects a \$12.8 million of cash used for purchases of property and equipment, \$15.3 million of advances to related parties, and \$31.8 million for acquisitions, partially offset by proceeds from the sale of property and equipment of \$0.7 million and \$8.0 million of cash becoming unrestricted during the period. Cash used in financing activities includes a \$2.7 million net reduction in debt and \$30.7 million in shareholder distributions.

In the first three months of 2007, Tutor-Saliba generated \$19.9 million from investing activities while using \$0.4 million for operating activities and funding \$5.5 million in financing activities for a net increase in its cash balance of \$14.0 million. Tutor-Saliba s use of cash in operating activities primarily reflects increases in accounts receivable and unbilled work of \$42.6 million and \$8 million, respectively, a deduction from net income for a gain on sale of marketable securities of \$11.4 million, and an increase in other assets of \$2.4 million, offset by an increase in accounts payable and other liabilities of \$38.3 million, and an increase in deferred contract revenue of \$1.6 million. Cash provided from investing activities was primarily from proceeds from sales of marketable securities of \$23.8 million, partially offset by net purchases of property and equipment of \$3.9 million. Cash flow from financing activities primarily reflects net proceeds from long-term debt of \$2.4 million, offset by principal payments of long-term debt of \$6.1 million and distributions to shareholders of \$1.8 million.

A summary of cash flows for each of the years ended December 31, 2007, 2006 and 2005 is set forth below:

	Year	Year Ended December 31,		
	2005	2006 (in millions)	2007	
Cash flows from:				
Operating activities	\$ 6.3	\$ (28.8)	\$115.2	
Investing activities	35.2	0.6	104.4	
Financing activities	(28.0)	10.0	(64.4)	
Net increase (decrease) in cash	13.5	(18.2)	155.2	
Cash at beginning of year	42.3	55.8	37.6	
Cash at end of year	\$ 55.8	\$ 37.6	\$ 192.8	

During 2007, Tutor-Saliba generated \$115.2 million from operating activities and \$104.4 million from investing activities while using \$64.4 million from financing activities, for an increase in its cash balance of \$155.2 million. Cash provided by operating activities resulted primarily from net income being offset by the non-operating gain on sale of securities of \$94.1 million and a \$59 million increase in receivables due to new projects started during the year, offset in part by depreciation and amortization of \$6.6 million, an increase in accounts payable of \$111.2 million and a \$4.1 million increase in deferred contract revenue. Cash provided by investing activities reflects proceeds from sales of marketable securities of \$147.7 million and proceeds from the sale of property and equipment of \$4.6 million, partially offset by purchases of property and equipment of \$23.8 million and \$24 million of cash being designated as restricted. The use of cash for financing activities primarily reflects principal payments of long-term debt of \$21.1 million, paydown on credit line of \$10 million, and distributions to shareholders totaling \$51.8 million, partially offset by proceeds from long-term debt of \$18.5 million.

During 2006, Tutor-Saliba generated \$10 million in cash flow from financing activities and \$0.6 million from investing activities, while using \$28.8 million for operating activities for a net decrease in its cash balance

of \$18.2 million. Cash flow from financing activities primarily reflects a \$30 million increase in net borrowings offset in part by \$20 million in shareholder distributions. Cash used in operating activities resulted primarily from net income being offset by the non-operating gain on sale of securities of \$8.9 million and a \$59.1 million increase in receivables due to new projects started during the year, offset in part by depreciation and amortization of \$5.4 million, an increase in accounts payable of \$9.4 million, and a \$16 million increase in deferred contract revenue, and a \$2.2 million decrease in unbilled work. The cash flow from investing activities reflects net purchases of property and equipment of \$19.3 million and advances to related parties of \$1.3 million, offset by proceeds from the sale of marketable securities of \$21.2 million.

During 2005, Tutor-Saliba generated \$35.2 million from investing activities and \$6.3 million from operating activities and consumed \$28 million in financing activities for a net increase its cash balance of \$13.5 million. Cash flow from operating activities resulted primarily from net income being offset by the non-operating gains on the sale of securities, property and equipment, and derivatives of \$21.4 million, \$8.3 million, and \$16.9 million, respectively, and an increase in other assets of \$4.8 million partially offset by a decrease in receivables of \$21.1 million, a decrease in unbilled work of \$1.8 million, an increase in accounts payable and other liabilities of \$4.8 million, an increase in deferred contract revenue of \$5.2 million, and non-cash depreciation and amortization expense of \$4.4 million. Cash flow from investing activities primarily reflects proceeds from sale of marketable securities in the amount of \$29.3 million, and the net proceeds from the sale of equipment in the amount of \$6.4 million. The cash flow used in financing activities comes from the net reduction of long-term debt of \$13.2 million and shareholder distributions of \$14.8 million.

#### **Revolving Credit Agreement**

On September 17, 2007, Tutor-Saliba entered into a Revolving Credit Agreement with Comerica Bank, as administrative agent, and two participant lenders (referred to in this proxy statement as the Credit Agreement). Subsequently, on May 12, 2008, Tutor-Saliba executed the First Amendment to Revolving Credit Agreement.

The Credit Agreement as amended provides for a secured revolving credit facility (the Revolving Facility ) of up to \$50 million with an \$8 million sublimit for commercial and standby letters of credit and a term loan in the amount of \$25 million. This represents an increased borrowing capacity from Tutor-Saliba s prior agreement, which provided for a revolving credit facility of \$25 million, plus a term loan in the original amount of \$10 million, all of which was outstanding at December 31, 2006. The \$10 million term loan was paid in full on June 4, 2007.

Tutor-Saliba s interest rate alternatives include a prime-based rate, as well as Eurocurrency rate-based options. The Credit Agreement also supports letters of credit of up to \$8 million, which reduces availability under the Revolving Facility on a dollar-for-dollar basis. The termination date of the Revolving Facility is August 31, 2009, and the \$25 million term loan maturity date is May 1, 2011, with mandatory principal payments of \$5 million on May 1, 2009 and \$10 million on May 1, 2010 and 2011.

The Credit Agreement as amended requires Tutor-Saliba to comply with certain financial and other covenants at the end of each fiscal quarter, including:

tangible net worth of at least \$130 million as of March 31, 2008, increased to \$150 million as of the first fiscal quarter ending September 30, 2008, and again increased to \$185 million as of the fiscal quarter ending December 31, 2008 through maturity;

a minimum current ratio of at least 1.20:1.00;

minimum core operating profitability of at least \$25 million for the two quarter period ending June 30, 2008; \$100 million for the fiscal year ending December 31, 2008 and \$150 million for the fiscal year-to-date period ending December 31, 2009 and thereafter; and

maintain, at all times, a minimum liquidity (cash and/or marketable securities) of \$40 million.

The Credit Agreement also includes operational covenants customary for facilities of this type, including limitations on incurring additional indebtedness and liens, as well as restrictions on types of investments and the purchase and sale of assets outside of the normal course of business. Tutor-Saliba s obligations under the Credit Agreement are guaranteed by Mr. Tutor, and Tutor-Saliba s subsidiary, Black Construction, and are secured by substantially all of Tutor-Saliba s assets. Tutor-Saliba expects to substantially amend or replace its credit facility prior to completion of the merger. Pursuant to the terms of the Merger Agreement, Perini has agreed to use its commercially reasonable efforts to remove or release Mr. Tutor and his affiliates as obligors, guarantors or sureties of any obligations of Tutor-Saliba or its subsidiaries under certain contacts, including the Credit Agreement, and any similar obligation incurred from and after the date of the Merger Agreement.

#### **Off-Balance Sheet Arrangements**

Tutor-Saliba does not have any financial partnerships with unconsolidated entities, such as entities often referred to as structured finance, special purpose entities or variable interest entities which are often established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes. Accordingly, Tutor-Saliba is not exposed to any financing, liquidity, market or credit risk that could arise if Tutor-Saliba had such relationships.

### Long-Term Debt

Total debt at March 31, 2008, including current maturities, was \$38.7 million, a decrease of \$35.1 million from March 31, 2007, due primarily to distribution of the debt associated with a commercial office building that was distributed to an affiliate of Mr. Tutor in the amount of \$24.4 million, and pay off of the term loan and the outstanding line of credit under Tutor-Saliba s prior credit agreement of \$6 million and \$10 million, respectively. These pay offs were offset by the financing of additional purchases of more equipment to meet the demands of increased project work.

#### **Contractual Obligations**

Tutor-Saliba s outstanding contractual obligations as of March 31, 2008 are summarized in the following table:

	Total	Less	Fhan 1 Year (in t	1 3 years thousands)	3 5 years	>5 Years
Long-term debt (1)	\$46,981	\$	14,629	\$ 22,814	\$ 9,538	\$
Capital leases	1,547		615	853	79	
Operating leases (2)	43,985		5,754	10,838	10,833	16,560
Other long-term liabilities (3)	7,008		2,628	4,380		
Total contractual obligations	\$ 99,521	\$	23,626	\$ 38,885	\$ 20,450	\$ 16,560

(1) Includes estimated interest payments, which are based on our projected interest rates and estimated principal amounts outstanding for the period presented.

(2) Operating leases primarily consist of a corporate aircraft and our corporate office space in Sylmar, California

(3) Consists of payments related to legal settlement with the City and County of San Francisco (see footnote 13)

#### **Recent Accounting Pronouncements**

In July 2006, the Financial Accounting Standards Board (FASB) issued FASB Interpretation No. 48, *Accounting for Uncertainty in Income Taxes-an Interpretation of FASB Statement No. 109* (FIN 48). FIN 48 increases the relevancy and comparability of financial reporting by clarifying the way companies account for uncertainty in measuring income taxes. FIN 48 prescribes a comprehensive model for how a company should recognize, measure, present, and disclose in its financial statements uncertain tax positions that the company has taken or expects to take on a tax return. FIN 48 only allows a favorable tax position to be included in the calculation of tax liabilities and expenses if a company concludes that it is more likely than not that its adopted tax position will prevail if challenged by tax authorities. FIN 48 also provides guidance on the accounting for and recording of interest and penalties on uncertain tax positions. Tutor-Saliba adopted FIN 48 on January 1, 2007, and the adoption of FIN 48 did not have a material impact on Tutor-Saliba s financial condition and results of operations.

In September 2006, the FASB issued SFAS No. 157, *Fair Value Measurements* (SFAS No. 157), which clarifies the definition of fair value, describes methods used to appropriately measure fair value, and expands fair value disclosure requirements. SFAS No. 157 applies under other accounting pronouncements that currently require or permit fair value measurements. SFAS No. 157 is effective as of the beginning of the first fiscal year after November 15, 2007. In February 2008, the FASB issued FASB Staff Position (FSP) No. FAS 157-1 and FSP No. FAS 157-2, affecting implementation of FASB Statement No. 157. FSP No. FAS 157-1 excludes FASB Statement No. 13, *Accounting for Leases*, and other accounting pronouncements that address fair value measurements under FASB Statement No. 13, from the scope of FASB Statement No. 157. FSP No. FAS 157-2 delays the effective date of FASB Statement No. 157 for nonfinancial assets and nonfinancial liabilities, except for items that are recognized or disclosed at fair value on a recurring basis, to fiscal years beginning after November 15, 2008. The adoption of FASB Statement No. 157 did not have a material impact on Tutor-Saliba s financial condition and results of operations.

In February 2007, the FASB issued SFAS No. 159, *The Fair Value Option for Financial Assets and Financial Liabilities* (SFAS No. 159). SFAS No. 159 permits entities to choose to measure many financial instruments and certain other items at fair value. The objective of the guidance is to improve financial reporting by providing entities with the opportunity to mitigate volatility in reported earnings caused by measuring related assets and liabilities differently without having to apply complex hedge accounting provisions. SFAS No. 159 is effective as of the beginning of the first fiscal year after November 15, 2007. The adoption of SFAS No. 159 did not have a material impact on Tutor-Saliba s financial condition and results of operations.

In December 2007, the FASB issued SFAS No. 141 (revised 2007), *Business Combinations* (SFAS No. 141(R)). SFAS No. 141(R) establishes principles and requirements for how an acquirer recognizes and measures in its financial statements the identifiable assets acquired, the liabilities assumed, any noncontrolling interest in the acquiree and the goodwill acquired. SFAS No. 141(R) also establishes disclosure requirements to enable the evaluation of the nature and financial effects of the business combination. SFAS No. 141(R) is effective for fiscal years beginning after December 15, 2008. Tutor-Saliba will apply the provisions of SFAS No. 141(R) prospectively as of that date.

In December 2007, the FASB issued SFAS No. 160, *Noncontrolling Interests in Consolidated Financial Statements-an amendment of Accounting Research Bulletin No. 51* (SFAS No. 160). SFAS No. 160 establishes accounting and reporting standards for ownership interests in subsidiaries held by parties other than the parent, the amount of consolidated net income attributable to the parent and to the noncontrolling interest, changes in a parent s ownership interest, and the valuation of retained noncontrolling equity investments when a subsidiary is deconsolidated. SFAS No. 160 also establishes disclosure requirements that clearly identify and distinguish between the interests of the parent and the interests of the noncontrolling owners. SFAS No. 160 is effective for fiscal years beginning after December 15, 2008. Tutor-Saliba is currently evaluating the potential impact, if any, of the adoption of SFAS No. 160 on Tutor-Saliba s financial condition and results of operations.

#### **Selected Financial Information**

The selected financial data set forth below is derived in part from and should be read in conjunction with Tutor-Saliba s consolidated financial statements, the related notes and Management s Discussion and Analysis of Financial Condition and Results of Operations included elsewhere in this proxy statement. The consolidated statement of income data for each of the years ended December 31, 2005, 2006 and 2007 and the consolidated balance sheet data as of December 31, 2006 and 2007 were derived from Tutor-Saliba s audited consolidated financial statements appearing elsewhere in this proxy statement. The consolidated statement of income data for the three-month periods ended March 31, 2007 and 2008 and the consolidated balance sheet data as of March 31, 2007 and 2008 were derived from Tutor-Saliba s unaudited consolidated financial statements appearing elsewhere in this proxy statement. This information is unaudited but, in management s opinion, has been prepared on the same basis as the audited consolidated financial statements and related notes included elsewhere in this proxy statement and includes all adjustments, consisting only of normal recurring adjustments, that Tutor-Saliba s management considers necessary for a fair presentation of the information for the periods presented. Historical results are not necessarily indicative of results to be expected for future periods.

		onths Ended rch 31, 2007	2007 (in thous	Year 1 2006 ands, except sl	Ended December 2005	r 31, 2004	2003
Consolidated statement of income data:			(in thous	ands, except si	lare data)		
Revenues:							
Domestic civil	\$ 34,196	\$ 56,135	\$ 174,277	\$ 141,875	\$ 117,046	\$ 194,082	\$ 208,543
Domestic building	341,470	160,346	\$ 174,277 896,124	320,403	228,905	250,836	\$ 208,343 204,275
International	19,297	15,579	890,124	52,179	41.639	71,118	66,126
International	19,297	15,579	01,425	52,179	41,039	/1,110	00,120
T-4-1	204.062	222.060	1 151 904	514 457	297 500	516.026	479.044
Total	394,963	232,060	1,151,824	514,457	387,590	516,036	478,944
Cost of revenues	360,359	211,404	1,063,603	485,434	371,334	495,103	448,338
Gross profit	34,604	20,656	88,221	29,023	16,256	20,933	30,606
Gain on sale of property and equipment	(149)	(50)	(446)	(454)	(7,524)		
Cost of contract litigation settlement					14,652(4)		
General and administrative expenses	13,557	7,754	36,237	26,823	26,352	22,739	21,702
Income (loss) from operations	21,196	12,952	52,430	2,654	(17,224)	(1,806)	8,904
Other income, net (1)	1,263	12,142	99,206	11,479	40,504	17,114	11,593
Interest expense	(779)	(1,005)	(4,197)	(5,257)	(3,414)	(3,750)	(4,519)
•							
Income before minority interest and income							
taxes	21,680	24,089	147,439	8,876	19,866	11,558	15,978
Provision for income taxes (3)	(619)		(4,399)	(1,663)	(849)	(309)	(2,166)
rovision for medine taxes (5)	(01)	(121)	(1,3)))	(1,005)	(01))	(30))	(2,100)
Income before minority interest	21.061	23.292	143.040	7.213	19.017	11.249	13.812
Minority interest	21,001	(20)	(111)	(40)	(16)	(204)	(29)
wintomy interest		(20)	(111)	(40)	(10)	(204)	(29)
T C / · · /·	01.0(1	22.272	1 42 020	7 172	10.001	11.045	10 700
Income from continuing operations	21,061	23,272	142,929	7,173	19,001	11,045	13,783
Income (loss) from discontinued operations, net			226	(20)	1 405	11.000	(70)
of taxes (2)		(39)	226	(36)	1,407	11,803	(79)
Net income	\$ 21,061	\$ 23,233	\$ 143,155	\$ 7,137	\$ 20,408	\$ 22,848	\$ 13,704
Income per share from continuing operations	\$ 23.40	\$ 25.85	\$ 158.80	\$ 7.97	\$ 21.11	\$ 12.27	\$ 15.31
Income (loss) per share from discontinued							
operations, net of taxes		(0.04)	0.25	(0.04)	1.56	13.11	(0.09)
Net income per share	\$ 23.40	\$ 25.81	\$ 159.05	\$ 7.93	\$ 22.67	\$ 25.38	\$ 15.22
i vet meome per share	φ 25.40	φ 25.01	φ 157.05	Ψ 1.75	φ <i>22.01</i>	φ 25.50	φ 10.22

## Edgar Filing: IDEX CORP /DE/ - Form 10-K

Book value per share	\$ 205.01	\$ 208.99	\$ 235.13	\$ 179.67	\$ 183.26	\$ 177.71	\$ 146.37

- (1) Primarily reflects unrealized holding gains and gains on sales of marketable securities.
- (2) Reflects the results of operations associated with equipment and real estate operations unrelated to construction activities, including a gain on sale of equipment of \$11.8 million in 2004.
- (3) Since January 1, 1996, Tutor-Saliba has been a subchapter S Corporation and has been exempt from paying federal income taxes. In addition, from and after the day Tutor-Saliba elected or was otherwise treated as a subchapter S corporation for state tax purposes, Tutor-Saliba has paid certain state taxes at a reduced rate.
- (4) Represents charge recorded related to settlement of contract litigation on the San Francisco International Airport project.

	March 31,						
	2008	2007	2007	2006	2005	2004	2003
				(in thousands)	1		
Consolidated balance sheet data:							
Working capital	\$ 96,311	\$ 151,352	\$ 147,541	\$ 126,885	\$ 133,659	\$ 137,381	\$ 47,305
Current ratio	1.26x	1.61x	1.45x	1.60x	1.80x	1.94x	1.26x
Long term debt, less current maturities	26,905	60,038	53,617	60,221	38,456	50,398	61,582
Shareholders equity	184,515	188,096	211,630	161,713	164,939	159,948	131,740
Ratio of long-term debt to equity	.15x	.32x	.25x	.37x	.23x	.32x	.47x
Total assets	603,980	503,741	601,493	441,757	382,748	364,956	379,476

#### UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS

The unaudited pro forma condensed combined balance sheet as of March 31, 2008 and the unaudited pro forma condensed combined statements of income for the year ended December 31, 2007 and the three months ended March 31, 2008 are based on the separate historical consolidated financial statements of Perini and Tutor-Saliba. These unaudited pro forma condensed combined financial statements reflect the merger and related events using the purchase method of accounting and apply the assumptions and adjustments described in the accompanying notes to the unaudited pro forma condensed combined financial statements. The unaudited pro forma condensed combined balance sheet as of March 31, 2008 reflects the merger and related events as if they had been consummated on March 31, 2008. The unaudited pro forma condensed combined statements of income for the year ended December 31, 2007 and the three months ended March 31, 2008 reflect the merger and related events as if they had been consummated on March 31, 2008. The unaudited pro forma condensed combined statements of income for the year ended December 31, 2007 and the three months ended March 31, 2008 reflect the merger and related events as if they had been consummated on March 31, 2008 reflect the merger and related events as if they had been consummated on March 31, 2008 reflect the merger and related events as if they had been consummated on March 31, 2008 reflect the merger and related events as if they had been consummated on January 1, 2007, the beginning of Perini s 2007 fiscal year.

The pro forma adjustments are based upon available information and assumptions that the managements of Perini and Tutor-Saliba believe reasonably reflect the merger. We present the unaudited pro forma condensed combined financial statements for informational purposes only. The pro forma condensed combined financial statements are not necessarily indicative of what our financial position or results of operations actually would have been had we completed the merger as of the dates indicated. In addition, the unaudited pro forma condensed combined financial statements do not purport to project the future financial position or operating results of the combined company. You should read this information together with the following:

the accompanying notes to the unaudited pro forma condensed combined financial statements;

the separate historical unaudited financial statements of Perini as of and for the three months ended March 31, 2008 included in Perini s Quarterly Report on Form 10-Q for the quarterly period ended March 31, 2008, which are incorporated by reference into this proxy statement;

the separate historical audited financial statements of Perini as of and for the fiscal year ended December 31, 2007 included in Perini s Annual Report on Form 10-K for the fiscal year ended December 31, 2007, which are incorporated by reference into this proxy statement;

the separate historical unaudited financial statements of Tutor-Saliba as of and for the three months ended March 31, 2008, which are included in the annexes to this proxy statement; and

the separate historical audited financial statements of Tutor-Saliba as of and for the fiscal year ended December 31, 2007, which are included in the annexes to this proxy statement.

We prepared the unaudited pro forma condensed combined financial statements using the purchase method of accounting, with Perini as the acquirer. Accordingly, the total estimated purchase price, calculated as described in Note 1 to the unaudited pro forma condensed combined financial statements, is allocated to the net tangible and identifiable intangible assets of Tutor-Saliba acquired in connection with the merger, based on their respective fair values. Should there be an increase in the fair value of the Tutor-Saliba tangible and/or intangible assets as of the closing date of the merger, the amount of the purchase price allocated to these assets will increase accordingly, resulting in a decrease in the amount of goodwill recorded and an increase in depreciation expense and/or amortization expense. A 10% increase in the fair value of the depreciable tangible assets could increase depreciation expense by approximately \$0.1 million per year. A 10% increase in the fair value of amortizable intangible assets could increase amortization expense by \$1.1 million per year.

The allocation is dependent upon valuations and other studies that have not progressed to a stage where there is sufficient information to make a definitive allocation. Accordingly, the purchase price allocation pro forma adjustments are preliminary and have been made solely for the purpose of providing unaudited pro forma condensed combined financial statements. The final purchase price allocation, which will be determined subsequent to the closing of the merger, and its effect on results of operations may differ significantly from the pro forma amounts included in the unaudited pro forma condensed combined financial statements. These amounts represent the managements best estimate as of the date of this proxy statement. In order to provide a definitive accounting of the purchase price allocation as of the date of the closing of the merger, Perini will retain

valuation specialists to help establish the fair value of the net tangible and intangible assets of Tutor-Saliba as of the closing date. These valuations will primarily include valuations of the fair value of fixed assets, intangible assets such as trade name, existing customer relationships, favorable lease terms on existing leases and existing construction contract backlog. In addition, Perini will review and adjust the effective tax rate as required, and adjust estimated transaction costs to actual. Statement of Financial Accounting Standards No. 141 Business Combinations allows the acquiring company one year to complete the final analysis and accounting for the purchase price allocation related to a business combination.

In connection with the plan to integrate the operations of Perini and Tutor-Saliba, we anticipate that non-recurring charges, such as costs associated with systems implementation, relocation expenses, severance and other costs associated with exit or disposal activities, will be incurred. We are not able to determine the timing, nature and amount of these charges as of the date of this proxy statement/prospectus. However, these charges could affect the combined results of operations of Perini and Tutor-Saliba, as well as those of the combined company following the merger, in the period in which they are recorded. The unaudited pro forma condensed combined financial statements do not include the effects of the costs associated with any restructuring or integration activities resulting from the transaction, as they are non-recurring in nature and not factually supportable at the time that the unaudited pro forma condensed combined financial statements were prepared. In addition, the unaudited pro forma condensed combined financial statements were prepared. In addition, the unaudited pro forma condensed combined financial statements were prepared. In addition, the unaudited pro forma condensed combined financial statements were prepared. In addition, the unaudited pro forma condensed combined financial statements were prepared. In addition, the unaudited pro forma condensed combined financial statements were prepared. In addition, the unaudited pro forma condensed combined financial statements were prepared. In addition, the unaudited pro forma condensed combined financial statements were prepared. In we are save as a statements were prepared. In the transaction, nor do they include any potential incremental revenues and earnings that may be achieved with the combined capabilities of the companies.

# UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET

# MARCH 31, 2008

	Perini Corporation (historical)	Tutor-Saliba (historical)	Pro Forma Adjustments (In thousands)	Note 2	Pro Forma Combined
ASSETS:					
Cash and cash equivalents	\$ 349,749	\$ 110,328	\$ (64,290)	(g)	\$ 395,496
			(291)	(h)	
Restricted cash		16,000			16,000
Short-term investments	110,337				110,337
Accounts receivable, including retainage	1,013,399	286,906	(8,002)	(i)	1,291,903
			(400)	(g)	
Costs and estimated earnings in excess of billings	93,577	32,386			125,963
Advances to related parties		15,310	(15,310)	(g)	
Deferred income taxes	5,964	193			6,157
Other current assets	3,764	11,290	3,300	(a)	18,354
Total current assets	1,576,790	472,413	(84,993)		1,964,210
Property and equipment, net	98,698	85,198	16,133	(a)	197,412
		,	(2,617)	(h)	,
Goodwill	27,268	21,181	746,397	(b)	773,665
			(21,181)	(b)	
Purchased intangible assets, net	3,916	10,689	234,900	(c)	238,816
			(10,689)	(c)	
Mineral right assets		12,850			12,850
Other assets	23,507	1,649	(294)	(f)	24,862
	\$ 1,730,179	\$ 603,980	\$ 877,656		\$ 3,211,815

# UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET (continued)

# MARCH 31, 2008

	Perini Corporation (historical)	Tutor-Saliba (historical)	Pro Forma Adjustments (In thousands)	Note 2	Pro Forma Combined
LIABILITIES AND STOCKHOLDERS EQUITY:					
Current maturities of long-term debt	\$ 5,683	\$ 11,822			\$ 17,505
Accounts payable, including retainage	959,168	278,850	(8,002)	(i)	1,230,016
Billings in excess of costs and estimated earnings	186,470	57,613			244,083
Accrued expenses	128,447	27,817	30,212	(f)	186,476
Total current liabilities	1,279,768	376,102	22,210		1,678,080
Long-term debt, less current maturities included above	13,635	26,905	(291)	(h)	40,249
Deferred income taxes	471	1,311	87,395	(d)	107,770
			18,593	(d)	
Royalty liabilities		11,361	, i i i i i i i i i i i i i i i i i i i	, í	11,361
Other long-term liabilities	39,951	3,786	55,000	(g)	98,737
Stockholders equity:					
Common stock	27,147	6,757	22,987	(e)	50,134
			(6,757)	(e)	
Additional paid-in capital	163,371	11,076	856,277	(e)	1,019,648
			(11,076)	(e)	
Retained earnings	223,353	166,581	(135,000)	(g)	223,353
			(11,065)	(e)	
			(2,617)	(h)	
			(18,593)	(d)	
			694	(f)	
Accumulated other comprehensive income (loss)	(17,517)	101	(101)	(e)	(17,517)
Total stockholders equity	396,354	184,515	694,749		1,275,618
	\$ 1,730,179	\$ 603,980	\$ 877,656		\$ 3,211,815

# UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF INCOME

# FOR THE THREE MONTHS ENDED MARCH 31, 2008

		Perini istorical)	tor-Saliba istorical) (in thousand	Ad	o Forma justments ept per share	Note 2 e data)		ro Forma ombined
Revenues	\$ 1	,256,336	\$ 394,963	\$	(11,113)	(0)	\$ 1	,640,186
Cost of operations	1	,189,774	360,210		2,793	(k)	1	,540,677
					253	(1)		
					(11,113)	(0)		
					(1,240)	(j)		
Gross profit		66,562	34,753		(1,806)			99,509
General and administrative expenses		27,599	13,557		256	(r)		40,718
					(694)	(q)		
Income from construction operations		38,963	21,196		(1,368)			58,791
Other income (expense), net		1,505	1,263					2,768
Interest expense		(355)	(779)		(688)	(p)		(1,822)
Income before income taxes		40,113	21,680		(2,056)			59,737
Provision for income taxes		(14,960)	(619)		(7,533)	(s)		(22,339)
					773	(t)		
Net income	\$	25,153	\$ 21,061	\$	(8,816)		\$	37,398
Basic earnings per common share	\$	0.93					\$	0.75
Diluted earnings per common share	\$	0.91					\$	0.74
Weighted average common shares outstanding:								
Basic		27,145			22,987	(u)		50,132
Diluted		27,653			22,987			50,640

# UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF INCOME

## FOR THE YEAR ENDED DECEMBER 31, 2007

	(h	Perini istorical)		itor-Saliba historical) (in thousand	Adj	o Forma ustments pt per share	Note 2 data)		ro Forma Combined
Revenues	\$	1,628,358	\$	1,151,824	\$	127,499	(n)	\$	5,875,295
Revenues	φ-	r,020,330	ψ	1,131,024	ψ	(32,386)	(II) (0)	ψ.	5,675,295
Cost of operations	4	1,379,464		1,063,157		11,170	(b)		5,510,638
		.,,		1,000,107		761	(1)		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
						(883)	(m)		
						90,586	(n)		
						(32,386)	(0)		
						(1,231)	(j)		
Gross profit		248,894		88,667		27,096			364,657
General and administrative expenses		107,913		36,237		1,069	(r)		159,495
						14,276	(n)		
Income from construction operations		140,981		52,430		11,751			205,162
Other income (expense), net		15,361		99,206		63	(n)		114,630(1)
Interest expense		(1,947)		(4,197)		(2,750)	(p)		(8,962)
						(68)	(n)		
Income before minority interest and income taxes		154,395		147,439		8,996			310,830
Provision for income taxes		(57,281)		(4,399)		(51,038)	(s)		(116,100)
						(3,382)	(t)		
Income before minority interest		97,114		143,040		(45,424)			194,730
Minority interest				(111)					(111)
Income from continuing operations		97,114		142,929		(45,424)			194,619
Income from discontinued operations, net of tax				226		(226)	(m)		
Net income	\$	97,114	\$	143,155	\$	(45,650)		\$	194,619
Basic earnings per common share	\$	3.62						\$	3.91
Diluted earnings per common share	\$	3.54						\$	3.86(2)
Weighted average common shares outstanding:		04 6 1 0				22.005			10.001
Basic		26,819				22,987	(u)		49,806
Diluted		27,419				22,987			50,406

(1) Includes \$94,105 non-recurring gain on sale of marketable securities.

(2) Pro forma diluted earnings per share excluding the non-recurring gain on sale of marketable securities (see Note (1) above) is \$2.70.

#### NOTES TO UNAUDITED PRO FORMA CONDENSED

# COMBINED FINANCIAL STATEMENTS

#### 1. Basis of Pro Forma Presentation

On April 2, 2008, Perini and Tutor-Saliba entered into the Merger Agreement, pursuant to which Tutor-Saliba will merge into a wholly owned subsidiary of Perini, with Perini continuing as the surviving corporation. The transaction is to be accounted for using the purchase method of accounting. For purposes of these unaudited pro forma condensed combined financial statements, Perini has assumed the total preliminary purchase consideration in the merger to be approximately \$898 million, consisting of shares of Perini common stock valued at \$879 million, and approximately \$19 million in transaction costs, to be paid by Perini.

Under the terms of the Merger Agreement, Perini will issue 22,987,293 shares of Perini common stock to the Tutor-Saliba shareholders in exchange for their shares of Tutor-Saliba common stock.

The unaudited pro forma condensed combined balance sheet contains a preliminary estimate of the purchase price allocation, assuming a per share value of Perini common stock of \$38.25, the closing market price of Perini common stock on April 2, 2008.

The preliminary unaudited pro forma condensed combined financial statements have been prepared assuming that the merger is accounted for using the purchase method of accounting, which is referred to as purchase accounting, with Perini as the acquiring entity. Accordingly, under purchase accounting, the assets, liabilities and commitments of Tutor-Saliba are adjusted to their fair values. The preliminary unaudited pro forma condensed combined financial statements do not reflect the impact of possible revenue enhancements, cost and expense efficiencies, synergies or asset dispositions. The preliminary unaudited pro forma condensed combined financial statements do not reflect to be determined or charges or credits that are not expected to have a continuing impact after twelve months succeeding the merger.

The preliminary unaudited pro forma adjustments represent each management s estimates based on information available as of the time this proxy statement was prepared and are subject to revision as additional information becomes available and as additional analyses are performed.

The final allocation of the purchase price will be determined after the merger is consummated and after completion of a thorough analysis to determine the fair values of Tutor-Saliba s tangible and identifiable intangible assets and liabilities. Accordingly, the final purchase accounting adjustments could be materially different from the preliminary unaudited pro forma adjustments presented herein. Any increase or decrease in the fair values of Tutor-Saliba s assets, liabilities, and other items, as compared to the information shown herein, will change the portion of the purchase price allocable to goodwill and will impact the combined income statement due to adjustments in amortization or accretion related to the adjusted assets or liabilities.

Based on Perini s shares of common stock and equity awards outstanding as of April 2, 2008, and assuming that all of the equity awards outstanding as of April 2, 2008 remain outstanding as of the effective time of the merger, the total preliminary estimated purchase price is as follows:

#### Stock consideration

Shares of Perini common stock to be issued for Tutor-Saliba common stock outstanding based on the closing market		
price of Perini common stock on April 2, 2008 of \$38.25	22,987	\$ 879,264
Total gross consideration		\$ 879,264
Estimated transaction costs, to be paid by Perini		19,200
Total preliminary estimated purchase price		\$ 898,464

#### NOTES TO UNAUDITED PRO FORMA CONDENSED

#### COMBINED FINANCIAL STATEMENTS (Continued)

Under the purchase method of accounting, the total preliminary estimated purchase price as shown in the table above is allocated to Tutor-Saliba s tangible and intangible assets and liabilities based on their estimated fair values as of the date of completion of the merger. The total preliminary estimated purchase price is allocated herein as follows:

	Amounts (In thousands)
Net tangible assets as of March 31, 2008 at estimated fair value	\$ 4,562
Identifiable intangible assets:	
Trade name	\$ 137,000
Customer relationships	51,200
Favorable leases	23,000
Contract backlog acquired	11,500
Nevada contractor s licenses	12,200
Total amount allocated to identifiable intangible assets	234,900
Deferred tax liabilities	(87,395)
Amount allocated to goodwill	746,397
Total preliminary estimated purchase price	\$ 898,464

A preliminary estimate of \$4.6 million has been allocated to net tangible assets acquired, \$85.7 million has been allocated to amortizable intangible assets acquired, and \$149.2 million has been allocated to non-amortizable intangible assets. The depreciation and amortization related to the fair value adjustment to net tangible assets and the amortization related to the amortizable intangible assets are reflected as pro forma adjustments to the unaudited pro forma condensed combined statements of income.

*Identifiable intangible assets.* Of the total estimated purchase price, \$234.9 million has been allocated to trade name, customer relationships, favorable leases, contract backlog acquired and other. This adjustment is preliminary and based on managements estimate. The amount that will ultimately be allocated to identifiable intangible assets may differ materially from this preliminary allocation. Customer relationships are amortized using the straight-line method over an estimated useful life of fifteen years based on an estimate of repeat business with certain major customers. Contract backlog acquired is amortized using the straight-line method over an estimated useful life of 2.5 years. Favorable leases are amortized over the remaining terms of the leases. The Tutor-Saliba trade name and Nevada contractor s licenses held by a subsidiary of Tutor-Saliba are the only non-amortizable intangible assets anticipated to be acquired by Perini pursuant to the merger. The Tutor-Saliba trade name was determined to have an indefinite life on the basis that Tutor-Saliba and its subsidiaries intend to continue to operate under their existing names with no existing or contemplated plan to initiate any material changes to their licensed business operations. The Nevada contractor s licenses were determined to have indefinite lives on the basis that the licenses are critical to the success of the subsidiary s operations for an indefinite period of time and that the subsidiary would not be able to successfully operate without these licenses in the state of Nevada, the only state in which it operates. Although these licenses require renewal every two years, they are renewable as a matter of course for a minimal fee and without requiring material modification of any existing terms or conditions of the licenses.

*Deferred tax liabilities.* The deferred tax liabilities reflect the estimated deferred tax liabilities associated with purchase accounting. Such deferred tax liabilities are primarily associated with the step-up to fair value of identifiable intangible assets. This determination is preliminary and subject to change based upon the final determination of the fair values of identifiable intangible assets acquired.

#### NOTES TO UNAUDITED PRO FORMA CONDENSED

#### COMBINED FINANCIAL STATEMENTS (Continued)

Goodwill. Goodwill represents the excess of the purchase price over the fair value of the underlying net tangible and intangible assets.

The preliminary estimated amount of goodwill resulting from the merger of \$746.4 million generally represents the value of Tutor-Saliba s geographic market presence, accumulated experience as a general contractor, relationships with suppliers and subcontractors, management team and assembled workforce and the ability of these elements to contribute to the generation of significant future cash flows. Perini believes that the merger will provide substantial strategic and financial benefits to the company beyond what another potential market participant could realize, including the following:

increased scale and greater diversification of our business;

entry into additional high growth and high-margin markets and projects;

consolidation of Ronald N. Tutor s, our chief executive officer and chairman, management activities on the growth and development of the combined company, and elimination of risk that he might leave Perini to focus on Tutor-Saliba;

additional management depth and enhanced management capabilities;

enhanced commercial building and civil business operations, due to the complementary and synergistic strengths of the two companies in these market segments;

greater opportunities to win new, substantial contracts to drive accelerated growth;

ability to use the strength of Perini s balance sheet to win additional large civil and public works projects that required surety capacity in excess of what Tutor-Saliba was able to obtain; and

opportunities to realize significant synergies.

Please see Questions and Answers About the Merger and the Meeting Why has Perini decided to merge with Tutor-Saliba? beginning on page 1 and Reasons for the Merger beginning on page 47 for a detailed discussion of the reasons for and benefits of the merger and Opinion of UBS Securities LLC beginning on page 59 for information regarding the Special Committee s valuation of Tutor-Saliba.

In accordance with Statement of Financial Accounting Standards No. 142, *Goodwill and Other Intangible Assets*, goodwill will not be amortized, but instead will be tested for impairment at least annually and whenever events or circumstances have occurred that may indicate a possible impairment. In the event the combined management determines that the value of goodwill has become impaired, the combined company will incur an accounting charge for the amount of the impairment during the period in which the determination is made.

#### 2. Pro Forma Adjustments

Adjustments included in the column under the heading Pro Forma Adjustments in the unaudited pro forma condensed combined financial statements correspond to the following descriptions:

#### Pro Forma Adjustments to Condensed Combined Balance Sheet

(a) Adjust Tutor-Saliba s tangible fixed assets to fair value based upon a preliminary valuation study performed by management using available information from equipment dealers and financing institutions. The step-up to fair value in property and equipment primarily reflects a favorable market for used construction equipment and the overall good condition of the Tutor-Saliba equipment fleet. The step-up to fair value in other current assets reflects a favorable market for used steel maintained in inventory.

(b) Eliminate Tutor-Saliba s historical goodwill and record preliminary goodwill resulting from the merger. See Note 1 for a more detailed discussion.

1	2	n
T	4	υ

#### NOTES TO UNAUDITED PRO FORMA CONDENSED

#### COMBINED FINANCIAL STATEMENTS (Continued)

(c) Eliminate Tutor-Saliba s historical intangible assets and record the preliminary estimated identifiable intangible assets, which include the acquired trade name, customer relationships, contract backlog acquired, favorable leases and other. See Note 1 for a more detailed discussion.

(d) Record deferred income tax liabilities of \$87,395 at 37.6% of the estimated identifiable intangible assets of \$232,433. Pursuant to the merger, also record a pro forma adjustment of \$18,593 to revalue the deferred income tax liabilities of Tutor-Saliba due to a change in tax rate from the subchapter S corporation election valuation rate to a subchapter C corporation valuation rate. The pro forma adjustments to the deferred income tax liabilities result in a corresponding increase in goodwill.

(e) Eliminate Tutor-Saliba s historical equity balances and record shares of Perini common stock issued as a result of the merger.

(f) Accrue estimated remaining incremental direct and external transaction costs of \$18.2 million (net of \$0.3 million included in Other Assets as deferred costs and \$0.7 million recorded as expense in 2008) comprised of investment banking fees (\$11.5 million), legal fees (\$4.3 million), accounting fees (\$0.6 million), due diligence expenses (\$0.6 million), and filing, printing and other costs related to the merger (\$1.2 million). These estimated transaction costs were developed based on actual invoices received and managements estimates of the various remaining professional services to be provided. Also includes a \$12 million accrual for the payment of contingent consideration by Tutor-Saliba related to the acquisition of Desert Plumbing & Heating Co. consummated by Tutor-Saliba in January 2008 described under Information About Tutor-Saliba Recent Developments and Expected 2008 Events beginning on page 94. The contingent consideration payment of \$12 million becomes fixed and accelerated upon Tutor-Saliba s merger with Perini.

(g) Record the estimated distributions to be paid to Tutor-Saliba shareholders prior to the closing of the merger transaction as described under Pre-closing Distribution of Property beginning on page 70 (excluding the distributions of the office building, which was completed prior to March 31, 2008, and the Idaho residence, which is considered in Note 2(h) below) and Pre-Closing Payment of Dividends; Dividend Notes beginning on page 79. The estimated amounts of these distributions were developed by management based on the Tutor-Saliba retained earnings and cash balances at December 31, 2007, the estimated April 2008 subchapter S corporation tax liabilities of the Tutor-Saliba shareholders, actual cash distributions made to shareholders of Tutor-Saliba in 2008, and a projection of the amount of Tutor-Saliba cash available for distribution as of the closing date.

(h) Record distribution of Ketchum, Idaho property to the Tutor-Saliba shareholders prior to the merger.

(i) Eliminate intercompany balances between Perini and Tutor-Saliba.

#### NOTES TO UNAUDITED PRO FORMA CONDENSED

#### COMBINED FINANCIAL STATEMENTS (Continued)

#### Pro Forma Adjustments to Condensed Combined Statements of Income

(j) Reverse Tutor-Saliba s amortization of intangible assets.

(k) Record the amortization of the purchased intangible assets resulting from the merger. The purchased intangible assets consist of the estimated fair value of the acquired trade name, customer relationships, favorable leases, contract backlog acquired and other. (See Note 1)

	Three Months Ended March 31, 2008 (In the	Fiscal Year Ended December 31, 2007 ousands)		
Amortization of purchased intangible assets:	· · · · · · · · · · · · · · · · · · ·		, 	
Customer relationships	\$ 854	\$	3,413	
Favorable leases	789		3,157	
Contract backlog acquired	1,150		4,600	
	\$ 2,793	\$	11,170	
Trade name and other intangible assets	n.a.		n.a.	

(1) Record the estimated incremental depreciation expense resulting from the step-up to fair value of the fixed assets acquired in the merger.

(m) Reverse the impact of certain properties distributed to Tutor-Saliba s shareholders prior to the merger.

(n) Record the impact of Tutor-Saliba s acquisitions of companies in September 2007 and January 2008, which are described under Information About Tutor-Saliba Recent Developments and Expected 2008 Events beginning on page 94, as if they were both acquired as of January 1, 2007. The pro forma amounts were derived from financial statements of the acquired companies as of the respective closing dates of the acquisition transactions.

- (o) Eliminate intercompany balances between Perini and Tutor-Saliba.
- (p) Record interest expense on distribution payable to Tutor-Saliba shareholders.
- (q) Reverse transaction costs included in Tutor-Saliba s G&A expense.

(r) Record estimated impact of the new employment agreement between Perini and Ronald N. Tutor, chairman and chief executive officer.

(s) Record the tax effect of an assumed statutory income tax rate of 37.6% on the historical pretax income of Tutor-Saliba.

(t) Record the tax effect of an assumed statutory tax rate of 37.6% on the pro forma adjustments made.

(u) The pro forma basic and diluted earnings per common share is based on the historical weighted-average number of shares of Perini common stock used in computing basic and diluted net income per share, plus approximately 23 million shares of Perini common stock assumed to be issued in connection with the merger based on the number of shares of Perini common stock outstanding as of April 2, 2008.

#### **PROPOSAL 3: ELECTION OF DIRECTORS**

The Perini board of directors (the Board ) has nominated four (4) Class III directors to serve until the 2011 annual meeting of shareholders. In accordance with our bylaws, each director nominee will be elected to serve for a three-year term, unless he or she resigns, dies or is removed before his or her term expires, or until his or her successor has been duly elected and qualified.

The following individuals are the nominees for election to the Board:

Name	Age	<b>Director Since</b>
Class III Nominees for Election		
Marilyn A. Alexander	56	
Peter Arkley	53	2000
Raymond R. Oneglia	60	2000
Donald D. Snyder	60	

The principal occupation and business experience of each director nominee for the last five years is set forth below:

*Marilyn A. Alexander* is a nominee for director. She founded and has served as a principal of Alexander & Friedman LLC, a management consulting company, since 2006. Prior to that, she was senior vice president and chief financial officer of The Disneyland Resort since 2000. She is also a member of the board of governors of Chapman University, a not-for-profit organization, a member of the board of regents of Chapman University, the president of the board of directors of the Breast Health Awareness Foundation, a not-for-profit foundation, and a member of the board of advisors of Walkstyles, Incorporated, a privately held company.

*Peter Arkley* has served as a director since May 2000. He has served as the President/CEO of AON Construction Services Group, an insurance and bonding brokerage firm, since 2006 and prior to that was Managing Principal of Aon Risk Services, Inc. since 1994. He is also a director of Valley Crest Companies, a privately held company, and of the Greater Los Angeles Zoo Association, a non-profit organization.

*Raymond R. Oneglia* has served as a director since March 2000. He has also served as Vice Chairman of the board of directors of O&G Industries, Inc., a Connecticut corporation engaged in the construction industry, since 1997 and has served in various operating and administrative capacities since 1970.

**Donald D. Snyder** is a nominee for director. He was a director and the president of Boyd Gaming Corp. from 1997 until his retirement in 2005. He presently serves as a director of Sierra Pacific Resources, a public utility listed on the NYSE, Western Alliance Bancorp., a commercial bank holding company listed on the NYSE, Cash Systems, Inc., a corporation listed on NASDAQ, Bank of Nevada, a commercial bank and subsidiary of Western Alliance Bancorp., and Switch Communications Group, LLC, a privately held company. He is presently on the board of directors of the following not-for-profit entities: Las Vegas Performing Arts Center Foundation, Nevada Development Authority, Council for a Better Nevada, University of Nevada-Las Vegas Foundation, and the Nathan Adelson Hospice.

In 2007, we hired Spencer Stuart, an executive and director search consulting firm, to assist in identifying and/or evaluating potential director nominees. Spencer Stuart identified Mr. Snyder and Ms. Alexander as nominees for director.

Our Corporate Governance and Nominating Committee has recommended Messrs. Arkley and Oneglia for re-election, and Ms. Alexander and Mr. Snyder for election, as Class III Directors. Unless otherwise noted thereon, proxies solicited hereby will be voted for the election of the director nominees to hold office until the

2011 annual meeting of shareholders, and until their successors are chosen and qualified. Each nominee has consented to being named in this proxy statement, and if elected, each nominee has consented to serve as a director until his or her successor is duly elected and qualified. The Board does not contemplate that any nominee will be unable to serve as a director for any reason, but if that should occur prior to the meeting, proxies solicited hereby may be voted either for a substitute nominee designated by the Board or recommended by the Corporate Governance and Nominating Committee, or the Board may determine to reduce the number of directors.

#### **Board Recommendation**

# THE PERINI BOARD OF DIRECTORS UNANIMOUSLY RECOMMENDS A VOTE FOR EACH OF THE BOARD OF DIRECTORS NOMINEES FOR ELECTION AS A CLASS III DIRECTOR.

#### **BOARD OF DIRECTORS**

The following table shows, as of June 30, 2008, the names and ages of our current directors and director nominees.

Name	Age	Position (as of June 30)	Term of Office
Ronald N. Tutor	67	Chairman of the Board, and Chief Executive Officer	2010
Robert Band	60	Director, President, and Chief Operating Officer	2009
Michael R. Klein	65	Vice Chairman of the Board, Director	2009
Peter Arkley	53	Director	2008
Raymond R. Oneglia	59	Director	2008
Robert L. Miller	67	Director	2009
Willard W. Brittain, Jr.	60	Director	2010
Robert A. Kennedy	72	Director	2010
Marilyn A. Alexander	57	Nominee for Director	
Donald D. Snyder	60	Nominee for Director	

For biographical summaries of the Class III directors, Messrs. Arkley and Oneglia, and the other director nominees, Mr. Snyder and Ms. Alexander, see the list of nominees above.

*Ronald N. Tutor* has served as our chief executive officer since March 2000, as chairman since July 1999, and as a director since January 1997. Mr. Tutor also serves as chairman, president and chief executive officer of Tutor-Saliba Corporation, a privately held California corporation engaged in the construction industry.

*Robert Band* has served as a director since May 1999. He has also served as chief operating officer since March 2000 and as president since May 1999. He has served as president of Perini Management Services, Inc. since 1996. He has served in various operating and financial positions with Perini Corporation since 1973, including executive vice president and chief financial officer from 1997-1999. He also serves as a director of Jewish Family Services of Metrowest, a not-for-profit entity.

*Michael R. Klein* has served as a director since January 1997 and as vice chairman of our board of directors since September 2000. He is also the designated lead director. Mr. Klein, a private investor, serves as chairman of the board of directors and member of the Nominating Committee of CoStar Group, Inc., a publicly held provider of commercial real estate information, as chairman of the Sunlight Foundation, a non-profit organization and as chairman of the board of directors of Le Paradou, LLC, a privately held company. He is also the lead director and member of the Governance Committee of SRA International, Inc., a publicly-traded provider of technology and strategic consulting services and solutions, and a director of AStar Air Cargo, Inc., and OZ Fitness, Inc., which are privately held. Mr. Klein was a partner of the law firm Wilmer Cutler Pickering from 1974 until 2004, and when Wilmer Cutler Pickering merged with the law firm Hale and Dorr LLP in 2004 became a partner of Wilmer Cutler Pickering Hale and Dorr LLP until his retirement in 2005.

*Robert L. Miller* has served as a director since 2004. In 1979, he co-founded West Venture Development Co., a homebuilding and commercial real estate company, and acted as president until its sale in 1991. Previously he was a construction manager with Morrison-Knudsen Inc. He was chairman of the board of Monroc Corp., a public concrete and aggregate company based in Salt Lake City, Utah from 1995-1998. He is presently on the valuations committee of Caltius Mezzanine Partners, a capital source company in Los Angeles. Since 2000, he is a principal in Robert L. Miller & Assoc., Inc., a real estate development firm.

*Willard W. Brittain, Jr.* became a director in November 2004. He has served as chairman and chief executive officer of Professional Resources on Demand, a private senior executive staffing company, since 2002. He previously served as chief operating officer of PwC Consulting since 2000, and chief operating officer of PricewaterhouseCoopers, LLP since 1997. Mr. Brittain also serves on the board of Analysts International, where he is a member of the audit and compensation committees, DaVita Corporation, where he serves on the compliance and public policy committees, and Convergys Corporation. All are publicly held companies.

*Robert A. Kennedy* has served as a director since March 2000. He has been an independent financial consultant since 2003. From 1993 to 2003, Mr. Kennedy served in various capacities, including as vice president special projects, for The Union Labor Life Insurance Company, a provider of insurance and financial services to its union members and related trust funds.

#### **Board Composition**

The Board currently consists of eight directors. In accordance with our bylaws and the requirements of the Massachusetts Business Corporations Act, the Board is divided into three classes, with each director serving for a term of three years. As a consequence, the term of only one class of directors expires each year. At each annual meeting of shareholders, the successors to one class of directors then serving are elected to serve from the time of their election and qualification until the third annual meeting following their election or until their successors have been duly elected and qualified, or until their earlier resignation, removal or death.

As a result of the resignation of James A. Cummings on December 18, 2007, the Board has had only two Class III directors and one vacant Class III seat. (Class I and Class II each consist of three directors.) On March 19, 2008, at the recommendation of the Corporate Governance and Nominating Committee, the Board unanimously determined to increase the size of the board from nine to eleven directors. An additional seat was added to Class III; there are now two vacant Class III seats. In addition, there is one open seat that will be filled by a designee of the shareholder representative upon completion of the merger. The Board, at the recommendation of the Corporate Governance and Nominating Committee, has unanimously nominated Ms. Alexander and Mr. Snyder for election as Class III directors. The two incumbent Class III directors have been re-nominated.

If the Perini shareholders approve the share issuance proposal (Proposal 1) and the articles amendment proposal (Proposal 2) and the merger is completed, the Board intends for the open seat to exist following the annual meeting to accommodate the addition of a director designated by Mr. Tutor pursuant to the terms of the Shareholders Agreement that will become effective upon completion of the merger. Under the Shareholders Agreement, Mr. Tutor (as shareholder representative) will have the right to designate up to two nominees for appointment to the Board (and thereafter, for nomination for election), subject to certain limitations contained in the Shareholders Agreement. In addition, for so long as Mr. Tutor serves as our chief executive officer, he will be nominated for election to the Board. For a detailed description of the post-merger board composition, please see Post-Merger Governance and Management Board of Directors beginning on page 90.

Mr. Tutor has advised the Corporate Governance and Nominating Committee that he has selected C.L. Max Nikias as one of his designees, to be appointed to the Board immediately following completion of the merger. The Corporate Governance and Nominating Committee has, after review of his qualifications, recommended Mr. Nikias appointment to the Board, and the full board unanimously approved the expansion of the board to

eleven members and the appointment of Mr. Nikias to fill the eleventh seat, subject to, and effective immediately following completion of, the merger. Mr. Nikias would be appointed as a Class II director, and he would serve the remaining two years of the term of a Class II director.

As of the date of this proxy statement, Mr. Tutor has not elected to exercise his right to nominate a second director for election, although he has not waived the right to do so in the future. Under the Shareholders Agreement, Mr. Tutor has the right to designate two of eleven directors. Mr. Tutor has advised the Board that should he choose to designate a second person for appointment to the board of directors at a time when the board already includes eleven members, he would support a temporary expansion of the board to twelve members to accommodate such additional member. Such expansion would continue until the next meeting of shareholders at which directors are elected, at which time he would expect the size of the board to be reduced back to eleven members (as contemplated by the Shareholders Agreement) and the slate of nominees for election to be adjusted accordingly.

The principal occupation and business experience of Mr. Nikias for the last five years is set forth below.

*C.L. Max Nikias* is provost and senior vice president of the University of Southern California since 2005 and prior to that was the Dean of the University of Southern California Viterbi School of Engineering since 2001. He also serves on the executive committee of the Chadwick School, a private primary and secondary school. As of the date of this proxy statement, Mr. Nikias does not beneficially own any shares of Perini or Tutor-Saliba common stock.

#### **Director Independence**

The Board has determined that Mr. Oneglia, Mr. Klein, Mr. Brittain, Mr. Miller, Mr. Kennedy, Mr. Arkley, Ms. Alexander and Mr. Snyder are independent in accordance with the corporate governance standards for companies listed on the NYSE. In determining independence pursuant to NYSE standards, each year the Board determines whether directors have a direct or indirect material relationship with Perini, including its subsidiaries, that may interfere with their ability to exercise their independence from Perini. In anticipation of the completion of the merger, the Board has made a determination that none of the independent directors will have a direct or indirect material relationship with the combined company, including its subsidiaries, that would interfere with their ability to exercise their independence from the combined company. Material relationships can include commercial, industrial, banking, consulting, legal, accounting, charitable, and familial relationships, among others.

In evaluating the independence of each non-employee director, the Board considered several factors. With respect to Mr. Oneglia, the board considered the relationship between O&G Industries, Inc., of which Mr. Oneglia is Vice Chairman of the board or directors and a principal shareholder, and Perini including the construction joint ventures between Perini and O&G Industries. The Board determined that the joint ventures did not impact Mr. Oneglia s independence from Perini management because (1) the joint ventures are formed for the limited purposes of performing specific contractual requirements for owners as is commonplace in the construction business, (2) Mr. Oneglia is not personally involved in the management of these joint ventures and (3) Perini and O&G have an equal vote in the governance of such joint ventures. With respect to Mr. Arkley, the Board considered the relationship between AON Risk Services (AON), of which Mr. Arkley is President of the Construction Services Group, and Perini, an insurance and bonding client of AON. The Board has determined that his independence from Perini management is not impacted because (1) services provided by AON are supplied to Perini on terms similar to AON s other clients and (2) Mr. Arkley is not involved in the day to day management of the Perini relationship. No other independent directors had material relationships with Perini other than in their capacities as directors. The Perini board of directors has made no formal findings regarding the independence of Mr. Nikias but plans to make such a determination prior to appointing him to the board.

#### Communications with the Board

The Board welcomes the submission of any comments or concerns from shareholders and other interested parties. A Perini shareholder who wishes to communicate with our board of directors may submit such communication in writing to Perini Corporation, 73 Mt. Wayte Avenue, Framingham, MA 01701 and marked to

the attention of the Board or any of its committees or individual directors. All comments or concerns from shareholders and other interested parties will be forwarded to the Chair of our Audit Committee.

In order to facilitate communications with the independent directors, we have a secure telephone number (800-489-8689) whereby interested parties can communicate directly and confidentially with the independent directors, the Audit Committee or the Corporate Governance and Nominating Committee.

#### CORPORATE GOVERNANCE

#### **Committees and Meetings of the Board of Directors**

The Board met four times during 2007. During 2007, all of our directors attended at least 75% of (i) the total number of meetings of the Board and (ii) the total number of meetings held by all committees on which such director served. The members of the Board are encouraged to attend our annual shareholders meetings. Seven of the eight current directors attended the 2007 annual shareholders meeting.

Our bylaws authorize the Board to appoint one or more committees, each consisting of one or more directors. The Board currently has three standing committees: an Audit Committee, a Corporate Governance and Nominating Committee and a Compensation Committee. As described above, the Board has also created a Special Committee to explore and evaluate potential strategic transactions that might be available to Perini, including in particular a business combination transaction with Tutor-Saliba, and to discuss and negotiate the terms of any such transactions.

#### Audit Committee

Our board of directors has an Audit Committee, which consists of Willard W. Brittain, Jr. (Chair), Michael R. Klein, Raymond R. Oneglia and Robert A. Kennedy. Each of the members of the Audit Committee meets the independence and experience requirements of the rules of the NYSE and the Securities and Exchange Commission, as affirmed by the Board. Based upon review of his qualifications, our board of directors has designated Mr. Brittain as the Committee s financial expert, as defined by the rules of the Securities and Exchange Commission.

The primary duties and responsibilities of the Audit Committee are to:

- 1. Oversee the integrity of our internal controls, financial systems and financial statements;
- 2. Review the quarterly unaudited and annual audited financial statements with management and the independent auditor;
- 3. Appoint and evaluate the independent auditor and monitor and evaluate the auditor s qualifications and independence;
- 4. Oversee compliance with legal and regulatory requirements;
- 5. Meet with the independent auditor in executive session at least annually;
- 6. Monitor the performance of both our internal and external auditors; and

The Audit Committee has the authority to retain special legal, accounting or other consultants to advise the Committee. The Audit Committee met ten times in 2007.

<sup>7.</sup> Annually review the Audit Committee s charter and performance.

## **Corporate Governance and Nominating Committee**

Our board of directors has a Corporate Governance and Nominating Committee, which consists of Michael R. Klein (Chair), Peter Arkley, and Raymond R. Oneglia. Each member of the Corporate Governance and Nominating Committee is an independent director, as defined by the NYSE and as affirmed by our board of directors. The duties of the Corporate Governance and Nominating Committee include:

1. Identifying individuals qualified to become directors and recommending to the full Board the persons to be nominated for election as directors;

- 2. Recommending director nominees for each committee of our board of directors and nominees for Chair of each committee;
- 3. Evaluating the independence of each director and so advising our board of directors;
- 4. Conducting a review and update as necessary of the Corporate Governance Guidelines and the Code of Business Conduct and Ethics;
- 5. Conducting evaluations of the performance of our board of directors and each committee, including a self-evaluation; and

6. Nominating a Lead Director whose duties shall include presiding at executive sessions of the non-management directors. The Corporate Governance and Nominating Committee met four times in 2007.

The Corporate Governance and Nominating Committee has the authority to retain consultants or other experts as it considers necessary to assist in the performance of its duties.

The independent directors have designated Michael Klein, Chair of the Corporate Governance and Nominating Committee, to act as the Lead Director. In his capacity as Lead Director, Mr. Klein has the following duties and authority:

chairing any meeting of the independent members of the Board in executive session;

meeting with any director who is not adequately performing his duties as a member of the board or any committee;

serving as a liaison between the chairman of the Board and the independent directors;

working with the chairman to prepare the agenda for Board meetings and determining the need for special meetings of the Board; and

consulting with the chairman on matters relating to corporate governance and Board performance. Compensation Committee

Our board of directors has a Compensation Committee, which consists of Raymond R. Oneglia (Chair), Michael R. Klein and Robert L. Miller. Each member of the Compensation Committee is an independent director, as defined by the NYSE and as affirmed by our board of directors.

The principal powers and duties of the Compensation Committee as established by our board of directors are to:

1. Review the executive compensation programs and policies and to employ outside expert assistance, if required, to analyze our compensation practices to assure that they are consistent with corporate goals and objectives, and competitive with those of comparable firms in the construction industry;

2.

Review and approve corporate goals and objectives relevant to the compensation of the chairman and chief executive officer, to evaluate his performance in light of those goals and objectives, and to determine and approve his compensation level based on this evaluation;

- 3. Make recommendations to our board of directors with respect to executive officer compensation;
- 4. Recommend to the board of directors annual profit and other targets for Perini for the purpose of determining incentive compensation awards under the provisions of the Amended and Restated General Incentive Compensation Plan and the Construction Business Unit Incentive Compensation Plan (the Incentive Compensation Plan );
- 5. Administer the Special Equity Incentive Plan and the 2004 Stock Option and Incentive Plan (together, the Stock Option Plans) and the Incentive Compensation Plan; such administration includes power to (i) approve participants participation in the Stock Option Plans, (ii) establish performance goals,

(iii) determine if and when any bonuses shall be paid, (iv) pay out any bonuses, in cash or stock or a combination thereof, as the Committee shall determine from year to year, (v) construe and interpret the Incentive Compensation Plan and the Stock Option Plans, and (vi) establish rules and regulations and perform all other acts it believes reasonable and proper; and

6. Review the investment performance of the Perini Corporation Pension Plan and make changes in investment managers and allocations, as the Compensation Committee deems necessary.

The Compensation Committee has the authority to retain special consultants to advise the committee as it considers necessary. The Compensation Committee met seven times in 2007.

Perini maintains on its website, *http://www.perini.com*, copies of the charters of each of the committees of our Board. We have also developed Corporate Governance Guidelines and a Code of Business Conduct and Ethics to outline our commitment to carefully govern the operation of our business and compliance with applicable laws and regulations, while maintaining the highest ethical standards. The Code applies to all of our officers, directors and employees, including our principal executive officer, principal financial officer, principal accounting officer, and persons performing similar functions. Perini s Corporate Governance Guidelines and Code of Business Conduct and Ethics are available on our website at *http://www.perini.com*. In order to access this portion of our website, click on the Corporate Governance tab. Interested parties may obtain printed copies of these documents by writing to the Investor Relations Department of the Company at 73 Mt. Wayte Avenue, Framingham, MA 01701. Any amendments to, or waivers of, the Code of Business Conduct and Ethics which apply to our directors, chief executive officer, chief operating officer, chief financial officer or any person performing similar functions will be disclosed on our website promptly following the date of such amendment or waiver.

#### **Nominations for Director**

The Board seeks candidates who are independent, possess relevant business, professional or board experience to make a significant contribution to the Board and have sufficient availability to attend to the business of our company. Annually, the Corporate Governance and Nominating Committee conducts an evaluation of the Board to determine whether it is functioning effectively, and recommends to the full Board the slate of director-nominees to be nominated for election at the next annual meeting of shareholders. Potential candidates for the Board may include candidates nominated by shareholders in accordance with our bylaws, those identified by a search firm retained for such purpose, or candidates recommended by other persons, including current directors or executive officers. Pursuant to the Corporate Governance and Nominating Committee charter, the process and criteria for considering the recommendations of shareholders. The minimum qualifications and specific qualities and skills required for directors are set forth in the Corporate Governance Guidelines, a copy of which is maintained on our website *http://www.perini.com*.

A shareholder who wishes to recommend a director-nominee to the Corporate Governance and Nominating Committee for the 2009 annual meeting of shareholders should submit the recommendation in writing to Perini Corporation, 73 Mt. Wayte Avenue, Framingham, Massachusetts 01701-9160, Attn: Corporate Secretary, so it is received not less than 75 days nor more than 180 days prior to the anniversary date of the 2008 Perini annual meeting of shareholders. However, if the 2009 annual meeting of shareholders is held more than seven (7) days earlier than the anniversary date of the 2008 annual meeting then notice must be delivered or received no later than 5 p.m. eastern time on (a) the 20th day following the earlier of (i) the day on which such notice of the date of the annual meeting is mailed or (ii) the day on which public disclosure of the dated the annual meeting is made, or (b) if such date of notice or public disclosure occurs more than 75 days prior to the scheduled date of such meeting, then the later of (i) the 20th day following the first to occur of such notice or such public disclosure or (ii) the 75th day prior to such scheduled date of such meeting.

## THE AUDIT COMMITTEE REPORT

Pursuant to rules adopted by the SEC designed to improve disclosures related to the functioning of corporate audit committees and to enhance the reliability and credibility of financial statements of public companies, the Audit Committee of the Board submits the following report.

The primary duties and responsibilities of the Audit Committee (the Committee ), which met ten times during the past fiscal year, are to oversee:

- 1. The integrity of Perini s internal controls, financial systems and financial statements;
- 2. Compliance by Perini with legal and regulatory requirements; and

3. The independence and performance of both Perini s internal and external auditors. We meet with management periodically to consider the adequacy of Perini s internal controls, as well as compliance with Sarbanes Oxley Section 404, and the objectivity of Perini s financial reporting. We discuss these matters with Perini s independent auditors and with appropriate Company financial personnel and internal auditors.

We meet privately with both the independent auditors and the internal auditors, as required, each of whom has unrestricted access to the Committee.

We also appoint the independent auditors and review periodically their performance and independence from management. As in prior years, the independent auditors are invited to be present at our annual meeting of shareholders.

The directors who currently serve on the Committee meet the independence and experience requirements of the NYSE, and have been so affirmed by the Board. In connection therewith, the Board has determined that none of us has a relationship with Perini Corporation that may interfere with our independence from Perini and its management. The Board has designated Willard W. Brittain, Jr. as the Audit Committee Financial Expert, based in part on review of his qualifications.

The Board has adopted a written charter setting forth the duties and responsibilities the Committee is to perform, which we review annually and revise as appropriate.

Management has primary responsibility for Perini s financial statements and the overall reporting process, including Perini s system of internal controls, and compliance with Sarbanes Oxley Section 404.

The independent auditors audit the effectiveness of the internal controls over financial reporting as well as annual financial statements prepared by management, express an opinion as to whether those financial statements fairly present the financial position, results of operations and cash flows of Perini in conformity with accounting principles generally accepted in the United States and discuss with us any issues they believe should be raised with us.

This year, we reviewed Perini s audited financial statements and met with both management and Deloitte & Touche LLP, Perini s independent auditors, to discuss those financial statements. Management has represented to us that the financial statements were prepared in accordance with accounting principles generally accepted in the United States.

We reviewed with the independent auditors, who are responsible for expressing an opinion on the conformity of the audited financial statements with generally accepted accounting principles, their judgments as to the quality, not just the acceptability, of our accounting principles and such other matters as are required to be

discussed with the Audit Committee under generally accepted accounting standards. In addition, we have received from and discussed with Deloitte & Touche LLP the written disclosure and the letter required by Independence Standards Board Standard No. 1, Independence Discussions with Audit Committees . These items relate to that firm s independence from Perini. We also discussed with Deloitte & Touche LLP any matters requiring discussion per the standards of the Public Company Accounting Oversight Board, including those required by Statement on Auditing Standards No. 61, Communication with Audit Committees, as amended.

We have considered and determined that the provision of the non-audit services included in Fees Paid to Audit Firm on page 145 is compatible with maintaining Deloitte & Touche LLP s independence.

Based on these reviews and discussions, we recommended to the Board that Perini s audited financial statements be included in Perini s Annual Report on Form 10-K for the fiscal year ended December 31, 2007.

AUDIT COMMITTEE Willard W. Brittain, Jr., Chair Robert A. Kennedy Michael R. Klein Raymond R. Oneglia

### COMPENSATION OF DIRECTORS AND EXECUTIVE OFFICERS

#### **Compensation Philosophy**

Our executive compensation program is intended to attract and retain talented executive officers and key employees who will continue to contribute to Perini s long-term success. We believe that it is important to tie compensation to our operating and financial goals, thereby aligning more closely the interest of management with that of our shareholders. We are mindful of the limited pool of available talent and consider competitive conditions when determining compensation.

In recognition of the variability of the building industry, we have historically believed that compensation focusing on shorter-term corporate goals is more appropriate for Perini and our shareholders and more effective in retaining and motivating our key executive talent. As a result, our compensation practices for our named executive officers have emphasized annual cash compensation (base salary and annual cash incentive awards). Long-term awards have not historically been granted on a regular basis, but instead have been granted when the Compensation Committee has determined an award to be appropriate based on the circumstances prevailing at the time. During 2008, however, we have commenced a project to review our executive compensation practices to determine whether our practices continue to best serve Perini and our shareholders. We expect this review to be more extensive, and our compensation policies and practices to be subject to more substantial revision, if we complete the announced merger with Tutor-Saliba, as discussed in this proxy statement.

Consistent with this approach, we have historically aimed to have the target annual cash compensation for our named executive officers fall within the 50th to 75th percentile for comparable executives of our relevant competitors. The Compensation Committee slots each of our named executive officers into this range based on the Compensation Committee s evaluation of the value of that officer to Perini and the amount of compensation necessary to encourage that officer to remain employed by Perini and to appropriately focus that officer on achieving Perini s corporate goals. To ensure that a significant amount of each named executive officer s total cash compensation is at risk and earned only if Perini s goals are achieved, approximately 40-60% of each such officer s target annual cash compensation is provided in the form of an annual incentive bonus opportunity, with the balance provided in the form of a fixed base salary.

To execute this strategy, the Compensation Committee has engaged Watson Wyatt to perform an executive compensation review with respect to Perini s named executive officers. The most recent review was completed in 2007. Reviewing the relevant data for each of Washington Group International, Shaw Group, Peter Kiewit Sons, Granite Construction, URS Corp., Aecom Technology Corp., Chicago Bridge & Iron Co., Emcor Group, Inc., Foster Wheeler, Ltd., Jacobs Engineering Group, Inc. and Tetra Tech, Inc., Watson Wyatt established a range of annual cash compensation for each of our named executive officers at the time. Using this data and consistent with the methodology described above, the Compensation Committee set the target annual cash compensation for our named executive officers to fall within the 50-75<sup>th</sup> percentile of annual cash compensation for comparable officers of these companies. As described in more detail below, this methodology was not applied to Mr. Burk, who was hired by Perini in 2007 (following completion of Watson Wyatt s study).

#### **Elements of Compensation**

As noted above, our executive compensation program primarily relies on annual cash compensation to retain and motivate our named executive officers. Accordingly, base salaries and annual incentive awards are the primary components of our program. In addition, the Compensation Committee has used ad hoc grants of long-term incentive awards when deemed appropriate by the Compensation Committee based on conditions prevailing at the time. Finally, Perini provides certain retirement benefits to our named executive officers to encourage long-term service to Perini.

#### **Base Salary**

The base salary of each of our named executive officers is fixed compensation that is not directly tied to the performance of Perini. We pay base salaries to fairly compensate our named executive officers for the services they provide during the year and because we believe our executive compensation program would not be competitive and effective in retaining our named executive officers if we did not pay base salaries. With the exception of Mr. Burk, who was hired in 2007, the base salaries of our named executive officers were initially established in 2004 at the time Watson Wyatt prepared an initial benchmarking study. At that time, the Compensation Committee determined the appropriate level of each named executive officer s target total annual cash compensation based on Watson Wyatt s study and the Compensation Committee s evaluation of the appropriate position for each such named executive officer within the desired range for our relevant competitors. The Compensation Committee then established the base salary for each named executive officer (other than Mr. Burk) was the appropriate amount for base salaries.

Since this initial establishment of the base salaries of our named executive officers in 2004, the Compensation Committee has annually reviewed them to determine whether any increases are appropriate. Base salaries are generally increased in March or April of each year, with any increase reflected only prospectively. (As a result, the amount of salary paid to an executive in a fiscal year is less than the increased annual salary amount set by the Compensation Committee during that year.) As part of the annual review process in 2007, the Compensation Committee also relied on the study prepared by Watson Wyatt in 2007 to confirm that base salaries of our named executive officers (other than Mr. Burk), along with their annual incentive award, would remain within the 50-75<sup>th</sup> percentile for total annual cash compensation for comparable officers of companies included in the study. The Compensation Committee also continued to include 40-60% of target annual cash compensation in the performance-based annual incentive opportunity. In 2007, the base salaries of three of our named executive officers were increased by the following amounts: Mr. Tutor \$100,000; Mr. Band \$50,000 and Mr. Caspers \$75,000. The base salary of our other named executive officers did not change in 2007. The Compensation Committee determined these increases to be appropriate because of its view that Mr. Tutor has unique talents that have been demonstrated throughout his tenure, Mr. Band has led the recent above average performance for both Perini as a whole and as President of Perini Management Services, Inc., and Mr. Caspers is critical to our successful gaming and hospitality business, and works in the highly competitive Las Vegas market. In addition, the Compensation Committee considered Mr. Casper s promotion to President of Perini Building Company when deciding on the increase in his salary.

Although the above discussion indicates that Mr. Tutor receives a base salary from Perini, Mr. Tutor s base salary was not, in fact, paid directly to him by Perini in 2007. Rather, he was compensated through a Management Agreement with Tutor-Saliba. However, the Compensation Committee applies similar methodology to determining payments under this agreement as it does to setting executive salary levels.

Mr. Burk s base salary was established when he was hired in 2007. At the time of his employment, the Compensation Committee believed that it was appropriate and important for Mr. Burk to receive a significant long-term equity award to more significantly tie his earnings to the performance of Perini. Taking this award into account, Perini established Mr. Burk s annual base salary at \$375,000, which is between the 25-50th percentiles for base salaries of comparable officers included in the last compensation study prepared by Watson Wyatt.

#### Incentive Compensation Plans Annual Awards

As described above, our historic executive compensation program has emphasized the importance of annual cash compensation in motivating and retaining our named executive officers. To provide appropriate incentives to our named executive officers, approximately 40-60% of their target annual cash compensation is comprised of an annual incentive bonus opportunity that is paid only if Perini achieves pre-established performance goals set by the Compensation Committee.

For 2007, the Compensation Committee established a target annual bonus for each named executive officer that was payable only if Perini achieved the applicable performance goal established by the committee. If Perini achieved 80% of this goal, each named executive officer would receive 80% of his target annual bonus amount. If Perini achieved between 80% and 100% of this goal, each named executive officer would receive between 80% and 100% of his target annual bonus amount. Each named executive officer s annual bonus was capped at 100% of his applicable target bonus, regardless of whether Perini exceeded the applicable performance goal.

The table below shows the threshold, target and maximum bonus opportunities as a percentage of actual base compensation paid in 2007 for our named executive officers:

	Threshold	Target	Maximum
R. Tutor	80%	100%	100%
K. Burk	60%	75%	75%
M. Ciskey	60%	75%	75%
C. Shaw	80%	100%	100%
R. Band	80%	100%	100%
M. Caspers	80%	100%	100%

The dollar amounts corresponding to these percentages are included in the table captioned Grants of Plan-Based Awards Table on page 137.

For 2007, the Compensation Committee selected Perini s pre-tax income as the applicable performance goal for the annual bonuses for our named executive officers because this goal encourages executives to both obtain new projects for Perini and to complete Perini s projects on a cost efficient basis. The applicable target goal set by the Compensation Committee for 2007 was \$116,500,000 of pre-tax income. Because Perini exceeded this goal and earned \$154,395,000 of pre-tax income in 2007, each named executive officer was paid his maximum target bonus. Mr. Burk received a target annual bonus amount equal to 75% of his prorated base compensation in 2007, which is the maximum bonus for which he was eligible.

#### **Long-Term Incentives**

As noted earlier, regular grants of long-term incentives have not played a significant role in our executive compensation because of our historic belief that year-to-year incentives better focus our executives on achieving Perini s performance objectives. Accordingly, with the exception of Mr. Burk, we did not grant long-term incentive awards to our named executive officers in 2007. This decision was due in part to the fact that our named executive officers had received a significant grant of long-term equity awards in 2006 and in part to the Compensation Committee s determination that another grant of long-term incentives in 2007 was not necessary to appropriately incent our named executive officers.

Upon his employment, Mr. Burk received a grant of 50,000 restricted stock units, each of which represents the right to receive one share of our common stock upon vesting. Fifty percent of these awards vested on January 5, 2008 with the balance vesting on January 5, 2009 if Mr. Burk is then employed by Perini. The Compensation Committee granted this award to Mr. Burk to immediately provide him with a significant equity stake in Perini so that his earnings are directly and materially tied to the performance of Perini. As noted earlier, this award was taken into account in determining Mr. Burk s base salary and 2007 annual bonus opportunity.

#### **Retirement Benefits**

In addition to providing annual compensation to our named executive officers, we believe it is imperative that employees provide for their retirement years and believe that it is our obligation to contribute as well. Attractive retirement benefits are essential when competing in the marketplace for talented and qualified employees.

We have a tax-qualified Section 401(k) Retirement Plan covering all of our executive, professional, administrative and clerical employees who are over 21 years of age and who have completed three months of service with us. Employer contributions into the 401(k) plan are based on a non-discretionary match of employees contributions, as defined. The Compensation Committee reviews the employer match formula periodically, comparing the formula to that of similar companies, and improves the formula when it deems appropriate; the most recent change became effective in January 2007.

We also have a defined benefit pension plan for all of our full-time employees who were employed prior to June 1, 2004 when the plan was frozen, meaning that final average earnings and years of service will remain at the June 1, 2004 level for purposes of calculating future benefits. To the extent covered remuneration is limited by the Internal Revenue Code of 1986, as amended, certain pension benefits payable have been augmented through our Benefit Equalization Plan, which was also frozen at June 1, 2004. Following the freezing of the pension plan in 2004, the Compensation Committee voted to improve Perini s contribution toward employees retirement by enhancing the 401(k) match formula.

#### **Role of Executive Officers in Setting Compensation**

We believe that our chief executive officer is best positioned to evaluate both the performance of our other named executive officers and the competitive market for senior executives in our industry. Accordingly, the Compensation Committee works closely with Mr. Tutor in establishing the compensation of our other named executive officers. For 2007, Mr. Tutor recommended to the Compensation Committee the increases in base salaries and the target annual bonus amounts that he believed appropriate for our named executive officers (other than himself), all within the general framework of our compensation philosophy discussed above. After discussion with Mr. Tutor and careful evaluation of his recommendations, the Committee approved these recommendations.

#### Impact of Accounting and Tax Treatment

We believe that the primary goals of our executive compensation program are to attract and retain valued and important named executive officers, to clearly identify for our named executive officers the corporate goals and objectives important to Perini, to motivate our named executive officers to achieve these goals and to fairly reward our named executive officers for achieving these goals. Accordingly, the accounting and tax treatment of our executive compensation program, while important, is not a determining factor in structuring our program. We appropriately account for our executive compensation and, to the extent consonant with the goals of our executive compensation program, we attempt to structure our executive compensation program to preserve the deductibility of amounts paid to our named executive officers. With respect to 2007, we believe that the cash compensation paid to our named executive officers is fully deductible.

### **Recent Events**

As described in this proxy statement, in April 2008, we entered into a merger agreement with Tutor-Saliba. If the merger is completed, we will be a significantly larger and more diverse company. In connection with the signing of the merger agreement, we negotiated and signed a five-year employment agreement with Mr. Tutor that will become effective only if we consummate that transaction. In addition, if we consummate the transaction, we may implement other changes to our executive compensation philosophy and approach. We intend to work with Watson Wyatt in evaluating potential changes to levels and elements of compensation, as well as to our compensation programs and policies in light of the impact of the merger on our business. We also are considering the possibility of entering into employment agreements with some of our executives in connection with the merger.

#### **Summary Compensation Table**

The table below summarizes the total compensation paid to or earned by each of our named executive officers for the fiscal years ended December 31, 2007 and December 31, 2006.

				Stock		Iı	n-Equity ncentive Plan	Change in Pension Value and Nonqualified Deferred Compensatio	n Al	l Other		
Name and Principal Position	Year	Salary (6)	Bonus	Awards (1)	Option Awards	Con	pensation (2)	Earnings (3)	Com	pensation (4)		Total
Ronald N. Tutor Chairman and Chief Executive Officer (5)	2007 2006	(0)	\$ 800,000	\$ 3,996,300 \$ 9,149,500	11 wur us	\$ \$	976,900 879,000	(0)	\$ \$	976,900	\$ :	5,950,100 1,707,700
Kenneth R. Burk	2007	\$ 122,600		\$ 1,596,600		\$	91,950		\$	26,300	\$	1,837,450
Sr. Vice President and Chief Financial Officer (beginning Sept. 2007)												
Michael E. Ciskey Vice President, Chief Financial Officer (until Sept. 2007-currently SVP-Civil)	2007 2006	\$ 325,000 \$ 314,000		\$ 379,100 \$ 284,400		\$ \$	243,750 235,800			62,000 50,900	\$ \$	1,096,850 890,800
Craig W. Shaw Chairman and CEO Perini Building Company	2007 2006	\$ 500,000 \$ 484,100	\$ 150,000	\$ 1,895,700 \$ 1,421,800		\$ \$	478,750 463,600			92,100 84,000		3,315,350 2,459,600
Robert Band President and Chief Operating Officer	2007 2006	\$ 538,500 \$ 484,100	\$ 250,000	\$ 1,263,800 \$ 947,900		\$ \$	538,500 484,100			· · · ·		2,721,600 2,221,200
Mark A. Caspers President, Perini Building Company	2007 2006	\$ 482,700 \$ 401,000	\$ 150,000	\$ 947,900 \$ 710,900		\$ \$	462,200 307,100	\$ 26,150	\$ \$	56,700 79,400		2,125,650 1,498,400

(1) With the exception of Mr. Burk s awards, amounts are based on the value of restricted stock units granted on April 5, 2006 at \$31.55, which was the closing market price of our common stock on that date, amortized during the indicated fiscal year in accordance with FAS 123(R). Mr. Burk s units were based upon the closing market price of \$53.22 on September 26, 2007, which was the grant date of his award. The awards were granted under the 2004 Stock Option and Incentive Plan.

(2) These amounts represent payments made in 2008 and 2007, based on attainment of pretax income goals for 2007 and 2006 under our incentive compensation plans discussed above under the heading Incentive Compensation Plans . For 2007, Mr. Shaw and Mr. Caspers each received a discretionary payment of \$150,000 in addition to the amounts payable under the plan formula, as reflected in the Bonus column. For 2006, Mr. Tutor and Mr. Band also received discretionary payments of \$800,000 and \$250,000, respectively, which has been similarly reflected.

Messrs. Shaw and Caspers, with the concurrence of the Compensation Committee, had a portion of their annual incentive payments allocated to a bonus pool for the benefit of employees who work for Perini Building Company and who are otherwise not eligible to participate in the incentive compensation plan. As a result, the incentive payments made to Messrs. Shaw and Caspers for 2007 were reduced by \$21,250 and \$20,500 (\$20,600 and \$13,600 for 2006), respectively, to fund these allocations.

(3) Perini has a non-contributory defined benefit pension plan for all of our full-time employees prior to June 1, 2004 when the plan was frozen, meaning that final average earnings and years of service will remain at the June 1, 2004 level for purposes of calculating future benefits. Certain pension benefits payable have been augmented by a benefits equalization plan, or BEP, which was also frozen on June 1, 2004. The amounts presented here represent the difference between the present value of the benefits payable from the pension plan and the BEP as of December 31, 2007 and 2006, as compared to December 31, 2006 and 2005. The present values were calculated using the discount rates used to compute our pension benefit obligations at year end, which were 6.41%, 5.86% and 5.62% for December 31, 2007, 2006 and 2005, respectively. As the plans are frozen, the change in pension value above is primarily caused by the change in the discount rate and the present value effect of the individual being one year closer to normal retirement age. The change in 2007 is additionally impacted to reflect the earliest retirement age for which benefits are unreduced in conformance with recent SEC guidance. The change in pension value attributable to the change in the retirement age assumption is \$369,200 for Mr. Band, \$50,600 for Mr. Caspers, \$101,300 for Mr. Ciskey and \$273,300 for Mr. Shaw. Mr. Tutor and Mr. Burk do not participate in these plans.

(4) The \$976,900 listed for Mr. Tutor represents management service fees paid to Tutor-Saliba Corporation of which Mr. Tutor is the chairman, president, chief executive officer and primary beneficial owner. The management agreement between Perini and Tutor-Saliba is discussed below beginning on page 141 under Tutor-Saliba Management Agreement. If we complete the recently announced merger with Tutor-Saliba, Mr. Tutor s ongoing employment and compensation will be governed by the terms of a new five-year employment agreement. This agreement is discussed under The Employment Agreement beginning on page 89.

The \$26,300 listed for Mr. Burk includes temporary housing expenses (\$15,800) and vehicle allowance (\$3,600). It also includes life insurance premiums.

The \$62,000 listed for Mr. Ciskey includes temporary housing expenses (\$27,300) and company vehicle use (\$13,800). It also includes his 401(k) match (\$9,000), tax gross-up (\$9,700) and life insurance premiums.

The 92,100 listed for Mr. Shaw includes temporary housing expenses (45,300), company vehicle use (23,800), and club membership dues (10,500). It also includes his 401(k) match (9,000) and life insurance premiums.

The \$35,800 listed for Mr. Band includes company vehicle use (\$12,400), his 401(k) match (\$9,000), tax gross-up (\$8,900) and life insurance premiums.

The 56,700 listed for Mr. Caspers includes company vehicle use (\$23,250), temporary housing expenses (\$12,100), club membership dues (\$7,300), and travel allowance. It also includes his 401(k) match (\$9,000) and life insurance premiums.

- (5) We negotiated and signed a five-year employment agreement with Mr. Tutor that will become effective only if we consummate the pending merger between Perini and Tutor-Saliba.
- (6) The salary amounts reflect actual amounts received, which for Messrs. Tutor, Band and Caspers reflect a prorated amount of their respective annual current base salaries because salary increases were not determined or made effective until March 2007. The base salary for Mr. Burk was prorated from his start date of September 1, 2007. The current annual base salaries for our named executive officers (which have not been increased since March 2007) are: Mr. Tutor, \$1,000,000; Mr. Burk, \$375,000; Mr. Ciskey, \$325,000; Mr. Shaw, \$500,000, Mr. Band, \$550,000; and Mr. Caspers, \$500,000.

# Grants of Plan-Based Awards Table

		Under N	ted Future Ion-Equity Plan Awar	Incentive ds		nated Future ler Equity Ir Plan Awar	icentive	All Other Stock Awards: # of Shares	All Other Option Awards: Underlying # of Securities	Exercise or Base Price of Option	Grant Date Fair Value of Stock and Option
Name	Grant Date	hold (\$)	Target (\$)	Maximum (\$)	hold (\$)	Target (\$)	Maximum (\$)	or units (#)	Options (#)	Awards (\$/Sh)	Awards (\$)
R. Tutor	3/14/2007	781,520	976,900	976,900							
K. Burk	9/1/2007 9/26/2007	73,560	91,950	91,950		2,661,000					2,661,000
M. Ciskey	3/14/2007	195,000	243,750	243,750							
C. Shaw	3/14/2007	400,000	500,000	500,000							
R. Band	3/14/2007	430,800	538,500	538,500							
M. Caspers	3/14/2007	386,160	482,700	482,700				1			

The Non-Equity Incentive Plan is discussed beginning on page 133. These awards were granted in March 2007 (awards to Mr. Burk were granted upon commencement of his employment in September 2007) contingent upon the attainment of 2007 pretax income goals. The related goals were established by the Compensation Committee following consultation with management, and were set at a level that the Compensation Committee believed was achievable with a high level of effort. The goals were met, and the Compensation Committee voted to make the maximum payout according to the plan formula to the above individuals in April 2008.

The Equity Incentive Plan, the 2004 Stock Option and Incentive Plan, is discussed under Long-Term Incentives beginning on page 134. The restricted stock units awarded above are valued at \$53.22, the closing price of our common stock on the measurement date, in accordance with

FAS123(R). These awards are subject to time-based vesting: 50% vested in January 2008 and 50% is scheduled to vest in January 2009.

# Outstanding Equity Awards at Fiscal Year-End Table

		Optio	ons Awards				Stoc	x Awards (1)	
							Market Value	Equity Incentive Plan Awards:	Equity Incentive Plan Awards:
			Equity				of	Number	Market or
			Incentive			NT I	Shares	of	Payout
			Plan Awards			Number of Shares	or Units	Unearned Shares,	Value of Unearned
	Number of	Number of	Number of			or Units	of of	Units or	Shares,
	Securities	Securities	Securities			of Stock	Stock	Rights	Units or
	Underlying	Underlying	Underlying			That	That	That	Other
	Unexercised	Unexercised	Unexercised	Option	Option	Have	Have	Have Not	<b>Rights That</b>
	Options	Options	Unearned	Exercise	Expiration	Not	Not	Vested	Have Not
Name	Exercisable	Unexercisable	Options	Price	Date	Vested	Vested	(2)	Vested
Ronald N. Tutor				\$			\$	150,000	\$ 6,213,000
Kenneth R. Burk				\$			\$	50,000	\$ 2,071,000
Michael E. Ciskey				\$			\$	30,000	\$ 1,242,600
Craig W. Shaw				\$			\$	150,000	\$ 6,213,000
Robert Band				\$			\$	100,000	\$4,142,000
Mark A. Caspers				\$			\$	75,000	\$ 3,106,500

(1) Value is based on the closing market price of \$41.42 on December 31, 2007.

(2) Vesting is scheduled according to the following table:

	Jan. 2008	June 2008	Jan. 2009	Jan. 2010	Total
Ronald N. Tutor	2000	150,000(P)	2007	2010	150,000
Kenneth R. Burk	25,000(T)		25,000(T)		50,000
Michael E. Ciskey	10,000(T)		10,000(T)	10,000(T)	30,000
Craig W. Shaw	50,000(P)		50,000(P)	50,000(P)	150,000
Robert Band	33,333(P)		33,333(P)	33,334(P)	100,000
Mark A. Caspers	25,000(T)		25,000(T)	25,000(T)	75,000

(T) Unitare time-vested

(P) Unitare performance-vested

#### **Option Exercises and Stock Vested Table**

Options	Awards	Stock Awards			
Number of Shares Acquired on Exercise (1)	Value Realized on Exercise	Number of Shares Acquired on Vesting (2)	Value Realized on Vesting		

Name

Ronald N. Tutor			150,000	\$ 9,229,500
Kenneth R. Burk				
Michael E. Ciskey				
Craig W. Shaw	50,000	\$ 2,054,200		
Robert Band	100,000	\$4,700,250		
Mark A. Caspers				

- Mr. Shaw exercised options priced at \$4.50/share, and realized market value on our common stock ranging from \$29.42 \$63.89 per share. Mr. Band exercised options priced at \$4.50/share, and realized market value on our common stock ranging from \$39.31 \$73.49 per share. These options exercises and sales were made pursuant to the terms of Rule 10b5-1 trading plans.
- (2) Represents restricted stock units awarded to Mr. Tutor under the 2004 Stock Option and Incentive Plan. Vesting was based 50% on Mr. Tutor s continuing employment as our CEO, and 50% based on the attainment, in the judgment of our Compensation Committee, of certain management succession initiatives. These awards vested June 29, 2007.

## Pension Benefits for 2007 Fiscal Year

Name	Plan Name	Number of Years of Credited Service	Present Value of Accumulated Benefit (1)	Payments During Last Fiscal Year
Ronald N. Tutor			\$	\$
Kenneth R. Burk			\$	\$
Michael E. Ciskey	Pension Plan	26	\$ 414,918	\$
Craig W. Shaw	Pension Plan	27	\$ 340,218	\$
	BEP	27	\$ 700,007	
Robert Band	Pension Plan	31	\$ 540,195	\$
	BEP	31	\$ 1,033,599	
Mark A. Caspers	Pension Plan	23	\$ 169,190	\$

(1) Assumes retirement occurs at the later of age 62 or current age, and a discount rate of 6.41%. Based on RP2000 mortality tables, projected to 2006 by Scale AA.

Perini has a defined benefit pension plan that covers its executive, professional, administrative and clerical employees, subject to certain specified service requirements. The plan is noncontributory and benefits are based on an employee s years of service and final average earnings, as defined. The plan provides reduced benefits for early retirement and takes into account offsets for social security benefits. All employees are vested after five years of service. Perini also has an unfunded supplemental retirement plan (the Benefits Equalization Plan, or BEP) for certain employees whose benefits under the defined benefit pension plan were reduced because of compensation limitations under federal tax laws.

The normal retirement benefit under these plans is equal to:

.75% of final average earnings, as defined, not in excess of covered compensation (as defined), multiplied by years of service, up to 25; plus

1.5% of final average earnings, as defined, in excess of covered compensation (as defined), multiplied by years of service, up to 25.

Our plans provide for early retirement upon either the attainment of age 55 and 10 years of service, or the completion of 25 years of service. Accordingly, Mr. Ciskey, Mr. Shaw and Mr. Band are all currently eligible for early retirement benefits. Under our plans, a participant who elects early retirement may elect to receive either an immediate early retirement income equal to 91% of his or her normal retirement benefit or a deferred benefit. Upon the attainment of age 62 and completion of 25 years of service, the participant may receive an unreduced pension equal to his or her normal retirement benefit. A reduced benefit is available for a participant who elects early retirement and wishes to receive benefits prior to age 62.

Effective June 1, 2004, all benefit accruals under Perini s pension plans were frozen; however, the current vested benefits will be preserved. Accordingly, our named executive officers will not earn additional pension benefits, but they may become eligible for an early retirement benefit (which will be based on their frozen normal retirement benefit) based on service after June 1, 2004.

None of our named executive officers have written employment agreements with Perini or any arrangement that provides termination benefits.

#### **Director Compensation**

The following table sets forth compensation information for 2007 for each member of our Board.

	Fees Earned or Paid in Cash	Stock Awards	Option Awards	Non-Stock Incentive Plan Compensation	Change in Pension Value and Nonqualified Deferred Compensation Earnings	All Other Compensation	Total
Name	(\$)	(\$) (c)	(\$)	(\$)	(\$) (d)	(\$)	(\$)
Peter Arkley	49,900	49,820					99,720
Robert Band	(a)	(a)	(a)	(a)	(a)	(a)	(a)
Willard W. Brittain, Jr.	63,100	49,820					112,920
James A. Cummings	(b)	(b)		312,500	9,300	355,800	677,600
Robert A. Kennedy	54,100	49,820					103,920
Michael R. Klein	62,400	49,820					112,220
Robert L. Miller	44,500	49,820					94,320
Raymond R. Oneglia	62,400	49,820					112,220
Ronald N. Tutor	(a)	(a)	(a)	(a)	(a)	(a)	(a)

- (a) Mr. Band and Mr. Tutor are named executive officers, whose compensation appears on the Summary Compensation Table. They do not receive director s fees.
- (b) Mr. Cummings was the CEO of James A. Cummings Inc., a wholly owned subsidiary of Perini until his retirement in January 2008. He is not a named executive officer, and so is not included in the Summary Compensation Table. His remuneration for his role as CEO of James A. Cummings Inc. is reflected in the table above. All Other Compensation includes salary (\$312,500), use of company vehicle (\$14,700), club memberships (\$15,700) and life insurance premiums (\$3,900), along with his 401(k) match (\$9,000). Mr. Cummings did not receive director s fees. He resigned from the Board in December 2007 in anticipation of his retirement from Perini in January 2008.
- (c) Based on closing price of \$49.82 on the grant date, May 17, 2007.
- (d) See note (3) to Summary Compensation Table. The change in pension value attributable to the change in retirement age assumption is \$9,100 for Mr. Cummings.

Our Compensation Committee determines the level of compensation to be paid to our Board. Periodically, this committee reviews the functions being performed by the Board and its committees, as well as the board compensation paid by similar companies, in order to determine whether an adjustment should be made.

Fees for our outside directors consist of an annual retainer fee of \$40,000, payable in cash or common stock at each director s option, plus 1,000 shares of common stock. Fees paid in company stock are included in the Stock Awards column above. Directors also receive \$900 per Board meeting attended in person and \$300 per meeting attended telephonically. Members of the Audit Committee receive \$2,000 per meeting attended telephonically. The Audit Committee Chair receives an additional annual retainer of \$10,000. In May 2008, the Compensation Committee voted to increase the annual retainer fee from \$40,000 to \$80,000 based on a review of other public companies board compensation as well as a recognition of recent and continuing growth of Perini. All other components of Board compensation remained unchanged.

Ronald N. Tutor, our chairman and chief executive officer, does not receive director fees because he is party to a management agreement described in Certain Relationships and Related Party Transactions and Director Independence.

No directors had outstanding stock options at December 31, 2007.

#### **Director and Officer Indemnification**

Our amended and restated articles of organization provide that no director shall be personally liable to us or to our shareholders for monetary damages for breach of fiduciary duty as a director, except for liability for any breach of the director s duty of loyalty to us or our shareholders, for acts or omissions not in good faith, for acts or omissions involving intentional misconduct or a knowing violation of law or for any transaction from which the director derived an improper personal benefit. Our bylaws provide that our directors and officers will be indemnified against liabilities that arise from their service as directors and officers, subject to certain exceptions. We have obtained insurance which insures our directors and officers against certain losses and which insures us against our obligations to indemnify our directors and officers.

#### CERTAIN RELATIONSHIPS AND RELATED PARTY TRANSACTIONS

We have adopted the Code of Business Conduct and Ethics for all executive officers, directors and employees that addresses potential conflict of interest situations, including related party transactions. Under this policy, any questions are required to be directed to our chief compliance officer, and suspected violations are required to be reported to the chief compliance officer or the Chair of the Audit Committee. In addition, our Audit Committee is responsible for reviewing and evaluating potential transactions with related parties, and then advising the Board whether such transactions are appropriate.

The transactions described below were reviewed and approved by the Audit Committee in accordance with our policies. In addition, we believe that the transactions described below were on terms that were at least as favorable to us as we would have expected to negotiate with other unaffiliated third parties at the point in time these transactions were consummated.

#### **Tutor-Saliba Management Agreement**

In January 1997, we entered into a management agreement with Tutor-Saliba and Ronald N. Tutor, chief executive officer and primary beneficial owner of Tutor-Saliba, to provide certain management services. The management agreement has been renewed annually by our Compensation Committee, which consists entirely of independent directors, under the same basic terms and conditions as the initial agreement except that the amount of the fee payable thereunder by us to Tutor-Saliba has been increased. In March 2007, the Compensation Committee voted to increase the annual payment to \$1,000,000.

#### Joint Ventures

*Tutor-Saliba Joint Ventures*. Historically, we have participated in joint ventures with Tutor-Saliba and currently participate in certain joint ventures with them. Some of these joint venture projects have been managed by Tutor-Saliba and others have been managed by us. These joint ventures generated total revenues of \$96.0 million in 2007 of which our share contributed \$70.6 million to our consolidated revenues for the year ended December 31, 2007.

*O&G Joint Ventures.* We also have participated in certain joint ventures with O&G Industries, Inc., of which Raymond R. Oneglia, one of our directors, is vice chairman of the board of directors. These joint ventures generated total revenues of \$4.4 million in 2007 of which our share contributed \$3.1 million to our consolidated revenues for the year ended December 31, 2007.

#### Subcontractor

In January 2008 Tutor-Saliba acquired a plumbing contractor which at the time of such acquisition had subcontracts with Perini totaling approximately \$63.7 million. This acquisition was not reviewed by our Audit Committee because it occurred prior to the merger.

#### **Merger Agreement and Related Transactions**

The merger and the related transactions and agreements were recommended to the Board by the Special Committee, which included only independent and disinterested directors. They were then approved by the full Board (with Mr. Tutor not participating or voting).

As described in this proxy statement, on April 2, 2008, we entered into the Merger Agreement with Tutor-Saliba, pursuant to which Tutor-Saliba will merge with and into one of our wholly owned subsidiaries. Mr. Tutor, our chairman and chief executive officer, is the chairman and chief executive officer of Tutor-Saliba. Two trusts that he controls own 96% of the outstanding capital stock of Tutor-Saliba. As described above, Mr. Tutor has various interests in this transaction that are different from other Perini shareholders. For a detailed discussion of the terms of the Merger Agreement, the Shareholders Agreement and the Employment Agreement and Mr. Tutor s interests in this transaction please see Additional Interests of Directors, Executive Officers and Certain Beneficial Owners beginning on page 70. The Merger Agreement beginning on

Additional Interests of Directors, Executive Officers and Certain Beneficial Owners' beginning on page 70, The Merger Agreement beginning of page 73, The Shareholders Agreement beginning on page 86 and The Employment Agreement beginning on page 89.

#### COMPENSATION COMMITTEE INTERLOCKS AND INSIDER PARTICIPATION

No member of the Compensation Committee has served as one of our officers or employees at any time. None of our executive officers currently serves, or in the past fiscal year has served, as a member of the board or compensation committee of any entity that has one or more executive officers serving on our board of directors or Compensation Committee.

#### SECTION 16(a) BENEFICIAL OWNERSHIP REPORTING COMPLIANCE

Section 16(a) of the Securities Exchange Act of 1934, as amended, requires our executive officers (as defined in regulations issued by the SEC) and directors, and persons who own more than ten percent of a registered class of Perini s equity securities (collectively, Insiders), to file initial reports of ownership and reports of changes in ownership of our common stock (including options and warrants to acquire common stock) with the SEC. Officers, directors and greater than ten percent shareholders are required by SEC regulation to furnish us with copies of all Section 16(a) forms they file.

Based solely on a review of the copies of such reports of ownership received by us and certifications from our Executive Officers and Directors, we believe that during fiscal year 2007 all filing requirements applicable to our Insiders were met.

#### OWNERSHIP OF COMMON STOCK BY DIRECTORS, EXECUTIVE OFFICERS AND

#### PRINCIPAL SHAREHOLDERS

The following table sets forth certain information concerning beneficial ownership as of June 30, 2008 of our common stock by each Director; each Executive Officer named in the summary compensation table; all Directors and Executive Officers as a Group; and all persons we know to hold in excess of 5% of our common stock.

In preparing the following table, we relied upon statements filed with the SEC by beneficial owners of more than 5% of the outstanding shares of our common stock pursuant to Section 13(d) or 13(g) of the Exchange Act, unless we knew or had reason to believe that the information contained in such statements was not complete or accurate, in which case we relied upon information which we considered to be accurate and complete. Unless otherwise indicated, the address of each of the individuals and entities named below is: c/o Perini Corporation, 73 Mt. Wayte Avenue, Framingham, Massachusetts 01701.

	2	Shares of Common Stock Beneficially	
	Owned on June	Owned on June 30, 2008	
	(1) (2)		
Name and Address	Number	Percent	
Directors and executive officers:			
Ronald N. Tutor	150,000	**	
Michael R. Klein	106,031	**	
Robert L. Miller	78,736	**	
Craig W. Shaw	32,120	**	
Mark A. Caspers	30,887	**	
Robert Band	24,212	**	
Kenneth R. Burk	16,887	**	
Raymond R. Oneglia	14,736	**	
Willard W. Brittain, Jr.	11,036(3)	**	
Peter Arkley	8,408	**	
Robert A. Kennedy	3,500	**	
Michael E. Ciskey		**	
Marilyn A. Alexander			
Donald D. Snyder			
All Directors and Executive Officers as a Group (14 persons)	476,553	1.74%	
Beneficial Ownership of 5% or More			
Jeffrey L. Gendell	2,030,860(4)	7.43%	
State Street Bank and Trust Company	1,692,264(5)	6.19%	
Mutuelle AXA	1,460,341(6)	5.34%	
Barclays Global Investors N.A.	1,383,729(7)	5.06%	
Total beneficial owners of more than 5% of Perini Common Stock	6,567,194	24.03%	

\*\* Less than 1%

(1) Beneficial ownership is determined in accordance with the rules of the Securities and Exchange Commission and generally includes voting or investment power with respect to securities. Shares of common stock and options or warrants that are currently exercisable or exercisable within 60 days of May 1, 2008 are deemed to be outstanding and to be beneficially owned by the person holding such options for the purpose of computing the percentage ownership of such person, but are not treated as outstanding for the purpose of computing the percentage ownership of any other person.

(2) Based on 27,328,927 shares of common stock outstanding as of June 30, 2008.

- (3) Includes 1,600 shares held by a partnership in which Mr. Brittain is a 57.3% owner.
- (4) Based on information contained in a Schedule 13G filed on January 22, 2008 by Tontine Capital Partners, L.P., Tontine Capital Management, L.L.C., Tontine Overseas Associates, L.L.C., and Jeffrey L. Gendell, the address of each of which is 55 Railroad Avenue, Greenwich, Connecticut 06830. Includes 1,697,961 shares owned by Tontine Capital Partners, L.P. (TCP), and 332,899 shares owned by Tontine Capital Overseas Master Fund, L.P. (TCO). Tontine Overseas Associates, L.L.C., a limited liability company organized under the laws of the State of Delaware (TOA), serves as investment manager to TCO with respect to the shares of Common Stock directly owned by TCO. Tontine Capital Management, L.L.C., the general partner of TCP, has the power to direct the affairs of TCP, including decisions respecting the disposition of the proceeds from the sale of the shares of the Company. Mr. Gendell is the managing member of TCM and TOA, and in that capacity directs their operations. Each of the clients of TOA has the power to direct the receipt of dividends from or the proceeds of sale of such shares.
- (5) Based on information contained in a Schedule 13G filed on May 5, 2008 (for the period ending March 31, 2008) by State Street Bank and Trust Company, the address for which is State Street Financial Center, One Lincoln Street, Boston, MA 02111.
- (6) Based on information contained in a Schedule 13G filed on February 14, 2008 by AXA Financial, Inc., AXA Assurances I.A.R.D. Mutuelle, AXA Assurances Vie Mutuelle, AXA Courtage Assurances Mutuelle and AXA, the address for all of which is 1290 Avenue of the Americas, New York, NY 10104. Includes 567,569 shares owned by entities controlled by the Mutuelles AXA, including 544,669 shares owned by AXA Rosenberg Investment Management LLC and 21,400 shares owned by Alliance Bernstein, and 1,460,341 shares held by unaffiliated third-party client accounts managed by Alliance Capital Management L.P., as investment adviser. Alliance Capital Management L.P. is a majority-owned subsidiary of AXA Financial, Inc., which is controlled by Mutuelles AXA.
- (7) Based on the most recent publicly available data according to Thomson Financial. The address for Barclays Global Investors N.A. is 45 Fremont St., San Francisco, CA 94105.

#### **PROPOSAL 4: RATIFICATION OF APPOINTMENT OF AUDITORS**

Our Audit Committee has selected the firm of Deloitte & Touche LLP, independent registered public accounting firm, as our auditors for the fiscal year ending December 31, 2008. Although shareholder approval of the selection of Deloitte & Touche LLP is not required by law, our board of directors believes that it is advisable to give shareholders an opportunity to ratify this selection. If this proposal is not approved by our shareholders at the 2008 annual meeting, our Audit Committee will reconsider their selection of Deloitte & Touche LLP.

Representatives of Deloitte & Touche LLP will be present at the annual meeting, will have the opportunity to make a statement if they so desire and will be available to answer appropriate questions.

#### FEES PAID TO AUDIT FIRM

During the years ended December 31, 2007 and 2006, we retained Deloitte & Touche LLP to provide services in the following categories and amounts:

	2007	2006
Audit Fees	\$ 1,858,443	\$ 1,478,550
Audit Related Fees	\$ 108,600	\$ 108,700
Tax Fees	\$	\$ 7,808
Total Fees	\$ 1,967,043	\$ 1,595,058

The Audit Committee has considered the nature of the tax services provided by Deloitte & Touche LLP in 2006 and concluded they are compatible with maintaining the auditors independence.

# THE PERINI BOARD OF DIRECTORS RECOMMENDS A VOTE FOR RATIFICATION OF THE SELECTION OF DELOITTE & TOUCHE LLP AS INDEPENDENT AUDITORS FOR PERINI FOR THE FISCAL YEAR ENDING DECEMBER 31, 2008.

#### **PROPOSAL 5: APPROVAL OF THE PLAN AMENDMENT**

#### Discussion

The Perini board of directors has approved an increase in the number of shares authorized for issuance under our 2004 Stock Option and Incentive Plan (the 2004 Plan ) from 3 million to 5.5 million.

The Perini board of directors determined that it was in the best interests of Perini and our shareholders to increase the number of shares available for issuance of awards under the 2004 Plan to permit the grant of long-term incentive awards under the 2004 Plan to a larger group of executives and key employees, who will be responsible for a substantially larger business, if the merger is completed. The board of directors also took into account that the total number of shares of Perini common stock that will be outstanding upon completion of the merger will be approximately 50.3 million (as compared to approximately 27.3 million as of July 30, 2008). Currently, the 2004 Plan as approved by the shareholders provides for the issuance of approximately 3 million shares of Perini common stock. Of this amount, awards covering a total of 1.3 million shares have been granted. A total of 1.7 million shares remain available for future grants.

Based on the increase in the number of employees, the increase in the size of the business and the number of shares of Perini common stock that will be issued and outstanding immediately following the completion of the merger, the Perini board of directors has recommended that the 2004 Plan be amended to provide for the grant of awards covering a total of up to 5.5 million shares of Perini common stock, representing an increase of 2.5 million shares. If this amendment is approved, there would be a total 5.5 million shares of Perini common stock available for grant pursuant to awards under the 2004 Plan.

The Perini board of directors has determined that this amendment, even if approved by the Perini shareholders, will not be implemented unless the merger is completed.

#### **Summary Description of 2004 Plan**

The following summary of the principal terms of the 2004 Plan is qualified in its entirety by reference to the full text of the plan attached as *Annex G* to this proxy statement.

2004 Plan Administration. The 2004 Plan provides for administration by the Compensation Committee of the Board of Directors (the Administrator ). The Administrator has full power to select, from among the individuals eligible for awards, the individuals to whom awards will be granted, to make any combination of awards to participants, and to determine the specific terms and conditions of each award, subject to the provisions of the 2004 Plan. The Administrator may permit common stock, and other amounts payable pursuant to an award, to be deferred. In such instances, the Administrator may permit interest, dividends or deemed dividends to be credited to the amount of deferrals. In addition, the Administrator may not reprice outstanding options, other than to appropriately reflect changes in the capital structure of Perini.

*Eligibility and Limitations on Grants.* All full-time and part-time officers, employees, non-employee directors and other key persons of Perini and its subsidiaries are eligible to participate in the 2004 Plan, subject to the discretion of the Administrator.

The maximum award of stock options or stock appreciation rights granted to any one individual will not exceed 200,000 shares of common stock (subject to adjustment for stock splits and similar events) for any calendar year period. If any award of restricted stock or deferred stock granted to an individual is intended to qualify as performance based compensation under Section 162(m) of the Internal Revenue Code, then the maximum award shall not exceed 200,000 shares of common stock (subject to adjustment for stock splits and similar events) to any one such individual in any performance cycle.

*Stock Options*. Options granted under the 2004 Plan may be either incentive stock options (within the meaning of Section 422 of the Internal Revenue Code) or non-qualified stock options. Incentive options may be granted only to employees of Perini or any subsidiary. Options granted under the 2004 Plan will be non-qualified options if they (i) fail to qualify as incentive options, (ii) are granted to a person not eligible to receive incentive options under the Internal Revenue Code, or (iii) otherwise so provide. Non-qualified options may be granted to any persons eligible to receive incentive stock options and to non-employee directors and other key persons.

*Other Option Terms.* The Administrator has authority to determine the terms of options granted under the 2004 Plan. Generally, options are granted with an exercise price that is not less than the fair market value of the shares of common stock on the date of the option grant, except for options that are granted in lieu of cash compensation.

The term of each option will be fixed by the Administrator and may not exceed ten years from the date of grant. The Administrator will determine at what time or times each option may be exercised and, subject to the provisions of the 2004 Plan, the period of time, if any, after retirement, death, disability or termination of employment during which options may be exercised. Options may be made exercisable in installments, and the exercisability of options may be accelerated by the Administrator. In general, unless otherwise permitted by the Administrator, no option granted under the 2004 Plan is transferable by the optionee other than by will or by the laws of descent and distribution, and options may be exercised during the optionee s lifetime only by the optionee, or by the optionee s legal representative or guardian in the case of the optionee s incapacity.

Options granted under the 2004 Plan may be exercised for cash or by transfer to Perini (either actually or by attestation) of shares of common stock that are not then subject to restrictions under any Perini stock plan, and that have been held by the optionee for at least six months or were purchased on the open market, and that have a fair market value equivalent to the option exercise price of the shares being purchased. Subject to applicable law, options granted under the 2004 Plan also may be exercised by compliance with certain provisions pursuant to which a securities broker delivers the purchase price for the shares to us.

To qualify as incentive options, options must meet additional requirements under the Internal Revenue Code, including a \$100,000 limit on the value of shares subject to incentive options which first become exercisable in any one calendar year, and a shorter term and higher minimum exercise price in the case of certain large stockholders.

*Stock Appreciation Rights.* The Administrator may award a stock appreciation right either as a freestanding award or in tandem with a stock option. Upon exercise of the stock appreciation right, the holder will be entitled to receive an amount equal to the excess of the fair market value on the date of exercise of one share of common stock over the exercise price of the stock appreciation right (or more than the option exercise price per share if such stock appreciation right was granted in tandem with a stock option) times the number of shares of common stock with respect to which the stock appreciation right is exercised. This amount may be paid in cash, in shares of common stock, or a combination of cash and common stock, as determined by the Administrator. The exercise price per share of stock appreciation rights may not be less than 100% of the fair market value of the shares of common stock on the date of grant.

*Restricted Stock Awards.* The Administrator may grant shares, at a purchase price (which may be zero) determined by the Administrator, of common stock to any participant subject to such conditions and restrictions as the Administrator may determine. These conditions and restrictions may include the achievement of pre-established performance goals and/or continued employment with Perini through a specified vesting period. The vesting period shall be determined by the Administrator. However, in the event these awards have a performance-based goal, the restriction period will be at least one year, and in the event these awards have a time-based restriction, the restriction period will be at least three years. If the applicable performance goals and other restrictions are not attained, the participant will forfeit his or her award of restricted stock.

*Unrestricted Stock Awards*. The Administrator may also grant shares (at par value or a higher purchase price determined by the Administrator) of common stock that are free from any restrictions under the 2004 Plan. Unrestricted stock may be granted to any participant in recognition of past services or other valid consideration, and may be issued in lieu of cash compensation due to such participant.

*Deferred Stock Awards*. The Administrator also may award phantom stock units as deferred stock awards to participants. The deferred stock awards are ultimately payable in the form of shares of common stock and may be subject to such conditions and restrictions as the Administrator may determine. These conditions and restrictions may include the achievement of certain performance goals and/or continued employment with Perini through a specified vesting period. However, in the event these awards have a performance-based goal, the restriction period will be at least one year, and in the event these awards have a time-based restriction, the restriction period will be at least three years. During the deferral period, subject to terms and conditions imposed by the Administrator, the deferred stock awards may be credited with dividend equivalent rights (discussed below). Subject to the consent of the Administrator, a participant may make an advance election to receive a portion of his or her compensation or restricted stock award otherwise due in the form of a deferred stock award.

*Dividend Equivalent Rights.* The Administrator may grant dividend equivalent rights that entitle the recipient to receive credits for dividends that would be paid if the recipient had held specified shares of common stock. Dividend equivalent rights may be granted as a component of another award or as a freestanding award. Dividend equivalent rights credited under the 2004 Plan may be paid currently or be deemed to be reinvested in additional shares of common stock, that may thereafter accrue additional dividend equivalent rights at fair market value at the time of deemed reinvestment or on the terms then governing the reinvestment of dividends under our dividend reinvestment plan, if any. Dividend equivalent rights may be settled in cash, shares of common stock or a combination thereof, in a single installment or installments, as specified in the award.

*Tax Withholding*. Participants under the 2004 Plan are responsible for the payment of any federal, state or local taxes that we are required by law to withhold upon any option exercise or vesting of any awards. Subject to approval by the Administrator, participants may elect to have the minimum tax withholding obligations satisfied either by authorizing us to withhold shares of common stock to be issued pursuant to an option exercise or other award, or by transferring to us shares of common stock having a value equal to the amount of such taxes.

*Adjustments for Stock Dividends, Mergers, etc.* The Plan authorizes the Administrator to make appropriate adjustments to the number of shares of common stock that are subject to the 2004 Plan and to any outstanding awards to reflect any reorganization, recapitalization, reclassification stock dividends, stock splits and other similar change in the common stock. In the event of certain transactions, such as a merger, consolidation, dissolution, sale of assets, liquidation of Perini or certain other sale events, all stock options and stock appreciation rights will automatically become fully exercisable and the restrictions and conditions on all other stock based awards will automatically be deemed waived. In addition, upon the effective time of any such transaction, the 2004 Plan and all awards will terminate unless the parties to the transaction, in their discretion, provide for appropriate substitutions or adjustments of outstanding stock options or other awards.

*Amendments and Termination.* Our board of directors may at any time amend or discontinue the 2004 Plan and the Administrator may at any time amend or cancel any outstanding award for the purpose of satisfying changes in law or for any other lawful purpose, but no such action shall adversely affect the rights under any outstanding awards without the holder s consent. Any amendments that materially change the terms of the 2004 Plan, including any amendments that increase the number of shares reserved for issuance under the 2004 Plan, expand the type of awards available, materially expand the eligibility to participate or materially extend the term of the 2004 Plan, or materially change the method of determining Fair Market Value, will be subject to approval by our Stockholders. To the extent required by the Internal Revenue Code to ensure that options granted under the 2004 Plan qualify as incentive options or that compensation earned under awards granted under the 2004 Plan qualify as performance-based compensation under the Internal Revenue Code, Plan amendments shall be subject to approval by our stockholders.

The following table sets forth grants of stock options received under the 2004 Plan during the calendar year ended December 31, 2007 by (i) the named executive officers, (ii) all current executive officers as a group, (iii) all current directors who are not executive officers as a group, and (iv) all employees, who are not executive officers, as a group. The granting of options is discretionary with the Compensation Committee, and the benefits to be received by the individuals and classes listed in the table are not determinable.

		No. of
Name and Position	Dollar Values (\$)	Restricted Stock Awards Granted
Ronald N. Tutor, Chief Executive Officer		
Kenneth R. Burk, Chief Financial Officer		
Michael E. Ciskey, Senior VP of Civil Construction		
Craig W. Shaw, Chief Executive Officer of Perini West		
Robert Band, Chief Operating Officer		
Mark A. Caspers, President of Perini Building Company		
All current executive officers as a group (14 persons)		
All Current Directors who are not executive officer as a group (8 persons) <sup>1</sup>		
All employees, including all current officers who are not executive officers, as a group		

#### (1) Includes director nominees.

The following table sets forth grants of shares of restricted stock received under the 2004 Plan during the calendar year ended December 31, 2007 by (i) the named executive officers, (ii) all current executive officers as a group, (iii) all current directors who are not executive officers as a group, and (iv) all employees, who are not executive officers, as a group. The granting of restricted stock is discretionary with the Compensation Committee, and the benefits to be received by the individuals and classes listed in the table are not determinable.

Name and Position Ronald N. Tutor, Chief Executive Officer	Dollar Value \$	No. of Restricted Stock Awards Granted
Kenneth R. Burk, Chief Financial Officer	2,661,000	50,000
Michael E. Ciskey, Senior VP of Civil Construction	2,001,000	50,000
Craig W. Shaw, Chief Executive Officer Perini Building Company		
Robert Band, Chief Operating Officer		
Mark A. Caspers, President of Perini Building Company		
All current executive officers as a group (14 persons)	2,661,000	50,000
All Current Directors who are not executive officer as a group (8 persons) <sup>1</sup>		
All employees, including all current officers who are not executive officers, as a group		

(1) Includes director nominees.

The following table sets forth grants of deferred stock awards received under the 2004 Plan during the calendar year ended December 31, 2007 by (i) the named executive officers, (ii) all current executive officers as a group, (iii) all current directors who are not executive officers as a group, and (iv) all employees, who are not executive officers, as a group. The granting of deferred stock awards is discretionary with the Compensation Committee, and the benefits to be received by the individuals and classes listed in the table are not determinable.

Name and Position	Dollar Values (\$)	No. of Restricted Stock Awards Granted
Ronald N. Tutor, Chief Executive Officer		
Kenneth R. Burk, Chief Financial Officer		
Michael E. Ciskey, Senior VP of Civil Construction		
Craig W. Shaw, Chief Executive Officer of Perini West		
Robert Band, Chief Operating Officer		
Mark A. Caspers, President of Perini Building Company		
All current executive officers as a group (14 persons)		
All Current Directors who are not executive officer as a group (8 persons) <sup>1</sup>		
All employees, including all current officers who are not executive officers, as a group		

(1) Includes director nominees.

**Board Recommendation** 

THE PERINI BOARD OF DIRECTORS APPROVED THE ADOPTION OF THE AMENDMENT TO THE 2004 PLAN TO INCREASE THE NUMBER OF AUTHORIZED SHARES FROM 3 MILLION TO 5.5 MILLION AND RECOMMENDS A VOTE FOR APPROVAL OF THE AMENDMENT TO THE 2004 PLAN.

#### **PROPOSAL 6: ADJOURNMENT OR POSTPONEMENT OF THE ANNUAL MEETING**

#### Discussion

We propose to approve the adjournment or postponement of the annual meeting by the Perini shareholders if necessary to solicit additional proxies or as otherwise necessary.

#### **Board Recommendation**

#### THE PERINI BOARD OF DIRECTORS RECOMMENDS THAT THE PERINI SHAREHOLDERS VOTE FOR THE ADJOURNMENT OR POSTPONEMENT OF THE ANNUAL MEETING IF NECESSARY TO SOLICIT ADDITIONAL PROXIES OR AS OTHERWISE NECESSARY.

#### SHAREHOLDER PROPOSALS FOR 2009 ANNUAL MEETING

Any proposal of a shareholder submitted pursuant to Exchange Act Rule 14a-8 for inclusion in Perini s proxy statement and form of proxy for its 2009 annual meeting of shareholders must be received by Perini on or before [] in order to be considered for inclusion in its proxy statement and form of proxy. Such proposals must comply with the requirements as to form and substance established by the SEC if such proposals are to be included in the proxy statement and form of proxy. Any such proposal should be mailed to: Perini Corporation, 73 Mt. Wayte Avenue, Framingham, Massachusetts 01701-9160, Attn: Corporate Secretary.

Perini s bylaws require that Perini be given advance written notice of matters that shareholders wish to present for action at an annual meeting of shareholders (other than matters included in Perini s proxy materials in accordance with Rule 14a-8 under the Exchange Act). Any proposal of a shareholder intended to be presented at Perini s 2009 annual meeting of shareholders, other than shareholder proposals submitted pursuant to Exchange Act Rule 14a-8, must be received by us not earlier than [], nor later than []. If a shareholder fails to provide timely notice of a proposal to be presented at the 2009 annual meeting, the proxies designated by the Perini board of directors will have discretionary authority to vote on any such proposal that may come before the meeting. In addition, shareholder proposals must comply with the requirements of our bylaws. Any such proposal should be mailed to: Perini Corporation, 73 Mt. Wayte Avenue, Framingham, Massachusetts 01701-9160, Attn: Corporate Secretary.

Please see Nominations for Director for a description of the requirements for submitting a candidate for nomination as a director at the 2009 annual meeting of shareholders.

#### **OTHER MATTERS**

The board of directors knows of no other matters that are likely to be brought before the meeting. However, if any other matters of which the board of directors is not aware are presented to the meeting for action, it is the intention of the persons named in the accompanying form of proxy to vote said proxy in accordance with their judgment on such matters.

Perini will bear the cost of solicitation of proxies. The solicitation of proxies by mail may be followed by telephone or oral solicitation of certain shareholders and brokers.

#### HOUSEHOLDING OF ANNUAL MEETING MATERIALS

Some banks, brokers and other nominee record holders may be participating in the practice of householding proxy statements and annual reports. This means that only one copy of our proxy statement or annual report may have been sent to multiple shareholders in your household. We will promptly deliver a separate copy of either document to you if you call or write us at the following address or telephone number: Perini Corporation, 73 Mt. Wayte Avenue, Framingham, MA 01701, Attention: Susan C. Mellace, Secretary, (508) 628-2000. If you want to receive separate copies of the annual report and proxy statement in the future, or if you are receiving multiple copies and would like to receive only one copy for your household, you should contact your bank, broker, or other nominee record holder, or you may contact us at the above address and telephone number.

#### WHERE YOU CAN FIND ADDITIONAL INFORMATION

Perini files annual, quarterly and current reports, proxy statements and other information with the SEC. You may read and copy any reports, proxy statements or other information that we file with the SEC at the following location of the SEC:

Public Reference Room

100 F. Street, N.E.

Washington, D.C. 20549

Please call the SEC at 1-800-SEC-0330 for further information on the public reference rooms. You may also obtain copies of reports, proxy statements or other information concerning us, including any document incorporated by reference in this proxy statement, without charge, by written or telephonic request directed to us at Perini Corporation, 73 Mt. Wayte Avenue, Framingham, MA 01701, Attention: Susan C. Mellace, Corporate Secretary, (508) 628-2000. If you would like to request documents, please do so by [], 2008, in order to receive them before the annual meeting.

#### **INCORPORATION BY REFERENCE**

The SEC allows us to incorporate by reference into this proxy statement documents that we file with the SEC. This means that we can disclose important information to you by referring you to those documents. The information incorporated by reference is considered to be part of this proxy statement, and later information that we file with the SEC will update and supersede that information. We incorporate by reference the documents listed below and any documents filed pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of this proxy statement and prior to the date of the annual meeting:

Company Filings: Annual Reports on Forms 10-K and 10-K/A Quarterly Report on Form 10-Q Current Reports on Form 8-K Period (if applicable): Year ended December 31, 2007 Quarter ended March 31, 2008 Filed April 2, 2008; April 3, 2008; April 7, 2008;

May 19, 2008; June 30, 2008 and July 29, 2008

## PROXIES AND VOTING PROCEDURES

If you are a shareholder of record, you may vote your shares over the Internet at www.investorvote.com/PCR or telephonically 1-800-652-VOTE (1-800-652-8683) or by following the instructions on the enclosed proxy card. Proxies submitted via the Internet or by telephone must be received by 2:00 a.m. eastern time on [], 2008.

If the shares you own are held in street name by a bank or brokerage firm, your bank or brokerage firm will provide a vote instruction form to you with this proxy statement, which you may use to direct how your shares will be voted. Many banks and brokerage firms also offer the option of voting via the Internet or by telephone, instructions for which would be provided by your bank or brokerage firm on your vote instruction form.

Annex A

**EXECUTION VERSION** 

## AGREEMENT AND PLAN OF MERGER

BY AND AMONG

## PERINI CORPORATION,

## TRIFECTA ACQUISITION LLC,

## TUTOR-SALIBA CORPORATION,

## **RONALD N. TUTOR**

## AND

## SHAREHOLDERS OF TUTOR-SALIBA CORPORATION

## DATED AS OF APRIL 2, 2008

## TABLE OF CONTENTS

		Page
ARTIC	<u>CLE I THE MERGER; ALTERNATIVE STRUCTURE</u>	A-2
1.1	The Merger	A-2
1.2	Closing	A-2
1.3	Effective Time	A-2
1.4	Effect of the Merger	A-3
1.5	Surviving Entity Organizational Documents	A-3
1.6	Surviving Entity Members	A-3
1.7	Conversion of Company Shares	A-3
1.8	Taking of Necessary Action; Further Action	A-3
1.9	Parent Charter Amendment	A-4
	Effective Time Parent Board	A-4
1.11	<u>Alternative Structure</u>	A-4
	CLE II EXCHANGE OF CERTIFICATES IN THE MERGER	A-4
2.1	Exchange of Certificates	A-4
2.2	Dissenting Shares	A-6
2.3	Withholding	A-6
2.4	Adjustments	A-6
2.5	Acknowledgement by Recipient of Merger Consideration	A-6
	CLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY	A-6
3.1	Organization and Qualification	A-6
3.2	Subsidiaries	A-7
3.3	Authority	A-7
3.4	<u>No Conflict</u>	A-8
3.5	Governmental Consents	A-8
3.6	Capitalization	A-8
3.7	Financial Statements	A-9
3.8	No Undisclosed Liabilities	A-9
3.9	Absence of Certain Changes	A-9
	Compliance with Laws and Permits	A-10
	Litigation	A-10
	Assets	A-10
	Material Contracts	A-11
	Government Contracts	A-11
	Real Property	A-13
	Intellectual Property	A-14
	Insurance	A-15
	Environmental Matters	A-15
	Tax Matters	A-15
	Labor Matters	A-17
	Employment Matters	A-17
	Interested Party Transactions	A-19
	Certain Payments: Suitability	A-19
	Indemnification	A-19
	Required Vote	A-19
	Certain Approvals; Takeover Laws	A-20
	Brokers and Finders Fees	A-20
3.28	Ownership of Parent Equity Interests	A-20

## TABLE OF CONTENTS

		Page
ARTIC	CLE IV REPRESENTATIONS AND WARRANTIES OF THE SHAREHOLDERS	A-20
4.1	Ownership of Subject Shares	A-20
4.2	Authority	A-20
4.3	No Conflict	A-20
4.4	Investment	A-21
4.5	S Corporation Status	A-21
4.6	Accredited Investor	A-21
4.7	Ownership of Parent Equity Interests	A-21
ARTIC	CLE V REPRESENTATIONS AND WARRANTIES OF PARENT	A-21
5.1	Organization and Qualification	A-21
5.2	Subsidiaries	A-22
5.3	Authority	A-22
5.4	No Conflict	A-23
5.5	Governmental Consents	A-23
5.6	Capitalization	A-24
5.7	SEC Reports; Financial Statements	A-24
5.8	No Undisclosed Liabilities	A-25
5.9	Absence of Certain Changes	A-25
5.10	Compliance with Laws and Permits	A-25
5.11	Litigation	A-26
5.12	Assets	A-26
5.13	Material Contracts	A-26
	Government Contracts	A-27
5.15	Real Property	A-29
	Intellectual Property	A-30
	Insurance	A-30
5.18	Environmental Matters	A-30
	Tax Matters	A-31
5.20	Labor Matters	A-32
	Employment Matters	A-33
	Interested Party Transactions	A-34
	Certain Payments; Suitability	A-34
	Indemnification	A-35
	Required Vote	A-35
	Certain Approvals; Takeover Laws	A-35
	Brokers and Finders Fees	A-35
	Opinion of Financial Advisor	A-35
	Merger Sub	A-35
ARTIC	CLE VI COVENANTS	A-35
6.1	Conduct of the Company and its Subsidiaries	A-35
6.2	Conduct of Parent and its Subsidiaries	A-38
6.3	No Solicitation by the Company or Shareholders; Shareholder Lock-Up	A-39
6.4	No Solicitation by Parent	A-40
6.5	Parent Board Recommendation	A-41
6.6	Proxy Statement; Parent Shareholders Meeting	A-42
6.7	Obligations to Consummate the Merger	A-43
6.8	Access	A-44

## TABLE OF CONTENTS

		Page
6.9	Notices of Certain Events	A-44
6.10	Public Announcements	A-45
6.11	Section 16 Matters	A-45
6.12	NYSE Listing	A-45
6.13	<u>Tax Treatment</u>	A-45
6.14	Tax Matters.	A-46
6.15	Shareholder Representative	A-46
6.16	Non-Competition; Non-Hire	A-47
6.17	Compliance with Obligations	A-48
6.18	Signatory Shareholders Joinder	A-48
6.19	Guarantees	A-49
6.20	Pre-Closing Dividend and Dividend Notes	A-49
-	E VII CONDITIONS PRECEDENT TO THE CLOSING	A-49
7.1	Mutual Conditions to Merger	A-49
7.2	Parent s Conditions to the Merger	A-50
7.3	The Company s Conditions to the Merger	A-51
7.4	No Frustration of Closing Conditions	A-52
ARTICLI	E VIII TERMINATION	A-52
8.1	Termination	A-52
8.2	Effect of Termination; Controlling Trusts Guarantee	A-54
8.3	Expenses: Termination Fee	A-54
ARTICLI	E IX SURVIVAL; INDEMNIFICATION	A-55
9.1	<u>Survival</u>	A-55
9.2	Indemnification	A-55
9.3	Limitations	A-56
9.4	<u>Claims</u>	A-58
9.5	Adjustments to Merger Consideration	A-59
9.6	Waiver of Reliance on Excluded Information	A-59
9.7	Nature of Shareholder Indemnification Obligations	A-60
9.8	Sole and Exclusive Remedy	A-60
ARTICLI	E X DEFINITIONS, CONSTRUCTION, ETC.	A-60
10.1	Certain Definitions	A-60
10.2	Index of Defined Terms	A-68
10.3	Construction	A-72
ARTICLI	E XI GENERAL PROVISIONS	A-73
11.1	Notices	A-73
11.2	Entire Agreement	A-75
11.3	Severability	A-75
11.4	Amendment	A-75
11.5	Extension; Waiver	A-75
11.6	Specific Performance	A-76
11.7	Other Remedies	A-76
11.8	Successors and Assigns; Assignment; No Third Party Beneficiaries	A-76
11.9	GOVERNING LAW	A-76
11.10	Consent to Jurisdiction; Service of Process	A-76
11.11	WAIVER OF JURY TRIAL	A-77
11.12	Counterparts; Facsimile Delivery	A-77

## TABLE OF CONTENTS

Exhibit A Company Capitalization

- Annex A Company Shareholder Written Consent
- Annex B Shareholders Agreement
- Annex C Form of Employment Agreement
- Annex D Form of Articles of Amendment
- Annex E Form of Dividend Note

Company Schedule Parent Schedule

A-iv

#### AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER, dated as of April 2, 2008 (this <u>Agreement</u>), is entered into by and among **Perini Corporation**, a Massachusetts corporation (<u>Parent</u>), **Trifecta Acquisition LLC**, a California limited liability company and a wholly-owned subsidiary of Parent (<u>Merger Sub</u>), **Tutor-Saliba Corporation**, a California corporation (the <u>Company</u>), **Ronald N. Tutor**, a resident of California and a trustee under each of the Controlling Trusts (as defined below) (in the capacity as Shareholder Representative as provided in <u>Section 6.15</u> or in any other capacity contemplated hereby, the <u>Shareholder Representative</u>), and shareholders of the Company signatory hereto (including the Shareholder Representative, each a <u>Signatory Shareholder</u> and collectively, the <u>Signatory Shareholder</u> and collectively, the <u>Shareholders</u>) to become a Signatory Shareholder in accordance with Section 6.18. Parent, Merger Sub, the Company and the Signatory Shareholders are referred to herein each individually as a <u>Party</u> and collectively as the <u>Pa</u>rties . All capitalized terms used in this Agreement shall have the respective meanings ascribed thereto in <u>Article X</u>.

#### RECITALS

WHEREAS, the board of directors of Parent (the <u>Parent Board</u>), acting upon the recommendation of a special committee of independent directors thereof (the <u>Special Committee</u>), and the respective boards of directors of Merger Sub and the Company have each determined that it is in the best interests of their respective shareholders or members, as appropriate, to enter into a business combination transaction;

WHEREAS, Parent, Merger Sub and the Company intend to effect a merger of the Company with and into Merger Sub (the <u>Merger</u>), with Merger Sub to be the surviving entity of the Merger and a wholly-owned Subsidiary of Parent, on the terms and subject to the conditions set forth herein and in accordance with the General Corporation Law of the State of California and the Limited Liability Company Act of the State of California (collectively, the <u>CGCL</u>);

**WHEREAS**, the board of directors of Parent, acting upon the recommendation of the Special Committee, and the respective boards of directors of Merger Sub and the Company have each (i) determined that this Agreement and the transactions contemplated hereby, including the Merger, are advisable, fair to and in the best interests of their respective shareholders or members, as appropriate, and (ii) approved this Agreement and the transactions contemplated hereby, including the Merger, on the terms and subject to the conditions set forth herein;

**WHEREAS,** as a condition to the Company s willingness to enter into this Agreement, simultaneously with or immediately after the execution and delivery of this Agreement, Parent, as the sole member of Merger Sub, is executing and delivering or will execute and deliver an unconditional unanimous written consent to the approval of this Agreement and the transactions contemplated hereby, including the Merger;

**WHEREAS,** as a condition to Parent s and Merger Sub s willingness to enter into this Agreement, simultaneously with or immediately after the execution and delivery of this Agreement, the Shareholders, who collectively own all of the issued and outstanding capital stock of the Company, are executing and delivering or will execute and deliver an unconditional unanimous written consent to the approval of this Agreement and the transactions contemplated hereby, including the Merger, substantially in the form attached as <u>Annex A</u> hereto (the <u>Company Shareholder Written Consent</u>);

**WHEREAS,** as a condition to Parent s and Merger Sub s willingness to enter into this Agreement, simultaneously with the execution and delivery of this Agreement, Parent and the Signatory Shareholders party hereto on the date hereof are entering into the Shareholders Agreement in the form attached as <u>Annex B</u> hereto

(the <u>Shareholders Agreement</u>), with all other Shareholders other than the Signatory Shareholders party hereto on the date hereof to become a party thereto in accordance with <u>Section 6.18</u>, which Shareholders Agreement shall become effective upon the closing of the Merger;

**WHEREAS,** as a condition to Parent s and Merger Sub s willingness to enter into this Agreement, simultaneously with the execution and delivery of this Agreement, Parent and the Shareholder Representative are entering into the Employment Agreement in the form attached as <u>Annex C</u> hereto (the <u>Employment Agreement</u>), which Employment Agreement shall become effective upon the closing of the Merger;

**WHEREAS,** for United States federal income tax purposes, Parent, Merger Sub, the Company and the Shareholders intend that the Merger shall qualify as a reorganization within the meaning of Section 368(a) of the Code; and

**WHEREAS**, Parent, Merger Sub, the Company and the Shareholders desire to make certain representations, warranties, covenants and agreements in connection with, and to prescribe various conditions to, the Merger and the other transactions contemplated hereby.

#### AGREEMENT

**NOW, THEREFORE,** in consideration of the foregoing premises and the representations, warranties, covenants and agreements set forth herein, as well as other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, and intending to be legally bound hereby, Parent, Merger Sub, the Company and the Shareholders hereby agree as follows:

#### ARTICLE I

## THE MERGER; ALTERNATIVE STRUCTURE

1.1 <u>The Merger</u>. Upon the terms and subject to the conditions set forth in this Agreement and the applicable provisions of the CGCL, at the Effective Time, the Company shall be merged with and into Merger Sub, the separate corporate existence of the Company shall thereupon cease, and Merger Sub shall continue as the surviving entity of the Merger. Merger Sub, as the surviving entity of the Merger, is sometimes hereinafter referred to as the <u>Surviving Entity</u>.

1.2 <u>Closing</u>. The consummation of the Merger (the <u>Closing</u>) shall take place at a closing to occur at the offices of Kirkland & Ellis LLP located at 655 Fifteenth Street, N.W., Washington, D.C. 20005 at 10:00 a.m. (local time) on the third (3rd) Business Day after the satisfaction or waiver (to the extent permitted hereunder) of the conditions set forth in <u>Article VII</u> hereof (other than those conditions that by their terms are to be satisfied at the Closing, but subject to the satisfaction or waiver (to the extent permitted hereunder) of such conditions), or at such other location, date and time as Parent and the Company shall mutually agree upon in writing (the date upon which the Closing shall actually occur pursuant hereto being referred to herein as the <u>Closing Date</u>).

1.3 <u>Effective Time</u>. Upon the terms and subject to the conditions set forth in this Agreement, on the Closing Date, Parent, Merger Sub and the Company shall cause the Merger to be consummated in accordance with the CGCL by filing a certificate of merger in customary form and substance (the <u>Certificate of Merger</u>) with the Secretary of State of the State of California (the <u>California Secretary of State</u>) in accordance with the applicable provisions of the CGCL (the time of such filing and acceptance by the California Secretary of State, or such later time as may be agreed in writing by Parent and the Company and specified in the Certificate of Merger, being referred to herein as the <u>Effective Time</u>).

1.4 Effect of the Merger. At the Effective Time, the effect of the Merger shall be as provided in this Agreement and the applicable provisions of the CGCL. Without limiting the generality of the foregoing, and

subject thereto, at the Effective Time, all rights and property of the Company and Merger Sub shall vest in the Surviving Entity, and all debts and liabilities of the Company and Merger Sub shall become the debts and liabilities of the Surviving Entity.

1.5 <u>Surviving Entity Organizational Documents</u>. At the Effective Time, the articles of organization and operating agreement of Merger Sub, as in effect immediately prior to the Effective Time, shall be the articles of organization and operating agreement of the Surviving Entity until thereafter amended in accordance with Applicable Law and such articles of organization and operating agreement; <u>provided</u>, <u>however</u>, that, at the Effective Time, the articles of organization of the Surviving Entity shall be amended so that the name of the Surviving Entity shall be Tutor-Saliba LLC .

1.6 <u>Surviving Entity Members</u>. At the Effective Time, the initial members of the Surviving Entity shall be the members of Merger Sub immediately prior to the Effective Time, each to hold office in accordance with the articles of organization and any operating agreement of the Surviving Entity until his or her respective successor is duly elected or appointed and qualified. In order to effectuate the foregoing, the Company and each of the Shareholders shall use its commercially reasonable efforts to procure, in connection with the Closing, the resignation of each of the Company s directors to be effective immediately prior to the Effective Time.

1.7 <u>Conversion of Company Shares</u>. Upon the terms and subject to the conditions set forth in this Agreement, at the Effective Time, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company or the holders of any of the following securities (including the Shareholders), the following shall occur:

(a) <u>Membership Interests of Merger Sub</u>. Each membership interest of Merger Sub that is issued and outstanding immediately prior to the Effective Time shall, by virtue of the Merger and without any action on the part of the holder thereof, be converted into the right to receive one membership interest of the Surviving Entity, so that at the Effective Time, Parent shall be the holder of all of the issued and outstanding membership interests of the Surviving Entity. Each certificate evidencing ownership of such membership interests of Merger Sub shall cease to have any rights with respect thereto except that thereafter it shall evidence ownership of a membership interest of the Surviving Entity.

(b) <u>Owned Company Common Stock</u>. Each share of common stock, no par value per share, of the Company (the <u>Company Common Stock</u>) owned by Parent or any other direct or indirect, wholly-owned Subsidiary of Parent or the Company immediately prior to the Effective Time shall, by virtue of the Merger and without any action on the part of the holder thereof, be automatically canceled and retired and shall cease to exist, and no payment or other consideration shall be made with respect thereto.

(c) <u>Conversion of Company Common Stock</u>. Each share of Company Common Stock issued and outstanding immediately prior to the Effective Time (other than those shares cancelled and retired pursuant to <u>Section 1.7(b)</u> or any Dissenting Shares) shall, by virtue of the Merger and without any action on the part of the holder thereof, be automatically canceled and retired and shall cease to exist and shall be converted into the right to receive a number of duly authorized, validly issued, fully paid and nonassessable shares of common stock, par value \$1.00 per share, of Parent (the <u>Parent Common Stock</u>) equal to the quotient (the <u>Exchange Ratio</u>) obtained by dividing (A) 22,987,293 shares of Parent Common Stock by (B) the Outstanding Common Stock Number (the shares of Parent Common Stock issuable pursuant to this <u>Section 1.7(c)</u>, collectively, the <u>Parent Shares</u>). Set forth <u>on Exhibit</u> A hereto is a calculation of the aggregate Parent Shares to be issuable to each Shareholder as a result of the cancellation, retirement and conversion of the shares of Company Common Stock pursuant to this <u>Section 1.7(c)</u>. As of the Effective Time, each certificate evidencing ownership of such shares of Company Common Stock shall cease to have any rights with respect thereto except the right to receive the Parent Shares to be issued in consideration therefor upon the surrender of such certificate in accordance with <u>Section 2.1</u>.

1.8 Taking of Necessary Action: Further Action. If, at any time after the Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest the Surviving Entity with full

right, title and possession to all assets, property, rights, privileges, powers and franchises of the Company and Merger Sub, the directors and officers of the Surviving Entity shall take all such lawful and necessary action, consistent with this Agreement, on behalf of the Company and Merger Sub.

1.10 <u>Effective Time Parent Board</u>. The Parties will take all actions required such that as of the Effective Time (a) the Parent Board shall consist of the directors determined in compliance with Section 2(b) of the Shareholders Agreement and (b) the committees of the Parent Board shall be comprised of such directors determined in compliance with Section 2(d) of the Shareholders Agreement.

1.11 <u>Alternative Structure</u>. If (i) the Parent Shareholder Approval has not been obtained and (ii) Parent and the Company determine that an alternative structure of the transactions contemplated hereby is necessary to avoid undue delay of the consummation of the Merger and the other transactions contemplated hereby and the other Related Agreements or otherwise to address legal, commercial or regulatory concerns of the other transactions contemplated hereby that would reasonably be avoided by the use of such alternative structure, then each Party hereto will cooperate in good faith to make changes to the terms hereof and the other Related Agreements or the greatest extent possible, the economic, business and other purposes of the terms hereof and the other Related Agreements.

#### **ARTICLE II**

#### EXCHANGE OF CERTIFICATES IN THE MERGER

#### 2.1 Exchange of Certificates.

(a) Exchange Agent. Immediately following the Effective Time, Parent shall deposit with a bank or trust company of national recognition (the Exchange Agent ), for the benefit of the holders of Company Common Stock immediately prior to the Effective Time for exchange in accordance with this Article II through the Exchange Agent certificates representing the Parent Shares issuable pursuant to Section 1.7(c) in exchange for outstanding shares of Company Common Stock canceled, retired and converted pursuant to Section 1.7(c) (such shares of Parent Common Stock, together with any dividends or distributions with respect thereto, being hereinafter referred to as the Exchange Fund ). In lieu of a bank or trust company, Parent or any of its Subsidiaries may serve as the Exchange Agent hereunder with the consent of the Company prior to the Effective Time.

(b) Exchange Procedures. As soon as reasonably practicable after the Effective Time, the Exchange Agent shall mail to each holder of record of a certificate or certificates that immediately prior to the Effective Time represented outstanding shares of Company Common Stock that were canceled, retired and converted pursuant to Section 1.7(c) (i) a letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to the certificates shall pass, only upon delivery of the certificates to the Exchange Agent and shall be in such form and have such other provisions as Parent may reasonably specify) and (ii) instructions for use in surrendering the certificates in exchange for the Parent Shares into which the shares of Company Common Stock were converted. Upon the surrender to the Exchange Agent of a certificate that formerly represented shares of

Company Common Stock in accordance with the provisions of this <u>Section 2.1</u>, together with such letter of transmittal, duly executed, and such other documents as may reasonably be required by the Exchange Agent, the Exchange Agent shall deliver to the holders of certificates that immediately prior to the Effective Time represented outstanding shares of Company Common Stock certificates representing the number of shares of Parent Common Stock into which such shares shall have been converted in accordance with <u>Section 1.7(c)</u>, and the certificate so surrendered shall forthwith be cancelled.

(c) <u>No Further Ownership Rights in the Company</u>. At and after the Effective Time, each holder of shares of Company Common Stock immediately prior to the Effective Time shall cease to have any rights as a shareholder of the Company except for the right to surrender such shareholder s certificates that immediately prior to the Effective Time represented outstanding shares of Company Common Stock in exchange for receipt of Parent Common Stock in accordance with <u>Section 1.7(c)</u>.

(d) <u>Transfers of Ownership</u>. After the Effective Time, no transfer of shares of Company Common Stock which were outstanding immediately prior to the Effective Time shall be made on the stock transfer books of the Company.

(e) Lost, Stolen or Destroyed Certificates. In the event that any certificates representing shares of Company Common Stock immediately prior to the Effective Time shall have been lost, stolen or destroyed, the Exchange Agent or Parent, as applicable, shall issue in exchange for such lost, stolen or destroyed certificates, upon the making of an affidavit of that fact by the holder thereof, the Parent Shares deliverable in respect thereof pursuant to Section 1.7(c); provided, however, that Parent may, in its discretion and as a condition precedent to the delivery of such shares, require the owners of such lost, stolen or destroyed certificates to deliver a bond in such sum as it may reasonably direct as indemnity against any claim that may be made against Parent, the Surviving Entity or the Exchange Agent with respect to the certificates alleged to have been lost, stolen or destroyed.

(f) <u>No Fractional Shares</u>. No fraction of a share of Parent Common Stock shall be issued in the Merger. In lieu of any fractional shares, the fractional amount of Parent Common Stock to which any holder of Company Common Stock is entitled to receive under <u>Section 1.7(c)</u> shall be rounded down to the nearest whole number and the holder thereof shall be entitled to receive an amount in cash, without interest, equal to the product obtained by multiplying (x) the fraction of a share of Parent Common Stock to which such holder (after taking into account all shares of Company Common Stock and all certificates held immediately prior to the Effective Time by such holder) would otherwise be entitled to and (y) the closing price per share of Parent Common Stock as reported on the NYSE Composite Transaction Tape (as reported in The Wall Street Journal, or, if not reported thereby, any other authoritative source) on the last trading day prior to but not including the date of the Effective Time.

(g) <u>No Liability</u>. Notwithstanding anything to the contrary set forth in this Agreement, none of the Escrow Agent, Parent or the Surviving Entity shall be liable to a holder of shares of Company Common Stock immediately prior to the Effective Time for any amount or property properly paid or transferred to a public official pursuant to any applicable abandoned property, escheat or similar law.

(h) <u>Distribution of Exchange Fund to Parent</u>. Any portion of the Exchange Fund that remains undistributed to the holders of certificates representing shares of Company Common Stock immediately prior to the Effective Time on the date that is twelve (12) months after the Effective Time shall be delivered to Parent upon demand, and any holders of shares of Company Common Stock that were issued and outstanding immediately prior to the Merger who have not theretofore surrendered such certificates evidencing such shares of Company Common Stock for exchange pursuant to the provisions of this <u>Article II</u> shall thereafter look for issuance of the Parent Shares in respect of the shares of Company Common Stock evidenced by such certificates solely to Parent, as general creditors thereof, for any claim to the applicable merger consideration to which such holders may be entitled pursuant to the provisions of this <u>Article II</u>.

(i) <u>Distributions with Respect to Unexchanged Shares</u>. No dividends or other distributions with respect to Parent Common Stock with a record date after the Effective Time shall be paid to a holder of shares of Company Common Stock immediately prior to the Effective Time with respect to the shares of Parent Common Stock into which they were converted until surrender of shares of Company Common Stock in accordance with this <u>Article II</u>. Following surrender of any such shares of Company Common Stock, there shall be paid to the holder of the certificate representing shares of Parent Common Stock issued in exchange therefor, without interest, the amount of dividends or other distributions with a record date after the Effective Time payable with respect to such shares of Parent Common Stock.

2.2 <u>Dissenting Shares</u>. Pursuant to the delivery of the Company Shareholder Written Consent, all holders of shares of Company Common Stock shall waive any right to demand appraisal for such shares in accordance with the CGCL. If for any reason any holder of shares of Company Common Stock outstanding immediately prior to the Effective Time who has not voted in favor of the Merger or consented thereto in writing (such shares, <u>Dissenting Shares</u>) has properly demanded appraisal for such shares in accordance with the CGCL, such shares of Company Common Stock shall not be converted into a right to receive Parent Shares, unless such holder fails to perfect, withdraws or otherwise loses such holder s right to appraisal under the CGCL. If, after the Effective Time, such holder fails to perfect, withdraws or otherwise loses such holder s right to appraisal, each such share shall be treated as if it had been converted as of the Effective Time into a right to receive Parent Shares. The Company shall give Parent (i) prompt notice of (A) any demands for appraisal pursuant to the CGCL or otherwise received by the Company (B) withdrawals of such demands and (ii) the opportunity to direct all negotiations and proceedings with respect to all such demands for appraisal under the CGCL or otherwise prior to the Effective Time. The Company shall not, except with the prior written consent of Parent or as otherwise required by any Applicable Law, make any payment with respect to any such demands for appraisal or offer to settle or settle any such demands.

2.3 <u>Withholding</u>. Each of Parent, the Surviving Entity and Exchange Agent, as applicable, shall be entitled to deduct and withhold from any amounts payable by it pursuant to this Agreement any withholding Taxes or other amounts required by any Applicable Law to be deducted and withheld. To the extent that any such amounts are so deducted or withheld, such amounts will be treated for all purposes of this Agreement as having been paid prior to the Closing to the Person in respect of which such deduction and withholding was made.

2.4 <u>Adjustments</u>. The numbers of shares of Parent Common Stock issuable pursuant to <u>Section 1.7(c)</u> shall be equitably adjusted to reflect appropriately the effect of any stock split, reverse stock split, stock dividend (including any dividend or distribution of securities convertible into Parent Common Stock), extraordinary cash dividends, merger, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to Parent Common Stock occurring on or after the date hereof.

2.5 <u>Acknowledgement by Recipient of Merger Consideration</u>. Without limiting any other provision hereof, each holder of certificates representing shares of Company Common Stock converted into Parent Shares pursuant to <u>Section 1.7(c)</u> receiving such Parent Shares shall be deemed to have acknowledged and agreed to be bound by its obligations set forth in this Agreement including, without limitation, the obligation to indemnify Parent Indemnified Parties pursuant to <u>Article IX</u>.

#### ARTICLE III

#### REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to Parent and Merger Sub as of the date hereof as follows:

3.1 <u>Organization and Qualification</u>. The Company is a corporation duly organized, validly existing and in good standing under the laws of the State of California. The Company has all requisite corporate power and

authority to own, lease and operate its properties and to carry on its business. The Company is duly qualified or licensed to do business and is in good standing as a foreign corporation in each jurisdiction in which the conduct of its business or the ownership, leasing, holding or use of its properties makes such qualification necessary, except such other jurisdictions where the failure to be so qualified or licensed or in good standing would not have a Company Material Adverse Effect. The Company has made available to Parent a true and correct copy of its articles of incorporation and bylaws, each as amended to date and in full force and effect on the date hereof.

#### 3.2 Subsidiaries.

(a) <u>Section 3.2(a) of the Company Schedule</u> sets forth a true and complete list of each Subsidiary of the Company, together with (i) the jurisdiction of formation of each such Subsidiary, (ii) the percentage of the outstanding Equity Interests of such Subsidiary owned by the Company or any of its other Subsidiaries, and (iii) if applicable, each other holder of Equity Interests of such Subsidiary. Neither the Company nor any of its Subsidiaries owns any Equity Interest in any Person other than the Subsidiaries set forth in <u>Section 3.2(a) of the Company</u> <u>Schedule</u>. All Equity Interests of each Subsidiary of the Company owned by the Company or any of its other Subsidiaries are owned free and clear of all Liens other than Permitted Liens.

(b) Each Subsidiary of the Company is duly organized, validly existing and in good standing (to the extent applicable) under the Applicable Laws of its jurisdiction of formation. Each Subsidiary of the Company has all requisite power and authority to own, lease and operate its properties and to carry on its business. Each Subsidiary of the Company is duly qualified or licensed to do business and is in good standing (to the extent applicable) as a foreign organization in each jurisdiction in which the conduct of its business or the ownership, leasing, holding or use of its properties makes such qualification necessary, except such other jurisdictions where the failure to be so qualified or licensed or in good standing would not have a Company Material Adverse Effect. The Company has made available to Parent a true and correct copy of the certificate of incorporation and bylaws or other comparable organizational documents of each Subsidiary of the Company, each as amended to date and in full force and effect on the date hereof.

(c) All of the outstanding share capital of each of the Company s Subsidiaries has been duly authorized and validly issued and is fully paid and non-assessable.

3.3 Authority. The Company has all requisite power and authority to execute and deliver this Agreement and all other Related Agreements to which it is or is to be a party and subject, in the case of the consummation of the Merger, to the receipt of the Company Shareholder Written Consent, to consummate the transactions contemplated hereby and thereby and to perform its obligations hereunder and thereunder. The execution and delivery by the Company of this Agreement and the other Related Agreements to which it is or is to be a party and the consummation by the Company of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action on the part of the Company, and no additional corporate proceedings on the part of the Company are necessary to authorize this Agreement or any other Related Agreement or the consummation of the transactions contemplated hereby or thereby, other than in the case of the consummation of the Merger, obtaining the Company Shareholder Written Consent. Without limiting the generality of the foregoing, the board of directors of the Company has duly and validly adopted resolutions (i) determining that this Agreement and the other Related Agreements to which it is or is to be a party and the Merger and the other transactions contemplated hereby and thereby are advisable, fair to and in the best interests of the Company and its shareholders; (ii) approving this Agreement and the other Related Agreements to which it is or is to be a party and the Merger and the other transactions contemplated hereby and thereby; and (iii) recommending that the Shareholders deliver the Company Shareholder Written Consent, which resolutions have not been subsequently rescinded, modified or withdrawn in any way. This Agreement and the other Related Agreements to which the Company is or is to be a party have been duly executed and delivered by the Company and, assuming the due authorization, execution and delivery by Parent and Merger Sub and any other counterparty thereto, constitutes a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as (a) may be limited by applicable bankruptcy, insolvency, reorganization, moratorium and other similar laws affecting or relating to creditors rights generally, and (b) is subject to general principles of equity.

3.4 <u>No Conflict</u>. None of the execution, delivery or performance by the Company of this Agreement or any other Related Agreements to which it is to be a party, the consummation by the Company of the transactions contemplated hereby or thereby or the compliance by the Company with any of the provisions hereof or thereof: (i) violate or conflict with any provision of the articles of incorporation or bylaws of the Company; (ii) violate or conflict with any provision or bylaws or other comparable organizational documents of any of the Company s Subsidiaries; (iii) except as set forth in <u>Section 3.4 of the Company Schedule</u> and subject to obtaining the Consents set forth in <u>Section 3.5 of the Company Schedule</u>, violate, conflict with, or result in the breach of or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or result in the termination of, or accelerate the performance required by or modification to the terms or conditions of, or result in a right of termination, acceleration or modification under, any Contract to which the Company or any of its Subsidiaries is a party or by which the Company, any of its Subsidiaries or any of their properties or assets may be bound; (iv) assuming compliance with the matters referred to in <u>Section 3.5 of the Company Schedule</u> and, in the case of the consummation of the Merger, subject to obtaining the Company Shareholder Written Consent, violate or conflict with any Applicable Law or Decree applicable to the Company or any of its Subsidiaries, except in the case of each of clauses (ii), (iii), (iv) and (v) above, for such violations, conflicts, breaches, defaults, terminations, accelerations or Liens that would not have a Company Material Adverse Effect.

3.5 <u>Governmental Consents</u>. Except as set forth on <u>Section 3.5 of the Company Schedule</u>, no Consent of any Governmental Entity is required on the part of the Company, any of its Subsidiaries or any of the Shareholders in connection with the execution, delivery or performance by the Company or the Shareholders of this Agreement or any other Related Agreement to which it is or is to be a party or the consummation by the Company or the Shareholders of the transactions contemplated hereby or thereby, except (a) the filing and recordation of the Certificate of Merger with the California Secretary of State and such filings with Governmental Entities to satisfy the Applicable Laws of states in which the Company and its Subsidiaries are qualified to do business; (b) such filings and approvals as may be required by any federal or state securities laws, including compliance with any applicable requirements of the Exchange Act; (c) compliance with any applicable requirements of the HSR Act and any applicable foreign antitrust, competition or merger control laws (collectively, <u>Antitrust Laws</u>); and (d) such other Consents, the failure of which to obtain would not have a Company Material Adverse Effect.

#### 3.6 Capitalization.

(a) The authorized capital stock of the Company consists of 1,000,000 shares of Company Common Stock. Exhibit A hereto sets forth a true and complete list of all Equity Interests of the Company issued or outstanding and the record and beneficial holders thereof. Except as set forth on <u>Section 3.6(a) of the Company Schedule</u>, the shares of Company Common Stock set forth on <u>Exhibit A</u> hereto are the only Equity Interests of the Company issued or outstanding and the shareholders set forth thereon are the only holders of Equity Interests of the Company. All outstanding shares of Company Common Stock are validly issued, fully paid, nonassessable and free of any preemptive rights.

(b) The Company has reserved for issuance no additional shares of Company Common Stock other than those shares of Company Common Stock issued and outstanding as of the date hereof.

(c) All of the outstanding Equity Interests of the Company have been issued in compliance in all material respects with Applicable Law and the articles of incorporation and bylaws of the Company.

(d) Except as set forth on <u>Section 3.6(d) of the Company Schedule</u>, neither the Company nor any of its Subsidiaries, nor any of the Shareholders, is a party to any Contract restricting the transfer of, relating to the voting of, requiring registration of, or granting any preemptive rights, anti-dilutive rights or rights of first refusal or similar rights with respect to any Equity Interests of the Company or any other shareholders agreement or other similar agreement restricting the election or appointment of directors of the Company.

## 3.7 Financial Statements.

(a) <u>Section 3.7(a) of the Company Schedule</u> contains (i) the unaudited consolidated balance sheets as of December 31, 2006 and 2007 and the related consolidated statements of income, shareholders equity and cash flows of the Company and its Subsidiaries for the fiscal years ended December 31, 2005, 2006 and 2007 (the most recent balance sheet set forth therein, the <u>Most Recent Company Balance Sheet</u>; and the date thereof, the <u>Most Recent Company Fiscal Year End</u>; and such financial statements collectively, the <u>Unaudited Company Financial Statements</u>); and (ii) the unaudited consolidated balance sheet of the Company and its Subsidiaries as of January 31, 2008 and the related unaudited consolidated income statements, changes in shareholders equity and cash flow of the Company and its Subsidiaries as of and for the one-month period then ended (the financial statements referred to in items (i) and (ii), and, upon the delivery of the Audited Company Financial Statements, the Audited Company Financial Statements, collectively, the <u>Company Financial Statements</u>).

(b) Except as set forth on <u>Section 3.7(b) of the Company Schedule</u>, the Company Financial Statements (i) were prepared in accordance with GAAP as in effect on the respective dates thereof applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes thereto); and (ii) fairly present in accordance with GAAP (subject, in the case of the unaudited interim financial statements, to normal, recurring year-end audit adjustments that were not or are not expected to be, individually or in the aggregate, materially adverse to the Company) the consolidated financial position of the Company and its Subsidiaries as of the respective dates thereof and the consolidated results of their operations and cash flows for the respective periods then ended.

(c) Except as set forth on <u>Section 3.7(c) of the Company Schedule</u>, the Company maintains a system of internal accounting controls sufficient to provide reasonable assurance that (i) transactions are executed in accordance with management s general or specific authorizations; (ii) access to assets is permitted only in accordance with management s general or specific authorization; and (iii) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. Except as set forth on <u>Section 3.7(c) of the Company Schedule</u>, since December 31, 2007, the Company s outside auditors and the audit committee of the board of directors of the Company have not been advised of (A) any significant deficiencies or material weaknesses in the design or operation of internal control over financial reporting which adversely affect the Company s ability to record, process, summarize and report financial information, or (B) any fraud, whether or not material, that involves management or other employees who have a significant role in the Company s internal control over financial reporting.

3.8 <u>No Undisclosed Liabilities</u>. Neither the Company nor any of the Company s Subsidiaries has any Liability that would be required by GAAP, as in effect on the date thereof, to be reflected on a consolidated balance sheet of the Company and its Subsidiaries except for (a) Liabilities reflected or reserved against in the Most Recent Company Balance Sheet; (b) Liabilities incurred in the ordinary course of business consistent with past practice since the Most Recent Company Fiscal Year End; (c) Liabilities that arose under, or were incurred in connection with the transactions contemplated by, this Agreement; or (d) Liabilities that would not have a Company Material Adverse Effect. Except as set forth on Section 3.8 of the Company Schedule, neither the Company nor any of its Subsidiaries is a party or subject to, or has any commitment to become a party to any off-balance sheet arrangements (as defined in Item 303(a) of Regulation S-K).

#### 3.9 Absence of Certain Changes.

(a) Since the Most Recent Company Fiscal Year End, except as set forth on <u>Section 3.9 of the Company Schedule</u> or except for actions expressly contemplated by this Agreement or any other Related Agreement or expressly consented to in writing by Parent from and after the date hereof,
(i) the business of the Company and its Subsidiaries has been conducted, in all material respects, in the ordinary course consistent with past practice and (ii) there has not been a Company Material Adverse Effect.

(b) Without limiting the generality of <u>Section 3.9(a)</u>, the Company and its Subsidiaries have not since the Most Recent Company Fiscal Year End and prior to the date of this Agreement taken any action, or failed to take any action, which if taken from and after the date hereof would be restricted by <u>Section 6.1</u>.

## 3.10 Compliance with Laws and Permits.

(a) Except with respect to matters that are the subjects of <u>Section 3.18</u>, <u>Section 3.19</u> and <u>Section 3.20</u> and except as set forth on <u>Section 3.10 of the Company Schedule</u>, the Company and each of its Subsidiaries is and during the last three (3) years has been in compliance with all Applicable Laws and Decrees, except for such noncompliance that would not have a Company Material Adverse Effect. Except as set forth on <u>Section 3.10 of the Company Schedule</u>, none of the Company or any of its Subsidiaries has received any written, or to the Knowledge of the Company, oral, notice (i) of any non-routine administrative, civil or criminal investigation or audit (other than Tax audits, which are addressed in <u>Section 3.19</u>) by any Governmental Entity relating to the Company or any of its Subsidiaries or (ii) from any Governmental Entity alleging that the Company or any of its Subsidiaries is not or has not been in compliance with any Applicable Law or Decree, that, if adversely determined (as to both subsections (i) and (ii) above), would have a material adverse effect on the Company.

(b) Except with respect to matters that are the subjects of <u>Section 3.18</u>, <u>Section 3.19</u> and <u>Section 3.20</u>, the Company and each of its Subsidiaries has and during the last three (3) years has had in effect all Permits necessary for it to own, lease or otherwise hold and operate its properties and assets and to carry on its businesses and operations as now conducted, and no suspension or cancellation of any such Permits is pending or, to the Knowledge of the Company, threatened in writing, except for such lack of Permits, noncompliance, suspensions or cancellations that would not have a Company Material Adverse Effect. Except with respect to matters that are the subjects of <u>Section 3.18</u>, <u>Section 3.19</u> and <u>Section 3.20</u>, there are no defaults (with or without notice or lapse of time or both) under, violations of, or events giving rise to any right of termination, amendment or cancellation of, any such Permits, except for such defaults, violations, terminations, amendments or cancellations that would not have a Company Material Adverse Effect.

3.11 Litigation. Except with respect to matters that are the subjects of Section 3.18, Section 3.19 and Section 3.20, and except as set forth in Section 3.11 of the Company Schedule, there is no Legal Proceeding pending or, to the Knowledge of the Company, threatened, (a) against the Company, any of its Subsidiaries, the Shareholder Representative, or any of their respective properties that (i) involves an amount in controversy in excess of \$1,000,000, or (ii) seeks to impose any material legal restraint on or prohibition against or limit the Surviving Entity s or any of its Affiliates ability to operate the business of the Company and its Subsidiaries substantially as it was operated immediately prior to the date of this Agreement; (b) against any current director or officer of the Company or any of its Subsidiaries (in their respective capacities as such) or, to the Knowledge of the Company, any of its Subsidiaries or any Shareholder or any of their respective properties that challenges, or that has the effect of preventing, delaying, making illegal or otherwise materially interfering with, the Merger or any of the other transactions contemplated hereby. Neither the Company nor any of its Subsidiaries is subject to any outstanding Decree that materially impairs the Company s or such Subsidiary s ability to operate in the same manner it operates on the date hereof. Section 3.11 of the Company Schedule sets forth, for each Legal Proceeding listed thereon, a brief description and the status thereof.

3.12 <u>Assets</u>. Except as set forth on <u>Section 3.12 of the Company Schedule</u>, the Company and its Subsidiaries have (i) good and valid title to all of the assets and properties (whether real, personal or mixed, or tangible or intangible) material to the operation of their business, taken as a whole (including all assets and properties recorded on the Most Recent Company Balance Sheet, other than assets and properties disposed of in the ordinary course of business since the Most Recent Company Fiscal Year End), free and clear of any Liens, other than Permitted Liens, and (ii) valid leasehold interests in, or any other valid rights under Contract to use, all of the assets and properties which the Company or any of its Subsidiaries lease or otherwise use, except where the failure to have such title or interest would not have a Company Material Adverse Effect.

## 3.13 Material Contracts.

(a) <u>Section 3.13 of the Company Schedule</u> lists, as of the date hereof, all of the Material Contracts to which the Company or any of its Subsidiaries is a party or that is otherwise binding on the Company, any of its Subsidiaries or any of the assets or property of the Company or any of its Subsidiaries (each such listed Material Contract or Material Contract otherwise required to be listed, or any Material Contract that would be required to be listed if entered into from and after the date hereof and prior to the Effective Time, a <u>Company Material Contract</u>).

(b) Except as set forth on <u>Section 3.13(b) of the Company Schedule</u>, each Company Material Contract, other than any Company Material Contract that by its terms has expired or been terminated since the date hereof, is valid and binding on the Company (and/or each Subsidiary of the Company party thereto) and, to the Knowledge of the Company, is valid and binding on each other party thereto and is in full force and effect, except as may be limited by bankruptcy, insolvency, reorganization, preference, fraudulent transfer, moratorium or similar laws relating to or affecting the rights and remedies of creditors and by general principles of equity regardless of whether considered in a proceeding in equity or at law. Except as set forth on <u>Section 3.13(b) of the Company Schedule</u>, neither the Company nor any of its Subsidiaries party thereto, nor, to the Knowledge of the Company, any other party thereto, is in breach of, or default under, any Company Material Contract, and no event has occurred that with notice or lapse of time or both would constitute such a breach or default thereunder by, the Company or any of its Subsidiaries, or, to the Knowledge of the Company, any other party thereto, or would give rise to the right to declare a default or exercise any remedy under, or to accelerate the maturity of, or to cancel, terminate or modify any Company Material Contract, except for such failures to be in full force and effect and such breaches, defaults or events that would not have a Company Material Adverse Effect.

(c) Except as set forth on <u>Section 3.13(c) of the Company Schedule</u>, the Company has made available to Parent true, accurate and complete copies of all Company Material Contracts.

#### 3.14 Government Contracts.

(a) Except as set forth on <u>Section 3.14(a) of the Company Schedule</u>, with respect to each Government Contract of the Company: (i) each of the Company and its Subsidiaries has complied in all material respects with all material terms and conditions and all Applicable Laws; (ii) no written notice has been received by either the Company or any of its Subsidiaries (and, to the Knowledge of the Company, none has been threatened) alleging that the Company, any of its Subsidiaries, or any director, officer or employee of the Company or any of its Subsidiaries, is in material breach or violation of any Applicable Law or contractual requirement thereunder; and (iii) no written notice of termination has been received by the Company or any of its Subsidiaries thereunder.

(b) Except as set forth in <u>Section 3.14(b) of the Company Schedule</u>, no Governmental Entity nor any prime contractor, subcontractor or vendor has asserted in writing any material claim or initiated any material dispute proceeding against the Company or any of its Subsidiaries relating to any Government Contract of the Company or Government Bid of the Company, nor is the Company or any of its Subsidiaries asserting any material claim or initiating any material dispute proceeding directly concerning any such Government Contract or Government Bid.

(c) With respect to each Government Contract of the Company and Government Bid of the Company, except as set forth in <u>Section 3.14(c) of</u> the Company Schedule:

(i) there are no assignments of revenues or anticipated revenues pursuant to the federal Assignment of Claims Act, 31 U.S.C. § 3727;

(ii) all written representations and certifications submitted by the Company or any of its Subsidiaries in order to induce the award of a Government Contract or to induce payments under a Government Contract were current, accurate and complete in all respects as of their respective effective dates;

(iii) material pricing discounts have been properly reported to and credited to the customer;

(iv) the Company and its Subsidiaries have maintained systems of internal controls that are and have been in material compliance with all requirements of such Government Contracts and Applicable Law;

(v) to the Knowledge of the Company, as of the date hereof, the Company and its Subsidiaries fully expect and intend to fully perform all of the obligations under (and the Company and its Subsidiaries have all material Governmental Entity authorizations and all material third-party certifications and approvals required in order to perform) each such Government Contract;

(vi) neither the Company nor any of its Subsidiaries, nor any of their respective partners, principals or officers (nor, to the Knowledge of the Company, any of its or its Subsidiaries employees, agents or consultants) have possessed or have accessed confidential or non-public information to which they were not lawfully entitled; and

(vii) the Company and each of its Subsidiaries and, to the Knowledge of the Company, each of their respective officers or directors is in compliance with Applicable Laws and other administrative and contractual restrictions associated with the employment of (or discussions concerning possible employment with) current or former officials or employees of a Governmental Entity (regardless of the branch of government), in each case, in all respects material to the Company and its Subsidiaries taken as a whole.

(d) Except as set forth in Section 3.14(d) of the Company Schedule, with respect to each Government Contract of the Company and Government Bid of the Company, neither the Company nor any of its Subsidiaries, nor, to the Knowledge of the Company, any of their respective corporate directors or officers (i) is or during the past five (5) years has been under investigation, indictment or audit by, any Governmental Entity, nor (ii) has conducted or initiated any investigation or made any disclosure to a Governmental Entity with respect to any alleged irregularity, misstatement or non-compliance, other than, in each of the cases in clause (i) or (ii), for routine audits or non-material disclosures.

(e) Neither the Company nor any of its Subsidiaries, nor, to the Knowledge of the Company, any of their respective shareholders, officers or directors, is or within the past three (3) years has been debarred, suspended or deemed non-responsible or otherwise formally excluded from participation in the award of a Government Contract of any Person, nor is there any pending debarment, suspension or exclusion proceeding that has been initiated against the Company or any of its Subsidiaries or, to the Knowledge of the Company, any of their predecessors, shareholders, officers or directors.

(f) Neither the Company nor any of its Subsidiaries, nor, to the Knowledge of the Company, any of their respective directors or officers, have (i) used any funds of the Company or its Subsidiaries to offer or provide any unlawful contribution, payment, kickback, bribe, gift, gratuity or entertainment, or (ii) to the Knowledge of the Company made any unlawful expenditures relating to political activity. Neither the Company nor any of its Subsidiaries has received written (or, to the Knowledge of the Company, oral) notice of any payment identified in (i) or (ii) above, (hereinafter referred to as an <u>Unlawful Payment</u>), and the Company and its Subsidiaries have adequate financial controls to detect and prevent any such Unlawful Payments. The Company and its Subsidiaries have been in compliance in all material respects and have, during all periods for which any applicable statute of limitations has not expired, complied with the applicable provisions of the U.S. Foreign Corrupt Practices Act, as amended, and other foreign Applicable Laws relating to corrupt practices and similar matters.

(g) <u>Section 3.14(g) of the Company Schedule</u> sets forth as of the date hereof a current, complete and accurate list of all of Government Contracts of the Company that are currently active in performance. Unless listed on <u>Section 3.14(g) of the Company Schedule</u> as being closed, each Government Contract of the Company listed on <u>Section 3.14(g) of the Company Schedule</u> is in full force and effect and constitutes a legal,

valid and binding agreement, enforceable in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, preference, fraudulent transfer, moratorium or similar laws relating to or affecting the rights and remedies of creditors and by general principles of equity regardless of whether considered in a proceeding in equity or law. No Government Contract of the Company listed on Section 3.14(g) of the Company Schedule was awarded on the basis of any qualification as a small business or other set aside or preferential prime contractor or subcontractor bidding status (collectively, a <u>Preferred Bidder Status</u>). The Company has made available to Parent complete and correct copies of those Government Contracts of the Company listed on Section 3.14(g) of the Company Schedule, together with all draft or final audit reports from a Governmental Entity or other customer as received by the Company pertaining to such Government Contracts.

(h) <u>Section 3.14(h) of the Company Schedule</u> sets forth as of the date set forth in such schedule (or if not indicated as of the date hereof) a current, accurate and complete list of each of the Government Bids which the Company or its Subsidiaries have submitted to a Governmental Entity (or a prime contractor) for which no notice of award decision has been received by the Company or its Subsidiaries.

#### 3.15 Real Property.

(a) <u>Section 3.15(a) of the Company Schedule</u> sets forth the address and a general description of all material real property owned by the Company or any of its Subsidiaries as of the date of this Agreement (the <u>Company Owned Real Property</u>). With respect to all Company Owned Real Property (other than Distributable Property):

(i) the Company or one of its Subsidiaries has good and marketable fee simple title, free and clear of all Liens, other than Permitted Liens;

(ii) except as set forth on <u>Section 3.15(a) of the Company Schedule</u> and except for Permitted Liens, neither the Company nor any of its Subsidiaries has leased or otherwise granted to any Person the right to use or occupy such Company Owned Real Property or any material portion thereof; and

(iii) there are no outstanding options, rights of first offer or rights of first refusal to purchase such Company Owned Real Property or any portion thereof or interest therein.

(b) Section 3.15(b) of the Company Schedule contains a complete and accurate list of the following:

(i) the address of each item of material Leased Real Property used or occupied by the Company or any of its Subsidiaries (the <u>Company Leased</u> <u>Real Property</u> and together with the Company Owned Real Property, the <u>Company Real Property</u>) pursuant to a Lease (the <u>Company Leases</u>), and a true and complete list of all Company Leases. The Company has made available to Parent a true and complete copy of each of the Company Leases and, in the case of any oral Company Lease, a written summary of the terms of such Company Lease; and

(ii) all Contracts or options granted by the Company or any of its Subsidiaries, or contractual obligations on the part of the Company or any of its Subsidiaries, to purchase or acquire any interest in real property material to the Company and its Subsidiaries taken as a whole.

(c) Except as set forth on <u>Section 3.15(c) of the Company Schedule</u>, with respect to each Company Lease, (i) such Lease is valid and binding on the Company (and/or each Subsidiary of the Company party thereto) and, to the Knowledge of the Company, is valid and binding on each other party thereto and is in full force and effect, except as may be limited by bankruptcy, insolvency, reorganization, preference, fraudulent transfer, moratorium or similar laws relating to or affecting the rights and remedies of creditors and by general principles of equity regardless of whether considered in a proceeding in equity or law; (ii) neither the Company nor any of its Subsidiaries party thereto, nor, to the Knowledge of the Company, any other party thereto, is in breach of, or default under, such Lease, and no event has occurred that with notice or lapse of time or both would

constitute such a breach or default thereunder by, the Company or any of its Subsidiaries, or, to the Knowledge of the Company, any other party thereto, or would give rise to the right to declare a default or exercise any remedy under, or to accelerate the maturity of, or to cancel, terminate or modify such Lease; (iii) neither the Company or any of its Subsidiaries owes, or has any obligation to owe in the future, any brokerage commissions or finder s fees with respect to such Lease; (iv) neither the Company nor any of its Subsidiaries has collaterally assigned or granted any other security interest in such Lease or any interest therein; and (v) there are no Liens, other than Permitted Liens, on the estate or interest created by such Lease, except in the case of clauses (i) through (v) as would not have a Company Material Adverse Effect.

(d) Except as would not materially interfere with the use or operation thereof, the Company Real Property has received all required approvals of Governmental Entities (including Permits and a certificate of occupancy or other similar certificate permitting lawful occupancy of the Company Real Property) required in connection with the operation thereof. Except as would not materially interfere with the use or operation of any Company Real Property, all buildings, structures, improvements, fixtures, building systems and equipment, and all components thereof, included in the Company Real Property, including leasehold improvements (the <u>Company Improvements</u>), are, to the Knowledge of the Company, (x) in good operating condition and repair, subject to ordinary wear and tear, (y) sufficient for the operation of the Company s or its Subsidiaries business as presently conducted, and (z) in conformity with all Applicable Laws.

(e) Except as set forth on <u>Section 3.15(e) of the Company Schedule</u>, neither the Company nor any of its Subsidiaries has received any written notice that it is in violation of any zoning, use, occupancy or building regulation, ordinance or other Applicable Law or Decree relating to the Company Real Property, and there is no condemnation, expropriation or other proceeding in eminent domain pending, or to the Knowledge of the Company, threatened, affecting the Company Real Property or any portion thereof or interest therein.

3.16 Intellectual Property. Section 3.16 of the Company Schedule contains a complete and accurate list of all of the following that constitutes material Company Intellectual Property (identifying for each, the owner, and, if not exclusively owned by the Company, the license pursuant to which the Company has the right to use such Intellectual Property): (i) patented or registered Intellectual Property, (ii) pending patent applications or applications for registration of Intellectual Property, (iii) all computer software (other than commercially available, off-the-shelf software with a replacement cost and/or annual license fee of less than \$100,000), and (iv) trade or corporate names and material unregistered trademarks and service marks. Except as set forth on Section 3.16 of the Company Schedule or as would not have a Company Material Adverse Effect:

(a) the Company and/or each of its Subsidiaries owns, free and clear of all Liens, other than Permitted Liens, or has a valid and enforceable license to use, all Company Intellectual Property;

(b) the Company Intellectual Property is valid, enforceable and subsisting and no loss of any Company Intellectual Property is reasonably foreseeable;

(c) the Company and each of its Subsidiaries have taken all customary actions necessary to maintain, protect, and enforce the Company Intellectual Property, including the confidentiality of its trade secrets and other confidential information;

(d) the conduct of the business of the Company or any of its Subsidiaries does not infringe, misappropriate or otherwise conflict with, and has not within the last six (6) years infringed, misappropriated or otherwise conflicted with, any Intellectual Property of any third party, and to the Knowledge of the Company, no third party is infringing, misappropriating or otherwise conflicting with, or has within the last six (6) years infringed, with, any of the Company Intellectual Property; and

(e) there are no claims against the Company or any of its Subsidiaries that were either made within the past six (6) years or are presently pending or, to the Company s Knowledge, threatened, contesting the ownership or use of any of the Company Intellectual Property.

3.17 Insurance. The Company and its Subsidiaries have all policies of insurance covering the Company, its Subsidiaries or any of their respective employees, properties or assets, which are reasonably customary for the operation of its business. All such insurance policies are in full force and effect, no notice of cancellation has been received, and there is no existing default or event which, with the giving of notice or lapse of time or both, would constitute a default, by any insured thereunder, except for such defaults that would not have a Company Material Adverse Effect. There is no material claim by the Company or any of its Subsidiaries or any Affiliate thereof pending under any of such insurance policies as to which coverage has been questioned, denied or disputed by the underwriters of such policies other than for routine matters in the ordinary course of business. Except as set forth on Section 3.17 of the Company Schedule, there is no pending claim that will exceed the policy limits under any such insurance policy. None of the Company or any of its Subsidiaries maintain, sponsor, participate in or contribute to any self-insurance plan or program.

#### 3.18 Environmental Matters. Except as set forth on Section 3.18 of the Company Schedule:

(a) Each of the Company, its Subsidiaries and their respective predecessors and Affiliates have at all times complied and are in compliance with all Environmental Laws, except for such noncompliance that would not have a Company Material Adverse Effect.

(b) Without limiting the generality of the foregoing, each of the Company, its Subsidiaries and their respective predecessors and Affiliates have obtained and at all times complied with, and are in compliance with, all Permits that are required pursuant to Environmental Law for the occupation of their facilities and the operation of their business, except for such noncompliance that would not have a Company Material Adverse Effect.

(c) Neither the Company, nor any of its Subsidiaries, nor their respective predecessors or Affiliates has received any written or, to the Company s Knowledge, oral notice, report or other information regarding any actual or alleged violation of Environmental Law, or any Liability, including any investigatory, remedial or corrective obligations, relating to any of them or their facilities or the conduct of their respective businesses arising under Environmental Law, except for such violations or Liability that would not have a Company Material Adverse Effect.

(d) Neither the Company, nor any of its Subsidiaries, nor their respective predecessors or Affiliates has treated, stored, disposed of, arranged for or permitted the disposal of, transported, handled, manufactured, distributed, exposed any person to, or Released any substance, including without limitation any Hazardous Materials, or owned, operated or conducted their business at or upon any property or facility (and no such property or facility is contaminated by any such substance) so as to give rise to any current or future material Liabilities, including any material Liability for fines, penalties, response costs, corrective action costs, personal injury, property damage, natural resources damages or attorneys fees, pursuant to the CERCLA, the Solid Waste Disposal Act or any other Environmental Law.

(e) Neither the Company nor any of its Subsidiaries has assumed, undertaken, or provided an indemnity with respect to any material Liability, including any obligation for corrective or remedial action, of any Person relating to any Environmental Law.

(f) The Company has made available to Parent true and correct copies of all environmental audits, reports and assessments, and all other documents materially bearing on environmental, health or safety liabilities, in each case relating to the past or current operations, facilities or business of the Company and its Subsidiaries, in each case which are in its possession or under its reasonable control.

#### 3.19 Tax Matters. Except as set forth on Section 3.19 of the Company Schedule:

(a) The Company and each of its Subsidiaries have filed or have caused to be timely filed all material Tax Returns required to be filed by it (the <u>Company Tax Returns</u>). All such Company Tax Returns were correct and complete in all material respects.

(b) All material Taxes due and owing by the Company and each of its Subsidiaries (whether or not shown on any Company Tax Return) have been paid or adequate reserves therefor have been established on the Most Recent Company Balance Sheet in accordance with GAAP.

(c) The Company and each of its Subsidiaries have timely withheld or collected all material Taxes that they were required to withhold or collect under Applicable Law from their employees, customers, shareholders, creditors and others from whom they are or were required to withhold Taxes and have timely paid all such withheld amounts to the appropriate taxing authorities.

(d) Neither the Company nor any of its Subsidiaries is currently the subject of an audit, judicial proceeding or other examination in respect of Taxes by the tax authorities of any nation, state or locality (and, to the Knowledge of the Company, no such audit, judicial proceeding or other examination is contemplated). The Company has made available to Parent correct and complete copies of all examination reports, closing agreements and statements of deficiencies assessed against or agreed to by the Company or any of its Subsidiaries filed or received since December 31, 2003.

(e) Neither the Company nor any of its Subsidiaries has consented in writing to extend the statutory period of limitations applicable to any claim for, or the period for the collection or assessment of, Taxes of the Company or any of its Subsidiaries due for any taxable period.

(f) Neither the Company nor any of its Subsidiaries has received written notice of any claim by any taxing authority in a jurisdiction where such Company or Subsidiary does not file Company Tax Returns that such Company or Subsidiary is or may be subject to taxation by that jurisdiction.

(g) No material Liens for Taxes exist with respect to any of the assets or properties of the Company or any of its Subsidiaries, except for Permitted Liens.

(h) Neither the Company nor any of its Subsidiaries is a party to or bound by any Tax indemnity agreement, Tax sharing agreement or Tax allocation agreement.

(i) Neither the Company nor any of its Subsidiaries is a party to any Contract which, individually or collectively with respect to any Person, could give rise to the payment of any amount that would not be deductible by the Company or any of its Subsidiaries by reason of Section 280G of the Code (or any corresponding provision of United States or non-United States federal, state and local Tax law) as a result of the transactions contemplated hereby or by any other Related Agreement.

(j) Neither the Company nor any of its Subsidiaries has engaged in any listed transaction, or any reportable transaction the principal purpose of which was tax avoidance, within the meaning of Section 6011, Section 6111 and Section 6112 of the Code.

(k) Neither the Company nor any of its Subsidiaries is, or has been, a United States real property holding company (as defined in Section 897(c)(2) of the Code) during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.

(1) In the past five (5) years, neither the Company nor any of its Subsidiaries has been either a controlled corporation or a distributing corporation (within the meaning of Section 355(a)(1)(A) of the Code) with respect to a transaction that was described in, or intended to qualify as a Tax-free transaction pursuant to Section 355 of the Code.

(m) Since December 31, 2004, each plan, program, arrangement or agreement that constitutes in any part a nonqualified deferred compensation plan within the meaning of Section 409A of the Code has been operated and maintained in accordance with the requirements of IRS Notice 2005-1 and a good faith, reasonable interpretation of Section 409A of the Code with respect to amounts deferred (within the meaning of Section 409A of the Code) after December 31, 2004.

(n) The Company has been a validly electing S corporation within the meaning of Section 1361 and 1362 of the Code (and similar provisions of state and local law) at all times since January 1, 1996, and the Company will be an S corporation up to an including the Closing Date.

(o) Section 3.19(o) of the Company Schedule identifies each Subsidiary of the Company that is a qualified subchapter S subsidiary within the meaning of Code Section 1361(b)(3)(B). Each Subsidiary of the Company so identified has been a qualified subchapter S subsidiary at all times since the date shown on such schedule up to and including the Closing Date.

3.20 Labor Matters. Except as set forth on Section 3.20 of the Company Schedule, as it relates to the Company and its Subsidiaries: (i) as of the date hereof, there is no collective bargaining agreement or relationship covering employees of the Company or any of its Subsidiaries; (ii) there are no pending or, to the Knowledge of the Company, threatened, material labor or employment controversies or disputes, including any Legal Proceeding alleging alleged unlawful harassment, employment discrimination, unfair labor practices, unpaid wages, unsafe workplace, unlawful wage or immigration practices, or unlawful tax withholding practices; (iii) there is no strike, slowdown, work stoppage, lockout or other material labor dispute underway, or threatened, and no such labor dispute has occurred within the past three (3) years; (iv) the Company and its Subsidiaries are in compliance with all laws affecting labor and employment, except for instances of noncompliance that would not, individually or in the aggregate, result in a Company Material Adverse Effect; (v) with respect to the transactions contemplated by this Agreement, all bargaining obligations with any employee representative have been or prior to Closing will be satisfied; and (vi) within the past three (3) years, neither the Company nor any of its Subsidiaries has implemented any employee layoffs in violation of the Worker Adjustment and Retraining Notification Act of 1988 or any similar Applicable Laws (collectively, the WARN Act ), and no such action will be implemented without advance notification to Parent. Except as set forth on Section 3.20 of the Company Schedule, neither the Company nor any of its Subsidiaries party thereto, nor, to the Knowledge of the Company, any other party thereto, is in breach of, or default under, any collective bargaining agreement or relationship covering employees of the Company or any of its Subsidiaries, and no event has occurred that with notice or lapse of time or both would constitute such a breach or default thereunder by, the Company or any of its Subsidiaries, or, to the Knowledge of the Company, any other party thereto, or would give rise to the right to declare a default or exercise any remedy under, or to accelerate the maturity of, or to cancel, terminate or modify any such collective bargaining contract, except for such failures to be in full force and effect and such breaches, defaults or events that would not have a Company Material Adverse Effect.

### 3.21 Employment Matters.

(a) <u>Section 3.21(a) of the Company Schedule</u> sets forth each material Company Employee Plan. Neither the Company nor any of its Subsidiaries has any stated plan, intention or commitment to establish any new Company Employee Plan, to modify any Company Employee Plan (except to the extent required by Applicable Law or to conform any such Company Employee Plan to the requirements of any Applicable Law, in each case as previously disclosed to Parent in writing), or to enter into or terminate any Company Employee Plan.

(b) The Company has made available to Parent (i) correct and complete copies of each Company Employee Plan and each summary plan description and summary of material modifications thereto; (ii) the three (3) most recent annual reports (Series 5500 and all schedules thereto), if any, required under ERISA or the Code in connection with each Company Employee Plan or related trust; (iii) if any Company Employee Plan is funded, the most recent annual and periodic accounting of Company Employee Plan assets; and (iv) each trust, insurance policy or other funding mechanism which implements each Company Employee Plan.

(c) The Company and each of its Subsidiaries has performed in all material respects all obligations required to be performed by it under each Company Employee Plan and each Company Employee Plan has been established, maintained, funded and operated in accordance with its terms, the terms of any applicable collective

bargaining agreement and in compliance with all Applicable Law, including ERISA and the Code. Each Company Employee Plan intended to qualify under Section 401(a) of the Code and each trust intended to qualify under Section 501(a) of the Code is so qualified and has either received a favorable determination letter or opinion letter from the IRS with respect to such Company Employee Plan as to its qualified status under the Code, and to the Knowledge of the Company, nothing has occurred since the date of the last such determination as to each Company Employee Plan which has resulted or is likely to result in the revocation of such determination or which requires or could require action under the compliance resolution programs of the IRS to preserve such qualification. There are no Legal Proceedings pending, or, to the Knowledge of the Company Employee Plan can be amended, terminated or otherwise discontinued after the Effective Time in accordance with its terms, without liability to the Company, any of its Subsidiaries, Parent or any of its ERISA Affiliates (other than ordinary administration expenses typically incurred in a termination event). There are no audits, inquiries investigations or proceedings pending or, to the Knowledge of the Company, threatened by the IRS, DOL or other Governmental Entity with respect to any Company Employee Plan. All annual reports and other filings required by the DOL or the IRS to be made with respect to each Company Employee Plan have been timely and completely made.

(d) Except as set forth on <u>Section 3.21(d) of the Company Schedule</u>, none of the Company, any of its Subsidiaries or ERISA Affiliates now, or has ever, maintained, established, sponsored, participated in, or contributed to, any plan that is subject to Title IV of ERISA or Section 412 of the Code, other than a Multiemployer Plan. None of the Company, any of its Subsidiaries or ERISA Affiliates has incurred, nor do they reasonably expect to incur, any liability with respect to any transaction described in Section 4069 of ERISA. Except as set forth on <u>Section 3.21(d) of the Company Schedule</u>, no Company Employee Plan is a multiple employer plan as defined in Section 210 of ERISA.

(e) None of the Company, any of its Subsidiaries or any of its ERISA Affiliates (i) has incurred any Liability on account of a partial withdrawal or complete withdrawal from any Multiemployer Plan (as described in Sections 4205 and 4203 of the Code, respectively), no such Liability has been asserted, and there are no events or circumstances which have occurred which could result in any such partial or complete withdrawal; or (ii) is bound by any contract or agreement or has any obligation or Liability under Section 4204 of ERISA.

(f) The execution and delivery by the Company of this Agreement and any other Related Agreement to which the Company is a party, and the consummation of the transactions contemplated hereby and thereby, will not conflict with or result in any violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation, modification or acceleration of any obligation or loss of any benefit under, any Company Employee Plan, trust or loan that could reasonably be expected to result in any payment (whether of severance pay or otherwise), acceleration, forgiveness of indebtedness, vesting, distribution, increase in benefits or obligation to fund benefits with respect to any Employee.

(g) No Company Employee Plan provides, or has any Liability to provide, life insurance, medical or other employee welfare benefits to any Employee upon his or her retirement or termination of employment for any reason, except as may be required by Applicable Law.

(h) All contributions and premium payments required to be made to or with respect to each Company Employee Plan prior to the Effective Time have been timely made in accordance with each such Company Employee Plan and Applicable Law and all contributions and premium payments not yet due to or with respect to each Company Employee Plan have been made or properly accrued. There have been no Prohibited Transactions with respect to any Company Employee Plan which could result in a material liability to the Company. The Company, its Subsidiaries and any ERISA Affiliates have complied with the requirements of COBRA in all material respects.

### 3.22 Interested Party Transactions.

(a) Except as set forth on <u>Section 3.22(a) of the Company Schedule</u>, to the Knowledge of the Company, no officer or director of the Company or any of its Subsidiaries (nor any ancestor, sibling, descendant or spouse of any of such Person, or any trust, partnership or corporation in which any of such Persons has an economic interest in excess of five percent (5%) of the ownership interests therein), has, directly or indirectly, (i) an economic interest in any Person that furnishes or sells, or has furnished or sold during or since January 1, 2007, services or products that the Company or any of its Subsidiaries, furnishes or sells; (ii) an economic interest in any Person that purchases from or sells or furnishes to, the Company or any of its Subsidiaries, any goods or services; or (iii) a beneficial interest in any material Contract to which the Company or any of its Subsidiaries is a party or by which they or their properties are bound; <u>provided</u>, <u>however</u>, that ownership of debt or equity interests not exceeding one percent (1%) of the outstanding voting stock of an entity shall not be deemed an economic interest in any entity for purposes of this <u>Section 3.22</u>.

(b) Except as set forth on <u>Section 3.22(b) of the Company Schedule</u>, there are no material receivables of the Company or any of its Subsidiaries owing by any officer or director of the Company or any of its Subsidiaries (or any ancestor, sibling, descendant or spouse of any of such Person, or any trust, partnership or corporation in which any of such Persons has an economic interest in excess of five percent (5%) of the ownership interests therein), other than advances in the ordinary and usual course of business for reimbursable business expenses (as determined in accordance with the Company s established employee reimbursement policies and consistent with past practice). Except with respect to the personal guarantees by the Shareholder Representative set forth in <u>Section 3.22(b) of the Company Schedule</u>, none of the Shareholders has agreed to, or assumed, any obligation or duty to guaranty or otherwise assume or incur any obligation or liability of the Company or any of its Subsidiaries.

### 3.23 Certain Payments; Suitability.

(a) None of Company nor any of its Subsidiaries, nor to the Company s Knowledge, any of their respective corporate officers or directors, acting alone or together, has performed any of the following acts: (i) the making of any contribution, payment, remuneration, bribe, rebate, influence payment, payoff, kickback, gift or other form of economic benefit, whether in money, property, or services (a <u>Payment</u>), to or for the private use of any governmental official, employee or agent where the Payment or the purpose of the Payment was in violation of Applicable Law, or (ii) the establishment or maintenance of any unrecorded fund, asset or liability for any purpose or the making of any false or artificial entries on its books or records.

(b) None of the Company or any of its Subsidiaries, nor to the Company s Knowledge, any of their respective corporate officers or directors, acting alone or together: (i) has ever been indicted for or convicted of any felony or any crime involving fraud or misrepresentation; (ii) is subject to any outstanding Decree barring, suspending or otherwise limiting the right of the Company or any of its Subsidiaries or such Person to engage in any activity conducted as part of the business of the Company or any of its Subsidiaries as currently conducted; or (iii) to the Company s Knowledge, has ever been denied any Permit affecting the Company s or any of its Subsidiary s or such Person s ability to conduct any activity conducted as part of the business of the Company or any of its Subsidiaries.

3.24 <u>Indemnification</u>. Other than pursuant to its articles of incorporation or bylaws or other comparable organizational documents, neither the Company nor any of its Subsidiaries is a party to any material indemnification agreement with any of its present or former managers, officers, directors, employees or other Persons who serve or served in any other official capacity with the Company, any Subsidiary of the Company or any other enterprise at the request of the Company or any of its Subsidiaries.

3.25 <u>Required Vote</u>. The affirmative vote of the holders of a majority of the outstanding shares of Company Common Stock, voting together or acting by written consent as a class, for the approval of this Agreement, is the only vote or action of the holders of any class or series of the Equity Interests of the Company necessary to adopt

or approve this Agreement and the transactions contemplated hereby, which shall be satisfied in full by the delivery of the Company Shareholder Written Consent.

3.26 <u>Certain Approvals; Takeover Laws</u>. The Company, including the board of directors of the Company, has taken all action required to be taken by it in order to exempt this Agreement and the other Related Agreements and the transactions contemplated hereby and thereby from, and this Agreement and the other Related Agreements are exempt from, the requirements of all applicable moratorium, control share, fair price and other anti-takeover Applicable Laws (each a <u>Takeover Law</u>).

3.27 <u>Brokers and Finders</u> Fees. Except as set forth <u>on Section 3.27 of the Company Sched</u>ule, no broker, finder, agent, investment banker, financial advisor or other Person is entitled to any broker s, finder s, agent s, investment banker s, financial advisor s or other similar fee or commission in connection with the Merger or any other transactions contemplated by this Agreement or any other Related Agreement (including for these purposes any inquiry of any sale of any Equity Interests of the Company considered in lieu thereof) based upon arrangements made by or on behalf of the Company, any of its Subsidiaries or any of the Shareholders.

3.28 <u>Ownership of Parent Equity Interests</u>. Neither the Company nor any of its Affiliates beneficially owns any Equity Interests of Parent (for avoidance of doubt, other than the Shareholders right to acquire Parent Shares pursuant to the transactions contemplated hereby and the other Related Agreements), other than, in the case of the Shareholder Representative, the beneficial ownership of any Equity Interests of Parent as a result of his employment by Parent as of the date hereof. Neither the Company nor any of its Affiliates is, or has been within the two-year period immediately prior to the date of this Agreement, an interested shareholder of Parent as those terms are defined in any applicable Takeover Law.

### ARTICLE IV

### REPRESENTATIONS AND WARRANTIES OF THE SHAREHOLDERS

Each Shareholder hereby represents and warrants to Parent and Merger Sub as of the date hereof as follows:

4.1 <u>Ownership of Subject Shares</u>. Except as set forth on <u>Section 4.1 of the Company Schedule</u>, Shareholder has good and marketable title to the Equity Interests of the Company corresponding to Shareholder on <u>Exhibit A</u> hereto (with respect to any such Shareholder, the <u>Subject Shares</u>) free and clear of any and all Liens. Except for the Subject Shares, Shareholder does not own or have any interest in any other Equity Interests of the Company or any of its Subsidiaries.

4.2 <u>Authority</u>. Except as set forth on <u>Section 4.2 of the Company Schedule</u>, Shareholder has all requisite power and authority to execute and deliver this Agreement and all other Related Agreements to which it is or is to be a party and to consummate the transactions contemplated hereby and thereby and to perform its obligations hereunder and thereunder. The execution and delivery by Shareholder of this Agreement and the other Related Agreements to which it is or is to be a party and the consummation by Shareholder of the transactions contemplated hereby and thereby have been duly authorized by all necessary action on the part of Shareholder (and for any Shareholder that is not a natural person, no additional corporate or trust proceedings on the part of Shareholder are necessary to authorize this Agreements to which Shareholder is a party has been duly executed and delivered by Shareholder and, assuming the due authorization, execution and delivery by the Company, Parent and Merger Sub and any other counterparty thereto, constitutes a legal, valid and binding obligation of Shareholder, enforceable against Shareholder in accordance with its terms, except as (a) may be limited by applicable bankruptcy, insolvency, reorganization, moratorium and other similar laws affecting or relating to creditors rights generally and (b) is subject to general principles of equity.

4.3 <u>No Conflict</u>. Except as set forth on <u>Section 4.3 of the Company Schedule</u>, none of the execution, delivery or performance by Shareholder of this Agreement or any other Related Agreements to which it is or is to be a party, the consummation by Shareholder of the transactions contemplated hereby or thereby, or the compliance by Shareholder with any of the provisions hereof or thereof: (i) for any Shareholder that is not a natural person, violate or conflict with any provision of its certificate of incorporation or bylaws or other comparable organizational documents; (ii) violate, conflict with, or result in the breach of or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or result in the termination of, or accelerate the performance required by or modification to the terms or conditions of, or result in a right of termination, acceleration or modification under, any Contract to which Shareholder is a party or by which Shareholder or any of its properties or assets may be bound; (iii) violate or conflict with any Applicable Law or Decree applicable to Shareholder or by which its properties or assets are bound; or (iv) result in the creation of any Lien upon any of the properties or assets of Shareholder.

4.4 <u>Investment</u>. Shareholder is not acquiring any Parent Shares with a view to or for sale in connection with any distribution thereof within the meaning of the Securities Act. Shareholder has been advised that the Parent Shares it will obtain as a result of the transactions contemplated hereby have not been registered under the Securities Act or any state securities laws and, therefore, cannot be resold unless they are registered under the Securities laws or unless an exemption from such registration requirements is available.

 $4.5 \underline{S Corporation Status}$ . Shareholder is an eligible S corporation shareholder under Section 1361(b)(1)(B), Section 1361(c)(2) or Section 1361(c)(6) of the Code.

4.6 <u>Accredited Investor</u>. Shareholder is an accredited investor as defined in Regulation D under the Securities Act, is experienced in evaluating companies such as Parent, is able to independently evaluate the transactions contemplated by this Agreement, has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of its investment and has the ability to bear the economic risks of its investment.

4.7 <u>Ownership of Parent Equity Interests</u>. Except as set forth on <u>Section 4.7 of the Company Schedule</u>, neither Shareholder nor any of its Affiliates beneficially owns any Equity Interests of Parent (for avoidance of doubt, other than the right to acquire Parent Shares pursuant to the transactions contemplated hereby and the other Related Agreements), other than, in the case of the Shareholder Representative, the beneficial ownership of any Equity Interests of Parent as a result of his employment by Parent as of the date hereof. Neither Shareholder nor any of its Affiliates is, or has been within the two-year period immediately prior to the date of this Agreement, an interested shareholder of Parent as those terms are defined in any applicable Takeover Law.

### ARTICLE V

### **REPRESENTATIONS AND WARRANTIES OF PARENT**

Parent hereby represents and warrants to the Company and the Shareholders as of the date hereof as follows, it being understood that such representations and warranties are made subject to any exception thereto reasonably apparent in any disclosure set forth in any Parent SEC Reports filed by Parent with, or furnished by Parent to, the SEC at any time on or prior to the date hereof:

5.1 <u>Organization and Qualification</u>. Parent is a corporation duly organized, validly existing and in good standing under the laws of the State of Massachusetts. Merger Sub is a limited liability company duly organized, validly existing and in good standing under the laws of the State of California. Each of Parent and Merger Sub has all requisite corporate or limited liability power and authority to own, lease and operate its properties and to carry on its business. Each of Parent and Merger Sub is duly qualified or licensed to do business and is in good standing as a foreign corporation in each jurisdiction in which the conduct of its business or the ownership,

leasing, holding or use of its properties makes such qualification necessary, except such other jurisdictions where the failure to be so qualified or licensed or in good standing would not have a Parent Material Adverse Effect. Each of Parent and Merger Sub has made available to the Company a true and correct copy of its articles of organization and bylaws or other comparable organizational documents, each as amended to date and in full force and effect on the date hereof.

### 5.2 Subsidiaries.

(a) <u>Section 5.2(a) of the Parent Schedule</u> sets forth a true and complete list of each Subsidiary of Parent, together with (i) the jurisdiction of formation of each such Subsidiary, (ii) the percentage of the outstanding Equity Interests of such Subsidiary owned by Parent or any of its other Subsidiaries, and (iii) if applicable, each other holder of Equity Interests of such Subsidiary. Neither Parent nor any of its Subsidiaries owns any Equity Interest in any Person other than the Subsidiaries set forth in <u>Section 5.2(a) of the Parent Schedule</u>. All Equity Interests of each Subsidiary of Parent or any of its other Subsidiaries are owned free and clear of all Liens other than Permitted Liens.

(b) Each Subsidiary of Parent is duly organized, validly existing and in good standing (to the extent applicable) under the Applicable Laws of its jurisdiction of formation. Each Subsidiary of Parent has all requisite power and authority to own, lease and operate its properties and to carry on its business. Each Subsidiary of Parent is duly qualified or licensed to do business and is in good standing (to the extent applicable) as a foreign organization in each jurisdiction in which the conduct of its business or the ownership, leasing, holding or use of its properties makes such qualification necessary, except such other jurisdictions where the failure to be so qualified or licensed or in good standing would not have a Parent Material Adverse Effect. Parent has made available to the Company a true and correct copy of the certificate of incorporation and bylaws or other comparable organizational documents of each Subsidiary of Parent, each as amended to date and in full force and effect on the date hereof.

(c) All of the outstanding share capital of each of Parent s Subsidiaries has been duly authorized and validly issued and is fully paid and non-assessable.

### (d) There are no Subsidiaries of Merger Sub.

5.3 Authority. Each of Parent and Merger Sub has all requisite power and authority to execute and deliver this Agreement and all other Related Agreements to which it is or is to be a party and subject, in the case of the consummation of the Share Issuance and the Charter Amendment, to the receipt of the Parent Shareholder Approval, to consummate the transactions contemplated hereby and thereby and to perform its obligations hereunder and thereunder. The execution and delivery by each of Parent and Merger Sub of this Agreement and the other Related Agreements to which it is or is to be a party and the consummation by each of Parent and Merger Sub of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action on the part of each of Parent and Merger Sub, and no additional corporate proceedings on the part of Parent and Merger Sub are necessary to authorize this Agreement or any other Related Agreement or the consummation of the transactions contemplated hereby or thereby, other than in the case of the consummation of the Share Issuance and the Charter Amendment, obtaining the Parent Shareholder Approval. Without limiting the generality of the foregoing, (a) the Parent Board, acting upon the recommendation of the Special Committee, has duly and validly adopted resolutions (i) determining that this Agreement and the other Related Agreements to which it is or is to be a party and the Merger, the Share Issuance, the Charter Amendment and the other transactions contemplated hereby and thereby are advisable, fair to and in the best interests of Parent and the Parent Shareholders; (ii) approving this Agreement and the other Related Agreements to which it is or is to be a party and the Merger, the Share Issuance, the Charter Amendment and the other transactions contemplated hereby and thereby; and (iii) recommending that the Parent Shareholders approve the Share Issuance and the Charter Amendment at the Parent Shareholders Meeting, which resolutions have not been subsequently rescinded, modified or withdrawn in any way; and (b) the sole member of Merger Sub has duly and validly

adopted resolutions and, simultaneously with the execution and delivery of this Agreement has executed and delivered (or immediately thereafter will execute and deliver), its written consent (i) determining that this Agreement and the other Related Agreements to which Merger Sub is or is to be a party and the Merger and the other transactions contemplated hereby and thereby are advisable, fair to and in the best interests of Merger Sub and its members; and (ii) approving this Agreement and the other Related Agreements to which it is or is to be a party and the other transactions contemplated hereby, which resolutions have not been subsequently rescinded, modified or withdrawn in any way. This Agreement and the other Related Agreements to which Parent or Merger Sub is or is to be a party have been duly executed and delivered by each of Parent and Merger Sub, as appropriate, and, assuming the due authorization, execution and delivery by the Company and the Shareholders and any other counterparty thereto, constitutes a legal, valid and binding obligation of Parent and Merger Sub, as appropriate, in accordance with its terms, except as (a) may be limited by applicable bankruptcy, insolvency, reorganization, moratorium and other similar laws affecting or relating to creditors rights generally, and (b) is subject to general principles of equity.

5.4 <u>No Conflict</u>. None of the execution, delivery or performance by Parent or Merger Sub of this Agreement or any other Related Agreements to which it is or is to be a party, the consummation by Parent or Merger Sub of the transactions contemplated hereby or thereby or the compliance by Parent or Merger Sub with any of the provisions hereof or thereof do or will: (i) violate or conflict with any provision of its certificate of incorporation or bylaws or other comparable organizational documents; (ii) violate or conflict with any provision of the certificate of incorporation or bylaws or other comparable organizational documents; (ii) violate or conflict with any provision of the certificate of incorporation or bylaws or other comparable organizational documents of any of Parent s Subsidiaries (other than Merger Sub); (iii) except as set forth in <u>Section 5.4 of the Parent Schedule</u> and subject to obtaining the Consents set forth in <u>Section 5.5 of the Parent Schedule</u>, violate, conflict with, or result in the breach of or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or result in the termination of, or accelerate the performance required by or modification to the terms or conditions of, or result in a right of termination, acceleration or modification under, any Contract to which Parent or any of its Subsidiaries is a party or by which Parent, any of its Subsidiaries or any of their properties or assets may be bound; (iv) assuming compliance with the matters referred to in <u>Section 5.5 of the Parent Schedule</u> Approval, and the other approvals referenced in <u>Section 5.3</u>, as applicable, violate or conflict with any Applicable Law or Decree applicable to Parent or any of its Subsidiaries or by which any of their properties or assets are bound; or (v) result in the creation of any Lien upon any of the properties or assets of Parent or any of the case of each of clauses (ii), (iii), (iv) and (v) above, for such violations, conflicts, breaches, defaults,

5.5 <u>Governmental Consents</u>. Except as set forth in <u>Section 5.5 of the Parent Schedule</u>, no Consent of any Governmental Entity is required on the part of Parent or any of its Subsidiaries in connection with the execution, delivery or performance by Parent or Merger Sub of this Agreement or any other Related Agreement to which it is or is to be a party or the consummation by Parent or Merger Sub of the transactions contemplated hereby or thereby, except (a) the filing and recordation of the Certificate of Merger with the California Secretary of State, the filing and recordation of the Articles of Amendment with the Massachusetts Secretary of State and such filings with Governmental Entities to satisfy the Applicable Laws of states in which Parent and its Subsidiaries are qualified to do business; (b) such filings and approvals as may be required by any federal or state securities laws, including compliance with any applicable requirements of the Exchange Act; (c) compliance with any applicable requirements of the Antitrust Laws; (d) such filings and approvals as may be required by the rules and regulations of the SEC and the NYSE; and (e) such other Consents, the failure of which to obtain would not have a Parent Material Adverse Effect.

# 5.6 Capitalization.

(a) The authorized capital stock of Parent consists of (i) 40,000,000 shares of Parent Common Stock and (ii) 1,000,000 shares of preferred stock, par value \$1.00 per share, of Parent (the <u>Parent Preferred Stock</u>). As of March 20, 2008: (A) 27,147,000 shares of Parent Common Stock were issued and outstanding, and (B) no shares of Parent Preferred Stock were issued and outstanding. All outstanding shares of Parent Common Stock are validly issued, fully paid, nonassessable and free of any preemptive rights. Since March 20, 2008, Parent has not issued any shares of Parent Capital Stock other than (i) pursuant to the exercise of options to purchase Parent Common Stock or the vesting of Parent RSUs outstanding as of March 20, 2008 or issued since that date in the ordinary course of business, (ii) pursuant to agreements existing as of the date hereof set forth in the Parent Schedule, or (iii) in the ordinary course of business in accordance with <u>Section 6.2</u>.

(b) As of March 20, 2008, Parent had reserved 2,509,506 shares of Parent Common Stock for issuance under the Parent Equity Plans. As of March 20, 2008, (i) 36,500 shares of Parent Common Stock were issuable upon the exercise of outstanding Parent Options (whether or not vested), and (ii) 925,001 shares of Parent Common Stock were issuable upon vesting of outstanding restricted stock units to acquire Parent Common Stock (<u>Parent RSUs</u>), and, since such date, Parent has not granted, committed to grant or otherwise created or assumed any obligation with respect to any Parent Options or Parent RSUs, other than as permitted by <u>Section 6.2</u>.

(c) All of the outstanding Equity Interests of Parent have been issued in compliance in all material respects with Applicable Law and the articles of organization and bylaws of Parent.

(d) Except as set forth on <u>Section 5.6 of the Parent Schedule</u>, neither Parent nor any of its Subsidiaries is a party to any Contract restricting the transfer of, relating to the voting of, requiring registration of, or granting any preemptive rights, anti-dilutive rights or rights of first refusal or similar rights with respect to any Equity Interests of Parent or any other shareholders agreement or other similar agreement restricting the election or appointment of directors of the Company.

### 5.7 SEC Reports; Financial Statements.

(a) Parent has timely filed or furnished (as applicable) all forms, reports and documents with the SEC that have been required to be so filed or furnished (as applicable) by it since January 1, 2005 under the Securities Act or the Exchange Act (all such forms, reports and documents, together with any other forms, reports or other documents filed or furnished (as applicable) by Parent with the SEC during such period, whether or not required to be so filed or furnished, the <u>Parent SEC Reports</u>). Each Parent SEC Report complied as of its filing date in all material respects with the applicable requirements of the Securities Act or the Exchange Act, as the case may be, each as in effect on the date such SEC Report was filed. As of its filing date (or, if amended or superseded by a filing, on the date of such amended or superseded filing), each Parent SEC Report to make the statements made therein, in the light of the circumstances under which they were made, not misleading. Each of the principal executive officer and the principal financial officer of Parent (or each former principal executive officer and principal financial officer of Parent, as applicable) has made all certifications required by Rule 13a-14 or 15d-14 promulgated under the Exchange Act and Sections 302 and 906 of SOX with respect to the Parent SEC Reports. For purposes of the immediately preceding sentence, principal executive officer and principal financial officer of parent SOX.

(b) The Parent SEC Reports contain the audited consolidated balance sheets and the related audited consolidated income statements, changes in shareholders equity and cash flow of Parent and its Subsidiaries as of and for the fiscal years ended December 31, 2005, 2006 and 2007 and the opinion of Deloitte & Touche LLP, Parent s independent auditor, thereon (the most recent audited balance sheet set forth therein <u>Most Recent Parent Audited Balance Sheet</u> and the date as of, the <u>Most Recent Parent Fiscal Year End</u>; the financial statements referred to in this <u>Section 5.7(b)</u>, the <u>Parent Financial Statements</u> ).

(c) The Parent Financial Statements (i) were prepared in accordance with GAAP as in effect on the respective dates thereof applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes thereto and except with respect to unaudited statements as permitted by the rules of the SEC for a Quarterly Report on Form 10-Q) and (ii) fairly present in accordance with GAAP (subject, in the case of the unaudited interim financial statements, to normal, recurring year-end audit adjustments that were not or are not expected to be, individually or in the aggregate, materially adverse to Parent) the consolidated financial position of Parent and its Subsidiaries as of the respective dates thereof and the consolidated results of their operations and cash flows for the respective periods then ended.

(d) Except as set forth on <u>Section 5.7(d) of the Parent Schedule</u>, Parent maintains a system of internal accounting controls sufficient to provide reasonable assurance that (i) transactions are executed in accordance with management s general or specific authorizations; (ii) access to assets is permitted only in accordance with management s general or specific authorization; and (iii) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. Except as set forth on <u>Section 5.7(d) of the Parent Schedule</u>, since December 31, 2007, Parent s outside auditors and the audit committee of the board of directors of Parent have not been advised of (A) any significant deficiencies or material weaknesses in the design or operation of internal control over financial reporting which adversely affect the Company s ability to record, process, summarize and report financial information, or (B) any fraud, whether or not material, that involves management or other employees who have a significant role in the Company s internal control over financial reporting.

5.8 <u>No Undisclosed Liabilities</u>. Neither Parent nor any of Parent s Subsidiaries has any Liability that would be required by GAAP, as in effect on the date thereof, to be reflected on a consolidated balance sheet of Parent and its Subsidiaries except for (a) Liabilities reflected or reserved against in the Most Recent Parent Audited Balance Sheet; (b) Liabilities incurred in the ordinary course of business consistent with past practice since the Most Recent Parent Fiscal Year End; (c) Liabilities that arose under, or were incurred in connection with the transactions contemplated by, this Agreement; or (d) Liabilities that would not have a Parent Material Adverse Effect. Except as set forth on <u>Section 5.8 of the Parent</u> <u>Schedule</u>, neither Parent nor any of its Subsidiaries is a party or subject to, or has any commitment to become a party to any off-balance sheet arrangements (as defined in Item 303(a) of Regulation S-K).

### 5.9 Absence of Certain Changes.

(a) Since the Most Recent Parent Fiscal Year End, except for actions expressly contemplated by this Agreement or any other Related Agreement, expressly consented to in writing by the Company from and after the date hereof or set forth on <u>Section 5.9 of the Parent Schedule</u>,
(i) the business of Parent and its Subsidiaries has been conducted, in all material respects, in the ordinary course consistent with past practice and
(ii) there has not been a Parent Material Adverse Effect.

(b) Without limiting the generality of <u>Section 5.9(a)</u>, Parent and its Subsidiaries have not since the Most Recent Parent Fiscal Year End and prior to the date of this Agreement taken any action, or failed to take any action, which if taken from and after the date hereof would be restricted by <u>Section 6.2</u>.

### 5.10 Compliance with Laws and Permits.

(a) Except with respect to matters that are the subjects of <u>Section 5.18</u>, <u>Section 5.19</u> and <u>Section 5.20</u>, Parent and each of its Subsidiaries is and during the last three (3) years has been in compliance with all Applicable Laws and Decrees, except for such noncompliance that would not have a Parent Material Adverse Effect. None of Parent or any of its Subsidiaries has received any written, or the Knowledge of Parent, oral, notice (i) of any non-routine administrative, civil or criminal investigation or audit (other than Tax audits, which are address in <u>Section 5.19</u>) by any Governmental Entity relating to Parent or any of its Subsidiaries or (ii) from any Governmental Entity alleging that Parent or any of its Subsidiaries is not or has not been in compliance with

any Applicable Law or Decree, that, if adversely determined (as to both subsections (i) and (ii) above), would have a material adverse effect on Parent.

(b) Except with respect to matters that are the subjects of <u>Section 5.18</u>, <u>Section 5.19</u> and <u>Section 5.20</u>, Parent and each of its Subsidiaries has and during the last three (3) years has had in effect all Permits necessary for it to own, lease or otherwise hold and operate its properties and assets and to carry on its businesses and operations as now conducted, and no suspension or cancellation of any such Permits is pending or, to the Knowledge of Parent, threatened in writing, except for such lack of Permits, noncompliance, suspensions or cancellations that would not have a Parent Material Adverse Effect. Except with respect to matters that are the subjects of <u>Section 5.18</u>, <u>Section 5.19</u> and <u>Section 5.20</u>, there are no defaults (with or without notice or lapse of time or both) under, violations of, or events giving rise to any right of termination, amendment or cancellation of, any such Permits, except for such defaults, violations, terminations, amendments or cancellations that would not have a Parent Material Adverse Effect.

5.11 Litigation. Except with respect to matters that are the subjects of Section 5.18, Section 5.19 and Section 5.20, and except as set forth in Section 5.11 of the Parent Schedule, there is no Legal Proceeding pending or, to the Knowledge of Parent, threatened, (a) against Parent, any of its Subsidiaries or any of their respective properties that (i) involves an amount in controversy in excess of \$1,000,000, or (ii) seeks to impose any material legal restraint on or prohibition against or limit the Surviving Entity s or any of its Affiliates ability to operate the business of Parent and its Subsidiaries (including the Company and its Subsidiaries from and after the Effective Time) substantially as it was operated immediately prior to the date of this Agreement; (b) against any current director or officer of Parent or any of its Subsidiaries (in their respective capacities as such) or, to the Knowledge of Parent, any former director or officer of Parent or any of its Subsidiaries (in their respective capacities as such) or, to the Knowledge of Parent, any former director or officer of Parent or any of its Subsidiaries (as such); or (c) against Parent or any of its Subsidiaries or any of their respective properties that challenges, or that has the effect of preventing, delaying, making illegal or otherwise materially interfering with, the Merger or any of the other transactions contemplated hereby. Neither Parent nor any of its Subsidiaries is subject to any outstanding Decree that materially impairs Parent s or such Subsidiary s ability to operate in the same manner it operates on the date hereof. Section 5.11 of the Parent Schedule sets forth, for each Legal Proceeding listed thereon, a brief description and the status thereof.

5.12 <u>Assets</u>. Parent and its Subsidiaries have (i) good and valid title to all of the assets and properties (whether real, personal or mixed, or tangible or intangible) material to the operation of their business, taken as a whole (including all assets and properties recorded on the Most Recent Parent Audited Balance Sheet, other than assets and properties disposed of in the ordinary course of business since the Most Recent Parent Fiscal Year End), free and clear of any Liens, other than Permitted Liens, and (ii) valid leasehold interests in, or any other valid rights under Contract to use, all of the assets and properties which Parent or any of its Subsidiaries lease or otherwise use, except where the failure to have such title or interest would not have a Parent Material Adverse Effect.

### 5.13 Material Contracts.

(a) <u>Section 5.13 of the Parent Schedule</u> lists, as of the date hereof, all of the Material Contracts to which Parent or any of its Subsidiaries is a party or that is otherwise binding on Parent, any of its Subsidiaries or any of the assets or property of Parent or any of its Subsidiaries (each such listed Material Contract or Material Contract otherwise required to be listed, or any Material Contract that would be required to be listed if entered into from and after the date hereof and prior to the Effective Time, a <u>Parent Material Contract</u>).

(b) Each Parent Material Contract, other than any Parent Material Contract that by its terms has expired or been terminated since the date hereof, is valid and binding on Parent (and/or each Subsidiary of Parent party thereto) and, to the Knowledge of Parent, is valid and binding on each other party thereto and is in full force and effect, except as may be limited by bankruptcy, insolvency, reorganization, preference, fraudulent transfer, moratorium or similar laws relating to or affecting the rights and remedies of creditors and by general principles

of equity regardless of whether considered in a proceeding in equity or at law. Neither Parent nor any of its Subsidiaries party thereto, nor, to the Knowledge of Parent, any other party thereto, is in breach of, or default under, any Parent Material Contract, and no event has occurred that with notice or lapse of time or both would constitute such a breach or default thereunder by, Parent or any of its Subsidiaries, or, to the Knowledge of Parent, any other party thereto, or would give rise to the right to declare a default or exercise any remedy under, or to accelerate the maturity of, or to cancel, terminate or modify any Parent Material Contract, except for such failures to be in full force and effect and such breaches, defaults or events that would not have a Parent Material Adverse Effect.

(c) Parent has made available to the Company true, accurate and complete copies of all Parent Material Contracts.

### 5.14 Government Contracts.

(a) Except as set forth on Section 5.14(a) of the Parent Schedule, with respect to each Government Contract of Parent: (i) each of Parent and its Subsidiaries has complied in all material respects with all material terms and conditions and all Applicable Laws; (ii) no written notice has been received by either Parent or any of its Subsidiaries (and, to the Knowledge of Parent, none has been threatened) alleging that the Parent, any of its Subsidiaries, or any director, officer or employee of Parent or any of its Subsidiaries, is in material breach or violation of any Applicable Law or contractual requirement thereunder; and (iii) no written notice of termination has been received by Parent or any of its Subsidiaries thereunder.

(b) Except as set forth in <u>Section 5.14(b) of the Parent Schedule</u>, no Governmental Entity nor any prime contractor, subcontractor or vendor has asserted in writing any material claim or initiated any material dispute proceeding against Parent or any of its Subsidiaries relating to any Government Contract of Parent or Government Bid of Parent, nor is Parent or any of its Subsidiaries asserting any material claim or initiating any material dispute proceeding any such Government Contract or Government Bid.

(c) With respect to each Government Contract of Parent and Government Bid of Parent, except as set forth in <u>Section 5.14(c) of the Parent</u> <u>Schedule</u>:

(i) there are no assignments of revenues or anticipated revenues pursuant to the federal Assignment of Claims Act, 31 U.S.C. § 3727;

(ii) all written representations and certifications submitted by Parent or any of its Subsidiaries in order to induce the award of a Government Contract or to induce payments under a Government Contract were current, accurate and complete in all respects as of their respective effective dates;

(iii) to the Knowledge of Parent, all material pricing discounts have been properly reported to and credited to the customer;

(iv) Parent and its Subsidiaries have maintained systems of internal controls that are and have been in material compliance with all requirements of such Government Contracts and Applicable Law;

(v) to the Knowledge of Parent, as of the date hereof, Parent and its Subsidiaries fully expect and intend to fully perform all of the obligations under (and Parent and its Subsidiaries have all material Governmental Entity authorizations and all material third-party certifications and approvals required in order to perform) each such Government Contract;

(vi) neither Parent nor any of its Subsidiaries, nor any of their respective partners, principals or officers (nor, to the Knowledge of Parent, any of its or its Subsidiaries employees, agents or consultants) have possessed or have accessed confidential or non-public information to which they were not lawfully entitled; and

(vii) Parent and each of its Subsidiaries and, to the Knowledge of Parent, each of their respective officers or directors is in compliance with Applicable Laws and other administrative and contractual restrictions

associated with the employment of (or discussions concerning possible employment with) current or former officials or employees of a Governmental Entity (regardless of the branch of government), in each case, in all respects material to Parent and its Subsidiaries taken as a whole.

(d) Except as set forth in Section 5.14(d) of the Parent Schedule, with respect to each Government Contract of Parent and Government Bid of Parent, neither Parent nor any of its Subsidiaries, nor, to the Knowledge of Parent, any of their respective corporate directors or officers (i) is or during the past five (5) years has been under investigation, indictment or audit by, any Governmental Entity, nor (ii) has conducted or initiated any investigation or made any disclosure to a Governmental Entity with respect to any alleged irregularity, misstatement or non-compliance, other than, in each of the cases in clause (i) or (ii), for routine audits or non-material disclosures.

(e) Neither Parent nor any of its Subsidiaries, nor, to the Knowledge of Parent, any of their respective shareholders, officers or directors, is or within the past three (3) years has been debarred, suspended or deemed non-responsible or otherwise formally excluded from participation in the award of a Government Contract of any Person, nor is there any pending debarment, suspension or exclusion proceeding that has been initiated against Parent or any of its Subsidiaries or, to the Knowledge of Parent, any of their predecessors, shareholders, officers or directors.

(f) Neither Parent nor any of its Subsidiaries, nor, to the Knowledge of Parent, any of their respective officers or directors, have (i) used any funds of Parent or its Subsidiaries to offer or provide any unlawful contribution, payment, kickback, bribe, gift, gratuity or entertainment, or (ii) to the Knowledge of Parent made any unlawful expenditures relating to political activity. Neither Parent nor any of its Subsidiaries has received written (or, to the Knowledge of Parent, oral) notice of any Unlawful Payment, and Parent and its Subsidiaries have adequate financial controls to detect and prevent any such Unlawful Payments. Parent and its Subsidiaries have been in compliance in all material respects and have, during all periods for which any applicable statute of limitations has not expired, complied with the applicable provisions of the U.S. Foreign Corrupt Practices Act, as amended, and other foreign Applicable Laws relating to corrupt practices and similar matters.

(g) <u>Section 5.14(g) of the Parent Schedule</u> sets forth as of the date hereof a current, complete and accurate list of all of Government Contracts of Parent that are currently active in performance. Unless listed on <u>Section 5.14(g) of the Parent Schedule</u> as being closed, each Government Contract of Parent listed on <u>Section 5.14(g) of the Parent Schedule</u> is in full force and effect and constitutes a legal, valid and binding agreement, enforceable in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, preference, fraudulent transfer, moratorium or similar laws relating to or affecting the rights and remedies of creditors and by general principles of equity regardless of whether considered in a proceeding in equity or law. No Government Contract of Parent listed on <u>Section 5.14(g) of the Parent Schedule</u>, together with all draft or final audit reports from a Governmental Entity or other customer as received by Parent pertaining to such Government Contracts.

(h) <u>Section 5.14(h) of the Parent Schedule</u> sets forth as of the date set forth in such schedule (or if not indicated as of the date hereof) a current, accurate and complete list of each of the Government Bids which Parent or its Subsidiaries have submitted to a Governmental Entity (or a prime contractor) for which no notice of award decision has been received by Parent or its Subsidiaries.

# 5.15 Real Property.

(a) <u>Section 5.15(a) of the Parent Schedule</u> sets forth the address and a general description of all material real property owned by Parent or any of its Subsidiaries as of the date of this Agreement (the <u>Parent Owned Real Property</u>). With respect to all Parent Owned Real Property:

(i) Parent or one of its Subsidiaries has good and marketable fee simple title, free and clear of all Liens, other than Permitted Liens;

(ii) except as set forth on <u>Section 5.15(a) of the Parent Schedule</u> and except for Permitted Liens, neither Parent nor any of its Subsidiaries has leased or otherwise granted to any Person the right to use or occupy such Parent Owned Real Property or any material portion thereof; and

(iii) there are no outstanding options, rights of first offer or rights of first refusal to purchase such Parent Owned Real Property or any portion thereof or interest therein.

(b) Section 5.15(b) of the Parent Schedule contains a complete and accurate list of the following:

(i) the address of each item of material Leased Real Property used or occupied by Parent or any of its Subsidiaries (the <u>Parent Leased Real</u> <u>Property</u> and together with the Parent Owned Real Property, the <u>Parent Real Property</u>) pursuant to a Lease (the <u>Parent Leases</u>), and a true and complete list of all Parent Leases. Parent has made available to the Company a true and complete copy of each of the Parent Leases and, in the case of any oral Parent Lease, a written summary of the terms of such Parent Lease;

(ii) all Contracts or options granted by Parent or any of its Subsidiaries, or contractual obligations on the part of Parent or any of its Subsidiaries, to purchase or acquire any interest in real property material to Parent and its Subsidiaries taken as a whole.

(c) Except as set forth on <u>Section 5.15(c) of the Parent Schedule</u>, with respect to each Parent Lease, (i) such Lease is valid and binding on Parent (and/or each Subsidiary of Parent party thereto) and, to the Knowledge of Parent, is valid and binding on each other party thereto and is in full force and effect, except as may be limited by bankruptcy, insolvency, reorganizations, preference, fraudulent transfer, moratorium or similar laws relating to or affecting the rights and remedies of creditors and by general principles of equity regardless of whether considered in a proceeding in equity or law; (ii) neither Parent nor any of its Subsidiaries party thereto, nor, to the Knowledge of Parent, any other party thereto, is in breach of, or default under, such Lease, and no event has occurred that with notice or lapse of time or both would constitute such a breach or default thereunder by, Parent or any of its Subsidiaries, or, to the Knowledge of Parent, any other party thereto, or would give rise to the right to declare a default or exercise any remedy under, or to accelerate the maturity of, or to cancel, terminate or modify such Lease; (iii) neither Parent nor any of its Subsidiaries has collaterally assigned or granted any other security interest in such Lease or any interest therein; and (v) there are no Liens, other than Permitted Liens, on the estate or interest created by such Lease, except in the case of clauses (i) through (v) as would not have a Parent Material Adverse Effect.

(d) Except as would not materially interfere with the use or operation thereof, the Parent Real Property has received all required approvals of Governmental Entities (including Permits and a certificate of occupancy or other similar certificate permitting lawful occupancy of the Parent Real Property) required in connection with the operation thereof. Except as would not materially interfere with the use or operation of any Parent Real Property, all buildings, structures, improvements, fixtures, building systems and equipment, and all components thereof, included in the Parent Real Property, including leasehold improvements (the <u>Parent Improvements</u>), are, to the Knowledge of Parent, (x) in good operating condition and repair, subject to ordinary wear and tear, (y) sufficient for the operation of Parent s or its Subsidiaries business as presently conducted, and (z) in conformity with all Applicable Laws.

(e) Except as set forth on <u>Section 5.15(e) of the Parent Schedule</u>, neither Parent nor any of its Subsidiaries has received any written notice that it is in violation of any zoning, use, occupancy or building regulation, ordinance or other Applicable Law or Decree relating to the Parent Real Property, and there is no condemnation, expropriation or other proceeding in eminent domain pending, or to the Knowledge of the Company, threatened, affecting the Parent Real Property or any portion thereof or interest therein.

5.16 Intellectual Property. Section 5.16 of the Parent Schedule contains a complete and accurate list of all of the following that constitutes material Parent Intellectual Property (identifying for each, the owner, and, if not exclusively owned by Parent, the license pursuant to which Parent has the right to use such Intellectual Property): (i) patented or registered Intellectual Property, (ii) pending patent applications or applications for registration of Intellectual Property, (iii) all computer software (other than commercially available, off-the-shelf software with a replacement cost and/or annual license fee of less than \$100,000), and (iv) trade or corporate names and material unregistered trademarks and service marks. Except as would not have a Parent Material Adverse Effect:

(a) Parent and/or each of its Subsidiaries owns, free and clear of all Liens, other than Permitted Liens, or has a valid and enforceable license to use, all Parent Intellectual Property;

(b) the Parent Intellectual Property is valid, enforceable and subsisting and no loss of any Parent Intellectual Property is reasonably foreseeable;

(c) Parent and each of its Subsidiaries have taken all customary actions necessary to maintain, protect, and enforce the Parent Intellectual Property, including the confidentiality of its trade secrets and other confidential information;

(d) the conduct of the business of Parent or any of its Subsidiaries does not infringe, misappropriate or otherwise conflict with, and has not within the last six (6) years infringed, misappropriated or otherwise conflicted with, any Intellectual Property of any third party, and to the Knowledge of Parent, no third party is infringing, misappropriating or otherwise conflicting with, or has within the last six (6) years infringed, misappropriated or otherwise conflicted with, any Intellectual Property of any third party, and to the Knowledge of Parent, no third party is infringing, misappropriating or otherwise conflicting with, or has within the last six (6) years infringed, misappropriated or otherwise conflicted with, any of the Parent Intellectual Property; and

(e) there are no claims against Parent or any of its Subsidiaries that were either made within the past six (6) years or are presently pending or, to Parent s Knowledge, threatened, contesting the ownership or use of any of the Parent Intellectual Property.

5.17 <u>Insurance</u>. Parent and its Subsidiaries have all policies of insurance covering Parent, its Subsidiaries or any of their respective employees, properties or assets, which are reasonably customary for the operation of its business. All such insurance policies are in full force and effect, no notice of cancellation has been received, and there is no existing default or event which, with the giving of notice or lapse of time or both, would constitute a default, by any insured thereunder, except for such defaults that would not have a Parent Material Adverse Effect. There is no material claim by Parent or any of its Subsidiaries or any Affiliate thereof pending under any of such insurance policies as to which coverage has been denied or disputed by the underwriters of such policies other than for routine matters in the ordinary course of business. There is no pending claim that will exceed the policy limits under any such insurance policy. None of Parent or any of its Subsidiaries maintain, sponsor, participate in or contribute to any self-insurance plan or program.

### 5.18 Environmental Matters.

(a) Each of Parent, its Subsidiaries and their respective predecessors and Affiliates have at all times complied and are in compliance with all Environmental Laws, except for such noncompliance that would not have a Parent Material Adverse Effect.

(b) Without limiting the generality of the foregoing, each of Parent, its Subsidiaries and their respective predecessors and Affiliates have obtained and at all times complied with, and are in compliance with, all Permits that are required pursuant to Environmental Law for the occupation of their facilities and the operation of their business, except for such noncompliance that would not have a Parent Material Adverse Effect.

(c) Neither Parent, nor any of its Subsidiaries, nor their respective predecessors or Affiliates has received any written or, to Parent s Knowledge, oral notice, report or other information regarding any actual or alleged violation of Environmental Law, or any Liability, including any investigatory, remedial or corrective obligations, relating to any of them or their facilities or the conduct of their respective businesses arising under Environmental Law, except for such violations or Liability that would not have a Parent Material Adverse Effect.

(d) Neither Parent, nor any of its Subsidiaries, nor their respective predecessors or Affiliates has treated, stored, disposed of, arranged for or permitted the disposal of, transported, handled, manufactured, distributed, exposed any person to, or Released any substance, including without limitation any Hazardous Materials, or owned, operated conducted their business at or upon any property or facility (and no such property or facility is contaminated by any such substance) so as to give rise to any current or future material Liabilities, including any material Liability for fines, penalties, response costs, corrective action costs, personal injury, property damage, natural resources damages or attorneys fees, pursuant to the CERCLA, the Solid Waste Disposal Act or any other Environmental Law.

(e) Neither Parent nor any of its Subsidiaries has assumed, undertaken, or provided an indemnity with respect to any material Liability, including any obligation for corrective or remedial action, of any Person relating to any Environmental Law.

(f) Parent has made available to the Company true and correct copies of all environmental audits, reports and assessments, and all other documents materially bearing on environmental, health or safety liabilities, in each case relating to the past or current operations, facilities or business of Parent and its Subsidiaries, in each case which are in its possession or under its reasonable control.

5.19 Tax Matters. Except as set forth on Section 5.19 of the Parent Schedule:

(a) Parent and each of its Subsidiaries have filed or have caused to be filed all material Tax Returns required to be filed by it (the <u>Parent Tax</u> <u>Returns</u>). All such Parent Tax Returns were correct and complete in all material respects.

(b) All material Taxes due and owing by Parent and each of its Subsidiaries (whether or not shown on any Parent Tax Return) have been paid or adequate reserves therefor have been established on the Most Recent Parent Audited Balance Sheet in accordance with GAAP.

(c) Parent and each of its Subsidiaries have timely withheld or collected all material Taxes that they were required to withhold or collect under Applicable Law from their employees, customers, shareholders, creditors and others from whom they are or were required to withhold Taxes and have timely paid all such withheld amounts to the appropriate taxing authorities.

(d) Neither Parent nor any of its Subsidiaries is currently the subject of an audit, judicial proceeding or other examination in respect of Taxes by the tax authorities of any nation, state or locality (and, to the Knowledge of Parent, no such audit, judicial proceeding or other examination is contemplated). Parent has made available to the Company correct and complete copies of all examination reports, closing agreements and statements of deficiencies assessed against or agreed to by Parent or any of its Subsidiaries filed or received since December 31, 2003.

(e) Neither Parent nor any of its Subsidiaries has consented in writing to extend the statutory period of limitations applicable to any claim for, or the period for the collection or assessment of, Taxes of Parent or any of its Subsidiaries due for any taxable period.

(f) Neither Parent nor any of its Subsidiaries has received written notice of any claim by any taxing authority in a jurisdiction where such Parent or Subsidiary does not file Parent Tax Returns that such Parent or Subsidiary is or may be subject to taxation by that jurisdiction.

(g) No material Liens for Taxes exist with respect to any of the assets or properties of Parent or any of its Subsidiaries, except for Permitted Liens.

(h) Neither Parent nor any of its Subsidiaries is a party to or bound by any Tax indemnity agreement, Tax sharing agreement or Tax allocation agreement.

(i) Neither Parent nor any of its Subsidiaries is a party to any Contract which, individually or collectively with respect to any Person, could give rise to the payment of any amount that would not be deductible by Parent or any of its Subsidiaries by reason of Section 280G of the Code (or any corresponding provision of United States or non-United States federal, state and local Tax law) as a result of the transactions contemplated hereby or by any other Related Agreement.

(j) Neither the Company nor any of its Subsidiaries has engaged in any listed transaction, or any reportable transaction the principal purpose of which was tax avoidance, within the meaning of Section 6011, Section 6111 and Section 6112 of the Code.

(k) Neither Parent nor any of its Subsidiaries is, or has been, a United States real property holding company (as defined in Section 897(c)(2) of the Code) during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.

(1) In the past five (5) years, neither Parent nor any of its Subsidiaries has been either a controlled corporation or a distributing corporation (within the meaning of Section 355(a)(1)(A) of the Code) with respect to a transaction that was described in, or intended to qualify as a Tax-free transaction pursuant to Section 355 of the Code.

(m) Since December 31, 2004, each plan, program, arrangement or agreement that constitutes in any part a nonqualified deferred compensation plan within the meaning of Section 409A of the Code has been operated and maintained in accordance with the requirements of IRS Notice 2005-1 and a good faith, reasonable interpretation of Section 409A of the Code with respect to amounts deferred (within the meaning of Section 409A of the Code) after December 31, 2004.

5.20 Labor Matters. Except as set forth on Section 5.20 of the Parent Schedule, as it relates to Parent and its Subsidiaries: (i) as of the date hereof, there is no collective bargaining agreement or relationship covering employees of Parent or any of its Subsidiaries; (ii) there are no pending or, to the Knowledge of Parent, threatened, material labor or employment controversies or disputes, including any Legal Proceeding alleging alleged unlawful harassment, employment discrimination, unfair labor practices, unpaid wages, unsafe workplace, unlawful wage or immigration practices, or unlawful tax withholding practices; (iii) there is no strike, slowdown, work stoppage, lockout or other material labor dispute underway, or threatened, and no such labor dispute has occurred within the past three (3) years; (iv) Parent and its Subsidiaries are in compliance with all laws affecting labor and employment, except for instances of noncompliance that would not, individually or in the aggregate, result in a Parent Material Adverse Effect; (v) with respect to the transactions contemplated by this Agreement, all bargaining obligations with any employee representative have been or prior to Closing will be satisfied; and (vi) within the past three (3) years, neither Parent nor any of its Subsidiaries has implemented any employee layoffs in violation of the WARN Act, and no such action will be implemented without advance notification to the Company. Neither Parent nor any of its Subsidiaries party thereto, nor, to the Knowledge of Parent, any other party thereto, is in breach of, or default under, any collective bargaining agreement or relationship covering employees of Parent or any of its Subsidiaries, and no event has occurred that with notice or lapse of time or both would constitute such a breach or default thereunder by, Parent or any of its Subsidiaries, or, to the Knowledge of Parent, any other party thereto, or would give rise to the right to declare a default or exercise any remedy under, or to accelerate the maturity of, or to cancel, terminate or modify any such collective bargaining contract, except for such failures to be in full force and effect and such breaches, defaults or events that would not have a Parent Material Adverse Effect.

### 5.21 Employment Matters.

(a) <u>Section 5.21(a) of the Parent Schedule</u> sets forth each material Parent Employee Plan. Neither Parent nor any of its Subsidiaries has any stated plan, intention or commitment to establish any new Parent Employee Plan, to modify any Parent Employee Plan (except to the extent required by Applicable Law or to conform any such Parent Employee Plan to the requirements of any Applicable Law, in each case as previously disclosed to the Company in writing), or to enter into or terminate any Parent Employee Plan.

(b) Parent has made available to the Company (i) correct and complete copies of each Parent Employee Plan and each summary plan description and summary material modification thereto; (ii) the three (3) most recent annual reports (Series 5500 and all schedules thereto), if any, required under ERISA or the Code in connection with each Parent Employee Plan or related trust; (iii) if any Parent Employee Plan is funded, the most recent annual and periodic accounting of Parent Employee Plan assets; and (iv) each trust, insurance policy or other funding mechanism which implements each Parent Employee Plan.

(c) Parent and each of its Subsidiaries has performed in all material respects all obligations required to be performed by it under each Parent Employee Plan and each Parent Employee Plan has been established, maintained, funded and operated in accordance with its terms, the terms of any applicable collective bargaining agreements and in compliance with all Applicable Law, including ERISA and the Code. Each Parent Employee Plan intended to qualify under Section 401(a) of the Code and each trust intended to qualify under Section 501(a) of the Code is so qualified and has either received a favorable determination letter or opinion letter from the IRS with respect to such Parent Employee Plan as to its qualified status under the Code, and to the Knowledge of the Company, nothing has occurred since the date of the last such determination as to each Parent Employee Plan which has resulted or is likely to result in the revocation of such determination or which requires or could require action under the compliance resolution programs of the IRS to preserve such qualification. There are no Legal Proceedings pending, or, to the Knowledge of Parent, threatened or anticipated (other than routine claims for benefits) against any Parent Employee Plan or fiduciary thereto or against the assets of any Parent Employee Plan. Each Parent Employee Plan can be amended, terminated or otherwise discontinued after the Effective Time in accordance with its terms, without liability to Parent, any of its Subsidiaries, the Company or any of its ERISA Affiliates (other than ordinary administration expenses typically incurred in a termination event). There are no audits, inquiries investigations or proceedings pending or, to the Knowledge of Parent, threatened by the IRS, DOL or other Governmental Entity with respect to any Parent Employee Plan have been timely made.

(d) None of Parent, any of its Subsidiaries or ERISA Affiliates now, or has ever, maintained, established, sponsored, participated in, or contributed to, any plan that is subject to Title IV of ERISA or Section 412 of the Code, other than Multiemployer Plans. None of Parent, any of its Subsidiaries or ERISA Affiliates has incurred, nor do they reasonably expect to incur, any liability with respect to any transaction described in Section 4069 of ERISA. Except as set forth on <u>Section 5.21(d) of the Parent Schedule</u>, no Parent Employee Plan is a multiple employer plan as defined in Section 210 of ERISA.

(e) None of the Parent, any of its Subsidiaries or any of its ERISA Affiliates (i) has incurred any Liability on account of a partial withdrawal or complete withdrawal from any Multiemployer Plan (as described in Sections 4205 and 4203 of the Code, respectively), no such Liability has been asserted, and there are no events or circumstances which have occurred which could result in any such partial or complete withdrawal; or (ii) is bound by any contract or agreement or has any obligation or Liability under Section 4204 of ERISA .

(f) All contributions and premium payments required to be made to or with respect to each Parent Employee Plan prior to the Effective Time have been timely made in accordance with each such Parent Employee Plan and Applicable Law and all contributions and premium payments not yet due to or with respect to each Parent Employee Plan have been made or properly accrued. There have been no Prohibited Transactions

with respect to any Parent Employee Plan which could result in a material liability to the Parent. The Parent, its Subsidiaries and any ERISA Affiliates have complied with the requirements of COBRA in all material respects.

(g) No Parent Employee Plan provides, or has any Liability to provide, life insurance, medical or other employee welfare benefits to any Employee upon his or her retirement or termination of employment for any reason, except as may be required by Applicable Law.

(h) The execution and delivery by Parent of this Agreement and any other Related Agreement to which Parent is a party, and the consummation of the transactions contemplated hereby and thereby, will not conflict with or result in any violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation, modification or acceleration of any obligation or loss of any benefit under, any Parent Employee Plan, trust or loan that could reasonably be expected to result in any payment (whether of severance pay or otherwise), acceleration, forgiveness of indebtedness, vesting, distribution, increase in benefits or obligation to fund benefits with respect to any Employee.

### 5.22 Interested Party Transactions.

(a) Except as set forth on Section 5.22 of the Parent Schedule, to the Knowledge of Parent, no officer or director of Parent or any of its Subsidiaries (nor any ancestor, sibling, descendant or spouse of any of such Person, or any trust, partnership or corporation in which any of such Persons has an economic interest in excess of five percent (5%) of the ownership interests therein), has, directly or indirectly, (i) an economic interest in any Person that furnishes or sells, or has furnished or sold during or since January 1, 2007, services or products that Parent or any of its Subsidiaries, furnishes or sells; (ii) an economic interest in any Person that purchases from or sells or furnishes to, Parent or any of its Subsidiaries, any goods or services; or (iii) a beneficial interest in any material Contract to which Parent or any of its Subsidiaries is a party or by which they or their properties are bound; provided, however, that ownership of debt or equity interests not exceeding one percent (1%) of the outstanding voting stock of an entity shall not be deemed an economic interest in any entity for purposes of this Section 5,22.

(b) Except as set forth on Section 5.22 of the Parent Schedule, there are no material receivables of Parent or any of its Subsidiaries owing by any officer or director of Parent or any of its Subsidiaries (or any ancestor, sibling, descendant or spouse of any of such Person, or any trust, partnership or corporation in which any of such Persons has an economic interest in excess of five percent (5%) of the ownership interests therein), other than advances in the ordinary and usual course of business for reimbursable business expenses (as determined in accordance with Parent s established employee reimbursement policies and consistent with past practice).

### 5.23 Certain Payments; Suitability.

(a) None of Parent nor any of its Subsidiaries, nor to Parent s Knowledge, any of their respective corporate officers or directors, acting alone or together, has performed any of the following acts: (i) the making of any Payment, to or for the private use of any governmental official, employee or agent where the Payment or the purpose of the Payment was in violation of Applicable Law, or (ii) the establishment or maintenance of any unrecorded fund, asset or liability for any purpose or the making of any false or artificial entries on its books or records.

(b) None of Parent or any of its Subsidiaries, nor to Parent s Knowledge, any of their respective corporate officers or directors, acting alone or together: (i) has ever been indicted for or convicted of any felony or any crime involving fraud or misrepresentation; (ii) is subject to any outstanding Decree barring, suspending or otherwise limiting the right of Parent or any of its Subsidiaries or such Person to engage in any activity conducted as part of the business of Parent or any of its Subsidiaries as currently conducted; or (iii) to Parent s Knowledge, has ever been denied any Permit affecting Parent s or any of its Subsidiary s or such Person s ability to conduct any activity conducted as part of the business of Parent or any of its Subsidiaries.

5.24 <u>Indemnification</u>. Other than pursuant to its certificate of incorporation or bylaws or other comparable organizational documents, neither Parent nor any of its Subsidiaries is a party to any material indemnification agreement with any of its present or former managers, officers, directors, employees or other Persons who serve or served in any other official capacity with Parent, any Subsidiary of Parent or any other enterprise at the request of Parent or any of its Subsidiaries.

5.25 <u>Required Vote</u>. The (a) affirmative vote of the holders of a majority of the outstanding shares of the Parent Common Stock, voting together as a class, for the approval of the Charter Amendment, and (b) the affirmative vote of a majority of all shares of Parent Common Stock casting votes (provided that the total votes cast represent over 50% in interest of all Parent Capital Stock entitled to vote) at the Parent Shareholders Meeting, for the approval of the issuance of shares of Parent Common Stock pursuant to the Merger (the <u>Share Issuance</u>; the affirmative votes contemplated by clause (a) and clause (b) hereof, the <u>Parent Shareholder Approval</u>), are the only votes or action of the holders of any class or series of the Equity Interests of Parent necessary to adopt or approve this Agreement and the transactions contemplated hereby.

5.26 <u>Certain Approvals; Takeover Laws</u>. Assuming the truth and accuracy of <u>Section 3.28</u> and <u>Section 4.7</u>, Parent, including the Parent Board, has taken all action required to be taken by it in order to exempt this Agreement and the other Related Agreements and the transactions contemplated hereby and thereby from, and this Agreement and the other Related Agreements are exempt from, the requirements of all applicable Takeover Laws.

5.27 <u>Brokers and Finders</u> Fees. Except for UBS Securities LLC, no broker, finder, agent, investment banker, financial advisor or other Person is entitled to any broker s, finder s, agent s, investment banker s, financial advisor s or other similar fee or commission in connection with the Merger or any other transactions contemplated by this Agreement or any other Related Agreement based upon arrangements made by or on behalf of Parent or any of its Subsidiaries.

5.28 <u>Opinion of Financial Advisor</u>. The Special Committee has received the opinion of its financial advisor, UBS Securities LLC, dated the date hereof, to the effect that, as of such date and subject to the matters set forth in the opinion, the Exchange Ratio provided for in the Merger is fair, from a financial point of view, to Parent, and, as of the date of this Agreement, such opinion has not been withdrawn, revoked or modified.

5.29 <u>Merger Sub</u>. Parent owns all of the issued and outstanding equity interests of Merger Sub. Merger Sub was formed solely for the purpose of engaging in the transactions contemplated by this Agreement and the other Related Agreements, and Merger Sub has not conducted any business or activity other than in connection with the Merger and the other transactions contemplated by this Agreement and the other Related Agreements. Except for obligations or liabilities incurred in connection with its incorporation and the transactions contemplated by this Agreement and the other Related Agreement and the other Related Agreements, Merger Sub has not incurred, directly or indirectly, any obligations or liabilities or engaged in any business activities of any type or kind whatsoever.

### ARTICLE VI

#### **COVENANTS**

### 6.1 Conduct of the Company and its Subsidiaries.

(a) From the date hereof until the earlier of the Effective Time or the termination of this Agreement in accordance with its terms, except as contemplated hereby, as set forth on <u>Section 6.1 of the Company Schedule</u>, as required by Applicable Law or to the extent that Parent shall otherwise consent to in writing (which consent shall not be unreasonably withheld, conditioned or delayed), the Company will and will cause its Subsidiaries to use their respective commercially reasonable efforts to, and Shareholders will cause the Company and its Subsidiaries to use their commercially reasonable efforts to, (i) operate, invest in and conduct its business in all

material respects in the ordinary course consistent with past practice; (ii) continue, in a manner consistent with the past practice, to (A) keep available the services of its respective officers, (B) pay accounts payable and collect accounts receivable, (C) comply in all material respects with Applicable Law and Permits, (D) maintain with financially responsible insurance companies insurance in such amounts and against such risks and losses as are customary for comparable companies engaged in its business, (E) maintain and preserve intact its business in all material respects, and (F) maintain in all material respects the ordinary and customary relationships of its business with its suppliers, customers, subcontractors, licensees, licensors, consultants and others having business relationships with the Company and its Subsidiaries; and (iii) prevent a material breach of any representations or warranty of the Company set forth herein.

(b) Without limiting the generality of <u>Section 6.1(a)</u>, from the date hereof until the earlier of the Effective Time or the termination of this Agreement in accordance with its terms, except as contemplated hereby, as set forth on <u>Section 6.1 of the Company Schedule</u>, as required by Applicable Law or to the extent that Parent shall otherwise consent to in writing (which consent shall not be unreasonably withheld, conditioned or delayed), the Company will not, and will cause its Subsidiaries not to, and Shareholders will cause the Company and its Subsidiaries not to:

(i) issue, deliver or sell or authorize or propose the issuance, delivery or sale of any Equity Interests;

(ii) except as contemplated by <u>Section 6.20</u>, split, combine, repurchase or reclassify any Equity Interests, declare, set aside or pay any dividend or other distribution (whether in cash, shares or property or any combination thereof) in respect of any Equity Interests (other than the payment of any dividend by a wholly-owned Subsidiary), or make any other actual, constructive or deemed distribution in respect of any Equity Interests;

(iii) except as contemplated by <u>Section 6.20</u>, (A) incur, assume or otherwise become liable for any Debt (including by way of guarantee or the issuance and sale of Debt securities or rights to acquire Debt securities) or (B) enter into or modify any Contract with respect to the foregoing (other than consents to the transactions contemplated by this Agreement), other than in each such case in the ordinary course of business consistent with past practices;

(iv) incur or commit to any material capital expenditures outside of the ordinary course of business consistent with past practices (excluding, for the avoidance of doubt, any expenditures in connection with the Company s performance of services for any third-party);

(v) cancel, compromise, release or assign any material Liability owed to it or any material claims held by it, other than in the ordinary course of business consistent with past practice;

(vi) enter into or accelerate, terminate, modify, renew or cancel any Company Material Contract other than (in the case of a Material Contract of the kind described in clause (i), (ii), (iii), (v), (vii) or (xi) of the definition thereof set forth in <u>Section 10.1</u>) in the ordinary course of business consistent with past practice;

(vii) sell, lease, transfer, license, mortgage, pledge or otherwise dispose of or encumber, except for any Permitted Liens, or grant any option to do any of the foregoing, any of its material properties or assets (other than the sale of assets or grant of such options in the ordinary course of business consistent with past practice); <u>provided</u>, <u>however</u>, that the Company may transfer the Distributable Property prior to the Closing Date provided that it does not incur any material liabilities in connection therewith;

(viii) acquire or agree to acquire by merging or consolidating with, or acquire or agree to acquire by purchasing a substantial portion of the Equity Securities or assets of, or in any other manner, any business or Person (other than acquisitions of assets with a value not greater than \$5,000,000 in the aggregate);

(ix) implement any layoffs that could implicate the WARN Act;

(x) wind-up, liquidate or dissolve, adopt a plan of complete or partial wind-up, liquidation or dissolution or resolutions providing for or authorizing such wind-up, liquidation or dissolution;

(xi) enter into, consummate, or adopt resolutions providing for or authorizing, any merger, consolidation, restructuring, recapitalization or other reorganization, other than the transactions contemplated herein;

(xii) amend its certificate of incorporation or bylaws or other comparable organizational documents;

(xiii) modify or amend any right of any holder of outstanding Equity Interests of the Company of any of its Subsidiaries;

(xiv) change its auditor or materially change its methods of accounting in effect as of the date hereof except as required by changes in GAAP or Applicable Law or by a Governmental Entity;

(xv) make, change, or rescind any Tax election, change any accounting period, adopt or change any accounting method for Tax purposes, file any amended Tax Return, enter into any closing agreement, settle or compromise any Tax claim or assessment relating to any of the Company or its Subsidiaries, surrender any right to claim a refund of Taxes, consent to any extension or waiver of the limitation period applicable to any Tax claim or assessment relating to any of the Company or its Subsidiaries, or take any other similar action relating to the filing of any Tax Return or the payment of any Tax, if, in all cases, such election, adoption, change, amendment, agreement, settlement, surrender, consent or other action would have the effect of increasing the Tax liability of any of the Company or its Subsidiaries for any period ending after the Closing Date or decreasing any Tax attribute of any of the Company or its Subsidiaries existing on the Closing Date by a material amount;

(xvi) take any action or make or revoke any election that would result in the termination of the Company s status as a validly electing S corporation within the meaning of Section 1361 and 1362 of the Code;

(xvii) (A) settle or compromise any Legal Proceeding on terms which require it or any of its Subsidiaries to pay, incur or assume any Liability in excess of \$5,000,000;

(xviii) (A) enter into, amend, modify or renew any Contract regarding employment, consulting, severance or similar arrangements with any of its directors, officers, employees or consultants, or grant any salary, wage or other increase in compensation or increase any employee benefit (including incentive, profit sharing or bonus payments), other than, in the case of employees or consultants, in the ordinary course of business consistent with past practice; (B) grant any severance or termination pay or benefit to any directors, officers, employees or consultants except pursuant to written agreements or severance policies in effect of the date hereof and disclosed in the applicable Company Schedules, other than, in the case of employees or consultants, other than in the ordinary course of business consistent with past practice not related to the transactions contemplated hereby; or (C) adopt or amend any Company Employee Plan except to the extent required by Applicable Law or to the extent the adoption or amendment of such Company Employment Plan does not materially increase the aggregate cost of the Company s Employee Plans;

(xix) except as disclosed in the Company Schedule, enter into or modify any transaction or arrangement with, or for the benefit of any of its Affiliates or any of its or their directors, former directors, officers or shareholders; or

(xx) agree in writing or otherwise to take any of the foregoing actions.

### 6.2 Conduct of Parent and its Subsidiaries.

(a) From the date hereof until the earlier of the Effective Time or the termination of this Agreement in accordance with its terms, except as contemplated hereby, as set forth on <u>Section 6.2 of the Parent Schedule</u>, as required by Applicable Law or to the extent that the Company shall otherwise consent to in writing (which consent shall not be unreasonably withheld, conditioned or delayed), Parent will and will cause its Subsidiaries to use their respective commercially reasonable efforts to, (i) operate, invest in and conduct its business in all material respects in the ordinary course consistent with past practice; (ii) continue, in a manner consistent with the past practice, to (A) keep available the services of its respective officers, (B) pay accounts payable and collect accounts receivable, (C) comply in all material respects with Applicable Law and Permits, (D) maintain with financially responsible insurance companies insurance in such amounts and against such risks and losses as are consistent with past practice or customary for comparable companies engaged in its business, (E) maintain and preserve intact its business in all material respects, and (F) maintain in all material respects the ordinary and customary relationships of its business with its suppliers, customers, subcontractors, licensees, licensors, consultants and others having business relationships with Parent and its Subsidiaries; and (iii) prevent a material breach of any representations or warranty of Parent set forth herein.

(b) Without limiting the generality of <u>Section 6.2(a)</u>, from the date hereof until the earlier of the Effective Time or the termination of this Agreement in accordance with its terms, except as contemplated hereby, as set forth on <u>Section 6.2 of the Parent Schedule</u>, as required by Applicable Law or to the extent that the Company shall otherwise consent to in writing (which consent shall not be unreasonably withheld, conditioned or delayed), Parent will not, and will cause its Subsidiaries not to:

(i) issue, deliver or sell or authorize or propose the issuance, delivery or sale of any Equity Interests other than in the ordinary course consistent with past practice or pursuant to issued and outstanding Equity Interests;

(ii) split, combine, repurchase or reclassify any Equity Interests, declare, set aside or pay any dividend or other distribution (whether in cash, shares or property or any combination thereof) in respect of any Equity Interests (other than the payment of any dividend by a wholly-owned Subsidiary), or make any other actual, constructive or deemed distribution in respect of any Equity Interests;

(iii) acquire or agree to acquire by merging or consolidating with, or acquire or agree to acquire by purchasing a substantial portion of the Equity Securities or assets of, or in any other manner, any business or Person (other than acquisitions of assets with a value not greater than \$5,000,000 in the aggregate);

(iv) wind-up, liquidate or dissolve, adopt a plan of complete or partial wind-up, liquidation or dissolution or resolutions providing for or authorizing such wind-up, liquidation or dissolution;

(v) enter into, consummate, or adopt resolutions providing for or authorizing, any merger, consolidation, restructuring, recapitalization or other reorganization, other than the transactions contemplated herein;

(vi) amend its articles of organization or bylaws or other comparable organizational documents;

(vii) make, change, or rescind any Tax election, change any accounting period, adopt or change any accounting method for Tax purposes, file any amended Tax Return, enter into any closing agreement, settle or compromise any Tax claim or assessment relating to any of Parent or its Subsidiaries, surrender any right to claim a refund of Taxes, consent to any extension or waiver of the limitation period applicable to any Tax claim or assessment relating to any of Parent or its Subsidiaries, or take any other similar action relating to the filing of any Tax Return or the payment of any Tax, if, in all cases, such election, adoption, change, amendment, agreement, settlement, surrender, consent or other action would have the effect of increasing the Tax liability of

any of Parent or its Subsidiaries for any period ending after the Closing Date or decreasing any Tax attribute of any of Parent or its Subsidiaries existing on the Closing Date by a material amount;

(viii) modify or amend any right of any holder of outstanding Equity Interests of Parent or Merger Sub; or

(ix) agree in writing or otherwise to take any of the foregoing actions.

6.3 No Solicitation by the Company or Shareholders; Shareholder Lock-Up.

(a) <u>No Solicitation</u>. At all times during the period commencing with the execution and delivery of this Agreement and continuing until the earlier to occur of the termination of this Agreement or the Effective Time, neither the Company nor any Shareholder shall, and the Company and each Shareholder shall use its reasonable best efforts to cause its Representatives not to:

(i) solicit, initiate, knowingly encourage, or knowingly induce the making, submission or announcement of, an Acquisition Proposal with respect to the Company;

(ii) furnish to any Person any non-public information relating to the Company or its Subsidiaries in connection with an Acquisition Proposal with respect to the Company (for avoidance of doubt, it being understood that the foregoing shall not prohibit the Company or any of its Representatives from furnishing any non-public information to any actual or potential customer, supplier, distributor, subcontractor, licensor, licensee, partner or other Person with which it has or may have business dealings in the ordinary course of business consistent with past practice (solely in such capacity) or to a Governmental Entity);

(iii) participate or engage in discussions or negotiations with any Person with respect to an Acquisition Proposal with respect to the Company (for avoidance of doubt, it being understood that the foregoing shall not prohibit the Company or any of its Representatives from making a Person making an Acquisition Proposal with respect to the Company aware of the restrictions of this <u>Section 6.3(a)(iii)</u> in response to the receipt of such an Acquisition Proposal); or

(iv) enter into any agreement, letter of intent or other similar instrument with any Person providing for an Acquisition Transaction with respect to the Company.

(b) <u>Notice of Acquisition Proposal</u>. In addition to the obligations of the Company set forth in <u>Section 6.3(a)</u>, the Company and each Shareholder shall promptly, and in all cases within one (1) Business Day of its receipt, advise Parent and the Special Committee orally (to be confirmed in writing as soon as practicable thereafter) of the receipt by the Company or such Shareholder, as appropriate, prior to the Effective Time or termination of this Agreement of (i) any Acquisition Proposal with respect to the Company, or (ii) any request for information or inquiry that is reasonably to be expected to lead to an Acquisition Proposal with respect to the Company, including the material terms and conditions of such Acquisition Proposal, request or inquiry, and in either case any developments with respect thereto.

(c) <u>Company Shareholder Written Consent</u>. The Shareholders shall cause the Company Shareholder Written Consent substantially in the form of the draft thereof previously provided to Parent to be executed and delivered to the Company and Parent simultaneously with or immediately after the execution and delivery of this Agreement.

(d) Shareholder Lock-Up. From the date hereof until the earlier of the Effective Time or the termination of this Agreement:

(i) <u>No Transfer of Equity Interests of the Company</u>. Each Shareholder agrees that it shall not transfer, sell, offer, exchange, pledge or otherwise dispose of or encumber any of its Equity Interests of the Company (except pursuant to the transactions contemplated hereby and the other Related Agreements).

(ii) <u>No Voting Trusts</u>. Each Shareholder agrees that it will not, and will not permit any Person under Shareholder s control to, deposit any of its Equity Interests of the Company in a voting trust, grant any proxies with respect to its Equity Interests of the Company or subject any of its Equity Interests of the Company to any arrangement with respect to the voting of its Equity Interests of the Company (except pursuant to the transactions contemplated hereby and the other Related Agreements).

(iii) <u>No Proxy Solicitations</u>. Each Shareholder agrees that it shall not, and shall not permit any Person under Shareholder s control to, (a) solicit proxies or become a participant in a solicitation (as such terms are defined in Regulation 14A under the Exchange Act) in opposition to or competition with the consummation of any of the transactions contemplated hereby or otherwise encourage or assist any party in taking or planning any action which could compete with, impede, interfere with or attempt to discourage any of the transactions contemplated hereby or inhibit the timely consummation of any of the transactions contemplated hereby in accordance with the terms hereof; (b) directly or indirectly encourage, initiate or cooperate in a shareholders vote or action by consent of the Parent s shareholders in opposition to or in competition with the consummation of any of the transactions contemplated hereby; or (c) become a member of a group (as such term is used in Rule 13d-5 under the Exchange Act) with respect to any voting securities of Parent for the purpose of opposing or competing with the consummation of any of the transactions contemplated hereby.

(iv) <u>No Acquisition of Parent Equity Interests</u>. Each Shareholder agrees that it shall not acquire, offer to acquire, or agree to acquire, directly or indirectly, by purchase or otherwise, Equity Interests of Parent (except pursuant to the transactions contemplated hereby and the other Related Agreements or other agreements or employment arrangements in effect as of the date hereof).

### 6.4 No Solicitation by Parent.

(a) <u>No Solicitation</u>. At all times during the period commencing with the execution and delivery of this Agreement and continuing until the earlier to occur of the termination of this Agreement or the Effective Time, Parent shall not, and shall use its reasonable best efforts to cause its Representatives not to:

(i) solicit, initiate, knowingly encourage, or knowingly induce the making, submission or announcement of, an Acquisition Proposal with respect to Parent;

(ii) furnish to any Person any non-public information relating to Parent or its Subsidiaries in connection with an Acquisition Proposal with respect to Parent (for avoidance of doubt, it being understood that the foregoing shall not prohibit Parent or any of its Representatives from furnishing any non-public information to any actual or potential customer, supplier, distributor, subcontractor, licensor, licensee, partner or other Person with which it has or may have business dealings in the ordinary course of business consistent with past practice (solely in such capacity) or to a Governmental Entity);

(iii) participate or engage in discussions or negotiations with any Person with respect to an Acquisition Proposal with respect to Parent (for avoidance of doubt, it being understood that the foregoing shall not prohibit Parent or any of its Representatives from making a Person making an Acquisition Proposal with respect to Parent aware of the restrictions of this <u>Section 6.4(a)(iii)</u> in response to the receipt of such an Acquisition Proposal); or

(iv) enter into any agreement, letter of intent or other similar instrument with any Person providing for an Acquisition Transaction with respect to Parent except as expressly permitted by this Agreement;

provided, however, that notwithstanding the foregoing, at any time prior to the Effective Time or termination of this Agreement, the Parent Board (or the Special Committee on its behalf) may, directly or indirectly through Representatives, (A) engage or participate in discussions or negotiations with any Person that has made a *bona fide* Acquisition Proposal with respect to Parent that the Parent Board or the Special Committee reasonably determines in good faith (after consultation with a financial advisor of nationally recognized standing and its

outside legal counsel) constitutes or is reasonably likely to lead to a Superior Proposal and/or (B) furnish to any such Person any non-public information relating to Parent or any of its Subsidiaries pursuant to a confidentiality agreement the terms of which, taken as a whole, are no less favorable to Parent than those contained in the Confidentiality Agreement; provided that in the case of any action taken pursuant to the foregoing clauses (A) or (B), (x) the Parent Board or the Special Committee received such Acquisition Proposal other than as a result of a material breach or violation of the terms of this Section 6.4(a); (y) at least forty-eight (48) hours prior to engaging or participating in any such discussions or negotiations with, or furnishing any non-public information to, such Person, Parent gives the Company written notice of its receipt of such Acquisition Proposal and of Parent s (or the Special Committee s) intention to engage or participate in discussions or negotiations with, or furnishing and of Parent s (or the Special Committee s) intention to engage or participate in discussions or negotiations with, or furnish non-public information to, such Person; and (z) contemporaneously with furnishing any non-public information to such Person, Parent or the Special Committee furnishes such non-public information to the Company (to the extent such information has not been previously made available to the Company or any of its Representatives).

(b) <u>Notice of Acquisition Proposal</u>. In addition to the obligations of Parent set forth in <u>Section 6.4(a)</u>, Parent or the Special Committee shall promptly, and in all cases within one (1) Business Day of its receipt, advise the Company orally (to be confirmed in writing as soon as practicable thereafter) of the receipt by Parent or the Special Committee prior to the Effective Time or termination of this Agreement of (i) any Acquisition Proposal with respect to Parent, or (ii) any request for information or inquiry that is reasonably to be expected to lead to an Acquisition Proposal with respect to Parent, including the material terms and conditions of such Acquisition Proposal, request or inquiry. Parent or the Special Committee shall keep the Company informed on a reasonably current basis of the status and material terms and conditions (including all amendments or proposed amendments) of any such Acquisition Proposal, request or inquiry.

### 6.5 Parent Board Recommendation.

(a) Except as contemplated by <u>Section 6.5(b)</u> or <u>Section 6.5(c)</u>, neither the Parent Board nor any committee thereof (including the Special Committee) shall (i) (A) withdraw or modify, in a manner adverse to the Company, the recommendation by the Parent Board in favor of the Parent Shareholder Approval, or (B) recommend that the Parent Shareholders reject the approval of the Share Issuance or the approval of the Charter Amendment; (ii) determine that the Share Issuance or the Charter Amendment is no longer advisable; (iii) approve or adopt, or resolve, agree or propose publicly to approve or adopt, or recommend the approval or adoption by the Parent Shareholders of, any Acquisition Proposal with respect to Parent; (iv) cause or permit Parent to enter into any Contract providing for an Acquisition Transaction with respect to Parent (an <u>Acquisition Agreement</u>) (for avoidance of doubt other than a confidentiality agreement referred to in, and in accordance with, <u>Section 6.4(a)</u>); or (v) resolve, agree or propose publicly to take any of the foregoing actions (any action set forth in this <u>Section 6.5(a)</u> being referred to herein as a <u>Parent Change of Recommendation</u>).

(b) Notwithstanding anything to the contrary set forth herein, at any time prior to the receipt of the Parent Shareholder Approval (the <u>Applicable Period</u>), the Parent Board or any committee thereof (including the Special Committee) may make a Parent Change of Recommendation, (i) if it determines in good faith, after consulting with outside counsel, that the failure to take such action would be reasonably likely to result in a breach of the fiduciary obligations of the Parent Board or such committee thereof to the Parent Shareholders under Applicable Law and (ii) if such Parent Change of Recommendation is or is being made to lead to the actions contemplated by clause (iii) or clause (iv) of <u>Section 6.5(a)</u> (a <u>Qualifying Parent Change of Recommendation</u>), (A) the Acquisition Proposal giving rise thereto shall constitute a Superior Proposal and (B) the Parent Board or such committee thereof shall have provided to the Company and the Shareholders three (3) Business Days prior written notice of its intent to effect the Qualifying Parent Change of Recommendation) and, if requested by the Company or the Shareholders, negotiated in good faith with the Company and the Shareholders during such three (3) Business Day period regarding revisions to this Agreement that would avoid such Qualifying Parent Change of Recommendation; provided, however, that it is understood

and agreed for avoidance of doubt that Parent may, simultaneously with the termination of this Agreement pursuant to <u>Section 8.1(g)</u>, enter into an Acquisition Agreement or otherwise accept an Acquisition Proposal providing for an Acquisition Transaction.

(c) Nothing in this Agreement shall prohibit the Parent Board or any committee thereof from (i) taking and disclosing to the Parent Shareholders a position contemplated by Rule 14e-2(a) promulgated under the Exchange Act or complying with the provisions of Rule 14d-9 promulgated under the Exchange Act or (ii) making any disclosures to the Parent Shareholders that the Parent Board or such committee thereof determines in good faith (after consultation with outside legal counsel) that the Parent Board or such committee thereof is required to make in order to comply with its fiduciary duties to the Parent Shareholders under Applicable Law.

### 6.6 Proxy Statement; Parent Shareholders Meeting.

(a) <u>Proxy Statement; Cooperation</u>. As soon as reasonably practicable after the date hereof, Parent shall prepare and shall cause to be filed with the SEC a proxy statement relating to the Parent Shareholders Meeting (together with any amendments thereof or supplements thereto, the <u>Proxy Statement</u>) in preliminary form. The Parties shall cooperate and consult with each other in preparation of the Proxy Statement. Without limiting the generality of the foregoing, no filing of, or amendment or supplement to, the Proxy Statement shall be made by Parent without providing the Company a reasonable opportunity to review and comment thereon. Parent shall advise the Company promptly after it receives notice of any request by the SEC or its staff for an amendment or revisions to the Proxy Statement, or comments thereon and responses thereto, or requests by the SEC or its staff for additional information in connection therewith. Parent shall use commercially reasonable efforts to respond promptly to any comments by the SEC staff in respect of the Proxy Statement. Parent shall cause the Proxy Statement to comply as to form in all material respects with the provisions of the Exchange Act and the Securities Act and to satisfy all rules of the NYSE with respect thereto.

(b) <u>Information in the Proxy Statement</u>. Parent will cause the Proxy Statement, at the time of the mailing thereof and at the time of the Parent Shareholders Meeting, to not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading, it being understood that Parent s obligations in such regard shall not extend to information supplied by the Company or the Shareholders for inclusion in the Proxy Statement. The Company and Shareholders will cause the information relating to the Company or any of the Shareholders supplied by it or them for inclusion in the Proxy Statement, at the time of the mailing thereof and at the time of the Parent Shareholders Meeting, not to contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The Company and Shareholders shall each furnish to Parent any information relating to it required by the Exchange Act and the rules and regulations promulgated thereunder to be set forth in the Proxy Statement or requested by the SEC and shall promptly inform Parent, and correct as promptly as practicable, any information previously provided by it for use in the Proxy Statement which shall have become false or misleading.

(c) <u>Audited Financial Statements</u>. The Company (i) shall use reasonable best efforts to deliver to Parent within thirty (30) days after the date of this Agreement historical unaudited and, to the extent required, audited financial statements of the Company and its consolidated Subsidiaries necessary for the Proxy Statement) and shall cooperate with Parent in connection with the preparation of related pro forma financial statements, in each case that comply with either (A) the requirements of Regulation S-X (as interpreted by the staff of the SEC) for financial statements that would be required to be included in a Definitive Proxy Statement filed pursuant to Regulation 14A of the Exchange Act, or (B) the requirements set forth in clause (A) except as the staff of the SEC may permit Parent by waiver of such requirements, (ii) shall take such actions as are reasonably necessary to cause any financial statements contemplated by clause (i) covering any interim periods to be subject to SAS 100 review by the Company s auditors, and (iii) shall provide and make reasonably available upon reasonable

notice the senior management employees of the Company to discuss the materials prepared and delivered pursuant to this Section 6.6(c).

(d) <u>Mailing of Proxy Statement: Amendments</u>. As promptly as reasonably practicable after the Proxy Statement has been cleared by the SEC, Parent shall mail to the Parent Shareholders as of the record date established for the Parent Shareholders Meeting the Proxy Statement contained in the Proxy Statement. If at any time prior to the Effective Time any fact, event or circumstance relating to a Party should become known by a Party that is required to be forth in an amendment or a supplement to the Proxy Statement pursuant to any Applicable Law, such Party shall promptly inform the other Parties.

(e) <u>Parent Shareholders Meeting</u>. Parent shall take all action necessary under Applicable Law to call, give notice of and hold a meeting of the holders of the Parent Common Stock (the <u>Parent Shareholders</u>) to vote on the proposals to obtain the Parent Shareholder Approval (including any adjournment thereof, the <u>Parent Shareholders Meeting</u>), and shall submit such proposals to such holders at the Parent Shareholders Meeting. Parent shall ensure that all proxies solicited in connection with the Parent Shareholders Meeting are solicited in compliance with Applicable Law. Parent s obligation to call, give notice of and hold the Parent Shareholders Meeting in accordance with this <u>Section 6.6(e)</u> shall not be limited or otherwise affected by the commencement, disclosure, announcement or submission of any Superior Proposal or other Acquisition Proposal, or by any withdrawal or modification of the recommendation by the Parent Board in favor of the Parent Shareholder Approval as provided in <u>Section 6.5(b)</u> unless this Agreement shall have been terminated.

### 6.7 Obligations to Consummate the Merger.

(a) <u>Efforts</u>. Upon the terms and subject to the conditions set forth in this Agreement (including <u>Section 6.5(b)</u>), each of the Parties shall use its reasonable best efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other Parties in doing, all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the Merger and the other transactions contemplated hereby. Without limiting the generality of the foregoing:

(i) (A) Parent and Merger Sub shall each use its reasonable best efforts to cause the conditions set forth in <u>Section 7.1</u> and <u>Section 7.3</u> to be satisfied or fulfilled, and (B) the Company and each Shareholder shall each use its reasonable best efforts to cause the conditions set forth in <u>Section 7.1</u> and <u>Section 7.2</u> to be satisfied or fulfilled;

(ii) each Party shall use its commercially reasonable efforts to obtain all necessary, proper or advisable Consents under any Contracts or Permits to which a Party is a party in connection with this Agreement and the other Related Agreements and the consummation of the transactions contemplated hereby and thereby so as to maintain and preserve the benefits under such Contracts and Permits following the consummation of the transactions contemplated hereby and thereby; provided, however, that except as expressly provided in <u>Article VII</u>, no such consents shall constitute conditions to Closing;

(iii) each Party shall use its reasonable best efforts to obtain all necessary Consents and other actions or non-actions and Decrees from Governmental Entities, seek the expiration or termination of all applicable waiting periods under Antitrust Laws, and make all necessary registrations, declarations and filings with Governmental Entities necessary to consummate and make effective the Merger and the other transactions contemplated hereby; <u>provided</u>, <u>however</u>, that except as expressly provided in <u>Article VII</u>, no such consents, actions, non-actions or Decrees shall constitute conditions to Closing; and

(iv) each Party shall execute or deliver any additional instruments reasonably necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement.

(b) <u>Governmental Entities</u>. Without limiting the generality of <u>Section 6.7(a)</u>, each of Parent and the Company shall file, as soon as may be reasonably practicable (and in any event within ten (10) Business Days)

following the execution and delivery of this Agreement, with the United States Federal Trade Commission (the <u>FTC</u>) and the Antitrust Division of the United States Department of Justice (the <u>DOJ</u>) a premerger notification and report form relating to this Agreement and the transactions contemplated hereby as required by the Hart-Scott-Rodino Antitrust Improvements Act of 1976 (the <u>HSR Act</u>) and, as soon as may be reasonably practical following the execution and delivery of this Agreement, any comparable pre-merger notification filings, forms and submissions with any foreign Governmental Entity that is required by the merger notification or control laws and regulations of any applicable foreign jurisdiction. Each Party shall promptly (i) cooperate and coordinate with the other in the making of such filings; (ii) supply the other with any information that may be required in order to effectuate such filings; and (iii) supply any additional information that reasonably may be required or requested by the FTC, the DOJ or the competition or merger control authorities of any other jurisdiction. Each Party shall promptly inform the other Parties of any communication from any Governmental Entity regarding any of the transactions contemplated by this Agreement or any other Related Agreement. If any Party or Representative thereof receives a request for additional information or documentary material from any such Governmental Entity with respect to the transactions contemplated by this Agreement, then such Party shall use reasonable best efforts to make, or cause to be made, as soon as reasonably practicable and after consultation with the other Parties, an appropriate response in compliance with such request.

(c) <u>Takeover Laws</u>. Without limiting the generality of <u>Section 6.7(a)</u>, in the event that any Takeover Law is or becomes applicable to this Agreement or any other Related Agreement or any of the transactions contemplated hereby or thereby, each Party hereto shall take all actions to ensure that the transactions contemplated hereby may be consummated as promptly as practicable on the terms and subject to the conditions set forth in this Agreement, and otherwise to minimize the effect of such statute or regulation on this Agreement and the other Related Agreements and the transactions contemplated hereby or thereby.

### 6.8 Access.

(a) <u>Access</u>. Except as required pursuant to any confidentiality obligation to which the Company or Parent is a party as of the date hereof, from the date hereof until the earlier to occur of the termination of this Agreement or the Effective Time, each of the Company and Parent shall afford the other Party and its Representatives reasonable access during normal business hours, upon reasonable notice, to the properties, Contracts, books and records and corporate officers of such Party and its Subsidiaries to enable the other Party to obtain all information concerning its business, properties, results of operations and personnel as such other Party may reasonably request; <u>provided</u>, <u>however</u>, that neither Party shall be required to provide any Person information or access if (x) providing such information or access would reasonably be expected to result in a waiver of the attorney/client privilege; (y) providing such information or access would reasonably be expected to result in a violation of Applicable Law, including any Antitrust Laws; or (z) is deemed in good faith by such Party to be competitively sensitive.

(b) <u>Effect of Review</u>. No review pursuant to this <u>Section 6.8</u> shall have an effect for the purpose of determining the accuracy of any representation or warranty given by any Party hereunder.

(c) <u>Confidentiality</u>. All information obtained pursuant to this <u>Section 6.8</u> shall be Evaluation Material (as such term is defined in the Confidentiality Agreement, dated as of January 25, 2008 (the <u>Confidentiality Agreement</u>), by and between the Company and Parent) subject to the terms and conditions of the Confidentiality Agreement.

6.9 <u>Notices of Certain Events</u>. From the date hereof until the earlier to occur of the termination of this Agreement or the Effective Time, each of Parent and the Company shall promptly notify the other of:

(a) the existence of any fact or circumstance, or the occurrence of any event, of which it has Knowledge which would be reasonably likely to cause a condition to a Party s obligations to consummate the transactions contemplated hereby set forth in <u>Article VII</u> not to be satisfied as of a reasonably foreseeable Closing Date,

(b) any notice or other communication from any Person alleging that a material Consent of such Person is or may be required in connection with any of the transactions contemplated hereby;

(c) any notice or other communication from any Governmental Entity in connection with the transactions contemplated hereby regarding any material matter; or

(d) any Legal Proceeding commenced or, to its Knowledge, threatened relating to the consummation of the transactions contemplated by this Agreement or any other Related Agreement.

The delivery of any such notice pursuant to this <u>Section 6.9</u> shall not be deemed to amend or supplement this Agreement and the failure to deliver any such notice shall not constitute a waiver of any right or condition to the consummation of the transactions contemplated hereby by any Party.

6.10 <u>Public Announcements</u>. No Party shall issue any press release or make any public announcement from the date hereof until the earlier to occur of the termination of this Agreement or the Effective Time relating to the existence or subject matter of this Agreement or the transactions contemplated hereby without the prior consent of the other Parties; <u>provided</u>, <u>however</u>, that any Party may make any public disclosure it determines in good faith is required by Applicable Law (in which case the disclosing Party will use its reasonable best efforts to advise the other Parties and provide them with a reasonable opportunity to review and comment thereon and consider such comments in good faith prior to making the disclosure, to the extent practicable and permissible under Applicable Law).

6.11 Section 16 Matters. Prior to the Effective Time, the board of directors or other comparable governing body, or a duly authorized committee thereof, of each of Parent, Merger Sub and the Company shall adopt resolutions as necessary to specifically approve and take all other actions required pursuant to Rules 16b-3(d) and 16b-3(e) under the Exchange Act to cause the disposition in the Merger of the Company Common Stock and derivative securities (as defined in Rule 16a-1(c) under the Exchange Act) relating thereto and the acquisition in the Merger of Parent Shares and derivative securities relating thereto, in either such case to be exempt from the provisions of Section 16(b) of the Exchange Act.

6.12 <u>NYSE Listing</u>. Parent shall use its reasonable best efforts to cause the shares of Parent Common Stock issuable to holders of Company Common Stock immediately prior to the Effective Time pursuant to <u>Section 1.7(c)</u> to be approved for listing on the NYSE, subject to official notice of issuance, as promptly as practicable after the date of this Agreement, and in any event prior to the Closing Date.

### 6.13 Tax Treatment.

(a) The Parties intend that, for federal income tax purposes, the Merger shall qualify as a reorganization within the meaning of Section 368(a) of the Code. The Parties (i) are aware of no facts or circumstances that would prevent the Merger from constituting a transaction qualifying under Section 368(a) of the Code, (ii) shall act in accordance with the treatment of the Merger as a reorganization within the meaning of Section 368(a) of the Code in the filing of all Tax Returns and in the course of any Tax audit, Tax review or Tax litigation relating thereto and (iii) shall take no position and shall cause their Affiliates to take no position inconsistent with the treatment of the Merger as a reorganization within the meaning of Section 368(a) of the Code.

(b) Parent and the Company shall use their reasonable best efforts, and shall cause their respective Affiliates to use their reasonable best efforts, to take or cause to be taken any action (whether before or after the Merger) that is required to cause the Merger to qualify as a reorganization within the meaning of Section 368(a) of the Code.

(c) The Parties shall cooperate and use their commercially reasonable efforts in order for the Parent to obtain the opinion of Kirkland & Ellis LLP described in Section 7.2(i) and for the Company to obtain the opinion

of Latham & Watkins LLP described in <u>Section 7.3(h)</u>. In connection therewith, both Parent (together with Merger Sub) and the Company shall deliver to Kirkland & Ellis LLP and Latham & Watkins LLP representation letters, dated and executed as of the dates of such opinions, in a form reasonably acceptable to Kirkland & Ellis LLP and Latham & Watkins LLP.

### 6.14 Tax Matters.

(a) <u>Tax Returns Filed After the Closing Date</u>. The Surviving Entity shall prepare or cause to be prepared and file or cause to be filed all Tax Returns for each of the Company and its Subsidiaries for all Tax periods ending on or prior to the Closing Date which are filed after the Closing Date. The Surviving Entity shall permit the Shareholder Representative to review and comment on each such Tax Return described in the preceding sentence prior to filing thereof and shall make such revisions as are reasonably requested by the Shareholder Representative, unless otherwise required by Applicable Law.

(b) <u>Tax Returns for Tax Periods Beginning Before and Ending After the Closing Date</u>. Parent, shall prepare or cause to be prepared and file or cause to be filed any Tax Returns of any of the Surviving Entity and its Subsidiaries for all Tax periods that begin and end after the Closing Date. Parent shall permit the Shareholder Representative to review and comment on each such Tax Return described in the preceding sentence prior to filing thereof and shall make such revisions as are reasonably requested by the Shareholder Representative with respect to items relating to the Pre-Closing Period, unless otherwise required by Applicable Law.

(c) <u>Cooperation on Tax Matters</u>. Parent, the Shareholder Representative, the Company and the Surviving Entity shall cooperate fully, as and to the extent reasonably requested by the other party, in connection with the preparation and filing of any Tax Return (including any report required pursuant to Section 6043 of the Code and all Treasury Regulations promulgated thereunder), any audit, litigation or other proceeding with respect to Taxes. Such cooperation shall include the retention and (upon the other party s request) the provision of records and information which are reasonably relevant to any such audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Parent, the Shareholder Representative, the Company and the Surviving Entity agree (i) to retain all books and records with respect to Tax matters pertinent to the Company and its Subsidiaries relating to any Pre-Closing Period, and to abide by all record retention agreements entered into with any Taxing authority, and (ii) to give the other party reasonable written notice prior to destroying or discarding any such books and records and, if the other party so requests, Parent, the Shareholder Representative, the Company or the Surviving Entity, as the case may be, shall allow the other party to take possession of such books and records.

(d) <u>Certain Tax Refunds</u>. Any Tax refund or credit of the Company or its Subsidiaries relating to a Pre-Closing Period described on <u>Section 6.14(d) of the Company Schedule</u> shall be for the benefit of the Shareholders and each of Parent, the Surviving Entity and its Subsidiaries shall pay to the Shareholder Representative for the benefit of the Shareholders the amount of such refund or credit (whether or not such refund or credit is applied to or used to offset the Post Closing Period Taxes of Parent, the Surviving Entity and its Subsidiaries) within fifteen (15) days after such refund or credit is received.

### 6.15 Shareholder Representative.

(a) <u>Authority</u>. Each Shareholder other than the Shareholder Representative hereby appoints the Shareholder Representative as its and their true and lawful agent and attorney-in-fact on behalf of each and every Shareholder on the terms and subject to the conditions set forth in this <u>Section 6.15</u>. In such capacity, the Shareholder Representative may take any action on behalf of, and may bind, any or all Shareholders from and after the date hereof under this Agreement or any other Related Agreement, including (i) entering into any such other Related Agreement, any amendment or modification hereof or any waiver of any obligation hereunder or thereunder; (ii) giving and receiving notices and communications to or from Parent or Merger Sub relating to this Agreement or any other Related Agreement or any other Related hereby or thereby (except to the extent that this Agreement or any other Related Agreement expressly contemplates that any such notice or

communication shall be given or received by a Shareholder individually without reliance on the Shareholder Representative); (iii) making any claim on behalf of any Shareholder Indemnified Party (provided that recovery of which shall be made or directed to such Shareholder Indemnified Party) pursuant to Article IX, consenting to the payment to any Parent Indemnified Party of any amounts payable pursuant to Article IX, making any objection to, or determine to waive or fail to take any action to object to, any claim for indemnification made by any Parent Indemnified Party pursuant to Article IX (including any notice contemplated by Section 9.4(a)); (iv) consenting or agreeing to, negotiate, entering into settlements and compromises of, and agreeing to arbitration and complying with orders of courts and awards of arbitrators with respect to, any claims contemplated by clause (ii) or (iii); and (v) taking all actions necessary or appropriate in the judgment of the Shareholder Representative for the accomplishment of the foregoing, in each case without having to seek or obtain the consent of any Shareholder under any circumstance; provided, however, that, except with respect to claims on behalf of any Shareholder Indemnified Party, the Shareholder Representative may not take any action that has the effect of providing disparate treatment to any Shareholder or group of Shareholders as compared to the other Shareholders (other than the Shareholder Representative in such capacity). The Person serving as the Shareholder Representative may be replaced upon his death, incapacity or resignation by the other Shareholders (with each Shareholder having such number of votes equal to the number of shares of Company Common Stock held by such Shareholder (for such purposes determined immediately prior to the Effective Time) and the Shareholders (without reliance on the actions of the Shareholder Representative) shall provide notice thereof to Parent. No bond shall be required of the Shareholder Representative, and the Shareholder Representative shall receive no compensation for its services hereunder. The Shareholder Representative accepts its appointment hereunder. Parent and their respective Representatives shall be entitled to rely on the authority of the Shareholder Representative for actions taken under this Section 6.15(a) and shall have no Liability to the Company or any Shareholder for such reliance.

(b) <u>Liability</u>. The Shareholder Representative shall not be liable to any other Shareholder for any act done or omitted hereunder as the Shareholder Representative while acting in accordance with the limitations set forth in <u>Section 6.15(a)</u> in good faith. Any act done or omitted pursuant to the advice of counsel shall be conclusive evidence of such good faith for such purposes.

### 6.16 Non-Competition; Non-Hire.

(a) <u>Non-Competition</u>. Each Shareholder shall not, and shall cause its respective Affiliates to not, directly or indirectly, as an owner, equityholder, manager, operator, consultant, member, partner, licensor, contractor, subcontractor, financing source, agent or in any other capacity, engage in any Competing Operation in any country in the world in which the Company or any of its Subsidiaries or Parent or any of its Subsidiaries conducts as of the date hereof or the Closing Date, or in the prior two (2) years has conducted, or conducts during the Restricted Period (the <u>Restricted Territory</u>), including the United States of America and its territories, the Philippines, Iraq, Afghanistan and the Pacific and Micronesian Islands, during the Restricted Period; <u>provided</u>, <u>however</u>, that a Shareholder may, without having been deemed to have breached this covenant, own as a passive investment not in excess of an aggregate of three percent (3%) of the outstanding capital stock of a Person which engages in a Competing Operation if such capital stock is a security which is actively traded on an established national or international securities exchange. A <u>Competing Operation</u> means any contracting or subcontracting of, or any other provision or promotion of, construction services or products, in either case including general contracting, plumbing, electrical, concrete forming or placing, steel erection, engineering, construction management, design-build services, self-performed construction services, whether for Governmental Entities, private customers or otherwise.

(b) <u>Non-Solicitation</u>. During the Restricted Period, no such Shareholder shall, nor shall it permit any Affiliate thereof to, directly or indirectly, for itself or on behalf of any other Person, solicit for hire or hire any officer, director or employee of the Company or any of its Subsidiaries, or any such individual who had been so employed within six (6) months prior to such solicitation or hiring by a Shareholder or any Affiliate of such Shareholder, or induce or attempt to induce any officer, director or employee to leave his or her employment with

the Company or any of its Subsidiaries; <u>provided</u>, <u>however</u>, that nothing in this <u>Section 6.16(b)</u> shall restrict or preclude the ability of any Shareholder to make generalized searches for employees by use of advertisements (in any medium) or search firms that do not target employees of the Company or any of its Subsidiaries.

(c) <u>Interference</u>. During the Restricted Period, no Shareholder shall, nor shall it permit any Affiliate thereof to, directly or indirectly, for itself or on behalf of any other Person, cause or attempt to cause any of the Company s or any of its Subsidiaries customers or suppliers to cease doing business or reduce the level of business with the Company or any of its Subsidiaries.

(d) <u>Confidential Information</u>. No Shareholder shall, nor shall it permit any Affiliate thereof to, at any time use or disclose to any Person other than Parent or any of its Subsidiaries, any information, knowledge or data relating to the business of the Company or its Subsidiaries (including information relating to accounts, financial dealings, transactions, trade secrets, Company Intellectual Property, customer lists, pricing lists, catalogues, brochures, service manuals, processes, formulae, plans and proposals) whether or not marked or otherwise identified as confidential or secret; except that any Shareholder or any of its Affiliates may disclose such information, knowledge or data: (i) that is or becomes generally available to the public other than as a result of disclosure by the Shareholders or (ii) when required by Applicable Law, subject to the following sentence. In the event any Shareholder or any of its Affiliates is requested pursuant to, or required by, Applicable Law to disclose any such information, knowledge or data, such Shareholder or Affiliate shall provide Parent with prompt notice of each such request or requirement so that Parent may seek an appropriate protective order.

(e) <u>Specific Performance</u>. Shareholders acknowledge that, in view of the nature of the business of the Company and its Subsidiaries and the business objectives of Parent in entering into the transactions contemplated hereby, and the consideration paid to the Shareholders hereby, the restrictions contained in this <u>Section 6.16</u> are reasonably necessary to protect the legitimate business interests of Parent and that any violation of such restrictions may result in irreparable injury to Parent and the business Parent has acquired hereunder, for which damages may not be an adequate remedy. Shareholders acknowledge that, if any such restrictions are violated, Parent shall be entitled to seek preliminary and permanent injunctive relief as well as to an equitable accounting of earnings, profits and other benefits arising from such violation, and shall not be required to post bond or other security to enforce their rights, any requirement respecting which is hereby waived by the Shareholders.

(f) <u>Severability</u>. Should any provision of this <u>Section 6.16</u> be held under any circumstances in any jurisdiction to be invalid or unenforceable, such invalidity or unenforceability shall not affect the validity or enforceability of any other provision or other part of such provision, or of such provision or part thereof under any other circumstances or in any other jurisdiction. It is acknowledged and agreed that the Restricted Territory is reasonable in scope. Notwithstanding the foregoing, if a court of competent jurisdiction decides that any provision of this <u>Section 6.16</u> is overbroad, the remaining provisions shall continue to be binding on the parties hereto.

(g) Exceptions. This Section 6.16 shall not be deemed to prohibit or restrict any action taken by a Shareholder as an officer or director of Parent or any of its Subsidiaries if taken on behalf of such entity.

6.17 <u>Compliance with Obligations</u>. Parent shall take all action necessary to cause Merger Sub and the Surviving Entity to perform their respective obligations under this Agreement and the other Related Agreements and to consummate the transactions contemplated hereby and thereby upon the terms and subject to the conditions set forth in this Agreement. Shareholders shall take all action necessary to cause the Company to perform its obligations under this Agreement and the other Related Agreements and to consummate the transactions contemplated hereby and thereby upon the terms and subject to the conditions set forth in this Agreements and to consummate the transactions contemplated hereby and thereby upon the terms and subject to the conditions set forth in this Agreement.

6.18 <u>Signatory Shareholders Joinder</u>. The Company and the Signatory Shareholders shall cause all Shareholders who are not Signatory Shareholders to execute and deliver to the other parties hereto a joinder

signature page to become a party to each of this Agreement and the Shareholders Agreement as promptly as practicable and in no event later than five (5) Business Days from and after the date hereof. Upon the execution and delivery thereof, such Shareholder shall become a Signatory Shareholder, the representations and warranties of the Shareholders made as of the date hereof shall be deemed for all purposes to be made by such Shareholder as of the date hereof (notwithstanding the actual date of execution and delivery). Prior to such joinder, the Company shall be liable for any deemed breach of any covenant hereunder by any Shareholder who is not a Signatory Shareholder hereto and shall take all reasonable actions to prevent any such deemed breach. This Section shall not limit <u>Section 2.5</u> or any other provision hereof.

6.19 <u>Guarantees</u>. Parent and the Company shall cooperate to obtain such consents and seek amendments to Contracts to remove and release the Shareholder Representative party hereto and his Affiliates as an obligor, guarantor or surety of any obligations of the Company or its Subsidiaries pursuant to those obligations set forth in <u>Section 3.22(b) of the Company Schedule</u> or any similar obligation incurred from and after the date hereof in accordance with the terms hereof.

6.20 Pre-Closing Dividend and Dividend Notes. Notwithstanding the limitations set forth in clause (ii) of Section 6.1(b), the Company may from time to time declare and distribute one or more dividends payable in cash to the holders of Company Common Stock an amount equal to the lesser of (x) \$120 million in the aggregate and (y) the amount of Available Cash (such dividends collectively, the <u>Cash Dividends</u>); provided, however, that the Company may not declare or distribute more than \$50 million in the aggregate of Cash Dividends prior to the date that is one (1) Business Day prior to the Closing; and provided further that the Company may not make any Cash Dividends unless the amount of Available Cash as of the Closing Date (after giving effect to such dividends, borrowings under the Company Credit Agreement and ongoing operations of the Company) will be at least zero. Notwithstanding the limitations set forth in clause (ii) or (iii) of Section 6.1(b), the Company may declare and distribute a dividend payable in the form of one or more notes substantially in the form of Annex E hereto, with such changes thereto as are reasonably requested by Parent prior to the issuance thereof in connection with any consent under, amendment to or replacement of any credit facilities or borrowings of Parent or the Company in connection with the transactions contemplated hereby (the \_Dividend Note ), to the holders of Company Common Stock in principal amount equal to no more than the amount by which \$120 million exceeds the aggregate amount of the Cash Dividends (such dividend, the <u>Note Dividend</u>), subject to any offset for amounts contemplated by Section 3.22(b) of the Company Schedule to be repaid prior to the Effective Time which have not been repaid by the date of the Note Dividend. The Note Dividend may not be declared or distributed more than one (1) Business Day prior to the Closing. Each of the Cash Dividends and the Note Dividend may only be declared and distributed subject to the limitations set forth in this Section 6.20 and to the extent permitted by Applicable Law. To the extent permitted by Applicable Law, the Cash Dividend and the Note Dividend shall be distributed pro rata among the Shareholders, unless the Shareholders shall agree to a different allocation distribution of the Cash Dividend and the Note Dividend. The Company shall reasonably consult with and cooperate with Parent in the determination of the dividend of the Dividend Note prior to declaring or distributing such dividend. <u>Available Cash</u> means the amount of unrestricted cash held by the Company as of any date of determination, which may include any cash obtained from borrowings under the Company Credit Agreement from and after the date hereof.

# ARTICLE VII

### CONDITIONS PRECEDENT TO THE CLOSING

7.1 <u>Mutual Conditions to Merger</u>. The respective obligations of Parent, Merger Sub and the Company to consummate the Merger and the other transactions contemplated hereby are subject to the satisfaction, on or before the Closing Date, of each of the following conditions, any one or more of which may be waived in writing by Parent, Merger Sub or the Company to the extent permitted by Applicable Law:

(a) <u>Shareholder Approval</u>. Parent shall have obtained the Parent Shareholder Approval, and the Shareholders shall have delivered the Company Shareholder Written Consent.

(b) <u>Governmental Approvals and Filings</u>. (i) Any applicable waiting period under the HSR Act relating to the Merger shall have expired or been terminated; (ii) any other Consents of any Governmental Entity required to consummate the transactions contemplated hereby, the failure of which to be obtained, individually or in the aggregate, would be material to any Party, shall have been obtained; and (iii) any other filings to be made with any Governmental Entity required to consummate the transactions contemplated hereby, the failure of which to be obtained, individually or in the aggregate, would be material to any Party, shall have been obtained, the failure of which to be obtained, individually or in the aggregate, would be material to any Party, shall have been obtained.

(c) <u>No Injunctions or Restraints</u>. No Governmental Entity of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any Applicable Law or Decree which is then in effect which (i) challenges the Merger or the other transactions contemplated hereby or has the effect of making the Merger or the other transactions contemplated hereby illegal or otherwise preventing the consummation of any of the Merger or the other transactions contemplated hereby or (ii) relates to any Legal Proceeding that seeks to restrain or prohibit the consummation of, or seeks the rescission of, the Merger or any other transactions contemplated hereby.

(d) <u>NYSE Listing</u>. The shares of Parent Common Stock issuable to holders of Company Common Stock immediately prior to the Effective Time pursuant to <u>Section 1.7(c)</u> shall have been approved for listing on the NYSE, subject to official notice of issuance.

(e) Charter Amendment. The Articles of Amendment shall have been filed and shall have become effective.

7.2 <u>Parent s Conditions to the Merg</u>er. The respective obligations of Parent and Merger Sub to consummate the Merger and the other transactions contemplated hereby are subject to the satisfaction, on or before the Closing Date, of each of the following further conditions, any one or more of which may be waived in writing by Parent and Merger Sub:

(a) <u>Representations and Warranties</u>. (i) The representations and warranties of the Company and the Shareholders (including any representation deemed made by any Shareholder who is not a Signatory Shareholder) contained in <u>Section 3.6</u> and <u>Section 4.1</u>, respectively, shall be true and correct in all respects (other than for *de minimis* differences in amounts) on the date of this Agreement and on the Closing Date as though made on the Closing Date (other than those representations and warranties which speak of an earlier date, which representations and warranties shall have been true and correct in all respects (other than for *de minimis* differences in amounts) as of such earlier date); and (ii) the other representations and warranties of the Company and the Shareholders (including any representation deemed made by any Shareholder who is not a Signatory Shareholder) contained in this Agreement shall be true and correct on the date of this Agreement and on the Closing Date as though made on the Closing Date (other than those representations and warranties which speak of an earlier date, which representations Date as though made on the Closing Date (other than those representations and warranties which speak of an earlier date, which representations and warranties shall be true and correct on the date of this Agreement and on the Closing Date as though made on the Closing Date (other than those representations and warranties which speak of an earlier date, which representations and warranties shall have been true and correct as of such earlier date), except in the case of this <u>Section 7.2(a)(ii)</u> for such failures to be true and correct which have not had and are not reasonably likely to have a Company Material Adverse Effect, it being agreed that any materiality, Company Material Adverse Effect and other similar qualifications in such representations and warranties shall be disregarded in determining any inaccuracies for purposes of this <u>Section 7.2(a)(ii)</u>;

(b) <u>Covenants and Agreements</u>. The Company and the Shareholders shall have complied in all material respects with all covenants and agreements contained in this Agreement to be performed by the Company or the Shareholders (whether or not any Shareholder is a Signatory Shareholder) prior to the Closing.

(c) <u>Company Material Adverse Effect</u>. Since the date hereof, there shall not have been any changes, events or developments that have had a Company Material Adverse Effect.

(d) <u>Officer s Certificate</u>. Parent shall have received a certificate signed by the Chief Executive Officer or Chief Financial Officer of the Company and dated as of the Closing Date to the effect that the conditions set forth in <u>Section 7.2(a)</u>, <u>Section 7.2(b)</u> and <u>Section 7.2(c)</u> have each been satisfied.

(e) <u>Consents</u>. The Company shall have obtained all necessary, proper or advisable Consents under any Contracts or Permits to which the Company, any of its Subsidiaries or any Shareholder is a party in connection with this Agreement and the consummation of the transactions contemplated hereby so as to maintain and preserve the benefits under such Contracts or Permits following the consummation of the transactions contemplated hereby, except those Consents the failure of which to obtain would not have a Company Material Adverse Effect.

(f) <u>Shareholders Agreement</u>. Each Shareholder shall have executed and delivered the Shareholders Agreement, and the Shareholders Agreement shall continue to be in full force and effect.

(g) <u>Employment Agreement</u>. The Shareholder Representative shall have executed and delivered the Employee Employment Agreement, and the Shareholder Representative shall not have provided actual, constructive or anticipatory notice of termination of service thereunder.

(h) <u>FIRPTA Certification</u>. The Company shall have provided to Parent and Merger Sub a certification dated as of the Closing Date and in accordance with Sections 1.897-2(h) and 1.1445-2(c)(3) of the Treasury regulations to the effect that Company is not, nor has it been at any time during the period specified in Section 897(c)(1)(A)(ii), a United States Real Property Holding Corporation as that term is defined in Section 897(c)(2) of the Code.

(i) <u>Tax Opinion</u>. Parent shall have received an opinion of Kirkland & Ellis LLP, dated the Closing Date and based on customary facts, representations and assumptions set forth or described in such opinion, to the effect that the Merger will qualify as a reorganization within the meaning of Section 368(a) of the Code. In rendering such opinion, Kirkland & Ellis LLP will be entitled to receive and rely upon customary certificates and representations of officers of the Company and Parent.

(j) <u>Audited Company Financial Statements</u>. (i) The Company shall have delivered to Parent the audited consolidated balance sheets as of December 31, 2006 and 2007 and the related audited consolidated statements of income, shareholders equity and cash flows of the Company and its Subsidiaries for the fiscal years ended December 31, 2005, 2006 and 2007 (the <u>Audited Company Financial Statements</u>), together with the unqualified audit opinion (other than for such qualification for a material weakness set forth in the Unaudited Company Financial Statements, by April 30, 2008, and (ii) the respective Audited Company Financial Statements shall not differ in any material respect (whether by way of reserve, footnote disclosure or otherwise), including any material change in any of the total current assets, total assets, total current liabilities, total liabilities, shareholders equity, total shareholders equity, net tangible book value, income from operations or net income line items, from the Unaudited Company Financial Statements as of and for the company Schedule.

7.3 <u>The Company</u> s Conditions to the Merger. The obligations of the Company to consummate the Merger and the other transactions contemplated hereby are subject to the satisfaction, on or before the Closing Date, of each of the following further conditions, any one or more of which may be waived in writing by the Company:

(a) <u>Representations and Warranties</u>. (i) The representations and warranties of Parent contained in <u>Section 5.6</u> shall be true and correct in all respects (other than for *de minimis* differences in amounts) on the date of this Agreement and on the Closing Date as though made on the Closing Date (other than those representations and warranties which speak of an earlier date, which representations and warranties shall have been true and correct in all respects (other than for *de minimis* differences in amounts) as of such earlier date); and (ii) the other representations and warranties of Parent contained in this Agreement shall be true and correct on the date of this Agreement and on the Closing Date as though made on the Closing Date (other than those representations and warranties which speak of an earlier date, which representations and warranties shall be true and correct on the date of this Agreement and on the Closing Date as though made on the Closing Date (other than those representations and warranties which speak of an earlier date, which representations and warranties shall have been true and correct as of such earlier date), except in the case of this <u>Section 7.3(a)(ii)</u> for such failures to be true and correct which

have not had and are not reasonably likely to have a Company Material Adverse Effect, it being agreed that any materiality, Parent Material Adverse Effect and other similar qualifications in such representations and warranties shall be disregarded in determining any inaccuracies for purposes of this <u>Section 7.3(a)(ii)</u>;

(b) <u>Covenants and Agreements</u>. Parent and Merger Sub shall have complied in all material respects with all covenants and agreements contained in this Agreement to be performed by Parent and Merger Sub prior to the Closing.

(c) <u>Parent Material Adverse Effect</u>. Since the date hereof, there shall not have been any changes, events or developments that have had a Parent Material Adverse Effect.

(d) <u>Officer s Certificate</u>. The Company shall have received a certificate signed by the Chief Executive Officer or Chief Financial Officer of Parent and dated as of the Closing Date to the effect that the conditions set forth in <u>Section 7.3(a)</u>, <u>Section 7.3(b)</u> and <u>Section 7.3(c)</u> have each been satisfied.

(e) <u>Consents</u>. Parent shall have obtained all necessary, proper or advisable Consents under any Contracts or Permits to which Parent or any of its Subsidiaries is a party in connection with this Agreement and the consummation of the transactions contemplated hereby so as to maintain and preserve the benefits under such Contracts and Permits following the consummation of the transactions contemplated hereby, except those Consents the failure of which to obtain would not have a Parent Material Adverse Effect.

(f) <u>Shareholders Agreement</u>. Parent shall have executed and delivered the Shareholders Agreement, and the Shareholders Agreement shall continue to be in full force and effect.

(g) Employment Agreement. Parent shall have executed and delivered the Employment Agreement.

(h) <u>Tax Opinion</u>. The Company shall have received an opinion of Latham & Watkins LLP, dated the Closing Date and based on customary facts, representations and assumptions set forth or described in such opinion, to the effect that the Merger will qualify as a reorganization within the meaning of Section 368(a) of the Code. In rendering such opinion, Latham & Watkins LLP will be entitled to receive and rely upon customary certificates and representations of officers of the Company and Parent.

7.4 <u>No Frustration of Closing Conditions</u>. No Party may rely on or assert the failure of any condition to its obligation to consummate the transactions contemplated hereby set forth in this <u>Article VII</u>, as appropriate, to be satisfied if such failure was caused by such Party s failure to use its reasonable best efforts to satisfy the conditions to the consummation of the transactions contemplated hereby or any other material breach of any other provision hereunder.

### ARTICLE VIII

### TERMINATION

8.1 <u>Termination</u>. This Agreement may be terminated and the Merger abandoned at any time prior to the Closing Date regardless of whether the Parent Shareholder Approval has been obtained:

(a) by mutual written agreement of the Company and Parent;

(b) by either Parent or the Company, if the Merger has not occurred by September 30, 2008 (the <u>Initial Termination Date )</u>; provided, however, that if (x) the Merger has not occurred by such date by reason of non-satisfaction of any of the conditions set forth in <u>Section 7.1(a)</u>, <u>Section 7.1(b)</u> or <u>Section 7.1(e)</u> and (y) as of such date, all other conditions set forth in <u>Article VII</u> have been satisfied or waived or are then capable of being satisfied, then such right of termination shall not be available to Parent or the Company unless the Merger has not

occurred by December 31, 2008 (the <u>Extended Termination Date</u>; the Initial Termination Date or, if applicable hereto, the Extended Termination Date shall be referred to as the <u>Termination Date</u>); and provided further that the right to terminate this Agreement pursuant to this Section 8.1(b) shall not be available to any Party whose failure to fulfill in any material respect any obligation hereunder has been the cause of, or resulted in, the failure of the Merger to have become effective on or before the Termination Date and such action or failure constitutes a breach of this Agreement;

(c) by Parent or the Company, if a Governmental Entity of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any Applicable Law or Decree having any of the effects set forth in <u>Section 7.1(c)</u> that is in effect and has become final and non-appealable; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to this <u>Section 8.1(c)</u> shall not be available to any Party who shall have failed to use its reasonable best efforts to resist or prevent the entry or passage, or to seek the removal or lifting of, such Applicable Law or Decree;

(d) by Parent, in the event (i) that any representation or warranty of the Company or the Shareholders (including any representation deemed made by any Shareholder who is not a Signatory Shareholder) set forth in this Agreement shall have been inaccurate when made or shall not be capable of being made as of the Termination Date or (ii) of a material breach of any covenant or agreement on the part of the Company or any Shareholder (whether or not any Shareholder is a Signatory Shareholder) set forth in this Agreement, in the case of either clause (i) or (ii) such that the conditions set forth in <u>Section 7.2(a)</u> or <u>Section 7.2(b)</u>, respectively, could not be satisfied as of the Termination Date; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, in the event that such inaccuracies in the representations and warranties of the Company or the Shareholders or such breaches of covenant by the Company or the Shareholders are curable by the Company or the Shareholders through the exercise of reasonable best efforts, then Parent shall not be permitted to terminate this Agreement pursuant to this <u>Section 8.1(d)</u> until the earlier to occur of (x) the expiration of a twenty (20) Business Day period after delivery of written notice from Parent to the Company of such breach or inaccuracy or breach (it being understood that Parent may not terminate this Agreement pursuant to this <u>Section 8.1(d)</u> if such inaccuracy or breach by the Company or the Shareholders is cured within such twenty (20) Business Day period); and <u>provided further</u> that Parent may not terminate this Agreement pursuant to this <u>Section 8.1(d)</u> if the inaccuracy or breach by the Company or the Shareholders is cured within such twenty (20) Business Day period); and <u>provided further</u> that Parent may not terminate this Agreement pursuant to this <u>Section 8.1(d)</u> if the inaccuracy or breach by the Company or the Shareholders is cured within such twenty (20) Business Day period); and <u>provided further</u> that Parent may not terminate this Agreement purs

(e) by the Company, in the event (i) that any representation or warranty of Parent set forth in this Agreement shall have been inaccurate when made or shall not be capable of being made as of the Termination Date or (ii) of a material breach of any covenant or agreement on the part of Parent or Merger Sub set forth in this Agreement, in the case of either clause (i) or (ii) such that the conditions set forth in <u>Section 7.3(a)</u> or <u>Section 7.3(b)</u>, respectively, could not be satisfied as of the Termination Date; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, in the event that such inaccuracies in the representations and warranties of Parent or such breaches of covenant by the Parent or Merger Sub are curable by Parent or Merger Sub through the exercise of reasonable best efforts, then the Company shall not be permitted to terminate this Agreement pursuant to this <u>Section 8.1(e)</u> until the earlier to occur of (x) the expiration of a twenty (20) Business Day period after delivery of written notice from the Company to Parent of such breach or inaccuracy, as applicable, or (y) the ceasing by Parent or Merger Sub, as applicable, to exercise reasonable best efforts to cure such inaccuracy or breach (it being understood that the Company may not terminate this Agreement pursuant to this <u>Section 8.1(e)</u> if such inaccuracy or breach by the Parent or Merger Sub is cured within such twenty (20) Business Day period); and <u>provided further</u> that the Company may not terminate this Agreement pursuant to this <u>Section 8.1(e)</u> if such inaccuracy or breach by the Parent or Merger Sub is cured within such twenty (20) Business Day period); and <u>provided further</u> that the Company may not terminate this Agreement pursuant to this <u>Section 8.1(e)</u> if the inaccuracy or breach by the material inaccuracy or breach of any representation or warranty or covenant or agreement hereunder, as appropriate, by any of the Company or the Shareholders;

(f) by Parent or the Company, if the Parent Shareholder Approval shall not have been obtained at the Parent Shareholders Meeting, or at any adjournment or postponement thereof, at which the vote was taken thereon; or

(g) (i) by the Company, if Parent shall have effected any Parent Change of Recommendation in accordance with <u>Section 6.5(b)</u>, or (ii) by Parent, if Parent shall have effected a Qualifying Parent Change of Recommendation in accordance with <u>Section 6.5(b)</u>.

8.2 Effect of Termination: Controlling Trusts Guarantee. Except as otherwise set forth in this Section 8.2, any termination of this Agreement pursuant to Section 8.1 will be effective immediately upon the delivery of written notice as provided therein. Upon a termination of this Agreement pursuant to Section 8.1, this Agreement shall forthwith become void and of no further force or effect, and there shall be no liability under this Agreement on the part of any Party, provided, however, that (a) nothing herein shall relieve any Party from liability for any breach of its covenants or agreements or willful breach of any of its representations or warranties set forth in this Agreement prior to such termination; (b) the terms of Section 6.8(b), Section 6.15, Section 6.17, Section 8.3, Section 9.6, Article X and Article XI and this Section 8.2 shall survive any termination of this Agreement; and (c) no termination of this Agreement shall affect the obligations of the Parties set forth in the Confidentiality Agreement, all of which obligations shall survive termination of this Agreement in accordance with its terms. The Controlling Trusts shall jointly and severally unconditionally guarantee for the benefit of Parent and its Subsidiaries the representations, warranties, covenants and obligations of the Company and each Shareholder (regardless of whether such Shareholder is or becomes a Signatory Shareholder) hereunder and shall be liable for any inaccuracy or breach thereof occurring prior to the termination of this Agreement, which guarantee and obligation shall survive the termination of this Agreement, which

#### 8.3 Expenses; Termination Fee.

(a) <u>General</u>. Except as set forth in this <u>Section 8.3</u> or as otherwise expressly provided herein, all costs, fees and expenses incurred in connection with this Agreement and the transactions contemplated hereby (including legal fees and expenses) shall be paid by the Party incurring such expenses, whether or not the Merger is consummated; <u>provided</u>, <u>however</u>, that Parent and the Company shall each bear and pay one-half of the costs and expenses incurred in connection with the filing fees for the premerger notification and report form relating required by the HSR Act with respect to this Agreement.

(b) <u>Termination Fee</u>. Parent shall pay to the Company a fee equal to Thirty Million Dollars (\$30,000,000) in the event that this Agreement is terminated as follows:

(i) if (A) Parent shall terminate this Agreement pursuant to  $\underline{\text{Section 8.1(g)}}$  and (B) simultaneously therewith or immediately thereafter shall enter into an Acquisition Agreement, then in such case the fee shall be paid upon the termination of this Agreement;

(ii) if (A) Parent or the Company shall terminate this Agreement pursuant to <u>Section 8.1(g)</u> other than in the context contemplated by <u>Section 8.3(b)(i)</u>, and (B) Parent enters into an Acquisition Agreement, or an Acquisition Transaction with respect to Parent is otherwise consummated, in either case, within twelve (12) months following the termination of this Agreement (it being understood that, solely for purposes of this clause (B), references to twenty percent (20%) and eighty percent (80%) in the definition of Acquisition Transaction (and indirectly the corresponding application thereto in the definition of Acquisition Agreement) shall be deemed to be replaced with thirty-five percent (35%) and sixty-five percent (65%) ), then in such case the fee shall be paid within one (1) Business Day following the consummation of the Acquisition Transaction; or

(iii) if (A) Parent or the Company shall terminate this Agreement pursuant to <u>Section 8.1(f)</u>, (B) at any time after the date of this Agreement and before the Parent Shareholders Meeting (or at any adjournment or postponement thereof at which a vote on the Parent Shareholder Approval was taken), an Acquisition Proposal with respect to Parent shall have been publicly announced and not publicly withdrawn, and (C) Parent enters into

an Acquisition Agreement, or an Acquisition Transaction with respect to Parent is otherwise consummated, in either case, within twelve (12) months following the termination of this Agreement (it being understood that, solely for purposes of this clause (C), references to twenty percent (20%) and eighty percent (80%) in the definition of Acquisition Transaction (and indirectly the corresponding application thereto in the definition of Acquisition Agreement) shall be deemed to be replaced with thirty-five percent (35%) and sixty-five percent (65%) ), then in such case the fee shall be paid within one (1) Business Day following the consummation of the Acquisition Transaction.

In no event shall more than one fee become payable pursuant to this <u>Section 8.3(b)</u>. The fee contemplated by this <u>Section 8.3(b)</u> shall be paid by wire transfer of immediately available funds to an account or accounts designated in writing by the Company on the terms and subject to the conditions set forth herein.

(c) <u>Reimbursement of Company Expenses</u>. If (i) Parent or the Company shall terminate this Agreement pursuant to <u>Section 8.1(f)</u> or (ii) (x) Parent shall have effected a Parent Change of Recommendation in accordance with <u>Section 6.5(b)</u> and (y) thereafter Parent or the Company shall terminate this Agreement pursuant to <u>Section 8.1(g)</u>, Parent shall reimburse the Company for its reasonable, documented, out-of-pocket expenses related to this Agreement and the transactions contemplated hereby (including those related to the preparation of any proposed filings with the SEC by the Company prior to the date hereof); <u>provided</u>, <u>however</u>, that in no event shall Parent be required to reimburse the Company pursuant to this <u>Section 8.3(c)</u> for any expenses in excess of Five Million Dollars (\$5,000,000) in the aggregate. Payment for the reimbursement of expenses contemplated by this <u>Section 8.3(c)</u> shall be paid by wire transfer of immediately available funds to an account or accounts designated in writing by the Company on the terms and subject to the conditions set forth herein promptly following demand therefor.

(d) <u>Enforcement</u>; <u>Effect</u>. Parent acknowledges and hereby agrees that the provisions of <u>Section 8.3(b)</u> and <u>Section 8.3(c)</u> are an integral part of the transactions contemplated by this Agreement and that, without such provisions, the Company would not have entered into this Agreement.

#### ARTICLE IX

#### SURVIVAL; INDEMNIFICATION

#### 9.1 Survival.

(a) <u>Representations and Warranties</u>. The respective representations and warranties of the Company, the Shareholders and Parent set forth in this Agreement made as of the execution and delivery hereof or deemed made as of the Closing Date shall survive the Closing and will remain in full force and effect thereafter until the first (1st) anniversary of the Closing Date (the <u>Initial Survival Termination Date</u>), after which time they shall be of no further force or effect and upon which no claim for indemnification may be made with respect to breaches thereof; <u>provided</u>, <u>however</u>, that (i) the representations and warranties set forth in <u>Section 3.6</u>, <u>Section 3.19</u>, <u>Section 4.1</u>, <u>Section 4.4</u>, <u>Section 4.5</u>, <u>Section 5.6</u> and <u>Section 5.19</u> shall survive the Closing and will remain in full force and effect thereafter until thirty (30) calendar days following the expiration of the applicable statute of limitations with respect thereto, (ii) the representations and warranties set forth in <u>Section 3.18</u> shall survive the Closing Date will remain in full force and effect thereafter until the fifth (5<sup>th</sup>) anniversary of the Closing Date, and (iii) the representations and warranties set forth in <u>Section 3.1</u>, <u>Section 3.25</u>, <u>Section 4.2</u>, <u>Section 5.1</u>, <u>Section 5.3</u> and <u>Section 5.25</u> shall survive indefinitely.

(b) <u>Covenants and Agreements</u>. The respective covenants and agreements of the Parties set forth in this Agreement shall survive the Closing in accordance with their terms.

# 9.2 Indemnification.

(a) <u>Indemnification by the Shareholders</u>. Subject to the other provisions of this <u>Article IX</u> (including <u>Section 9.7</u>), from and after the Effective Time, the Shareholders shall indemnify, defend and hold harmless

Parent and its Subsidiaries (for avoidance of doubt, including the Surviving Entity and its Subsidiaries) (the <u>Parent Indemnified Parties</u>) from and against any and all Losses incurred or suffered by any such Parent Indemnified Parties directly or indirectly as a result of, with respect to or in connection with:

(i) any misrepresentation or breach of any representation or warranty of the Company or any Shareholder (including any representation deemed made by any Shareholder who is not a Signatory Shareholder) set forth in this Agreement made as of the execution and delivery hereof or deemed made as of the Closing Date set forth in this Agreement;

(ii) any failure to perform any covenant or agreement of the Company or any Shareholder (whether or not any Shareholder is a Signatory Shareholder) set forth in this Agreement; or

(iii) any (A) qualification to the audit opinion from Deloitte & Touche with respect to the Audited Company Financial Statements (other than for such qualification for a material weakness set forth in the Unaudited Company Financial Statement as of and for the fiscal year ended December 31, 2007), or (B) any difference (whether by way of line item amount, reserve, footnote disclosure or otherwise) from the Unaudited Company Financial Statements as of and for the comparable dates and periods, except for the item set forth on Section 7.2(j) of the Company Schedule.

(b) <u>Indemnification by Parent</u>. Subject to the other provisions of this <u>Article IX</u>, from and after the Effective Time, Parent shall indemnify, defend and hold harmless the Shareholders, (the <u>Shareholder Indemnified Parties</u>) from and against any and all Losses incurred or suffered by any such Shareholder Indemnified Parties directly or indirectly as a result of, with respect to or in connection with:

(i) any misrepresentation or breach of any representation or warranty of Parent set forth in this Agreement made as of the execution and delivery hereof or deemed made as of the Closing Date set forth in this Agreement; or

(ii) any failure to perform any covenant or agreement of Parent or Merger Sub set forth in this Agreement.

#### 9.3 Limitations.

(a) Notwithstanding anything contained herein to the contrary, the maximum aggregate liability of the Shareholders to all Parent Indemnified Parties taken together for all Losses pursuant to Section 9.2(a)(i) shall be limited to an amount equal to sixty million dollars (\$60,000,000) (the Liability Cap Amount ); provided, however, that this limitation shall not apply to Losses of any Parent Indemnified Parties with respect to or related to the Company s or any Shareholder s breach of any representation or warranty set forth in Section 3.1, Section 3.3, Section 3.6, Section 3.19, Section 3.25, Section 3.26, Section 3.27, Section 4.1, Section 4.2, Section 4.4 or Section 4.5, for which the maximum aggregate liability of the Shareholders to all Parent Indemnified Parties taken together for all Losses pursuant to Section 9.2(a)(i) shall be the Calculated Purchase Price.

(b) Notwithstanding anything contained herein to the contrary, the maximum aggregate liability of Parent to all Shareholder Indemnified Parties taken together for all Losses pursuant to <u>Section 9.2(b)(i)</u> shall be limited to the Liability Cap Amount; <u>provided</u>, <u>however</u>, that this limitation shall not apply to Losses of any Shareholder Indemnified Parties with respect to or related to Parent s breach of any representation or warranty set forth in <u>Section 5.1</u>, <u>Section 5.3</u>, <u>Section 5.6</u>, <u>Section 5.19</u>, <u>Section 5.24</u>, <u>Section 5.25</u> and <u>Section 9.2(b)(i)</u> shall be the Calculated Parties taken together for all Losses pursuant to <u>Section 9.2(b)(i)</u> shall be the Calculated Purchase Price.

(c) Notwithstanding anything contained herein to the contrary, the Shareholders shall not be obligated to make any payment pursuant to Section 9.2(a)(i) unless: (i) the individual claim (together with any group of related claims) for Losses thereunder is in excess of two hundred fifty thousand dollars (\$250,000) (the

<u>Minimum Claim Amount</u>): provided, however, that the limitations set forth in this Section 9.3(c)(i) shall not be applicable to Losses related to a breach of any representation or warranty of the Company or any Shareholder set forth in this Agreement that includes or is subject to any Material Adverse Effect, materiality or any other similar qualification (including as a result of the application of Section 9.3(e)), and (ii) unless and until the aggregate Losses of the Parent Indemnified Parties for which indemnification is permitted thereunder collectively exceed an amount equal to five million dollars (\$5,000,000) (the <u>Basket</u>), and then any indemnification with respect to such Losses shall be required to be made by the Shareholders only to the extent of such excess over the Basket; provided, however, that the limitations set forth in this <u>Section 9.3(c)</u> shall not apply to Losses of any Parent Indemnified Parties with respect to or related to the Company s or any Shareholder s breach of any representation or warranty set forth in <u>Section 3.6</u>, <u>Section 3.19</u>, <u>Section 4.1</u> or <u>Section 4.5</u>.

(d) Notwithstanding anything contained herein to the contrary, Parent shall not be obligated to make any payment pursuant to <u>Section 9.2(b)(i)</u> unless: (i) the individual claim (together with any group of related claims) for Losses thereunder is in excess of the Minimum Claim Amount; <u>provided, however</u>, that the limitations set forth in this <u>Section 9.3(d)(i)</u> shall not be applicable to Losses related to a breach of any representation or warranty of Parent set forth in this Agreement that includes or is subject to any Material Adverse Effect, materiality or any other similar qualification (including as a result of the application of <u>Section 9.3(f)</u>), and (ii) unless and until the aggregate Losses of the Shareholder Indemnified Parties for which indemnification is permitted thereunder collectively exceed the Basket, and then any indemnification with respect to such Losses shall be required to be made by Parent only to the extent of such excess over the Basket.

(e) Notwithstanding anything contained herein to the contrary, for purposes of <u>Section 9.2(a)(i)</u>, in determining whether there has been a breach of any representation or warranty of the Company or any Shareholder set forth in this Agreement or the amount of any Loss related to a breach of any such representation or warranty, references to Company Material Adverse Effect shall be deemed references to material to the Company and its Subsidiaries.

(f) Notwithstanding anything contained herein to the contrary, for purposes of <u>Section 9.2(b)(i)</u>, in determining whether there has been a breach of any representation or warranty of Parent set forth in this Agreement or the amount of any Loss related to a breach of any such representation or warranty, references to Parent Material Adverse Effect shall be deemed references to material to the Parent and its Subsidiaries.

(g) The obligations to indemnify, defend and hold harmless a Party pursuant to <u>Section 9.2</u> shall cease to survive when the applicable representation or warranty ceases to survive pursuant to <u>Section 9.1(a)</u>; provided, however, that such obligations to indemnify, defend and hold harmless shall not terminate with respect to any item as to which the Indemnified Party shall have, before the expiration of the applicable survival period, previously made a claim by delivering a written notice (stating in reasonable detail the basis of such claim) to the Party required to provide such indemnification.

(h) No Indemnified Party shall be entitled to recover more than the full amount of any Loss incurred by such Indemnified Party under the provisions of this Agreement in respect of any such Loss. Without limiting the generality of the foregoing, the amount of any Losses subject to indemnification under <u>Section 9.2</u> shall be reduced by (1) the amounts actually recovered by the Indemnified Party (or its related Persons) incurring such Loss under applicable insurance policies with respect to claims related to such Losses and (2) the amounts of any net tax benefits actually realized by such Person with respect to such Losses.

(i) The waiver by any Party hereto of any conditions set forth in <u>Article VII</u> will not affect or limit the provisions of this <u>Article IX</u> in any manner.

(j) Except as otherwise payable pursuant to a Third Party Claim, no Indemnifying Party shall be liable pursuant to <u>Section 9.2</u> to any Indemnified Party for any consequential, incidental or punitive Losses or Losses for lost profits.

(k) Each Shareholder Indemnified Party who is an officer or director of the Company or any of its Subsidiaries waives any right of contribution or other similar right against the Surviving Company or such Subsidiary, respectively, and its respective officers, directors, employees, assigns, successors and Affiliates that such Shareholder Indemnified Party may have by virtue of his or her status as an officer or director of the Company or such Subsidiary with respect to any claim for indemnification by Parent arising out of a breach of the representations, warranties, covenants or agreements of the Company or Shareholder contained herein. Each Shareholder Indemnified Party also agrees not to make a claim (except on behalf of the Surviving Entity) for recovery against any directors and officers liability insurance policy of the Surviving Entity or any of its Affiliates (including Parent) with respect to matters covered by the foregoing waiver.

#### 9.4 Claims.

(a) <u>Notice of Claim</u>. Promptly after the receipt by a Parent Indemnified Party or a Shareholder Indemnified Party, as appropriate (the <u>Indemnified Party</u>), of a notice of any claim, action, suit or proceeding by any third party (<u>a Third Party Claim</u>) or upon becoming aware of any other facts, circumstances or events, in either case that would reasonably give rise to a right to indemnification pursuant to this <u>Article IX</u>, such Indemnified Party shall give written notice of such claim to the indemnifying party hereunder (the <u>Indemnifying Party</u>), stating the nature and basis of the claim and the amount thereof, to the extent known, along with copies of the relevant documents evidencing the claim and the basis for indemnification sought. Failure of the Indemnified Party to give such notice shall not relieve the Indemnifying Party from liability on account of such indemnification, except if and to the extent that the Indemnifying Party is actually prejudiced thereby. Thereafter, the Indemnified Party shall promptly deliver to the Indemnifying Party copies of all notices and documents (including court papers) received by the Indemnified Party relating to the Third Party Claim or such other claim.

(b) Assumption of Defense of Third Party Claims. The Indemnifying Party shall have the right to assume the defense of the Indemnified Party against any Third Party Claim with counsel of its choice reasonably satisfactory to the Indemnified Party and control the defense of such Third Party Claim (x) if it elects to do so within twenty (20) Business Days following receipt of the notice with respect thereto and (y) so long as the Indemnifying Party conducts such defense actively and diligently. The Indemnifying Party may not assume the defense of a Third Party Claim if the Third Party Claim of which it seeks to assume control (i)(x) seeks non-monetary relief or (y) if such Third Party Claim, if determined adversely to the Indemnified Party, could have a material adverse effect on such Indemnified Party; (ii) involves criminal or quasi-criminal allegations; (iii) involves a claim which the Indemnifying Party failed or is failing to vigorously prosecute or defend; or (iv) involves a claim that is reasonably expected to result in Liability to the Indemnified Party in excess of the Indemnifying Party s indemnification obligation hereunder in light of the limitations set forth in this Article IX; provided, however, that, in the event that the Indemnified Party assumes the defense of a Third Party Action pursuant to the provisions of this sentence, the Indemnifying Party shall have the right to participate in the defense of the Third Party Claim, and may retain separate co-counsel at its sole cost and expense and the Indemnified Party shall not compromise or settle such Third Party Claim without the prior written consent of the Indemnifying Party. If the Indemnifying Party elects to assume the defense of the Third Party Claim, then, notwithstanding anything to the contrary in this Article IX, the Indemnifying Party shall not be required to pay or otherwise indemnify the Indemnified Party against attorneys fees or other expenses incurred on behalf of the Indemnified Party in connection with the defense of such action following the Indemnifying Party s election to assume the defense. So long as the Indemnifying Party has assumed the defense of the Third Party Claim in accordance herewith and notified the Indemnified Party in writing thereof, (i) the Indemnified Party may retain separate co-counsel at its sole cost and expense and participate in the defense of the Third Party Claim, it being understood that the Indemnifying Party shall pay all reasonable costs and expenses of counsel for the Indemnified Party for all periods prior to such time as the Indemnifying Party has notified the Indemnified Party that it has assumed the defense of such Third Party Claim; (ii) the Indemnified Party shall not file any papers or consent to the entry of any judgment or enter into any settlement with respect to the Third Party Claim without the prior written consent of the Indemnifying Party (not to be unreasonably withheld or delayed); and (iii) the Indemnifying Party will not consent to the entry of any judgment or enter into any settlement with respect to the

Third Party Claim (other than a judgment or settlement that is solely for money damages and is accompanied by a release of all indemnifiable claims against the Indemnified Party) without the prior written consent of the Indemnified Party, which consent shall not be unreasonably withheld or delayed. Whether or not the Indemnifying Party shall have assumed the defense, such Indemnifying Party shall not be obligated to indemnify the Indemnified Party hereunder for any settlement entered into without the Indemnifying Party s prior written consent, which consent shall not be unreasonably withheld or delayed. If the Indemnifying Party elects not to assume the defense of an action contemplated by this Section 9.4(b) within the manner contemplated above, or, if such action is assumed by the Indemnifying Party, the Indemnifying Party fails to conduct the defense actively and diligently, (A) the Indemnified Party may defend against the action in any manner it reasonably may deem appropriate; provided, however, that the Indemnified Party shall consult with the Indemnifying Party in connection therewith; (B) the Indemnifying Party will reimburse the Indemnified Party promptly and periodically for the reasonable costs of defending against the action (including reasonable attorneys fees and expenses) to the extent such costs are Losses for which the Indemnified Party is entitled to indemnification therefor hereunder; and (C) the Indemnifying Party will remain responsible for any costs the Indemnified Party may incur resulting from the action to the extent such costs are Losses for which the Indemnified Party is entitled to indemnification therefor hereunder; provided, however, that, in the event that it is determined that the Indemnified Party was not entitled to indemnification for any costs that are Losses hereunder, such Indemnified Party shall promptly refund to the Indemnifying Party that portion of the amounts paid or reimbursed by the Indemnifying Party to such Indemnified Party in respect of reimbursement of such costs. In the event that both the Indemnified Party and the Indemnifying Party are named as defendants in an action and/or the representation of both Persons by the same counsel would be appropriate under applicable standards of professional conduct, they shall both be represented by the same counsel (on whom they shall agree), unless such counsel, the Indemnified Party or the Indemnifying Party shall determine that such counsel has a conflict of interest in representing both the Indemnified Party and the Indemnifying Party in the same action or proceeding and the Indemnified Party and the Indemnifying Party do not waive such conflict to the satisfaction of such counsel, in which case the reasonable expense of separate counsel for the Indemnified Party shall be paid by the Indemnifying Party.

(c) <u>Payment</u>. On each occasion that any Indemnified Party shall be entitled to indemnification or reimbursement under this <u>Article IX</u>, the Indemnifying Party shall, at each such time, promptly pay the amount of such indemnification or reimbursement.

9.5 <u>Adjustments to Merger Consideration</u>. The Company, Parent, the Surviving Entity and the Shareholders agree to treat each indemnification payment pursuant to this <u>Article IX</u> as an adjustment to the consideration to be issued pursuant to the Merger for all Tax purposes and shall take no position contrary thereto unless required to do so by applicable Tax Law pursuant to a determination as defined in Section 1313(a) of the Code.

9.6 <u>Waiver of Reliance on Excluded Information</u>. (a) Other than for purposes of <u>Section 3.19</u> or <u>Section 4.5</u>, the Company and the Shareholders, on one side, and Parent and Merger Sub, on the other, each disclaim and waive for all purposes hereunder reliance on, any claim under, or failure to satisfy a condition set forth in <u>Section 7.3</u> related to, any fact, circumstance or event related to any of the other Parties or any of their Subsidiaries (for these purposes, expressly including joint ventures) that was actually Known (for avoidance of doubt, subject to the provisions set forth in <u>Section 10.3(m)</u>) by any such waiving Party (including, in the case of the Shareholders or the Company, the Shareholders Representative); and (b) the Company and the Shareholders each disclaim and waive for all purposes hereunder reliance on, any claim under, or failure to satisfy a condition set forth in <u>Section 7.3</u> related to, any action or failure to take action by the Company, Parent or any of their respective Subsidiaries (for these purposes, expressly including joint ventures) that was taken by or at the direction or with the consent (which need not be in writing), of the Shareholder Representative, including in his capacity as an officer, director, contractor or fiduciary of the Company, Parent or any of their respective Subsidiaries (any fact, circumstance or event or action or failure to take action contemplated by clause (a) through (b) as applicable to those Parties having provided the applicable disclaimer and waiver with respect thereto, <u>Excluded Information</u> ). Without limiting the foregoing, none of Parent, the Company or the Shareholders may (x) rely upon any Excluded Information to form the basis of any failure to satisfy any

condition set forth in <u>Section 7.3</u> (including <u>Section 7.3(a)</u> or <u>Section 7.3(b)</u>), (y) rely upon any Excluded Information to form the basis of any right of termination pursuant to <u>Section 8.1</u> (including <u>Section 8.1(e)</u>), or (z) make a claim for indemnification pursuant to this <u>Article IX</u> related to any Excluded Information.

# 9.7 Nature of Shareholder Indemnification Obligations.

(a) <u>Obligations Regarding the Company</u>. The obligations of the Shareholders to indemnify the Parent Indemnified Parties pursuant to this <u>Article</u> <u>IX</u> shall be several and not joint to the extent the obligation arises pursuant to <u>Section 9.2(a)(i)</u>, <u>Section 9.2(a)(ii)</u> or <u>Section 9.2(a)(iii)</u> with respect to any misrepresentation or breach of any representation or warranty made or deemed made by the Company, any failure to perform any covenant or agreement of the Company or the Audited Company Financial Statements, respectively, in either case such that each Shareholder s obligation therefor shall be pro rata with respect to the Equity Interests of the Company as set forth on <u>Exhibit A</u> hereto; <u>provided</u>, <u>however</u>, that (i) such obligations of the Controlling Trusts shall be joint and several obligations of the Controlling Trusts and not pro rata and (ii) the Controlling Trusts shall jointly and severally unconditionally guarantee for the benefit of the Parent Indemnified Parties the payment of all such obligations of any Shareholder (regardless of whether such Shareholder is or becomes a Signatory Shareholder).

(b) <u>Obligations Regarding a Shareholder</u>. The obligations of the Shareholders to indemnify the Parent Indemnified Parties pursuant to this <u>Article IX</u> shall be several and not joint to the extent the obligation arises pursuant to <u>Section 9.2(a)(i)</u> or <u>Section 9.2(a)(ii)</u> with respect to any misrepresentation or breach of any representation or warranty made or deemed made by a Shareholder or any failure to perform any covenant or agreement of a Shareholder, in either case such that obligation therefor shall be a liability of the Shareholder making such representation or warranty or failing to perform any covenant or agreement, respectively, giving rise to the obligation; <u>provided</u>, <u>however</u>, that any such obligations of any Shareholder shall be joint and several obligations of the Controlling Trusts with such Shareholder notwithstanding any Liability therefor otherwise.

9.8 <u>Sole and Exclusive Remedy</u>. Except as otherwise provided below, the indemnification provided for in this <u>Article VII</u>, subject to the limitations set forth herein or therein, shall be the exclusive post-Closing remedy available to any Indemnified Party in connection with any Losses arising out of the matters set forth in this Agreement or the transactions contemplated hereby; <u>provided</u>, <u>however</u>, that nothing herein will limit in any way any such Indemnified Party s (A) remedies in respect of fraud or willful breach of any representation, warranty, covenant or agreement herein, or (B) rights hereunder to injunctive or other equitable relief to enforce its rights under this Agreement or in connection with the transactions contemplated hereby.

# ARTICLE X

# DEFINITIONS, CONSTRUCTION, ETC.

10.1 <u>Certain Definitions</u>. For all purposes of and under this Agreement, the following capitalized terms shall have the following respective meanings:

<u>Acquisition Proposal</u> means, when used with respect to any Party, any offer or proposal to enter into or consummate an Acquisition Transaction with respect to such Party.

<u>Acquisition Transaction</u> means, when used with respect to any Party, any transaction or series of related transactions (other than the transactions contemplated by this Agreement) resulting in: (i) any acquisition or purchase from such Party by any Person or group (as defined in or under Section 13(d) of the Exchange Act), directly or indirectly, of more than twenty percent (20%) of the total outstanding voting securities of such Party or any of its Subsidiaries, or any tender offer or exchange offer that, if consummated, would result in the Person or group (as defined in or under Section 13(d) of the Exchange Act) making such offer beneficially owning twenty percent (20%) or more of the total outstanding voting securities of such Party or any of its Subsidiaries; (ii) any merger, consolidation, business combination or other similar transaction involving such Party or any of

its Subsidiaries pursuant to which the shareholders of such Party immediately prior to the consummation of such transaction would hold less than eighty percent (80%) of the equity interests in the surviving or resulting entity of such transaction immediately after consummation thereof; (iii) any sale, exchange, transfer, license, acquisition or disposition (in each case other than in the ordinary course of business) of more than twenty percent (20%) of the assets of the Party and/or its Subsidiaries taken on a consolidated basis (measured by either book or fair market value thereof); or (iv) any liquidation, dissolution, recapitalization or other significant corporate reorganization of the Party; provided, however, that any reference to an Acquisition Transaction with respect to the Company shall also include the issuance of any shares of Equity Interests of the Company or any of its Subsidiaries or any transaction which would be prohibited pursuant to <u>Section 6.3(d)</u> if consummated prior to the termination of this Agreement or the Effective Time.

<u>Affiliate</u> means, with respect to the Person to which it refers, any other Person which directly or indirectly controls, is controlled by or is under common control with such Person. For purposes of the immediately preceding sentence, the term control (including, with correlative meanings, the terms controlling, controlled by and under common control with ), as used with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities, by Contract or otherwise.

<u>Applicable Law</u> means, with respect to any Person, (i) any foreign, national, federal, provincial, state, municipal or local law, statute, constitution, principle of common law, resolution, ordinance, code, edict, treaty, decree, rule, regulation, ruling or other similar requirement enacted, adopted, promulgated, applied or otherwise put into effect by or under the authority of a Governmental Entity that is binding upon or applicable to such Person, assets or set of facts, and (ii) the rules, regulations and listing agreements of any national securities exchange or automated quotation system on which securities of such Person are traded or listed.

<u>Business Day</u> means any day, other than a Saturday, Sunday and any day which is a legal holiday under the laws of the State of New York or is a day on which banking institutions located in New York are authorized or required by Applicable Law to close.

<u>Calculated Purchase Price</u> means the amount equal to (x) the total number of Parent Shares contemplated to be issued by Section 1.7(c) multiplied by (y) the Fixed Parent Share Price.

CERCLA means the Comprehensive Environmental Response, Compensation and Liability Act of 1980.

COBRA means the Consolidated Omnibus Budget Reconciliation Act of 1985.

<u>Code</u> means the United States Internal Revenue Code of 1986.

<u>Company Credit Agreement</u> means the Revolving Credit Agreement, dated September 17, 2007, by and among the financial institutions from time to time signatory thereto, Comerica Bank and the Company, including any waivers thereto with respect to the transactions contemplated hereby.

Company Employee Plan means any Employee Plan of the Company.

<u>Company Intellectual Property</u> means any Intellectual Property that is used, or held for use in the business of the Company or any of its Subsidiaries as currently conducted or as currently proposed to be conducted.

Company Material Adverse Effect means a Material Adverse Effect on the Company.

<u>Consent</u> means any approval, consent (including a negative consent), order, waiver, filing, authorization, license, sublicense, permit, notice, report of, or filing or registration with, or notification to, any Person or similar item.

<u>Contract</u> means any legally binding agreement, contract, mortgage, indenture, lease, sublease, license, sublicense, understanding, arrangement, instrument, note, guaranty, indemnity, representation, warranty, franchise, letter of intent, deed, assignment, power of attorney, certificate, purchase order, work order, insurance policy, benefit plan, permission, commitment, covenant, assurance, dealer and distributorship agreement, supply agreement, development agreement, joint venture agreement, promotion agreement, partnership agreement or other undertaking of any nature, whether written, oral or implied, and each and every amendment, extension, exhibit, attachment, schedule, addendum, appendix, statement of work, change order, and any other similar instrument or document relating thereto, and including any right or obligation under any of the foregoing.

Controlling Trusts means Ronald N. Tutor Separate Property Trust and the Ronald N. Tutor 2006 QuickGRAT, each a Shareholder.

<u>Debt</u> means, without duplication, with respect to any Person (i) all obligations for borrowed money or extensions of credit (including bank overdrafts and advances); (ii) all obligations evidenced by bonds, debentures, notes or other similar instruments; (iii) all obligations to pay the deferred purchase price of property or services, except trade accounts payable arising in the ordinary course of business; (iv) all obligations as lessee capitalized in accordance with GAAP; (v) all obligations of others secured by a Lien on any asset, whether or not such obligations are assumed; (vi) all obligations, contingent or otherwise, directly or indirectly guaranteeing any obligations of any other Person, all obligations to reimburse the issuer in respect of letters of credit or under performance or surety bonds, or other similar obligations; (vii) all obligations in respect of bankers acceptances and under reverse repurchase agreements; and (viii) all obligations in respect of futures contracts, swaps, other financial contracts and other similar obligations (determined on a net basis as if such contract or obligation was being terminated early on such date).

<u>Decree</u> means any judgment, decision, decree, prohibition, injunction, ruling, writ, assessment or order of any Governmental Entity that is binding on any Person or its property under Applicable Law.

<u>Distributable Property</u> means the properties and other assets identified <u>on Section 10.1 of the Company Sched</u>ule.

<u>DO</u>L means the United States Department of Labor.

<u>Employee Plan</u> of any Person means any plan, program, policy, practice, contract, agreement or other arrangement (written or oral) providing for deferred compensation, profit sharing, bonus, severance, termination pay, performance awards, stock option, share appreciation right or other stock-related awards, fringe benefits, group or individual health, dental, medical, life insurance, survivor benefit or other welfare, pension or other employee benefits or remuneration of any kind, whether formal or informal, funded or unfunded, including each employee benefit plan within the meaning of Section 3(3) of ERISA, whether or not subject to ERISA, which is or has been maintained, contributed to, or required to be contributed to, by such Person or any of its Subsidiaries or ERISA Affiliates for the benefit of any Employee, or pursuant to which such Person or any of its Subsidiaries has or may have any Liability.

<u>Environmental Law</u> means, whenever in effect, any and all Applicable Laws and Permits issued, promulgated or entered into by any Governmental Entity relating to the environment, workplace health or safety, or the treatment, storage, disposal, management, Release or threatened Release of, or exposure to, hazardous materials, substances or waste.

<u>Equity Interests</u> means, with respect to any Person, any of (i) its capital stock, partnership interests (general or limited), limited liability company interests, trust interests or other securities which entitle the holder thereof to participate in the earnings of such Person or to receive dividends or distributions on liquidation, winding up or dissolution of such Person, or to vote for the election of directors or other management of such Person, or to exercise other rights generally afforded to shareholders of a corporation; (ii) any stock appreciation,

phantom stock, profit participation, or similar rights with respect to any capital stock of such Person; or (iii) any security convertible into, exchangeable for or evidencing the right to subscribe for or acquire any security contemplated by clause (i) or clause (ii) hereof, including any common stock, preferred stock, option, warrant, subscription right, purchase right, conversion right, exchange right or other similar Contract.

ERISA means the Employee Retirement Income Security Act of 1974.

<u>ERISA Affiliate</u> means any Person that, together with the Company or the Parent (as applicable) or any of their respective Subsidiaries, would be treated as a single employer under Section 414 of the Code or Section 4001 of ERISA and the regulations thereunder within the past 6 years

Exchange Act means the Securities Exchange Act of 1934.

<u>Fixed Parent Share Price</u> means the average closing price per share of Parent Common Stock as reported on the NYSE Composite Transaction Tape (as reported in The Wall Street Journal, or, if not reported thereby, any other authoritative source) for the last five (5) trading days prior to but not including the Closing Date, as adjusted pursuant to <u>Section 2.4</u>.

<u>GAAP</u> means generally accepted accounting principles effective in the United States.

<u>Government Bid</u> of a Person means any quotation, bid or proposal by such Person or any of its Subsidiaries that, if accepted or awarded, would lead to a Government Contract.

<u>Government Contract</u> of a Person means any Contract that is (a) between such Person or any of its Subsidiaries and a Governmental Entity or (b) is entered into by such Person or any of its Subsidiaries as a subcontractor (at any tier) in connection with a Contract between another entity and a Governmental Entity, in each case for the provision of services by such Person or any of its Subsidiaries for payments in excess of \$5,000,000.

<u>Governmental Entity</u> means any government, any governmental, regulatory or administrative entity or body, department, commission, board, agency or instrumentality, any court, tribunal or judicial body or any other governmental, self-regulatory or quasi-governmental authority of any nature (including any governmental department, division, agency, bureau, office, branch, court, commission, tribunal, or other governmental instrumentality) or any political or other subdivision or part of any of the foregoing, in each case whether federal, state, territorial, commonwealth, province, county, provincial, municipal, district, local or foreign.

<u>Hazardous Material</u> means any material, substance, or waste as to which liability or standards of conduct may be imposed pursuant to any Environmental Law.

<u>Intellectual Property</u> means any or all of the following and all rights in, arising out of, or associated therewith: (i) all United States, international and foreign patents and applications therefor and all reissues, divisions, divisionals, renewals, extensions, provisionals, continuations and continuations-in-part thereof, and all patents, applications, documents and filings claiming priority to or serving as a basis for priority thereof; (ii) all inventions (whether or not patentable), invention disclosures, improvements, trade secrets, proprietary process information, computer software programs (in both source code and object code form), technology, business methods, technical data and customer lists, tangible or intangible proprietary information, and all documentation relating to any of the foregoing; (iii) all copyrights, copyrights registrations and applications therefor throughout the world; (iv) all trade names, logos, common law trademarks and service marks, trademark and service mark registrations and applications therefor throughout the world; (vi) all databases and data collections and all rights therein throughout the world; (vii) all moral and economic rights of authors and inventors, however denominated, throughout the world; (viii) all Web addresses, sites and domain names and numbers; and (ix) any similar or equivalent rights to any of the foregoing anywhere in the world.

<u>IRS</u> means the United States Internal Revenue Service.

<u>Knowledge</u> (and other words of similar import) means the actual knowledge after reasonable inquiry (which, for avoidance of doubt, shall not include accessing the online data room made available by the Company) of (i) when used with respect to the Company, Ronald N. Tutor, Jack Frost, William B. Sparks, John D. Barrett and Josh Randall; (ii) when used with respect to Parent, Robert Band, Kenneth R. Burk, Anthony J. Buccitelli and Susan Mellace; or (iii) when used with respect to any other Person, the executive officers and directors (or other persons holding similar positions) of such Person (or if such Person is a natural person, such natural person), or, if any person named in this definition shall no longer be employed by its employer as of the date hereof or no longer shall have the responsibilities in such capacity as processed as of the date hereof, the person holding the comparable office or job with comparable experience and responsibility of such named person.

<u>Lease</u> means all of the applicable party s right, title and interest in all leases, subleases, licenses, concessions and other agreements (written or oral), pursuant to which such party holds a leasehold or subleasehold estate in, or is otherwise granted the right to use or occupy, the Leased Real Property, including the right to all security deposits and other amounts and instruments deposited thereunder.

<u>Leased Real Property</u> means any land, buildings, structures, improvements, fixtures or other interest in real property which are used or occupied pursuant to a Lease.

<u>Legal Proceeding</u> means any action, cause of action, claim (or counterclaim), demand, suit, litigation, proceeding (public or private), criminal prosecution, audit, citation, summons, subpoena, inquiry, arbitration, mediation, alternative dispute resolution procedure or investigation of any nature, civil, criminal, administrative, regulatory or otherwise, by or before any Governmental Entity or other tribunal, arbitrator or mediator.

<u>Liability</u> means any and all claims, debts, liabilities, obligations and commitments of whatever nature, whether asserted or unasserted, fixed, absolute or contingent, matured or unmatured, accrued or unaccrued, liquidated or unliquidated or due or to become due, and whenever or however arising (including those arising out of any Contract or tort, whether based on negligence, strict liability or otherwise) and whether or not the same would be required by GAAP to be reflected as a liability in financial statements or disclosed in the notes thereto.

<u>Lien</u> means any lien (including any lien for unpaid Taxes), pledge, mortgage, deed of trust, security interest, claim, lease, sublease, license, charge, option, right of first refusal, easement, restriction, condition, reservation, servitude, proxy, voting trust or agreement, transfer restriction under any shareholder or similar agreement, or encumbrance or adverse claims or rights of any nature whatsoever.

<u>Losses</u> shall mean, without duplication for purposes of recovery, claims, losses, liabilities, deficiencies, royalties, damages, diminution of value, debt, interest and penalties, costs and expenses, including reasonable out-of-pocket attorneys fees and expenses and other expenses of investigation and defense.

<u>Material Adverse Effect</u> means, when used with respect to any Party, any fact, circumstance, change or effect that, individually or when taken together with all other facts, circumstances, changes or effects in the aggregate, has had or would be reasonably expected to have or result in a material adverse effect on (a) the business, assets, liabilities, properties, financial condition, or results of operations of such Party and its Subsidiaries, taken as a whole, other than facts, circumstances, changes or effects resulting from: (i) general market, economic or political conditions (including any changes arising out of acts of terrorism or war, weather conditions or other force majeure events) in the jurisdictions in which such Party or any of its Subsidiaries does business that do not disproportionately affect such Party and its Subsidiaries, taken as a whole, in any material respect compared to other Persons operating similar businesses in such jurisdictions; (ii) general market, legal or economic conditions in the engineering or construction industries that do not disproportionately affect such Party and its Subsidiaries, taken as a whole, in any material respect compared to other participants in such industries; (iii) the condition of the financial or securities markets in the countries in which such Party or any of its

Subsidiaries does business that do not disproportionately affect such Party and its Subsidiaries, taken as a whole, in any material respect compared to other Persons in such markets; (iv) any change in the stock price or trading volume of Equity Interests of such Party traded or listed on any national securities exchange or quotation system (but not any circumstance, change or effect that may be underlying such change to the extent such circumstance, change or effect would otherwise constitute a Material Adverse Effect); (v) the pendency of this Agreement or any of the transactions contemplated hereby, including the Merger, or the taking of any action contemplated hereby or expressly permitted by the other Party hereto; or (vi) changes in GAAP, or (b) the ability of such Party (or (x) in the case of Parent or Merger Sub or (y) in the case of the Company, any Shareholder) to perform its obligations pursuant to this Agreement and the other Related Agreements and to consummate the transactions contemplated hereby and thereby in a timely manner.

Material Contract means any of the following:

(i) employment or independent personal services contractor agreement involving compensation in excess of \$250,000 per year;

(ii) lease of personal property having a value individually in excess of \$1,000,000;

(iii) Contract of guaranty to any third party in excess of \$5,000,000;

(iv) Contract containing any covenant limiting the freedom of the Person or any of its Affiliates (for avoidance of doubt for purposes of this clause (iv), determined before or after the Closing) to engage in any line of business or in any geographic territory or to compete with any other Person, or pursuant to which the Person or any of its Affiliates has granted to any other Person any exclusivity to any geographic territory, any customer, or any product or service or expressly limits in any material respect the right of the Person or any of its Affiliates to engage in any activities with respect to its business;

(v) Contract relating to capital expenditures and involving future payments in excess of \$5,000,000 in the aggregate;

(vi) Contract not already fully performed relating to the acquisition or disposition of assets with a value in excess of \$5,000,000 or any interest with a value in excess of \$5,000,000 in any business enterprise outside the ordinary course of the Person s or any of its Subsidiaries businesses;

(vii) Contract relating to the borrowing of money or the extension of credit or evidencing any Debt or securing such Debt with a principal amount in excess of \$5,000,000;

(viii) other than Contracts covered by other clauses, settlement, conciliation or similar Contract with any Governmental Entity or any other Person, in either case pursuant to which the subject Person would reasonably be expected to be required after the date hereof to pay consideration in excess of \$5,000,000;

(ix) any Contract pursuant to which a Person has advanced or loaned any amount to any of its shareholders, officers or directors, other than for the reimbursement of business expenses in ordinary course of business consistent with past practice;

(x) a material joint venture, partnership, strategic alliance or other Contract involving the sharing of profits, losses, costs or liabilities with any Person (including any earn out or other similar payments) that, in terms of size or terms, is not consistent with past practice;

(xi) a Contract with the owner of a project, the performance of which involves payment or the provision of services to or from such subject Person in excess of \$20,000,000 in any year; or

(xii) other Contract that alone, or in connection with other related Contracts, for which such Person or any of its Subsidiaries has aggregate surviving obligations in excess of \$10,000,000 that, in terms of size or terms, is not consistent with past practice.

Multiemployer Plan has the meaning set forth in Section 3(37) of ERISA.

<u>NYS</u>E means the New York Stock Exchange.

<u>Outstanding Common Stock Number</u> means the sum of the total number of shares of Company Common Stock issued and outstanding as of immediately prior to the Effective Time.

Parent Capital Stock means any Parent Common Stock or Parent Preferred Stock.

Parent Employee Plan means any Employee Plan of Parent.

<u>Parent Intellectual Property</u> means any Intellectual Property that is necessary for, used, or held for use in the business of Parent or any of its Subsidiaries as currently conducted or as currently proposed to be conducted.

Parent Material Adverse Effect means a Material Adverse Effect on Parent.

<u>Parent Equity Plans</u> means Parent s 2004 Stock Option and Incentive Plan, Parent s Special Equity Incentive Plan, Parent s Amended And Restated (2004) General Incentive Compensation Plan, Parent s Amended and Restated 2004 Construction Business Unit Incentive Compensation Plan and any predecessor or successor plan to any of the foregoing.

<u>Parent Option</u> means any option or other right to purchase Parent Common Stock, whether issued pursuant to the Parent Equity Plans or otherwise.

<u>Permits</u> means approvals, permits, notifications, licenses, grants, authorizations, Consents, easements, notifications, certificates, approvals, orders and franchises issued, granted, given, required or otherwise made available by or from any Governmental Entity.

<u>Permitted Liens</u> shall mean (a) Liens for Taxes not yet due and payable; (b) Liens of landlords and Liens of carriers, warehousemen, mechanics and materialmen, equipment providers and other like Liens arising in the ordinary course of business for sums not yet due and payable; (c) statutory Liens claimed or held by any Governmental Entity that are related to obligations that are not due or delinquent; (d) restrictions on resale of securities imposed by applicable Legal Requirements; (e) security given in the ordinary course of business to any public utility, Governmental Entity or other statutory or public authority; (f) with respect to leased or licensed personal property and Intellectual Property, the terms and conditions of the lease or license applicable thereto; (g) with respect to real property, zoning, building codes and other land use laws regulating the use or occupancy of such real property or the activities conducted thereon which are imposed by any Governmental Entity having jurisdiction over such real property which are not violated by the current use or occupancy of such real property or the operation of the applicable business; (h) easements, covenants, conditions, restrictions and other similar matters affecting title to real property and other encroachments and title and survey defects that do not or would not materially impair the use or occupancy of such real property in the operation of the applicable business as currently conducted; and (i) other Liens (other than those securing indebtedness) that do not materially interfere with the use or operation of the property subject thereto.

<u>Person</u> means any individual, corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including any limited liability company or joint stock company), firm or other enterprise, association, organization, entity or Governmental Entity, including the media.

<u>Pre-Closing Period</u> means any taxable period or portion thereof ending on or before the Closing Date. If a taxable period begins on or before the Closing Date and ends after the Closing Date, then the portion of the taxable period through the end of the Closing Date shall constitute a Pre-Closing Period.

Prohibited Transaction has the meaning set forth in Section 406 of ERISA and Section 4975 of the Code.

<u>Regulation S-K</u> means Regulation S-K promulgated under the rules and regulations of the SEC.

<u>Regulation S-X</u> means Regulation S-X promulgated under the rules and regulations of the SEC.

<u>Related Agreements</u> means this Agreement, the Shareholders Agreement, the Company Shareholder Written Consent, the Employment Agreement, the Charter Amendment, the Dividend Notes, and any other Contracts, documents, schedules, certifications or other instruments being delivered pursuant to or in connection with this Agreement or the any other Related Agreement.

<u>Release</u> has the meaning set forth in Section 101(22) of CERCLA.

<u>Representative</u> means, when used with respect to any Person, such Person s Affiliates (including Subsidiaries) and such Person s or any of the foregoing Persons respective shareholders, partners, members, managers, directors (or committees thereof), officers, employees, trustees, trust beneficiaries, agents, representatives, advisors (including financial advisors, bankers, consultants, counsel and accountants) and controlling Persons.

<u>Restricted Period</u> means the period from the Effective Time until the fifth (5th) anniversary thereof.

<u>SEC</u> means the Securities and Exchange Commission.

Securities Act means the Securities Act of 1933.

<u>SOX</u> means the Sarbanes-Oxley Act of 2002.

<u>Subsidiary</u> of any Person shall mean (a) a corporation more than fifty percent (50%) of the combined voting power of the outstanding voting stock of which is owned, directly or indirectly, by such Person or by one of more other Subsidiaries of such Person or such Person and one or more other Subsidiaries thereof; (b) a partnership of which such Person, or one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries thereof, directly or indirectly, is the general partner and has the power to direct the policies, management and affairs of such partnership; (c) a limited liability company of which such Person or one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries thereof, directly or indirectly, is the managing member and has the power to direct the policies, management and affairs of such company; (d) any other Person (other than a corporation, partnership or limited liability company) in which such Person, or one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries of such Person, or one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries thereof, directly or indirectly, is the managing member and has the power to direct the policies, management and affairs of such company; (d) any other Person or such Person and one or more other Subsidiaries of such Person or such Person or such Person and one or more other Subsidiaries of such Person or such Person and one or more other Subsidiaries of such Person or such P

<u>Superior Proposal</u> shall mean any *bona fide* Acquisition Proposal (it being understood that, solely for purposes of this definition, references to twenty percent (20%) and eighty percent (80%) in the definition of

Acquisition Transaction shall be deemed to be replaced with forty-five percent (45%) and fifty-five percent (55%) for purposes of the reference to the Acquisition Proposal contemplated thereby) relating to an Acquisition Transaction that the Parent Board shall have reasonably determined in good faith (after consultation with a financial advisor of nationally recognized standing and its outside legal counsel, and after taking into account the financial, legal and regulatory aspects of such Acquisition Transaction, strategic considerations relating to the Parent (including whether such Acquisition Transaction is more favorable from a long-term strategic standpoint), the identity of the Person making the Acquisition Proposal, the conditions and likelihood of consummation of the Acquisition Transaction and the transactions contemplated hereby, any revisions to the terms hereof proposed by the Company and the Shareholders pursuant to clause (B) of Section 6.5(b) and any other factors deemed in good faith to be appropriate thereto, is more favorable to the Parent Shareholders (in their capacity as such) than the transactions contemplated by this Agreement.

<u>Tax</u> means all (i) United States federal, state or local or non United States taxes, assessments, charges, duties, levies or other similar governmental charges of any nature, including all income, franchise, profits, capital gains, capital stock, transfer, sales, use, occupation, property, excise, severance, windfall profits, stamp, stamp duty reserve, license, payroll, withholding, ad valorem, value added, alternative minimum, environmental, customs, social security (or similar), unemployment, sick pay, disability, registration and other taxes, assessments, charges, duties, fees, levies or other similar governmental charges of any kind whatsoever, whether disputed or not, together with all estimated taxes, deficiency assessments, additions to tax, penalties and interest; (ii) any liability for the payment of any amount of a type described in clause (i) arising as a result of being or having been a member of any consolidated, combined, unitary or other group or being or having been included or required to be included in any Tax Return related thereto; and (iii) any liability for the payment of any amount of a type described in clause (i) or clause (ii) as a result of any obligation to indemnify or otherwise assume or succeed to the liability of any other Person.

<u>Tax Returns</u> means all returns, declarations, reports, claims for refund, information statements and other documents relating to Taxes, including all schedules and attachments thereto, and including all amendments thereof.

10.2 <u>Index of Defined Terms</u>. The following capitalized terms shall have the respective meanings ascribed thereto in the respective sections of this Agreement set forth opposite each of the capitalized terms below:

Term	<b>Reference</b>
Acquisition Agreement	Section 6.5(a)
Acquisition Proposal	Section 10.1
Acquisition Transaction	Section 10.1
Affiliate	Section 10.1
Agreement	Preamble
Antitrust Laws	Section 3.5
Applicable Law Applicable Period Articles of Amendment	Section 5.5 Section 10.1 Section 6.5(b) Section 1.9
Audited Company Financial Statements	Section 7.2(j)
Available Cash	Section 6.20
Basket	Section 9.3(c)
Business Day	Section 10.1
Calculated Purchase Price	Section 10.1
California Secretary of State	Section 1.3
Cash Dividend	Section 6.20
CERCLA	Section 10.1
Certificate of Merger	Section 1.3
CGCL	Recitals

Term	Reference
Charter Amendment	Section 1.9
Closing	Section 1.2
Closing Date	Section 1.2
COBRĂ	Section 10.1
Code	Section 10.1
Company	Preamble
Company Common Stock	Section 1.7(b)
Company Credit Agreement	Section 10.1
Company Employee Plan	Section 10.1
Company Financial Statements	Section 3.7(a)
Company Improvements	Section 3.15(d)
Company Intellectual Property	Section 10.1
Company Leased Real Property	Section 3.15(b)(i)
Company Leases	Section 3.15(b)(i)
Company Material Adverse Effect	Section 10.1
Company Material Contract	Section 3.13(a)
Company Owned Real Property	Section 3.15(a)
Company Real Property	Section 3.15(b)(i)
Company Schedule	Section 10.3(k)
Company Shareholder Written Consent	Recitals
Company Tax Returns	Section 3.19(a)
Competing Operation	Section 6.16(a)
Confidentiality Agreement	Section 6.8(c)
Consent	Section 10.1
Contract	Section 10.1
Controlling Trusts	Section 10.1
Debt	Section 10.1
Decree	Section 10.1
Dissenting Shares	Section 2.2
Distributable Property	Section 10.1
Dividend Note	Section 6.20
DOJ	Section 6.7(b)
DOL	Section 10.1
Effective Time	Section 1.3
Employee Plan	Section 10.1
Employment Agreement	Recitals
Environmental Law	Section 10.1
	Section 10.1
Equity Interest ERISA	Section 10.1
ERISA ERISA Affiliate	Section 10.1
Exchange Act	Section 10.1
Exchange Agent	Section 2.1(a)
Exchange Fund	Section 2.1(a) Section $1.7(a)$
Exchange Ratio Excluded Information	Section 1.7(c) Section 9.6
Excluded Information Extended Termination Date	
	Section 8.1(b)
Fixed Parent Share Price FTC	Section 10.1
	Section 6.7(b)
GAAP Coursement Bid	Section 10.1 Section 10.1
Government Bid	
Government Contract	Section 10.1

Ferm	Reference
Governmental Entity	Section 10.1
Hazardous Material	Section 10.1
HSR Act	Section 6.7(b)
Indemnified Party	Section 9.4(a)
Indemnifying Party	Section 9.4(a)
Initial Survival Termination Date	Section 9.1(a)
Initial Termination Date	Section 8.1(b)
Intellectual Property	Section 10.1
IRS	Section 10.1
Knowledge	Section 10.1
Lease	Section 10.1
Leased Real Property	Section 10.1
Legal Proceeding	Section 10.1
Liability	Section 10.1
Liability Cap Amount	Section 9.3(a)
Lien	Section 10.1
Losses	Section 10.1
Massachusetts Secretary of State	Section 1.9
Material Adverse Effect	Section 10.1
Material Contract	Section 10.1
Merger	Recitals
Merger Sub	Preamble
Minimum Claim Amount	Section 9.3(c)
Most Recent Company Balance Sheet	Section 3.7(a)
Most Recent Company Balance Sheet Most Recent Company Fiscal Year End	Section 3.7(a)
Most Recent Parent Audited Balance Sheet	Section 5.7(a) Section 5.7(b)
Most Recent Parent Fiscal Year End	Section 5.7(b)
Multiemployer Plan	Section 10.1
Note Dividend	Section 6.20
NYSE	Section 10.1
	Section 10.1
Outstanding Common Stock Number Parent	Preamble
Parent Board	Recitals
Parent Capital Stock	Section 10.1
Parent Change of Recommendation	Section 6.5(a)
Parent Common Stock	Section 1.7(c)
Parent Employee Plan	Section 10.1
Parent Equity Plans	Section 10.1
Parent Financial Statements	Section 5.7(b)
Parent Improvements	Section 5.15(d)
Parent Indemnified Parties	Section 9.2(a)
Parent Intellectual Property	Section 10.1
Parent Leased Real Property	Section 5.15(b)(i)
Parent Leases	Section 5.15(b)(i)
Parent Material Adverse Effect	Section 10.1
Parent Material Contract	Section 5.13(a)
Parent Option	Section 10.1
Parent Owned Real Property	Section 5.15(a)
	Section 5.6(a)
Parent Preferred Stock Parent Real Property Parent RSU	Section 5.15(b)(i) Section 5.6(b)

Term	Reference
Parent Schedule	Section 10.3(k)
Parent SEC Report	Section 5.7(a)
Parent Shareholder Approval	Section 5.25
Parent Shareholders	Section 6.6(e)
Parent Shareholders Meeting	Section 6.6(e)
Parent Shares	Section 1.7(c)
Parent Tax Returns	Section 5.19(a)
Party	Preamble
Payment	Section 3.23(a)
Permit	Section 10.1
Permitted Lien	Section 10.1
Person	Section 10.1
Pre-Closing Period	Section 10.1
Pre-Closing Taxes	Section 10.1
Preferred Bidder Status	Section 3.14(g)
Prohibited Transaction	Section 10.1
Proxy Statement	Section 6.6(a)
Qualifying Parent Change of Recommendation	Section 6.5(b)
Regulation S-K	Section 10.1
Regulation S-X	Section 10.1
Related Agreements	Section 10.1
Release	Section 10.1
Representatives	Section 10.1
Restricted Period	Section 10.1
Restricted Territory	Section 6.16(a)
SEC	Section 10.1
Securities Act	Section 10.1
Share Issuance	Section 5.25
Shareholder Indemnified Parties	Section 9.2(b)
Shareholder Representative	Preamble
Shareholders	Preamble
Shareholders Agreement	Recitals
Signatory Shareholders	Preamble
SOX	Section 10.1
Special Committee	Recitals
Subject Shares	Section 4.1
Subsidiary	Section 10.1
Superior Proposal	Section 10.1
Surviving Entity	Section 1.1
Takeover Law	Section 3.26
Tax	Section 10.1
Tax Return	Section 10.1
Termination Date	Section 8.1(b)
Third Party Claim	Section 9.4(a)
Unaudited Company Financial Statements	Section 3.7(a)
Unlawful Payment	Section 3.14(f)
WARN Act	Section 3.20

10.3 Construction. Unless otherwise indicated to the contrary:

(a) The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms. Whenever the context may require, any pronouns used herein shall include the corresponding masculine, feminine or neuter forms, and the singular form of names and pronouns shall include the plural and vice versa.

(b) The words include and including, and other words of similar import when used herein shall not be deemed to be terms of limitation but rather shall be deemed to be followed in each case by the words without limitation.

(c) The word if and other words of similar import when used herein shall be deemed in each case to be followed by the phrase and only if.

(d) Except as otherwise indicated, all references in this Agreement to Articles, Sections, Annexes, Exhibits and Schedules are intended to re to Articles, Sections, Annexes, Exhibits and Schedules of or to this Agreement, as applicable.

(e) The words herein, hereto, and hereby and other words of similar import in this Agreement shall be deemed in each case to refer to this Agreement as a whole and not to any particular Article, Section or other subdivision of this Agreement.

(f) Any reference herein to dollars or \$ shall mean United States dollars.

(g) The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.

(h) Any reference herein to Applicable Law shall be deemed to include reference to all Applicable Law, and any reference herein to a specific Applicable Law (or, with respect to any statute, ordinance, code, rule or regulation, any provision thereof) shall be deemed to include reference to such Applicable Law and any Applicable Law promulgated thereunder (or provision thereof, as applicable), in either case including any successor(s) thereto, respectively, and as may be amended from time to time. Any reference herein to a Governmental Entity shall be deemed to include reference to any successor thereto.

(i) The Parties agree that they have been represented by counsel during the negotiation and execution of this Agreement and the other Related Agreements and have participated jointly in the negotiation and drafting of this Agreement and the other Related Agreements and hereby waive the application of any law, regulation, holding, rule of construction or other Applicable Law providing that ambiguities in an agreement or other document will be construed against the party drafting such agreement or document. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement or any other Related Agreement.

(j) Whenever this Agreement requires the disclosure of an agreement on the Company Schedule or the Parent Schedule or the delivery to a Party of an agreement, that disclosure requirement or delivery requirement, as applicable, shall also require the disclosure or delivery of each and every amendment, extension, exhibit, attachment, schedule, addendum, appendix, statement of work, change order, and any other similar instrument or document relating to that agreement.

(k) The Company shall deliver to Parent a disclosure schedule (the <u>Company Schedule</u>) setting forth certain exceptions to, and providing certain required supplementary information with respect to, the Company s and the Shareholders representations, warranties, covenants and agreements hereunder in connection with the

execution and delivery of this Agreement, which shall be arranged in order corresponding to the appropriate provisions of this Agreement. Parent shall deliver to the Company a disclosure schedule (the <u>Parent Schedule</u>) setting forth certain exceptions to, and providing certain required supplementary information with respect to, Parent s representations, warranties, covenants and agreements hereunder in connection with the execution and delivery of this Agreement, which shall be arranged in order corresponding to the appropriate provisions of this Agreement. Neither the Company nor Parent shall have any right to update or supplement the Company Schedule or the Parent Schedule, respectively, without the written consent of the other Party, which may be withheld or delayed at its sole discretion. Any such updates or supplements thereto made shall be disregarded for all purposes until such written consent is delivered, if at all. All capitalized terms not defined in the Company Schedule or the Parent Schedule shall have the meanings ascribed to them in this Agreement. The disclosure of any matter in the Company Schedule or the Parent Schedule shall be deemed to be a disclosure for all purposes of this Agreement to which such matter is readily apparent on the face thereof. The listing of any matter in the Company Schedule or the Parent Schedule shall expressly not be deemed to constitute an admission by the Party delivering such schedule, or to otherwise imply, that any such matter is material, is required to be disclosed under this Agreement or falls within relevant minimum thresholds or materiality standards set forth in this Agreement. No disclosure in the Company Schedule or the Parent Schedule relating to any possible breach or violation of any Contract or Applicable Law shall be construed as an admission or indication that any such breach or violation exists or has actually occurred. In no event shall the listing of any matter in the Company Schedule or the Parent Schedule be deemed or interpreted to expand the scope of the Company s, the Shareholders or Parent s representations, warranties, covenants and/or agreements set forth in this Agreement, respectively. All attachments to the Company Schedule or the Parent Schedule are incorporated by reference into such schedule.

(1) The representations, warranties, covenants and obligations of the Parties, and the rights and remedies that may be exercised by the Indemnified Parties pursuant to <u>Section 9.2</u> based on such representations, warranties, covenants and agreements hereunder, will not be limited or affected by any investigation conducted by any Party or any of its Representatives, or any knowledge acquired (or capable of being acquired) by such Party or any of its Representatives at any time, whether before or after the execution and delivery of this Agreement or the Closing, with respect to the accuracy or inaccuracy of or compliance with any such representation, warranty, covenant or agreement, except in each case as otherwise provided in <u>Section 9.6</u>.

(m) For avoidance of doubt, any Knowledge by the Shareholder Representative in his capacity as an officer or director of the Company or any of its Subsidiaries shall not be deemed Knowledge of any of Parent or Merger Sub.

#### ARTICLE XI

#### GENERAL PROVISIONS

11.1 <u>Notices</u>. All notices, requests, demands, claims and other communications hereunder will be in writing except as expressly provided herein. Any notice, request, demand, claim or other communication hereunder shall be deemed duly given (i) when delivered personally or by commercial delivery service to the recipient; (ii) one (1) Business Day after being sent to the recipient by reputable overnight courier service (charges prepaid); (iii) when delivered (with acknowledgment of a complete transmission) if sent by facsimile transmission; or (iv) three (3) Business Days after being mailed to the recipient by certified or registered mail, return receipt requested and postage prepaid. Notices shall be deemed to be properly addressed to any party hereto if addressed to the following addresses (or at such other address for a Party as shall be specified by like notice):

(a) if to Parent or Merger Sub, to:

Perini Corporation 73 Mt. Wayte Avenue Framingham, Massachusetts 01701 Attention: Robert Band, President & Chief Operating Officer Facsimile: (508) 628-2010

with a copy (which shall not constitute notice to Parent or Merger Sub) to:

Kirkland & Ellis LLP 655 Fifteenth Street, N.W. Washington, D.C. 20005-5793 Attention: George P. Stamas, Esq. Facsimile: (202) 879-5200

(b) if to the Company, to:

Tutor-Saliba Corporation 15901 Olden Street Sylmar, California 91342 Attention: Ronald N. Tutor, President & Chief Executive Officer Facsimile: (818) 364-9574

with a copy (which shall not constitute notice to the Company) to each of the following:

Latham & Watkins LLP 633 West Fifth Street, Suite 4000 Los Angeles California 90071-2007 Attention: Mark A. Stegemoeller, Esq.

Steven B. Stokdyk, Esq. Facsimile: (213) 891-8763

Castle & Associates 1925 Century Park East, Suite 210 Los Angeles, California 90067 Attention: Nomi L. Castle, Esq. Facsimile: (310) 286-3404

(c) if to any Shareholder, subject to <u>Section 6.15</u>, to the address of such Shareholder set forth on <u>Exhibit A</u> hereto, with a copy (which shall not constitute notice to the Shareholder) to each of the following:

Latham & Watkins LLP

633 West Fifth Street, Suite 4000

Los Angeles California 90071-2007 Attention: Mark A. Stegemoeller, Esq. Steven B. Stokdyk, Esq. Facsimile: (213) 891-8763

Castle & Associates 1925 Century Park East, Suite 210 Los Angeles California 90067 Attention: Nomi L. Castle, Esq. Facsimile: (310) 286-3404

(d) if to the Shareholder Representative, to the address of the Shareholder Representative set forth on Exhibit A hereto, with a copy (which shall not constitute notice to the Shareholder Representative) to each of the following:

Latham & Watkins LLP 633 West Fifth Street, Suite 4000 Los Angeles California 90071-2007 Attention: Mark A. Stegemoeller, Esq. Steven B. Stokdyk, Esq. Facsimile: (213) 891-8763

Castle & Associates 1925 Century Park East, Suite 210 Los Angeles California 90067 Attention: Nomi L. Castle, Esq. Facsimile: (310) 286-3404

11.2 Entire Agreement. This Agreement, the other Related Agreements and the Confidentiality Agreement constitute the entire agreement between the Parties and supersede any prior understandings, agreements or representations (whether written or oral) by or between the Parties, written or oral, with respect to the subject matter hereof.

11.3 <u>Severability</u>. In the event that any provision of this Agreement or the application thereof becomes or is declared by a court of competent jurisdiction to be illegal, void or unenforceable, the remainder of this Agreement will continue in full force and effect and the application of such provision to other Persons or circumstances will be interpreted so as reasonably to effect the intent of the Parties hereunder. The Parties further agree to replace such void or unenforceable provision of this Agreement with a valid and enforceable provision that will achieve, to the greatest extent possible, the economic, business and other purposes of such void or unenforceable provision.

11.4 <u>Amendment</u>. No amendment of any provision of this Agreement shall be valid unless the same shall be in writing and signed by each Party (subject to <u>Section 6.15</u>) except as expressly provided herein. Subject to Applicable Law and subject to the other provisions of this Agreement, this Agreement may be amended by the Parties at any time by execution of an instrument in writing signed on behalf of each Party (subject to <u>Section 6.15</u>); provided, however, that, in the event that this Agreement has been adopted and approved by the Parent Shareholders in accordance with Applicable Law, no amendment shall be made to this Agreement that requires the approval of such Parent Shareholders without such approval; and provided further that this Agreement may not be amended by Parent at any time without the approval of the Special Committee or, if such committee shall no longer be constituted, a majority of the directors of the Parent Board who are considered independent directors within the meaning of the rules of the NYSE and applicable federal securities laws.

11.5 Extension; Waiver. At any time and from time to time prior to the Effective Time, any Party may, to the extent legally allowed and except as otherwise set forth herein, (a) extend the time for the performance of any of the obligations or other acts of any other Party, as applicable; (b) waive any inaccuracies in the representations and warranties made to such Party hereto contained herein or in any document delivered pursuant hereto; or (c) waive compliance with any of the agreements or conditions for the benefit of such Party providing such extension or waiver by any Party shall be valid unless the same shall be in writing and signed by the Party providing such extension or waiver, nor shall such extension or waiver be deemed to extend to any prior or subsequent default, misrepresentation or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent default, misrepresentation or breach of warranty or covenant. Any delay in exercising any right under this Agreement shall not constitute a waiver of such right. No conditions,

course of dealing or performance, understanding or agreement purporting to modify, vary, explain or supplement the terms or conditions of this Agreement shall be binding unless this Agreement is amended or modified in writing pursuant to the first sentence of <u>Section 11.4</u>.

11.6 <u>Specific Performance</u>. The Parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the Parties shall be entitled to seek an injunction or injunctions to prevent breaches of this Agreement and to seek to enforce specifically the terms and provisions of this Agreement in any state or federal court sitting in New York County, New York, this being in addition to any other remedy to which they are entitled at law, in equity or otherwise.

11.7 <u>Other Remedies</u>. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a Party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or in equity, upon such Party, and the exercise by a Party of any one remedy will not preclude the exercise of any other remedy.

#### 11.8 Successors and Assigns; Assignment; No Third Party Beneficiaries.

(a) This Agreement shall be binding upon each Shareholder that is a natural person and each of its respective personal representatives, executors, administrators, estates, heirs, successors and assigns, if any, and each of Parent, Merger Sub, the Company and each Shareholder that is not a natural person and its respective successors and assigns, if any. This Agreement shall be binding upon and inure to the benefit of the Parties named herein and their respective successors and permitted assigns.

(b) No Party may assign either this Agreement or any of its rights, interests or obligations hereunder without the prior written approval of the other Parties.

(c) Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person other than a Party any rights, interests, benefits or other remedies of any nature under or by reason of this Agreement.

11.9 <u>GOVERNING LAW</u>. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH (A) THE LAWS OF THE STATE OF CALIFORNIA WITH RESPECT TO MATTERS, ISSUES AND QUESTIONS RELATING TO THE MERGER OR THE FIDUCIARY DUTIES OF THE BOARD OF DIRECTORS OR OTHER COMPARABLE GOVERNING BODY AND OFFICERS OR MANAGERS OF THE COMPANY OR MERGER SUB, (B) THE LAWS OF THE STATE OF MASSACHUSETTS WITH RESPECT TO MATTERS, ISSUES AND QUESTIONS RELATING TO THE CHARTER AMENDMENT OR THE FIDUCIARY DUTIES OF THE BOARD OF DIRECTORS OF PARENT, AND (C) THE LAWS OF THE STATE OF NEW YORK WITH RESPECT TO ALL OTHER MATTERS, ISSUES AND QUESTIONS REGARDLESS OF THE LAWS OF THE STATE OF NEW YORK WITH RESPECT TO ALL OTHER MATTERS, ISSUES AND QUESTIONS REGARDLESS OF THE LAWS THAT MIGHT OTHERWISE GOVERN UNDER APPLICABLE PRINCIPLES OF CONFLICT OF LAWS WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICT OF LAW PROVISION OR RULE (WHETHER OF THE STATE OF NEW YORK OR ANY OTHER JURISDICTION) THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY JURISDICTION OTHER THAN THE STATE OF NEW YORK.

11.10 <u>Consent to Jurisdiction: Service of Process</u>. Each of the Parties irrevocably and unconditionally submits to the exclusive jurisdiction of any state or federal court sitting in New York County, New York in any Legal Proceeding arising out of or relating to this Agreement and agrees that all claims in respect of such Legal Proceeding may be heard and determined in any such court. Each Party also agrees not to bring any Legal Proceeding arising out of or relating to this Agreement in any other court. Each of the Parties irrevocably and unconditionally waives any objection to the laying of venue in, and any defense of inconvenient forum to the maintenance of, any Legal Proceeding so brought and waives any bond, surety or other security that might be required of any other Party with respect thereto. Any Party may make service on the other Party by sending or

delivering a copy of the process to the Party to be served at the address and in the manner provided for the giving of notices in <u>Section 11.1</u>; <u>provided</u>, <u>however</u>, that nothing in this <u>Section 11.10</u> shall affect the right of any Party to serve legal process in any other manner permitted by Applicable Law or in equity. Each Party agrees that a final judgment in any Legal Proceeding so brought shall be conclusive and may be enforced by Legal Proceeding or in any other manner provided by law or in equity. The Parties intend that all foreign jurisdictions will enforce any Decree of any state or federal court sitting in New York County, New York in any Legal Proceeding arising out of or relating to this Agreement or any other Related Agreement.

11.11 <u>WAIVER OF JURY TRIAL</u>. EACH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ALL RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDINGS (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY OTHER RELATED AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY.

11.12 <u>Counterparts; Facsimile Delivery</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument and shall become effective when one or more counterparts have been signed by each of the Parties and delivered to the other Parties. Any signature page delivered by facsimile or electronic image transmission shall be binding to the same extent as an original signature page. Any Party that delivers a signature page by facsimile or electronic image transmission shall deliver an original counterpart to any other Party that requests such original counterpart.

{Remainder of page intentionally left blank.}

IN WITNESS WHEREOF, the Parties hereto have executed and delivered this Agreement as of the date first written above.

### Parent:

#### PERINI CORPORATION

By:	/s/ Robert Band
Name:	Robert Band
Title:	President & Chief Operating Officer

## Merger Sub:

# TRIFECTA ACQUISITION LLC

By:	/s/ Robert Band
Name:	Robert Band
Title:	President

#### TRIFECTA ACQUISITION LLC

By:/s/ROSEMARY A. ORTEGAName:Rosemary A. OrtegaTitle:Secretary

#### Company:

#### TUTOR-SALIBA CORPORATION

By:	/s/ Ronald N. Tutor
Name:	Ronald N. Tutor
Title:	President & Chief Executive Officer

#### **TUTOR-SALIBA CORPORATION**

By: /s/ WILLIAM B. SPARKS Name: William B. Sparks Title: Secretary

Shareholder Representative:

/s/ RONALD N. TUTOR Ronald N. Tutor

#### Shareholders:

# RONALD N. TUTOR SEPARATE PROPERTY TRUST

By: Name: Title: /s/ RONALD N. TUTOR Ronald N. Tutor Trustee

# **RONALD N. TUTOR 2006 QUICKGRAT**

By: Name: Title: /s/ RONALD N. TUTOR Ronald N. Tutor Trustee

{Signature page to Agreement and Plan of Merger}

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

Additi	Additional Signatory Shareholders:	
Name:	/s/ JAMES A. FROST James A. Frost	
Name:	/s/ WILLIAM B. SPARKS William B. Sparks	
Name:	/s/ JOHN D. BARRETT John D. Barrett	
Name:	/s/ ROGER K. SEXTON Roger K. Sexton	
Name:	/s/ DAVID L. RANDALL David L. Randall	
Name:	/s/ Gerald W. Brown Gerald W. Brown	
Name:	/s/ ROBERT STEPHEN LEWIS Robert Stephen Lewis	
Name:	/s/ Michael J. Kerchner Michael J. Kerchner	
Name:	/s/ JAMES FOSTER James Foster	
Name:	/s/ Leonard Kaae Leonard Kaae	
Name:	/s/ MARK S. FISHBACH Mark S. Fishbach	
Name:	/s/ Joseph A. Guglielmo Joseph A. Guglielmo	
Name:	/s/ THOMAS E. ANDERSON Thomas E. Anderson	
Name: {Signature page to Agreement and Plan of Me		

	Name:	/s/ Mark Mamczarz Mark Mamczarz
	Name:	/s/ Eric Carlin Eric Carlin
{Signature page to Agreement and Pla	n of Merger}	

Annex AA

**Execution Version** 

#### AMENDMENT NO. 1 TO

#### THE AGREEMENT AND PLAN OF MERGER

THIS AMENDMENT NO. 1 (this <u>Amendment</u>), dated as of May 28, 2008, to the Agreement and Plan of Merger, dated as of April 2, 2008 (the <u>Original Agreement</u>), is entered into by and among **Perini Corporation**, a Massachusetts corporation (<u>Parent</u>), **Trifecta Acquisition LLC**, a California limited liability company and a wholly-owned subsidiary of Parent (<u>Merger Sub</u>), **Tutor-Saliba Corporation**, a California corporation (the <u>Company</u>), **Ronald N. Tutor**, a resident of California and a trustee under each of the Controlling Trusts (as defined in the Original Agreement) (in the capacity as Shareholder Representative as provided in <u>Section 6.15</u> of the Original Agreement or in any other capacity contemplated hereby, the <u>Shareholder Representative</u>), and the shareholders of the Company (including the Shareholder Representative, each a <u>Shareholder</u> and collectively, the <u>Shareholders</u>), by the Shareholder Representative as their respective agent and attorney-in-fact pursuant to Section 6.15(a) of the Original Agreement. All capitalized terms used but not otherwise defined in this Amendment shall have the respective meanings ascribed thereto in the Original Agreement.

#### RECITALS

**WHEREAS**, the Parties entered into the Original Agreement on April 2, 2008, and Section 11.4 of the Original Agreement provides that, subject to Applicable Law and subject to the other provisions of the Original Agreement, the Original Agreement may be amended by the Parties at any time by execution of an instrument in writing signed on behalf of each Party;

**WHEREAS**, the Parties desire to amend the Original Agreement to clarify the understanding of the Parties that the Company will be permitted to make distributions to its Shareholders for taxes as a result of income of the Company from January 1, 2008 through the Closing, on the terms and subject to the conditions set forth herein; and

**WHEREAS**, the board of directors of Parent, acting upon the recommendation of the Special Committee, and the respective boards of directors of Merger Sub and the Company have each duly and validly adopted resolutions (i) determining that this Amendment and the other transactions contemplated hereby are advisable, fair to and in its and its respective shareholders or members , as appropriate, best interests; and (ii) approving the execution and delivery of this Amendment and the consummation of the transactions contemplated hereby.

**NOW, THEREFORE,** in consideration of the foregoing premises and the representations, warranties, covenants and agreements set forth herein, as well as other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, and intending to be legally bound hereby, Parent, Merger Sub, the Company and the Shareholders hereby agree as follows:

#### ARTICLE I

#### AMENDMENTS TO THE ORIGINAL AGREEMENT

The Parties hereby agree to amend the Original Agreement as follows:

1.1 <u>Amendment to Section 10.1 of the Company Schedule of the Original Agreement</u>. Clause 2 of Section 10.1 of the Company Schedule to the Original Agreement is hereby amended and restated in its entirety to read as follows:

A cash distribution of up to \$15,000,000 to the shareholders pro-rata to pay income taxes on the net taxable income of the Company (determined on a basis consistent with net taxable income in the definition of

Final Separate Year Tax Amount included in Section 10.1 of the Merger Agreement) for the period from January 1, 2008 to March 31, 2008; <u>provided</u>, <u>however</u>, that no further distributions of cash as Distributable Property shall be made from and after the date of the Amendment. Such distribution shall not be subject to the restrictions set forth in Section 6.20 of the Agreement.

1.2 <u>Amendment to Add Section 6.21 to the Original Agreement</u>. The Original Agreement is amended to add a new Section 6.21 to read as follows:

#### 6.21 Adjustment for Pre-Closing Taxes.

(a) Notwithstanding the limitations set forth in Section 6.1(b)(ii) or Section 6.1(b)(vii), the Company may declare and pay in cash to the Shareholders as a tax distribution an amount equal to the Final Separate Tax Year Amount within one (1) Business Day of the Closing following the final determination of such amount as determined pursuant to this Section 6.21. Prior to declaring and distributing any amount under this subsection (a), the Company shall prepare and deliver (or shall cause to be prepared and delivered) to Parent within four (4) Business Days of the Closing a statement showing the Company s calculation of the Final Separate Tax Year Amount (the Draft Final Separate Tax Year Amount Statement ), together with such supporting calculations and workpapers with respect thereto and any other documents in connection therewith as Parent shall reasonably request. If Parent does not object to the Draft Final Separate Tax Year Amount Statement within three (3) Business Days after receipt thereof, then the Draft Final Separate Tax Year Amount Statement shall become final and binding on the Parties. The Company shall give Parent and its authorized Representatives reasonable access, during normal business hours, to all Representatives (including attorneys and accountants), personnel, books and records of the Company as reasonably requested by Parent to assist it in its review of the Draft Final Separate Tax Year Amount Statement. Any objection to the Draft Final Separate Tax Year Amount Statement shall be made in writing and shall set forth the basis for such objection in reasonable detail. If Parent objects to the Draft Final Separate Tax Year Amount Statement within three (3) Business Days of the delivery thereof, then Parent and the Company shall negotiate in good faith to resolve promptly any such objection(s). Any resolution of any such objection(s) shall be in writing and shall be conclusive and binding on the Parties. If Parent and the Company do not reach a final resolution within five (5) Business Days after the Company has received Parent s statement of objection(s), then all objections remaining unresolved will be submitted promptly to a nationally-recognized independent accounting firm as is mutually acceptable to Parent and the Shareholder Representative (the <u>Neutral Arbitrator</u>). The resolution of the objections(s) by the Neutral Arbitrator shall be set forth in writing and shall be conclusive and binding upon the Parties. The Parties shall instruct the Neutral Arbitrator to, and take all reasonable actions requested by the Neutral Arbitrator so that it may, finally determine the objections(s) within ten (10) Business Days of the submission of the objection(s) to the Neutral Arbitrator. Parent, on the one hand, and the Shareholders, on the other hand, shall share equally the fees and expenses of the Neutral Arbitrator, which allocation of fees shall be included in any final payment as a result of the resolution of the objection(s). The Draft Final Separate Tax Year Amount Statement, as finally determined hereunder, whether by agreement, deemed agreement or determination by the Neutral Arbitrator, shall be binding on all Parties. Pending final determination of the Draft Final Separate Tax Year Amount Statement, in order that the Closing may occur prior to such determination, upon request of Parent or the Company, the Company within one (1) Business Day following such request shall declare and pay in cash to the Shareholders as a tax distribution an amount equal to the Final Separate Tax Year Amount giving effect to the resolution of all then unresolved objections in favor of Parent, and upon the final determination of the Draft Final Separate Tax Year Amount Statement in the manner contemplated hereby following the Effective Time, Parent shall cause the Company to declare and pay in cash to the Shareholders as a tax distribution the amount, if any, by which the Final Separate Tax Year Amount in the manner contemplated by the other provisions of this Section 6.21(a) exceeds the Final Separate Tax Year Amount used for purposes of the first distribution contemplated by this sentence.

(b) Following completion and filing of all of the Company s Tax Returns that include any Pre-Closing Period, in the event that Parent or the Shareholder Representative determines that (x) the Final

Separate Tax Year Amount as calculated and distributed pursuant to Section 6.21(a) (the <u>Closing Final Separate Tax Year Amount</u>) was different from (y) the amount that would have been determined and distributed by the Company if the Final Separate Tax Year Amount had been calculated using the income reported on all such Company Tax Returns for the Final Separate Tax Year (taking into account the adjustments required by the definition of Final Separate Tax Year Amount) (the <u>Post-Closing Final Separate Tax Year Amount</u>), either Parent or the Shareholder Representative may notify the other of such difference (such notification, the <u>Post-Closing Tax Year Amount Notice</u>) within six (6) months of the filing of such of the Company s Tax Returns. If the recipient of the Post-Closing Tax Year Amount Notice does not object to the calculations therein within ten (10) Business Days of the delivery of such notification, together with all supporting documentation, then the calculation set forth in the Post-Closing Tax Year Amount Notice shall be conclusive and binding on the Parties. If the recipient of the Post-Closing Tax Year Amount Notice timely objects to the calculation therein, then Parent and the Shareholder Representative shall resolve such differences in the manner contemplated for resolving disagreements with respect to the Draft Final Separate Tax Year Amount Statement in Section 6.21(a). The Post-Closing Final Separate Tax Year Amount, as finally determined hereunder, whether by agreement, deemed agreement or determination by the Neutral Arbitrator, shall be binding on all Parties. Within five (5) Business Days of the final determination of the Post-Closing Final Separate Tax Year Amount, (i) if the Post-Closing Final Separate Tax Year Amount exceeds the Closing Final Separate Tax Year Amount, Parent shall cause the Company to pay in cash to the Shareholder Representative to be distributed to all Shareholders the amount of such excess, or (ii) if the Closing Final Separate Tax Year Amount exceeds the Post-Closing Final Separate Tax Year Amount, the Shareholders shall pay in cash to Parent the amount of such excess, or, at the Shareholder Representative s election if delivered to Parent within three (3) days of the final determination, or at the election of Parent if the Shareholders have failed to make payment within thirty (30) calendar days following the final determination, the aggregate principal amount of any Dividend Note(s) shall be reduced by such excess.

1.3 <u>Amendment to Section 10.1 of the Original Agreement</u>. Section 10.1 of the Original Agreement is hereby amended to add the following defined terms, inserted into such Section 10.1 in alphabetical order:

Amendment means Amendment No. 1 to this Agreement, dated as of May 28, 2008.

Assumed Tax Rate means forty percent (40%).

Final Separate Tax Year means the period commencing on and including January 1, 2008 and ending on and including the Closing Date.

<u>Final Separate Tax Year Amount</u> means an amount equal to the product of (i) the aggregate net taxable income of the Company for the Final Separate Tax Year and (ii) the Assumed Tax Rate, less the sum of (x) all amounts paid by the Company in reliance on Clause 2 of Section 10.1 of the Company Schedule prior to the Amendment, which the Company currently represents and warrants equals \$11,571,226, plus (y) all other amounts, if any, previously distributed to the Shareholders by the Company in respect of the Shareholders income tax liabilities arising from inclusion of the Company s Final Separate Tax Year income in their taxable income. For purposes of this definition, the net taxable income of the Company (A) shall exclude any income or gain of the Company realized in respect of the transactions contemplated by this Agreement or otherwise not in the ordinary course of the Closing Date, and (C) shall exclude any gain realized on the distribution of appreciated assets (whether included as part of Distributable Property or otherwise).

1.4 <u>Amendment to Section 10.2 of the Original Agreement</u>. Section 10.2 of the Original Agreement is hereby amended to add the following terms to the list of defined terms:

Term	Reference
Amendment	Section 10.1
Assumed Tax Rate	Section 10.1
Closing Final Separate Tax Year Amount	Section 6.21(b)
Draft Final Separate Tax Year Amount Statement	Section 6.21(a)
Final Separate Tax Year	Section 10.1
Final Separate Tax Year Amount	Section 10.1
Neutral Arbitrator	Section 6.21(a)
Post-Closing Final Separate Tax Year Amount	Section 6.21(b)
Post-Closing Tax Year Amount Notice	Section 6.21(b)

#### ARTICLE II

#### EFFECT OF AMENDMENT; GENERAL PROVISIONS

2.1 <u>Representations and Warranties</u>. Each Party hereby represents and warrants to each other Party as of the date of this Amendment as follows:

(a) Such Party has all requisite power and authority to execute and deliver this Amendment, to consummate the transactions contemplated hereby and to perform its obligations hereunder. The execution and delivery by such Party of this Amendment and the consummation by such Party of the transactions contemplated hereby have been duly authorized by all necessary action on the part of such Party, and, other than for any Shareholder that is a natural person, no additional corporate proceedings on the part of such Party are necessary to authorize this Amendment or the consummation of the transactions contemplated hereby.

(b) This Amendment has been duly executed and delivered by such Party, and, assuming the due authorization, execution and delivery by all other Parties, constitutes a legal, valid and binding obligation of such Party enforceable against such Party in accordance with its terms, except as (i) may be limited by applicable bankruptcy, insolvency, reorganization, moratorium and other similar laws affecting or relating to creditors rights generally, and (ii) is subject to general principles of equity.

2.2 <u>References: Dates</u>. Unless otherwise indicated to the contrary, each reference to hereof, herein, hereunder, hereby and this Agreement sh from and after the date hereof, refer to the Original Agreement as amended by this Amendment. Notwithstanding the foregoing, references to the date of the Original Agreement, as amended hereby, shall in all instances continue to refer to April 2, 2008, references to the date hereof and the date of this Agreement shall continue to refer to April 2, 2008 and references to the date of the Amendment and as of the date of the Amendment shall refer to May 28, 2008.

2.3 <u>Effect of Amendment</u>. This Amendment shall form a part of the Original Agreement for all purposes, and each Party thereto and hereto shall be bound hereby. From and after the execution of this Amendment by the Parties, any reference to the Original Agreement shall be deemed a reference to the Original Agreement as amended hereby.

2.4 <u>No Further Amendment</u>. Except as expressly amended hereby, the Original Agreement and all the terms, conditions, and provisions thereof shall remain in full force and effect. This Amendment is limited precisely as written and shall not be deemed to be an amendment to any other term or condition of the Original Agreement or any of the documents referred to therein. Except as expressly amended by this Amendment, this Amendment does not constitute a waiver of any condition or other provision of the Original Agreement.

2.5 <u>Other Miscellaneous Terms</u>. Section 10.3 of the Original Agreement (other than clauses (j), (k) or (l) thereof) and Article XI of the Original Agreement are each hereby incorporated by reference into this Amendment and shall each apply to this Amendment equally as if set forth fully herein with respect to this Amendment.

{Remainder of page intentionally left blank.}

IN WITNESS WHEREOF, the Parties hereto have executed and delivered this Amendment as of the date first written above.

Parent:

# PERINI CORPORATION

By: /s/ ROBERT BAND Name: Robert Band

Title: President & Chief Operating Officer

Merger Sub:

# TRIFECTA ACQUISITION LLC

By: /s/ ROBERT BAND Name: Robert Band

Title: President

# TRIFECTA ACQUISITION LLC

By: /s/ ROSEMARY A. ORTEGA Name: Rosemary A. Ortega

 Title:
 Secretary

 {Signature page to Amendment No. 1 to the Agreement and Plan of Merger}

Company:

## TUTOR-SALIBA CORPORATION

By: /s/ RONALD N. TUTOR Name: Ronald N. Tutor

Title: President & Chief Executive Officer

# TUTOR-SALIBA CORPORATION

By: /s/ WILLIAM B. SPARKS Name: William B. Sparks

Title: Secretary

Shareholder Representative:

/s/ RONALD N. TUTOR Ronald N. Tutor {Signature page to Amendment No. 1 to the Agreement and Plan of Merger}

Shareholders:

RONALD N. TUTOR SEPARATE PROPERTY TRUST RONALD N. TUTOR 2006 QUICKGRAT JAMES A. FROST WILLIAM B. SPARKS JOHN D. BARRETT ROGER K. SEXTON DAVID L. RANDALL GERALD W. BROWN ROBERT STEPHEN LEWIS MICHAEL J. KERCHNER JAMES FOSTER LEONARD KAAE MARK S. FISHBACH JOSEPH A. GUGLIELMO

THOMAS E. ANDERSON

JOHN MCSWEENEY

MARK MAMCZARZ

ERIC CARLIN

By: /s/ RONALD N. TUTOR Name: Ronald N. Tutor

Title: Shareholder Representative, Agent

and Attorney-in-Fact

{Signature page to Amendment No. 1 to the Agreement and Plan of Merger}

AA-8

Annex B

**Execution Version** 

### SHAREHOLDERS AGREEMENT

# BY AND AMONG

# PERINI CORPORATION,

# **RONALD N. TUTOR**

AND

# THE SHAREHOLDERS LISTED ON ANNEX A HERETO

# DATED AS OF APRIL 2, 2008

# TABLE OF CONTENTS

1.	Definitions	B-1
2.	Board of Directors.	B-5
3.	Voting	B-7
4.	Standstill Agreement	B-7
5.	Transfer Restrictions	B-9
6.	Registration Rights	B-9
7.	Representations and Warranties	B-16
8.	Legend	B-17
9.	Amendment and Waiver	B-17
10.	Shareholder Representative	B-18
11.	<u>Severability</u>	B-18
12.	Successors and Assigns	B-18
13.	Counterparts	B-18
14.	Remedies	B-18
15.	Notices	B-18
16.	Governing Law; Jurisdiction	B-19
17.	Waiver of Jury Trial	B-19
18.	Business Days	B-19
19.	Construction	B-19
20.	No Third Party Beneficiaries	B-19
21.	Delivery by Facsimile or Email	B-19
22.	Effectiveness	B-19

#### SHAREHOLDERS AGREEMENT

THIS SHAREHOLDERS AGREEMENT (this <u>Agreement</u>) is made as of April 2, 2008 by and among **Perini Corporation**, a Massachusetts corporation (the <u>Company</u>), **Ronald N. Tutor**, a resident of California, in the capacity as the initial Shareholder Representative as provided in <u>Section 10</u> or in any other capacity contemplated hereby, and the shareholders listed on the schedule of shareholders (the <u>Schedule of</u> <u>Shareholders</u>), attached as <u>Exhibit</u> A hereto (including the Shareholder Representative, each <u>a</u> <u>Shareholder</u> and collectively, <u>the</u> <u>Shareholders</u>). The Company, the Shareholder Representative and the Shareholders are referred to herein each individually as a <u>Party</u> and collectively as the <u>Parties</u>.

WHEREAS, immediately following the merger contemplated by the Merger Agreement (the <u>Merger</u>), the Shareholders shall Beneficially Own shares of the Company Common Stock (as defined below) as provided in the Merger Agreement;

WHEREAS, the Parties hereto desire to enter into this Agreement to establish certain arrangements with respect to the Company Common Stock that will be Beneficially Owned by the Shareholders following the Merger, as well as restrictions on certain activities in respect of the Company Common Stock, corporate governance and other related corporate matters; and

WHEREAS, the Merger Agreement contemplates that this Agreement will be executed concurrently with the execution of the Merger Agreement and will become effective upon the Closing.

NOW, THEREFORE, in consideration of the mutual covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties to this Agreement hereby agree as follows:

1. <u>Definitions</u>. As used in this Agreement, the following terms shall have the respective meanings ascribed to them in this <u>Section 1</u>, and capitalized terms that are used but not defined herein shall have the respective meanings ascribed to them in that certain Agreement and Plan of Merger, dated as of the date hereof, by and among, the Company, Tutor-Saliba Corporation, a California corporation (<u>Tutor-Saliba</u>), and the other parties thereto (the <u>Merger Agreement</u>):

<u>Affiliate</u> of any Person means any other Person, directly or indirectly controlling, controlled by or under common control with such Person, where control means the possession, directly or indirectly, of the power to direct the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. For the avoidance of doubt, no Management Shareholder shall be deemed to be an Affiliate of Ronald N. Tutor or any member of the Shareholder Representative Group.

Agreement has the meaning set forth in the preamble hereto.

<u>Applicable Law</u> means, with respect to any Person, (i) any foreign, national, federal, provincial, state, municipal or local law, statute, constitution, principle of common law, resolution, ordinance, code, edict, treaty, decree, rule, regulation, ruling or other similar requirement enacted, adopted, promulgated, applied or otherwise put into effect by or under the authority of a Governmental Entity that is binding upon or applicable to such Person, assets or set of facts, and (ii) the rules, regulations and listing agreements of any national securities exchange or automated quotation system on which securities of such Person are traded or listed.

<u>Beneficially Own</u> with respect to any securities shall mean having beneficial ownership of such securities (as determined pursuant to Rule 13d-3 under the Exchange Act). The terms <u>Beneficially Own</u>, <u>Beneficially Owned and Beneficially Owning</u> shall have correlative meanings.

<u>Board</u> has the meaning set forth in Section 2.

<u>Business Day</u> means any day, other than a Saturday, Sunday and any day which is a legal holiday under the laws of the State of New York or is a day on which banking institutions located in New York are authorized or required by Applicable Law to close.

<u>CEO Director</u> has the meaning set forth in Section 2(c)(i).

<u>Company</u> has the meaning set forth in the preamble hereto.

<u>Company Common Stock</u> means the common stock of the Company, par value \$1.00 per share, and any other common stock of the Company that may be issued from time to time.

<u>Demand Registration</u> has the meaning set forth in Section 6(a)(i).

Discretionary Shares has the meaning set forth in Section 3(b).

<u>Excess Shares</u> has the meaning set forth in Section 3(b).

Exchange Act means the Securities Exchange Act of 1934, as amended from time to time.

<u>Governing Documents</u> means the certificate of incorporation and the by-laws of the Company (or their equivalents).

<u>Governmental Entity</u> means any government, any governmental, regulatory or administrative entity or body, department, commission, board, agency or instrumentality, any court, tribunal or judicial body or any other governmental, self-regulatory or quasi-governmental authority of any nature (including any governmental department, division, agency, bureau, office, branch, court, commission, tribunal, or other governmental instrumentality) or any political or other subdivision or part of any of the foregoing, in each case whether federal, state, territorial, commonwealth, province, county, provincial, municipal, district, local or foreign.

<u>Group</u> has the meaning assigned to it in Section 13(d)(3) of the Exchange Act.

<u>Holdback Period</u> means, with respect to any registered offering covered by this Agreement, (a) ninety (90) days after and during the ten (10) days before, the effective date of the related Registration Statement or, in the case of a takedown from a shelf registration statement, ninety (90) days after the date of the prospectus supplement filed with the SEC in connection with such takedown and during such prior period (not to exceed ten (10) days) as the Company has given reasonable written notice to the holder of Registrable Securities or (b) such shorter period as the Shareholder Representative, the Company and the underwriter of such offering, if any, shall agree.

<u>Independent Third Party</u> means any Person who, immediately prior to the contemplated transaction, is not a member of the Shareholder Representative Group.

Management Shareholder or Management Shareholders means those Persons noted as Management Shareholders on Exhibit A.

Merger has the meaning set forth in the recitals hereto.

Merger Agreement has the meaning set forth in the preamble to this Section 1.

<u>Nominating and Governance Committee</u> shall mean the Nominating and Governance Committee of the Board or any other committee of the Board that has the responsibility and power for recommending nominees for appointment or election to the Board.

<u>NYSE</u> shall mean the New York Stock Exchange.

<u>Other Directors</u> has the meaning set forth in Section 2(c).

<u>Ownership Cap</u> has the meaning set forth in Section 4(a).

<u>Party</u> and <u>Parties</u> have the meanings set forth in the preamble hereto.

<u>Permitted Transferee</u> means (i) a spouse or lineal descendant (whether natural or adopted), sibling, parent, heir, executor, administrator, testamentary trustee, lifetime trustee or legatee of such Shareholder, (ii) any trust, the trustees of which include only Persons named in clause (i) and the beneficiaries of which include only the Shareholder and one or more Persons named in clause (i), or (iii) if such Shareholder is a trust, the beneficiaries authorized or entitled to receive distributions from such trust.

<u>Person</u> means an individual, a partnership (including a limited partnership), a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization or other entity or a Governmental Entity or any department, agency or political subdivision thereof.

Piggyback Registration has the meaning set forth in Section 6(b)(i).

<u>Registrable Securities</u> means (i) all Company Common Stock, (ii) any other stock or securities that the Shareholders of the Company Common Stock may be entitled to receive, or will have received pursuant to such Shareholders ownership of the Company Common Stock, in lieu of or in addition to the Company Common Stock, or (iii) any equity securities issued or issuable directly or indirectly with respect to the securities referred to in the foregoing clauses (i) or (ii) by way of conversion or exchange thereof or share dividend or share split or in connection with a combination of shares, recapitalization, reclassification, merger, amalgamation, arrangement, consolidation or other reorganization. As to any particular securities constituting Registrable Securities, such securities will cease to be Registrable Securities when (w) they have been effectively registered or qualified for sale by prospectus filed under the Securities Act and disposed of in accordance with the Registration Statement covering therein, (x) they have been sold to the public pursuant to Rule 144 or Rule 145 or other exemption from registration under the Securities Act or (y) they have been acquired by the Company.

<u>Registration Expenses</u> has the meaning set forth in Section 6(d)(i).

<u>Registration Reque</u>st has the meaning set forth in Section 6(a)(i).

Registration Statement means the prospectus and other documents filed with the SEC to effect a registration under the Securities Act.

Schedule of Shareholders has the meaning set forth in the preamble hereto.

<u>SEC</u> means the United States Securities and Exchange Commission.

Securities Act means the Securities Act of 1933, as amended from time to time.

<u>Selling Expenses</u> means all underwriting discounts, selling commissions and transfer taxes applicable to the sale of Registrable Securities hereunder and any other Registration Expenses required by Applicable Law to be paid by a selling Shareholder.

Short-Form Registration has the meaning set forth in Section 6(a)(iii).

Shareholder or Shareholders has the meaning set forth in the preamble hereto.

Shareholder Designees has the meaning set forth in Section 2(c)(i).

Shareholder Representative means Ronald N. Tutor and any successor appointed in accordance with the provisions of Section 10.

<u>Shareholder Representative Group</u> means the Shareholder Representative, Ronald N. Tutor Separate Property Trust, Ronald N. Tutor 2006 QuickGRAT and any other Affiliate of any of the foregoing.

<u>Shareholder Representative Percentage Interest</u> means, as of any date of determination, the percentage of Total Voting Power that is Beneficially Owned by any member of the Shareholder Representative Group and their respective Affiliates, in the aggregate.

Shareholder s Counsel has the meaning set forth in Section 6(d)(ii).

<u>Special Registration</u> means the registration of (a) equity securities and/or options or other rights in respect thereof solely registered on Form S-4 or Form S-8 (or successor form) or (b) shares of equity securities and/or options or other rights in respect thereof to be offered to directors, members of management, employees, consultants, customers, lenders or vendors of the Company or its direct or indirect Subsidiaries or in connection with dividend reinvestment plans.

<u>Standstill Termination Date</u> means the later of (a) the third (3rd) anniversary of the Effective Time and (b) the date on which the aggregate number of shares of Company Common Stock Beneficially Owned by the members of the Shareholder Representative Group is less than twenty percent (20%) of the Total Voting Power as of such date.

<u>Subsidiary</u> means, with respect to any Person, any corporation, limited liability company, partnership, association or other business entity of which (a) if a corporation, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, or (b) if a limited liability company, partnership, association or other business entity, a majority of the limited liability company, partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by any Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons shall be deemed to have a majority ownership interest in a limited liability company, partnership, association or other business entity if such Person or Persons shall be allocated a majority of limited liability company, partnership, association or other business entity gains or losses or shall be or control the managing director or general partner of such limited liability company, partnership, association or other business entity.

<u>Total Voting Power</u> means, at any time, the total number of votes then entitled to be cast by the holders of the outstanding shares of Company Common Stock and any other securities that either (a) are entitled, in the ordinary course, to vote or take action generally in the election of directors (and not solely upon the occurrence and during the continuation of certain specified events), or (b) in instances where Total Voting Power is being measured with respect to a matter to be voted or acted upon on that is other than the election of directors, any other securities then entitled to vote or take action on such matter.

<u>Transfer</u> means a sale, transfer, hypothecation, negotiation, pledge, assignment, encumbrance, grant any option, warrant or other right to purchase, or otherwise disposition of, or entering into any swap or any other agreement or any transaction that transfers, in whole or in part, directly or indirectly, the economic consequence of ownership of the Company Common Stock.

<u>Transfer Restriction Termination Date</u> means the later of (a) the fifth (5th) anniversary of the Effective Time and (b) the date on which the aggregate number of shares of Company Common Stock Beneficially Owned by the members of the Shareholder Representative Group is less than twenty percent (20%) of the Total Voting Power as of such date.

<u>Tutor-Saliba</u> has the meaning set forth in the preamble to this Section 1.

2. Board of Directors.

(a) <u>Board Composition</u>. The members of the Board of Directors of the Company (the <u>Board</u>) shall be nominated and elected (or appointed, as applicable) in accordance with the Governing Documents and the provisions of this Agreement.

(b) Effective Time Board Composition. At the Effective Time (as defined in the Merger Agreement), the Board will be comprised of (i) C.L. Max Nikias and, if the Shareholder Representative shall elect, one other Person to be designated by the Shareholder Representative prior to the Effective Time, who are the Shareholder Designees, (ii) the CEO Director, and (iii) eight (8) additional Other Directors determined and nominated consistent with the manner provided in Section 2(c)(i) (or serving in such capacity immediately prior to the Effective Time). Any such directors serving on the Board immediately prior to the Effective Time who are to be members of the Board as of the Effective Time pursuant to this Section 2(b) shall continue to be directors of the Board in the classes with remaining terms of service in which they are serving at such time, and any other Persons to be appointed or elected as directors to the Board pursuant to this Section 2(b) shall be so appointed or elected in accordance with the Governing Documents among the classes so as to maintain the number of directors in each class as nearly equal as possible.

#### (c) Designees.

(i) The Shareholder Representative shall have the right to designate individuals for nomination for election to the Board, and the Company shall cause such individuals to be nominated for election to the Board as follows: (1) for so long as the Shareholder Representative Percentage Interest equals or exceeds 22.5%, the Shareholder Representative shall be entitled to designate two persons for nomination and election to the Board; and (2) at such time as the Shareholder Representative Percentage Interest falls below 22.5%, but is not less than 11.25%, the Shareholder Representative shall be entitled to designate one person for nomination and election to the Board. Persons designated by the Shareholder Representative in accordance with the foregoing sentence shall be referred to as the <u>Shareholder Designees</u>. In addition, for so long as Ronald N. Tutor serves as the Chief Executive Officer of the Company, he shall be nominated for election to the Board (the <u>CEO Director</u>). The remaining members of the Board shall be nominated by the Nominating and Governance Committee in accordance with the Governing Documents of the Company (or for the composition of the Board as of the Effective Time pursuant to <u>Section 2(b)</u>, selected by the Nominating and Governance Committee of the Company in accordance with the Governing Documents of the Company) (the <u>Other Directors</u>).

(ii) At each meeting of the shareholders of the Company held after the Effective Time at which directors of the Company are to be elected, the Company agrees to nominate for election, and recommend that the shareholders elect, to the Board each Shareholder Designee that the Shareholder Representative is entitled to designate for nomination and election at that time, in accordance with the provisions of <u>Section 2(c)(i)</u>, and, subject to the requirements of <u>Section 2(c)(i)</u>, the CEO Director; <u>provided</u>, <u>however</u>, that notwithstanding anything to the contrary in this <u>Section 2</u>, the Company (and its Board) shall be under no obligation to recommend to the shareholders or vote in favor of a Shareholder Designee to the extent that (1) the Board (or the Nominating and Governance Committee thereof) determines in good faith that the nomination or recommendation of such nominee by the Board (A) would be prohibited by, or cause the composition of the Board as a whole to fail to satisfy, any Applicable Law or any applicable eligibility, listing, or governance standard or requirement of the NYSE (or any other securities exchange on which the Company Common Stock is subsequently listed

or is sought to be listed); or (B) would reasonably be expected to violate the Board s duties under Applicable Law because (I) such nominee is unfit to serve as a director of an NYSE-listed company or (II) service by such nominee as a director would reasonably be expected to violate Applicable Law; or (2) if the Shareholder Representative is not Ronald N. Tutor, in the good faith judgment of the Board (or the Nominating and Governance Committee thereof), in light of the Company s then applicable eligibility criteria for nominees to the Board, such Shareholder Designee lacks suitable professional qualifications or an appropriate level of experience for service as a member of a board of directors of a publicly traded company of the size and stature of the Company or otherwise does not meet such eligibility criteria in any material respect.

(iii) In the event that the Shareholder Representative loses the right to designate to the Board one or more designees provided for in Section 2(c)(i), such designee(s) shall resign immediately upon receiving notice from the Nominating and Corporate Governance Committee of the Board that such committee has identified a replacement director, and, in any event, shall resign no later than 60 days after the Shareholder Representative loses the right to designate such designee(s) to the Board. In such event, the Board seat formerly occupied by such designee shall become a seat for a director to be selected solely by the Nominating and Corporate Governance Committee or the Board. At its option, the Board may fill the vacancy in accordance with the Governing Documents or, subject to the terms of the Governing Documents and Applicable Law, may reduce its size by the number of vacated board seats. In addition, at such time as Ronald N. Tutor ceases to serve as the Chief Executive Officer of the Company, (A) the Shareholder Representative shall request one Shareholder Designee selected by the Shareholder Representative to resign, (B) upon the resignation of such director as contemplated by clause (A) or the resignation of the other Shareholder Designee, the Shareholder Representative may nominate for election Ronald N. Tutor to be a Shareholder Designee to fill such vacancy as provided in Section 2(c)(iv) in lieu of his then existing directorship, which shall be subject to being filled as provided in clause (D) below, (C) if no Shareholder Designee shall have resigned as contemplated by clause (C) within sixty (60) days of his ceasing to serve as the Chief Executive Officer of the Company, then Ronald N. Tutor shall resign, and (D) the Board seat previously held by Ronald N. Tutor as the CEO Director shall become a seat for a director to be selected solely by the Nominating and Corporate Governance Committee of the Board and the provisions of this Section 2(c)(iii) shall apply to the seat that he previously held. To the extent necessary to give effect to the terms of this Section 2(c)(iii) in a manner that complies with Applicable Law, each person who is a Shareholder Designee and the CEO Director shall, as a condition of his or her appointment or nomination (both initially and annually thereafter), be required to provide to the Nominating and Governance Committee a signed, undated letter of resignation that shall be available to be accepted by the Nominating and Governance Committee only when required to give effect to the terms of this Section 2(c)(iii).

(iv) As long as the Shareholder Representative has any right to designate one or more persons for nomination for election to the Board, as specified in Section 2(c)(i), at any time at which a vacancy shall be created on the Board as a result of the death, disability, retirement, resignation, removal or otherwise of such designee, the Shareholder Representative shall be entitled to designate for appointment by the remaining directors of the Company under the Governing Documents an individual to fill such vacancy and to serve as a director on the Board; provided, that such designee shall be subject to satisfying the qualification standards set forth in the proviso to Section 2(c)(ii) to the same extent as a nominee for election to the Board.

(v) Each Shareholder agrees to vote, in person or by proxy, or to act by written consent (if applicable) with respect to, all shares of Company Common Stock owned by it to cause the election of the CEO Director and each of the Shareholder Designees and the Other Directors (in each case in accordance with the requirements, and subject to the limitations, of Section 2(c)(i)) when nominated for election to the Board and to take all other steps within such Shareholder s power to ensure that the composition of the Board is as set forth in this Section 2.

(d) <u>Committees</u>. One Shareholder Designee shall serve on each committee of the Board if and to the extent the Nominating and Governance Committee determines that such designee, to the extent required by Applicable Law or any applicable eligibility, listing, or governance standard or requirement of the NYSE, qualifies as an Independent Director within the meaning of Section 303A.02 of the Listed Company Manual of the NYSE and Rule 10A-3 under the Exchange Act, as applicable, as such rules may be amended from time to time. If the Company Common Stock is subsequently listed, or sought to be listed, on a securities exchange other than the NYSE, this <u>Section 2(d)</u> shall be applied in a manner to ensure that the composition of the Board committees complies with the applicable eligibility, listing and governance standards of such other securities exchange.

3. <u>Voting</u>. At all times prior to the termination or lapsing of the restrictions of <u>Section 4</u> below, the Shareholder Representative and each member of the Shareholder Representative Group party hereto covenants and agrees that, each such party shall, and shall cause each other member of the Shareholder Representative Group to, take all actions so that:

(a) on any action to elect or appoint any directors of the Company, all of the shares of Company Common Stock that are Beneficially Owned by any member of the Shareholder Representative Group shall be voted or action shall be taken in accordance with Section 2(c)(v); and

(b) on any matter to be voted or acted upon by the Company s shareholders other than those contemplated by Section 3(a), (i) the amount of shares of Company Common Stock equal to in the aggregate up to twenty percent (20%) of the Total Voting Power that are Beneficially Owned by any member of the Shareholder Representative Group (the <u>Discretionary Shares</u>) may be voted (or action with respect to such shares may be taken) in the discretion of the Shareholder Representative Group, and (ii) all shares of Company Common Stock that are Beneficially Owned by any member of the Shareholder Representative Group other than the Discretionary Shares (the <u>Excess Shares</u>) shall be voted (or action with respect to such shares shall be taken) for or against such matter in the same proportion as all other shares of Company Common Stock entitled to vote or take action on such matter are voted or action is taken on such matter (determined without inclusion of the voting or taking of action with respect to the Discretionary Shares and prior to giving effect to the voting or taking of action with respect to the Excess Shares); and

(c) the Shareholder Representative and each member of the Shareholder Representative Group party hereto, and to the extent practicable each other member of the Shareholder Representative Group Beneficially Owning any shares of Company Common Stock, shall be present, in person or by proxy, at all meetings of the Shareholders of the Company so that all such securities Beneficially Owned by any member of the Shareholder Representative Group and any of his Affiliates may be counted for the purposes of determining the presence of a quorum at such meeting.

The Shareholder Representative and each member of the Shareholder Representative Group party hereto covenants and agrees that it shall not, and shall not permit any other member of the Shareholder Representative Group to, grant any proxies with respect to the Company Common Stock or subject any of the Company Common Stock to any arrangement with respect to the voting of or taking action with respect to shares of Company Common Stock that are Beneficially Owned by any member of the Shareholder Representative Group other than as provided under this Agreement. In addition, notwithstanding any other provision of this <u>Section 3</u> to the contrary, at all times prior to the termination or lapsing of the restrictions of <u>Section 4</u> below, no member of the Shareholder Representative Group shall vote, or permit to be voted, any shares of Common Stock that are Beneficially Owned by any member of the Shareholder Representative Group in any manner that would be inconsistent with the restrictions contained in this <u>Section 3</u> or <u>Section 4</u>.

4. <u>Standstill Agreement</u>. Except as expressly provided in this Agreement or as otherwise requested or consented to by the Company or required by Applicable Law, or for matters solely among the Shareholder Representative Group, the Shareholder Representative and each member of the Shareholder Representative Group party hereto covenants and agrees that, from and after the date hereof until the Standstill Termination

Date, each such party shall not, and shall cause each other member of the Shareholder Representative Group not to, singly or as part of a partnership, limited partnership, syndicate or other group (as those terms are defined in Section 13(d)(3) of the Exchange Act), directly or indirectly:

(a) acquire, offer to acquire, or agree to acquire, by purchase, gift or otherwise, directly or indirectly, the beneficial ownership of any additional equity securities of the Company (or any warrants, options, or other rights to purchase or acquire, or any securities convertible into, or exchangeable for, any equity securities of the Company) that has or could have the effect of increasing the Beneficial Ownership of the Shareholder Representative Group on a percentage basis in the outstanding Common Stock of the Company above the percentage interest held by the members of the Shareholder Representative Group as of the date of the Closing (<u>Ownership Cap</u>), except pursuant to a stock split, stock dividend, rights offering, recapitalization, reclassification or similar transaction or grant or issuance approved by the Board;

(b) make, or in any way participate, directly or indirectly, in any solicitation of proxies to vote (as such terms are defined in Rule 14a-1 under the Exchange Act), solicit any consent or communicate with or seek to advise or influence any person or entity with respect to the voting of any securities of the Company or become a participant in any election contest (as such terms are defined in the Exchange Act) with respect to the Company;

(c) form, join, encourage or in any way participate in the formation of, any person or group within the meaning of Section 13(d)(3) of the Exchange Act with respect to any shares of Company Common Stock;

(d) grant or agree to grant any proxy or other voting power to any Person other than the Company or other Persons designated by the Company to vote at any meeting of the shareholders of the Company, or deposit any shares of Company Common Stock into a voting trust or subject any shares of Company Common Stock to any arrangement or agreement with respect to the voting thereof;

(e) initiate, propose or otherwise solicit shareholders for the approval of one or more shareholder proposals with respect to the Company as described in Rule 14a-8 under the Exchange Act or otherwise, or induce or attempt to induce any other person to initiate any shareholder proposal;

(f) seek election to or seek to place a representative on the Board or seek removal of any member of the Board;

(g) seek to have called any meeting of the shareholders of the Company;

(h) make any public announcement or proposal whatsoever with respect to, any form of business combination transaction involving the Company (other than the transactions contemplated by the Merger Agreement), including, without limitation, a merger, exchange offer, or sale or liquidation of the Company s assets, or any restructuring, recapitalization or similar transaction with respect to the Company;

(i) seek publicly to have the Company waive, amend or modify any of the provisions contained in this Section 4;

(j) publicly disclose or announce any intention, plan or arrangement to do any of the foregoing; or

(k) advise, assist, instigate or encourage any third party to do any of the foregoing;

<u>provided</u>, <u>however</u>, that this <u>Section 4</u> shall not prohibit or restrict (x) any action taken by the Shareholder Designees as members of the Board in such capacity, or (y) the exercise by any Shareholder of his, her or its voting rights with regard to shares of Company Common Stock to the extent contemplated by <u>Sections 2 or 3</u>.

# 5. Transfer Restrictions.

(a) No Shareholder may Transfer any shares of Company Common Stock (x) with respect to all Shareholders, prior to the six (6) month anniversary of the Effective Time, except to a Permitted Transfere of such proposed transferor or pursuant to a Piggyback Registration, or (y) with respect to any member of the Shareholder Representative Group, following the six (6) month anniversary of the Closing Date and prior to the Transfer Restriction Termination Date, except (i) to a Permitted Transferee of such proposed transferor, (ii) to one or more third parties so long as the Shareholder Representative Group continues to (and at all times) Beneficially Owns at least seventy percent (70%) of the Company Common Stock acquired by the Shareholder Representative Group at the Closing; <u>provided</u>, <u>however</u>, that, notwithstanding the restrictions set forth in this <u>Section 5(a)(y)(ii)</u>, from and after the fifth (5<sup>th</sup>) anniversary of the Effective Time (or as otherwise contemplated by Section 10(i) of the Employment Agreement (as defined in the Merger Agreement)), the members of the Shareholder Representative Group may Transfer shares of Company Common Stock so long as any such Transfers are not directly or indirectly in the form of a Transfer of shares, through one or a series of related transactions, equal to or greater than fifteen percent (15%) of the Total Voting Power to any person or group within the meaning of Section 13(d)(3) of the Exchange Act, or (iii) in a transaction approved by a majority of the directors of the Company other than Shareholder Designees and the CEO Director. Any attempt by any Shareholder to Transfer any Company Common Stock not in compliance with this Agreement shall be null and void, and the Company shall not, and shall cause any transfer agent not to, give effect in the Company s stock records to any such attempted Transfer.

(b) No share of Company Common Stock shall be Transferred pursuant to <u>Section 5(a)</u> to any Permitted Transferee of the applicable Shareholder unless and until such Permitted Transferee shall have agreed in writing, in a manner acceptable in form and substance to the Company, (i) to accept the shares of Company Common Stock Transferred to it subject to the terms and conditions of this Agreement and (ii) to be bound by this Agreement and to agree and acknowledge that such Person shall constitute a Shareholder for all purposes of this Agreement. Each Shareholder is, and will remain, obligated for the performance by any of such party s Permitted Transferees of its obligations hereunder.

(c) This <u>Section 5</u> shall terminate and be of no further force or effect on the first Business Day following the Transfer Restriction Termination Date, provided that such termination shall not relieve any Party of liability for such party s breach of this <u>Section 5</u> prior to such termination. This <u>Section 5</u> shall not apply to shares of Company Common Stock granted or issued to any Shareholder and approved by the Board (or a committee thereof) after the Effective Time; <u>provided</u>, that the Shareholder Designees and the CEO Director shall not have voted on such grants or issuances.

#### 6. Registration Rights.

#### (a) Demand Registrations.

(i) <u>Requests for Registration</u>. The Shareholder Representative may request in writing, on behalf of the Shareholder Representative Group, that the Company effect the registration of all or any part of the Registrable Securities held by the Shareholder Representative Group (a <u>Registration</u> <u>Request</u>) at any time following the six (6) month anniversary of the Effective Time, <u>provide</u>d, that prior to the Transfer Restriction Termination Date, the number of shares of Company Common Stock to be sold by the Shareholder Representative Group pursuant to a Registration Request (together with all shares of Company Common Stock Transferred publicly in accordance with <u>Section 5(a)</u>) shall not exceed thirty percent (30%) of the shares of Company Common Stock acquired by the Shareholder Representative Group upon the consummation of the Closing (except, from and after the fifth (5th) anniversary of the Effective Time, as contemplated by the proviso set forth in <u>Section 5(a)(y)(ii)</u>). Promptly after its receipt of any Registration Request, the Company will give written notice of such request to all other Shareholders, and will use its reasonable best efforts to register, in accordance with the provisions of this Agreement, all Registrable Securities that have been requested to be registered in the Registration Request or by any other Shareholders by written notice to the Company given within fifteen

(15) Business Days after the date the Company has given such Shareholders notice of the Registration Request. The Company will pay all Registration Expenses incurred in connection with any registration pursuant to this <u>Section 6(a)</u>. Any registration requested by the Shareholder Representative pursuant to <u>Section 6(a)(i)</u> or <u>6(a)(ii)</u> is referred to in this Agreement as a <u>Demand Registration</u>.

(ii) Limitation on Demand Registrations. The Shareholder Representative will be entitled to initiate no more than three (3) Demand Registrations (including Short-Form Registrations permitted pursuant to Section 6(a)(iii)). No request for registration will count for the purposes of the limitations in this Section 6(a)(ii) if (i) the Shareholder Representative determines in good faith to withdraw the proposed registration prior to the effectiveness of the Registration Statement relating to such request due to marketing conditions or regulatory reasons relating to the Company, (ii) the Registration Statement relating to such request is not declared effective within one hundred and eighty (180) days of the date such Registration Statement is first filed with the SEC (other than solely by reason of the Shareholder Representative having refused to proceed) and the Shareholder Representative withdraws its Registration Request prior to such Registration Statement being declared effective, (iii) prior to the sale of at least ninety percent (90%) of the Registrable Securities included in the applicable registration relating to such request, such registration is adversely affected by any stop order, injunction or other order or requirement of the SEC or other Governmental Entity or court for any reason and the Company fails to have such stop order, injunction or other order or requirement removed, withdrawn or resolved to the Shareholder Representative s reasonable satisfaction within thirty (30) days of the date of such order, (iv) more than ten percent (10%) of the Registrable Securities requested by the Shareholder Representative to be included in the registration are not so included pursuant to Section 6(a)(vi), or (v) the conditions to closing specified in the underwriting agreement or purchase agreement entered into in connection with the registration relating to such request are not satisfied (other than as a result of a material default or breach thereunder by the Shareholder Representative Group). Notwithstanding the foregoing, the Company will pay all Registration Expenses in connection with any request for registration pursuant to Section 6(a) regardless of whether or not such request counts toward the limitation set forth above.

(iii) <u>Short-Form Registrations</u>. The Company will use its reasonable best efforts to qualify for registration on Form S-3 or any comparable or successor form or forms or any similar short-form registration (<u>Short-Form Registrations</u>), and, if requested by the Shareholder Representative and available to the Company, such Short-Form Registration will be a shelf registration statement providing for the registration of, and the sale on a continuous or delayed basis of the Registrable Securities, pursuant to Rule 415. In no event shall the Company be obligated to effect any shelf registration other than pursuant to a Short-Form Registration. The Company will pay all Registration Expenses incurred in connection with any Short-Form Registration.

(iv) <u>Restrictions on Demand Registrations</u>. If the filing, initial effectiveness or continued use of a registration statement, including a shelf registration statement pursuant to Rule 415, with respect to a Demand Registration would (1) require the Company to make a public disclosure of material non-public information, which disclosure in the good faith judgment of the Board (A) would be required to be made in any Registration Statement so that such Registration Statement would not be materially misleading, (B) would not be required to be made at such time but for the filing, effectiveness or continued use of such Registration Statement and (C) would in the good faith judgment of the Board reasonably be expected to have a material adverse effect on the Company or its business if made at such time, or (2) would in the good faith and judgment of the Board reasonably be expected to have a material adverse effect on the Company or its business or on the Company s ability to effect a planned or proposed acquisition, disposition, financing, reorganization, recapitalization or similar transaction, then the Company may upon giving prompt written notice of such action to the participants in such registration (each of whom hereby agrees to maintain the confidentiality of all information disclosed to such participants) delay the filing or initial effectiveness of, or suspend use of, such

Registration Statement, provided, that the Company shall not be permitted to do so (x) more than three (3) times during any twelve-month period or (y) for periods exceeding, in the aggregate, one hundred twenty-five (125) days during any twelve-month period. In the event the Company exercises its rights under the preceding sentence, such Shareholders agree to suspend, promptly upon their receipt of the notice referred to above, their use of any prospectus relating to such registration in connection with any sale or offer to sell Registrable Securities. If the Company so postpones the filing of a prospectus or the effectiveness of a Registration Statement, the Shareholder Representative will be entitled to withdraw such request and, if such request is withdrawn, such registration request will not count for the purposes of the limitation set forth in Section 6(a)(ii). The Company will pay all Registration Expenses incurred in connection with any such aborted registration or prospectus.

#### (v) Selection of Underwriters.

(A) If the Shareholder Representative intends that the Registrable Securities covered by his Registration Request shall be distributed by means of an underwritten offering, the Shareholder Representative will so advise the Company as a part of the Registration Request, and the Company will include such information in the notice sent by the Company to the other Shareholders with respect to such Registration Request. In such event, the lead underwriter to administer the offering will be chosen by the Shareholder Representative subject to the prior written consent, not to be unreasonably withheld or delayed, of the Company.

(B) If the offering is underwritten, the right of any Shareholder to registration pursuant to <u>Section 6(a)</u> will be conditioned upon such Shareholder s participation in such underwriting and the inclusion of such Shareholder s Registrable Securities in the underwriting, and each such Shareholder will (together with the Company and the other Shareholders distributing their securities through such underwriting) enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting. If any Shareholder disapproves of the terms of the underwriting, such Shareholder may elect to withdraw therefrom by written notice to the Company, the managing underwriter and the Shareholder Representative.

(vi) <u>Priority on Demand Registrations</u>. The Company will not include in any underwritten registration pursuant to <u>Section 6(a)</u> any securities that are not Registrable Securities, without the prior written consent of the Shareholder Representative. If the managing underwriter advises the Company or the Shareholder Representative that in its reasonable opinion the number of Registrable Securities (and, if permitted hereunder, other securities requested to be included in such offering) exceeds the number of securities that can be sold in such offering without adversely affecting the marketability of the offering (including an adverse effect on the per share offering price), the Company or the Shareholder Representative will include in such offering only such number of securities that in the reasonable opinion of such underwriters can be sold without adversely affecting the marketability of the offering (including an adverse effect on the per share offering price), which securities will be so included in the following order of priority: (1) first, Registrable Securities of the Shareholder Representative Group and (2) second, Registrable Securities of any other Shareholders who have delivered written requests for registration pursuant to <u>Section 6(a)(i)</u>, pro rata on the basis of the aggregate number of Registrable Securities owned by each such Shareholder and (3) any other securities of the Company that have been requested to be so included, subject to the terms of this Agreement.

(vii) <u>Effective Registration Statement</u>. A registration requested pursuant to <u>Section 6(a)(i)</u> shall not be deemed to have been effected unless it is declared effective by the SEC and remains effective for the period specified in <u>Section 6(c)(ii)</u>.

# (b) Piggyback Registrations.

(i) <u>Right to Piggyback</u>. Whenever the Company proposes to register any of its securities, other than a registration pursuant to <u>Section 6(a)</u> or a Special Registration, and the registration form to be filed may be used for the registration or qualification for distribution of Registrable Securities, the Company will give prompt written notice (and in any event no later than fifteen (15) Business Days prior to the filing of a Registration Statement with respect to such registration) to all Shareholders of its intention to effect such a registration and, subject to <u>Section 6(b)(iv)</u>, will include in such registration all Registrable Securities with respect to which the Company has received written requests for inclusion therein within ten (10) Business Days after the date of the Company s notice (<u>a Piggyback Registration</u>). Any Shareholder that has made such a written request may withdraw its Registrable Securities from such Piggyback Registration by giving written notice to the Company and the managing underwriter, if any, on or before the fifth (5th) Business Day prior to the planned effective date of such Piggyback Registration. The Company may terminate or withdraw any registration under this <u>Section 6(b)</u> prior to the effectiveness of such registration, whether or not any Shareholder has elected to include Registrable Securities in such registration, and except for the obligation to pay Registration Expenses pursuant to <u>Section 6(b)(iii)</u> the Company will have no liability to any Shareholder in connection with such termination or withdrawal.

(ii) <u>Underwritten Registration</u>. If the registration referred to in <u>Section 6(b)(i)</u> is proposed to be underwritten, the Company will so advise the Shareholders as a part of the written notice given pursuant to <u>Section 6(b)(i)</u>. In such event, the right of any Shareholder to registration pursuant to this <u>Section 6(b)</u> will be conditioned upon such Shareholder s participation in such underwriting and the inclusion of such Shareholder s Registrable Securities in the underwriting, and each such Shareholder will (together with the Company and the other Shareholders and other holders of securities distributing their securities through such underwriting) enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting by the Company. If any Shareholder disapproves of the terms of the underwriting, such Shareholder may elect to withdraw therefrom by written notice to the Company, the managing underwriter and the Shareholder Representative.

(iii) <u>Piggyback Registration Expenses</u>. The Company will pay all Registration Expenses in connection with any Piggyback Registration, whether or not any registration or prospectus becomes effective or final.

(iv) <u>Priority on Primary Registrations</u>. If a Piggyback Registration relates to an underwritten primary offering on behalf of the Company, and the managing underwriters advise the Company that in their reasonable opinion the number of securities requested to be included in such registration exceeds the number which can be sold without adversely affecting the marketability of such offering (including an adverse effect on the per share offering price), the Company will include in such registration or prospectus only such number of securities that in the reasonable opinion of such underwriters can be sold without adversely affecting the marketability of the offering (including an adverse effect on the per share offering price), which securities will be so included in the following order of priority: (1) first, the securities the Company proposes to sell, (2) second, Registrable Securities of the Shareholder Representative Group and (3) third, Registrable Securities of any other Shareholders who have requested registration of Registrable Securities pro rata on the basis of the aggregate number of such securities or shares owned by each such Shareholder.

(c) <u>Registration Procedures</u>. Subject to <u>Section 6(a)(iv)</u>, whenever the Shareholders of Registrable Securities have requested that any Registrable Securities be registered pursuant to <u>Sections 6(a)</u> or <u>6(b)</u> of this Agreement, the Company will use its commercially reasonable efforts to effect the registration and sale of such Registrable Securities as soon as reasonably practicable in accordance with the intended method of disposition thereof and pursuant thereto. The Company shall use its reasonable best efforts to as expeditiously as possible:

(i) prepare and file with the SEC a Registration Statement with respect to such Registrable Securities, make all required filings with the Financial Industry Regulatory Authority and thereafter use

its reasonable best efforts to cause such Registration Statement to become effective as soon as reasonably practicable, provided that before filing a Registration Statement or any amendments or supplements thereto, the Company will, in the case of a Demand Registration, furnish to Shareholders Counsel copies of all such documents proposed to be filed, which documents will be subject to review of such counsel at the Company s expense;

(ii) prepare and file with the SEC such amendments and supplements to such Registration Statement as may be necessary to keep such Registration Statement effective for a period of either (1) not less than (A) three (3) months, if such Registration Statement relates to an underwritten offering, or such longer period as a prospectus is required by Applicable Law to be delivered in connection with sales of Registrable Securities by an underwriter or dealer or (B) three (3) years in the case of shelf registration statements (or in each case such shorter period ending on the date that the securities covered by such shelf registration statement case to constitute Registrable Securities) or (3) such shorter period as will terminate when all of the securities covered by such Registration Statement (but in any event not before the expiration of any longer period required under the Securities Act), and comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such Registration Statement three been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such Registration Statement (but in any event not before the expiration of any longer period required under the Securities Act), and comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such Registration Statement until such time as all of such securities have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such Registration statement;

(iii) furnish to each seller of Registrable Securities such number of copies, without charge, of such Registration Statement, each amendment and supplement thereto, including each preliminary prospectus, final prospectus, any other prospectus (including any prospectus filed under Rule 424, Rule 430A or Rule 430B under the Securities Act and any issuer free writing prospectus as such term is defined under Rule 433 promulgated under the Securities Act), all exhibits and other documents filed therewith and such other documents as such seller may reasonably request including in order to facilitate the disposition of the Registrable Securities owned by such seller;

(iv) register or qualify such Registrable Securities under such other securities or blue sky laws of such jurisdictions as any seller reasonably requests and do any and all other acts and things that may be reasonably necessary or reasonably advisable to enable such seller to consummate the disposition in such jurisdictions of the Registrable Securities owned by such seller (provided that the Company will not be required to (1) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this subsection, (2) subject itself to taxation in any such jurisdiction or (3) consent to general service of process in any such jurisdiction);

(v) notify each seller of such Registrable Securities and Shareholders Counsel, at any time when a prospectus relating thereto is required to be delivered under the Securities Act, upon discovery that, or upon the discovery of the happening of any event as a result of which, the prospectus contains an untrue statement of a material fact or omits any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made, and, as soon as reasonably practicable, prepare and furnish to such seller a reasonable number of copies of a supplement or amendment to such prospectus so that, as thereafter delivered to the purchasers of such Registrable Securities, such prospectus will not contain an untrue statement of a material fact or omit to state any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made;

(vi) notify each seller of any Registrable Securities covered by such Registration Statement and Shareholders Counsel (1) when such Registration Statement or the prospectus or any prospectus supplement or post-effective amendment has been filed and, with respect to such Registration Statement or any post-effective amendment, when the same has become effective, (2) of any request by

the Commission for amendments or supplements to such Registration Statement or to amend or to supplement such prospectus or for additional information, and (3) of the issuance by the Commission of any stop order suspending the effectiveness of such Registration Statement or the initiation of any proceedings for any of such purposes;

(vii) cause all such Registrable Securities to be listed on each securities exchange on which similar securities issued by the Company are then listed or, if no similar securities issued by the Company are then listed on any securities exchange, use its reasonable best efforts to cause all such Registrable Securities to be listed on the New York Stock Exchange or the NASDAQ Stock Market, as determined by the Company;

(viii) provide a transfer agent and registrar for all such Registrable Securities not later than the effective date of such Registration Statement;

(ix) enter into such customary agreements (including underwriting agreements and, subject to Section 6(g), lock-up agreements in customary form, and including provisions with respect to indemnification and contribution in customary form) and take all such other customary actions as the Shareholder Representative, the selling Shareholders or the underwriters, if any, reasonably request in order to expedite or facilitate the disposition of such Registrable Securities (including, without limitation, making members of senior management of the Company available to participate in road show and other customary marketing activities);

(x) make available for inspection by any seller of Registrable Securities and Shareholders Counsel, any underwriter participating in any disposition pursuant to such Registration Statement and any attorney, accountant or other agent retained by any such seller or underwriter, all financial and other records, pertinent corporate documents and documents relating to the business of the Company, and cause the Company s officers, directors, employees and independent accountants to supply all information reasonably requested by any such seller, underwriter, attorney, accountant or agent in connection with such Registration Statement, <u>provided</u> that it shall be a condition to such inspection and receipt of such information that the inspecting Person (1) enter into a confidentiality agreement in form and substance reasonably satisfactory to the Company and (2) agree to minimize the disruption to the Company s business in connection with the foregoing;

(xi) timely provide to its security holders earning statements satisfying the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder;

(xii) in the event of the issuance of any stop order suspending the effectiveness of a Registration Statement, or of any order suspending or preventing the use of any related prospectus or ceasing trading of any securities included in such Registration Statement for sale in any jurisdiction, use every reasonable effort to promptly obtain the withdrawal of such order;

(xiii) obtain one or more comfort letters, addressed to the underwriters, if any, dated the effective date of such Registration Statement and the date of the closing under the underwriting agreement for such offering, signed by the Company s independent public accountants in customary form and covering such matters of the type customarily covered by comfort letters as such underwriters shall reasonably request; and

(xiv) provide legal opinions of the Company s counsel, addressed to the underwriters, if any, dated the date of the closing under the underwriting agreement, with respect to the Registration Statement, each amendment and supplement thereto (including the preliminary prospectus) and such other documents relating thereto as the underwriter shall reasonably request in customary form and covering such matters of the type customarily covered by legal opinions of such nature.

As a condition to registering Registrable Securities, the Company may require each Shareholder of Registrable Securities as to which any registration is being effected to furnish the Company with such information regarding such Shareholder and pertinent to the disclosure requirements relating to the registration and the distribution of such securities as the Company may from time to time reasonably request in writing.

#### (d) Registration Expenses.

(i) Except as otherwise provided in this Agreement, all expenses incidental to the Company s performance of or compliance with this Agreement, including, without limitation, all registration and filing fees, fees and expenses of compliance with securities or blue sky laws, word processing, duplicating and printing expenses, messenger and delivery expenses, and fees and disbursements of counsel for the Company and all independent certified public accountants, underwriters and other Persons retained by the Company (all such expenses, <u>Registration Expenses</u>), will be borne by the Company. The Company will, in any event, pay its internal expenses (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), the expenses of any annual audit or quarterly review, the expenses of any liability insurance and the expenses and fees for listing the securities to be registered on each securities exchange on which similar securities issued by the Company are then listed or on the New York Stock Exchange or the NASDAQ stock market. All Selling Expenses will be borne by the holders of the securities so registered pro rata on the basis of the amount of proceeds from the sale of their shares so registered.

(ii) In connection with each Demand Registration and each Piggyback Registration in which members of the Shareholder Representative Group participate, the Company will reimburse the Shareholder Representative for the reasonable fees and disbursements of one counsel (<u>Shareholders Counsel</u>).

#### (e) Participation in Underwritten Registrations.

(i) No Shareholder may participate in any registration hereunder that is underwritten unless such Shareholder (1) agrees to sell its Registrable Securities on the basis provided in any underwriting arrangements approved by the Shareholder Representative (including, without limitation, pursuant to the terms of any over-allotment or green shoe option requested by the managing underwriter(s), provided that no Shareholder will be required to sell more than the number of Registrable Securities that such Shareholder has requested the Company to include in any registration),
(2) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents reasonably required under the terms of such underwriting arrangements, and (3) cooperates with the Company s reasonable requests in connection with such registration or qualification (it being understood that the Company s failure to perform its obligations hereunder, which failure is caused by such Shareholder s failure to cooperate with such reasonable requests, will not constitute a breach by the Company of this Agreement).

(ii) Each Shareholder that is participating in any registration hereunder agrees that, upon receipt of any notice from the Company of the happening of any event of the kind described in Section 6(c)(v), such Shareholder will forthwith discontinue the disposition of its Registrable Securities pursuant to the Registration Statement until such Shareholder receives copies of a supplemented or amended prospectus as contemplated by such Section 6(c)(v). In the event the Company gives any such notice, the applicable time period mentioned in Section 6(c)(i) during which a Registration Statement is to remain effective will be extended by the number of days during the period from and including the date of the giving of such notice pursuant to this Section 6(e)(i) to and including the date when each seller of a Registrable Security covered by such Registration Statement will have received the copies of the supplemented or amended prospectus contemplated by Section 6(c)(v).

(f) <u>Rule 144</u>. The Company will use its reasonable best efforts to timely file all reports and other documents required to be filed by it under the Securities Act and the Exchange Act and the rules and regulations adopted by the SEC thereunder (or, if the Company is not required to file such reports, it will, upon the request of the Shareholder Representative, make publicly available such information as necessary to permit sales pursuant to Rule 144), and will take such further action as the Shareholder Representative may reasonably request, all to the extent required from time to time to enable the Shareholders to sell shares of Registrable Securities without registration under the Securities Act within the limitation of the exemptions provided by Rule 144. Upon the request of the Shareholder Representative, the Company will deliver to the Shareholder Representative a written statement as to whether it has complied with such information requirements.

(g) <u>Holdback</u>. In consideration for the Company agreeing to its obligations under this Agreement, each Shareholder agrees in connection with any registration of the Company s securities (whether or not such Shareholder is participating in such registration) upon the request of the Company and the underwriters managing any underwritten offering of the Company s securities, not to effect (other than pursuant to such registration) any public sale or distribution of Registrable Securities, including, but not limited to, any sale pursuant to Rule 144 or Rule 144A, or make any short sale of, loan, grant any option for the purchase of, or otherwise dispose of any Registrable Securities, any other equity securities of the Company or any securities convertible into or exchangeable or exercisable for any equity securities of the Company without the prior written consent of the Company or such underwriters, as the case may be, during the Holdback Period, <u>provided</u> that nothing herein will prevent any Shareholder that is a partnership or corporation from making a distribution of Registrable Securities to the partners or shareholders thereof or a Transfer to an Affiliate that is otherwise in compliance with applicable securities laws, so long as such distributees agree to be so bound. With respect to such underwritten offering of Registrable Securities covered by a registration pursuant to <u>Sections 6(a)</u> or <u>6(b)</u>, the Company further agrees not to effect (other than pursuant to such registration or pursuant to a Special Registration) any public sale or any securities convertible into or exchangeable or securities, during the Holdback Period with respect to such underwritten offering, if required by the managing underwriter, <u>provided</u> that notwithstanding anything to the contrary herein, the Company s obligations under this <u>Section 6(g)</u> shall not apply during any twelve-month period for more than an aggregate of ninety (90) days.

7. <u>Representations and Warranties</u>. Each Shareholder (as to himself, herself or itself only) represents and warrants to the Company and each other Shareholder that, as of the date hereof:

(a) upon consummation of the Merger, such Shareholder will be the record owner of the number of the Company shares set forth opposite such Shareholder s name on the Schedule of Shareholders attached as Exhibit A hereto;

(b) this Agreement has been duly authorized, executed and delivered by such Shareholder and constitutes the valid and binding obligation of such Shareholder, enforceable against such Shareholder in accordance with its terms;

(c) such Shareholder has not granted and is not a party to any proxy, voting trust or other agreement which is inconsistent with, conflicts with or violates any provision of this Agreement; and

(d) the execution, delivery and performance by such Shareholder of this Agreement and the consummation by such Shareholder of the transactions contemplated hereby will not, with or without the giving of notice or lapse of time, or both, (i) violate any provision of Applicable Law to which such Shareholder is subject, (ii) violate any order, judgment or decree applicable to such Shareholder or (iii) conflict with, or result in a breach or default under, any term or condition of any agreement or other instrument to which such Shareholder is a party or by which such Shareholder or any of such Shareholder s assets or properties is bound.

#### 8. Legend.

(a) The Shareholders agrees that all certificates or other instruments representing the Company Common Stock subject to this Agreement will bear a legend substantially to the following effect:

(i) THE SECURITIES REPRESENTED BY THIS INSTRUMENT HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR SECURITIES LAWS OF ANY STATE AND MAY NOT BE TRANSFERRED, SOLD OR OTHERWISE DISPOSED OF EXCEPT WHILE A REGISTRATION STATEMENT RELATING THERETO IS IN EFFECT UNDER SUCH ACT AND APPLICABLE STATE SECURITIES LAWS OR PURSUANT TO AN EXEMPTION FROM REGISTRATION UNDER SUCH ACT OR SUCH LAWS. (ii) THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO TRANSFER AND OTHER RESTRICTIONS SET FORTH IN A SHAREHOLDERS AGREEMENT, DATED AS OF APRIL 2, 2008, COPIES OF WHICH ARE ON FILE WITH THE SECRETARY OF THE ISSUER. NO TRANSFER, SALE, ASSIGNMENT, PLEDGE, HYPOTHECATION OR OTHER DISPOSITION OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE, DIRECTLY OR INDIRECTLY, MAY BE MADE EXCEPT IN ACCORDANCE WITH THE PROVISIONS OF SUCH SHAREHOLDERS AGREEMENT. THE HOLDER OF THIS CERTIFICATE, BY ACCEPTANCE OF THIS CERTIFICATE, AGREES TO BE BOUND BY ALL OF THE PROVISIONS OF SUCH SHAREHOLDERS AGREEMENT.

(b) Upon request of a Shareholder, the Company shall promptly cause clause (i) of the legend to be removed from any certificate for any Company Common Stock to be so transferred upon receipt by the Company of an opinion of counsel reasonably satisfactory to the Company to the effect that such legend is no longer required (or, in the case of a Shareholder that is not an Affiliate of the Company, upon receipt by the Company of a certification to that effect at least six months after the Closing Date) and clause (ii) of the legend shall be removed upon the expiration of such transfer and other restrictions set forth in this Agreement. The Shareholders acknowledge that the Company Common Stock has not been registered under the Securities Act or under any state securities laws and agree that they will not sell or otherwise dispose of any of the Company Common Stock, except in compliance with the registration requirements or exemption provisions of the Securities Act and any other applicable securities laws.

#### 9. Amendment and Waiver.

(a) Except as otherwise provided herein, the provisions of this Agreement may be amended or waived only upon the prior written consent of the Company (to the extent approved by a majority of directors who are not Shareholder Designees) and the Shareholder Representative. A copy of each such amendment shall be sent to each Shareholder and shall be binding upon each Party hereto, <u>provided</u> that the failure to deliver a copy of such amendment shall not impair or affect the validity of such amendment.

(b) Notwithstanding the foregoing provision of <u>Section 9(a)</u>, the Board may authorize the Company to amend or modify this Agreement at any time without the prior written consent of the Shareholder Representative, to (i) enter into agreements with permitted assignees pursuant to the terms of this Agreement, providing in substance that such permitted assignees will be bound by this Agreement, and (ii) amend this Agreement (A) to satisfy any requirements, conditions, guidelines or opinions contained in any opinion, directive, order, ruling or regulation of the SEC, the Internal Revenue Service or any other United States federal or state agency, or in any United States federal or state statute, and to cure any ambiguity or correct or supplement any provision of this Agreement that may be incomplete or inconsistent with any other provision contained herein, so long as any amendment under this clause (ii) does not adversely affect the investment in the Company of the Shareholder Representative or the rights, duties or obligations of the Shareholder Representative hereunder, provided that no amendment of this Agreement shall change the provisions of this <u>Section 9</u> without the prior written consent of the Shareholder Representative.

(c) No failure to exercise and no delay in exercising any right, power or privilege granted under this Agreement shall operate as a waiver of such right, power or privilege. No single or partial exercise of any right, power or privilege granted under this Agreement shall preclude any other or further exercise thereof or the

exercise of any other right, power or privilege. The rights and remedies provided in this Agreement are cumulative and are not exclusive of any rights or remedies provided by Applicable Law.

10. <u>Shareholder Representative</u>. Ronald N. Tutor shall serve as the Shareholder Representative until the earlier of his resignation as the Shareholder Representative, death, or mental or physical incapacity (such that he is unable to fulfill the obligations of the Shareholder Representative). At such time as Ronald N. Tutor ceases to serve in such capacity, and thereafter in the case of the resignation as the Shareholder Representative, death or mental or physical incapacity of any successor Shareholder Representative. The Beneficial Owners of a majority of the total number of shares of Company Stock then Beneficially Owned by the Shareholder Representative Group shall appoint an individual to serve as a successor Shareholder Representative and shall give prompt written notice to the Company of the selection and identity of such successor. Any such successor Shareholder Representative Group (and shall be deemed to have resigned as the Shareholder Representative upon such successor s death or mental or physical incapacity).

11. <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under Applicable Law, but if any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any Applicable Law in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement in such jurisdiction or affect the validity, legality or enforceability of any provision in any other jurisdiction, but this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

12. <u>Successors and Assigns</u>. Except as otherwise expressly provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, the successors, assigns, heirs, executors and administrators of the Parties hereto. Except as otherwise expressly provided herein, none of the Shareholders may assign any of their rights or obligations hereunder without the prior written consent of the Company.

13. <u>Counterparts</u>. This Agreement may be executed in multiple counterparts, each of which shall be an original and all of which taken together shall constitute one and the same agreement.

14. <u>Remedies</u>. The Company and the Shareholders shall be entitled to enforce their rights under this Agreement specifically, to recover damages by reason of any breach of any provision of this Agreement and to exercise all other rights existing in their favor. The Parties hereto agree and acknowledge that money damages would not be an adequate remedy for any breach of the provisions of this Agreement and that the Company and any Shareholder may in its sole discretion apply to any court of law or equity of competent jurisdiction for specific performance or injunctive relief (without posting a bond or other security) in order to enforce or prevent any violation of the provisions of this Agreement.

15. <u>Notices</u>. Any notice provided for in this Agreement shall be in writing and shall be either personally delivered, or sent via facsimile or mailed first class mail (postage prepaid) or sent by reputable overnight courier service (charges prepaid) to the Company at the address set forth below and to any other recipient at the address indicated on the Schedule of Shareholders at such address as indicated by the Company s records, or at such address or to the attention of such other Person as the recipient party has specified by prior written notice to the sending party. Notices shall be deemed to have been given hereunder when delivered personally, when sent via facsimile (as evidenced by a printed confirmation) if sent prior to 5:00 p.m. (local time of recipient) on a Business Day or, if not, the next succeeding Business Day), three (3) Business Days after deposit in the United States mail and one (1) Business Day after deposit with a reputable overnight courier service. The Company s address as of the date hereof is:

Perini Corporation

73 Mt. Wayte Avenue

Framingham, Massachusetts 01701

Attention: Corporate Secretary

Facsimile: (508) 628-2010

16. <u>Governing Law; Jurisdiction</u>. THIS AGREEMENT SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF MASSACHUSETTS, WITHOUT REGARD TO THE CONFLICTS OF LAW PRINCIPLES THEREOF. ANY ACTION OR PROCEEDING AGAINST THE PARTIES RELATING IN ANY WAY TO THIS AGREEMENT MAY BE BROUGHT AND ENFORCED EXCLUSIVELY IN THE COURTS OF THE STATE OF MASSACHUSETTS OR (TO THE EXTENT SUBJECT MATTER JURISDICTION EXISTS THEREFOR) THE U.S. DISTRICT COURT FOR THE DISTRICT OF MASSACHUSETTS, AND THE PARTIES IRREVOCABLY SUBMIT TO THE JURISDICTION OF BOTH SUCH COURTS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING.

17. <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES HERETO WAIVES ANY RIGHT IT MAY HAVE TO TRIAL BY JURY IN RESPECT OF ANY LITIGATION BASED ON, ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR ANY COURSE OF CONDUCT, COURSE OF DEALING, VERBAL OR WRITTEN STATEMENT OR ACTION OF ANY PARTY HERETO.

18. <u>Business Days</u>. If any time period for giving notice or taking action hereunder expires on a day which is not a Business Day, the time period shall automatically be extended to the Business Day immediately following such day.

19. <u>Construction</u>. The Parties hereto have participated jointly in the negotiation and drafting of this Agreement; accordingly, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any Party hereto by virtue of the authorship of any provisions of this Agreement.

20. <u>No Third Party Beneficiaries</u>. This Agreement is intended for the benefit of the Parties hereto and their respective permitted successors and assigns and are not for the benefit of, nor may any provision hereof be enforced by, any other Person.

21. <u>Delivery by Facsimile or Email</u>. This Agreement, the agreements referred to herein, and each other agreement or instrument entered into in connection herewith or therewith or contemplated hereby or thereby, and any amendments hereto or thereto, to the extent signed and delivered by means of a facsimile machine or email with scan or facsimile attachment, shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person. At the request of any Party hereto or to any such agreement or instrument, each other Party hereto or thereto shall re-execute original forms thereof and deliver them to all other Parties. No Party hereto or to any such agreement or instrument shall raise the use of a facsimile machine or email to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of a facsimile machine or email as a defense to the formation or enforceability of a contract, and each such Party forever waives any such defense.

22. <u>Effectiveness</u>. This Agreement shall become effective on, and not effective until, the Closing Date (as defined in the Merger Agreement. This Agreement shall automatically terminate and be of not further force or effect upon termination of the Merger Agreement.

{Remainder of page intentionally left blank.}

IN WITNESS WHEREOF, the Parties hereto have executed this Agreement on the day and year first above written.

### PERINI CORPORATION

 By:
 /s/
 ROBERT BAND

 Name:
 Robert Band

 Title:
 President and Chief Operating Officer

#### **RONALD N. TUTOR**

/s/ RONALD N. TUTOR Name: Ronald N. Tutor

#### SHAREHOLDERS:

# RONALD N. TUTOR SEPARATE PROPERTY TRUST

By: /s/ RONALD N. TUTOR Name: Ronald N. Tutor Title: Trustee

# **RONALD N. TUTOR 2006 QUICKGRAT**

By:	/s/	RONALD N. TUTOR
Name:		<b>Ronald N. Tutor</b>
Title:		Trustee

Name:	/s/ JAMES A. FROST James A. Frost
Name:	/s/ WILLIAM B. SPARKS William B. Sparks
Name:	/s/ John D. Barrett John D. Barrett
Name:	/s/ ROGER K. SEXTON Roger K. Sexton
Name:	/s/ David L. Randall David L. Randall
Name:	/s/ Gerald W. Brown Gerald W. Brown
Name:	/s/ Robert Stephen Lewis Robert Stephen Lewis
Name:	/s/ Michael J. Kerchner Michael J. Kerchner
Name:	/s/ JAMES FOSTER James Foster
Name: Name:	
	James Foster /s/ Leonard Kaae
Name:	James Foster /s/ Leonard Kaae Leonard Kaae /s/ Mark S. Fishbach
Name: Name:	James Foster /s/ Leonard Kaae Leonard Kaae /s/ Mark S. Fishbach Mark S. Fishbach /s/ Joseph A. Guglielmo
Name: Name: Name:	James Foster /s/ Leonard Kaae Leonard Kaae /s/ Mark S. Fishbach Mark S. Fishbach /s/ Joseph A. Guglielmo Joseph A. Guglielmo
Name: Name: Name: Name:	James Foster /s/ LEONARD KAAE Leonard Kaae /s/ MARK S. FISHBACH Mark S. FISHBACH Mark S. Fishbach /s/ JOSEPH A. GUGLIELMO JOSEPH

Annex C

#### **Execution Version**

#### **EMPLOYMENT AGREEMENT**

This EMPLOYMENT AGREEMENT (this <u>Agreement</u>) is entered into as of this 2nd day of April, 2008, (the <u>Signing</u> Date), by and between **Perini Corporation**, a Massachusetts corporation (herein referred to as <u>Employer</u>), and **Ronald N. Tutor**, an individual (<u>Exec</u>utive).

WHEREAS, on the Signing Date, Employer is entering into that certain Agreement and Plan of Merger (the <u>Merger Agreement</u>) by and among the Employer, Trifecta Acquisition LLC, a California limited liability company and a wholly-owned subsidiary of Employer (<u>Merger Sub</u>), Tutor-Saliba Corporation, a California corporation (the <u>Company</u>), Executive and shareholders of the Company;

WHEREAS, Employer, Merger Sub and the Company intend to effect a merger of the Company with and into Merger Sub (the <u>Merger</u>), with Merger Sub to be the surviving entity of the Merger and a wholly-owned Subsidiary of Employer;

WHEREAS, as a condition to entering into the Merger Agreement, the Employer and Executive desire to enter into this Agreement (contingent on the closing of the transactions contemplated hereby) to set out the terms and conditions for the continued employment relationship of Executive with the Employer.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein and for other good and valuable consideration, the receipt and sufficiency of which hereby are acknowledged, the parties hereto agree as follows:

Section 1. Effectiveness. This Agreement shall become effective on, and not effective until, the Closing Date (as defined in the Merger Agreement) (the <u>Effective Date</u>). This Agreement shall automatically terminate upon termination of the Merger Agreement.

Section 2. <u>Employment Agreement</u>. On the terms and conditions set forth in this Agreement, the Employer agrees to continue to employ Executive and Executive agrees to continue to be employed by the Employer for the Employment Period set forth in <u>Section 3</u> and in the positions and with the duties set forth in <u>Section 4</u>. Terms used herein with initial capitalization not otherwise defined are defined in <u>Section 27</u>.

Section 3. <u>Term</u>. The initial term of employment under this Agreement shall be for a five-year period commencing on the Effective Date (the <u>Initial Term</u>). The term of employment shall be automatically extended for an additional consecutive 12-month period (the <u>Extended Term</u>) on the fifth anniversary of the Effective Date and each subsequent anniversary, unless and until the Employer or Executive provides written notice to the other party in accordance with <u>Section 14</u> hereof not less than 90 days before such anniversary date that such party is electing not to extend the term of employment under this Agreement (<u>Non-Renewal</u>), in which case the term of employment hereunder shall end as of the end of such Initial Term or Extended Term, as the case may be, unless sooner terminated as hereinafter set forth. Such Initial Term and all such Extended Terms are collectively referred to herein as the <u>Employment Period</u>.

Section 4. <u>Position and Duties</u>. During the Employment Period, Executive shall serve as the Chief Executive Officer of the Employer, as a member of the Employer s Board of Directors (the <u>Board</u>) and as Chairman of the Board. In such capacities, Executive shall report exclusively to the Board and shall be the most senior executive officer of the Employer and shall have the duties, responsibilities and authorities customarily associated with the positions of chairman of the board of directors and chief executive officer of a company the size and nature of the Employer, including, without limitation, oversight of the Employer s day-to-day operations. Executive shall use his good faith efforts to assist the Employer in developing a long-term succession

plan and assisting in its implementation. Executive shall devote Executive s reasonable best efforts and full business time to the performance of Executive s duties hereunder and the advancement of the business and affairs of the Employer; provided that Executive shall be entitled (i) with the consent of the Board (which shall not be unreasonably withheld), to serve as a member of the board of directors of a reasonable number of other companies, (ii) to serve on civic, charitable, educational, religious, public interest or public service boards (including, without limitation, the USC Board of Trustees), and (iii) to manage Executive s personal and family investments, in each case, to the extent such activities, individually or in the aggregate, do not materially interfere with the performance of Executive s duties and responsibilities hereunder.

Section 5. <u>Place of Performance</u>. During the Employment Period, Executive shall be based primarily at the offices of the Employer as of the Effective Time near Los Angeles, California, or from the principal executive offices of the Employer in Las Vegas, Nevada, at such time as the primary office of Employer is established in Las Vegas, except for reasonable travel on the Employer s business consistent with Executive s positions.

#### Section 6. Compensation and Benefits.

(a) <u>Base Salary</u>. During the Employment Period, the Employer shall pay to Executive a base salary (the <u>Base Salary</u>) at the rate of no less than \$1,500,000 per calendar year, less applicable deductions, and prorated for any partial year. The Base Salary shall be reviewed for increase by the Employer no less frequently than annually and shall be increased in the discretion of the Employer and any such adjusted Base Salary shall constitute the Base Salary for purposes of this Agreement. The Base Salary shall be paid in substantially equal installments in accordance with the Employer s regular payroll procedures. Executive s Base Salary may not be decreased during the Employment Period.

(b) <u>Annual Bonus</u>. Executive shall be paid an annual cash performance bonus (an <u>Annual Bonus</u>) in respect of each calendar year that ends during the Employment Period, to the extent earned based on performance against objective performance criteria. The performance criteria for any particular calendar year shall be established by the Compensation Committee of the Board (the <u>Compensation Committee</u>) no later than 90 days after the commencement of such calendar year. Executive s Annual Bonus for a calendar year shall equal 175% of his Base Salary for that year if target levels of performance for that year (as established by the Compensation Committee when the performance criteria for that year are established) are achieved, with greater or lesser amounts (including zero) paid for performance above and below target (such greater and lesser amounts to be determined by a formula established by the Compensation Committee for that year when it established the targets and performance criteria for that year). Executive s Annual Bonus for a calendar year shall be determined by the Compensation Committee after the end of the calendar year and shall be paid to Executive when annual bonuses for that year are paid to other senior executives of the Employer generally, but in no event later than March 15 of the following calendar year. In carrying out its functions under this <u>Section 6(b)</u>, the Compensation Committee shall at all times act reasonably and in good faith, and shall consult with Executive to the extent appropriate.

(c) <u>Equity Compensation</u>. Executive will be considered at a level appropriate for his positions with the Employer for participation in the Employer s company-wide equity incentive plan to be adopted after the Effective Date, including the potential grant of restricted stock units of the Employer (the <u>Restricted Stock Units</u>). Subject to the terms of this Agreement, any Restricted Stock Units that are granted shall be governed by a restricted stock unit agreement in substantially the form used by the Employer for awards of restricted stock units to other senior executives.

(d) <u>Other Incentives</u>. Executive shall be eligible for other or additional long-term incentives in the sole and absolute discretion of the Compensation Committee and/or the Board. Such incentive awards (if any) shall be at a level, and on terms and conditions, that are commensurate with Executive s positions and responsibilities at the Employer and appropriate in light of corresponding awards to other senior executives of the Employer (but without regard to any special or one-time grants to other senior executives, including any sign-on or special retention grants). Except as otherwise provided herein, Executive shall not be entitled to participate in any other compensation, bonus, retention or incentive program, except as may be explicitly determined by the Board or the Compensation Committee in its sole and absolute discretion.

(e) <u>Perquisites</u>. During the Employment Period, Executive shall be entitled to 150 hours of flying time per calendar year of personal use of the Business Boeing Jet 737-700 Reg. No. N315TS, S/N 30772 (<u>BBJ</u>) with any unused balance being carried forward to subsequent calendar years in the Employment Period. Executive shall also be provided with use of an automobile and driver, and use of an apartment in Las Vegas, Nevada, in each case on terms and conditions to be determined by the Board. During the Employment Period, Executive shall, in addition to the foregoing, also be entitled to (i) to participate in all fringe benefits and perquisites made available generally to senior executives of the Employer, such participation to be at levels, and on terms and conditions, that are commensurate with his positions and responsibilities at the Employer, and (ii) to receive such additional fringe benefits and perquisites as the Employer may, in its sole and absolute discretion, from time to time provide.

(f) <u>Vacation: Benefits</u>. Executive shall be entitled to 30 vacation days during each calendar year in the Employment Period, such vacation to be accrued, taken and carried over in accordance with the policies of the Employer. During the Employment Period, Executive will be entitled to participate in all pension, retirement, profit sharing, savings, 401(k), income deferral, life insurance, disability insurance, accidental death and dismemberment protection, travel accident insurance, hospitalization, medical, dental, vision and other employee benefit plans, programs and arrangements that may from time to time be made available generally to other senior executives of the Employer, all to the extent Executive is eligible under the terms of such plans, programs and arrangements. Executive s participation in all such plans, programs and arrangements shall be at a level, and on terms and conditions, that are commensurate with his positions and responsibilities at the Employer. In addition, the Employer shall, as promptly as reasonably practicable after the Effective Date and with Executive s full cooperation, obtain on behalf of Executive life insurance coverage under term or ordinary life insurance polici(es) (at Executive s choice) with an aggregate annual premium cost not to exceed \$175,000. Except for the immediately preceding sentence, nothing in this Section 6 shall be construed to require the Employer to establish or maintain any particular employee benefit plan, program or arrangement.

Section 7. <u>Expenses</u>. Executive is expected and is authorized to incur reasonable expenses in the performance of his duties hereunder. The Employer shall reimburse Executive for all such expenses reasonably and actually incurred in accordance with policies which may be adopted from time to time by the Employer promptly upon periodic presentation by Executive of an itemized account, including reasonable substantiation, of such expenses. Unless the Employer s financial condition materially declines from that existing at the Effective Date, the Employer shall maintain for Executive s business use an aircraft similar to the aircraft historically used by the Company prior to the Effective Date.

Section 8. <u>Confidentiality</u>, <u>Non-Disclosure and Non-Competition Agreement</u>. The Employer and Executive acknowledge and agree that during Executive s employment with the Employer, Executive will have access to and may assist in developing Confidential Information and will occupy a position of trust and confidence with respect to the Employer s affairs and business and the affairs and business of its Affiliates. Executive agrees that the following obligations are necessary to preserve the confidential and proprietary nature of Confidential Information and to protect the Employer and its Affiliates against harmful solicitation of employees and customers, harmful competition and other actions by Executive that would result in serious adverse consequences for the Employer and any of its Affiliates:

(a) <u>Non-Disclosure</u>. During and after Executive s employment with the Employer, Executive will not knowingly use, disclose or transfer any Confidential Information other than as authorized in writing by the Employer or within the scope of Executive s duties with the Employer as determined reasonably and in good faith by Executive. Anything herein to the contrary notwithstanding, the provisions of this <u>Section 8(a)</u> shall not apply (i) when disclosure is required by law or by any court, arbitrator, mediator or administrative or legislative body (including any committee thereof) with actual or apparent jurisdiction to order Executive to disclose or make accessible any information; (ii) to the extent necessary in connection with any other litigation, arbitration or mediation involving this Agreement, including, but not limited to, the enforcement of this Agreement; (iii) as to information that becomes generally known to the public or within the relevant trade or industry other than due

to Executive s violation of this <u>Section 8(a)</u>; or (iv) as to information that is or becomes available to Executive on a non-confidential basis from a source that is entitled to disclose it to Executive.

(b) <u>Materials</u>. Executive will not remove any Confidential Information or any other property of the Employer or any of its Affiliates from the Employer's premises or make copies of such materials except for normal and customary use in the Employer's business as determined reasonably and in good faith by Executive. The Employer acknowledges that Executive, in the ordinary course of his duties, routinely uses and stores Confidential Information at home and other locations. Executive will return to the Employer all Confidential Information and copies thereof and all other property of the Employer or any of its Affiliates at any time upon the request of the Employer and in any event promptly after termination of Executive's employment. Executive agrees to identify and return to the Employer any copies of any Confidential Information within Executive's control after Executive ceases to be employed by the Employer. Anything to the contrary notwithstanding, nothing in this Section 8 shall prevent Executive from retaining a home computer, papers and other materials of a personal nature, including diaries, calendars and Rolodexes, information relating to his compensation or relating to reimbursement of expenses, information that he reasonably believes may be needed for tax purposes, and copies of plans, programs and agreements relating to his employment.

(c) <u>Developments</u>. Executive shall, promptly upon reasonable request, disclose to the Employer all inventions (whether patentable or not), trade secrets, trademark concepts, and advertising and marketing concepts (collectively, hereinafter referred to as <u>Developments</u>), that he makes, alone or with others, during his employment with Employer or any of its Affiliates relating to any of their businesses. Employer will exclusively own all Developments. Executive hereby assigns to the Employer all rights that he has or acquires in any Developments, and he will exclusively own documents and take any actions as reasonably requested by the Employer necessary to effect that assignment. Executive need not incur any cost related to that assignment or the creation of any related intellectual property rights. The parties agree that Developments are Confidential Information. Both during the Employment Period and thereafter, Executive shall fully cooperate with the Employer or any of its Affiliates, whether under the terms of this Agreement or otherwise. This shall include, upon reasonable request by the Employer, executing, acknowledging, and delivering to Employer all documents or papers that may be necessary to enable Employer to publish or protect such intellectual property rights. The Employer to publish or protect such

(d) <u>Cooperation</u>. During the Employment Period and thereafter Executive will, upon reasonable request and subject to such reasonable condition as Executive may reasonably establish: (a) cooperate with the Employer in connection with any matter that arose during Executive s employment and that relates to the business or operations of the Employer or any of its Affiliates, or of which Executive may have any knowledge or involvement; and (b) consult with and provide information to the Employer and its representatives concerning such matters. Such cooperation shall be rendered at reasonable times and places and in a manner that does not unreasonably interfere with any other employment in which Executive may then be engaged. Nothing in this Agreement shall be construed or interpreted as requiring Executive to provide any testimony or affidavit that is not truthful.

(e) <u>No Solicitation or Hiring of Employees</u>. During the Non-Compete Period, Executive shall not solicit, entice, persuade or induce any individual who is employed by the Employer or any of its Affiliates (or who was so employed within 180 days prior to Executive s action) to terminate or refrain from continuing such employment or to become employed by or enter into contractual relations with any other individual or entity other than the Employer or any of its Affiliates, and Executive shall not hire, directly or indirectly, as an employee, consultant or otherwise, any such person. Anything to the contrary notwithstanding, the Employer agrees that (i) Executive s responding to an unsolicited request from any former employee of the Employer for advice on employment matters; and (ii) Executive s responding to an unsolicited request for an employment reference regarding any former employee of the Employer from such former employee, or from a third party, by

providing a reference setting forth his personal views about such former employee, shall not be deemed a violation of this Section 8(e).

#### (f) Non-Competition.

(i) During the Non-Compete Period, Executive shall not, directly or indirectly, (A) solicit or encourage any client or customer of the Employer or any of its Affiliates, or any person or entity who was such a client or customer within 180 days prior to Executive s action to terminate, reduce or alter in a manner adverse to the Employer or any of its Affiliates, any existing business arrangements with the Employer or any of its Affiliates or to transfer existing business from the Employer or any of its Affiliates to any other person or entity, (B) provide services in any capacity to any entity if (i) the entity competes with the Employer or any of its Affiliates by engaging in any business engaged in by the Employer or any of its Affiliates in any country in which the Employer or its Affiliates engages in such business, or (ii) the services to be provided by Executive are competitive with the Employer and substantially similar to those previously provided by Executive to the Employer or any of its Affiliates; or (C) own an interest in any entity described in subsection (B)(i) immediately above; provided, however, that Executive may own, as a passive investor, securities of any such entity that has outstanding publicly traded securities so long as his direct holdings in any such entity shall not in the aggregate constitute more than 5% of the voting power of such entity. Executive acknowledges that this covenant has a unique, very substantial and immeasurable value to the Employer, that Executive has sufficient assets and skills to provide a livelihood for Executive while such covenant remains in force and that, as a result of the foregoing, in the event that Executive breaches such covenant, monetary damages would be an insufficient remedy for the Employer and equitable enforcement of the covenant would be proper.

(ii) If the restrictions contained in Section 8(f)(i) shall be determined by any court of competent jurisdiction to be unenforceable by reason of their extending for too great a period of time or over too great a geographical area or by reason of their being too extensive in any other respect, Section 8(f)(i) shall be modified to be effective for the maximum period of time for which it may be enforceable and over the maximum geographical area as to which it may be enforceable and to the maximum extent in all other respects as to which it may be enforceable.

(g) <u>Publicity</u>. During the Employment Period, Executive hereby grants to the Employer the right to use, in a reasonable and appropriate manner, Executive s name and likeness, without additional consideration, on, in and in connection with technical, marketing or disclosure materials, or any combination thereof, published by or for the Employer or any of its Affiliates.

(h) <u>Conflicting Obligations and Rights</u>. Executive agrees to inform the Employer of any apparent conflicts between Executive s work for the Employer and any obligations Executive may have to preserve the confidentiality of another s proprietary information or related materials before using the same on the Employer s behalf. The Employer shall receive such disclosures in confidence and consistent with the objectives of avoiding any conflict of obligations and rights or the appearance of any conflict of interest.

(i) <u>Enforcement</u>. Executive acknowledges that in the event of any breach of this <u>Section 8</u>, the business interests of the Employer and its Affiliates will be irreparably injured, the full extent of the damages to the Employer and its Affiliates will be impossible to ascertain, monetary damages will not be an adequate remedy for the Employer and its Affiliates, and the Employer will be entitled to enforce this Agreement by a temporary, preliminary and/or permanent injunction or other equitable relief, without the necessity of posting bond or security, which Executive expressly waives. Executive understands that the Employer may waive some of the requirements expressed in this Agreement, but that such a waiver to be effective must be made in writing and should not in any way be deemed a waiver of the Employer s right to enforce any other requirements or provisions of this Agreement. Executive agrees that each of Executive s obligations specified in this Agreement

is a separate and independent covenant and that the unenforceability of any of them shall not preclude the enforcement of any other covenants in this Agreement. Executive further agrees that any breach of this Agreement by the Employer prior to the Date of Termination shall not release Executive from compliance with his obligations under this <u>Section 8</u>, so along as the Employer fully complies with <u>Section 10</u>, <u>Section 11</u>, <u>Section 12</u>, and <u>Section 13</u>.

Section 9. Termination of Employment.

(a) <u>Permitted Terminations</u>. Executive s employment hereunder may be terminated during the Employment Period under the following circumstances:

(i) Death. The Employment Period and Executive s employment hereunder shall terminate upon Executive s death;

(ii) By the Employer. The Employer may terminate the Employment Period and Executive s employment:

(A) <u>Disability</u>. If Executive has been substantially unable to perform Executive s material duties hereunder by reason of illness, physical or mental disability or other similar incapacity, which inability shall continue for 180 consecutive days or 270 days in any 24-month period (a <u>Disability</u>) (provided, that until such termination, Executive shall continue to receive his compensation and benefits hereunder, reduced by any benefits payable to him under any disability insurance policy or plan applicable to him or her); or

(B) Cause. For Cause or without Cause;

(iii) <u>By Executive</u>. Executive may terminate the Employment Period and his employment for any reason (including Good Reason) or for no reason.

(b) <u>Termination</u>. Any termination of Executive s employment by the Employer or Executive (other than because of Executive s death) shall be communicated by written Notice of Termination to the other party hereto in accordance with <u>Section 14</u> hereof. For purposes of this Agreement, a <u>Notice of Termination</u> shall mean a notice which shall indicate the specific termination provision in this Agreement relied upon, if any, and shall set forth in reasonable detail the facts and circumstances claimed to provide a basis for termination. Executive agrees, in the event of any dispute under <u>Section 9(a)(ii)(A)</u> as to whether a Disability exists, and if requested by the Employer, to submit to a physical examination by a licensed physician selected by mutual consent of the Employer and Executive (which shall not unreasonably be withheld), the cost of such examination to be paid by the Employer. The written medical opinion of such physician shall be conclusive and binding upon each of the parties hereto as to whether a Disability exists and the date when such Disability arose. This Section shall be interpreted and applied so as to comply with the provisions of the Americans with Disabilities Act and any applicable state or local laws.

#### Section 10. Compensation Upon Termination.

(a) <u>Death</u>. If Executive s employment is terminated during the Employment Period as a result of Executive s death, this Agreement and the Employment Period shall terminate without further notice or any action required by the Employer or Executive s legal representatives. Upon Executive s death during the Employment Period, the Employer shall pay or provide the following: (i) Executive s Base Salary due through the Date of Termination, (ii) all Accrued Benefits, if any, to which Executive is entitled as of the Date of Termination at the time such payments are due, and (iii) all outstanding equity awards held by Executive immediately prior to his termination shall immediately vest (with outstanding options remaining exercisable for the length of their remaining term). Except as set forth herein, the Employer shall have no further obligation to Executive under this Agreement.

(b) <u>Disability</u>. If the Employer terminates Executive s employment during the Employment Period because of Executive s Disability, the Employer shall pay or provide the following: (i) Executive s Base Salary

due through the Date of Termination, (ii) all Accrued Benefits, if any, to which Executive is entitled as of the Date of Termination at the time such payments are due, and (iii) all outstanding equity awards held by Executive immediately prior to his termination shall immediately vest (with outstanding options remaining exercisable for the length of their remaining term). Except as set forth herein, the Employer shall have no further obligations to Executive under this Agreement.

(c) <u>Termination by the Employer for Cause or by Executive without Good Reason</u>. If, during the Employment Period, the Employer terminates Executive s employment for Cause pursuant to <u>Section 9(a)(ii)(B)</u> or Executive terminates his employment without Good Reason, the Employer shall pay to Executive Executive s Base Salary due through the Date of Termination and all Accrued Benefits, if any, to which Executive is entitled as of the Date of Termination, at the time such payments are due, and Executive s rights with respect to equity or equity-related awards shall be governed by the applicable terms of the related plan or award agreement.

(d) <u>Termination by the Employer without Cause or by Executive with Good Reason</u>. Subject to <u>Section 10(e)</u>, if the Employer terminates Executive s employment during the Employment Period other than for Cause or Disability pursuant to <u>Section 9(a)</u> or if Executive terminates his employment hereunder with Good Reason: (i) the Employer shall pay Executive (A) Executive s Base Salary due through the Date of Termination, (B) a Pro Rata Bonus at the time other executives of the Employer receive annual bonuses for the calendar year in which the Date of Termination occurs, (C) all Accrued Benefits, if any, to which Executive is entitled as of the Date of Termination, in each case at the time such payments are due and (D) a cash lump sum in an amount equal to two times the sum of Executive s Base Salary and Target Bonus for the year of termination, (ii) all outstanding equity awards held by Executive immediately prior to his termination shall immediately vest (with outstanding options remaining exercisable for the length of their remaining term), and (iii) Executive and his covered dependents shall be entitled to continued participation on the same terms and conditions as applicable immediately prior to Executive s Date of Termination for 24 months; provided that if such continued coverage is not permitted under the terms of such benefit plans, the Employer shall pay Executive an additional amount that, on an after-tax basis, is equal to the cost of comparable coverage obtained by Executive.

(e) <u>Change in Control</u>. This <u>Section 10(e)</u> shall apply if there is (i) a termination of Executive s employment by the Employer other than for Cause or Disability pursuant to <u>Section 9(a)</u> or by Executive for Good Reason during the two-year period after a Change in Control or (ii) a termination of Executive s employment by the Employer prior to a Change in Control, if the termination was at the request of a third party or otherwise arose in anticipation of a Change in Control. If any such termination occurs, Executive shall receive benefits set forth in <u>Section 10(d)</u>, except that (i) in lieu of the lump-sum payment under <u>Section 10(d)(i)(D)</u>, Executive shall receive in a lump sum after the termination of his employment an amount equal to three multiplied by the sum of (A) Executive s Base Salary and (B) Executive s Target Bonus and (ii) the benefits described in <u>Section 10(d)(iii)</u> shall be continued for the greater of 36 months or the balance of the Employment Period. Notwithstanding anything to the contrary herein, this <u>Section 10(e)</u> shall not apply upon Executive s death.

(f) Liquidated Damages. The parties acknowledge and agree that damages which will result to Executive for termination by the Employer of Executive s employment without Cause or by Executive for Good Reason shall be extremely difficult or impossible to establish or prove, and agree that the amounts payable to Executive under Section 10(d)(i)(D) or Section 10(e)(i) (the Severance Payments ) shall constitute liquidated damages for any such termination. Executive agrees that, except for such other payments and benefits to which Executive may be entitled as expressly provided by the terms of this Agreement or any other applicable benefit plan, such liquidated damages shall be in lieu of all other claims that Executive may make by reason of any such termination of his employment and that, as a condition to receiving the Severance Payments, Executive will execute a release of claims substantially in the form attached hereto as Exhibit A. Within five business days of the Date of Termination, the Employer shall deliver to Executive the appropriate form of release of claims for Executive to execute. The Severance Payments shall be made within three business days of the expiration of the revocation period without the release being revoked.

(g) <u>No Offset</u>. In the event of termination of his employment, Executive shall be under no obligation to seek other employment and there shall be no offset against amounts due to him on account of any remuneration or benefits provided by any subsequent employment he may obtain. The Employer s obligation to make any payment pursuant to, and otherwise to perform its obligations under, this Agreement shall not be affected by any offset, counterclaim or other right that the Employer or its affiliates may have against him for any reason.

(h) <u>Section 409A</u>. Notwithstanding the timing of the payments pursuant to <u>Section 10</u> of this Agreement, to the extent Executive would otherwise be entitled to a payment during the six months beginning on the Date of Termination that would be subject to the additional tax imposed under Section 409A of the Internal Revenue Code of 1986, as amended (the <u>Code</u>), (i) the payment will not be made to Executive and instead will be made to an account established to fund such payments (provided that such funds shall be at all times subject to the creditors of the Employer) and (ii) the payment, together with interest thereon at the rate of prime plus 1%, will be paid to Executive on the six-month anniversary of Date of Termination. Similarly, to the extent Executive would otherwise be entitled to any benefit (other than a cash payment) during the six months beginning on the Date of Termination that would be subject to the additional tax under Section 409A of the Code, the benefit will be delayed and will begin being provided (together, if applicable, with an adjustment to compensate Executive for the delay, with such adjustment to be determined in the Employer s reasonable good faith discretion) on the six-month anniversary of the Date of Termination. The Employer will establish the account, as applicable, no later than ten days after Executive s Date of Termination. It is the intention of the parties that the payments and benefits to which Executive could become entitled in connection with termination of employment under this Agreement comply with Section 409A of the Code. In the event that the parties determine that any such benefit or right does not so comply, they will negotiate reasonably and in good faith to amend the terms of this Agreement such that it complies (in a manner that attempts to minimize the economic impact of such amendment on Executive and the Employer and its affiliates).

(i) In the event of a Termination of Executive s employment by Employer without Cause, notwithstanding the limitation on Transfer set forth in Section 5(a) of the Shareholders Agreement, dated April 2, 2008, by and among Employer, Executive, and certain other shareholders of Employer (the <u>Shareholders Agreement</u>), Executive shall be entitled to rely on the proviso set forth in clause (y)(ii) thereof from and after such termination (without regard to requirement the Executive wait until the 5 year anniversary of the Effective Time as set forth therein).

#### Section 11. Certain Additional Payments by the Employer.

(a) If it shall be determined that any benefit provided to Executive or payment or distribution by or for the account of the Employer to or for the benefit of Executive, whether provided, paid or payable or distributed or distributable pursuant to the terms of this Agreement or otherwise (a <u>Payment</u>) would be subject to the excise tax imposed by Section 4999 of the Code, or any interest or penalties are incurred by Executive with respect to such excise tax resulting from any action or inaction by the Employer (such excise tax, together with any such interest and penalties, collectively, the <u>Excise Tax</u>), then Executive shall be entitled to receive an additional payment <u>(a Gross-Up Payment</u>) in an amount such that after payment by Executive of the Excise Tax and all other income, employment, excise and other taxes that are imposed on the Gross-Up Payment, Executive retains an amount of the Gross-Up Payment equal to the sum of (A) the Excise Tax imposed upon the Payments and (B) the product of any deductions disallowed because of the inclusion of the Gross-Up Payment in Executive s adjusted gross income and the highest applicable marginal rate of federal income taxation for the calendar year in which the Gross-Up Payment is to be made.

(b) Subject to the provisions of <u>Section 11(c)</u>, all determinations required to be made under this <u>Section 11</u>, including whether and when a Gross-Up Payment is required and the amount of such Gross-Up Payment and the assumptions to be utilized in arriving at such determination, shall be made by the Employer s independent, certified public accounting firm or such other certified public accounting firm as may be designated by Executive and shall be reasonably acceptable to the Employer (the <u>Accounting Firm</u>) which shall provide detailed supporting calculations both to the Employer and Executive within 15 business days of the receipt of notice from Executive that there has been a Payment, or such earlier time as is requested by the Employer. If the

Accounting Firm is serving as accountant or auditor for the individual, entity or group effecting a change in the ownership or effective control (as defined for purposes of Section 280G of the Code) of the Employer, Executive shall appoint another nationally recognized accounting firm which is reasonably acceptable to the Employer to make the determinations required hereunder (which accounting firm shall then be referred to as the Accounting Firm hereunder). All fees and expenses of the Accounting Firm shall be borne solely by the Employer. Any Gross-Up Payment, as determined pursuant to this <u>Section 11</u>, shall be paid by the Employer to Executive within five days of the receipt of the Accounting Firm shall be binding upon the Employer and Executive. As a result of the uncertainty in the application of Section 4999 of the Code at the time of the initial determination by the Accounting Firm hereunder, it is possible that additional Gross-Up Payments shall be required to be made to compensate Executive for amounts of Excise Tax later determined to be due, consistent with the calculations required to be made hereunder (an <u>Underpayment</u>). If the Employer exhausts its remedies pursuant to <u>Section 11(c)</u> and Executive is required to make a payment of any Excise Tax, the Accounting Firm shall determine the amount of the Underpayment that has occurred and any such Underpayment shall be promptly paid by the Employer to or for the benefit of Executive.

(c) Executive shall notify the Employer in writing of any claim by the Internal Revenue Service that, if successful, would require the payment by the Employer of the Gross-Up Payment. Such notification shall be given as soon as practicable but no later than 10 business days after Executive is informed in writing of such claim and shall apprise the Employer of the nature of such claim and the date on which such claim is requested to be paid. Executive shall not pay such claim prior to the expiration of the 30-day period following the date on which it gives such notice to the Employer (or such shorter period ending on the date that any payment of taxes with respect to such claim is due). If the Employer notifies Executive in writing prior to the expiration of such period that they desire to contest such claim, Executive shall:

(i) give the Employer any information reasonably requested by the Employer relating to such claim;

(ii) take such action in connection with contesting such claim as the Employer shall reasonably request in writing from time to time, including, without limitation, accepting legal representation with respect to such claim by an attorney reasonably selected by the Employer;

(iii) cooperate with the Employer in good faith effectively to contest such claim; and

(iv) permit the Employer to participate in any proceedings relating to such claim; provided, however, that the Employer shall bear and pay directly all costs and expenses (including additional interest and penalties incurred in connection with such contest) and shall indemnify and hold Executive harmless, on an after-tax basis, for any Excise Tax or income tax (including interest and penalties with respect thereto) imposed as a result of such representation and payment of costs and expenses.

Section 12. Indemnification. During the Employment Period and thereafter, the Employer agrees to indemnify and hold Executive and Executive s heirs and representatives harmless, to the maximum extent permitted by law, against any and all damages, costs, liabilities, losses and expenses (including reasonable attorneys fees) as a result of any claim or proceeding (whether civil, criminal, administrative or investigative), or any threatened claim or proceeding (whether civil, criminal, administrative or investigative), or any threatened claim or proceeding (whether civil, criminal, administrative or investigative), against Executive that arises out of or relates to Executive s service as an officer, director or employee, as the case may be, of the Employer, or Executive s service in any such capacity or similar capacity with an affiliate of the Employer or other entity at the request of the Employer, both prior to and after the Effective Date, and to promptly advance to Executive or Executive s heirs or representatives such expenses upon written request with appropriate documentation of such expense upon receipt of an undertaking by Executive or on Executive s behalf to repay such amount if it shall ultimately be determined that Executive is not entitled to be indemnified by the Employer. During the Employment Period and thereafter, the Employer also shall provide Executive with coverage under its current directors and officers liability policy to the same extent that it provides such coverage to its other executive officers. If Executive has any knowledge of any actual or threatened action, suit or proceeding, whether

civil, criminal, administrative or investigative, as to which Executive may request indemnity under this provision, Executive will give the Employer prompt written notice thereof; provided that the failure to give such notice shall not affect Executive s right to indemnification. The Employer shall be entitled to assume the defense of any such proceeding and Executive will use reasonable efforts to cooperate with such defense. To the extent that Executive in good faith determines that there is an actual or potential conflict of interest between the Employer and Executive in connection with the defense of a proceeding, Executive shall so notify the Employer and shall be entitled to separate representation at the Employer s expense by counsel selected by Executive (provided that the Employer may reasonably object to the selection of counsel within ten (10) business days after notification thereof) which counsel shall cooperate, and coordinate the defense. This Section 12 shall continue in effect after the termination of Executive s employment or the termination of this Agreement.

Section 13. <u>Attorney s Fees</u>. The Employer shall advance Executive (and his beneficiaries) any and all costs and expenses (including without limitation attorneys fees and other charges of counsel) incurred by Executive (or any of his beneficiaries) in resolving any controversy, dispute or claim arising out of or relating to this Agreement, any other agreement or arrangement between Executive and the Employer, Executive s employment with the Employer, or the termination thereof; provided that Executive shall reimburse the Employer any advances on a net after-tax basis to cover expenses incurred by Executive for claims (a) brought by the Employer on account of Executive s alleged breach of <u>Section 8</u> of this Agreement, breach of Executive that are judicially determined to be frivolous or advanced in bad faith.

Section 14. <u>Notices</u>. All notices, demands, requests, or other communications which may be or are required to be given or made by any party to any other party pursuant to this Agreement shall be in writing and shall be hand delivered, mailed by first-class registered or certified mail, return receipt requested, postage prepaid, delivered by overnight air courier, or transmitted by facsimile transmission addressed as follows:

(i) If to the Employer:

Perini Corporation 73 Mt. Wayte Avenue Framingham, Massachusetts 01701 Attention: Corporate Secretary Facsimile: (508) 628-2010

#### (ii) If to Executive:

Ronald N. Tutor

Address last shown on the Employer s Records

Each party may designate by notice in writing a new address to which any notice, demand, request or communication may thereafter be so given, served or sent. Each notice, demand, request, or communication that shall be given or made in the manner described above shall be deemed sufficiently given or made for all purposes at such time as it is delivered to the addressee (with the return receipt, the delivery receipt, confirmation of facsimile transmission or the affidavit of messenger being deemed conclusive but not exclusive evidence of such delivery) or at such time as delivery is refused by the addressee upon presentation.

Section 15. <u>Severability</u>. The invalidity or unenforceability of any one or more provisions of this Agreement shall not affect the validity or enforceability of the other provisions of this Agreement, which shall remain in full force and effect.

Section 16. <u>Effect on Other Agreements</u>. The provisions of this Agreement shall supersede the terms of any plan, policy, agreement, award or other arrangement of the Employer (whether entered into before or after the Effective Date) to the extent application of the terms of this Agreement are more favorable to Executive.

Section 17. <u>Survival</u>. It is the express intention and agreement of the parties hereto that the provisions of <u>Section 8</u>, <u>Section 10</u>, <u>Section 11</u>, <u>Section 12</u>, <u>Section 13</u>, <u>Section 14</u>, <u>Section 16</u>, <u>Section 18</u>, <u>Section 20</u>, <u>Section 22</u>, <u>Section 22</u> and <u>Section 26</u> hereof and this <u>Section 17</u> shall survive the termination of employment of Executive. In addition, all obligations of the Employer to make payments hereunder shall survive any termination of this Agreement on the terms and conditions set forth herein.

Section 18. <u>Assignment</u>. The rights and obligations of the parties to this Agreement shall not be assignable or delegable, except that (i) in the event of Executive s death, the personal representative or legatees or distributees of Executive s estate, as the case may be, shall have the right to receive any amount owing and unpaid to Executive hereunder and (ii) the rights and obligations of the Employer hereunder shall be assignable and delegable in connection with any subsequent merger, consolidation, sale of all or substantially all of the assets or equity interests of the Employer or similar transaction involving the Employer or a successor corporation. The Employer shall require any successor to the Employer to expressly assume and agree to perform this Agreement in the same manner and to the same extent that the Employer would be required to perform it if no such succession had taken place.

Section 19. <u>Binding Effect</u>. Subject to any provisions hereof restricting assignment, this Agreement shall be binding upon the parties hereto and shall inure to the benefit of the parties and their respective heirs, devisees, executors, administrators, legal representatives, successors and assigns.

Section 20. <u>Amendment; Waiver</u>. This Agreement shall not be amended, altered or modified except by an instrument in writing duly executed by the party against whom enforcement is sought. Neither the waiver by either of the parties hereto of a breach of or a default under any of the provisions of this Agreement, nor the failure of either of the parties, on one or more occasions, to enforce any of the provisions of this Agreement or to exercise any right or privilege hereunder, shall thereafter be construed as a waiver of any subsequent breach or default of a similar nature, or as a waiver of any such provisions, rights or privileges hereunder.

Section 21. <u>Headings</u>. Section and subsection headings contained in this Agreement are inserted for convenience of reference only, shall not be deemed to be a part of this Agreement for any purpose, and shall not in any way define or affect the meaning, construction or scope of any of the provisions hereof.

Section 22. Governing Law; Venue. This Agreement, the rights and obligations of the parties hereto, and any claims or disputes relating thereto, shall be governed by and construed in accordance with the laws of the State of California (but not including any choice of law rule thereof that would cause the laws of another jurisdiction to apply). Except as otherwise provided in Section 8(i), each of the parties agrees that any dispute between the parties shall be resolved only in the courts of the State of California sitting in Los Angeles, California or the United States District Court for the Central District of California and the appellate courts having jurisdiction of appeals in such courts. In that context, and without limiting the generality of the foregoing (but subject to Section 8(i)), each of the parties hereto irrevocably and unconditionally (a) submits for himself or itself in any proceeding relating to this Agreement or Executive s employment by the Employer or any of its Affiliates, or for the recognition and enforcement of any judgment in respect thereof (a <u>Proceeding</u>), to the exclusive jurisdiction of the courts of the State of California sitting in Los Angeles, California, the court of the United States District Court for the Central District of California and appellate courts having jurisdiction of appeals from any of the foregoing, and agrees that all claims in respect of any such Proceeding shall be heard and determined in such California State court or, to the extent permitted by law, in such federal court; (b) consents that any such Proceeding may and shall be brought in such courts and waives any objection that he or it may now or thereafter have to the venue or jurisdiction of any such Proceeding in any such court or that such Proceeding was brought in an inconvenient court and agrees not to plead or claim the same; (c) waives all right to trial by jury in any Proceeding (whether based on contract, tort or otherwise) arising out of or relating to this Agreement or Executive s employment by the Employer or any of its Affiliates, or his or its performance under or the enforcement of this Agreement; (d) agrees that service of process in any such Proceeding may be effected by mailing a copy of such process by registered or certified mail (or any substantially similar form of mail), postage

prepaid, to such party at his or its address as provided in <u>Section 14</u>; and (e) agrees that nothing in this Agreement shall affect the right to effect service of process in any other manner permitted by the laws of the State of California.

Section 23. <u>Representations</u>. Executive represents, warrants and covenants to the Employer that: (i) on or prior to the date hereof, Executive has informed the Employer of any judgment, order, agreement or arrangement of which he is currently aware and which may affect his right to enter into this Agreement and to fully perform his duties hereunder; (ii) Executive is knowledgeable and sophisticated as to business matters, and that prior to assenting to the terms of this Agreement, or giving the representations and warranties herein, he has been given a reasonable time to review it and has consulted with counsel of his choice; (iii) in entering into this Agreement, Executive is not knowingly breaching or violating any provision of any law or regulation; and (iv) Executive has not knowingly provided to the Employer, nor been requested by the Employer to provide, any confidential or non-public document or information of a former employer that constitutes or contains any protected trade secret, and will not knowingly use any protected trade secrets of any former employer in the course of his employment hereunder.

Section 24. <u>Entire Agreement</u>. This Agreement constitutes the entire agreement between the parties respecting the employment of Executive, there being no representations, warranties or commitments except as set forth herein.

Section 25. <u>Counterparts</u>. This Agreement may be executed in two counterparts, each of which shall be an original and all of which shall be deemed to constitute one and the same instrument.

Section 26. <u>Withholding</u>. The Employer may withhold from any benefit payment under this Agreement all federal, state, city or other taxes as shall be required pursuant to any law or governmental regulation or ruling; provided that any withholding obligation arising in connection with the exercise of a stock option or the transfer of stock or other property shall be satisfied through withholding an appropriate number of shares of stock or appropriate amount of such other property.

#### Section 27. Definitions.

<u>Accrued Benefits</u> means (i) any compensation deferred by Executive prior to the Date of Termination and not paid by the Employer or otherwise specifically addressed by this Agreement; (ii) any amounts or benefits owing to Executive or to Executive s beneficiaries under the then applicable benefit plans of the Employer; (iii) any amounts owing to Executive for reimbursement of expenses properly incurred by Executive prior to the Date of Termination and which are reimbursable in accordance with <u>Section 7</u>; and (iv) any other benefits or amounts due and owing to Executive under the terms of any plan, program or arrangement of the Employer.

Affiliate means any entity controlled by, in control of, or under common control with, the Employer.

<u>Cause</u> shall be limited to the following events (i) Executive s conviction of, or plea of nolo contendere to, a felony (other than in connection with a traffic violation) under any state or federal law; (ii) Executive s willful and continued failure to substantially perform his essential job functions hereunder after receipt of written notice from the Employer that specifically identifies the manner in which Executive has substantially failed to perform his essential job functions and specifying the manner in which Executive may substantially perform his essential job functions in the future; (iii) a material act of fraud or willful and material misconduct with respect, in each case, to the Employer, by Executive; (iv) a willful and material breach this Agreement; (v) a material breach by Executive of any material written policy of the Employer; or (vi) a failure by Executive to cooperate in any investigation or audit regarding the accounting practices, financial statements, or business practices of the Employer or any of its Affiliates. For purposes of this provision, no act or failure to act, on the part of Executive, shall be considered willful unless it is done, or omitted to be done, by Executive

in bad faith or without reasonable belief that Executive s action or omission was in the best interests of the Employer. Anything herein to the contrary notwithstanding, Executive shall not be terminated for Cause hereunder unless (A) written notice stating the basis for the termination is provided to Executive, (B) as to clauses (ii), (iii), (iv), (v) or (vi) of this paragraph, he is given 10 days to cure the neglect or conduct that is the basis of such claim (it being understood that any errors in expense reimbursement may be cured by repayment), (C) if he fails to cure such neglect or conduct, Executive has an opportunity to be heard before the full Board prior to any vote regarding the existence of Cause and (D) there is a vote of a majority of the members of the Board to terminate him for Cause.

<u>Change in Control</u> means the occurrence of one or more of the following events: (i) any person (as such terms is used in Sections 3(a)(9) and 13(d) of the Securities Exchange Act of 1934 as amended (the <u>Act</u>)) or group (as such term is used in Section 14(d)(d) of the Act) (other than Executive or a group consisting of Executive) becomes a beneficial owner (as such term is used in Rule 13d-3 promulgated under the Act) of more than 30% of the Voting Stock of the Employer; (ii) the majority of the Board consists of individuals other than Incumbent Directors, which term means the members of the Board on the Effective Date; provided that any person becoming a director subsequent to such date whose election or nomination for election was supported by two-thirds of the directors who then comprised the Incumbent Directors shall be considered to be an Incumbent Director; (iii) the Employer adopts any plan of liquidation providing for the distribution of all or substantially all of its assets; (iv) the Employer transfers all or substantially all of its assets or business (unless the shareholders of the Employer immediately prior to such transaction beneficially own, directly or indirectly, in substantially the same proportion as they owned the Voting Stock of the Employer, all of the Voting Stock or other ownership interests of the entity or entities, if any, that succeed to the business of the Employer); or (v) any merger, reorganization, consolidation or similar transaction unless, immediately after consummation of such transaction, the shareholders of the Employer immediately prior to the transaction hold, directly or indirectly, more than 50% of the Voting Stock of the Employer or the Employer s ultimate parent company if the Employer is a subsidiary of another corporation (there being excluded from the number of shares held by such shareholders, but not from the Voting Stock of the combined company, any shares received by Affiliates of such other company in exchange for stock of such other company). For purposes of this Change in Control definition, the <u>Employer</u> shall include any entity that succeeds to all or substantially all of the business of the Employer and <u>Voting Stock</u> shall mean securities of any class or classes having general voting power under ordinary circumstances, in the absence of contingencies, to elect the directors of a corporation.

<u>Confidential Information</u> means information constituting trade secrets or proprietary information belonging to or regarding the Employer or any of its Affiliates or other confidential financial information, operating budgets, strategic plans or research methods, personnel data, projects or plans, or non-public information regarding the Employer or any of its Affiliates. Without limiting the foregoing, Confidential Information shall include, but shall not be limited to, any of the following information relating to the Employer: (i) information regarding the Employer s business proposals, (ii) manner of the Employer s operations, and methods of selling or pricing any products or services; (iii) the identity of persons or entities actually conducting or considering conducting business with the Employer, and any information in any form relating to such persons or entities and their relationship or dealings with the Employer; (iv) any trade secret or confidential information of or concerning any business operation or business relationship; (v) computer databases, software programs and information relating to the nature of the hardware or software and how said hardware or software are used in combination or alone; (vi) information concerning personnel, confidential financial information, customer or customer prospect information, information concerning subscribers, subscriber and customer lists and data, methods and formulas for estimating costs and setting prices, engineering design standards, testing procedures, research results (such as marketing surveys, programming trials or product trials), cost data (such as billing, equipment and programming cost projection models), compensation information and models, business or marketing plans or strategies, deal or business terms, budgets, vendor names, programming operations, product names, information on proposed acquisitions or dispositions, actual performance compared to budgeted performance, long-range plans, internal financial information (including but not limited to financial and operating results for certain offices, divisions, departments, and key market areas that are not disclosed to the

public in such form), results of internal analyses, computer programs and programming information, techniques and designs, and trade secrets; (vii) information concerning the Employer s employees, officers, directors and shareholders; and (viii) any other trade secret or information of a confidential or proprietary nature. For purposes hereof, <u>Employer</u> shall include the Employer and any and all of its Affiliates.

<u>Date of Termination</u> means (i) if Executive s employment is terminated by Executive s death, the date of Executive s death; (ii) if Executive s employment is terminated because of Executive s Disability pursuant to <u>Section 9(a)(ii)(A)</u>, 30 days after Notice of Termination, provided that Executive shall not have returned to the performance of Executive s duties on a full-time basis during such 30-day period; (iii) if Executive s employment is terminated by the Employer pursuant to <u>Section 9(a)(ii)(B)</u> or by Executive pursuant to <u>Section 9(a)(ii)(B)</u>, the date specified in the Notice of Termination; or (iv) if Executive s employment is terminated during the Employment Period other than pursuant to <u>Section 9(a)</u>, the date on which Notice of Termination is given.

Extended Term shall have the meaning set forth in Section 3.

<u>Good Reason</u> means, unless otherwise agreed to in writing by Executive, (i) any adverse change in Executive s titles; (ii) any reduction in Executive s Base Salary; (iii) a material diminution in Executive s authority, responsibilities or duties; (v) the assignment of duties materially inconsistent with Executive s position or status with the Employer as of the date hereof; (vi) a relocation of Executive s primary place of employment to a location more than 50 miles further from the offices of the Employer as of the Effective Time near Los Angeles, California; (vii) any other material breach of the terms of this Agreement or (viii) the failure of the Employer to obtain the assumption in writing of its obligations under this Agreement by any successor to all or substantially all of the assets of the Employer within 15 days after a merger, consolidation, sale or similar transaction. In order to invoke a termination for Good Reason, Executive must notify the Employer of the existence of an event of Good Reason within 90 days of the occurrence of such event, the Employer must fail to cure such event within 30 days of such notice and Executive must terminate his employment within 10 days of the expiration of such period.

<u>Non-Compete Period</u> means the period commencing on the Effective Date and ending twenty-four months after the expiration of the Employment Period; provided that except for purposes of <u>Section 8(e)</u>, in the event Executive s employment is terminated by Employer without Cause or terminated by the Executive for Good Reason the Non-Competition Period shall end on the Date of Termination.

<u>Pro Rata Bonus</u> means an amount equal to the product of (i) the Annual Bonus that would have been earned by Executive for the calendar year that includes the Date of Termination if his employment had not terminated and (ii) a fraction the numerator of which is the number of days that have elapsed as of the Date of Termination during the calendar year that includes the Date of Termination and the denominator of which is 365.

Target Bonus means an amount equal to 175% of Executive s Base Salary.

IN WITNESS WHEREOF, the undersigned have duly executed and delivered this Agreement, or have caused this Agreement to be duly executed and delivered on their behalf.

#### PERINI CORPORATION

 By:
 /s/
 ROBERT BAND

 Name:
 Robert Band

 Title:
 President and Chief Operating Officer

EXECUTIVE

/s/ RONALD N. TUTOR Ronald N. Tutor

#### EXHIBIT A

#### Form of Release

THIS RELEASE (this <u>Release</u>) is made as of this day <u>of</u>, by and between **Perini Corporation**, a Massachusetts corporation (herein referred to as <u>Company</u>), and **Ronald N. Tutor**, an individual (<u>Executive</u>).

#### PRELIMINARY RECITALS

A. Executive s employment with the Company has terminated.

B. Executive and the Company are parties to an Employment Agreement, dated as of the 2nd day of April, 2008 (the <u>Agreement</u>).

#### AGREEMENT

In consideration of the payments due Executive under the Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. Executive, intending to be legally bound, does hereby, on behalf of himself and his agents, representatives, attorneys, assigns, heirs, executors and administrators (collectively, the <u>Executive Parties</u>) REMISE, RELEASE AND FOREVER DISCHARGE the Company, its affiliates, subsidiaries, parents, joint ventures, and its and their officers, directors, shareholders, members, and managers, and its and their respective successors and assigns, heirs, executors, and administrators (collectively, the <u>Company Parties</u>) from all causes of action, suits, debts, claims and demands whatsoever in law or in equity, which Executive or any of the Executive Parties ever had, now has, or hereafter may have, by reason of any matter, cause or thing whatsoever, from the beginning of Executive s initial dealings with the Company to the date of this Release, and particularly, but without limitation of the foregoing general terms, any claims arising from or relating in any way to Executive s employment relationship with Company, the terms and conditions of that employment relationship, and the termination of that employment relationship, including, but not limited to, any claims arising under the Age Discrimination in Employment Act, as amended, 29 U.S.C. § 621 et seq. (ADEA), Title VII of The Civil Rights Act of 1964, as amended, 42 U.S.C. § 2000e et seq., the Civil Rights Act of 1966, 42 U.S.C. §1981, the Civil Rights Act of 1991, Pub. L. No. 102-166, the Americans with Disabilities Act, 42 U.S.C. §12101 et seq., the Age Discrimination in Employment Act, as amended, 29 U.S.C. §621 et seq., the Fair Labor Standards Act, 29 U.S.C. §201 et seq., the National Labor Relations Act, 29 U.S.C. §151 et seq., and any other claims under any federal, state or local common law, statutory, or regulatory provision, now or hereafter recognized, but not including such claims to payments and other rights provided Executive under the Agreement. This Release is effective without regard to the legal nature of the claims raised and without regard to whether any such claims are based upon tort, equity, implied or express contract or discrimination of any sort. Except as specifically provided herein, it is expressly understood and agreed that this Release shall operate as a clear and unequivocal waiver by Executive of any claim for accrued or unpaid wages, benefits or any other type of payment.

2. Executive expressly waives all rights afforded by any statute which limits the effect of a release with respect to unknown claims. Executive understands the significance of his release of unknown claims and his waiver of statutory protection against a release of unknown claims.

3. Executive agrees that he will not be entitled to or accept any benefit from any claim or proceeding within the scope of this Release that is filed or instigated by him or on his behalf with any agency, court or other government entity.

4. The parties agree and acknowledge that the Agreement, and the settlement and termination of any asserted or unasserted claims against the Company and the Company Parties pursuant to this Release, are not and shall not be construed to be an admission of any violation of any federal, state or local statute or regulation, or of any duty owed by the Company or any of the Company Parties to Executive.

5. Executive certifies and acknowledges as follows:

(a) That he has read the terms of this Release, and that he understands its terms and effects, including the fact that he has agreed to RELEASE AND FOREVER DISCHARGE the Company and all Company Parties from any legal action or other liability of any type related in any way to the matters released pursuant to this Release other than as provided in the Agreement and in this Release.

(b) That he understands the significance of his release of unknown claims and his waiver of statutory protection against a release of unknown claims. Accordingly, Executive expressly waives any and all rights and benefits under Section 1542 of the California Civil Code, which states:

A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM MUST HAVE MATERIALLY AFFECTED HIS SETTLEMENT WITH THE DEBTOR.

(c) That he is waiving all rights to sue or obtain equitable, remedial or punitive relief from any or all Company Parties of any kind whatsoever, including, without limitation, reinstatement, back pay, front pay, attorneys fees and any form of injunctive relief. Notwithstanding the above, he further acknowledges that he is not waiving and is not being required to waive (i) any right that cannot be waived under law, including the right to file an administrative charge or to participate in an administrative investigation or proceeding; provided, however, that he disclaims and waives any right to share or participate in any monetary award resulting from the prosecution of such charge or investigation or proceeding, (ii) any claim for indemnity pursuant to the Company s by-laws, articles of incorporation or Section 12 of the Agreement or (iii) any claim for Accrued Benefits (as defined in the Agreement, [or (iv) claim for benefits pursuant to Sections 10(c), 10(d) or 11] [Clause iv to be conformed to delete reference to Section 10(d) or 11 if a Change in Control has not occurred prior to termination date]

(d) That he has signed this Release voluntarily and knowingly in exchange for the consideration described herein, which he acknowledges is adequate and satisfactory to him and which he acknowledges is in addition to any other benefits to which he is otherwise entitled.

(e) That he has been and is hereby advised in writing to consult with an attorney prior to signing this Release.

(f) That he does not waive rights or claims that may arise after the date this Release is executed or those claims arising under the Agreement with respect to payments and other rights due Executive on the date of, or during the period following, the termination of his Employment.

(g) That the Company has provided him with adequate opportunity, including a period of twenty-one (21) days from the initial receipt of this Release and all other time periods required by applicable law, within which to consider this Release (it being understood by Executive that Executive may execute this Release less than 21 days from its receipt from the Company, but agrees that such execution will represent his knowing waiver of such 21-day consideration period), and he has been advised by the Company to consult with counsel in respect thereof.

(h) That he has seven (7) calendar days after signing this Release within which to rescind, in a writing delivered to the Company, the portion of this Release related to claims arising under ADEA or any other claim arising under any other federal, state or local law that requires extension of this revocation right as a condition to the valid release and waiver of such claim.

(i) That at no time prior to or contemporaneous with his execution of this Release has he filed or caused or knowingly permitted the filing or maintenance, in any state, federal or foreign court, or before any local, state, federal or foreign administrative agency or other tribunal, any charge, claim or action of any kind, nature and character whatsoever (<u>Claim</u>), known or unknown, suspected or unsuspected, which he may now have or has ever had against the Company Parties which is based in whole or in part on any matter referred to in Section 1 above; and, subject to the Company s performance under this Release, to the maximum extent permitted by law, Executive is prohibited from filing or maintaining, or causing or knowingly permitting the filing or maintaining, of any such Claim in any such forum. Executive hereby grants the Company his perpetual and irrevocable power of attorney with full right, power and authority to take all actions necessary to dismiss or discharge any such Claim. Executive further covenants and agrees that he will not encourage any person or entity, including but not limited to any current or former employee, officer, director or stockholder of the Company, to institute any Claim against the Company Parties or any of them, and that except as expressly permitted by law or administrative policy or as required by legally enforceable order he will not aid or assist any such person or entity in prosecuting such Claim.

#### 6. Miscellaneous

(a) This Release and the Agreement, and any other documents expressly referenced therein, constitute the complete and entire agreement and understanding of Executive and the Company with respect to the subject matter hereof, and supersedes in its entirety any and all prior understandings, commitments, obligations and/or agreements, whether written or oral, with respect thereto; it being understood and agreed that this Release and including the mutual covenants, agreements, acknowledgments and affirmations contained herein, is intended to constitute a complete settlement and resolution of all matters set forth in Section 1 hereof.

(b) The Company Parties are intended third-party beneficiaries of this Release, and this Release may be enforced by each of them in accordance with the terms hereof in respect of the rights granted to such Company Parties hereunder. Except and to the extent set forth in the preceding two sentences, this Release is not intended for the benefit of any Person other than the parties hereto, and no such other person or entity shall be deemed to be a third party beneficiary hereof. Without limiting the generality of the foregoing, it is not the intention of the Company to establish any policy, procedure, course of dealing or plan of general application for the benefit of or otherwise in respect of any other employee, officer, director or stockholder, irrespective of any similarity between any contract, agreement, commitment or understanding between the Company and such other employee, officer, director or stockholder, on the one hand, and any contract, agreement, commitment or understanding between the Company and Executive, on the other hand, and irrespective of any similarity in facts or circumstances involving such other employee, officer, director or stockholder, on the other hand.

(c) The invalidity or unenforceability of any provision of this Release shall not affect the validity or enforceability of any other provision of this Release, which shall otherwise remain in full force and effect.

(d) This Release may be executed in separate counterparts, each of which shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.

(e) The obligations of each of the Company and Executive hereunder shall be binding upon their respective successors and assigns. The rights of each of the Company and Executive and the rights of the Company Parties shall inure to the benefit of, and be enforceable by, any of the Company s, Executive s and the Company Parties respective successors and assigns. The Company may assign all rights and obligations of this Release to any successor in interest to the assets of the Company.

(f) No amendment to or waiver of this Release or any of its terms shall be binding upon any party hereto unless consented to in writing by such party.

(g) ALL ISSUES AND QUESTIONS CONCERNING THE CONSTRUCTION, VALIDITY, ENFORCEMENT AND INTERPRETATION OF THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE, WITHOUT GIVING EFFECT TO ANY CHOICE OF LAW OR CONFLICT OF LAW PROVISION OR RULE (WHETHER OF THE STATE OF DELAWARE OR ANY OTHER JURISDICTION) THAT WOULD CAUSE THE APPLICATION OF THE LAW OF ANY JURISDICTION OTHER THAN THE STATE OF DELAWARE.

\* \* \* \* \*

Intending to be legally bound hereby, Executive and the Company have executed this Release as of the date first written above.

RONALD N. TUTOR

By:

**Ronald N. Tutor** 

PERINI CORPORATION

By: Name: Title:

Robert Band President and Chief Operating Officer

#### **READ CAREFULLY BEFORE SIGNING**

I have read this Release and have been given adequate opportunity, including 21 days from my initial receipt of this Release, to review this Release and to consult legal counsel prior to my signing of this Release. I understand that by executing this Release I will relinquish certain rights or demands I may have against the Company Parties or any of them.

[Name]

Witness:

Annex D

#### PROMISSORY NOTE

\$[ ] \_\_\_\_\_\_\_, 2008
FOR VALUE RECEIVED, Tutor-Saliba Corporation, a California corporation (the <u>Company</u>) hereby promises to pay to [Shareholder] (<u>Holder</u>)
or its registered assigns the principal sum of \_\_\_\_\_\_ Dollars (\$[ ]), together with interest thereon calculated from the date hereof in accordance
with the provisions of this note (this <u>Note</u>).

This Note is issued in the manner contemplated by Section 6.20 of that certain Agreement and Plan of Merger, dated as of April 2, 2008 (the <u>Merger Agreement</u>), by and among Perini Corporation, a Massachusetts corporation (<u>Perini</u>), Trifecta Acquisition LLC, a California corporation and a wholly-owned subsidiary of Perini (<u>Merger Sub</u>), the Company, Ronald N. Tutor, a resident of California, and shareholders of the Company signatory thereto, pursuant to which, among other things, the Company shall merge with and into Merger Sub, with Merger Sub to continue as the surviving entity in the merger and as a wholly-owned subsidiary of Perini on the terms and subject to the conditions set forth in the Merger Agreement.

This Note evidences the absolute and unconditional obligation of the Company. The Company is issuing notes, together with this Note, equal to an aggregate principal amount of \_\_\_\_\_ Dollars ([]) (the <u>Aggregate Principal Amo</u>unt ) in connection with the issuance of this Note (all such notes, including the Note, the <u>Total Dividend Notes</u> ).

1. <u>Scheduled Payments</u>: Subject to the terms of Section 4:

(a) <u>Principal</u>. The entire unpaid principal balance of this Note (together with all accrued and unpaid interest thereupon) shall become due and payable in full on June 30, 2012 (the <u>Maturity Date</u>).

(b) <u>Interest</u>. Interest shall accrue on the unpaid principal amount of this Note at a rate per annum equal to 5% per annum. Interest shall be calculated on the basis of the actual number of days elapsed and a year of 365 days, and be payable in cash on the last day of each March, June, September and December of each applicable year (each a <u>Quarterly Payment Date</u>); provided that, in the event that the Company is prohibited from paying interest in cash pursuant to the terms of any Senior Indebtedness (defined below), such interest owing will be added to the principal amount outstanding on each Quarterly Payment Date and treated as principal for all purposes under this Note on and after such date.

(c) Excess Cash Flow Mandatory Prepayments. Not later than 90 days after the end of each fiscal year (or for the year in which the Closing (as defined in the Merger Agreement) occurs, partial fiscal year as described in clause (x) below) of the Company in which there is Excess Cash Flow of the Company, the Company, to the extent not prohibited by the terms of any Senior Indebtedness or Applicable Law, will make a prepayment of principal outstanding under this Note, in an amount equal to (x) 100% of Excess Cash Flow for such fiscal year (or for the fiscal year in which the Closing occurs, the partial fiscal year for the period from the Closing Date (as defined in the Merger Agreement) through the end of such fiscal year) multiplied by (y) a fraction, the numerator of which is the principal outstanding under the Total Dividend Notes.

(d) <u>Optional Prepayments</u>. The Company may at any time, and from time to time, prepay, without premium or penalty, all or any portion of the Company s obligations under this Note. All such prepayments shall be applied first to pay all accrued but unpaid interest and then to pay outstanding principal.

2. <u>Payment of Note</u>. All payments and prepayments of principal of and interest on this Note shall be made to the registered Holder, in lawful money of the United States of America by wire transfer of immediately

available funds to a United States bank account designated in writing by the Holder (or at such other place as the holder hereof shall notify the Company in writing).

3. <u>{Reserved}</u>. [To include any subordination provisions requested by any lender or surety of Perini or the Company.]

4. <u>Event of Default: Consequences</u>. If the Company fails to pay when due any amount (whether interest, principal or other amount) then payable on this Note, then, subject to the provisions of Section 3, the Holder may, by notice of default and acceleration given to the Company, declare the entire outstanding principal amount of this Note, together with all accrued and unpaid interest thereon, immediately due and payable.

5. <u>Waiver of Presentment</u>. The Company hereby waives presentment for payment, demand, protest, and notice of demand, protest and nonpayment, and any other notice that might be required by law, and consents to any and all renewals or extensions that might be made by the Holder as to the time of payment of this Note from time to time.

#### 6. Replacement and Cancellation.

(a) <u>Replacement of Lost Note</u>. Upon receipt of evidence reasonably satisfactory to the Company (an affidavit of the Holder shall be satisfactory) of the ownership and the loss, theft, destruction or mutilation of this Note and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to the Company (provided that, if the holder is an employee of the Company, its own agreement shall be satisfactory), or, in the case of any such mutilation, upon the surrender of such Note to the Company at its principal office, the Company shall (at its expense) execute and deliver, in lieu thereof, a new Note of the same class and representing the same rights represented by such lost, stolen, destroyed or mutilated Note and dated so that there will be no loss of interest on such Note. Any Note in lieu of which any such new Note has been so executed and delivered by the Company shall not be deemed to be an outstanding Note.

(b) <u>Cancellation</u>. After all principal, accrued interest and all other amounts at any time owed on this Note have been paid in full, this Note shall be surrendered to the Company for cancellation.

7. <u>Business Days</u>. If any payment is due, or any time period for giving notice or taking action expires, on a day which is not a business day, in the State of New York, the payment shall be due and payable on, and the time period shall automatically be extended to, the next business day immediately following, and interest shall continue to accrue at the required rate hereunder until any such payment is made.

8. <u>Governing Law</u>. This Note shall be governed by and construed in accordance with the internal laws of the State of New York, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of New York or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of New York.

9. <u>Merger Agreement</u>. This Note has been executed and delivered pursuant to and in accordance with the terms and conditions of the Merger Agreement, as in effect on the date hereof. Capitalized terms used in this Note without separate definition shall have the respective meanings given to them in the Merger Agreement.

10. <u>Successors and Assigns</u>. This Note may not be assigned or transferred by the Company or the Holder without the prior written consent of the other party hereto. Each assignment or transfer shall be recorded by the Company in the Register. Any transfer or assignment in violation of this Section 10 shall be void, and the Company shall not recognize such purported transferee as a holder of this Note. The terms and provisions of this Note shall inure to the benefit of the Agent and each of the other holders of the Designated Senior Indebtedness.

D-2

11. <u>Register</u>. The Company shall maintain a register (the <u>Register</u>) in which it records the name and address of each Holder of a Note and the principal amount and interest outstanding with respect thereto from time to time. The parties shall treat the registered holder of a Note as the Holder for all purposes, absent manifest error.

12. <u>Amendments</u>. The provisions of this Note may not be amended, supplemented, replaced or otherwise modified in any manner without the prior written consent of the Holder and the Company. To the extent that the Credit Agreement is replaced or refinanced, the Holder and the Company will consent to (and execute and deliver) any amendments reasonably necessary to give effect to the documentation governing such replacement or refinancing indebtedness to preserve the economic position of the parties hereto.

13. Defined Terms. The following terms shall have the following meanings when used herein:

Agent means the agent under the Credit Agreement. In the event the Credit Agreement is with a single lender, Agent shall refer to such lender.

<u>Bankruptcy Law</u> shall mean Title 11 of the United States Code entitled Bankruptcy, as now and hereafter in effect, or any successor statute, and any similar law for the relief of debtors.

<u>Consolidated Net Income</u> shall mean, for any period, the net income of the Company and its subsidiaries (the <u>Consolidated Group</u>) for such period determined on a consolidated basis in accordance with GAAP.

<u>Credit Agreement</u> shall mean any [insert any credit facility that will be pending as of the Closing], as amended, supplemented, amended and restated or otherwise modified or as refinanced, extended, renewed, defeased, restructured, replaced or refinanced from time to time.

\_\_\_\_\_Designated Senior Indebtedness means the Senior Indebtedness in respect of the Secured Obligations.

Excess Cash Flow shall mean, for any period, an amount equal to the excess of:

(a) the sum, without duplication, of

(i) Consolidated Net Income for such period,

(ii) an amount equal to the amount of all non-cash charges to the extent deducted in arriving at such Consolidated Net Income,

(iii) an amount equal to the provision for taxes based on income, profits or capital of the Consolidated Group, including federal, foreign, state, franchise, excise and similar taxes and foreign withholding taxes paid during such period (or accrued during such period and payable within 180 days after the last day of such period) to the extent deducted in arriving at such Consolidated Net Income, and

(iv) an amount equal to the aggregate consolidated net non-cash loss on the sale, lease, transfer or other disposition of assets by the Consolidated Group during such period (other than sales, leases, transfers or other dispositions in the ordinary course of business) to the extent deducted in arriving at such Consolidated Net Income,

less

(b) the sum, without duplication, of

(i) an amount equal to the amount of all non-cash credits included in arriving at such Consolidated Net Income and included in arriving at such Consolidated Net Income,

D-3

(ii) the consolidated amount of capital expenditures made in cash during such period, except to the extent that such capital expenditures were financed with the proceeds of indebtedness,

(iii) the aggregate consolidated amount of all principal payments of indebtedness of the Consolidated Group made during such period, including payments made under the Total Dividend Notes (other than in respect of any revolving credit facility to the extent there is not an equivalent permanent reduction in commitments thereunder), except to the extent financed with the proceeds of other Indebtedness,

(iv) an amount equal to the aggregate net non-cash gain on the sale, lease, transfer or other disposition of assets by the Consolidated Group during such period (other than sales in the ordinary course of business) to the extent included in arriving at such Consolidated Net Income,

(v) the amount of taxes (including penalties and interest) paid in cash in such period, and

(vi) the amount of cash expenses paid to effect the transaction described in the Merger Agreement.

<u>Note Payment</u> means any payment or distribution of any kind or character, whether direct or indirect, by setoff or otherwise, from any source, whether in cash, property or securities, on or in respect of the Note or any amounts due thereunder or on account of any purchase, redemption, rescission of purchase, defeasance or other acquisition of the Note.

<u>Obligations</u> shall mean any and all obligations (including guaranty obligations) with respect to the payment and performance of (a) any principal of or interest or premium on any indebtedness, including, without limitation, any reimbursement obligation in respect of any letter of credit, or any other liability, (b) any fees, indemnification obligations, expense reimbursement obligations or other liabilities payable under the documentation governing any indebtedness (including, without limitation, the retaking, holding, selling or otherwise disposing of or realizing on any collateral), (c) any obligation to post cash collateral in respect of letters of credit or any other obligations, and (d) all performance obligations under the documentation governing any indebtedness.

<u>Proceeding</u> shall mean (a) any voluntary or involuntary case or proceeding under any applicable Bankruptcy Law with respect to the Company, (b) any other voluntary or involuntary insolvency, reorganization, arrangement or bankruptcy case or proceeding, or any receivership, liquidation, reorganization, arrangement or other similar case or proceeding with respect to the Company or with respect to a material portion of its respective assets, (c) any liquidation, dissolution, reorganization or winding up of the Company whether voluntary or involuntary and whether or not involving insolvency or bankruptcy or (d) any assignment for the benefit of creditors or any other marshalling of assets and liabilities of the Company.

<u>Remedies</u> shall mean the acceleration of the Note or the exercise of any remedies in respect of any default thereunder (including, without limitation, any right of setoff or similar right, any right of collection, repayment, prepayment, repurchase, redemption or put right, the right to sue the Company or to file or participate in any involuntary Proceeding or take any other action under law, against the Company but excluding in any event the imposition additional interest pursuant to Section 1(b)).

Secured Obligations shall mean all Obligations outstanding under the Credit Agreement and the other Loan Documents (as defined therein).

<u>Senior Indebtedness</u> means all Secured Obligations and all other indebtedness for borrowed money owing by the Company other than any indebtedness which is expressly subordinated to the indebtedness owing under this Note and any amounts owing to trade creditors incurred in the ordinary course of business, in each case whether for (i) principal, (ii) interest (including interest accruing after the commencement of a Proceeding, whether or not allowed as a claim in such Proceeding), (iii) out-of-pocket costs, (iv) fees (including reasonable attorneys fees and out-of-pocket disbursements), (v) out-of-pocket expenses, and (vi) otherwise, in each case whether or not allowed as a claim in any Proceeding.

14. <u>Third Party Beneficiaries</u>. The holders of Designated Senior Indebtedness are intended third party beneficiaries hereof and the subordination provisions set forth herein are for the benefit of the holders of Designated Senior Indebtedness (and their successors and assigns) and shall be enforceable by them directly against the Holder (and its successors and assigns). This Section 14 shall constitute a continuing offer to all persons who become holders of or who continue to hold Designated Senior Indebtedness (whether such Designated Senior Indebtedness was created or acquired before or after the issuance of the Notes).

\* \* \* \* \* \*

IN WITNESS WHEREOF, the undersigned has duly executed and delivered this Note as of the date first written above.

# TUTOR-SALIBA CORPORATION

By: Title:

ACCEPTED AND AGREED:

## [SHAREHOLDER]

By: Title:

D-6

Annex E

UBS Securities LLC

Chicago, IL 60606

www.ubs.com

April 2, 2008

The Special Committee of the Board of Directors

Perini Corporation

73 Mt. Wayte Avenue

Framingham, MA 01701

Dear Members of the Special Committee:

We understand that Perini Corporation, a Massachusetts corporation (Perini or the Company), is considering a transaction (the Transaction) whereby Tutor-Saliba Corporation, a California corporation (TSC), will be merged with and into Trifecta Acquisition LLC, a California limited liability company and a wholly owned subsidiary of the Company (Merger Sub). Pursuant to the terms of an Agreement and Plan of Merger, draft dated as of April 2, 2008 (the Agreement), among the Company, Merger Sub, TSC, Ronald N. Tutor and other shareholders of TSC, (i) TSC will declare and distribute after the date hereof to the holders of common stock, no par value per share, of TSC (TSC Common Stock) a dividend of up to \$120 million consisting of cash or a combination of cash and one or more Dividend Notes (as defined in the Agreement); and (ii) TSC will merge with Merger Sub, with Merger Sub surviving the merger as a wholly owned subsidiary of the Company, and each issued and outstanding share of TSC Common Stock (other than certain shares specified in the Agreement) will be converted into the right to receive the number of shares of common stock, par value \$1.00 per share, of the Company (Company Common Stock) equal to the quotient (the Exchange Ratio) obtained by dividing (A) 22,987,293 by (B) the Outstanding Common Stock Number (as defined in the Agreement), which shares of Company Common Stock will be subject to the terms of the Shareholders Agreement (as defined in the Agreement). The terms and conditions of the Transaction are more fully set forth in the Agreement and the Shareholders Agreement.

You have requested our opinion as to the fairness, from a financial point of view, to the Company of the Exchange Ratio provided for in the Transaction.

UBS Securities LLC (UBS) has acted as financial advisor to the Special Committee of the Board of Directors of the Company (the Special Committee) in connection with the Transaction and will receive a fee for its services, a portion of which is payable in connection with this opinion and a significant portion of which is contingent upon consummation of the Transaction. In the past, UBS and its affiliates have provided investment banking services to the Company and TSC unrelated to the proposed Transaction, for which UBS and its affiliates have received compensation, including having acted as underwriter in an offering of 10,341,909 shares of Company Common Stock in December 2005, including 4,240,229 shares of Company Common Stock that were beneficially owned by TSC and Ronald N. Tutor, and as agent in September 2006 in a private placement of 400,000 shares of Company Common Stock that were beneficially owned by TSC and Ronald N. Tutor. UBS and its affiliates may receive compensation, although there are no such services currently contemplated. In the ordinary course of business, UBS and its affiliates may hold or trade, for their own accounts and the accounts of their customers, securities of the Company and, accordingly, may at any time hold a long or short position in such securities. The issuance of this opinion was approved by an authorized committee of UBS.

Our opinion does not address the relative merits of the Transaction as compared to other business strategies or transactions that might be available to the Company or the Company s underlying business decision to effect the Transaction. Our opinion does not constitute a recommendation to any shareholder as to how such shareholder should vote or act with respect to the Transaction. At your direction, we have not been asked to, nor do we, offer any opinion as to the terms, other than the Exchange Ratio to the extent expressly specified herein, of the Agreement or the form of the Transaction. In addition, we express no opinion as to the fairness of the amount or nature of, or any other aspects relating to, any compensation to be received by any officers, directors or employees of any parties to the Transaction, or any class of such persons, relative to the Exchange Ratio. We express no opinion as to what the value of Company Common Stock will be when issued pursuant to the Transaction or the prices at which Company Common Stock will trade at any time. In rendering this opinion, we have assumed, with your consent, that (i) the final executed form of each of the Agreement and the Shareholders Agreement will not differ in any material respect from the most recent draft that we have reviewed, (ii) the parties to the Agreement without any adverse waiver or amendment of any material term or condition thereof. We have also assumed that all governmental, regulatory or other consents and approvals necessary for the consummation of the Transaction will be obtained without any material adverse effect on the Company, TSC or the Transaction.

In arriving at our opinion, we have, among other things: (i) reviewed certain publicly available business and financial information relating to the Company; (ii) reviewed certain internal financial information and other data relating to the businesses and financial prospects of the Company that were provided to us by the Company and not publicly available, including the financial forecasts and estimates prepared by the management of the Company and the Special Committee that you have directed us to utilize for purposes of our analysis; (iii) reviewed certain internal financial information and other data relating to the businesses and financial prospects of TSC that were provided to us by the Company and not publicly available, including the financial forecasts and estimates prepared by the management of the Company, the management of TSC and the Special Committee that you have directed us to utilize for purposes of our analysis; (iv) reviewed certain estimates of synergies prepared by the management of the Company, the management of TSC and the Special Committee, that were provided to us by the Company and not publicly available, that you have directed us to utilize for purposes of our analysis; (v) conducted discussions with the Special Committee and members of the senior managements of the Company and TSC concerning the businesses and financial prospects of the Company and TSC; (vi) reviewed publicly available financial and stock market data with respect to certain other companies that are generally in the industry in which the Company and TSC operate; (vii) compared the financial terms of the Transaction with the publicly available financial terms of other transactions involving certain other companies that are generally in the industry in which the Company and TSC operate; (viii) performed discounted cash flow analyses of the Company and TSC in which we analyzed the future cash flows of the Company and TSC using the financial forecasts and estimates prepared by the management of the Company, the management of TSC and the Special Committee that you have directed us to utilize for purposes of our analysis; (ix) reviewed current and historical market prices of Company Common Stock; (x) considered certain pro forma effects of the Transaction on the Company s financial statements; (xi) reviewed the Agreement and the Shareholders Agreement; and (xii) conducted such other financial studies, analyses and investigations, and considered such other information, as we deemed necessary or appropriate.

E-2

As you are aware, the financial and operating characteristics of the Company and TSC (including the business segments in which they operate) cause their financial results to have limited comparability, for valuation purposes, to those of other companies with publicly available financial and stock market data that we have reviewed and of other transactions that we have reviewed for which financial terms are publicly available, and, accordingly, we have relied primarily on discounted cash flow analyses of the forecasts and estimates referred to above for purposes of our opinion.

In connection with our review, with your consent, we have assumed and relied upon, without independent verification, the accuracy and completeness in all material respects of the information provided to or reviewed by us for the purpose of this opinion. In addition, with your consent, we have not made any independent evaluation or appraisal of any of the assets or liabilities (contingent or otherwise) of the Company or TSC, nor have we been furnished with any such evaluation or appraisal. With respect to the financial forecasts, estimates, synergies and pro forma effects referred to above, we have assumed, at your direction, that they have been reasonably prepared on a basis reflecting the best currently available estimates and judgments of the Special Committee as to the future financial forecasts and estimates, including synergies, referred to above will be achieved at the times and in the amounts projected. We also have assumed, with your consent, that the Transaction will qualify for U.S. federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended. Our opinion is necessarily based on economic, monetary, market and other conditions as in effect on, and the information available to us as of, the date hereof.

Based upon and subject to the foregoing, it is our opinion that, as of the date hereof, the Exchange Ratio provided for in the Transaction is fair, from a financial point of view, to the Company.

This opinion is provided for the benefit of the Special Committee in connection with, and for the purpose of, its evaluation of the Transaction.

Very truly yours,

UBS SECURITIES LLC

E-3

Annex F

## THE COMMONWEALTH OF MASSACHUSETTS

William Francis Galvin

Secretary of the Commonwealth

One Ashburton Place, Boston, Massachusetts 02108-1512

ARTICLES OF AMENDMENT

(General Laws, Chapter 156D, Section 10.06)

We, Robert Band, President and Susan Mellace, Clerk of Perini Corporation (the Company ), located at 73 Mt. Wayte Avenue, Framingham, Massachusetts 01701, certify that these Articles of Amendment effecting article numbered 3 of the Restated Articles of Organization of the Company, as amended and corrected (the Articles of Organization ), were duly adopted at a meeting held on [\_\_\_\_\_], 2008, by vote of [\_\_\_\_\_] shares of Common Stock out of [\_\_\_\_\_\_] shares outstanding, being at least a majority of each type, class or series outstanding and entitled to vote thereon and that such approval by the shareholders of the Company constitutes approval of these Articles of Amendment in the manner required by Chapter 156D and by the Articles of Organization of the Company.

To change the number of shares and par value (if any) of any type, class or series of stock which the corporation is authorized to issue, fill in the following:

The total presently authorized is:

#### Without Par Value Туре Number of Shares Par Value Common: None Preferred: None With Par Value Number of Shares Par Value Туре Common: 40.000.000 1.00 Preferred: 1,000,000 1.00 Change the total authorized to: Without Par Value Number of Shares Par Value Type Common: None Preferred: None With Par Value Туре Number of Shares Par Value Common: 75,000,000 1.00 Preferred: 1,000,000 1.00

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

The foregoing amendments will become effective when these Articles of Amendment are filed in accordance with General Laws, Chapter 156D, unless these articles specify in accordance with the vote adopting

the amendment, a later effective date not more than ninety days after such filing, in which event the amendment will become effective on such later date.

Later effective date:

Signed under the penalties of perjury, this [\_\_\_\_] day of [\_\_\_\_], 2008.

By: Name: Title:

Robert Band President

By: Name: Title:

Susan Mellace Clerk

F-2

Annex G

#### PERINI CORPORATION

#### 2004 STOCK OPTION AND INCENTIVE PLAN

#### (as amended on May 18, 2006)

#### SECTION 1. GENERAL PURPOSE OF THE PLAN; DEFINITIONS

The name of the plan is the Perini Corporation 2004 Stock Option and Incentive Plan (the Plan ). The purpose of the Plan is to encourage and enable the officers, employees, Non-Employee Directors and other key persons (including consultants and prospective employees) of Perini Corporation (the Company ) and its Subsidiaries upon whose judgment, initiative and efforts the Company largely depends for the successful conduct of its business to acquire a proprietary interest in the Company. It is anticipated that providing such persons with a direct stake in the Company s welfare will assure a closer identification of their interests with those of the Company, thereby stimulating their efforts on the Company s behalf and strengthening their desire to remain with the Company.

The following terms shall be defined as set forth below:

Act means the Securities Act of 1933, as amended, and the rules and regulations thereunder.

Administrator is defined in Section 2(a).

Award or Awards, except where referring to a particular category of grant under the Plan, shall include Incentive Stock Options, Non-Qualified Stock Options, Stock Appreciation Rights, Deferred Stock Awards, Restricted Stock Awards, Unrestricted Stock Awards and Dividend Equivalent Rights.

*Board* means the Board of Directors of the Company.

*Code* means the Internal Revenue Code of 1986, as amended, and any successor Code, and related rules, regulations and interpretations.

Committee means the Committee of the Board referred to in Section 2.

Covered Employee means an employee who is a Covered Employee within the meaning of Section 162(m) of the Code.

Deferred Stock Award means Awards granted pursuant to Section 8.

Dividend Equivalent Right means Awards granted pursuant to Section 11.

*Effective Date* means the date on which the Plan is approved by stockholders as set forth in Section 17.

Exchange Act means the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder.

*Fair Market Value* of the Stock on any given date means the fair market value of the Stock determined by its closing price on the New York Stock Exchange. If there are no market quotations for such date, the determination shall be made by reference to the last date preceding such date for which there are market quotations.

Incentive Stock Option means any Stock Option designated and qualified as an incentive stock option as defined in Section 422 of the Code.

Non-Qualified Stock Option means any Stock Option that is not an Incentive Stock Option.

Option or Stock Option means any option to purchase shares of Stock granted pursuant to Section 5.

*Performance Cycle* means one or more periods of time, which may be of varying and overlapping durations, as the Administrator may select, over which the attainment of one or more performance criteria will be measured for the purpose of determining a grantee s right to and the payment of a Restricted Stock Award or Deferred Stock Award.

Restricted Stock Award means Awards granted pursuant to Section 7.

Stock means the Common Stock, par value \$1.00 per share, of the Company, subject to adjustments pursuant to Section 3.

Stock Appreciation Right means any Award granted pursuant to Section 6.

*Subsidiary* means any corporation or other entity (other than the Company) in which the Company has a controlling interest, either directly or indirectly.

Unrestricted Stock Award means any Award granted pursuant to Section 9.

#### SECTION 2. ADMINISTRATION OF PLAN; ADMINISTRATOR AUTHORITY TO SELECT GRANTEES AND DETERMINE AWARDS

(a) <u>Committee</u>. The Plan shall be administered by the Compensation Committee of the Board. (the Administrator ).

(b) <u>Powers of Administrator</u>. The Administrator shall have the power and authority to grant Awards consistent with the terms of the Plan, including the power and authority:

(i) to select the individuals to whom Awards may from time to time be granted;

(ii) to determine the time or times of grant, and the extent, if any, of Incentive Stock Options, Non-Qualified Stock Options, Stock Appreciation Rights, Restricted Stock Awards, Deferred Stock Awards, Unrestricted Stock Awards and Dividend Equivalent Rights, or any combination of the foregoing, granted to any one or more grantees;

(iii) to determine the number of shares of Stock to be covered by any Award;

(iv) to determine and modify from time to time the terms and conditions, including restrictions, not inconsistent with the terms of the Plan, of any Award, which terms and conditions may differ among individual Awards and grantees, and to approve the form of written instruments evidencing the Awards;

(v) to accelerate at any time the exercisability or vesting of all or any portion of any Award;

(vi) subject to the provisions of Section 5(a)(ii), to extend at any time the period in which Stock Options may be exercised;

(vii) to determine at any time whether, to what extent, and under what circumstances distribution or the receipt of Stock and other amounts payable with respect to an Award shall be deferred either automatically or at the election of the grantee and whether and to what extent the Company shall pay or credit amounts constituting interest (at rates determined by the Administrator) or dividends or deemed dividends on such deferrals; and

(viii) at any time to adopt, alter and repeal such rules, guidelines and practices for administration of the Plan and for its own acts and proceedings as it shall deem advisable; to interpret the terms and provisions of the Plan and any Award (including related written instruments); to make all determinations it deems advisable for the administration of the Plan; to decide all disputes arising in connection with the Plan; and to otherwise supervise the administration of the Plan.

All decisions and interpretations of the Administrator shall be binding on all persons, including the Company and Plan grantees.

(c) <u>Delegation of Authority to Grant Awards</u>. The Administrator, in its discretion, may delegate to the Chief Executive Officer of the Company all or part of the Administrator s authority and duties with respect to the granting of Awards, to individuals who are not subject to the reporting and other provisions of Section 16 of the Exchange Act or covered employees within the meaning of Section 162(m) of the Code. Any such delegation by the Administrator shall include a limitation as to the amount of Awards that may be granted during the period of the delegation and shall contain guidelines as to the determination of the exercise price of any Stock Option or Stock Appreciation Right, the conversion ratio or price of other Awards and the vesting criteria. The Administrator may revoke or amend the terms of a delegation at any time but such action shall not invalidate any prior actions of the Administrator s delegate or delegates that were consistent with the terms of the Plan.

(d) <u>Indemnification</u>. Neither the Board nor the Committee, nor any member of either or any delegatee thereof, shall be liable for any act, omission, interpretation, construction or determination made in good faith in connection with the Plan, and the members of the Board and the Committee (and any delegatee thereof) shall be entitled in all cases to indemnification and reimbursement by the Company in respect of any claim, loss, damage or expense (including, without limitation, reasonable attorneys fees) arising or resulting therefrom to the fullest extent permitted by law and/or under any directors and officers liability insurance coverage which may be in effect from time to time.

#### SECTION 3. STOCK ISSUABLE UNDER THE PLAN; MERGERS; SUBSTITUTION

(a) <u>Stock Issuable</u>. The maximum number of shares of Stock reserved and available for issuance under the Plan shall be 3,000,000 shares, subject to adjustment as provided in Section 3(b). For purposes of this limitation, the shares of Stock underlying any Awards (including any awards granted pursuant to the Company s Special Equity Incentive Plan which are forfeited, canceled, held back upon exercise of an Option or settlement of an Award to cover the exercise price or tax withholding, reacquired by the Company prior to vesting, satisfied without the issuance of Stock or otherwise terminated (other than by exercise)) shall be added back to the shares of Stock available for issuance under the Plan. Subject to such overall limitations, shares of Stock may be issued up to such maximum number pursuant to any type or types of Award; provided, however, that Stock Options or Stock Appreciation Rights with respect to no more than 200,000 shares of Stock may be granted to any one individual grantee during any one calendar year period. The shares available for issuance under the Plan may be authorized but unissued shares of Stock or shares of Stock reacquired by the Company.

(b) <u>Changes in Stock</u>. Subject to Section 3(c) hereof, if, as a result of any reorganization, recapitalization, reclassification, stock dividend, stock split, reverse stock split or other similar change in the Company s capital stock, the outstanding shares of Stock are increased or decreased or are exchanged for a different number or kind of shares or other securities of the Company, or additional shares or new or different shares or other securities of the Company or other solution, respectively.

if, as a result of any merger or consolidation, sale of all or substantially all of the assets of the Company, the outstanding shares of Stock are converted into or exchanged for a different number or kind of securities of the Company or any successor entity (or a parent or subsidiary thereof), the Administrator shall make an appropriate or proportionate adjustment in (i) the maximum number of shares reserved for issuance under the Plan, (ii) the number of Stock Options or Stock Appreciation Rights that can be granted to any one individual grantee and the maximum number of shares that may be granted under a Performance-based Award, (iii) the number and kind of shares or other securities subject to any then outstanding Awards under the Plan, (iv) the repurchase price, if any, per share subject to each outstanding Restricted Stock Award, and (v) the price for each share subject to any then outstanding Stock Options and Stock Appreciation Rights under the Plan, without changing the aggregate exercise price (i.e., the exercise price multiplied by the number of Stock Options and Stock Appreciation Rights) as to which such Stock Options and Stock shall be issued under the Plan resulting from any such adjustment, but the Administrator in its discretion may make a cash payment in lieu of fractional shares.

The Administrator may also adjust the number of shares subject to outstanding Awards and the exercise price and the terms of outstanding Awards to take into consideration material changes in accounting practices or principles, extraordinary dividends, acquisitions or dispositions of stock or property or any other event if it is determined by the Administrator that such adjustment is appropriate to avoid distortion in the operation of the Plan, provided that no such adjustment shall be made in the case of an Incentive Stock Option, without the consent of the grantee, if it would constitute a modification, extension or renewal of the Option within the meaning of Section 424(h) of the Code.

(c) Mergers and Other Transactions. In the case of and subject to the consummation of (i) the dissolution or liquidation of the Company, (ii) the sale of all or substantially all of the assets of the Company on a consolidated basis to an unrelated person or entity, (iii) a merger, reorganization or consolidation in which the outstanding shares of Stock are converted into or exchanged for a different kind of securities of the successor entity and the holders of the Company s outstanding voting power immediately prior to such transaction do not own 40 percent or more of the outstanding voting power of the successor entity immediately upon completion of such transaction, or (iv) the sale of 60 percent or more of the Stock of the Company to an unrelated person or entity (in each case, a Sale Event ), all Options and Stock Appreciation Rights that are not exercisable immediately prior to the effective time of the Sale Event shall become fully exercisable as of the effective time of the Sale Event and all other Awards shall become fully vested and nonforfeitable as of the effective time of the Sale Event, except as the Administrator may otherwise specify with respect to particular Awards in the relevant Award documentation. Upon the effective time of the Sale Event, the Plan and all outstanding Awards granted hereunder shall terminate, unless provision is made in connection with the Sale Event in the sole discretion of the parties thereto for the assumption or continuation of Awards theretofore granted by the successor entity, or the substitution of such Awards with new Awards of the successor entity or parent thereof, with appropriate adjustment as to the number and kind of shares and, if appropriate, the per share exercise prices, as such parties shall agree (after taking into account any acceleration hereunder). In the event of such termination, each grantee shall be permitted, within a specified period of time prior to the consummation of the Sale Event as determined by the Administrator, to exercise all outstanding Options and Stock Appreciation Rights held by such grantee, including those that will become exercisable upon the consummation of the Sale Event; provided, however, that the exercise of Options and Stock Appreciation Rights not exercisable prior to the Sale Event shall be subject to the consummation of the Sale Event.

Notwithstanding anything to the contrary in this Section 3(c), in the event of a Sale Event pursuant to which holders of the Stock of the Company will receive upon consummation thereof a cash payment for each share surrendered in the Sale Event, the Company shall have the right, but not the obligation, to make or provide for a cash payment to the grantees holding Options and Stock Appreciation Rights, in exchange for the cancellation

thereof, in an amount equal to the difference between (A) the value as determined by the Administrator of the consideration payable per share of Stock pursuant to the Sale Event (the Sale Price) times the number of shares of Stock subject to outstanding Options and Stock Appreciation Rights (to the extent then exercisable at prices not in excess of the Sale Price) and (B) the aggregate exercise price of all such outstanding Options and Stock Appreciation Rights.

(d) <u>Substitute Awards</u>. The Administrator may grant Awards under the Plan in substitution for stock and stock based awards held by employees, directors or other key persons of another corporation in connection with the merger or consolidation of the employing corporation with the Company or a Subsidiary or the acquisition by the Company or a Subsidiary of property or stock of the employing corporation. The Administrator may direct that the substitute awards be granted on such terms and conditions as the Administrator considers appropriate in the circumstances. Any substitute Awards granted under the Plan shall not count against the share limitation set forth in Section 3(a).

#### SECTION 4. ELIGIBILITY

Grantees under the Plan will be such full or part-time officers and other employees, Non-Employee Directors and key persons (including consultants and prospective employees) of the Company and its Subsidiaries as are selected from time to time by the Administrator in its sole discretion.

#### SECTION 5. STOCK OPTIONS

Any Stock Option granted under the Plan shall be in such form as the Administrator may from time to time approve.

Stock Options granted under the Plan may be either Incentive Stock Options or Non-Qualified Stock Options. Incentive Stock Options may be granted only to employees of the Company or any Subsidiary that is a subsidiary corporation within the meaning of Section 424(f) of the Code. To the extent that any Option does not qualify as an Incentive Stock Option, it shall be deemed a Non-Qualified Stock Option.

(a) <u>Grant of Stock Options</u>. The Administrator in its discretion may grant Stock Options to eligible employees, Non-Employee Directors and key persons of the Company or any Subsidiary. Stock Options granted pursuant to this Section 5(a) shall be subject to the following terms and conditions and shall contain such additional terms and conditions, not inconsistent with the terms of the Plan, as the Administrator shall deem desirable. If the Administrator so determines, Stock Options may be granted in lieu of cash compensation at the optionees election, subject to such terms and conditions as the Administrator may establish.

(i) <u>Exercise Price</u>. The exercise price per share for the Stock covered by a Stock Option granted pursuant to this Section 5(a) shall be determined by the Administrator at the time of grant but shall not be less than 100 percent of the Fair Market Value on the date of grant (other than options granted in lieu of cash compensation). If an employee owns or is deemed to own (by reason of the attribution rules of Section 424(d) of the Code) more than 10 percent of the combined voting power of all classes of stock of the Company or any parent or subsidiary corporation and an Incentive Stock Option is granted to such employee, the option price of such Incentive Stock Option shall be not less than 110 percent of the Fair Market Value on the grant date.

(ii) <u>Option Term</u>. The term of each Stock Option shall be fixed by the Administrator, but no Stock Option shall be exercisable more than 10 years after the date the Stock Option is granted. If an employee owns or is deemed to own (by reason of the attribution rules of Section 424(d) of the Code) more than 10 percent of the combined voting power of all classes of stock of the Company or any parent or subsidiary corporation and an Incentive Stock Option is granted to such employee, the term of such Stock Option shall be no more than five years from the date of grant.

(iii) <u>Exercisability: Rights of a Stockholder.</u> Stock Options shall become exercisable at such time or times, whether or not in installments, as shall be determined by the Administrator at or after the grant date. The Administrator may at any time accelerate the exercisability of all or any portion of any Stock Option. An optionee shall have the rights of a stockholder only as to shares acquired upon the exercise of a Stock Option and not as to unexercised Stock Options.

(iv) <u>Method of Exercise</u>. Stock Options may be exercised in whole or in part, by giving written notice of exercise to the Company, specifying the number of shares to be purchased. Payment of the purchase price may be made by one or more of the following methods to the extent provided in the Option Award agreement:

(A) In cash, by certified or bank check or other instrument acceptable to the Administrator;

(B) Through the delivery (or attestation to the ownership) of shares of Stock that have been purchased by the optionee on the open market or that have been beneficially owned by the optionee for at least six months and are not then subject to restrictions under any Company plan. Such surrendered shares shall be valued at Fair Market Value on the exercise date; or

(C) By the optionee delivering to the Company a properly executed exercise notice together with irrevocable instructions to a broker to promptly deliver to the Company cash or a check payable and acceptable to the Company for the purchase price; provided that in the event the optionee chooses to pay the purchase price as so provided, the optionee and the broker shall comply with such procedures and enter into such agreements of indemnity and other agreements as the Administrator shall prescribe as a condition of such payment procedure.

Payment instruments will be received subject to collection. The delivery of certificates representing the shares of Stock to be purchased pursuant to the exercise of a Stock Option will be contingent upon receipt from the optionee (or a purchaser acting in his stead in accordance with the provisions of the Stock Option) by the Company of the full purchase price for such shares and the fulfillment of any other requirements contained in the Option Award agreement or applicable provisions of laws. In the event an optionee chooses to pay the purchase price by previously-owned shares of Stock through the attestation method, the number of shares of Stock transferred to the optionee upon the exercise of the Stock Option shall be net of the number of shares attested to.

(v) <u>Annual Limit on Incentive Stock Options</u>. To the extent required for incentive stock option treatment under Section 422 of the Code, the aggregate Fair Market Value (determined as of the time of grant) of the shares of Stock with respect to which Incentive Stock Options granted under this Plan and any other plan of the Company or its parent and subsidiary corporations become exercisable for the first time by an optionee during any calendar year shall not exceed \$100,000. To the extent that any Stock Option exceeds this limit, it shall constitute a Non-Qualified Stock Option.

(b) <u>Non-transferability of Options</u>. No Stock Option shall be transferable by the optionee otherwise than by will or by the laws of descent and distribution and all Stock Options shall be exercisable, during the optionee s lifetime, only by the optionee, or by the optionee s legal representative or guardian in the event of the optionee s incapacity. Notwithstanding the foregoing, the Administrator, in its sole discretion, may provide in the Award agreement regarding a given Option that the optionee may transfer his Non-Qualified Stock Options to members of his immediate family, to trusts for the benefit of such family members, or to partnerships in which such family members are the only partners, provided that the transferee agrees in writing with the Company to be bound by all of the terms and conditions of this Plan and the applicable Option.

#### SECTION 6. STOCK APPRECIATION RIGHTS

(a) <u>Nature of Stock Appreciation Rights</u>. A Stock Appreciation Right is an Award entitling the recipient to receive an amount in cash or shares of Stock or a combination thereof having a value equal to the excess of the

Fair Market Value of the Stock on the date of exercise over the exercise price of the Stock Appreciation Right, which price shall not be less than 100 percent of the Fair Market Value of the Stock on the date of grant (or more than the option exercise price per share, if the Stock Appreciation Right was granted in tandem with a Stock Option) multiplied by the number of shares of Stock with respect to which the Stock Appreciation Right shall have been exercised, with the Administrator having the right to determine the form of payment.

(b) <u>Grant and Exercise of Stock Appreciation Rights</u>. Stock Appreciation Rights may be granted by the Administrator in tandem with, or independently of, any Stock Option granted pursuant to Section 5 of the Plan. In the case of a Stock Appreciation Right granted in tandem with a Non-Qualified Stock Option, such Stock Appreciation Right may be granted either at or after the time of the grant of such Option. In the case of a Stock Appreciation Right granted in tandem with an Incentive Stock Option, such Stock Appreciation Right may be granted only at the time of the grant of the Option.

A Stock Appreciation Right or applicable portion thereof granted in tandem with a Stock Option shall terminate and no longer be exercisable upon the termination or exercise of the related Option.

(c) <u>Terms and Conditions of Stock Appreciation Rights</u>. Stock Appreciation Rights shall be subject to such terms and conditions as shall be determined from time to time by the Administrator, subject to the following:

(i) Stock Appreciation Rights granted in tandem with Options shall be exercisable at such time or times and to the extent that the related Stock Options shall be exercisable.

(ii) Upon exercise of a Stock Appreciation Right, the applicable portion of any related Option shall be surrendered.

(iii) All Stock Appreciation Rights shall be exercisable during the grantee s lifetime only by the grantee or the grantee s legal representative.

#### SECTION 7. RESTRICTED STOCK AWARDS

(a) <u>Nature of Restricted Stock Awards</u>. A Restricted Stock Award is an Award entitling the recipient to acquire, at such purchase price (which may be zero) as determined by the Administrator, shares of Stock subject to such restrictions and conditions as the Administrator may determine at the time of grant ( Restricted Stock ). Conditions may be based on continuing employment (or other service relationship) and/or achievement of pre-established performance goals and objectives. The grant of a Restricted Stock Award is contingent on the grantee executing the Restricted Stock Award agreement. The terms and conditions of each such agreement shall be determined by the Administrator, and such terms and conditions may differ among individual Awards and grantees.

(b) <u>Rights as a Stockholder</u>. Upon execution of a written instrument setting forth the Restricted Stock Award and payment of any applicable purchase price, a grantee shall have the rights of a stockholder with respect to the voting of the Restricted Stock, subject to such conditions contained in the written instrument evidencing the Restricted Stock Award. Unless the Administrator shall otherwise determine, (i) uncertificated Restricted Stock shall be accompanied by a notation on the records of the Company or the transfer agent to the effect that they are subject to forfeiture until such Restricted Stock are vested as provided in Section 7(d) below, and (ii) certificated Restricted Stock shall be required, as a condition of the grant, to deliver to the Company such instruments of transfer as the Administrator may prescribe.

(c) <u>Restrictions</u>. Restricted Stock may not be sold, assigned, transferred, pledged or otherwise encumbered or disposed of except as specifically provided herein or in the Restricted Stock Award agreement. Except as may

otherwise be provided by the Administrator either in the Award agreement or, subject to Section 14 below, in writing after the Award agreement is issued, if any, if a grantee s employment (or other service relationship) with the Company and its Subsidiaries terminates for any reason, any Restricted Stock that has not vested at the time of termination shall automatically and without any requirement of notice to such grantee from or other action by or on behalf of, the Company be deemed to have been reacquired by the Company at its original purchase price from such grantee or such grantee s legal representative simultaneously with such termination of employment (or other service relationship), and thereafter shall cease to represent any ownership of the Company by the grantee or rights of the grantee as a shareholder. Following such deemed reacquisition of unvested Restricted Stock that are represented by physical certificates, grantee shall surrender such certificates to the Company upon request without consideration.

(d) <u>Vesting of Restricted Stock</u>. The Administrator at the time of grant shall specify the date or dates and/or the attainment of pre-established performance goals, objectives and other conditions on which the non-transferability of the Restricted Stock and the Company s right of repurchase or forfeiture shall lapse. Notwithstanding the foregoing, in the event that any such Restricted Stock shall have a performance-based goal, the restriction period with respect to such shares shall not be less than one year, and in the event any such Restricted Stock shall have a time-based restriction, the restriction period with respect to such shares shall not be less than three years. Subsequent to such date or dates and/or the attainment of such pre-established performance goals, objectives and other conditions, the shares on which all restrictions have lapsed shall no longer be Restricted Stock and shall be deemed vested. Except as may otherwise be provided by the Administrator either in the Award agreement or, subject to Section 14 below, in writing after the Award agreement is issued, a grantee s rights in any shares of Restricted Stock that have not vested shall automatically terminate upon the grantee s termination of employment (or other service relationship) with the Company and its Subsidiaries and such shares shall be subject to the provisions of Section 7(c) above.

#### SECTION 8. DEFERRED STOCK AWARDS

(a) <u>Nature of Deferred Stock Awards</u>. A Deferred Stock Award is an Award of phantom stock units to a grantee, subject to restrictions and conditions as the Administrator may determine at the time of grant. Conditions may be based on continuing employment (or other service relationship) and/or achievement of pre-established performance goals and objectives. The grant of a Deferred Stock Award is contingent on the grantee executing the Deferred Stock Award agreement. The terms and conditions of each such agreement shall be determined by the Administrator, and such terms and conditions may differ among individual Awards and grantees. Notwithstanding the foregoing, in the event that any such Deferred Stock Award shall have a performance-based goal, the restriction period with respect to such award shall not be less than one year, and in the event any such Deferred Stock Award shall have a time-based restriction, the restriction period with respect to such award shall not be less than three years. At the end of the deferral period, the Deferred Stock Award, to the extent vested, shall be paid to the grantee in the form of shares of Stock.

(b) <u>Election to Receive Deferred Stock Awards in Lieu of Compensation</u>. The Administrator may, in its sole discretion, permit a grantee to elect to receive a portion of the cash compensation or Restricted Stock Award otherwise due to such grantee in the form of a Deferred Stock Award. Any such election shall be made in writing and shall be delivered to the Company no later than the date specified by the Administrator and in accordance with rules and procedures established by the Administrator. The Administrator shall have the sole right to determine whether and under what circumstances to permit such elections and to impose such limitations and other terms and conditions thereon as the Administrator deems appropriate.

(c) <u>Rights as a Stockholder</u>. During the deferral period, a grantee shall have no rights as a stockholder; provided, however, that the grantee may be credited with Dividend Equivalent Rights with respect to the phantom stock units underlying his Deferred Stock Award, subject to such terms and conditions as the Administrator may determine.

(d) <u>Restrictions</u>. A Deferred Stock Award may not be sold, assigned, transferred, pledged or otherwise encumbered or disposed of during the deferral period.

(e) <u>Termination</u>. Except as may otherwise be provided by the Administrator either in the Award agreement or, subject to Section 14 below, in writing after the Award agreement is issued, a grantee s right in all Deferred Stock Awards that have not vested shall automatically terminate upon the grantee s termination of employment (or cessation of service relationship) with the Company and its Subsidiaries for any reason.

#### SECTION 9. UNRESTRICTED STOCK AWARDS

<u>Grant or Sale of Unrestricted Stock</u>. The Administrator may, in its sole discretion, grant (or sell at par value or such higher purchase price determined by the Administrator) an Unrestricted Stock Award to any grantee pursuant to which such grantee may receive shares of Stock free of any restrictions (Unrestricted Stock) under the Plan. Unrestricted Stock Awards may be granted in respect of past services or other valid consideration, or in lieu of cash compensation due to such grantee.

#### SECTION 10. PERFORMANCE-BASED AWARDS TO COVERED EMPLOYEES

Notwithstanding anything to the contrary contained herein, if any Restricted Stock Award or Deferred Stock Award granted to a Covered Employee is intended to qualify as Performance-based Compensation under Section 162(m) of the Code and the regulations promulgated thereunder (a Performance-based Award ), such Award shall comply with the provisions set forth below:

(a) <u>Performance Criteria</u>. The performance criteria used in performance goals governing Performance-based Awards granted to Covered Employees may include any or all of the following: (i) the Company s return on equity, assets, capital or investment: (ii) pre-tax or after-tax profit levels of the Company or any Subsidiary, a division, an operating unit or a business segment of the Company, or any combination of the foregoing; (iii) cash flow, funds from operations or similar measure; (iv) total shareholder return; (v) changes in the market price of the Stock; (vi) sales or market share; or (vii) earnings per share.

(b) <u>Grant of Performance-based Awards</u>. With respect to each Performance-based Award granted to a Covered Employee, the Committee shall select, within the first 90 days of a Performance Cycle (or, if shorter, within the maximum period allowed under Section 162(m) of the Code) the performance criteria for such grant, and the achievement targets with respect to each performance criterion (including a threshold level of performance below which no amount will become payable with respect to such Award). Each Performance-based Award will specify the amount payable, or the formula for determining the amount payable, upon achievement of the various applicable performance targets. The performance criteria established by the Committee may be (but need not be) different for each Performance Cycle and different goals may be applicable to Performance-based Awards to different Covered Employees.

(c) <u>Payment of Performance-based Awards</u>. Following the completion of a Performance Cycle, the Committee shall meet to review and certify in writing whether, and to what extent, the performance criteria for the Performance Cycle have been achieved and, if so, to also calculate and certify in writing the amount of the Performance-based Awards earned for the Performance Cycle. The Committee shall then determine the actual size of each Covered Employee s Performance-based Award, and, in doing so, may reduce or eliminate the amount of the Performance-based Award, such reduction or elimination is appropriate.

(d) <u>Maximum Award Payable</u>. The maximum Performance-based Award payable to any one Covered Employee under the Plan for a Performance Cycle is 200,000 Shares (subject to adjustment as provided in Section 3(b) hereof).

G-9

#### SECTION 11. DIVIDEND EQUIVALENT RIGHTS

(a) <u>Dividend Equivalent Rights</u>. A Dividend Equivalent Right is an Award entitling the grantee to receive credits based on cash dividends that would have been paid on the shares of Stock specified in the Dividend Equivalent Right (or other award to which it relates) if such shares had been issued to and held by the grantee. A Dividend Equivalent Right may be granted hereunder to any grantee as a component of another Award or as a freestanding award. The terms and conditions of Dividend Equivalent Rights shall be specified in the Award agreement. Dividend equivalents credited to the holder of a Dividend Equivalent Right may be paid currently or may be deemed to be reinvested in additional shares of Stock, which may thereafter accrue additional equivalents. Any such reinvestment shall be at Fair Market Value on the date of reinvestment or such other price as may then apply under a dividend reinvestment plan sponsored by the Company, if any. Dividend Equivalent Rights may be settled in cash or shares of Stock or a combination thereof, in a single installment or installments. A Dividend Equivalent Right granted as a component of another Award may provide that such Dividend Equivalent Right shall be settled upon exercise, settlement, or payment of, or lapse of restrictions on, such other award, and that such Dividend Equivalent Right shall expire or be forfeited or annulled under the same conditions as such other award. A Dividend Equivalent Right granted as a component of another Award may also contain terms and conditions different from such other award.

(b) <u>Interest Equivalents</u>. Any Award under this Plan that is settled in whole or in part in cash on a deferred basis may provide in the grant for interest equivalents to be credited with respect to such cash payment. Interest equivalents may be compounded and shall be paid upon such terms and conditions as may be specified by the grant.

(c) <u>Termination</u>. Except as may otherwise be provided by the Administrator either in the Award agreement or, subject to Section 14 below, in writing after the Award agreement is issued, a grantee s rights in all Dividend Equivalent Rights or interest equivalents granted as a component of another Award that has not vested shall automatically terminate upon the grantee s termination of employment (or cessation of service relationship) with the Company and its Subsidiaries for any reason.

#### SECTION 12. TAX WITHHOLDING

(a) <u>Payment by Grantee</u>. Each grantee shall, no later than the date as of which the value of an Award or of any Stock or other amounts received thereunder first becomes includable in the gross income of the grantee for Federal income tax purposes, pay to the Company, or make arrangements satisfactory to the Administrator regarding payment of, any Federal, state, or local taxes of any kind required by law to be withheld with respect to such income. The Company and its Subsidiaries shall, to the extent permitted by law, have the right to deduct any such taxes from any payment of any kind otherwise due to the grantee. The Company s obligation to deliver stock certificates to any grantee is subject to and conditioned on tax obligations being satisfied by the grantee.

(b) <u>Payment in Stock</u>. Subject to approval by the Administrator, a grantee may elect to have the minimum required tax withholding obligation satisfied, in whole or in part, by (i) authorizing the Company to withhold from shares of Stock to be issued pursuant to any Award a number of shares with an aggregate Fair Market Value (as of the date the withholding is effected) that would satisfy the withholding amount due, or (ii) transferring to the Company shares of Stock owned by the grantee with an aggregate Fair Market Value (as of the date the withholding is effected) that would satisfy the withholding amount due.

#### SECTION 13. TRANSFER, LEAVE OF ABSENCE, ETC.

For purposes of the Plan, the following events shall not be deemed a termination of employment:

(a) a transfer to the employment of the Company from a Subsidiary or from the Company to a Subsidiary, or from one Subsidiary to another; or

(b) an approved leave of absence for military service or sickness, or for any other purpose approved by the Company, if the employee s right to re-employment is guaranteed either by a statute or by contract or under the policy pursuant to which the leave of absence was granted or if the Administrator otherwise so provides in writing.

#### SECTION 14. AMENDMENTS AND TERMINATION

The Board may, at any time, amend or discontinue the Plan and the Administrator may, at any time, amend or cancel any outstanding Award for the purpose of satisfying changes in law or for any other lawful purpose, but no such action shall adversely affect rights under any outstanding Award without the holder s consent. Except as provided in Section 3(b) or 3(c), in no event may the Administrator exercise its discretion to reduce the exercise price of outstanding Stock Options or effect repricing through cancellation and re-grants. Any material Plan amendments (other than amendments that curtail the scope of the Plan), including any Plan amendments that (i) increase the number of shares reserved for issuance under the Plan, (ii) expand the type of Awards available, materially expand the eligibility to participate or materially extend the term of the Plan, or (iii) materially change the method of determining Fair Market Value, shall be subject to approval by the Company stockholders entitled to vote at a meeting of stockholders. In addition, to the extent determined by the Administrator to be required by the Code to ensure that Incentive Stock Options granted under the Plan are qualified under Section 422 of the Code or to ensure that compensation earned under Awards qualifies as performance-based compensation under Section 162(m) of the Code, Plan amendments shall be subject to approval by the Company stockholders. Nothing in this Section 14 shall limit the Administrator s authority to take any action permitted pursuant to Section 3(c).

#### SECTION 15. STATUS OF PLAN

With respect to the portion of any Award that has not been exercised and any payments in cash, Stock or other consideration not received by a grantee, a grantee shall have no rights greater than those of a general creditor of the Company unless the Administrator shall otherwise expressly determine in connection with any Award or Awards. In its sole discretion, the Administrator may authorize the creation of trusts or other arrangements to meet the Company s obligations to deliver Stock or make payments with respect to Awards hereunder, provided that the existence of such trusts or other arrangements is consistent with the foregoing sentence.

#### SECTION 16. GENERAL PROVISIONS

(a) <u>No Distribution: Compliance with Legal Requirements</u>. The Administrator may require each person acquiring Stock pursuant to an Award to represent to and agree with the Company in writing that such person is acquiring the shares without a view to distribution thereof.

No shares of Stock shall be issued pursuant to an Award until all applicable securities law and other legal and stock exchange or similar requirements have been satisfied. The Administrator may require the placing of such stop-orders and restrictive legends on certificates for Stock and Awards as it deems appropriate.

(b) <u>Delivery of Stock Certificates</u>. Stock certificates to grantees under this Plan shall be deemed delivered for all purposes when the Company or a stock transfer agent of the Company shall have mailed such certificates in the United States mail, addressed to the grantee, at the grantee s last known address on file with the Company. Uncertificated Stock shall be deemed delivered for all purposes when the Company or a Stock transfer agent of the Company shall have given to the grantee by United States mail, addressed to the grantee s last known address on file with the Company, notice of issuance and recorded the issuance in its records (which may include electronic book entry records).

(c) <u>Other Compensation Arrangements</u>; <u>No Employment Rights</u>. Nothing contained in this Plan shall prevent the Board from adopting other or additional compensation arrangements, including trusts, and such arrangements may be either generally applicable or applicable only in specific cases. The adoption of this Plan and the grant of Awards do not confer upon any employee any right to continued employment with the Company or any Subsidiary.

(d) <u>Trading Policy Restrictions</u>. Option exercises and other Awards under the Plan shall be subject to such Company s insider trading policy and procedures, as in effect from time to time.

(e) <u>Designation of Beneficiary</u>. Each grantee to whom an Award has been made under the Plan may designate a beneficiary or beneficiaries to exercise any Award or receive any payment under any Award payable on or after the grantee s death. Any such designation shall be on a form provided for that purpose by the Administrator and shall not be effective until received by the Administrator. If no beneficiary has been designated by a deceased grantee, or if the designated beneficiaries have predeceased the grantee, the beneficiary shall be the grantee s estate.

#### SECTION 17. EFFECTIVE DATE OF PLAN

This Plan shall become effective upon approval by the holders of a majority of the votes cast at a meeting of stockholders at which a quorum is present. Subject to such approval by the stockholders and to the requirement that no Stock may be issued hereunder prior to such approval, Stock Options and other Awards may be granted hereunder on and after adoption of this Plan by the Board. No grants of Stock Options and other Awards may be made hereunder after the tenth (10<sup>th</sup>) anniversary of the Effective Date and no grants of Incentive Stock Options may be made hereunder after the tenth (10<sup>th</sup>) anniversary of the date the Plan is approved by the Board.

#### SECTION 18. GOVERNING LAW

This Plan and all Awards and actions taken thereunder shall be governed by, and construed in accordance with, the laws of the Commonwealth of Massachusetts, applied without regard to conflict of law principles.

Tutor-Saliba Corporation

# and Subsidiaries

Consolidated Financial Statements as of

December 31, 2007 and 2006, and for the

Years Ended December 31, 2007, 2006 and 2005, and

Independent Auditors Report

H-1

#### INDEPENDENT AUDITORS REPORT

To the Board of Directors and Shareholders of Tutor-Saliba Corporation

Sylmar, CA

We have audited the accompanying consolidated balance sheets of Tutor-Saliba Corporation and subsidiaries (the Company ) as of December 31, 2007 and 2006, and the related consolidated statements of income, shareholders equity, and cash flows for each of the three years in the period ended December 31, 2007. These financial statements are the responsibility of the Company s management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company s internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, such consolidated financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2007 and 2006, and the results of its operations and its cash flows for each of the three years in the period ended December 31, 2007, in conformity with accounting principles generally accepted in the United States of America.

As discussed in Note 1 to the consolidated financial statements, in 2007, the Company changed its method of reporting its interests in construction joint ventures in the consolidated balance sheets and consolidated statements of cash flows from the equity method of accounting to the proportionate consolidation method and, retrospectively, adjusted the 2006 and 2005 consolidated financial statements for the change.

/s/Deloitte and Touche LLP

Los Angeles, California

April 25, 2008 (May 28, 2008, as to the effects of discontinued operations discussed in Note 14)

H-2

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

# CONSOLIDATED BALANCE SHEETS

	December 31, 2007 2006	
ASSETS	2007	2000
Current assets:		
Cash and cash equivalents	\$ 192,802,901	\$ 37,575,733
Restricted cash	24,000,000	
Marketable securities		93,424,299
Receivables	223,216,615	175,091,118
Unbilled work	29,018,934	19,280,817
Advances to related parties		4,595,513
Advances to affiliates	108,079	342,570
Inventory	2,478,893	1,374,194
Deferred income taxes	192,896	158,245
Other current assets	6,896,448	6,971,790
Total current assets	478,714,766	338,814,279
Property and equipment, net of depreciation	100,491,769	85,972,239
Noncurrent assets:		
Minority interest receivable	13,172,309	13,389,225
Goodwill	2,694,467	
Identified intangible asset, net	2,929,157	
Other assets	3,490,326	3,581,319
Total noncurrent assets	22,286,259	16,970,544
Total assets	\$ 601,492,794	\$ 441,757,062
LIABILITIES AND SHAREHOLDERS EQUITY		
Current liabilities:		
Current maturities of long-term debt	\$ 12,074,077	\$ 17,261,589
Accounts and subcontracts payable	236,970,144	122,937,180
Accrued liabilities	22,840,295	23,278,006
Advances from affiliates	523,948	576,200
Deferred contract revenue	56,229,238	47,476,186
Income taxes payable	2,536,311	399,805
Total current liabilities	331,174,013	211,928,966
Noncurrent liabilities:		
Long-term debt, less current maturities	53,616,849	60,220,891
Deferred income taxes	1,311,424	2,176,016
Other long-term liabilities	3,760,165	5,718,689
Total noncurrent liabilities	58,688,438	68,115,596
Total liabilities	389,862,451	280,044,562

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

Other commitments and Contingencies (Note 12)		
Shareholders equity:		
Capital stock, no par value: authorized 1,000,000 shares:		
issued and outstanding 900,043 shares	6,756,510	6,756,510
Additional paid-in capital	11,075,563	11,075,563
Accumulated other comprehensive income	144,570	38,901,604
Retained earnings	193,653,700	104,978,823
Total shareholders equity	211,630,343	161,712,500
Total liabilities and shareholders equity	\$ 601,492,794	\$441,757,062

See notes to consolidated financial statements.

H-3

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### CONSOLIDATED STATEMENTS OF INCOME

	Ye 2007	ear Ended December 31, 2006	2005
REVENUE:			
Construction revenue	\$ 1,150,872,378	\$ 513,445,293	\$ 386,755,250
Sales revenue	951,470	1,011,730	834,753
Total revenue	1,151,823,848	514,457,023	387,590,003
COST OF REVENUES:			
Cost of construction revenue	1,061,735,990	484,735,017	370,825,774
Amortization of intangible asset	1,230,843		
Cost of sales revenue	636,155	699,114	508,692
Total cost of revenues	1,063,602,988	485,434,131	371,334,466
Gross profit	88,220,860	29,022,892	16,255,537
Gain on sale of property and equipment Cost of contract litigation settlement	(446,325)	(453,750)	(7,523,793) 14,651,537
General and administrative expenses	36,237,164	26,823,295	26,352,043
Income (loss) from operations	52,430,021	2,653,347	(17,224,250)
OTHER INCOME (EXPENSES):			
Interest and other income	6,076,770	3,474,082	2,772,034
Net gain on sale of marketable securities	94,104,632	8,882,610	21,370,376
Gain on derivative			16,936,063
Interest expense	(4,196,814)	(5,256,848)	(3,414,423)
Other expense	(975,656)	(877,607)	(574,092)
Total other income, net	95,008,932	6,222,237	37,089,958
Income before minority interest and income taxes	147,438,953	8,875,584	19,865,708
Provision for income taxes	4,399,231	1,662,632	848,849
Income before minority interest	143,039,722	7,212,952	19,016,859
Minority interest	(110,705)	(40,268)	(16,103)
INCOME FROM CONTINUING OPERATIONS	142,929,017	7,172,684	19,000,756
DISCONTINUED OPERATIONS:			
Income (loss) from discontinued operations before minority interest and income taxes	337,053	(177,668)	392,660
Provision for income taxes	4,798	3,875	4,869
Income (loss) before minority interest	332,255	(181,543)	387,791
Minority interest	(106,211)	145,773	1,019,238

INCOME (LOSS) FROM DISCONTINUED OPERATIONS	226,044	(35,770)	1,407,029
NET INCOME	\$ 143,155,061	\$ 7,136,914	\$ 20,407,785

See notes to consolidated financial statements.

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### CONSOLIDATED STATEMENTS OF SHAREHOLDERS EQUITY

	Capital Stock	Additional Paid-In Capital	Accumulated Other Comprehensive Income	Retained Earnings	Total
BALANCE AT JANUARY 1, 2005	\$ 6,756,510	\$ 11,075,563	\$ 25,848,752	\$ 116,267,696	\$ 159,948,521
Comprehensive income:					
Net income				20,407,785	20,407,785
Other comprehensive income: Realized holding gains on marketable securities (net of tax of \$(320,555))			(21,049,820)		(21,049,820)
Unrealized holding gains arising during the period on marketable securities (net of tax of \$309,874)			20,348,419		20,348,419
Foreign currency translation adjustments (net of tax of \$27,185)			52,771		52,771
Total comprehensive income			(648,630)	20,407,785	19,759,155
Distributions to shareholder				(14,768,598)	(14,768,598)
BALANCE AT DECEMBER 31, 2005	6,756,510	11,075,563	25,200,122	121,906,883	164,939,078
Comprehensive income:				7 126 014	7 126 014
Net income				7,136,914	7,136,914
Other comprehensive income: Realized holding gains on marketable securities (net of tax of \$(133,239))			(8,749,371)		(8,749,371)
Unrealized holding gains arising during the period on marketable securities (net of tax of \$344,599)			22,628,669		22,628,669
Foreign currency translation adjustments (net of tax of \$(91,602))			(177,816)		(177,816)
Total comprehensive income			13,701,482	7,136,914	20,838,396
Distributions to shareholder				(24,064,974)	(24,064,974)
BALANCE AT DECEMBER 31, 2006	6,756,510	11,075,563	38,901,604	104,978,823	161,712,500
Comprehensive income:					
Net income				143,155,061	143,155,061
Other comprehensive income:					
Realized holding gains on marketable securities (net of tax of \$(1,411,569)) Unrealized holding gains arising during the period on			(92,693,063)		(92,693,063)
marketable securities (net of tax of \$814,393)			53,478,482		53,478,482
Foreign currency translation adjustments (net of tax of \$235,706)			457,547		457,547
Total comprehensive income			(38,757,034)	143,155,061	104,398,027
Distributions to shareholders				(54,480,184)	(54,480,184)

BALANCE AT DECEMBER 31, 2007

\$ 6,756,510 \$ 11,075,563 \$ 144,570 \$ 193,653,700 \$ 211,630,343

See notes to consolidated financial statements.

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### CONSOLIDATED STATEMENTS OF CASH FLOWS

		Year Ended December 31,		
CASH ELOWS EDOM OPEDATING ACTIVITIES.	2007	2006	2005	
CASH FLOWS FROM OPERATING ACTIVITIES: Net Income	\$ 142 155 061	\$ 7,136,914	\$ 20,407,785	
Adjustments to reconcile net income to net cash provided by (used in) operating activities:	\$ 143,155,061	\$ 7,136,914	\$ 20,407,785	
Depreciation and amortization	6,558,732	5,389,887	4,426,815	
Amortization of intangible asset	1,230,843	5,569,667	4,420,813	
Net gain from sale of property and equipment	(446,325)	(406,312)	(8,330,437)	
Net gain on sale of marketable securities	(94,104,632)	(8,882,610)	(21,370,376)	
Gain on derivative	(94,104,032)	(8,882,010)	(16,936,063)	
Deferred income taxes	(302,067)	491,828	(135,911)	
Minority interest in consolidated subsidiaries	216,916	(105,505)	(1,003,135)	
Increase (decrease) in cash attributed to changes in operating assets and liabilities:	210,910	(105,505)	(1,005,155)	
Receivables	(58,982,923)	(59,122,930)	21,146,222	
Inventory	(1,104,699)	(39,122,930) (381,115)	158,479	
Other assets	(1,104,099) (140,464)	350,900	(4,807,720)	
Advances from (to) joint ventures partners, net	(140,404) 182,240	(234,845)	957,325	
	2,097,251		951,525	
Advances from (to) related parties		(677,465)	1 902 002	
Unbilled work	1,577,606	2,190,316	1,803,003	
Accounts payable and other liabilities	111,195,217	9,385,058	4,773,308	
Deferred contract revenue	4,104,487	16,042,428	5,165,181	
Net cash provided by (used in) operating activities	115,237,243	(28,823,451)	6,254,476	
CASH FLOWS FROM INVESTING ACTIVITIES:				
Proceeds from sale of property and equipment	4,585,364	2,808,988	13,186,680	
Proceeds from sales of marketable securities	147,717,224	21,248,000	29,274,460	
Purchases of property and equipment	(23,815,351)	(22,138,835)	(6,748,292)	
Increase in restricted cash	(24,000,000)	(22,130,035)	(0,710,272)	
Advances to related parties	(173,728)	(1,309,664)	(511,565)	
Cash acquired from acquisition	83,494	(1,00),001)	(011,000)	
Net cash provided by investing activities	104,397,003	608,489	35,201,283	
CASH FLOWS FROM FINANCING ACTIVITIES:				
Proceeds from long-term debt	18,514,200	29,093,397	4,620,477	
Principal payments of long-term debt	(21,113,084)	(9,108,310)	(17,831,050)	
(Paydown) draw on line of credit	(10,000,000)	10,000,000		
Distributions to shareholders	(51,808,194)	(20,003,774)	(14,768,598)	
Net cash (used in) provided by financing activities	(64,407,078)	9,981,313	(27,979,171)	
Net increase (decrease) in cash and cash equivalents	155,227,168	(18,233,649)	13,476,588	
Cash and cash equivalents at beginning of year	37,575,733	55,809,382	42,332,794	
	51,515,155	33,007,302	12,332,791	
Cash and cash equivalents at end of year	\$ 192,802,901	\$ 37,575,733	\$ 55,809,382	
SUPPLEMENTAL CASH FLOW INFORMATION:				
Cash paid during the period for:				
Interest	\$ 6,274,740	\$ 10,772,006	\$ 5,298,228	
Income taxes	\$ 2,633,810	\$ 1,628,682	\$ 648,021	

NONCASH INVESTING AND FINANCING ACTIVITIES:			
Marketable securities distributed to shareholder	\$	\$ 4,061,200	\$
Advances to related parties distributed to shareholder	\$ 2,671,990	\$	\$
Capitalized lease obligation incurred for transportation equipment	\$ 521,234	\$ 1,260,566	\$ 375,677
Refinancing of long-term debt on equipment	\$	\$ 10,923,122	\$
Unrealized holding gains on marketable securities	\$ 53,478,482	\$ 22,628,669	\$ 20,348,419
Accrued liabilities incurred from purchase of marketable securities	\$	\$	\$ 19,682,351
Receivable arising from sale of marketable securities	\$ 12,678,428	\$	\$ 4,391,165

See notes to consolidated financial statements.

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES

#### Nature of Business:

Tutor-Saliba Corporation was incorporated in 1981 as a successor to businesses that had been providing construction services since 1949. Tutor-Saliba Corporation and its majority-owned subsidiaries (the Company ) provide diversified general contracting, design-build, and preconstruction services to public agencies and private clients. The Company s construction business is conducted through three basic segments or operations: domestic civil, domestic building, and international. The Company focuses on large, complex construction projects in the civil infrastructure and commercial building sectors. The Company manages all aspects of these construction projects and directly performs many significant construction specialties, including concrete forming and placement, site excavation, structural steel erection and electrical and mechanical services. The Company provides these services by using traditional general contracting arrangements, such as fixed price, guaranteed maximum price, and cost plus fee contracts.

In an effort to limit its financial and/or operational risk on certain of its larger and more complex projects, the Company participates in construction joint ventures, often as the sponsor or manager of the project, for the purpose of bidding and, if awarded, providing the agreed-upon construction services. Each joint venture participant usually agrees in advance to provide a predetermined percentage of capital, as required, and to share in the same percentage of profit or loss on the project.

#### Principles of Consolidation:

The consolidated financial statements include the accounts of Tutor-Saliba Corporation and its majority-owned and controlled subsidiaries. All intercompany balances and transactions have been eliminated in consolidation.

Prior to 2007, The Company s noncontrolling interests in construction joint ventures were accounted for on the equity method in the consolidated balance sheets and on the proportionate consolidation method in the consolidated statements of income. Beginning in 2007, construction joint venture interests are accounted for using the proportionate consolidation method in the consolidated balance sheets, as well as the consolidated statements of income, whereby the Company s proportionate share of each joint venture s assets, liabilities, revenues, and cost of operations are included in the appropriate classifications in the consolidated financial statements. The Company believes the change, which results in presenting all joint venture activity using a consistent methodology in both the consolidated balance sheets and consolidated statements of income, is preferable.

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

The change had no impact on the consolidated statements of income for any period presented. Although the change impacted various classifications within the consolidated balance sheets and consolidated statements of cash flows, there was no impact to shareholders equity. Consolidated balance sheets and consolidated statements of cash flows have been retrospectively adjusted to conform to the 2007 presentation. The impact on the consolidated financial statements is as follows (in thousands):

#### **Consolidated Balance Sheets**

			Decem	ber 31,		
		2007			2006	
	Equity Method	Proportionate Consolidation Method	Effect of Change	As Previously Reported	As Reported	Effect of Change
Cash and cash equivalents	173,739	192,803	19,064	19,956	37,576	17,620
Receivables	206,394	223,217	16,823	143,456	175,091	31,635
Unbilled work	28,492	29,019	527	18,914	19,281	367
Equity in and advances to construction joint ventures	4,289		(4,289)	2,478		(2,478)
Other current assets	6,752	6,896	144	6,940	6,972	32
Accounts and subcontracts payable	226,440	236,970	10,530	104,023	122,937	18,914
Accrued liabilities	18,482	22,840	4,358	18,673	23,278	4,605
Advances to affiliates		416	416		234	234
Deferred contract revenue	41,174	56,229	15,055	27,196	47,476	20,280
Other long-term liabilities	1,851	3,760	1,909	2,575	5,719	3,144
Consolidated Statements of Cash Flows						

	Equity Method	December 31, 2007 Proportionate Consolidation Method	Effect of Change
Net (gain) loss from joint ventures	(21,191)		21,191
Distributions from joint ventures	21,693		(21,693)
Receivables	(73,795)	(58,983)	14,812
Other assets	(23)	(3)	20
Advances to (from) joint ventures, net	2,011	182	(1,829)
Unbilled work	1,738	1,578	(160)
Accounts payable and other liabilities	121,062	111,195	(9,867)
Deferred contract revenue	9,329	4,104	(5,225)
Contributions to joint ventures	(4,328)		4,328

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

#### **Consolidated Statements of Cash Flows**

		••••	Decem	ber 31,		
	As Previously Reported	2006 As Reported	Effect of Change	As Previously Reported	2005 As Reported	Effect of Change
Net gain from sale of property and equipment	(454)	(406)	48	(7,524)	(7,507)	17
Net (gain) loss from joint ventures	(6,883)		6,883	3,794		(3,794)
Distributions from joint ventures	21,132		(21,132)	22,725		(22,725)
Receivables	(51,777)	(59,123)	(7,346)	(8,017)	21,146	29,163
Other assets	270	351	81	(4,835)	(4,808)	27
Advances (to) from joint ventures, net	(267)	(235)	32	(6,026)	957	6,983
Unbilled work	2,138	2,190	52	(16,030)	1,803	17,833
Accounts payable and other liabilities	1,540	9,385	7,845	7,234	4,773	(2,461)
Deferred contract revenue	6,463	16,042	9,579	4,379	5,165	786
Proceeds from sale of property and equipment	1,256	2,809	1,553			
Distributions from joint ventures				22,948		(22,948)
Contributions to joint ventures	(6,368)		6,368	(3,200)		3,200
Principal payments of long-term debt	(9,094)	(9,108)	(14)	(16,541)	(17,831)	(1,290)
of Estimates in the Preparation of Einancial Statements						

#### Use of Estimates in the Preparation of Financial Statements:

The consolidated financial statements have been prepared in conformity with accounting principles generally accepted in the United States of America, which require management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosures of contingent assets and liabilities at the date of the consolidated financial statements, as well as the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. On an ongoing basis, the Company reviews its estimates based on information that is currently available. Changes in facts and circumstances may result in revised estimates.

#### Method of Accounting for Contracts:

Revenues and profits from the Company s construction contracts are recognized by applying percentages of completion for the period to the total estimated profits for the respective contracts. Percentage of completion

is determined by relating the actual cost of the work performed to date to the current estimated total cost of the respective contracts. When the estimate on a contract indicates a loss, the Company s policy is to record the entire loss during the accounting period in which it is estimated. In the ordinary course of business, at a minimum on a quarterly basis, the Company prepares updated estimates of the total forecasted revenue, cost and profit or loss for each contract. The cumulative effect of revisions in estimates of the total forecasted revenue and costs, including unapproved change orders and claims, during the course of the work is reflected in the accounting period in which the facts that caused the revision become known. The financial impact of these revisions to any one contract is a function of both the amount of the revision and the percentage of completion of the contract. An amount equal to the costs incurred which are attributable to

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

unapproved change orders and claims are included in the total estimated revenue when realization is determined to be probable. Profits from unapproved change orders and claims are recorded in the total estimated revenue in the period such amounts are resolved.

The Company includes in current assets and current liabilities amounts related to construction contracts realizable and payable over a period in excess of one year. Deferred contract revenue represents excess of billings to date over the amount of contract costs and profits (or contract revenue) recognized to date on the percentage-of-completion accounting method on certain contracts. Unbilled work represents the excess of contract costs and profits (or contract revenue) recognized to date on the percentage-of-completion accounting method over billings to date on the remaining contracts. Unbilled work results when (1) the appropriate contract revenue amount has been recognized in accordance with the percentage-of-completion accounting method, but a portion of the revenue recorded cannot be billed currently due to the billing terms defined in the contract and/or (2) costs, recorded at estimated realized value, related to unapproved change orders or claims are incurred. Unbilled work on contracts consisted of the following:

	Decem	December 31,		
	2007	2006		
Unbilled costs and profits incurred to date	\$ 3,544,855	\$ 5,751,342		
Unapproved change orders	14,122,686	5,831,605		
Claims	11,351,393	7,697,870		
	\$ 29,018,934	\$ 19,280,817		

The prerequisite for billing Unbilled costs and profits incurred to date is provided in the defined billing terms of each of the applicable contracts. The prerequisite for billing Unapproved change orders or Claims is the final resolution and agreement between the parties. The amount of unbilled work at December 31, 2007 estimated by management to be collected beyond one year is approximately \$5,600,000.

#### Significant Estimates:

As outlined above, the Company s revenue is recognized on the percentage-of-completion basis. Consequently, construction revenue and gross margin for each reporting period are determined on a contract-by-contract basis by reference to estimates by the Company s management of expected costs to be incurred to complete each project. These estimates include provisions for known and anticipated cost overruns and cost recoveries, if any exist or are expected to occur. These estimates are subject to revision in the normal course of business and could be material.

#### Foreign Currency Translation:

The functional currency for foreign operations is the local currency. The assets and liabilities of the Company s international subsidiaries are translated into U.S. dollars using current exchange rates at the balance sheet date. Operating statement items are translated at average exchange rates prevailing during the period. The resulting cumulative translation adjustments of \$144,570 and \$(312,977) at December 31, 2007 and 2006, respectively, are recorded in the foreign currency translation adjustment account as part of accumulated other comprehensive income in shareholders equity. Foreign currency transaction gains and losses, if any, are included in operations as they occur.

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

#### Cash and Cash Equivalents:

Cash equivalents represent all highly liquid debt instruments with maturities, when purchased, of three months or less.

Cash and cash equivalents as reported in the accompanying consolidated balance sheets consist of amounts held by the Company that are available for general corporate purposes and the Company s proportionate share of amounts held by construction joint ventures that are available only for joint venture-related uses. Cash held by construction joint ventures is distributed from time to time to the Company and to the other joint venture participants in accordance with their percentage interest. Cash distributions received by the Company from its construction joint ventures are then available for general corporate purposes. Cash and cash equivalents consisted of the following:

	December 31,	
	2007	2006
Corporate cash and cash equivalents		
(available for general purposes)	\$ 173,738,595	\$ 19,955,991
Company s share of joint venture		
cash and cash equivalents (available only for joint venture purposes, including future		
distributions)	19,064,306	17,619,742
	\$ 192,802,901	\$ 37,575,733

#### Restricted Cash

As of December 31, 2007 and 2006, the Company had \$24,000,000 and \$0, respectively, of restricted cash. Restricted cash consists of amounts held as collateral on the revolving credit facility (see Note 5).

#### Marketable Securities:

The Company accounts for marketable securities, all of which are publicly traded equity securities of Perini Corporation (see Note 8), as available for sale, in accordance with Financial Accounting Standards Board (FASB) Statement No. 115, *Accounting for Certain Investments in Debt and Equity Securities*. As of December 31, 2007, the Company no longer held any marketable securities. Unrealized holding gains and losses for available-for-sale securities were reported as a part of accumulated other comprehensive income in shareholders equity until realized. The unrealized holding gains were \$0 and \$39,214,581 at December 31, 2007 and 2006, respectively.

#### Accounting for a Derivative:

On March 29, 2000, the Company entered into a shareholders agreement (the Agreement ) with certain shareholders of Perini Corporation, whereby the Company received the option to purchase 2,352,941 shares of Perini Corporation common stock commencing on the third anniversary of the closing of the Agreement and ending on the sixth anniversary of the closing. The call price on the option was set at \$4.25 per share and increased by 14% each year the option was outstanding. The option became exercisable on March 29, 2003. On December 30, 2005, the Company exercised its call option for an exercise price of \$8.365 per

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

share. The option was accounted for as a derivative and recorded at fair value. Unrealized gains and losses were recognized through earnings under the provisions of FASB Statement No. 133, *Accounting for Derivative Instruments and Hedging Activities*, as amended by FASB Statement No. 149, *Amendment of Statement 133 on Derivative Instruments and Hedging Activities*. For the year ended December 31, 2005, the Company recorded a gain of \$16,936,063 in other income for the change in fair value of the option.

#### Inventory:

Inventory consists of construction materials, equipment parts, and supplies that have not been assigned and charged to specific contracts. Inventory is stated at the lower of cost or market. Cost is determined generally on the specific identification method.

#### Property, Equipment, and Long-Lived Assets:

Property and equipment are stated at cost. Depreciation is provided using the straight-line method over its estimated useful lives ranging from three to forty years after an allowance for salvage.

All long-lived assets are reviewed by management in accordance with FASB Statement No. 144, *Accounting for the Impairment or Disposal of Long-Lived Assets*, for impairment whenever events or changes in circumstance indicate that the carrying value of such assets may not be recoverable from expected future undiscounted cash flows. Major renewals and betterments are capitalized and maintenance and repairs are charged to operations as incurred. When property and equipment are retired or otherwise disposed of, the cost and accumulated depreciation are removed from the appropriate account and any gain or loss is included in results of operations. No impairments were identified as of December 31, 2007 and 2006.

#### Goodwill and other Intangible Assets:

Goodwill is subject to annual impairment test pursuant to FASB Statement No. 142, *Goodwill and Other Intangible Assets*. The Company regularly evaluates whether events or circumstances have occurred which may indicate a possible impairment of goodwill. The Company believes the methodology used in testing impairment of goodwill, which includes significant judgements and estimates, provides a reasonable basis in determining whether an impairment charge should be taken.

#### Income Taxes:

The Company accounts for income taxes in accordance with FASB Statement No. 109, *Accounting for Income Taxes*, which requires the use of an asset and liability method of accounting for income taxes. Deferred income taxes are provided to reflect the tax effect of differences between the financial statement carrying amounts and the tax bases of assets and liabilities. The Company elected to be treated as an S corporation effective January 1, 1996. As a result, taxable income, loss and credits flow directly to the shareholder and tax-related assets and liabilities of the Company become the obligation of the shareholder of the S corporation and are no longer reflected in the consolidated financial statements. The deferred tax assets, liabilities, and provision reflected in the consolidated financial statements are those that do not flow through to the shareholder, as they relate to taxable subsidiaries.

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

In July 2006, the FASB issued FASB Interpretation (FIN) No. 48, *Accounting for Uncertainty in Income Taxes an interpretation of FASB Statement No. 109.* FIN No. 48 increases the relevancy and comparability of financial reporting by clarifying the way companies account for uncertainty in measuring income taxes. FIN No. 48 prescribes a comprehensive model for how a company should recognize, measure, present, and disclose in its financial statements uncertain tax positions that the company has taken or expects to take on a tax return. FIN No. 48 only allows a favorable tax position to be included in the calculation of tax liabilities and expenses if a company concludes that it is more likely than not that its adopted tax position will prevail if challenged by tax authorities. FIN No. 48 also provides guidance on the accounting for and recording of interest and penalties on uncertain tax positions. The Company adopted FIN No. 48 on January 1, 2007, and the adoption of FIN No. 48 did not have an impact on the financial condition and results of operations.

#### Fair Value of Financial Instruments:

Marketable securities approximate their fair value as determined by market quotes. The carrying value of receivables and other amounts arising out of normal contract activities, including retentions that may be settled beyond one year are estimated to approximate fair value. All significant debt obligations, interest rates, and their carrying value are considered to approximate market rates and fair value.

#### New Accounting Pronouncements:

In September 2006, the FASB issued FASB Statement No. 157, *Fair Value Measurements*. FASB Statement No. 157 defines fair value, establishes a framework for measuring fair value, and expands disclosures about fair value measurements. In February 2008, the FASB issued FASB Statf Position (FSP) No. FAS 157-1 and FSP No. FAS 157-2, affecting implementation of FASB Statement No. 157. FSP No. FAS 157-1 excludes FASB Statement No. 13, *Accounting for Leases*, and other accounting pronouncements that address fair value measurements under FASB Statement No. 13, from the scope of FASB Statement No. 157. FSP No. FAS 157-2 delays the effective date of FASB Statement No. 157 for nonfinancial liabilities, except for items that are recognized or disclosed at fair value on a recurring basis, to fiscal years beginning after November 15, 2008. The Company does not believe the adoption of FASB Statement No. 157, as amended by FSP No. FAS 157-1 and FSP No. FAS 157-2 will have a material impact on the financial condition and results of operations.

In February 2007, the FASB issued FASB Statement No. 159, *The Fair Value Option for Financial Assets and Financial Liabilities*. FASB Statement No. 159 permits entities to choose to measure many financial instruments and certain other items at fair value. The objective of the guidance is to improve financial reporting by providing entities with the opportunity to mitigate volatility in reported earnings caused by measuring related assets and liabilities differently without having to apply complex hedge accounting provisions. FASB Statement No. 159 is effective as of the beginning of the first fiscal year after November 15, 2007. The Company does not believe the adoption of FASB Statement No. 159 will have a material impact on the Company s financial condition and results of operations.

In December 2007, the FASB issued FASB Statement No. 141 (revised 2007), *Business Combinations*. FASB Statement No. 141(R) establishes principles and requirements for how an acquirer recognizes and measures in its financial statements the identifiable assets acquired, the liabilities assumed, any noncontrolling interest in the acquiree, and the goodwill acquired. FASB Statement No. 141(R) also

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

establishes disclosure requirements to enable the evaluation of the nature and financial effects of the business combination. FASB Statement No. 141(R) is effective for fiscal years beginning after December 15, 2008. The Company will apply the provision of FASB Statement No. 141(R) prospectively as of that date.

In December 2007, the FASB issued FASB Statement No. 160, *Noncontrolling Interests in Consolidated Financial Statements an amendment of Accounting Research Bulletin No. 51*. FASB Statement No. 160 establishes accounting and reporting standards for ownership interests in subsidiaries held by parties other than the parent, the amount of consolidated net income attributable to the parent and to the noncontrolling interest, changes in a parent s ownership interest, and the valuation of retained noncontrolling equity investments when a subsidiary is deconsolidated. FASB Statement No. 160 also establishes disclosure requirements that clearly identify and distinguish between the interests of the parent and the interests of the noncontrolling owners. FASB Statement No. 160 is effective for fiscal years beginning after December 15, 2008. The Company is currently evaluating the potential impact of the adoption of FASB Statement No. 160 on the Company s financial condition and results of operations.

### 2. RECEIVABLES

Receivables consisted of the following:

	Decem	ber 31,
	2007	2006
Receivables:		
Contract receivables	\$ 101,435,119	\$ 76,403,192
Retainage receivables	105,522,310	83,099,401
Trade and other	16,259,186	15,588,525
Total receivables	\$ 223,216,615	\$ 175,091,118

#### Retainage:

Many of the contracts under which the Company performs work contain retainage provisions. Retainage refers to that portion of revenue earned by the Company but held for payment by the customer pending satisfactory completion of the project. The Company believes that all amounts retained by customers under such provisions are fully collectible. Retainage on active contracts is classified as a current asset regardless of the term of the contract. Retainage is generally collected within one year of the completion of the contract. As of December 31, 2007 and 2006, the amount of retainage expected to be collected beyond one year was \$66,775,000 and \$38,650,000, respectively.

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

### 3. PROPERTY AND EQUIPMENT

Property and equipment, at cost, consisted of the following:

	Decem	ber 31,
	2007	2006
Land	\$ 6,591,263	\$ 6,591,263
Building and improvements	61,341,726	60,583,871
Transportation equipment	10,938,880	11,034,344
Construction equipment	80,305,067	65,338,114
Office equipment	2,897,427	2,857,491
Property and equipment, gross	162,074,363	146,405,083
Less accumulated depreciation	61,582,594	60,432,844
Property and equipment, net of depreciation	\$ 100,491,769	\$ 85,972,239

Included in property and equipment as of December 31, 2007 and 2006 are amounts recorded under capital leases for transportation equipment of \$3,350,866 and \$3,100,782 respectively. Accumulated depreciation on transportation equipment recorded under capital leases was \$1,553,557 and \$1,575,298 as of December 31, 2007 and 2006, respectively.

### 4. OPERATING LEASES

During 2005, the Company entered into a new non-cancelable operating lease for a corporate aircraft, after the buyout of an existing lease. The Company has the right to voluntarily terminate the lease and purchase the asset on any installment payment date after the first 12-month anniversary of the lease for the outstanding lease balance. Upon expiration of the initial lease term in December 2015, the Company has the option of either purchasing the asset for the outstanding lease balance or returning the asset to the lessor. The Company s shareholder has personally guaranteed this lease. Total rental expense under this lease and for the years ended December 31, 2007, 2006, and 2005 was \$3,456,380, \$3,456,380, and \$4,161,426 respectively.

The Company leases facilities from its principal shareholder and an affiliate owned by the principal shareholder under non-cancelable operating lease agreements with monthly payments of \$140,000 which increases at 3% per annum beginning July 1, 2007, expiring in July 31, 2016. Lease expense for these leases recorded on a straight-line basis for years ended December 31, 2007, 2006, and 2005 was \$1,926,636, \$1,823,871, and \$1,280,000 respectively.

Future minimum lease payments, including a guaranteed residual of \$18,162,015 on the transportation equipment lease at December 31, 2007 are as follows:

	Third-Party	<b>Related-Party</b>	Total
2008	\$ 3,456,380	\$ 1,756,356	\$ 5,212,736
2009	3,456,380	1,809,047	5,265,427
2010	3,456,380	1,863,318	5,319,698
2011	3,456,380	1,919,218	5,375,598

### Table of Contents

2012	3,456,380	1,976,794	5,433,174
Thereafter	28,531,155	7,577,533	36,108,688
Total	\$ 45,813,055	\$ 16,902,266	\$ 62,715,321

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

### 5. LONG-TERM DEBT

Long-term debt consisted of the following:

	Decen 2007	ıber 31, 2006
Notes payable collateralized by construction and transportation equipment payable in monthly principal installments of \$581,344, plus fixed interest ranging from 6% to 7.9%, through maturity, which extends to January 2013.	\$ 26,181,987	\$ 21,797,178
Notes payable collateralized by construction and transportation equipment payable in monthly principal and interest installments of \$283,645, with fixed interest ranging from 6.02% to 6.99%, through maturity, which extends to December 2012.	11,755,637	7,823,987
Revolving credit facility, secured by certain marketable securities and/or cash and other general assets, monthly interest at prime rate, which is 8.25% and 7.25% at December 31, 2006 and 2007, respectively or fixed at LIBOR plus 1.50%, at the Company s option, through maturity, which is November 2008. The effective rate at December 31, 2006 was 8.25%.		10,000,000
Term loan, secured by certain marketable securities, monthly interest at prime rate, which is 8.25%, or fixed at LIBOR, plus 1.5%, at the Company s option, which is 6.85% at December 31, 2006, through maturity, which was November 2007. The effective rate at December 31, 2006 was 6.85%.		10,000,000
Real estate note payable in monthly installments of \$199,365, including principal and interest fixed at 8.47% through maturity, which is July 2010.	24,370,345	24,656,525
Mortgage note, guaranteed by the principal shareholder, payable in monthly installments of interest only at prime rate, which is 7.25% at December 31, 2007, through maturity, which is October 2008.	1,460,000	1,460,000
Mortgage note payable in monthly installments of \$8,711, including principal and interest fixed at 6.5% through maturity, which is August 2011.	312,108	393,461
Transportation equipment capital leases payable through maturity, which is June 2011.	1,610,849	1,351,329
Long-term debt	65,690,926	77,482,480
Less current maturities	12,074,077	17,261,589
Long-term debt, less current maturities	\$ 53,616,849	\$ 60,220,891

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 5. LONG-TERM DEBT, Continued

The Company, in September 2007, entered into a new revolving credit facility, which expires on August 2009. The facility, which replaced a \$25,000,000 existing facility, provides for borrowings up to \$50,000,000, including a letter of credit facility of up to \$8,000,000. As of December 31, 2007 and 2006 outstanding balances under the credit facilities were \$0 and \$10,000,000, in addition to standby letters of credit in the amounts of \$6,531,696 and \$4,500,000 and availability under the credit facilities was \$43,468,304, and \$10,500,000 respectively. Certain of the Company s marketable securities and/or cash and other general assets secure the credit facility. The credit facility contains certain financial covenants, including maintenance of minimum levels of net worth and profitability and maintenance of certain debt-to-worth and current ratios. At December 31, 2007 and 2006, the Company was in compliance with its financial covenants.

The mortgage notes payable are collateralized by separate first trust deeds on real property. The real estate note payable is collateralized by a deed of trust on real property (commercial real estate).

Maturities of long-term debt at December 31, 2007 are as follows:

2008	\$ 12,074,077
2009	10,835,411
2010	33,962,152
2011	5,820,357
2012	2,989,875
Thereafter	9,054
Total	\$ 65,690,926

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

### 6. INCOME TAXES

The provision for income taxes includes the following:

	Ye	ear En	ded December	31,	
	2007		2006		2005
<u>Current</u>					
State	\$ 2,377,493	\$	240,711	\$	761,330
Federal	249,201		277,997		11,197
Foreign	2,079,402		655,971		217,102
Total current	4,706,096		1,174,679		989,629
Deferred					
State	(422,026)		181,161		(256,005)
Foreign	119,959		310,667		120,094
Total deferred	(302,067)		491,828		(135,911)
Total provision for income taxes	4,404,029		1,666,507		853,718
Less attributed to discontinued operations	4,798		3,875		4,869
Provision for income taxes from continuing operations	\$ 4,399,231	\$	1,662,632	\$	848,849

The table below reconciles the difference between the statutory federal income tax rate and the effective rate provided for income before income taxes in the consolidation statements of income.

	Y 2007	ear Ended December 31 2006	, 2005
Federal income taxes at the statutory rate	\$ 51,398,577	\$ 3,081,197	\$ 7,195,520
State income taxes, net of federal benefits	1,941,051	421,872	506,794
Permanent Differences			
Earnings subject to different federal corporate tax rates	(48,588,048)	(1,293,088)	(6,668,534)
Change in valuation allowance		(416,457)	(334,033)
Change in rates	(22,874)	(71,996)	(205,586)
Foreign earnings taxed at different rates	(217,476)	(80,108)	235,173
Other	(107,201)	25,087	124,384
Total provision for income taxes	4,404,029	1,666,507	853,718
Less attributed to discontinued operations	4,798	3,875	4,869
-			
Provision for income taxes from continuing operations	\$ 4,399,231	\$ 1,662,632	\$ 848,849

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

### 6. INCOME TAXES, Continued

The total income before taxes for the years ended is recorded as follows:

	Yea	r Ended December 3	31,
	2007	2006	2005
Income before taxes:			
Domestic	\$ 140,647,726	\$ 5,588,177	\$ 20,884,279
Foreign	6,911,364	3,215,244	377,224
Total income before taxes	147,559,090	8,803,421	21,261,503
Less attributed to discontinued operations	230,842	(31,895)	1,411,89
-			
Total income before taxes from continuing operations	\$ 147,328,248	\$ 8,835,316	\$ 19,849,60

Operating loss carryforwards of the Company s taxable subsidiaries, which totaled \$1,606,774, were fully utilized in 2006. The decrease in the valuation allowance for years ended December 31, 2006 and 2005, associated with these operating loss carryforwards were \$694,579 and \$334,405, respectively.

Components of the net deferred tax liability (asset) recognized by type of temporary difference are as follows:

	Decem	ber 31,
	2007	2006
Current		
Other	\$ (192,896)	\$ (158,245)
Total current	(192,896)	(158,245)
Non-Current		
Contract accounting	(16,442)	97,672
Depreciation	1,325,899	1,098,959
Unrealized gain on marketable securities		1,154,294
Sale of stock book/tax difference		(177,528)
Other	1,967	2,619
Total noncurrent	1,311,424	2,176,016
Total	\$ 1,118,528	\$ 2,017,771

As of January 1, 2007 and December 31, 2007, the Company identified and reviewed potential uncertainties related to taxes upon the adoption of FIN 48 and determined that the exposure to those uncertainties did not have a material impact on the Company s results of operations or financial condition.

The Company recognizes interest and penalties accrued related to uncertain tax positions as a component of the income tax provision. There were no material uncertain tax positions as of December 31, 2007 and 2006. For the years ended December 31, 2007, 2006, and 2005, there have been no interest and penalties recorded as a component of the income tax provision.

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

### 7. STOCK TRANSACTION

On October 1, 2007, certain executives of the Company entered into a Stock Purchase Agreement with the Company s principal shareholder (principal shareholder), in which the executives acquired 34,500 shares of common stock of the Company, representing 3.83% of the shares outstanding, for total consideration of \$9,582,179. In addition, the parties entered into a Buy-Sell Agreement (Buy-Sell) dated October 1, 2007, placing substantial restrictions on the ability of the executives to sell the shares acquired under the Agreement.

In the event that the executives terminate their employment with the Company, the Buy-Sell provides that the principal shareholder has a right to purchase the shares acquired by the executives at the original price. The Buy-Sell terminates on the occurrence of a reorganization, merger, or consolidation involving another corporation in which the Company is not the survivor, or the closing of a public offering of the shares of stock of the Company ( IPO ).

Because of the restrictions placed on transfer or sale of the shares, the executives rights to any excess of the fair value of the shares estimated at \$555 per share at the grant date over cost (in the amount of \$9,565,321) can only be realized through the termination of the agreement or at the sole discretion of the principal shareholder to pay fair market value under certain provisions of the agreement. Accordingly, no compensation expense will be recognized in connection with this transaction until such time that termination of the Buy-Sell Agreement becomes likely or the Buy-Sell Agreement is terminated.

#### 8. RELATED-PARTY TRANSACTIONS

At December 31, 2007 and 2006, the Company s principal shareholder has provided guarantees to the Company in the maximum amount of \$11,000,000 that support recovery of the Company s minority interest receivable related to the Company s commercial real estate.

At December 31, 2007 and 2006, the Company s principal shareholder also has provided a guarantee on a transportation equipment lease. In addition, the Company s principal shareholder has non-cancelable operating leases on facilities (see Note 4).

The Company s principal shareholder is Chairman and CEO of Perini Corporation. Included in the Company s consolidated financial statements are \$0 and \$93,424,299 as of December 31, 2007 and 2006, respectively, in marketable securities and \$0 and \$39,214,581 as of December 31, 2007 and 2006, respectively, in marketable securities, both in connection with the Company s ownership of Perini Corporation common shares.

As a condition to an investor group s acquisition of shares of Perini Corporation Series B Preferred Stock for an aggregate of \$30 million, which was approved by the shareholders in January 1997, Perini Corporation entered into an agreement with the Company and sole shareholder of the Company, to provide certain management services, as defined. The management agreement has been renewed annually by the Perini Corporation Board of Directors under the same basic terms and conditions as the initial agreement except that the amount of the annual fee payable thereunder to the Company was increased effective September 15, 2004, from \$375,000 to \$800,000, and effective March 15, 2006, from \$800,000 to \$900,000. Compensation for the sole shareholder to provide management services consisted of \$800,000 for the year ended December 31, 2005, and \$879,000 for the year ended December 31, 2006 and \$979,000 for the year ended December 31, 2007. Payments under the agreement paid to the Company were passed directly to the sole shareholder.

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 8. RELATED-PARTY TRANSACTIONS, Continued

Included in the Company s consolidated financial statements are \$25,267,005, \$18,986,733, and \$25,488,288 in revenues for the years ended December 31, 2007, 2006, and 2005, respectively, derived from the Company s interest in various construction joint ventures with Perini Corporation. Also included were total assets of \$13,893,235 and \$11,942,279 as of December 31, 2007 and 2006, respectively, relating to these construction joint ventures. One of these joint ventures is involved in a continuing dispute with the Los Angeles County Metropolitan Transportation Authority (see Note 12).

The Company s principal shareholder, and an affiliate of the principal shareholder, had \$0 and \$4,595,513 due to the company as of December 31, 2007 and 2006, respectively. The 2006 amounts consisted of construction services due from an affiliate of the shareholder, and the shareholder, as well as short-term advances.

An affiliate of the Company s principal shareholder had an outstanding note payable due to a joint venture, of which the Company s proportionate share was \$210,368 at December 31, 2007 and 2006, and is included in the accompanying consolidated balance sheets.

#### 9. RETIREMENT PLANS

The Company participates in multi-employer defined benefit pension plans in connection with union collective bargaining agreements. Contributions are based upon hours worked by employees covered under these agreements and are funded on a current basis. The Company contributed and charged to expense \$9,053,011, \$3,198,333, and \$2,317,922 for the years ended December 31, 2007, 2006, and 2005, respectively. The increase in 2007 was primarily due to increased operations from a single, large project located in Nevada.

The Company sponsors a 401(k) profit-sharing plan for non-union domestic employees. Effective July 1, 2006, the Company began contributing \$0.30 per dollar (an increase from \$0.25 per dollar) of employee contributions up to 10% of eligible salary. Participants are fully vested in their contributions and vest at a rate of 20% per year of participation in the plan, becoming fully vested in 5 years. The Company contributed and charged to expense \$458,555, \$369,497, and \$262,778 for the years ended December 31, 2007, 2006, and 2005, respectively.

The Company sponsors a 401(k) profit-sharing plan for non-union territory of Guam employees. The Company contributes \$0.25 per dollar of employee contributions up to 6% of eligible salary. Participants are fully vested in their contributions and vest at a rate of 20% per year of participation in the plan, becoming fully vested in 5 years. The Company contributed and charged to expense \$69,858, \$57,228, and \$43,867 for the years ended December 31, 2007, 2006, and 2005, respectively.

#### **10.** ACQUISITION Acquisition of Powerco Electric Corp.

On September 30, 2007, the Company purchased 100% of the outstanding stock of Powerco Electric Corp. (PEC), a commercial electrical subcontractor, for \$3,300,000 in cash. PEC, performs work in and around the Los Angeles area, specializing in commercial building projects. The Company has included PEC s assets and liabilities in the December 31, 2007 consolidated balance sheet. The results of operations beginning October 1, 2007 are included in the Company s consolidated financial statements.

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 10. ACQUISITION, Continued

The transaction was accounted for using the purchase method of accounting as required by FASB Statement No. 141, *Business Combinations*, including the allocation of the purchase to the tangible and intangible assets of Powerco Electric Corp., which has been finalized. The following table summarizes the fair value of the assets acquired and liabilities assumed as of September 30, 2007:

Current assets	\$ 21,684,107
Property and equipment, net	325,969
Goodwill	2,826,117
Intangibles	4,160,000
Total assets acquired	28,996,193
Current liabilities	(7,745,611)
Other long-term liabilities	(17,950,582)
-	
Total acquisition costs	\$ 3,300,000

The following table identifies the intangible asset acquired and its respective amortization period. The amount assigned to an intangible asset represents the Company s estimate of the fair value of the intangible asset acquired, which is being amortized in relation to contract revenue as of October 1, 2007.

	Amount	Amortization Period
Construction contract backlog	\$ 4,160,000	4 years
Total intangible asset acquired	\$ 4,160,000	

# 11. SIGNIFICANT RISKS AND UNCERTAINTIES Concentrations:

Concentrations:

A majority of the Company s cash balances are on deposit with financial institutions and exceed federally insured deposit limits. A substantial portion of the Company s labor force (40%) is subject to collective bargaining agreements, which expire in years 2008 and beyond. Revenues from various projects in California for University of California, Los Angeles totaled \$161,809,062 (or 31% of consolidated revenues) in 2006, and \$181,410,073 (or 46% of consolidated revenues) in 2005. Revenues from a single project in Nevada for Wynn Las Vegas, LLC totaled \$581,506,255 (or 50% of consolidated revenues) in 2007, and \$120,215,151 (or 23% of consolidated revenues) in 2006. Revenues from two projects for California Department of Transportation totaled \$89,910,092 (or 23% of consolidated revenues) in 2005. Revenues from a single project for Los Angeles World Airports totaled \$64,746,760 (or 12% of consolidated revenues) in 2006.

The Company had accounts receivable due from various projects for University of California, Los Angeles in the amount of \$39,284,017 (or 18% of receivables) in 2007, and \$62,323,735 (or 43% of receivables) in 2006. Accounts receivable due from a single project for Wynn Las Vegas, LLC totaled \$95,398,666 (or 43% of receivables) in 2007, and \$22,130,719 (or 15% of receivables) in 2006.

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

### 12. OTHER COMMITMENTS AND CONTINGENCIES

Contingent liabilities of the Company include the industry standard contractor liabilities for performance and completion of construction contracts. In addition, the Company is a defendant in various lawsuits. Except as discussed below, in the opinion of management, the resolution of these matters will not likely have a material adverse effect on the financial position, result of operations or cash flows of the Company.

The Company, from time to time, enters into certain types of contracts that contingently require the Company to indemnify parties against third-party claims. These contracts primarily relate to: (i) business sales and purchase agreements, under which the Company may provide customary indemnification, and (ii) certain lease agreements, under which the Company may be required to indemnify the lessor for environmental and other liabilities, and other claims arising from the Company s use of the applicable asset.

The terms of such obligations vary. Generally, a maximum obligation is not explicitly stated. Because the obligated amounts of these types of agreements often are not explicitly stated, the overall maximum amount of the obligations cannot be reasonably estimated. Historically, the Company has not been obligated to make significant payments for these obligations, and no liabilities have been recorded for these obligations on its consolidated balance sheets as of December 31, 2007 and 2006.

#### Tutor-Saliba/Perini JV v. Los Angeles MTA 1995:

Los Angeles Superior Court Case No. BC 123559

In March 1995, a joint venture, Tutor-Saliba/Perini (TSP), in which Tutor-Saliba Corporation is the 60% partner, filed a complaint in Los Angeles Superior Court against the Los Angeles County Metropolitan Transportation Authority (MTA) for breach of contract, which sought damages of approximately \$16 million to compensate TSP and its subcontractors for the costs of extra work, delays, and interest on late payments under four separate contracts with the MTA. In February 1999, the MTA filed a cross-complaint generally alleging violation of the California False Claims Act (CFCA) and several other federal and state statutory provisions.

Trial commenced before a jury in May 2001. During trial, the Judge ruled that TSP may have failed to produce certain documents to the MTA, and the Judge penalized TSP for its alleged non-compliance by dismissing TSP s claim and by ruling, without jury finding of fact, that TSP was liable to the MTA for damages on the MTA s counterclaim. Pursuant to the Judge s instructions, the Jury awarded the MTA approximately \$32 million in damages and prejudgment interest, subsequently, in March 2002, the judge awarded an additional \$31 million in legal fees and costs. Tutor-Saliba s share of the aggregate award being approximately \$37.8 million.

TSP appealed the verdict and the Court s rulings on motions and sanctions.

On January 25, 2005, the State of California Court of Appeal issued an opinion in which it reversed the entire \$63 million trial court s judgment and found that the trial court judge had abused his discretion and violated TSP s due process rights and imposed an impermissibly overbroad sanction in issuing terminating sanctions that prevented the joint venture from presenting its claims and severely limited TSP in defending itself against the MTA s lawsuit. The Court of Appeal also directed the trial court to dismiss MTA s claims that TSP had violated the Unfair Competition Law and remanded the remainder of the case to the trial court for further proceedings, including a new trial, with a new judge, on the joint venture s claims against the MTA.

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 12. OTHER COMMITMENTS AND CONTINGENCIES, Continued

The case was reassigned to the Complex Civil Division, Los Angeles Superior Court Central Civil West, with the Presiding Judge of that division being assigned to the case. The Court is seeking ways to resolve the case short of full trial, but in a manner that would be upheld by the Court of Appeal, elected to bifurcate the trial and try the MTA s suggested strongest claim first. That claim was a pass-through subcontractor claim that the MTA agreed to pay through an executed change order for approximately \$111,000, in or about 1995 (Tunnel Handrail claim). Trial commenced on the MTA s Tunnel Handrail claim on November 17, 2006, and was completed on December 18, 2006, with the Jury entering a verdict in favor of the MTA pertaining to the handrail issue. A final judgment with respect to this claim will not be entered until the entire case has been resolved and will be subject to appeal.

In February 2007, the court granted TSP a motion and precluded MTA in future proceedings from presenting its claims that TSP breached its contract and violated the CFCA.

On December 26, 2007 the Court issued an order and opinions regarding TSP s and MTA s affirmative claims, on the night restrictions issues, which determined that no claims have sufficient prima facie evidence to proceed to trial. MTA has filed a notice of its intent to seek a new trial as to the Court s denial of MTA s night restrictions claims in the entirety.

The court has indicated that it would like the parties to resolve the entire case through mediation. Those efforts to date have failed.

Due to the major issues of the case still remaining to be scheduled for retrial, the ultimate financial impact of the lawsuit is not yet determinable. Therefore, no provision for loss, if any, has been recorded in the consolidated financial statements.

### 13. LITIGATION SETTLEMENT

City and County of San Francisco v. Tutor-Saliba Corporation, Perini Corporation and

Buckley & Company Inc., A Joint Venture, et al. 2003:

United States District Court-Northern District Case No. C02-5286-CW

On November 1, 2002, the City and County of San Francisco (CCSF) filed a complaint in United States District Court-Northern District against the joint venture, Tutor-Saliba Corporation, Perini Corporation and Buckley & Company, Inc. (TSPB), its individual members, including Tutor-Saliba Corporation (the Company) and their respective bonding companies. The complaint was not served on the Company or the bonding companies until 2003. The complaint arose out of six San Francisco International Airport construction contracts with the San Francisco Airport Commission (Airport Commission) entered into during 1996 and 1997 that were subsequently successfully completed and closed-out under an agreement between TSPB, the Company, and the Airport Commission, in 2001.

On February 17, 2006, a settlement agreement was reached with the CCSF during mediation. The settlement agreement was reached in pursuit of the best interests of all parties involved, to put to rest divisive issues, and avoid the considerable continuing future cost and expense of litigation. The agreement resolves all issues of disputed liability between the parties and allows the parties to refocus their resources towards more productive purposes. The settlement agreement provided that the CCSF will be paid the total sum of \$19 million, payable over a period of years ending in 2009. The Company recorded a charge to net income of \$14,651,537 representing its portion of this settlement in the year ended December 31, 2005. The San Francisco Board of Supervisors and the Airport Commission approved the agreement on March 30, 2006. At

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 13. LITIGATION SETTLEMENT, Continued

December 31, 2007, the Company s present value portion of the remaining liability of \$5,551,486, of which \$2,179,960 is current, is included in the accompanying consolidated balance sheet.

#### 14. DISCONTINUED OPERATIONS

Effective January 1, 2008, the Company entered into an exchange transaction with an affiliate owned by the Company s majority shareholder. The Company transferred its partnership interest in a commercial office building in exchange for a 2% interest in one of the Company s consolidated subsidiaries that was held by the affiliated company. The December 31, 2007 consolidated balance sheet reflects \$26,759,142 in property and equipment, minority interest receivable of \$13,423,058, cash of \$3,855,279, other net assets of \$1,853,748 and mortgage debt of \$24,370,345 associated with the Company s interest in this partnership. The December 31, 2007 book value of the 2% interest in the Company s consolidated subsidiary is \$250,749. Based on this exchange transaction, the activities relating to the commercial office building have been reported as discontinued operations for all periods presented. Revenues of \$6,686,436, \$6,393,109 and \$6,342,813 and pretax income (loss) of \$230,842, \$(31,895) and \$1,411,898 for the years ended December 31, 2007, 2006 and 2005, respectively were reported as discontinued operations. The transaction had no impact to net income in the accompanying consolidated statements of income. The net book value of assets distributed over assets received was recorded as a distribution of \$21,270,133 on January 1, 2008.

### 15. SUBSEQUENT EVENTS

### Purchase of North Valley Commerce Center, Building G

On October 30, 2007, the Company entered into an agreement to purchase the North Valley Commerce Center, Building G in Sylmar California for \$6.5 million in cash. The property will consist of an office and warehouse facility, currently under construction. As of January 25, 2008, the Company has made initial deposits for the property totaling \$300,000 into an escrow account. The purchase is expected to close upon substantial completion of the building.

#### Acquisition of Desert Plumbing & Heating Co. Inc.

On January 4, 2008, the Company purchased 100% of the outstanding stock of Desert Plumbing & Heating Co. Inc. (DPH), a mechanical subcontractor located in Las Vegas, NV, for \$35.0 million cash and an earn-out based upon a percentage of DPH s earnings over the next three years, up to a maximum of \$4 million annually. DPH operates primarily in the state of Nevada and had approximately 25 contracts in progress, primarily in the Las Vegas hotel and gaming market, at the time of the acquisition.

In the event the Company becomes a public company by public offering or otherwise, the earn-out payments to the former DPH shareholders will be payable in shares of stock in the parent company in an amount equivalent to the maximum earn-out payment of \$12 million.

The transaction will be accounted for using the purchase method of accounting as required by FASB Statement No. 141 *Business Combinations*, including the allocation of the purchase to the tangible and intangible assets of DPH.

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### As of December 31, 2007 and 2006 and for each of the three years in the period ended December 31, 2007

#### 15. SUBSEQUENT EVENTS, Continued

#### All Star Aggregate, Inc.

On March 7, 2008, the Company entered into an agreement for the transfer and purchase of certain mineral material mining contracts, equipment, and material stockpiles used to operate All Star Aggregate, Inc. s existing business in Sloan Pit, located in Clark County, NV, for a purchase price of \$5,200,000.

#### Bureau of Land Management Leases

On March 12, 2008, the Company entered into a ten-year renewable contract with the Bureau of Land Management (BLM) for the extraction of 19,500,000 tons of limestone aggregate on land located at Mount Diablo Meridian, NV. Pursuant to the Company's September 12, 2007 high bid of \$0.90 per ton and a \$0.01 per ton administrative fee, the total contract price equates to \$17,745,000 and requires a 5% deposit in the amount of \$877,500 prior to beginning extraction. The balance of the contract price will be paid monthly over the contract term based upon the value of materials removed in the prior month. In addition, the Company is obligated under a separate agreement to pay a third party a finders fee related to the transaction in the amount of \$300,000.

#### Perini Merger

On April 2, 2008, the Company entered into a definitive agreement with Perini Corporation, a publicly traded construction company, under which the two companies would be combined, in tax-free, all-stock merger. The transaction is subject to customary closing conditions, including the approval of Perini s shareholders and receipt of regulatory approvals.

Annex I

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### CONSOLIDATED BALANCE SHEET (UNAUDITED)

	March 31, 2008
ASSETS	
Current assets:	
Cash and cash equivalents	\$ 110,327,937
Restricted cash	16,000,000
Receivables	286,833,794
Unbilled work	32,385,960
Advances to related parties	15,310,000
Advances to affiliates	72,528
Inventory	2,633,907
Deferred income taxes	192,896
Other current assets	8,656,322
Total current assets	472,413,344
Property and equipment, net of depreciation	85,197,703
Noncurrent assets:	21 101 4/7
Goodwill	21,181,467
Identified intangible asset, net	10,688,794
Mineral reserves	12,849,926
Other assets	1,648,790
Total noncurrent assets	46,368,977
Total assets	\$ 603,980,024
LIABILITIES AND SHAREHOLDERS EQUITY	
Current liabilities:	
Current maturities of long-term debt	\$ 11,821,768
Accounts and subcontracts payable	277,342,111
Accrued liabilities	27,817,245
Advances from affiliates	665,645
Deferred contract revenue	57,612,517
Income taxes payable	842,654
Total current liabilities	376,101,940
Noncurrent liabilities:	
Long-term debt, less current maturities	26,905,011
Deferred income taxes	1,311,424
Royalty payables	11,361,031
Other long-term liabilities	3,785,892
Total noncurrent liabilities	43,363,358

### Total liabilities

419,465,298
-------------

Commitments and Contingencies: (Note 13)	
Shareholders Equity:	
Capital stock, no par value: authorized 1,000,000 shares:	
issued and outstanding 900,043 shares	6,756,510
Additional paid-in capital	11,075,563
Accumulated other comprehensive income	101,255
Retained earnings	166,581,398
Total shareholders equity	184,514,726
Total liabilities and shareholders equity	\$ 603,980,024

See notes to consolidated financial statements

I-1

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### CONSOLIDATED STATEMENTS OF INCOME (UNAUDITED)

	Three Months F 2008	Three Months Ended March 31, 2008 2007		
REVENUE:				
Construction revenue	\$ 394,827,385	\$ 231,852,863		
Sales revenue	135,642	206,969		
Total revenue	394,963,027	232,059,832		
COST OF REVENUES:				
Cost of construction revenue	359,026,165	211,273,652		
Amortization of intangible asset	1,240,363			
Cost of sales revenue	92,150	130,227		
Total cost of revenues	360,358,678	211,403,879		
Gross profit	34,604,349	20,655,953		
Gain on sale of property and equipment	(148,486)	(49,789)		
General and administrative expenses	13,556,824	7,753,641		
Concrat and administrative expenses	10,000,021	7,755,611		
Income from operations	21,196,011	12,952,101		
OTHER INCOME (EXPENSES):				
Interest and other income	1,398,286	742,790		
Net gain on sale of marketable securities	, , , , , , , , , , , , , , , , , , ,	11,426,900		
Interest expense	(778,695)	(1,004,613)		
Other expense	(135,718)	(28,444)		
Total other income, net	483,873	11,136,633		
Income before minority interest and income taxes	21,679,884	24,088,734		
Provision for income taxes	618,854	796,424		
Income before minority interest	21,061,030	23,292,310		
Minority interest	21,001,050	(20,525)		
INCOME FROM CONTINUING OPERATIONS	21,061,030	23,271,785		
DISCONTINUED OPERATIONS:				
Loss from discontinued operations before minority interest and income taxes		(102,131)		
Provision for income taxes		800		
Loss before minority interest		(102,931)		
Minority interest		64,480		
LOSS FROM DISCONTINUED OPERATIONS		(38,451)		
NET INCOME	\$ 21,061,030	\$ 23,233,334		

See notes to consolidated financial statements

#### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### CONSOLIDATED STATEMENTS OF SHAREHOLDERS EQUITY (UNAUDITED)

### Three months ended March 31, 2008

	Capital Stock	Additional Paid-in Capital	Con	cumulated Other nprehensive Income	Retained Earnings	Total
BALANCE AT JANUARY 1, 2008	\$ 6,756,510	\$ 11,075,563	\$	144,570	\$ 193,653,700	\$ 211,630,343
Comprehensive income:						
Net income Three months ended March 31, 2008					21,061,030	21,061,030
Other comprehensive income:						
Foreign currency translation adjustments (net of tax of						
(\$649))				(43,315)		(43,315)
Total comprehensive income				(43,315)	21,061,030	21,017,715
Distributions to shareholder					(48,133,332)	(48,133,332)
BALANCE AT MARCH 31, 2008	\$ 6,756,510	\$ 11,075,563	\$	101,255	\$ 166,581,398	\$ 184,514,726

See notes to consolidated financial statements

### TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

### CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

	Three months ended March 31, 2008 2007		
CASH FLOWS FROM OPERATING ACTIVITIES:			
Net Income	\$ 21,061,030	\$ 23,233,334	
Adjustments to reconcile net income to net cash provided by (used in) operating activities:			
(used in) operating activities:			
Depreciation and amortization	1,673,313	1,528,065	
Amortization of intangible asset	1,240,363		
Net gain from sale of property and equipment	(148,486)	(49,789)	
Net gain on sale of marketable securities		(11,426,900)	
Minority interest in consolidated subsidiaries		(43,955)	
Increase (decrease) in cash attributed to changes in operating assets and liabilities:			
Receivables	(39,652,016)	(42,635,664)	
Inventory	508,325	(75,682)	
Other assets	(1,955,417)	(2,408,544)	
Advances from (to) joint ventures partners, net	177,247	(222,632)	
Advances to related parties		(173,728)	
Unbilled work	(1,686,759)	(7,959,964)	
Accounts payable and other liabilities	21,155,510	38,251,724	
Deferred contract revenue	(213,545)	1,571,369	
Net cash provided by (used in) operating activities	2,159,565	(412,366)	
CASH FLOWS FROM INVESTING ACTIVITIES:			
Proceeds from sale of property and equipment	657,941	524,344	
Proceeds from sales of marketable securities		23,791,000	
Purchases of property and equipment	(12,813,331)	(4,390,819)	
Cash paid on acquisition of mineral reserves	(1,488,895)		
Decrease in restricted cash	8,000,000		
Advances to related parties	(15,310,000)		
Cash paid on acquisition, net of cash acquired	(30,295,894)		
Net cash (used in) provided by investing activities	(51,250,179)	19,924,525	
CASH FLOWS FROM FINANCING ACTIVITIES:			
Proceeds from long-term debt		2,378,860	
Principal payments of long-term obligations	(2,665,872)	(6,078,529)	
Distributions to shareholders	(30,718,478)	(1,834,999)	
Net cash used in financing activities	(33,384,350)	(5,534,668)	
Net increase (decrease) in cash and cash equivalents	(82,474,964)	13,977,491	
Cash and cash equivalents at beginning of period	192,802,901	37,575,733	
Cash and cash equivalents at end of period	\$ 110,327,937	\$ 51,553,224	
SUPPLEMENTAL CASH FLOW INFORMATION:			
Cash paid during the period for:			
Interest	\$ 793,270	\$ 1,537,526	
	, , , , , , , , , , , , , , , ,	. ,,	

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

Income taxes	\$ 2,305,734	\$	360,553
NONCASH INVESTING AND FINANCING ACTIVITIES:			
Real estate assets and liabilities distributed to principal shareholder	\$ 17,414,854		
Capitalized lease obligation incurred for transportation equipment	\$ 72,070	\$	20,140
Unrealized holding gains on marketable securities		\$ 1	6,196,544
Royalty liabilities accrued from acquisition of mineral reserves	\$ 11,361,031		

See notes to consolidated financial statements

I-4

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

## As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 1. SIGNIFICANT ACCOUNTING POLICIES

#### Nature of Business:

Tutor-Saliba Corporation was incorporated in 1981 as a successor to businesses that had been providing construction services since 1949. Tutor-Saliba Corporation and its majority-owned subsidiaries (Tutor-Saliba or the Company) provide diversified general contracting, design-build, and preconstruction services to public agencies and private clients. The Company's construction business is conducted through three basic segments or operations: domestic civil, domestic building, and international. The Company focuses on large, complex construction projects in the civil infrastructure and commercial building sectors. The Company manages all aspects of these construction projects and directly performs many significant construction specialties, including concrete forming and placement, site excavation, structural steel erection, and electrical and mechanical services. The Company provides these services by using traditional general contracting arrangements, such as fixed price, guaranteed maximum price, and cost plus fee contracts.

In an effort to limit its financial and/or operational risk on certain of its larger and more complex projects, the Company participates in construction joint ventures, often as the sponsor or manager of the project, for the purpose of bidding and, if awarded, providing the agreed-upon construction services. Each joint venture participant usually agrees in advance to provide a predetermined percentage of capital, as required, and to share in the same percentage of profit or loss on the project.

#### Basis of Presentation:

The unaudited consolidated financial statements presented herein include the accounts of Tutor-Saliba Corporation and its majority-owned subsidiaries. The Company s interests in construction joint ventures are accounted for using the proportionate consolidation method. All intercompany balances and transactions have been eliminated in consolidation. These statements should read in conjunction with the Company s audited financial statements for the year ended December 31, 2007. In the opinion of management, the accompanying unaudited consolidated financial statements include all adjustments, consisting only of normal recurring adjustments, necessary to present fairly the Company s financial position as of March 31, 2008 and results of operations and cash flows for the three months ended March 31, 2008 and 2007. The results of operations for the three months ended March 31, 2008 may not be indicative of the results that may be expected for the year ending December 31, 2008 because, among other reasons, such results can vary depending on the timing of progress achieved and changes in estimated profitability of projects being reported.

#### Use of Estimates in the Preparation of Financial Statements:

The financial statements have been prepared in conformity with accounting principles generally accepted in the United States of America, which requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosures of contingent assets and liabilities at the date of the financial statements as well as the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. On an ongoing basis, the Company reviews its estimates based on information that is currently available. Changes in facts and circumstances may result in revised estimates.

I-5

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

#### Method of Accounting for Contracts:

Revenues and profits from the Company s construction contracts are recognized by applying percentages of completion for the period to the total estimated profits for the respective contracts. Percentage-of-completion is determined by relating the actual cost of the work performed to date to the current estimated total cost of the respective contracts. When the estimate on a contract indicates a loss, the Company s policy is to record the entire loss during the accounting period in which it is estimated. In the ordinary course of business, at a minimum on a quarterly basis, the Company prepares updated estimates of the total forecasted revenue, cost, and profit or loss for each contract. The cumulative effect of revisions in estimates of the total forecasted revenue and costs, including unapproved change orders and claims, during the course of the work is reflected in the accounting period in which the facts that caused the revision become known. The financial impact of these revisions to any one contract is a function of both the amount of the revision and the percentage-of-completion of the contract. An amount equal to the costs incurred, which are attributable to unapproved change orders and claims are included in the total estimated revenue when realization is determined to be probable. Profits from unapproved change orders and claims are recorded in the total estimated revenue in the period such amounts are resolved.

The Company includes in current assets and current liabilities amounts related to construction contracts realizable and payable over a period in excess of one year. Deferred contract revenue represents excess of billings to date over the amount of contract costs and profits (or contract revenue) recognized to date on the percentage-of-completion accounting method on certain contracts. Unbilled work represents the excess of contract costs and profits (or contract revenue) recognized to date on the percentage-of-completion accounting method over billings to date on the remaining contracts. Unbilled work results when (1) the appropriate contract revenue amount has been recognized in accordance with the percentage-of-completion accounting method, but a portion of the revenue recorded cannot be billed currently due to the billing terms defined in the contract and/or (2) costs, recorded at estimated realized value, related to unapproved change orders or claims are incurred. Unbilled work on contracts at March 31, 2008 consisted of the following:

Unbilled costs and profits incurred to date	\$ 6,849,259
Unapproved change orders	13,855,087
Claims	11,681,614

\$ 32,385,960

The prerequisite for billing Unbilled costs and profits incurred to date is provided in the defined billing terms of each of the applicable contracts. The prerequisite for billing Unapproved change orders or Claims is the final resolution and agreement between the parties. The amount of unbilled work at March 31, 2008, estimated by management to be collected beyond one year is approximately \$6,585,000.

#### Significant Estimates:

As outlined above, the Company s revenue is recognized on the percentage-of-completion basis. Consequently, construction revenue and gross margin for each reporting period are determined on a contract-by-contract basis by reference to estimates by the Company s management of expected costs to be incurred to complete each project. These estimates include provisions for known and anticipated cost overruns and cost recoveries, if any exist or are expected to occur. These estimates are subject to revision in the normal course of business and could be material.

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

#### Foreign Currency Translation:

The functional currency for foreign operations is the local currency. The assets and liabilities of the Company s international subsidiaries are translated into U.S. dollars using current exchange rates at the balance sheet date. Operating statement items are translated at average exchange rates prevailing during the period. The resulting cumulative translation adjustment of \$101,255 at March 31, 2008 is recorded in the foreign currency translation adjustment account as part of accumulated other comprehensive income in shareholder s equity. Foreign currency transaction gains and losses, if any, are included in operations as they occur.

#### Cash and Cash Equivalents:

Cash equivalents represent all highly liquid debt instruments with maturities, when purchased, of three months or less.

Cash and cash equivalents as reported in the accompanying consolidated balance sheets consist of amounts held by the Company that are available for general corporate purposes and the Company s proportionate share of amounts held by construction joint ventures that are available only for joint venture-related uses. Cash held by construction joint ventures is distributed from time to time to the Company and to the other joint venture participants in accordance with their percentage interest. Cash distributions received by the Company from its construction joint ventures are then available for general corporate purposes. Cash and cash equivalents at March 31, 2008 consisted of the following:

Corporate cash and cash equivalents (available for general purposes)	\$ 98,774,699
Company s share of joint venture cash and cash equivalents (available only for joint venture purposes, including future distributions)	11,553,238
	\$ 110,327,937

## Restricted Cash

As of March 31, 2008, the Company had \$16,000,000 of restricted cash. Restricted cash consists of amounts held as collateral on the revolving credit facility (see Note 5).

#### Inventory:

Inventory consists of construction materials, equipment parts, and supplies that have not been assigned and charged to specific contracts. Inventory is stated at the lower of cost or market. Cost is determined generally on the specific identification method.

#### Property, Equipment, and Long-Lived Assets:

Property and equipment are stated at cost. Depreciation is provided using the straight-line method over their estimated useful lives ranging from 3 to 40 years after an allowance for salvage.

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

All long-lived assets are reviewed by management in accordance with FASB Statement No. 144, *Accounting for the Impairment or Disposal of Long-Lived Assets*, for impairment whenever events or changes in circumstance indicate that the carrying value of such assets may not be recoverable from expected future undiscounted cash flows. Major renewals and betterments are capitalized and maintenance and repairs are charged to operations as incurred. When property and equipment are retired or otherwise disposed of, the cost and accumulated depreciation are removed from the appropriate account and any gain or loss is included in results of operations. No impairments were identified as of March 31, 2008.

#### Goodwill and other Intangible Assets:

Goodwill is subject to annual impairment test pursuant to FASB Statement No. 142, *Goodwill and Other Intangible Assets*. The Company regularly evaluates whether events or circumstances have occurred which may indicate a possible impairment of goodwill. The Company believes the methodology used in testing impairment of goodwill, which includes significant judgements and estimates, provides a reasonable basis in determining whether an impairment charge should be taken.

## Income Taxes:

The Company accounts for income taxes in accordance with FASB Statement No. 109, *Accounting for Income Taxes*, which requires the use of an asset and liability method of accounting for income taxes. Deferred income taxes are provided to reflect the tax effect of differences between the financial statement carrying amounts and the tax bases of assets and liabilities. The Company elected to be treated as an S corporation effective January 1, 1996. As a result, taxable income, loss and credits flow directly to the shareholder and tax-related assets and liabilities of the Company become the obligation of the shareholder of the S corporation and are no longer reflected in the financial statements. The deferred tax assets, liabilities, and provision reflected in the financial statements are those that do not flow through to the shareholder, as they relate to taxable subsidiaries.

In July 2006, the FASB issued FASB Interpretation (FIN) No. 48, *Accounting for Uncertainty in Income Taxes an Interpretation of FASB Statement No. 109.* FIN No. 48 increases the relevancy and comparability of financial reporting by clarifying the way companies account for uncertainty in measuring income taxes. FIN No. 48 prescribes a comprehensive model for how a company should recognize, measure, present, and disclose in its financial statements uncertain tax positions that the company has taken or expects to take on a tax return. FIN No. 48 only allows a favorable tax position to be included in the calculation of tax liabilities and expenses if a company concludes that it is more likely than not that its adopted tax position will prevail if challenged by tax authorities. FIN No. 48 also provides guidance on the accounting for and recording of interest and penalties on uncertain tax positions. The Company adopted FIN No. 48 on January 1, 2007, and the adoption of FIN No. 48 did not have an impact on the financial condition and results of operations.

#### Fair Value of Financial Instruments:

The carrying value of receivables and other amounts arising out of normal contract activities, including retentions that may be settled beyond one year, are estimated to approximate fair value. All significant debt obligations, interest rates, and their carrying value are considered to approximate market rates and fair value.

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

#### 1. SIGNIFICANT ACCOUNTING POLICIES, Continued

#### New Accounting Pronouncements:

In September 2006, the FASB issued FASB Statement No. 157, *Fair Value Measurements*. FASB Statement No. 157 defines fair value, establishes a framework for measuring fair value, and expands disclosures about fair value measurements. In February 2008, the FASB issued FASB Staff Position (FSP) No. FAS 157-1 and FSP No. FAS 157-2, affecting implementation of FASB Statement No. 157. FSP No. FAS 157-1 excludes FASB Statement No. 13, *Accounting for Leases*, and other accounting pronouncements that address fair value measurements under FASB Statement No. 13, from the scope of FASB Statement No. 157. FSP No. FAS 157-2 delays the effective date of FASB Statement No. 157 for nonfinancial liabilities, except for items that are recognized or disclosed at fair value on a recurring basis, to fiscal years beginning after November 15, 2008. The adoption of FASB Statement No. 157 did not have a material impact on the financial condition and results of operations.

In February 2007, the FASB issued FASB Statement No. 159, *The Fair Value Option for Financial Assets and Financial Liabilities including an Amendment of FASB No. 115. FASB Statement No. 159 permits entities to choose to measure many financial instruments and certain other items at fair value. The Company adopted FASB Statement No. 159 on January 1, 2008, as required. The Company did not elect the fair value measurement option for any of its financial assets or liabilities. Therefore, the adoption of FASB Statement No. 159 had no impact on the Company s financial statements.* 

In December 2007, the FASB issued FASB Statement No. 141 (revised 2007), *Business Combinations*. FASB Statement No. 141(R) establishes principles and requirements for how an acquirer recognizes and measures in its financial statements the identifiable assets acquired, the liabilities assumed, any noncontrolling interest in the acquiree and the goodwill acquired. FASB Statement No. 141(R) also establishes disclosure requirements to enable the evaluation of the nature and financial effects of the business combination. FASB Statement No. 141(R) is effective for fiscal years beginning after December 15, 2008. The Company will apply the provision of FASB Statement No. 141(R) prospectively as of that date.

In December 2007, the FASB issued FASB Statement No. 160, *Noncontrolling Interests in Consolidated Financial Statements an amendment of Accounting Research Bulletin No. 51*. FASB Statement No. 160 establishes accounting and reporting standards for ownership interests in subsidiaries held by parties other than the parent, the amount of consolidated net income attributable to the parent and to the noncontrolling interest, changes in a parent s ownership interest, and the valuation of retained noncontrolling equity investments when a subsidiary is deconsolidated. FASB Statement No. 160 also establishes disclosure requirements that clearly identify and distinguish between the interests of the parent and the interests of the noncontrolling owners. FASB Statement No. 160 is effective for fiscal years beginning after December 15, 2008. The Company is currently evaluating the potential impact, if any, of the adoption of FASB Statement No. 160 on the Company s financial condition and results of operations.

I-9

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

## As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 2. RECEIVABLES

Receivables at March 31, 2008 consisted of the following:

Receivables:	
Contract receivables	\$ 163,680,437
Retainage receivables	121,146,489
Trade and other	2,006,868
Total receivables	\$ 286,833,794

#### Retainage:

Many of the contracts under which the Company performs work contain retainage provisions. Retainage refers to that portion of revenue earned by the Company but held for payment by the customer pending satisfactory completion of the project. The Company believes that all amounts retained by customers under such provisions are fully collectible. Retainage on active contracts is classified as a current asset regardless of the term of the contract. Retainage is generally collected within one year of the completion of the contract. As of March 31, 2008, the amount of retainage expected to be collected beyond one year was \$69,088,317.

#### 3. PROPERTY AND EQUIPMENT

Property and equipment, at cost, at March 31, 2008 consisted of the following:

Land	\$ 4,496,263
Building and improvements	12,912,609
Transportation equipment	12,345,360
Construction equipment	91,169,147
Office equipment	3,209,285
Property and equipment, gross	124,132,664
Less accumulated depreciation	38,934,961
Property and equipment, net of depreciation	\$ 85,197,703

Included in property and equipment as of March 31, 2008, are amounts recorded under capital leases for transportation equipment of \$3,259,722. Accumulated depreciation on transportation equipment recorded under capital leases was \$1,562,897 as of March 31, 2008. Effective January 1, 2008, a commercial office building was transferred to an affiliate of the principal shareholder (see Note 9).

## 4. OPERATING LEASES

## Edgar Filing: IDEX CORP /DE/ - Form 10-K

During 2005, the Company entered into a new non-cancelable operating lease for a corporate aircraft after the buyout of an existing lease. The Company has the right to voluntarily terminate the lease and purchase the asset on any installment payment date after the first 12-month anniversary of the lease for the outstanding lease balance. Upon expiration of the initial lease term in December 2015, the Company has the option of either purchasing the asset for the outstanding lease balance or returning the asset to the lessor.

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 4. **OPERATING LEASES, Continued**

The Company s shareholder has personally guaranteed this lease. Total rental expense under this lease was \$864,095 for the three months ended March 31, 2008, and 2007.

The Company leases facilities from its principal shareholder and an affiliate owned by the principal shareholder under non-cancelable operating lease agreements with monthly payments of \$140,000, which increases at 3% per annum beginning July 1, 2007, expiring in July 31, 2016. Lease expense for these leases recorded on a straight-line basis was \$481,659 for three months ended March 31, 2008, and 2007.

Future minimum lease payments, including a guaranteed residual of \$18,162,015 on the transportation equipment lease at March 31, 2008, are as follows:

	Third Party	Related Party	Total
2008 (remaining 9 months)	\$ 2,592,285	\$ 1,323,756	\$ 3,916,041
2009	3,456,380	1,809,047	5,265,427
2010	3,456,380	1,863,318	5,319,698
2011	3,456,380	1,919,218	5,375,598
2012	3,456,380	1,976,794	5,433,174
Thereafter	28,531,155	7,577,533	36,108,688
Total	\$ 44,948,960	\$ 16,469,666	\$ 61,418,626

## 5. LONG-TERM DEBT

Long-term debt at March 31, 2008 consisted of the following:

Notes payable collateralized by construction and transportation equipment payable in monthly principle installments of \$581,344 plus fixed interest ranging from 6.00% to 7.95%, through maturity, which extends to	
January 2013.	\$ 24,350,708
Notes payable collateralized by construction and transportation equipment payable in monthly principal and interest installments of \$283,645, with fixed interest ranging from 6.02% to 6.99%, through maturity, which	
extends to December 2012.	11,092,570
Mortgage note, guaranteed by the principal shareholder, payable in monthly installments of interest only at prime rate, which is 6.0% at March 31, 2008, through maturity, which is October 2008.	1,460,000
Mortgage note payable in monthly installments of \$8,711, including principal and interest fixed at 6.5% through maturity, which is August 2011.	290,932
Transportation equipment capital leases payable through maturity, which is June 2011.	1,532,569
Long-term debt	38,726,779
Less current maturities	11,821,768

Long-term debt, less current maturities

The Company, in September 2007, entered into a new revolving credit facility, which expires August 2009. The facility, which replaced a \$25,000,000 existing facility, provides for borrowings up to \$50,000,000,

I-11

\$ 26,905,011

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 5. LONG-TERM DEBT, Continued

including a letter of credit facility of up to \$8,000,000. As of March 31, 2008 outstanding balances under the credit facilities were \$0, in addition to standby letters of credit in the amounts of \$6,531,696 and availability under the credit facilities was \$43,468,304. Certain of the Company s marketable securities and/or cash and other general assets secure the credit facility. The credit facility contains certain financial covenants, including maintenance of minimum levels of net worth and profitability and maintenance of certain debt-to-worth and current ratios. At March 31, 2008, the Company was in compliance with its financial covenants.

The mortgage notes payable are collateralized by separate first trust deeds on real property.

At March 31, 2008, maturities of long-term debt at March 31 for each fiscal year are as follows:

2009	\$ 11,821,768
2010	10,546,742
2011	9,215,964
2012	5,098,962
2013	2,043,343
Total	\$ 38,726,779

## 6. INCOME TAXES

The provision for income taxes includes the following:

	Three Months Ended March 31,	
	2008	2007
Current		
State	\$ 139,764	\$ 266,280
Federal	104,480	
Foreign	374,610	455,700
Total current	618,854	721,980
Deferred		
State		75,244
Foreign		
Total deferred		75,244
Total provision for income taxes	618,854	797,224
Less attributed to discontinued operations		800

Total provision for income taxes from continuing operations

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

## As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 6. INCOME TAXES, Continued

The table below reconciles the difference between the statutory federal income tax rate and the effective rate provided for income before income taxes in the Consolidation Statements of Income.

	Three Months E 2008	nded March 31, 2007
Federal income taxes at the statutory rate	\$ 7,587,959	\$ 8,410,695
State income taxes, net of federal benefits	139,764	341,741
Permanent differences		
Earnings subject to different federal corporate tax rates	(7,122,495)	(7,945,648)
Change in rates	(9,953)	
Foreign earnings taxed at different rates	23,579	(7,116)
Other		(2,448)
Total provision for income taxes	618,854	797,224
Less attributed to discontinued operations		800
Total provision for income taxes from continuing operations	\$ 618,854	\$ 796,424

The total income before taxes is recorded as follows:

		Three Months Ended March 31,	
	2008	2007	
Income before taxes:			
Domestic	\$ 20,676,937	\$ 22,701,851	
Foreign	1,002,947	1,328,707	
Total income before taxes	21,679,884	24,030,558	
Less attributed to discontinued operations		(37,651)	
Total income before taxes on continuing operations	\$ 21,679,884	\$ 24,068,209	

Components of the net deferred tax liability (asset) recognized by type of temporary difference at March 31, 2008 are as follows:

Current Other	\$ (102.806)
Ottler	\$ (192,896)
Total current	(192,896)

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

Noncurrent	
Contract accounting	(16,442)
Depreciation	1,325,899
Other	1,325,899 1,967
Total noncurrent	1,311,424
Total	\$ 1,118,528

I-13

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

#### 6. INCOME TAXES, Continued

As of January 1, 2007 and March 31, 2008, the Company identified and reviewed potential uncertainties related to taxes upon the adoption of FIN No. 48 and determined that the exposure to those uncertainties did not have a material impact on the Company s results of operations or financial condition.

The Company recognizes interest and penalties accrued related to uncertain tax positions as a component of the income tax provision. There were no material uncertain tax positions as of March 31, 2008. For the three months ended March 31, 2008, there have been no interest and penalties recorded as a component of the income tax provision.

#### 7. STOCK TRANSACTION

On October 1, 2007, certain executives of the Company entered into a Stock Purchase Agreement with the Company s principal shareholder (principal shareholder), in which the executives acquired 34,500 shares of common stock of the Company, representing 3.83% of the shares outstanding, for total consideration of \$9,582,179. In addition, the parties entered into a Buy-Sell Agreement (Buy-Sell) dated October 1, 2007, placing substantial restrictions on the ability of the executives to sell the shares acquired under the Agreement.

In the event that the executives terminate their employment with the Company, the Buy-Sell provides that the principal shareholder has a right to purchase the shares acquired by the executives at the original price. The Buy-Sell terminates on the occurrence of a reorganization, merger, or consolidation involving another corporation in which the Company is not the survivor, or the closing of a public offering of the shares of stock of the Company (IPO).

Because of the restrictions placed on transfer or sale of the shares, the executives rights to any excess of the fair value of the shares estimated at \$555 per share at the grant date over cost (in the amount of \$9,565,321) can only be realized through the termination of the agreement or at the sole discretion of the principal shareholder to pay fair market value under certain provisions of the agreement. Accordingly, no compensation expense will be recognized for the shares issued in connection with this transaction until such time that termination of the Buy-Sell Agreement becomes likely or the Buy-Sell Agreement is terminated. Distributions related to the shares held by the executives will be recorded as compensation expense in the period incurred. During the three months ended March 31, 2008, \$1,070,753 was recorded as compensation expense in General and administrative expenses in the accompanying consolidated statement of income.

## 8. RELATED-PARTY TRANSACTIONS

At March 31, 2008, the Company s principal shareholder has provided a guarantee on a transportation equipment lease. In addition, the Company s principal shareholder has non-cancelable operating leases on facilities (see Note 4).

The Company s principal shareholder is Chairman and CEO or Perini Corporation. As a condition to an investor group s acquisition of shares of Perini Corporation Series B Preferred Stock for an aggregate of \$30 million, which was approved by the shareholders in January 1997, Perini Corporation entered into an agreement with the Company and sole shareholder of the Company, to provide certain management services, as defined. The management agreement has been renewed annually by the Perini Corporation Board of Directors under the same basic terms and conditions as the initial agreement except that the

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

#### 8. RELATED-PARTY TRANSACTIONS, Continued

amount of the annual fee payable thereunder to the Company was increased effective September 15, 2004, from \$375,000 to \$800,000, and effective March 15, 2006, from \$800,000 to \$900,000. Compensation for the principal shareholder to provide management services consisted of \$233,333 for the three months ended March 31, 2007 and \$250,000 for the three months ended March 31, 2008. Payments under the agreement paid to the Company were passed directly to the principal shareholder.

Included in the Company s financial statements are \$7,675,245 and \$4,693,237 in revenues for the three months ended March 31, 2008 and 2007, respectively, derived from the Company s interest in various construction joint ventures with Perini Corporation. Also included were total assets of \$15,613,493 as of March 31, 2008, relating to these construction joint ventures. One of these joint ventures is involved in a continuing dispute with the Los Angeles County Metropolitan Transportation Authority (see Note 13).

The Company s principal shareholder had \$15,310,000 due to the Company as of March 31, 2008. This amount consisted of short term advances.

An affiliate of the Company s principal shareholder had an outstanding note payable due to a joint venture, of which the Company s proportionate share was \$210,368 as of March 31, 2008, and is included in the accompanying balance sheet.

## 9. DISCONTINUED OPERATIONS

Effective January 1, 2008 the Company entered into an exchange transaction with an affiliate owned by the Company s majority shareholder. The Company transferred its partnership interest in a commercial office building in exchange for a 2% interest in one of the Company s consolidated subsidiaries that was held by the affiliated company. The December 31, 2007 balance sheet reflects \$26,759,142 in property and equipment, minority interest receivable of \$13,423,058, cash of \$3,855,279, other net assets of

\$1,853,748 and mortgage debt of \$24,370,345 associated with the Company s interest in this partnership. The December 31, 2007 book value of the 2% interest in the Company s consolidated subsidiary is \$250,749. Based on this exchange transaction, the activities relating to the commercial office building have been reported as discontinued operations. Revenues were \$0 and \$1,616,382 and pretax income (loss) was \$0 and \$(37,651) for the three months ended March 31, 2008 and 2007, respectively, and were reported as discontinued operations. The transaction had no impact to net income in the accompanying consolidated statements of income. The net book value of assets distributed over assets received was recorded as a distribution of \$21,270,133 on January 1, 2008.

## **10. ACQUISITIONS**

## Acquisition of Powerco Electric Corp.

On September 30, 2007, the Company purchased 100% of the outstanding stock of Powerco Electric Corp. (PEC), a commercial electrical subcontractor, for \$3,300,000 in cash. PEC performs work in and around the Los Angeles area, specializing in commercial building projects.

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 10. ACQUISITIONS, Continued

The transaction was accounted for using the purchase method of accounting as required by FASB Statement No. 141, *Business Combinations*, including the allocation of the purchase to the tangible and intangible assets of Powerco Electric Corp., which has been finalized. The following table summarizes the fair value of the assets acquired and liabilities assumed as of September 30, 2007:

Current assets	\$ 21,684,107
Property and equipment, net	325,969
Goodwill	2,826,117
Intangibles	4,160,000
-	
Total assets acquired	28,996,193
Current liabilities	(7,745,611)
Other long-term liabilities	(17,950,582)
Total acquisition costs	\$ 3,300,000

The following table identifies the intangible asset acquired and its respective amortization period. The amount assigned to an intangible asset represents the Company s estimate of the fair value of the intangible asset acquired, which is being amortized in relation to contract revenue. Amortization expense was \$631,113 for the three months ended March 31, 2008.

		Amortization
	Amount	Period
Constructon contract backlog	\$ 4,160,000	4 years
Total intangible asset acquired	\$ 4,160,000	

#### Acquisition of Desert Plumbing & Heating Co. Inc.

On January 4, 2008 the Company purchased 100% of the outstanding stock of Desert Plumbing & Heating Co. Inc. (DPH), a mechanical subcontractor located in Las Vegas, NV, for \$35.0 million cash and an earn-out based upon a percentage of DPH s earnings over the next three years, up to a maximum of \$4 million annually. DPH operates primarily in the state of Nevada and had approximately 25 contracts in progress, primarily in the Las Vegas hotel and gaming market, at the time of the acquisition.

In the event the Company becomes a public company by public offering or otherwise, the earn-out payments to the former DPH shareholders will be payable in shares of stock in the parent company in an amount equivalent to the maximum earn-out payment of \$12 million.

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 10. ACQUISITIONS, Continued

The transaction was accounted for using the purchase method of accounting as required by FASB Statement No. 141, *Business Combinations*, including the allocation of the purchase to the tangible and intangible assets of DPH, which has been finalized. The following table summarizes the fair value of the assets acquired and liabilities assumed as of January 4, 2008:

Current assets	\$ 31,260,668
Property and equipment, net	933,964
Goodwill	18,487,000
Intangibles	9,000,000
Total assets acquired	59,681,632
Current liabilities	(24,681,632)
Total acquisition costs	\$ 35,000,000

The \$18,487,000 of Goodwill referred to above has been allocated to the building construction segment and is not deductible for tax purposes. Goodwill and intangible assets with an indefinite life recorded in the acquisition were tested in 2005, 2006 and 2007 and will be tested periodically in the future for impairment as required by FASB Statement No. 142, *Goodwill and Other Intangible Assets*.

The following table identifies the intangible assets acquired and their respective amortization periods. Amortization expense was \$609,250 for the three months ended March 31, 2008.

	Amount	Amortization Period
Trade names	4,100,000	19 years
Customer relationships	3,300,000	9 years
Construction contract backlog	\$ 1,600,000	2 years
Total intangible assets acquired	\$ 9,000,000	

#### All Star Aggregate, Inc.

On March 7, 2008, the Company entered into an agreement for the transfer and purchase of certain mineral material mining contracts, equipment, and material stockpiles used to operate All Star Aggregate, Inc. s existing business in Sloan Pit, located in Clark County, NV, for a purchase price of \$5,200,000. At March 31, 2008, the Company s present value portion of the remaining royalty payables of \$1,853,677 is included in the accompanying balance sheet.

#### Bureau of Land Management Leases

On March 12, 2008, the Company entered into a 10-year renewable contract with the Bureau of Land Management (BLM) for the extraction of 19,500,000 tons of limestone aggregate on land located at Mount Diablo Meridian, NV. Pursuant to the Company s September 12, 2007 high bid of \$0.90 per ton and a \$0.01 per ton administrative fee, the total contract price equates to \$17,745,000 and requires a five percent deposit in the amount of \$877,500 prior to beginning extraction. The balance of the contract price will be paid monthly over the contract term based upon the

## Table of Contents

# Edgar Filing: IDEX CORP /DE/ - Form 10-K

value of materials removed in the prior month. In addition, the Company is obligated under a separate agreement to pay a third party a finders fee related to

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 10. ACQUISITIONS, Continued

the transaction in the amount of \$300,000. At March 31, 2008, the Company s present value portion of the remaining royalty payables of \$9,507,354 is included in the accompanying balance sheet.

#### 11. RETIREMENT PLANS

The Company participates in multi-employer defined benefit pension plans in connection with union collective bargaining agreements. Contributions are based upon hours worked by employees covered under these agreements and are funded on a current basis. The Company contributed and charged to expense \$2,444,650 and \$1,633,804 for the three months ended March 31, 2008 and 2007, respectively.

The Company sponsors a 401(k) profit-sharing plan for non-union domestic employees. Effective July 1, 2006, the Company began contributing \$0.30 per dollar (an increase from \$0.25 per dollar) of employee contributions up to 10% of eligible salary. Participants are fully vested in their contributions and vest at a rate of 20% per year of participation in the plan, becoming fully vested in 5 years. The Company contributed and charged to expense \$170,786 and \$125,764 for the three months ended March 31, 2008 and 2007, respectively.

DPH adopted a 401(k) profit sharing plan in 1995. Participants may elect to contribute up to 12% of their earnings into the plan. DPH s contributions to the Plan are determined by management on an annual basis. For the three months ended March 31, 2008, DPH contributed \$11,717 in matching contributions to its plan.

The Company sponsors a 401(k) profit sharing plan for non-union territory of Guam employees. The Company contributes \$0.25 per dollar of employee contributions up to 6% of eligible salary. Participants are fully vested in their contributions and vest at a rate of 20% per year of participation in the plan, becoming fully vested in 5 years. The Company contributed and charged to expense \$23,156 and \$18,231 for the three months ended March 31, 2008 and 2007, respectively.

## 12. SIGNIFICANT RISKS AND UNCERTAINTIES

Concentrations:

A majority of the Company s cash balances are on deposit with financial institutions and exceed federally insured deposit limits. A substantial portion of the Company s labor force (40%) is subject to collective bargaining agreements, which expire in years 2008 and beyond. Revenues from various projects in California for University of California, Los Angeles totaled \$31,938,638 (or 14% of consolidated revenues) for the three months ended at March 31, 2007. Revenues from a single project in Nevada for Wynn Las Vegas, LLC totaled \$191,569,594 (or 48% of consolidated revenues) for the three months ended at March 31, 2007. Revenues from a single project in Nevada for Westgate Planet Hollywood Las Vegas, LLC totaled \$64,995,050 (or 16% of consolidated revenues) for the three months ended at March 31, 2008. Revenues from a single project for Los Angeles World Airports totaled \$37,464,130 (or 16% of consolidated revenues) for the three months ended at March 31, 2007.

The Company had accounts receivable due from various projects for University of California, Los Angeles, in the amount of \$37,531,270 (or 13% of receivables); a single project for Wynn Las Vegas, LLC totaled \$109,589,479 (or 38% of receivables); and a single project for Westgate Planet Hollywood Las Vegas, LLC totaled \$30,911,806 (or 11% of receivables) at March 31, 2008.

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 13. OTHER COMMITMENTS AND CONTINGENCIES

Contingent liabilities of the Company include the industry standard contractor liabilities for performance and completion of construction contracts. In addition, the Company is a defendant in various lawsuits. Except as discussed below, in the opinion of management, the resolution of these matters will not likely have a material adverse effect on the financial position, result of operations or cash flows of the Company.

The Company, from time to time, enters into certain types of contracts that contingently require the Company to indemnify parties against third-party claims. These contracts primarily relate to: (i) business sales and purchase agreements, under which the Company may provide customary indemnification, and (ii) certain lease agreements, under which the Company may be required to indemnify the lessor for environmental and other liabilities, and other claims arising from the Company s use of the applicable asset.

The terms of such obligations vary. Generally, a maximum obligation is not explicitly stated. Because the obligated amounts of these types of agreements often are not explicitly stated, the overall maximum amount of the obligations cannot be reasonably estimated. Historically, the Company has not been obligated to make significant payments for these obligations, and no liabilities have been recorded for these obligations in its balance sheet as of March 31, 2008.

## Tutor-Saliba/Perini JV v. Los Angeles MTA 1995:

Los Angeles Superior Court Case No. BC 123559

In March 1995, a joint venture, Tutor-Saliba/Perini (TSP), in which Tutor-Saliba Corporation is the 60% partner, filed a complaint in Los Angeles Superior Court against the Los Angeles County Metropolitan Transportation Authority (MTA) for breach of contract, which sought damages of approximately \$16 million to compensate TSP and its subcontractors for the costs of extra work, delays, and interest on late payments under four separate contracts with the MTA. In February 1999, the MTA filed a cross-complaint generally alleging violation of the California False Claims Act (CFCA) and several other federal and state statutory provisions.

Trial commenced before a jury in May 2001. During trial, the Judge ruled that TSP may have failed to produce certain documents to the MTA, and the Judge penalized TSP for its alleged non-compliance by dismissing TSP s claim and by ruling, without jury finding of fact, that TSP was liable to the MTA for damages on the MTA s counterclaim. Pursuant to the Judge s instructions, the Jury awarded the MTA approximately \$32 million in damages and prejudgment interest, subsequently, in March 2002, the judge awarded an additional \$31 million in legal fees and costs. Tutor-Saliba s share of the aggregate award being approximately \$37.8 million.

TSP appealed the verdict and the Court s rulings on motions and sanctions.

On January 25, 2005, the State of California Court of Appeal issued an opinion in which it reversed the entire \$63 million trial court s judgment and found that the trial court judge had abused his discretion and violated TSP s due process rights and imposed an impermissibly overbroad sanction in issuing terminating sanctions that prevented the joint venture from presenting its claims and severely limited TSP in defending itself against the MTA s lawsuit. The Court of Appeal also directed the trial court to dismiss MTA s claims that TSP had violated the Unfair Competition Law and remanded the remainder of the case to the trial court for further proceedings, including a new trial, with a new judge, on the joint venture s claims against the MTA.

## TUTOR-SALIBA CORPORATION AND SUBSIDIARIES

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### As of March 31, 2008 and for the three months ended March 31, 2008 and 2007

## 13. OTHER COMMITMENTS AND CONTINGENCIES, Continued

The case was reassigned to the Complex Civil Division, Los Angeles Superior Court Central Civil West, with the Presiding Judge of that division being assigned to the case. The Court is seeking ways to resolve the case short of full trial, but in a manner that would be upheld by the Court of Appeal, elected to bifurcate the trial and try the MTA s suggested strongest claim first. That claim was a pass-through subcontractor claim that the MTA agreed to pay through an executed change order for approximately \$111,000, in or about 1995 (Tunnel Handrail claim). Trial commenced on the MTA s Tunnel Handrail claim on November 17, 2006, and was completed on December 18, 2006, with the Jury entering a verdict in favor of the MTA pertaining to the handrail issue. A final judgment with respect to this claim will not be entered, until the entire case has been resolved and will be subject to appeal.

In February 2007, the court granted TSP a motion and precluded MTA in future proceedings from presenting its claims that TSP breached its contract and violated the CFCA.

On December 26, 2007 the Court issued an order and opinions regarding TSP s and MTA s affirmative claims, on the night restrictions issues, which determined that no claims have sufficient prima facie evidence to proceed to trial. MTA has filed a notice of its intent to seek a new trial as to the Court s denial of MTA s night restrictions claims in the entirety.

The court has indicated that it would like the parties to resolve the entire case through mediation. Those efforts to date have failed.

Due to the major issues of the case still remaining to be scheduled for retrial, the ultimate financial impact of the lawsuit is not yet determinable. Therefore, no provision for loss, if any, has been recorded in the financial statements.

## 14. SUBSEQUENT EVENTS Purchase of North Valley Commerce Center, Building G

On October 30, 2007, the Company entered into an agreement to purchase the North Valley Commerce Center, Building G in Sylmar California for \$6.5 million in cash. The property will consist of an office and warehouse facility, currently under construction. As of January 25, 2008, the Company has made initial deposits for the property totaling \$300,000 into an escrow account. The purchase is expected to close upon substantial completion of the building.

#### Perini Merger

On April 2, 2008, the Company entered into a definitive agreement with Perini Corporation, a publicly traded construction company, under which the two companies would be combined, in tax-free, all-stock merger. The transaction is subject to customary closing conditions, including the approval of Perini s shareholders and receipt of regulatory approvals.